

2014 Independent Environmental Audit

Tarrawonga Coal Pty Limited

22 December 2014



INDEPENDENT ENVIRONMENTAL AUDIT SUBMISSION FORM

Project

Consent Number	011-0047	
Description of Project	Tarrawonga Coal Project	
Project Address	Goonbri Road, Boggabri, NSW 2382	
Proponent	Tarrawonga Coal Pty Limited	
Proponent Address	PO Box 638, Newcastle, NSW 2300	

Independent Audit

Certificate	I certify that I have prepared the contents of the attached independent audit and to the best of my knowledge:
	 It is in accordance with relevant approval conditions; I have acted professionally, accurately and in an unbiased manner in conducting the audit; I am not related to any owner or operator of the project as a spouse, partner, child, sibling, employer, or in a contractual arrangement outside the audit; I do not have any pecuniary interest in the project, including where there is a reasonable likelihood or expectation of appreciable financial gain or loss to me or to a person to whom I am related; Neither I nor my employer have provided consultancy services to the project that were subject to this audit; I have not accepted, nor intend to accept any inducement, commission, gift or any other benefit (apart from fair payment) from any owner or operator of the project, their employees or any interested party. I have not knowingly allowed, nor intend to allow my colleagues to do so.
Signature	
Name	Peter Horn
Address	74 Hunter Street, Newcastle, NSW 2300
E-mail Address	Peter.horn@smec.com
Auditor Certification	Environmental Compliance and Environmental Management Systems Lead Auditor Certified through RAB-QSA
Date	22-12-2014

DOCUMENT CONTROL

File Location Name:	300123322/Final Report	
Project Name:	Tarrawonga IEA	
Project Number:	30012322	
Revision Number:	Revision A	

Revision History

Revision #	Date	Prepared by	Reviewed by	Approved for Issue by
Rev A	18-12-14	Peter Horn	Joy Duncan	Peter Horn
Rev B	22-12-14	Peter Horn	Joy Duncan	Peter Horn

SMEC Australia Pty Limited

74 Hunter Street

Newcastle NSW 2300

Tel:	+61 2 4925 9600	Fax:	+61 2 4925 3888
Email:	peter.horn@smec.com	Website:	<u>www.smec.com</u>

This report is confidential and is provided solely for the purposes of Tarrawonga Coal Pty Ltd. This report is provided pursuant to a Consultancy Agreement between SMEC Australia Pty Limited ("SMEC") and Tarrawonga Coal Pty Ltd under which SMEC undertook to perform a specific and limited task for Tarrawonga Coal Pty Ltd. This report is strictly limited to the matters stated in it and subject to the various assumptions, qualifications and limitations in it and does not apply by implication to other matters. SMEC makes no representation that the scope, assumptions, qualifications and exclusions set out in this report will be suitable or sufficient for other purposes nor that the content of the report covers all matters which you may regard as material for your purposes.

This report must be read as a whole. The executive summary is not a substitute for this.

Unless expressly agreed otherwise in writing, SMEC does not accept a duty of care or any other legal responsibility whatsoever in relation to this report, or any related enquiries, advice or other work, nor does SMEC make any representation in connection with this report, to any person other than Tarrawonga Coal Pty Ltd. Any other person who receives a draft or a copy of this report (or any part of it) or discusses it (or any part of it) or any related matter with SMEC, does so on the basis that he or she acknowledges and accepts that he or she may not rely on this report nor on any related information or advice given by SMEC for any purpose whatsoever.

TABLE OF CONTENTS

1. INTRODUCTION		ON	9	
	1.1.	Back	ground	9
	1.2.	Site	Description	9
	1.3.	Aud	it Work	9
	1.4.	Aud	it Approach	11
	1.5.	Repo	ort Structure	12
2.	CONSUL	TATIO	ON	13
3.	DOCUM	IENTS	AUDITED	15
4.			NTAL COMPLIANCE	
	4.1.	Issue	es Causing Non-Compliance	17
	4.2.	Proj	ect Approval PA 11_0047	21
	4.1.	State	ement of Commitments from the EA	31
	4.2.	Envi	ronmental Protection Licence	31
	4.3.	Min	ing Operations Plan	32
	4.4.	Min	ing Leases	34
	4.5.	EPB	C Approval	34
	4.6.	Envi	ronmental Management Strategy	37
	4.7.	Envi	ronmental Monitoring Program	38
	4.8.	Nois	e Management Plan	38
	4.9.	Nois	e Management Strategy	39
	4.10.	Blas	t Management Plan	39
	4.11.	Blas	t Monitoring Program	40
	4.12.	Blas	t Management Strategy	40
	4.13.	Air C	Quality and Greenhouse Gas Management Plan	40
	4.14.	Air C	Quality Management Strategy	41
	4.15.	Wat	er Management Plan	41
	4.16.	Grou	undwater Contingency Plan	44
	4.17.	Wat	er Management Strategy	45
	4.18.	Biod	liversity Management Plan	45
	4.19.	Wille	eroi Biodiversity Offset Area Management Plan	48
	4.20.		abilitation Management Plan	
	4.21.	Heri	tage Management Plan	50
	4.22.	Abo	riginal Heritage Conservation Strategy	51
	4.23.	Bush	nfire Management Plan	51
	4.24.	Was	te Management Plan	51
	4.25.	Road	d Closure Management Plan	52
	4.26.	Prev	ious Independent Environmental Audit	52
5.			EFFECTIVENESS OF ENVIRONMENTAL MANAGEMENT AND MANAGEMENT	53
20				
	5.1.		ogy	
	5.2.		abilitation	
	5.3.		ustics	
	5.4.	Air C	Quality	57
	5.4	.1.	Air Quality Management Plan	57
	5.4	.2.	Air Quality Management Strategy – Cumulative impacts from all mines	58

5.4	4.3. Assessment of Project implementation of best air quality management practice	e;58
5.4	4.4. Measures taken to minimise air quality impacts of the project during	
ex	traordinary events when the relevant air quality limits in this approval do not apply	59
5.4	4.5. Project compliance with EPL air quality requirements	59
5.4	4.6. Overall impressions – Observations	60
5.5.	Surface Water Quality	60
5.6.	Groundwater Resources	61
6. RECON	IMENDATIONS	62
6.1.	Biodiversity Offsets	62
6.1.	Bushfire Management	62
6.2.	Heritage	62
6.1.	Public Safety	62
6.2.	Rehabilitation	63
6.3.	Water Management	63
6.1.	Management Plans	64
6.1.	Reporting	64
APPENDIX A	A AUDIT TEAM APPROVAL	65
APPENDIX E	CONSULTATION WITH REGULATORY STAKEHOLDERS	66
APPENDIX (AUDIT PROTOCOL	67

LIST OF TABLES

Table 1.1 - List of Requirements for this IEA Report	9
Table 3.1 List of Documents Audited	15
Table 3.2 Tarrawonga Coal Approvals	
Table 4.1 TCPL Issues Causing Non-compliance	
Table 4.2 Non-Compliances for PA 11_0047	21
Table 4.3 Environmental Protection Licence	
Table 4.4 Mining Operations Plan	33
Table 4.5 EPBC Approval	34
Table 4.6 Environmental Management Strategy	37
Table 4.7 Noise Management Plan	38
Table 4.8 Blast Management Plan	39
Table 4.9 Air Quality and Greenhouse Gas Management Plan	40
Table 4.10 Water Management Plan	41
Table 4.11 Groundwater Contingency Plan	44
Table 4.12 Water Management Strategy	
Table 4.13 Biodiversity Management Plan	45
Table 4.14 Willeroi Biodiversity Offset Area Management Plan	48
Table 4.15 Rehabilitation Management Plan	48
Table 4.16 Heritage Management Plan	50
Table 4.17 Bushfire Management Plan	51
Table 4.18 Waste Management Plan	51
Table 5.1 Ecological attributes measured in condition plots (modified from DECC 2009* & Siv	ertsen
2009*)	54
Table 5.2: Sampling units	55

EXECUTIVE SUMMARY

SMEC Australia Pty Ltd (SMEC) was commissioned by Tarrawonga Coal Pty Ltd (TCPL) to conduct an Independent Environmental Audit (IEA) in accordance with the Project Approval 11_0047 (Schedule 5 Condition 10) for the Tarrawonga mine.

Tarrawonga mine is located approximately 13 km east north east of Boggabri New South Wales (NSW). TCPL is an open cut operation, with a history which dates back to 2005.

Project Approval (11_0047) was granted on 22 January 2013 to extend open cut operations. The project approval allows for the extraction of up to 3 million tonnes per annum of run of mine (ROM) coal until the end of December 2030. This IEA has been prepared to satisfy Condition 50 (Biodiversity Independent Audit) of Schedule 3 and Conditions 10 and 11 (Independent Environmental Audit) of Schedule 5 from the Project Approval 11_0047. The IEA was undertaken generally in accordance with AS/NZS ISO 19011:2003 – Guidelines for quality and/or environmental management systems auditing.

In the assessment of compliance, the status of each condition is described as:

	□ Compliant;
	□ Non-Compliant;
	□ Non-Compliant (Administrative);
	□ Observation; or
•	□ Not Applicable (used where conditions have not yet been activated (due to activities not being commenced or requests not being made for example).

A total of 1910 conditions and commitments were assessed as part of this audit. 48 issues resulted in 82 non-compliances. 31 of the issues were administrative (that is, the issue was caused by not submitting a document or keeping a document on file, not by the omission of an action or measurement). Many of the non-compliances noted in this audit relate to the same issue which, due to the duplication of commitments between consent documents and management plans, raise the same non-compliance several times.

A basic risk assessment was conducted for all non-compliances with Low/Medium/High/Extreme risk levels as results. The risk assessment details are presented in Appendix C. For the non-compliances that were not administrative (there were 46 administrative non-compliances), there were 4 Low, 34 Medium and 2 High results. Both high results relate to the same issue and a recommendation has been made that is aimed at reducing the risk. The high risk issue was related to water, water collected from the coal loading area is located in Dam SB4, which then flows through a series of dams to Licensed Discharge Point 1 (LDP1 - EPL 12365) and potentially is discharged off-site. The site is currently developing a solution to the issue.

It should be noted that the site is currently in a state of flux with many management plans in draft and not approved. None of the management strategies were approved at the time of the audit as the Maules Creek approval was only recently granted. As such, some of the measures detailed in the plans have not been implemented due to uncertainty surrounding the content of the management plan once approved. Where there is no direct environmental

impact associated with not implementing these measures, they have not been identified as "not compliant".

The audit has been able to give some direction to the site regarding environmental improvements and has identified a number of areas where documentation and reporting could be improved.

ABBREVIATIONS AND ACRONYMS

Abbreviation/ Acronym	Description
AEMR	Annual Environment Management Report
AMD	Acid Mine Drainage
AR	Annual Return (to the EPA)
AS/NZS	Australian Standard / New Zealand Standard
°C	Degrees Celsius
CCC	Community Consultative Committee
CMA	Catchment Management Authority
DP&E	NSW Department of Planning and Environment
DRE	Division of Resources and Energy
DTI	Department of Trade and Industry
EA	Environmental Assessment
EPA	Environment Protection Authority
EP&A Act	Environment Planning and Assessment Act 1979
EPL	Environment Protection License
EPBC	Environment Protection and Biodiversity Conservation Act 1999
GW	Groundwater Well
IEA	Independent Environmental Audit
LDP	Licensed Discharge Point
LLS	Local Land Services
m	Metres
ML	Mining Lease
MOP	Mining Operations Plan
NAG	Net Acid Generating
NATA	National Association of Testing Authorities, Australia
NOW	NSW Office of Water
OCE	Open Cut Examiner
OEH	NSW Office of Environment and Heritage
PA	Project Approval
PAG	Potentially Acid Generating
POEO	Protection of the Environment Operations Act 1997
SMEC	Snowy Mountains Engineering Corporation
TARP	Trigger, Action, Response Plan
TCPL	Tarrawonga Coal Pty Ltd

1. INTRODUCTION

1.1. Background

SMEC Australia Pty Ltd (SMEC) was commissioned by Tarrawonga Coal Pty Ltd (TCPL) to conduct an Independent Environmental Audit (IEA) in accordance with the Project Approval 11_0047 (Schedule 3, Condition 50 and Schedule 5 Condition 10) for the Tarrawonga mine.

The audit was designed and conducted to satisfy the planning approval conditions for TCPL and focused on the site's compliance with licences, approvals and supporting documents including management plans. This audit period is 22 January 2013 (date of approval of the project by NSW Department of Planning and Environment) to 10 September 2014.

1.2. Site Description

Tarrawonga mine is owned and operated by a joint venture between Whitehaven Coal Limited (70%) and Idemitsu Australian Resources Pty Ltd (30%). Project Approval was granted on 22 January 2013 to operate the Tarrawonga Coal Project under Project Approval 11_0047. Statements of compliance reported in this audit report are in relation to the conditions and commitments of Project Approval 11_0047.

Tarrawonga Mine is located approximately 13 km east north east of Boggabri New South Wales (NSW). TCPL is an open cut operation, with a history which dates back to 2005. The mine is operated on a two shift basis with mining conducted 7.00am till 3.30am the next morning Monday to Friday and 7.00am till 6.00pm on Saturdays. Maintenance can be conducted 24 hours per day 7 days a week.

Project Approval (11_0047) was granted on 22 January 2013 to extend open cut operations. The project approval allows for the extraction of up to 3 million tonnes per annum of run of mine (ROM) coal until the end of December 2030.

1.3. Audit Work

This IEA has been prepared to satisfy Condition 50 of Schedule 3 and Conditions 10 and 11, Schedule 5 of Project Approval 11_0047. Table 1 lists the requirements of this condition and shows where each is located in this IEA report.

Table 1.1 - .	List of F	equirements	s for this I	EA Report
----------------------	-----------	-------------	--------------	-----------

Condition	Condition Requirement	
Schedule 3		
50	By the end of December 2011, and every 3 years thereafter, unless the Director-General directs otherwise, the Proponent shall commission and pay the full cost of an Independent Environmental Audit of the project. This audit must:	This Audit
50 a)	Include consultation with OEH, Namoi CMA, DPI Catchments and Lands, SEWPaC, CCC and DRE	Section 2 and Appendix B
50 b)	Assess the performance of the revegetation in the rehabilitation area completed to date (and the Goonbri Creek Diversion, once commenced) against the	Sections 4.18, 4.19, 4.20, 5.1 and 5.2

Condition	Requirement	Location in report
	completion criteria in the Rehabilitation Management Plan	
50 c)	Assess the performance of management and restoration in the offsite Biodiversity Offset Strategy areas completed to date against the completion criteria in the Rehabilitation Management Plan	Sections 4.18, 4.19, 4.20, 5.1 and 5.2
50 d)	Identify any measures that should be implemented to improve the performance of rehabilitation management and restoration within the biodiversity offset areas	Sections 6.1 and 6.2
50 d)	Identify any additional measures that should be applied in the establishment of native vegetation around the realigned Goonbri Creek, both before and after the realignment is undertaken	Section 5.2
50 e)	If the completion criteria have not been met, or are not adequately trending towards being met, determine the likely ecological value of the rehabilitation and restoration once completed, and recommend additional measures to augment the Biodiversity Offset Strategy to ensure that it adequately offsets the projects impacts on biodiversity.	Sections 5.1 and 5.2
50	If the audit recommends the implementation of additional measures to augment the Biodiversity Offset Strategy in accordance with e) above then within 6 months of the completion of the audit the Proponent shall revise the Biodiversity Offset Strategy, in consultation with the Department, OEH and SEWPaC and to the satisfaction of the Director General	Noted
Schedule 5		
9	By the end of December 2011, and every 3 years thereafter, unless the Director-General directs otherwise, the Proponent shall commission and pay the full cost of an Independent Environmental Audit of the project. This audit must:	This Audit
9(a)	Be conducted by a suitably qualified, experienced and independent team of experts whose appointment has been endorsed by the Director-General;	Section 1.4
9(b)	Include consultation with the relevant agencies;	Section 2
9(c)	Assess the environmental performance of the project and assess whether it is complying with the requirements in this approval and any relevant EPL or Mining Lease (including any assessment, plan or program required under these approvals);	Section 4
9(d)	Review the adequacy of strategies, plans or programs required under the abovementioned approvals; and	Section 5
9(e)	Recommend appropriate measures or actions to improve the environmental performance of the project,	Section 6

Condition	Requirement	Location in report		
	and/or any assessment, plan or program required under the abovementioned approvals.			
Notes:	This audit team must be led by a suitably qualified auditor and include experts in surface water, groundwater and any other fields specified by the Director-General.	Appendix A		
10	Within 6 weeks of the completion of this audit, or as otherwise agreed by the Director-General, the Proponent shall submit a copy of the audit report to the Director-General, together with its response to any recommendations contained in the audit report.	Noted		

1.4. Audit Approach

This IEA was undertaken generally in accordance with AS/NZS ISO 19011:2003 – Guidelines for quality and/or environmental management systems auditing by the following personnel:

- Peter Horn (Environmental Manager) Lead Auditor and Rehabilitation Specialist from SMEC;
- Catherine Gallery (Senior Ecologist) Biodiversity and Offsets Specialist from SMEC;
- Glen Homes (Principal Acoustics) Acoustics from SMEC;
- Aleks Todoroski (Principal) Air Quality Specialist from Todoroski Air Sciences;
- Glenn Mounser (Water Manager) Surface Water Specialist from SMEC;
- Dr Yohannes Woldeyohannes (Senior Engineer) Groundwater Modelling Specialist from SMEC;
- Matthew Harland (Environmental Engineer) Assistant Auditor from SMEC; and
- Joy Duncan (Technical Principal Environment) Peer Review from SMEC.

The audit team were approved by the Department of Planning and Environment (DP&E) (on 23 July 2014) prior to conducting the audit (appended as Appendix A).

This IEA consisted of a detailed desktop review of documents supporting compliance, interviews with TCPL staff and a site inspection of Tarrawonga Mine from 8 - 10 September 2014. Interviewees included:

- Group Environment Manager;
- Environment Officer;
- Drill and Blast Engineer;
- Workshop Supervisor;

- Property Manager/Field Environmental Officer;
- Geologist;
- Safety and Training Coordinator;
- Manager Mining;
- Mine Planner; and
- An OCE.

Site opening and closing meetings were held with the site Environment team and Operations Manager in attendance with the audit team. The opening meeting discussed the approach and process of the audit while the closing meeting covered the findings to that point and the audit teams general impressions of the sites management.

The environmental conditions at the time of the audit were mild with daytime maximum temperatures in the mid 20's (degrees Celsius) and minimums between 0 and 10°C. There were mainly clear conditions during the site audit.

1.5. Report Structure

This report is structured as follows:

Executive Summary

Section 1.0 provides an introduction, background and description of Tarrawonga Mine, describes the requirements for the IEA and provides a guide to the structure of the report.

Section 2.0 discusses consultation with the relevant departments.

Section 3.0 lists the planning approvals in place at Tarrawonga Mine, provides a description of each and confirms those which have been the subject of this IEA.

Section 4.0 provides a discussion of non-compliances against the project approval, licences, permits and supporting documents.

Section 5.0 provides a review of the adequacy of the environmental management at the site both documented and observed

Section 6.0 provides recommendations for measures or actions to improve the environmental performance of TCPL.

2. CONSULTATION

The TCPL environment department notified the DP&E of the proposed scope of the areas requiring expert assessment for the audit. The DP&E confirmed the key scope areas requiring expert assessment to be rehabilitation, biodiversity, surface and groundwater, noise and air quality.

The audit team consulted the NSW Environment Protection Authority (EPA), Department of Trade and Investment, Regional Infrastructure and Services – Division of Mineral Resources (DRE), Tarrawonga Community Consultation Committee CCC), Department of Environment (Commonwealth) (DoE), NSW Office of Environment and Heritage requesting input into the audit scope and focus (appended as Appendix B). Local Land Services were contacted in place of the requirement to consult with the CMA but the suggested contact could not be reached for input in spite of several phone calls to establish contact.

Comments were as follows:

Department of Environment did not provide a response regarding the scope;

The EPA asked for a focus on:

- Surface water discharges;
- Particulate generation;
- Noise;
- The lack of rehabilitation on the southern emplacement; and
- Potential PRP for exposed areas.

DRE provided detailed feedback:

Desktop:

- Is there a current Mining Operations Plan (MOP) in place and has it been approved by DRE?
- Has the MOP been prepared in consultation with the relevant agencies as outlined in the Project Approval?
- Is the rehabilitation strategy as outlined in the MOP consistent with the Project Approval in terms of progressive rehabilitation schedule; and proposed final land use(s)?
- Has the rehabilitation objectives and completion criteria as outlined in the MOP been developed in accordance with the proposed final land(s) as outlined in the Project Approval?
- Has a rehabilitation monitoring program been developed and implemented to assess performance against the nominated objectives and completion criteria? – verified by reviewing monitoring reports and rehabilitation inspection records.

 Has a rehabilitation care and maintenance program been developed and implemented based on the outcomes of monitoring program? - verified by reviewing Annual Rehabilitation Programs or similar documentation.

Site Inspection:

- Are mining operations being conducted in accordance with the approved MOP (production, mining sequence etc.), including within the designated MOP approval boundary? – to be verified by site plans and site inspection.
- Is rehabilitation progress consistent with the approved MOP as verified by site plans and a site inspection? This should include an evaluation against rehabilitation targets and whether the final landform is being developed in accordance with conceptual final landform in Project Approval.
- Based on a visual inspection, are there any rehabilitation areas that appear to have failed or that have incurred an issue that may result in a delay in achieving the successful rehabilitation?

In addition to the above, the audit should note observations where rehabilitation procedures, practices and outcomes represent best industry practice.

OEH did not have any specific requests outside the scope already noted.

The Tarrawonga Community Consultative Committee (CCC) were contacted but chose not to offer any comment.

No other input was provided by the regulators prior to this audit report being finalised.

3. DOCUMENTS AUDITED

Table 3.1 lists the documents reviewed for compliance in this IEA along with where each document is addressed in the report. There were many other documents reviewed by the audit team as evidence or supporting information that are not listed here. Table 3.2 lists the sites approval documents.

Table 3.1 List of Documents Audited

Document	Location in Report
Project Approval 11_0047	4.2
Tarrawonga Coal Project Environmental Assessment - Statement of Commitments	4.1
EPL 12365	4.1
Mining Leases	4.4
Mining Operations Plan, November 2013 to October 2015	4.3
EPBC Approval, 2011	4.5
Environmental Management Strategy, May 2013	4.6
Environmental Monitoring Program, August 2009	4.7
Noise Management Plan, July 2014	4.8
Noise Management Strategy, Boggabri, Tarrawonga, Maules Creek Complex, March 2014	4.9
Blast Management Plan, September 2014	4.10
Blast Management Strategy, January 2013	4.12
Blast Monitoring Program, November 2008	4.11
Road Closure Management Plan, May 2013	4.25
Air Quality and Greenhouse Gas Management Plan, June 2014	4.13
Air Quality Management Strategy, March 2014	4.14
Water Management Plan, November 2013	4.15
Groundwater Contingency Plan, October 2006	4.16
Water Management Strategy, March 2013	4.17
Tarrawonga Biodiversity Management Plan, May 2013	4.18
Willeroi Biodiversity Offset Area Management Plan, February 2014	4.19

Document	Location in Report
Rehabilitation Management Plan, May 2013	4.20
Heritage Management Plan, January 2014	4.21
Aboriginal Cultural Heritage Strategy, August 2014	4.22
Bushfire Management Plan, April 2013	4.23
Waste Management Plan, May 2013	4.24

Table 3.2 Tarrawonga Coal Approvals

Approval	Regulator	Expiry Dates	
Project Approval 11_0047	DP&E	31 December 2030	
EPL 12365	EPA	Annual Renewal Date 09- 01-15	
Mining Lease ML1579	DTIRIS DRE Minerals		
Mining Lease ML 1685	DTIRIS DRE Minerals	Various	
Mining Lease ML 1693	DTIRIS DRE Minerals		

4. ENVIRONMENTAL COMPLIANCE

In the assessment of compliance, the status of each condition is described as:

- Compliant;
- Not Compliant;
- Not Compliant Administrative (the issue was caused by not submitting a document or keeping a document on file, not by the omission of an action or measurement);
- Observation; or
- Not Applicable (used where conditions have not yet been activated (due to activities not being commenced or requests not being made for example).

A total of 1910 conditions and commitments were assessed as part of this audit. 48 issues resulted in 82 non-compliances. 31 of the issues were administrative. Many of the non-compliances noted in this audit relate to the same issue which, due to the duplication of commitments between consent documents and management plans, raise the same non-compliance several times. These numbers do not include the assessment of compliance with recommendations from the previous IEA, these issues are addressed separately in Section 4.26.

A basic risk assessment was conducted for all non-compliances with Low/Medium/High/Extreme risk levels as results. For the non-compliances that were not administrative (there were 46 administrative non-compliances), there were 4 Low, 34 Medium and 2 High results. Both high results relate to the same issue and a recommendation has been made that is aimed at reducing the risk. The high risk issue was related to water, water collected from the coal loading area is located in Dam SB4, which then flows through a series of dams to Licensed Discharge Point 1 (LDP1 - EPL 12365) and potentially is discharged off-site. The site is currently developing a solution to the issue.

It should be noted that the site is currently in a state of flux with many management plans in draft and not approved. None of the management strategies were approved at the time of the audit as the Maules Creek approval was only recently granted. As such, some of the measures detailed in the plans have not been implemented due to uncertainty surrounding the content of the management plan once approved. Where there is no direct environmental impact associated with not implementing these measures, they have not been identified as "not compliant".

The audit has been able to give some direction to the site regarding environmental improvements and has identified a number of areas where documentation and reporting could be improved.

4.1. Issues Causing Non-Compliance

Each non-compliance was caused by an action, omission or event. These combined constitute the issues that the site needs to address to achieve compliance. For this reason, the issues are extracted from the non-compliances so they will be more readily addressed by TCPL.

The issues identified in this audit and the associated non-compliances are presented in Table 4.1.

Table 4.1 TCPL Issues Causing Non-compliance

Issue	Non-compliances
DA 88-4-2005 Not surrendered by December 2013	Administrative - PA 11-0047, Schedule 2 Condition 14
Enter into a Planning Agreement with Council by the end of March 2013	PA 11-0047, Schedule 2 Condition 21
The Groundwater Management Plan does not refer to stygo-fauna or groundwater dependent ecosystems	PA 11-0047, Schedule 3 Condition 39
Make arrangements for the long term security of the Willeroi offset area before the end of December 2013	Administrative - PA 11-0047, Schedule 3 Condition 46
By the end of May 2013, the Proponent shall lodge a Conservation and Biodiversity Bond with the Department to ensure that the biodiversity offset strategy is implemented. The details of the security bond are to be provided to the Minister DoE.	Administrative - PA 11-0047, Schedule 3 Condition 49 EPBC Approval, Condition 27
Soil testing and surveys not conducted prior to soil stripping	Administrative - PA 11-0047, Schedule 3 Condition 62 EPBC Approval, Condition 25 Tarrawonga Mine Biodiversity Management Plan, Section 4.3.2 Rehabilitation Management Plan, Section 6.4.2, 6.5.2,
The Rehabilitation Management Plan does not describe Interim rehabilitation techniques and triggers	Administrative - PA 11-0047, Schedule 3 Condition 64
This audit was commissioned after the due date of 30 th June 2014	Administrative - PA 11-0047, Schedule 3 Condition 50 and Schedule 4 Conditions 10 and 11 Willeroi Biodiversity Offset Area Management Plan Section 9.5
The Proponent shall, within 3 months of the date of this approval: a) make the following information for the project publicly available on its website, on a daily basis and in a clearly understandable form: daily weather forecasts for the coming week; proposed operational responses to these weather forecasts; real-time noise and air quality monitoring data (subject to any necessary caveats); This information is not placed on the website on a daily basis, generally up to date but not every day.	Administrative - PA 11-0047, Schedule 4 Condition 13
Exceed activity scale in the EPL for coal produced	Administrative - EPL 121365 - Preamble

	Administrative – PA 11-0047
Air quality results not reported online daily	Schedule 5, Conditions 12 and 13, MOP Section 3.2.1
Not all the requirements of the PA 11-0047 for the development of the AQGGMP have been addressed.	Administrative – PA 11-0047 Schedule 3, Condition 29
Overburden not tested for acid generating potential	MOP Section 3.2.5
Not all course rejects are not placed on the pit floor as required by the MOP but are placed at higher elevations, as part of the backfilling process.	Administrative - MOP Section3.2.12
Potential acid generating materials volumes managed not reported in the AEMR	MOP Section 3.3.3
Tubestock in the rehabilitated areas does not have tree guards nor is it mulched.	Administrative - MOP Section 5.4.5 Tarrawonga Mine Biodiversity Management Plan, Section 4.3.3, 4.7.2, 4.7.3, 5.2.1, 5.2.4, 5.3.1, 5.4.1 Rehabilitation Management Plan, Section 6.5.5
Performance of rehabilitation against key performance measures/indicators, regulatory requirements and TCPL commitments and trends in rehabilitation performance not reported in the AEMR	Administrative – MOP – Section 8.3
The Water Management Plan does not reference EPBC requirements or the National Water Quality Management Strategy.	Administrative – EPBC Approval, Condition 16
The Water Management Plan does not state the maximum amounts of drawdown in alluvial and hardrock aquifers risking not identifying a groundwater exceedence	EPBC Approval , Condition 17
No evidence of provision of the updated groundwater and surface water modelling and Management plans to DoE	Administrative – EPBC Approval, Condition 19
No evidence of the submission of the scoping report for the Willeroi Biodiversity Offset Area to DoE	Administrative – EPBC Approval, Condition 22
There is no evidence that Ecological survey data is collected in accordance with DoE Standards	Administrative – EPBC Approval, Condition 30
The AEMR reports on compliance with PA 11-0047, EPL 12365 and the Mining Leases but does not address compliance with the EPBC approval or water licences.	Administrative – EMS Section 4.8
Attended noise monitoring reports do not reference all	Administrative - Noise
the required standards. Attended noise monitoring reports do not record plant configuration when testing Sound Power levels of equipment	Management Plan, Section 4.4 Administrative - Noise Management Plan, Section 4.4
Attended Noise Monitoring reports do not include the names of the personnel conducting the monitoring	Administrative - Noise Management Plan, Section 6.1
Some blast monitoring events have not been recorded.	Blast Management Plan, Section 5.1.1

	DA 44 0047 C-11-1-2
Predictive air dispersion modelling is not used to inform operations planning	PA 11-0047 Schedule 3, Condition 28, Air Quality and Greenhouse Gas Management Plan, Section 3
No evidence of participation in an ACARP project to trial best practise measures for disturbing and handling overburden	Administrative – Air Quality and Greenhouse Gas Management Plan, Section 3
Deposited dust analysis to be conducted by a Nata certified laboratory, the analysis was not certified in the report provided as evidence	Administrative – Air Quality and Greenhouse Gas Management Plan, Section 4.3.2
The area around the coal loader drains to the dirty water system but this portion of the system has an off-site discharge point.	Water Management Plan, Section 2.3.2 Water Management Strategy, Section 4.1.1
The groundwater model has not been updated in the last two years.	Water Management Plan, Sections 2.3.2, 7.8
A series of in-stream monitoring locations and gauges have not been installed since the approval but were to be installed as soon as possible following approval.	Water Management Plan, Section 6.3
No evidence of water samplers being suitably qualified professionals	Administrative - Water Management Plan, Section 6.4
Groundwater wells are not purged prior to sampling	Water Management Plan, Section 7.5
Groundwater monitoring data not posted onto website within two weeks of sampling and not provided to NoW	Administrative - Water Management Plan, Section 7.5.1
The surface and groundwater response plan (TARP) does not detail how to manage complaints	Administrative - Water Management Plan, Section 8
Groundwater wells GW020432 and GW006013 were not recorded in the 2012-13 AEMR	Groundwater Contingency Plan, Section 2.3
In many places in the site documentation, information is to be provided to Narrabri Council and this does not occur	Administrative – Groundwater Contingency Plan, Section 2.3
Seed collection is currently done at the Whitehaven regional biodiversity bank area near Rocglen, no local seed is collected at Tarrawonga	Tarrawonga Mine Biodiversity Management Plan, Sections 4.1.2 and 4.3.3
The effects of salvaging and reusing habitat resources is not monitored annually	Tarrawonga Mine Biodiversity Management Plan, Section 4.3.3
Overburden is not tested for sodicity	Tarrawonga Mine Biodiversity Management Plan, Section 4.5.3
Vehicles are not inspected for weed seed propagules prior to entry to the offsets areas.	Tarrawonga Mine Biodiversity Management Plan, Section 4.7.2
Bushfire fuel load annual assessment is not documented	Administrative - Tarrawonga Mine Biodiversity Management Plan, Sections 4.10.2, 4.10.3, 5.2.2, 5.3.2 and 5.4.2.

	Bushfire Management Plan,
	Section 5.2.8, 5.3.4,
	Administrative - Tarrawonga
Internal audits will be conducted annually, none in the	Mine Biodiversity
audit period	Management Plan, Sections 10
	and 10.1
Tancail Stackwilds not signmented	Rehabilitation Management
Topsoil Stockpiles not signposted	Plan, Section 6.4.2
Mathadalagy for processation of constraint not strictly	Administrative - Heritage
Methodology for preservation of scar trees not strictly observed	Management Plan, Section
observed	4.1.1
The AEMR does not comment on waste minimisation	Administrative - Waste
effectiveness	Management Plan, Section 5

4.2. Project Approval PA 11_0047

Table 4.2 shows the conditions that were not compliant with the Project Approval PA 11_0047. An assessment of compliance for each condition in the PA 11_0047 is provided in the audit protocol in Appendix C.

Table 4.2 Non-Compliances for PA 11_0047

Schedule	Condition	Requirement	Finding
	14	By the end of December 2013, or as otherwise agreed by the Director-General, the Proponent shall surrender the existing development consent (DA-88-4-2005) for the Tarrawonga Coal Mine in accordance with	Not completed, work is underway to surrender the DA. Not Compliant
2	14	Section 104A of the EP&A Act. Prior to the surrender of this development consent, the conditions of this approval shall prevail to the extent of any inconsistency with the conditions of the development consent.	Administrative
2	21	21 By the end of March 2013, unless otherwise agreed by the Director-General, the Proponent shall enter into a planning agreement with Council in accordance with: a) Division 6 of Part 4 of the EP&A Act; and b) the terms of the Proponent's offer provided in Appendix 3.	No evidence provided to support the establishment of a planning agreement Not Compliant
3	25	The Proponent shall ensure that particulate matter emissions generated by the project do not exceed the criteria listed in Table 9 at any residence on privately-owned land or on more than 25 percent of any privately-owned land, except on property 49 in year 16.	HVAS run on non-EPA sanctioned day. HVAS exceedences not investigated. No publicly available information to assess compliance with PM ₁₀ and TSP criteria.
			Not Compliant Administrative

Schedule	Condition	Requirement	Finding
3	28	The Proponent shall: (a); (b) operate a comprehensive air quality management system on site that uses a combination of predictive meteorological forecasting, predictive and real time are dispersion modelling and real-time air quality monitoring data to guide the day-to-day planning of mining operations and implementation of both proactive and reactive air quality mitigation measures to ensure compliance with the relevant conditions of this approval;	B) Mine operates predictive meteorological forecasting, but does not appear to operate any predictive or realtime air dispersion modelling to guide dayto-day planning of the mining operations. Not Compliant Administrative
3	29	The proponent shall prepare and implement an Air Quality and Greenhouse Gas Management Plan for the project to the satisfaction of the Director-General. This plan must: (b) describe the measures that would be implemented to ensure: best practice management is being employed; the air quality impacts of the project are minimised during adverse meteorological conditions and extraordinary events; and compliance with the relevant conditions of this approval; (f) include and air quality monitoring program that: uses a combination of real-time monitors and supplementary monitors to evaluate the performance of the project; adequately supports the proactive and reactive air quality management system; includes PM2.5 monitoring; includes monitoring of occupied mine-owned residences and residences on the air quality affected land in Table 1, subject to the agreement of the tenant and/or landowner; evaluates and reports on the effectiveness of the air quality management system; includes sufficient random audits of operating responses to real time air quality management systems to determine the ongoing effectiveness of these responses in maintaining the project within the relevant criteria in this Schedule and the requirements of conditions 24 and 25, above; includes a protocol for determining any exceedences of the relevant conditions in this	b) AQGGMP incorporates such measures, but is not specific about extraordinary events, and does not contain a clear procedure to identify compliance with the 24-hour average PM ₁₀ criterion. F) AQMP includes most of these aspects, but does not appear to cover; - supplementary monitors; - protocol for determining exceedances of 24-hour PM10 and TSP criterion. AQMP includes an AQMS for the BTM complex, but the AQMP/S does not appear to cover; - a suitable control monitoring site at present (given that the "Will-gai" site is close to other mining activity, it is not a suitable control monitoring site representative of regional air quality.

Schedule	Condition	Requirement	Finding
		approval; and (g) includes a Leard Forest Mining Precinct Air Quality Management Strategy that has been prepared in consultation with the other coal mine in the Precinct to minimise the cumulative air quality impacts of all mines within the Precinct, that includes: systems and processes to ensure that all mines are managed to achieve their air quality criteria; a shared environmental monitoring network and data sharing protocol; control monitoring site (s) to provide real time data on background air quality levels (ie not influenced by mining in the Leard Forest Mining Precinct and representative of regional air quality); a shared predictive and real time air dispersion model covering the Lear Forest Mining Precinct to be used for assessment of cumulative impacts, optimising locations of the shared real time monitoring network, validation of air predictions and optimising mitigation measures; and procedures for identifying and apportioning the source/s and contribution/s to cumulative air impacts for both mines and other sources, using the air quality and meteorological monitoring network and appropriate investigative tools such as modelling of post incident plume dispersion, dual synchronised monitors and chemical methods of source apportionment (where possible).	Tamworth is not in the same air shed and is not a suitable control site either.) - AQMS is not clear that the proposed air dispersion model would cover all of these elements; - Conceptual strategy proposed in the AQMS to identify impacts may not be as effective as described in controlling day to day effects. AQMP includes an AQMS for the BTM complex, but the AQMP/S does not appear to cover; - a suitable control monitoring site at present (given that the "Will-gai" site is close to other mining activity, it is not a suitable control monitoring site representative of regional air quality. Tamworth is not in the same air shed and is not a suitable control site either.) - AQMS is not clear that the proposed air dispersion model would cover all of these elements; - Conceptual strategy proposed in the AQMS to identify impacts may not be as effective as described in controlling day to day effects. Not Compliant Administrative

Schedule	Condition	Requirement	Finding
3	39	i) Site Water Balance, that includes: details of: sources and security of water supply, including contingency for future reporting periods; water use and management on site; any off-site water discharges; reporting procedures, including the preparation of a site water balance for each calendar year; a program to validate the surface water model, including monitoring discharge volumes from the site and comparison of monitoring results with modelled predictions; and describes the measures that would be implemented to minimise clean water use on site; "ii) Surface Water Management Plan, that includes: detailed baseline data on surface water flows and quality in the water-bodies that could potentially be affected by the project; detailed baseline data on soils within the irrigation area; detailed baseline data on hydrology across the downstream drainage system of the Namoi River floodplain from the mine site to the Namoi River, including Barbers Lagoon and The Slush Holes; a detailed description of the water management system on site, including the: clean water diversion systems; erosion and sediment controls (mine water system); mine water management systems including irrigation areas; discharge limits in accordance with EPL requirements; and water storages; detailed plans, including design objectives and performance criteria for: design and management of final voids; design and management for the emplacement of reject materials, sodic and dispersible soils and acid or sulphate generating materials; the Goonbri Creek diversion and low permeability barrier; reinstatement of drainage lines on the rehabilitated areas of the site; and	There is a water balance included. The water balance meets these criteria though some of the criteria are met in the WMP. It is an updated version, created in 2014 in line with the requirement for annual updates but there is no specific reference to an annual update of the Water Balance. The WMP developed by URS in 2013 does not contain all of these details. The detailed baseline data on hydrology of the Namoi is not present. Not Compliant

Schedule Condition		Requirement	Finding
		with modelled predictions; and a plan to respond to any exceedences of the performance criteria	
3	46	The Proponent shall make suitable arrangements to provide appropriate longterm security for the offset areas: a) for the Willeroi Offset Area the long-term security shall be provided by way of: the Proponent entering into a conservation agreement of agreements pursuant to section 69B of the National Parks and Wildlife Act 1974, recording the obligations assumed by the Proponent under the conditions of this approval in relation to these offset, and registering the agreement(s) pursuant to section 69F of the National Parks and Wildlife Act 1974; or a tenure of higher conservation status such as a National Park, or Nature Reserve, under the National Parks and Wildlife Act 1974, the conservation agreement(s) must be registered by the end of December 2013 unless agreed otherwise by the Director-General after consultation b) by the end of December 2030 unless otherwise agreed by the Director-General, for the woodland to the satisfaction of the Director-General. Note: The Department acknowledges that the Proponent is investigating the potential to transfer part of all of the Willeroi Offset Area directly to the national park estate, and accepts that interim conservation measures may be implemented prior to this transfer.	At the time of the audit not compliant with December 2013 submission date but negotiating with DP&E for an extension to this date. Not Compliant Administrative
3	49	By the end of May 2013, the Proponent shall lodge a Conservation and Biodiversity Bond with the Department to ensure that the biodiversity offset strategy is implemented in accordance with the performance and completion criteria of the Biodiversity Management Plan. The sum of the bond shall be determined by: a) calculating the full cost of implementing	Biodiversity Management Plan is not approved, site therefore not able to calculate bond amount, not compliant with deadline stated. Not Compliant
		the biodiversity offset strategy (other than land acquisition costs); and b) employing a suitably qualified quantity surveyor to verify the calculated costs, to the satisfaction of the Director-General. If the offset strategy is completed generally in accordance with the completion criteria in the	Administrative

Schedule	Condition	Requirement	Finding
		Biodiversity Management Plan to the satisfaction of the Director-General, the Director-General will release the bond. If offset strategy is not completed generally in accordance with the completion criteria in the Biodiversity Management Plan, the Director-General will call in all, or part of, the conservation bond, and arrange for the satisfactory completion of the relevant works. With the agreement of the Director-General, this bond may be combined with rehabilitation security deposit administered by DRE. Notes: Alternative funding arrangements for long term management of the Biodiversity Offset Strategy, such as provision of capital and management funding as agreed by OEH as part of a Biobanking Agreement or transfer to conservation reserve estate can be used to reduce the liability of the conservation and biodiversity bond. The sum of the bond may be reviewed in conjunction with any revision to the	
3	50	"By the end of June 2014 and every 3 years thereafter, unless both the Director-General and OEH agree to a different timeframe, the Proponent shall commission suitably qualified, experiences and independent person/s, whose appointment has been approved by the Director-General, to undertake an audit of the revegetation of the rehabilitation area and management and restoration within the Biodiversity Offset Strategy areas to the satisfaction of the Director-General. This audit must: a) include consultation with OEH, Namoi CMA, DPI Catchments and Lands, SEWPaC, CCC and DRE; b) assess the performance of the revegetation in the rehabilitation area completed to date (and the Goonbri Creek Diversion, once commenced) against the completion criteria in the Rehabilitation Management Plan; c) assess the performance of management and restoration in the off-site Biodiversity Offset d) identify any additional measures that should be implemented to improve the	This audit, non-compliant with timing requirements Not Compliant Administrative

Schedule	Condition	Requirement	Finding
		performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be applied in the establishment of native vegetation, including riparian vegetation around the realigned Goonbri Creek, both before and after the realignment is undertaken;" "e) if the completion criteria have not been met, or are not adequately trending towards being met, determine the likely ecological value of the rehabilitation and restoration once completed, and recommend additional measures to augment the Biodiversity Offset Strategy to ensure that it adequately offsets the project's impacts on biodiversity. If the audit recommends the implementation of additional measures to augment the Biodiversity Offset Strategy in accordance with (e) above, then within 6 months of the completion of the audit the Proponent shall revise the Biodiversity Offset Strategy, in consultation with the Department, OEH and SEWPaC, and to the satisfaction of the Director-General.	
3	62	The Proponent shall, in consultation with the Namoi CMA: a) develop a detailed soil management protocol that identifies procedures for: comprehensive soil surveys prior to soil stripping; assessment of top-soil and sub-soil suitability for mine rehabilitation; and annual soil balances to manage soil handling including direct respreading and stockpiling; b) maximise the salvage of suitable top-soils and sub-soils and biodiversity habitat components such as bush rocks, tree hollows and fallen timber for rehabilitation of disturbed areas within the site and for enhancement of biodiversity offset areas; and c) ensure that coal reject, or any potentially acid forming interburden materials, are not emplaced at elevations in the pit shell where they may promote acid or sulphate species generation and migration beyond the pit shell.	The requirements detailed here have been met in the Tarrawonga Rehabilitation Management Plan and the Clearing and Prestrip Procedure apart from the comprehensive soil surveys prior to topsoil stripping. Not Compliant Administrative

Schedule	Condition	Requirement	Finding	
3	64	The Proponent shall prepare and implement a Rehabilitation Management Plan to the satisfaction of the Executive Director, Mineral Resources. This plan must: a) be prepared in consultation with the Department, Forests NSW, NOW, OEH, Namoi CMA and Council; b) be submitted to the Executive Director, Mineral Resources for approval by the end of May 2013; c) be prepared in accordance with any relevant DRE guideline; d) describe how the rehabilitation of the site would be integrated with: the implementation of the biodiversity offset strategy; and the final landform for the Boggabri coal mine; e) include detailed performance and completion criteria for evaluating the performance of the rehabilitation of the site, and triggering remedial action (if necessary); f) describe the measures that would be implemented to ensure compliance with the relevant conditions of this approval, and address all aspects of rehabilitation including mine closure, final landform and final land use; g) include interim rehabilitation where necessary to minimise the area exposed for dust generation; h) include a program to monitor, independently audit and report on the effectiveness of the rehabilitation measures, and progress against the detailed performance and completion criteria; and i) build to the maximum extent practicable on the other management plans required under the approval.	Interim rehabilitation techniques and triggers are not described but are described in the Biodiversity MP. Not Compliant Administrative	
5	10	"10 By the end of June 2014 and every 3 years thereafter, unless the Director-General directs otherwise, the Proponent shall commission and pay the full cost of an independent Environmental Audit of the project. This audit must: a) be conducted by a suitable qualified, experienced and independent team of experts whose appointment has been endorsed by the Director-General; b) include consultation with the relevant agencies;	This audit, non- compliant with timing requirements. Not Compliant Administrative	

Schedule	Condition	Requirement	Finding
		c) assess the environmental performance of the project and assess whether it is complying with the requirements in this approval, and any other relevant approvals, relevant EPL/s and/or Mining Lease/s (including any assessment, plan or program required under these approvals); d) assess whether the Proponent is implementing best noise, blasting and air quality management practice; e) investigate and report on the measures taken to minimise the noise and air quality impacts of the project during meteorological conditions and/or extraordinary events when the relevant noise and air quality limits in this approval do not apply, including: the effectiveness of these measures in maintaining impacts within the relevant criteria in this approval and/or the limits in the relevant EPL; and any additional measures available to mitigate impacts under such conditions; f) review the adequacy of any approved strategy, plan or program required under the abovementioned approvals; and g) recommend measures or actions to improve the environmental performance of the project and/or any strategy, plan or program required under these approvals.	
5	11	Within 3 months of commissioning this audit, or as otherwise agreed by the Director-General, the Proponent shall submit a copy of the audit report to the Director-General, together with its response to any recommendations contained in the audit report.	This audit, non- compliant with timing requirements Not Compliant Administrative
5	13	The Proponent shall, within 3 months of the date of this approval: a) make the following information for the project publicly available on its website, on a daily basis and in a clearly understandable form: daily weather forecasts for the coming week; proposed operational responses to these weather forecasts; real-time noise and air quality monitoring data (subject to any necessary caveats); and any operational responses that were taken in response to the noise and air quality	Checked website, all present except realtime data due to data validation concerns and weather forecasts/operation management, ongoing discussion with the department regarding how to address this requirement. Not Compliant

Schedule	Condition	Requirement	Finding
		monitoring data, and b) make provision on its website for the provision of on-line and/or email comments by members of the community regarding this information, to the satisfaction of the Director-General.	Administrative

4.1. Statement of Commitments from the EA

The commitments in the Statement of Commitments were assessed and all commitments were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each commitment in the Statement of Commitments is provided in the audit protocol in Appendix C.

4.2. Environmental Protection Licence

Table 4.3 shows the conditions that were not compliant with the Environmental Protection Licence 12365 (EPL 12365). An assessment of compliance for each condition in the EPL is provided in the audit protocol in Appendix C.

Table 4.3 Environmental Protection Licence

Condition	Requiren	nent				Finding
Preamble	Fee Based Activity Scale Coal works 0-2,000,000 T handled Mining for coal > 500,000-2,000,000 T produced					2013-14 AEMR identified 2,008,613t sent to Narrabri CHPP. Not Compliant
						Administrative
	Noise generated at the premises must not exceed the noise limits in the table below.					AEMR 2012/2013, Appendix 3 TABLE A3.2
	Locality and Location	Day- LAeq (15 minute)	Evening- LAeq (15 minute)	Night- LAeq (15 minute)	Night- LA1 (1 minute)	Compliance Review –
L4.1	All other surrounding residences	35	35	35	45	Environment Protection Licence 12365. AEMR/Annual Review 2012/2013, Section 3.10.3.2 June 2012 Noise Monitoring - Attended. AEMR/Annual Review 2012/2013, Section 3.10.3.4 September 2012 Noise Monitoring - Attended
						Not Compliant Administrative

4.3. **Mining Operations Plan**

Table 4.4 shows the requirements and commitments that were not compliant with the Mining Operations Plan. An assessment of compliance for all requirements and commitments in the Mining Operations Plan is provided in the audit protocol in Appendix C.

Table 4.4 Mining Operations Plan

Reference	Requirement	Finding
3.2.1	Air quality is monitored in accordance with the Air Quality and Greenhouse Gas Management Plan, with meteorology and air quality being continuously monitored. Results are reported on a daily basis online and on an annual basis in the Annual Review and the EPL Annual Return.	Results are not being updated on a daily basis online (Section 4.3.2). Not Compliant Administrative
3.2.5	Tarrawonga will conduct further material testing during soil stripping campaigns and will manage any occurrences of PAF-LC material by selectively handling and emplacing the material so it is covered with at least 15 m of NAF material.	No records on file of further material testing following that done for the EA for management of occurrence of PAF_LC material Not Compliant
3.2.12	• Emplacing coarse rejects (back hauled from the Whitehaven CHPP) in the pit floor to maximise cover with inert material. Sampling of coarse reject by Geo Environmental Management in 2011 identified an average total sulphur content of 0.4%. The material is managed by emplacement in the pit floor and subsequent coverage by the advancing overburden emplacement. This is in accordance with the existing Section 100 approval at the Tarrawonga site for reject emplacement.	Administrative Site observations indicate that the rejects are not placed on the pit floor as required by the MOP but are placed at higher elevations, as part of the backfilling process with suitable cover to prevent spon com/AMD. Recommendation - the wording of the MOP should change at the next revision. Not Compliant Administrative
3.3.3	Tarrawonga will undertake testing to identify any PAF materials. PAF material will be selectively emplaced under at least 15 m of NAF material to minimise the potential for ARD. Volumes of PAF material identified and selectively emplaced will be reported in the Annual Review.	AEMR 2012-2013 does not include information on PAF materials Not Compliant
5.4.5	Tubestock plantings are mulched, watered and protected from grazing by tubestock bags and/or fencing where required. Tubestock planting methodologies are described in the Rehabilitation	Tubestock is not mulched at the site but is incorporated into the topsoil, no evidence of excessive grazing of

Reference	Requirement	Finding
	Management Plan and Rehabilitation Monitoring Report (2012 Annual Review, Appendix 7). Further specific rehabilitation	tubestock in the site inspection.
	methodologies for establishing the woodland and agricultural final land-uses are under development and will be documented in the Rehabilitation Management Plan.	Recommendation - remove reference to guards and mulching from MOP. Not Compliant Administrative
8.3	Results of rehabilitation monitoring will be reported in the Annual Review. The Annual Review will record monitoring results and discuss rehabilitation performance against key performance measures/indicators; compliance with regulatory requirements and TCPL commitments. The Annual Review will also discuss identified trends and instances where potential rehabilitation failure has been identified triggering intervention in accordance with a Rehabilitation TARP (Section 9.2).	AEMR 2012-2013 includes rehab monitoring results but does not assess performance against key performance measures/indicators; compliance with regulatory requirements and TCPL commitments. AEMR 2012-2013 does not discuss identified trends and instances where potential rehabilitation failure has been identified triggering intervention in accordance with a Rehabilitation TARP. Not Compliant Administrative

4.4. **Mining Leases**

The conditions in the Mining Leases were assessed and all conditions were either Compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the mining leases is provided in the audit protocol in Appendix C.

4.5. **EPBC Approval**

Table 4.5 shows the conditions that were not compliant with the EPBC Approval. An assessment of compliance for all conditions in the EPBC Approval is provided in the audit protocol in Appendix C.

Table 4.5 EPBC Approval

Condition	Requirement	Finding	
-----------	-------------	---------	--

Condition	Requirement	Finding
16	The surface and groundwater management plans must be consistent with the National Water Quality Management Strategy.	The existing Water Management Plan is not approved but does not reference EPBC requirements or the National Water Quality Management Strategy. Not Compliant Administrative
17	The person taking the action must, within six months of this approval, in collaboration with the proponent to develop and operate the Maules Creek Mine (EPBC 2009/5566), provide written advice to the Minister demonstrating how the NSW government approved surface and groundwater management plans, address the cumulative impact of groundwater drawdown as a result of mining and how this may impact on the consequent health of the remnant native vegetation in the Leard State Forest, the Leard State Conservation Area and surrounding areas. In particular the advice must address the following matters: a. maximum amount of allowable drawdown in the alluvial aquifer b. drawdown in hard rock c. trigger levels pertaining to drawdown in the alluvial aquifer when corrective actions will be required to be undertaken d. identify the depth of root zone of the native vegetation e. monitoring to assess the ongoing quality and quantity of both surface and groundwater to identify impacts on the native vegetation	a) not stated in WMP, therefore there is a risk of not identifying GW impacts b) not stated in WMP, therefore there is a risk of not identifying GW impacts Not Compliant
18	The person taking the action must within 6 months of the date of this approval, or such other timeframe specified by the Minister, provide to the Minister a report on: a. any updated modelling of surface and groundwater impacts that has been undertaken in preparing the surface and groundwater management plans b. how the surface and groundwater management plans addressed groundwater and surface water impacts on matters of national environmental significance	No evidence provided to confirm that this has occurred. Not Compliant Administrative

Condition	Requirement	Finding
22	The person taking the action must implement the regional biodiversity strategy as required under condition 41 of the NSW state government project approval dated 22 January 2013 (application number 11_0047). The required scoping report for the development of the strategy must be submitted to the Minister for approval on or before 31 July 2013. The approved strategy must be implemented.	Evidence of the submission of the scoping strategy has not been provided. Not Compliant Administrative
25	The mine site rehabilitation plan must include, at a minimum, the following information: a. targets and performance indicators to achieve effective restoration of potential habitat for the regent honeyeater, swift parrot and greater long-eared bat and White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community, including weed management; b. details of the vegetation communities to be rehabilitated and the timing of progressive rehabilitation (commencing as soon as practicable following disturbance); c. detailed soil depth surveys and analysis to inform the effective placement and restoration of solid underlying the proposed rehabilitation sites; including mapping of soils across the disturbance sites and soil sampling at no less than one sample point per 20 ha of each soil type identified. Sampling must identify; type, depth, water holding capacity, structure and physiochemical properties of each of the soil and subsoil layers; d. processes and methodologies for the removal, storage and re-layering of the top soil and sub soil layer underlying the disturbed sites being prepared for rehabilitation. These processes and methodologies must ensure the replacement of top and sub soil layers as	c) This has not been done, some soil assessment was conducted in the EA. Not Compliant
27	provided in the Environment Assessment. The person taking the action is required to submit a Conservation and Biodiversity Bond under condition 49 of the New South Wales state government project approval dated 22 January 2013 (application 11_0047). It is noted that this bond may be	No evidence provided of provision of the details of the security bond to the Minister. Not Compliant

Condition	Requirement	Finding
	combined with the rehabilitation security deposit as required under the New South Wales Mining Act 1992. The person taking the action must submit details of this bond and the rehabilitation security deposit, to the Minister, within six months of this approval. If the Minister is not satisfied that the bond and rehabilitation security deposit lodged by the person taking the action is adequate to provide for the requirements referred to under conditions 19, 20, 22, 23 and 24, the Minister may require the person taking the action establish an additional bond or equivalent financial instrument in trust, under conditions approved in writing by the Minister.	Administrative
30	All survey data collected for the project must be recorded so as to conform to data standards notified from time to time by the department. When requested by the department, the proponent must provide to the department all species and ecological survey data and related survey information from ecological surveys undertaken for matters of national environmental significance. This survey data must be provided within 30 business days of request, or in a timeframe agreed to by the department in writing. the department may use the survey data for other purposes.	No evidence of compliance with the required survey standards Not Compliant Administrative

4.6. Environmental Management Strategy

Table 4.6 Environmental Management Strategy shows the conditions that were not compliant with the Environmental Management Strategy. An assessment of compliance for all conditions in the Environmental Management Strategy is provided in the audit protocol in Appendix C.

Table 4.6 Environmental Management Strategy

Condition	Requirement	Finding
4.8	A review of the mine's compliance with all conditions of PA 11_0047, ML 1579, EPL 12365 and all other approvals and licences will be undertaken prior to (and included within) each AEMR/Annual Review	This is done, but not for all approvals and licenses, water licences and EPBC approval not covered.
	submitted to the Director-General and DRE. The AEMR will also be provided to Council, relevant agencies, the Community Consultative Committee (CCC) and to the	Not Compliant Administrative

Condition	Requirement	Finding
	public on Whitehaven's website.	
4.7.1	Copies of all management plans/strategies or monitoring programs, together with the results of independent audits undertaken in accordance with PA 11_0047 will also be provided to the CCC and the Councils, and made publicly available on the Whitehaven website.	These are on the website and provided to the CCC. No evidence that they are provided to Councils Not Compliant Administrative
4.7.1	The AEMR/Annual Review will be provided to both NSC and GSC, relevant agencies and to the CCC members and also made available to the public on Whitehaven's website.	No evidence of provision of these reports to Councils Not Compliant Administrative
4.7.1	In addition to the above, Tarrawonga Coal may also utilise the local press to present feature articles on the mine's progress. Other forms of communication with the community will include the provision of newsletters to the landholders around the mine site, with copies made available at the Council offices and on the Whitehaven website. These newsletters will provide updates as to site developments and will be issued generally every 6 months or following specific events at the site for which community updates are warranted.	On website Newsletters not sent to Council Not Compliant Administrative

4.7. Environmental Monitoring Program

The Environmental Monitoring Program was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Environmental Monitoring Program is provided in the audit protocol in Appendix C.

4.8. Noise Management Plan

Table 4.7 shows the requirements and commitments that were not compliant with the Noise Management Plan. An assessment of compliance for all requirements and commitments in the Noise Management Plan is provided in the audit protocol in Appendix C.

Table 4.7 Noise Management Plan

Reference	Requirement	Finding
	Attended noise surveys will be	Attended Noise Monitoring –
	conducted as follows:	March 2014 Tarrawonga Coal
4.4	 All noise investigations will be 	Mine Boggabri only makes
	carried out in accordance with	reference to the Industrial Noise
	NSW EPA's Industrial Noise Policy,	policy

Reference	Requirement	Finding
	2000 (INP), Environmental Noise	Attended Noise Monitoring –
	Control Manual (ENCM) and	March 2014 Tarrawonga Coal
	applicable Australian Standards;	<i>Mine Boggabri</i> does not make
	 Noise levels will be measured in 	reference to this standard
	one-third octave bands using an	The noise monitoring consultant is
	instrument with IEC Type 1	very experienced and these issues
	characteristics as defined in	are likely to be an administrative
	Australian Standard AS IEC	oversight in the report.
	61672.1 – 2004 "Electroacoustics –	
	Sound Level Meters –	Not Complaint
	Specifications". The instrument	Administrative
	will have current calibration as per	
	manufacturer's instructions and	
	field calibration will be confirmed	
	before and after measurements	
	with a sound level calibrator;	
	 Details regarding plant 	Plant configuration is not recorded
	configuration, survey interval,	
	weather conditions, extraneous	Not Compliant
4.4	noise sources, monitoring	Administrative
7.7	locations and times of	
	measurement will be recorded for	
	inclusion in the noise monitoring	
	report.	
	Attended noise monitoring reports	Attended Noise Monitoring –
	issued by the acoustic consultant	March 2014
	will include details on the date and	Monitoring personnel were not
	time of monitoring, location of	included in the report, only the
6.1	monitoring and monitoring	names of author of the report and
	personnel (as required by	reviewer.
	Condition M1.3 of the EPL).	
		Not Compliant
		Administrative

4.9. Noise Management Strategy

The Noise Management Strategy was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Noise Management Strategy is provided in the audit protocol in Appendix C.

4.10. Blast Management Plan

Table 4.8 shows the requirements and commitments that were not compliant with the Blast Management Plan. An assessment of compliance for all requirements and commitments in the Blast Management Plan is provided in the audit protocol in Appendix C.

Table 4.8 Blast Management Plan

Reference Req	quirement	Finding
---------------	-----------	---------

Reference	Requirement				Finding
	monito Monito	ring parar		be monitored. taken for each 2.	Some events don't comply due to missed monitoring data - blast results records show missed results due to lack of triggering. Interview with EO indicated that these results were from a period where a contractor managed
	Parameter	Units of Measure	Frequency	Sampling Method	blasting including monitoring.
5.1.1	Blast Vibration	DB(Lin Peak)	Every Blast	Type 1 Noise Blast Logger	
J.1.1	Diss Vitragori	inns	Currently, blasting and monit managed by TCPL and there he no missed results since the collect site.	managed by TCPL and there have been no missed results since the contractor left site. No action required by TCPL as the issue has been resolved.	

4.11. Blast Monitoring Program

The Blast Monitoring Program was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Blast Monitoring Program is provided in the audit protocol in Appendix C.

4.12. Blast Management Strategy

The Blast Management Strategy was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Blast Management Strategy is provided in the audit protocol in Appendix C.

4.13. Air Quality and Greenhouse Gas Management Plan

Table 4.9 shows the requirements and commitments that were not compliant with the Air Quality and Greenhouse Gas Management Plan. An assessment of compliance for all requirements and commitments in the Air Quality and Greenhouse Gas Management Plan is provided in the audit protocol in Appendix C.

Table 4.9 Air Quality and Greenhouse Gas Management Plan

Reference	Requirement	Finding
3	Use predictive meteorological forecasting and predictive air dispersion modelling, together with real time monitoring data to inform operational practices in advance of the commencement of shift. This will aid in the day to day planning of mining activities giving due consideration to prevailing weather conditions and potential for dust	This is not done yet, predictive air dispersion modelling has not been implemented, the plan is not yet approved. Not Compliant

Reference	Requirement	Finding
	generation and dispersion.	
3	 Participation in an Australian Coal Association Research Project (ACARP) study to trial best practice measures for disturbing and handling overburden. 	No evidence provided to support the sites participation in the ACARP project. Not Compliant Administrative
4.3.2	For each deposited dust monitoring location, once each month the glass container used to capture the deposited dust will be removed, replaced and sent to a NATA accredited laboratory for analysis.	NATA certification is noted in laboratory reports Note the laboratory report provided as evidence notes that the results could not be certified as they claimed TCPL had collected the samples therefore the exposure time could not be verified, but the samples were collected by Lab staff who should have been able to verify exposure times from their records.
		Not Compliant Administrative

4.14. Air Quality Management Strategy

The Air Quality Management Strategy was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Air Quality Management Strategy is provided in the audit protocol in Appendix C.

4.15. Water Management Plan

Table 4.10 shows the requirements and commitments that were not compliant with the Water Management Plan. An assessment of compliance for all requirements and commitments in the Water Management Plan is provided in the audit protocol in Appendix C.

Table 4.10 Water Management Plan

Reference	Requirement	Finding
2.3.2	Saline groundwater, pumped from the mine pit, is deposited in the storage dams PW2, PW3 within the site. This water is used in dust suppression activities for the coal crusher or is allowed to evaporate. The use of pit water for dust suppression minimises the need to use clean water for this purpose and is seen as suitable way to recycle water onsite. The water will be applied at a suitable rate to ensure negligible runoff. Any water falling/applied onto the mine site	The area around the coal loader drains to dams that are in the dirty water circuit but have an offsite discharge point. Not Compliant

Reference	Requirement	Finding
	would be classed as 'dirty water' and collected in dirty water dams as discussed in Section 2.5. These dams are sized and operated to ensure that they only discharge in extreme high flows thus the water used for dust suppression has a minimal chance of discharging offsite. The water in these dirty water dams are regularly tested and monitored.	
2.3.2	The groundwater model will be updated every 2 years, using data recorded on site. If the estimates should show that additional groundwater licencing is required, this will be sought from the water markets.	The groundwater model was last updated for the EA for the current approval, report on model update is dated January 2012, an update of the groundwater model is overdue. Not Compliant
6.3	The water quality monitoring program has been designed to provide data on the flows leaving site as well as the flows in the receiving creeks. As the approved works are undertaken, the potential for impacting Goonbri Creek, Bollol Creek and the downstream watercourses is increased. For this reason the surface water monitoring program will be updated to include the following locations: • Two new surface water flow gauging stations on Goonbri Creek upstream (GC-U) and downstream (GC-D) of the permanent Goonbri Creek alignment; • Two water level (flood) gauges in the lower reaches and overflow areas of Bollol/Goonbri Creek (FG1 and FG2); • Water quality monitoring on Goonbri Creek at the two new gauging stations; • Water quality monitoring in new onsite water storages and licensed discharge points; and • Two pluviometers in the upper (PV1) and mid (PV2) reaches of the Goonbri Creek catchment. These locations will be introduced as soon as possible and will begin to	This has generally not been conducted. Not Compliant
6.4	establish baseline data. Monitoring is the responsibility of the Environmental Officer and should be conducted by a suitably qualified professional in accordance with the	There is no evidence that the ALS/ACIRL sampler is a suitably qualified professional. Letters of competency show no indication of professional

Reference	Requirement	Finding
	relevant Australian Standards. The frequency of monitoring and range of parameters analysed during flow and routine monitoring should be reviewed as part of the revision process for this document.	education. This is probably not necessary Recommendation - modifying the WMP to remove the requirement for sampling to be conducted by a professional. Not Compliant Administrative
7.5	 Bores will be purged of at least three well-volumes of groundwater using a submersible pump or bailer. The method of purging will be tailored for each site, dependant on the parameters being analysed to ensure the accuracy of results. Collection of groundwater samples will be undertaken following the purging of each bore. For groundwater samples, sampling devices will be dedicated and/or disposable for each sample or otherwise decontaminated between sampling locations. If rinsing is used, rinsate samples should be included in the QA/QC program as appropriate. 	this does not happen, reference the field sheet 20-02 year not recorded but samples were submitted 24-02-14 This is not clear from the analysis and sampling data provided, there is no record in the QA report from the lab of rinsate samples being tested. The water sampled should be from the aquifer not that exposed to the atmosphere in the well. Not Compliant
7.5.1	An organised internal approach to data management and monitoring documentation will significantly enhance the intended benefits of the monitoring program. When set up, it will facilitate the identification of potential issues of concern in a timely manner, such that appropriate contingency actions can be implemented if warranted. In addition to this, the raw monitoring data will be published on the mine's website within two weeks of sampling and will be given to the local NOW hydrogeologist.	The data is not posted up within two weeks and is not supplied to NOW. Not Complaint Administrative
7.8	The groundwater model will be recalibrated 2 years after the approval of this Plan. Recalibration will then occur every 5 years thereafter and at least 12 – 18 months prior to cessation of mining. Following the first review, if necessary, the groundwater model will be recalibrated and confirmatory forward impact predictions made.	This has not occurred, Ground water model updated for the EA in January 2012. Not Compliant
8	The surface and ground water response plan will include a protocol	There is no Surface and Groundwater Response Plan, there is a TARP but it

Reference	Requirement	Finding
	for managing and reporting any: • incidents; • complaints;	does not detail the management of complaints.
	 non-compliances with statutory requirements; and exceedances of the impact assessment criteria and or performance criteria; and; a protocol for periodic review of the plan. 	Not Compliant Administrative

4.16. Groundwater Contingency Plan

Table 4.11 shows the requirements and commitments that were not compliant with the Groundwater Contingency Plan. An assessment of compliance for all requirements and commitments in the Groundwater Contingency Plan is provided in the audit protocol in Appendix C.

Table 4.11 Groundwater Contingency Plan

Reference	Requirement	Finding
2.3	In order to monitor for natural variation in groundwater level and quality, the following bores will also be monitored. • GW044997 – within the alluvial aquifer on the "Templemore" property. • GW020432 – within the volcanics aquifer on the "Merriown" property. • GW006013 – within the Permian Coal Measures on the "Bollol Creek	2012-13 AEMR indicates that GW044997 is monitored but GW020432 and GW006013 are not recorded, no explanation is given in the text. Not Compliant
5	Station"property. The results of all groundwater monitoring will be made publicly available at the offices of Narrabri Shire Council and Gunnedah Shire Council and at TCPL's Boggabri office. These results will be updated at least every three months. Each year, the results of the groundwater monitoring program will be summarised and presented in the AEMR. This will include an analysis of the monitoring results against the trigger levels listed in Table 2, previous monitoring results and the monitoring results of background variance. Based on these results, trends in the groundwater levels, quality and availability will be identified and the requirement to initiate the contingency	No data is sent to the Councils offices. Not Compliant Administrative

Reference	Requirement	Finding
	measures identified in Section 4.	

4.17. Water Management Strategy

Table 4.12 shows the requirements and commitments that were not compliant with the Water Management Strategy. An assessment of compliance for all requirements and commitments in the Water Management Strategy is provided in the audit protocol in Appendix C.

Table 4.12 Water Management Strategy

Reference	Requirement	Finding
4.1.1	Contaminated water, containing suspended solids and soluble salts, will be generated from coal stockpiles and the mining void, as well as groundwater inflows to the mining void. In addition, dirty water containing suspended solids will be generated from runoff from disturbed areas within the mine sites, including from infrastructure areas, unshaped spoil dumps and haul roads. For all mines, contaminated water will be retained onsite for use, and dirty water will be retained in settlement ponds prior to discharge or on-site use.	The water collected from the coal loading area is located in Dam SB4, which then flows through a series of dams to Licensed Discharge Point 1 (LDP1 - EPL 12365) and is discharged offsite. Not Compliant

4.18. Biodiversity Management Plan

Table 4.13 shows the requirements and commitments that were not compliant with Biodiversity Management Plan. An assessment of compliance for all requirements and commitments in the Biodiversity Management Plan is provided in the audit protocol in Appendix C.

Table 4.13 Biodiversity Management Plan

Reference	Requirement	Finding
4.1.2	A seed collection and propagation program will be implemented so that areas to be revegetated are genetically comparable with those in the surrounding woodland areas.	There is no seed collection program at Tarrawonga but the Biobank site has a seed collection program and the offset site will have a seed collection program. The ecologist in the audit team indicated that the distance between the seed collection area and the site could be significant in terms of the risk of genetic variation, local seed should be sought. Not Compliant

Reference	Requirement	Finding
4.3.2	Subsoils and topsoils will be characterised prior to re-spreading to determine the type and application rates for any required soil ameliorants (e.g. lime, gypsum, fertiliser and organics) to maximise the availability of soil reserves for rehabilitation works.	Not done in the audit period, respreading has occurred in the audit period. Not Compliant
4.3.3	The progress and effects of salvaging and reusing habitat resources are monitored annually.	The Tarrawonga Monitoring Report 2013 notes the reuse of materials in the plots used for rehabilitation monitoring but there is no assessment of or comment on the effects of reusing these materials. The AEMR also does not address this issue. Not Compliant Administrative
4.3.3	TCPL will arrange for the collection of seed at the site at appropriate times after flowering.	This has not occurred Not Compliant
4.3.3	Planting will preferentially be undertaken in the Autumn period on previously established riplines followed by mulching around each seedling.	This does not occur, no mulching is conducted. Not Compliant
4.6.2	Overburden and interburden will be characterised prior to emplacement in waste dumps to ensure the final outer surfaces of the overburden emplacements (and structures such as drainage elements) are constructed with suitable non-sodic or low sodicity material where possible.	This does not occur, no characterisation of overburden has taken place in the audit period. Not Compliant
4.7.2	Minimise the potential for seed and organic matter transport to or from the rehabilitation area by ensuring all plant and equipment are weed free (via inspection of vehicles and plant and use of the site's vehicle wash bay).	This does not occur - no inspections have taken place specifically for weed vector control. Not Compliant
4.7.2	Installation of fauna exclusion fencing and/or tree guards for newly planted tubestock.	This does not occur. Not Compliant
4.7.3	Fauna exclusion fencing and/or tree guards are installed to exclude vertebrate pest species from rehabilitation areas / juvenile vegetation.	This does not happen. Not Compliant
4.10.3	Fuel load/fire security will be inspected at a minimum of 12 monthly intervals, occurring prior to the commencement	This assessment has not been documented.

Reference	Requirement	Finding
	of each bushfire season	Not Compliant
		Administrative
	Fauna exclusion fencing and/or tree	This is not conducted.
	guards are installed to exclude	
5.2.1	vertebrate pest species from	Not Compliant
	rehabilitation areas / juvenile	
	vegetation.	
	Fuel load/fire security will be inspected	This assessment has not been
	at a minimum of 12 monthly intervals,	documented.
5.2.2	occurring prior to the commencement	N . 6 . II .
	of each bushfire season	Not Compliant
	TCDI : !!! for the call of the call	Administrative
	TCPL will arrange for the collection of	See above, seed not collected at the
5.2.4	seed at the site at appropriate times	mine site.
5.2.4	after flowering and will extend this program to involve other contractors if	Not Compliant
	required.	
	Planting will preferentially be	See above, no mulching is done.
	undertaken in the Autumn period on	See above, no maierning is done.
5.2.4	previously established riplines followed	Not Compliant
	by mulching around each seedling.	. Tot compilant
	Fauna exclusion fencing and/or tree	This is not conducted.
	guards are installed to exclude	
5.3.1	vertebrate pest species from	Not Compliant
	rehabilitation areas / juvenile	
	vegetation.	
	Fuel load/fire security will be inspected	This assessment has not been
	at a minimum of 12 monthly intervals,	documented.
5.3.2	occurring prior to the commencement	
	of each bushfire season.	Not Compliant
		Administrative
	Fauna exclusion fencing and/or tree	This is not conducted.
F 4 4	guards are installed to exclude	Not Compliant
5.4.1	vertebrate pest species from	Not Compliant
	rehabilitation areas / juvenile	
	vegetation. Fuel load/fire security will be inspected	This assessment has not been
	at a minimum of 12 monthly intervals,	documented.
5.4.2	occurring prior to the commencement	documented.
3.1.2	of each bushfire season.	Not Compliant
	0. 000. 000	Administrative
	Whitehaven will undertake annual	Internal audits not conducted in the
	internal audits to align with its	audit period
10	AEMR/Annual Review and MOP	
	reporting requirements, as well as	Not Compliant
	independent audits every 3 years	Administrative
	An internal annual environmental audit	Internal audits not conducted in the
10.1	will be undertaken to align with	audit period
	reporting requirements for the MOP	

Reference	Requirement	Finding
	and AEMR/Annual Review. The annual	Not Compliant
	audit will focus on the compliance with	Administrative
	the requirements of the BMP and	
	where corrective action is required.	
	An annual audit and compliance report	
	will be prepared which will include the	
	methodology of the audit, actions	
	implemented in accordance with the	
	BMP, non-compliances with the BMP,	
	and corrective actions taken. A	
	summary of this report will be included	
	in the AEMR/Annual Review.	This could be a second by the state of the state of
	In accordance with Schedule 5,	This audit, not compliant with timing.
	Condition 10 of the Project Approval,	N. C. III.
10.2	an independent audit will be	Not Compliant
10.2	undertaken by the end of June 2014	Administrative
	and every 3 years thereafter, unless	
	the Director-General directs otherwise.	

4.19. Willeroi Biodiversity Offset Area Management Plan

Table 4.14 shows the requirements and commitments that were not compliant with the Willeroi Biodiversity Offset Area Management Plan. An assessment of compliance for all requirements and commitments in the Willeroi Biodiversity Offset Area Management Plan is provided in the audit protocol in Appendix C.

Table 4.14 Willeroi Biodiversity Offset Area Management Plan

Reference	Requirement	Finding
	In accordance with Schedule 5, Condition 10 of the DP&I PA, an	This audit, timing not compliant.
9.5	independent audit will be undertaken by the end of June 2014 and every 3 years thereafter, unless the Director- General directs otherwise.	Not Compliant Administrative

4.20. Rehabilitation Management Plan

Table 4.15 shows the requirements and commitments that were not compliant with the Rehabilitation Management Plan. An assessment of compliance for all requirements and commitments in the Rehabilitation Management Plan is provided in the audit protocol in Appendix C.

Table 4.15 Rehabilitation Management Plan

Reference	Requirement	Finding
6.4.2	Prior to soil stripping activities, additional investigations will be conducted to confirm the appropriate soil stripping depths for each soil type	This is done during stripping, not prior to. Not Compliant

Reference	Requirement	Finding
	recommended for salvage (refer to	
	Section 4.3.2 and Figure 7 in the	
	Biodiversity Management Plan).	
	"The soil stripping and handling	
	techniques described below have been	
	developed to maximise the soil	
	resource salvaged and minimise	
	deterioration of the soil resource.	
	Undertake soil testing to characterise	
	subsoils and topsoils prior to stripping	
	to determine the type and application	
	rates for any required soil ameliorants	
	(e.g. lime, gypsum, fertiliser and	
	organics);	
	Mulch all cleared vegetation not	
	retained for habitat augmentation on	
	site and mix into topsoils as a soil	
	conditioner;	
	Confirm the depth of soils to be	
	stripped in accordance with the	
	Biodiversity Management Plan and the	
	MOP. Topsoils and subsoils are	
	generally to be stripped and handled	
	separately;	
	Topsoil would be maintained in a	
	slightly moist condition during stripping	
	where possible. Material	
	will not be stripped in either an	
	excessively dry or wet condition; and	
	Soil would be graded or pushed into	
	windrows with excavators, graders or	
	dozers for loading into rear dump	
	trucks by front-end loaders, or will be	
	collected by scraper fleets and	
	replaced directly on rehabilitation	
	areas where available or placed in	
	temporary stockpile locations. "	New absenced to size 1111
	Clearly demarcate stockpiles with	Not observed in site visit
	signage. Install appropriate erosion	Not Compliant
6.4.2	and sediment controls (e.g.: bunds	Not Compliant
	along the upslope perimeter and	
	sediment fences along the toes of the	
	downslope batter;	This does not occur
	Tubestock plantings will be mulched,	This does not occur.
6.5.4	watered and protected from grazing by	Not Compliant
	tubestock bags and/or fencing	Not Compliant
	where required.	Not done
655	Install fencing and/or tree guards to	Not done
6.5.5	protect juvenile vegetation and tubestock from herbivores.	Not Compliant
	tubestock from herbivores.	Not Compliant

4.21. Heritage Management Plan

Table 4.16 shows the requirements and commitments that were not compliant with the Heritage Management Plan. An assessment of compliance for all requirements and commitments in the Heritage Management Plan is provided in the audit protocol in Appendix C.

Table 4.16 Heritage Management Plan

Reference	Requirement	Finding
4.1.1	 Depending on the structural integrity of the tree, the tree may need to be cut back to a manageable size before attempting to remove the tree. Preparation of the tree for removal, by attaching lengths of rigid timber to the outside of the padding, to effectively create 'splints' that will spread pressure evenly along the tree surface during movement. Lifting of the tree using a crane, with the tree supported by straps placed around the padded trunk or a heavy cargo net. During the initial stages of lifting, the crane would also support the tree when the final cut is made near the ground surface. After the base cut is complete, the tree would be suspended by the straps and manoeuvred into a horizontal position so that it is possible to examine the underside of the tree. If the core is too fragile and is likely to drop out, a cap would be secured in place to prevent loss. During this stage, any termite debris and loose materials would also be removed from the hollow base, which would then be stabilised by insertion of expanding polyurethane foam to fill the cavity. Movement of the tree onto the back of a flat bed truck, where it would be placed on a bed of foam and secured with straps for transport to the Cumbo Gunerah Keeping Place at the Red Chief Local Aboriginal Land Council office in Gunnedah or another suitable location agreed with the RAP's. Unloading the tree using a crane, with the tree supported by straps. The tree will be supported on a well-drained bed of rubber or similar protective surface. Periodic treatment for termites or other 	The methodology utilised to remove the tree was not precisely in accordance with the methodology described here due to a direction given by the arborist during the operation. The methodology in the HMP is too prescriptive making it difficult to comply with when on ground works occur and variations in tree size, location, tree condition and other factors need to be considered in the methodology for protection of the scar tree. Recommendation - at the next review of the HMP, the scar tree section be make more generic to allow for site conditions when recovering and preserving scar trees. Not Compliant Administrative

Reference	Requirement	Finding
	insect infestation would be undertaken as	
	required during storage.	

4.22. Aboriginal Heritage Conservation Strategy

The Aboriginal Heritage Conservation Strategy was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Aboriginal Heritage Conservation Strategy is provided in the audit protocol in Appendix C.

4.23. Bushfire Management Plan

Table 4.17 shows the requirements and commitments that were not compliant with the Bushfire Management Plan. An assessment of compliance for all requirements and commitments in the Bushfire Management Plan is provided in the audit protocol in Appendix C.

Table 4.17 Bushfire Management Plan

Reference	Requirement	Finding
5.2.8	 Fuel load / fire security will be inspected at minimum 12 monthly intervals, occurring prior to the 	This has not been documented. Not Compliant
	commencement of each bushfire season.	Administrative
5.3.4	 Fire breaks will be inspected at a minimum of 6 monthly intervals. 	This has not been documented.
5.3.4		Not Compliant
		Administrative

4.24. Waste Management Plan

Table 4.18 shows the requirements and commitments that were not compliant with the Waste Management Plan. An assessment of compliance for all requirements and commitments in the Waste Management Plan is provided in the audit protocol in Appendix C.

Table 4.18 Waste Management Plan

Reference	Requirement	Finding
5	Waste management information (including review of the effectiveness of waste minimisation and management) will be documented and reported in each AEMR/Annual Review, where applicable. Details will be provided on the implementation success of the Waste MP and any areas that require improvement.	The AEMR addresses waste management but does not address the requirement to comment on waste minimisation effectiveness. Not Compliant Administrative

4.25. Road Closure Management Plan

The Road Closure Management Plan was assessed and all conditions were either compliant or not applicable. No conditions were found to be "not compliant". An assessment of compliance for each condition in the Road Closure Management Plan is provided in the audit protocol in Appendix C.

4.26. Previous Independent Environmental Audit

The issues raised in the previous audit (UMWELT, 2011) were reviewed and none have been carried forward into the non-compliances or recommendations for this audit, generally because they have been rectified or are no longer applicable.

5. ADEQUACY / EFFECTIVENESS OF ENVIRONMENTAL MANAGEMENT AND MANAGEMENT DOCUMENTATION

From an environmental perspective, the key potential impacts resulting from activities at the Tarrawonga Mine are ecological sustainability and loss of habitat, rehabilitation, surface water quality, ground water extraction, air quality and noise. The majority of the management plans are relatively new and most were not final or approved at the time of the audit. As such, the comments in this section of the audit potentially offer more value to the site in improving environmental performance than addressing areas found not compliant with the management plans and associated documents.

This section looks into the adequacy of the mitigation measures and the on ground applicability of the management measures proposed in the site environmental management documentation.

5.1. Ecology

The offset area associated with the Tarrawonga Extension approval PA 11-0047 has been approved as Willeroi West. A draft management plan has been prepared for the Willeroi West Biodiversity Offset Area (ELA January 2014), the plan was not approved at the time of the audit. The plan addresses requirements set out in approval conditions relating to management of the offset site under the EP&A Act, PA 11_0047 and the EPBC Act, PA 2011/5923.

All management zones on the Willeroi BOA were inspected on 9 September 2014. Albeit very brief, the inspection confirmed the condition and composition of vegetation identified in the management zones in the Willeroi Biodiversity Offset Management Plan (WBOMP). Specific site issues such as weeds (Coolatai, Sweet Briar), feral pests (goats), gully erosion and bulldozed tree clearing windrows were observed. Cattle had recently been removed from the grassland zone (MZ3).

In general the WBOMP met all the requirements identified in the approval conditions. However, two key recommendations have been made to improve the environmental performance in relation to the requirements off the WBOMP.

EPBC approval 2011/5923 – condition 13e details ongoing ecological monitoring programs, performance criteria, targets and provisions for adaptive management, including but not limited to "a set of measurable ecological indicators for detecting changes to the White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community, including those that may be ascribed to ongoing water stress;"

It is recommended that Tables 5.2 and Table 5.3 in the WBOMP be revised in accordance with Table 5.1 Ecological attributes measured in condition plots (modified from DECC 2009* & Sivertsen 2009*) and Table 5.2: Sampling units below. These tables set out ecological attributes and the scale of measurement in vegetation condition monitoring plots.

The attributes & methods outlined in these tables are not consistent with BioBanking methodology for site attributes (OEH 2012). For example, transects should be 50 m long (not 100 m), cover recorded along transects should be measured as percent foliage cover (not projected crown cover), groundcover should be separately recorded for grasses, sub-shrubs and other vascular plants <1m high, coarse woody debris should be recorded at the baseplot

scale (50m x 20m), and the number of trees with hollows should be included to make comparisons against all ten Biometric benchmark attributes.

Changes in site attributes over time provide an indication of trends in condition of threatened ecological communities and threatened fauna species habitat. Impacts of climate change on grassy woodlands include death of trees from moisture stress, altered shrub-grass balance potentially in favour of shrubs, and changes in groundcover composition (Prober et al. 2010). It is recommended that 2 additional attributes (tree health, and density) should be monitored in Box-gum Woodland vegetation types on Willeroi (White Box grassy woodland of the Nandewar and Brigalow Belt South Bioregions & Yellow Box - Blakely's Red Gum grassy woodland of the Nandewar Bioregion). Suggested revisions of ecological attribute tables are incorporated into revised tables provided below:

Table 5.1 Ecological attributes measured in condition plots (modified from DECC 2009* & Sivertsen 2009*)

Theme	Attribute	Description
	Cover	Measured as Percent Foliage Cover along a 50m transect At 10 points along a 50-m transect (i.e. every 5 m), estimate the percentage foliage cover directly overhead. Divide the total by the number of points (i.e. 10) measured along the transect.
Native	Health	Categorised into categories based on proportion of canopy dieback: Normal healthy crown 100%, Slight dieback in crowns 70% - 99%, Moderate dieback in crowns 30% - 69%, Severe dieback in crowns 1% - 30%, Dead 0% for all native trees in baseplot (20m x 50m)
overstorey (canopy)	Density (native species only)	Recorded as tally of woody stem sizes (DBH): 5-10cm, 10-20cm, 20-30cm, 30-60cm, >60cm -old growth (tally live and dead trees separately) for all native trees in baseplot (20m x 50m)
	Number of trees with hollow	Count of the number of living and dead trees with at least one hollow for all native trees in baseplot (20m x 50m)
	Recruitment	Presence/absence of native trees in the juvenile and sapling diameter classes in baseplot (20m x 50m)
	Richness	List of native overstorey species (including emergents)
Native midstorey (shrub and small tree)	Cover	Measured as Percent Foliage Cover along a 50m transect At 10 points along a 50-m transect (i.e. every 5 m), estimate the percentage foliage cover in mid-storey. Divide the total by the number of points (i.e. 10) measured along the transect.
	Richness	List of native midstorey species
Native ground layer	Cover	Measured as Percent Foliage Cover of native groundcover plants (grasses, sub-shrubs, other – non-woody not grass, <1m) at each of 50 points along a 50 m transect. At 50 points along the 50-m transect (i.e. every 1 m), record whether native grass, sub-shrub, or other intersects that point. Divide the total of 'hits' by the number of points measured along the transect (i.e. 50).
	Richness	List of native groundcover species

Theme	Attribute	Description
Exotic species	Cover	Measured as Percent Foliage Cover along a 50m transect for exotic canopy and exotic midstorey species. At 10 points along a 50-m transect (i.e. every 5 m). Measured as Percent Foliage Cover. Recorded as a 'hit' or 'miss' at each of 50 points along a 50 m transect, then calculated as % cover exotic ground layer species.
	Richness	List of exotic flora species in all strata
	Large woody debris	Measured in the sub-plot as the total number and combined length of all sections of dead fallen timber ≥ 10 cm diameter, ≥ 0.5 m in length, and completely detached from living or dead standing trees in baseplot (20m x 50m)
	Organic litter	Recorded as a 'hit' or 'miss' at each of 50 points along a 50 m transect, then calculated as % litter cover
Groundcover	Cryptograms	Recorded as a 'hit' or 'miss' at each of 50 points along a 50 m transect, then calculated as % cryptogram cover
	Bare ground	Recorded as a 'hit' or 'miss' at each of 50 points along a 50 m transect, then calculated as % bare ground cover
	Rock	Recorded as a 'hit' or 'miss' at each of 50 points along a 50m transect, then calculated as % rock cover

^{1.} DECC 2009. Field data sheets for BioBanking: biobank / development site proposal package February 2009. (BioBanking09083fdsbbdev.pdf).

Table 5.2: Sampling units

Attributes	Sampling units
Photo Point	Point from which plot is located and oriented, and from which plot image is recorded 5 m back behind reference point (taken after plot laid out)
Reference point	Site marker post with ID tag and flagging tape attached
Alignment Point	Positioned 20 m from photo point along the transect
_	Native canopy health
Baseplot	Tree density (native species only)
(20 m x 50	Number of trees with hollows
m)	Recruitment of native canopy species
	Coarse woody debris (logs >10cm diameter)
Subplot (20 m x 20 m)	List of native and exotic species (all strata)
	Native and exotic canopy cover
	Native and exotic midstorey cover
Transect	Native and exotic groundcover (grasses, sub-shrubs, other)
	Organic litter cover
(50 m)	Cryptogram cover
	Rock cover
	Bare ground cover

^{2.} Sivertsen, D 2009 Native Vegetation Interim Type Standard, Department of Environment, Climate Change and Water NSW, Sydney. (10060nvinttypestand.pdf).

Additional	
area (within	
50m of	Additional native canony energies list
baseplot, ie.	Additional native canopy species list Additional recruitment of native canopy species
a one	Additional number of trees with hollows
hectare area	Additional number of trees with hollows
surrounding	
the plot)	

EP&A Act PA 11_0047– condition 47c "Detail the performance and completion criteria for evaluating the performance of the biodiversity offset strategy and for triggering remedial action"

It is recommended that a Trigger Action Response Plan (TARP) should be developed specifically for the Willeroi Biodiversity Offset Management Plan.

Section 7 of the Tarrawonga Biodiversity Management Plan (ELA 2013) describes the Trigger Action Response Plan (TARP) for managing risks to biodiversity management on the project and offset sites. Trigger values will be developed based on annual flora & fauna monitoring program outcomes for the Willeroi BOA.

5.2. Rehabilitation

The site has a long way to go time-wise with respect to rehabilitation and completion given the site will operate through until 2030. The rehabilitation that has been established is relatively stable but not of exceptional quality.

The opening up of the northern extension will further spread the rehabilitation effort reducing the operations ability to ensure rehabilitation keeps pace with mining. The southern emplacement presents an unrehabilitated emplacement face to the public, a steep unstable slope (angle of repose) to Goonbri Creek (120 metres distant) and a large disturbed area that allows the mobilisation of particulates.

Other issues identified in the audit included:

Lack of soil testing – this has impacts on the rehabilitation due to the lack of ability to target the use of soil ameliorants (gypsum, lime or dolomite etc), the site should consider implementing a testing regime for soils prior to topsoil spreading operations to allow the correct ameliorant to be applied at the time of spreading the topsoil. There was a section of recently established rehabilitation on the western end of the southern emplacement that showed evidence of rilling. Whilst not excessive at the time of the audit the correct ameliorant could reduce the potential for soil erosion prior to the establishment of competent vegetative cover. Correctly ameliorated soil will also assist with more rapid establishment of vegetation reducing erosion risks and reducing the area exposed to wind erosion and particulate generation.

Trending of Rehabilitation Progress and Failure of Rehabilitation – is not reported or tracked adequately in the Annual Rehabilitation Report. The lack of reporting and review of these issues reduces the opportunity to learn from poor results and ensure that the long term rehabilitation objectives will be met.

Local provenance seed collection – the ecologist on the audit team expressed concern that the seed used for rehabilitation at Tarrawonga was collected at Rocglen which is approximately 18 kilometres away. The risk of genetic variation over that distance should be considered and local seed from the mine site and surrounds used in preference. The sensitivity of this issue relates to maintaining the genetics of the Leard State Forest that the rehabilitation aims to support through replacement of species removed in the mining operation.

Rehabilitation Trials - Given the poor soils and rehabilitation quality to date, TCPL should commence trials of options for rehabilitation establishment at the site. Options to consider should include:

- Soil testing an ameliorant application;
- Fertiliser types and quantities;
- Seed application rates;
- Topsoil age some of the topsoil stockpile have been in place since the site was
 established, up to ten years. These soils should be used or trials conducted to ensure
 they remain viable.

5.3. Acoustics

The acoustics results are generally within the predictions made in the Environmental Assessment (EA). During the audit period, the Tarrawonga residence was acquired following consistently high noise results. The Tarrawonga residence was the only residence to display consistently elevated noise levels, as was predicted in the EA.

Generally, monitoring of road noise found levels were compliant with the requirements of the EPL and PA (and Noise Management Plan).

TCPL have been utilising real time monitoring to inform operational management of noise at the site. This has assisted with modification of operations to keep the site within the noise limits specified.

There are a few quality assurance issues with the acoustic consultant's report but none that place the results in doubt.

5.4. Air Quality

The air quality comments are from the subsidiary report by Todoroski Air Sciences.

5.4.1. Air Quality Management Plan

The AQMP generally complies with the relevant PA requirements however it does not contain a procedure that can be reasonably followed to determine compliance with 24-hour average impact assessment criteria.

Whilst the AQMP discusses a procedure for determining exceedances that can be applied to evaluate deposited dust results, there is no procedure for determining 24-hr average PM₁₀ compliance levels/ exceedances in the AQMP. The AQMP specifically excludes the TEOM data being used for compliance purposes. This aspect of the AQMP needs to be amended and a reasonable procedure for evaluating compliance with 24-hour average PM₁₀ criteria,

such as the analysis provided for the elevated level on 15 November 2014, needs to be included.

The AQMP also does not consider TSP. It is however reasonable to infer TSP compliance on the basis of PM₁₀ levels, as done at various other mines. It would be prudent to seek that approval for such a calculation procedure be confirmed in writing by DP&E.

As it is understood that a close receptor, Receptor 45 and other nearby receptors are now mine owned, it would appear that there are no privately owned receptors within 3 to 4 km of the mine, and hence the risk of adverse impacts at private receptors is relatively low. However, potential dust effects still need to be managed to minimise impacts at mine owned receptors that are occupied.

The risk profile for dust impacts has changed since the project was approved and the PA conditions developed.

This situation may have some effect on what is considered best practice in regard to dust control. Consideration of best practice includes financial and practical considerations.

5.4.2. Air Quality Management Strategy – Cumulative impacts from all mines

The AQMS is dated March 2014 and contains information generally regarding what actions are proposed to be carried out.

If the AQMS were to be implemented, it may technically meet many of the requirements of the PA in regard to managing cumulative impacts from the mines in the BTM complex. The AQMS outlines a conceptual strategy in regard to "procedures for identifying and apportioning the source/s and contribution/s to cumulative air impacts for both mines and other sources".

Caution should be used in assessing the likely benefit of the conceptual strategy proposed in the AQMS given that it is based on generic, annual average emission factor calculations and may not be as effective as described in controlling day to day effects.

5.4.3. Assessment of Project implementation of best air quality management practice;

Overall, the AQMP lists a range of measures, outlined in Section 3 of the AQMP that can be considered to represent best practice management measures. The aspects considered include consideration of the position of the mine relative to the most potentially affected privately owned receptors which indicate a low risk of significant ongoing dust impacts arising.

Evidence was provided that some of these measures are being implemented, for example the measures include daily dust inspections of the operations that can lead to mine staff deciding to cease certain activities in response to observed emissions of excessive dust from the activities. An example of such inspections was provided for 1 August 2014.

The mine operates a predictive weather forecasting system, and a TEOM monitor (Flixton).

It is noted that the TEOM is not used for compliance purposes.

5.4.4. Measures taken to minimise air quality impacts of the project during extraordinary events when the relevant air quality limits in this approval do not apply.

As outlined above, evidence was provided of daily dust inspections that are triggered under adverse weather conditions. This appears to be in response to EPL pollution reduction program requirements. The Mine has defined adverse weather conditions to be a wind speed above 7 m/s (or approx. 25 km/h).

The audit was not able to identify specific actions under extraordinary conditions and it is assumed that mitigation actions occur regardless of extraordinary events, such as bushfires and dust storms. In many cases extraordinary events may coincide with adverse conditions, but this may not always be the case.

The condition is interpreted to mean that the mine needs to mitigate its impact despite extraordinary events causing high dust or particulate levels, even though there is no requirement to meet criteria levels (as it may not be possible to do so during an extraordinary event).

The mine needs to develop a procedure for identifying extraordinary events, and keeping records which show that all normally applicable mitigation measures relevant to the conditions are applied during such circumstances.

5.4.5. Project compliance with EPL air quality requirements.

The EPL contains the following key requirements:

- Operate a real-time TEOM monitor per sampling and analysis Method AM-22, at the Flixton Property to monitor PM₁₀ on a continuous basis;
- Carry out all operations and activities in a manner that will minimise the emission of dust from the premises;
- Cover all trucks transporting coal from the premises;
- Operate a weather station at Point W1 (Templemore) per the EPA's standard requirements and sampling methods;
- Achieve and maintain a dust control efficiency of more than 80% on coal haul roads by 17 May 2013, per PRP U1 of the EPL;
- Alter or cease the use of equipment on overburden during adverse weather conditions, from 22 March 2013;
- Report on a trial of several measures to reduce dust arising from handling overburden.

The audit found evidence for the above as follows:

- A TEOM is operated at the Flixton Property to measure PM₁₀. Calibration records and monitoring data from January 2013 to December 2014 were provided to confirm this.
- Provided that the measures listed in the AQMP and AQMS are implemented, the operations would be carried out in a manner to minimise dust emissions from the premises. It is noted the measures in the AQMS (which related to cumulate

management of dust from three separate mines) are not presently implemented and it is assumed that this is due to the one of the mines not being operational.

The audit found that overall the mine management systems and documentation includes the majority of ameliorative controls that would need to be taken to manage the dust emissions from each of the activities, individually to an acceptable level.

- Procedures that include covering trucks were found to be in place.
- A weather station is operated at Point W1 (Templemore) and data from January 2013 to December 2014 were provided to confirm this.
- Documents showing that the Three PRP's relating to wheel generated dust, adverse conditions and a trial of mitigation to minimise dust from overburden handling were provided. It is clear that measures to mitigate impacts under adverse weather conditions had been carried out during the period covered by the audit.

5.4.6. Overall impressions – Observations

The audit found that much of the documentation related to management of air quality outlines what will be done, rather than what is being done. The PA is imprecise regarding when some of these actions need to be in place (rather than just existing in a management plan or strategy) thus it cannot be concluded that there is any non-compliance in this respect. However there is an impression that progress in implementing many required actions is slower than would be expected, and that perhaps this is due to delays in other Projects that form part of the cumulative management strategy.

There are also some areas where caution is advised in regard to the anticipated effectiveness of the conceptual cumulative management strategies being proposed. This is due in part to the inherent limitations of the proposed approach and the reduced risk of significant air quality impacts upon private receptors arising due to the acquisition of some of the nearest private properties since the time of the air assessment and development of the PA conditions.

In this context, given the small scale of the operations, the large distances to receptors, and thus small scope for actual impacts to arise, it may be reasonable for the mine to question the benefit to the community of some of the requirements in the PA. However, as some mitigation measures, especially those related to cumulative impacts do not yet appear to be in place, the issue cannot be examined by the audit due to the lack of hard data.

Whilst generally comprehensive and appropriate, a key deficiency in the AQMP is that there does not appear to be a procedure for determining whether compliance with 24-hour average PM₁₀ criteria is being achieved at receptors. This needs to be rectified.

Other technical issues such as reliance on other's HVAS monitors when the monitor is not run on an EPA run day, can be corrected easily but are beyond the absolute control of the mine.

5.5. Surface Water Quality

In the 2012-13 AEMR reporting year, there were two wet weather discharges (rainfall in excess of the site design criteria) and three controlled discharges from the site. On all occasions except one (3/3/13) the site was able to demonstrate that the impacts were

negligible. The 3rd March discharge was a wet weather discharge and is therefore not reportable. There was a single discharge of water from the site in the 2013-14 AEMR reporting year following a high rainfall event. The site collected upstream and downstream samples in Bollol and Nagero Creeks that indicated that site water was lower in turbidity than the receiving waters.

During the site inspection, it was apparent that the surface water runoff from around the coal loader area (described in the Water Management Plan as dirty water) drained to a series of sediment dams aimed at keeping water from the rehabilitated area on the northern emplacement onsite till sediment was removed. These dams drain offsite. A review of the management of surface water in this area is required.

The other key potential surface water issue related to the establishment of the Goonbri Creek diversion. The diversion is required when mining operations enter the alluvials along the creek, this has not yet occurred and is still some time off due to economic constraints associated with mining this area.

5.6. Groundwater Resources

Groundwater inflows to the pit are low as predicted in the EA. As such, the draw down to date has been in line with the EA predictions, that is, low in areas away from the pit and low but measurable in the wells closest to the pit. As yet the alluvials have not been intercepted and Standing Water Level measurements have not indicated any measureable impact on those aquifers in the audit period.

The other key potential ground water issue related to the establishment of the Goonbri Creek diversion and associated groundwater cut-off wall. The diversion is required when mining operations enter the alluvials along the creek, this has not yet occurred and is still some time off due to economic constraints.

Chemical analysis of groundwaters has shown no increase in contaminants in the audit period.

Absolute trigger levels need to be stated in the groundwater TARP to ensure triggers are not missed during routine groundwater level analysis. Currently the TARP limits quoted are dependent on the modelled numbers plus 15%, this changes over time and is difficult to keep track of. An absolute change in standing groundwater level should be developed and quoted in the TARP to simplify assessment of results and annual reporting.

The groundwater model has not been updated in over two years and is overdue for a review in accordance with the Water Management Plan.

6. RECOMMENDATIONS

There are a number of issues noted at the start of this report not addressed below as these have been addressed by TCPL and do not need to be repeated here. The following recommendations have resulted from:

- Observations made by the specialists on the audit team;
- Issues where non-compliance required a recommendation; and
- Those made to improve environmental performance.

Some of these recommendations have not been noted in the document prior to this section as they do not all result from lack of compliance.

6.1. Biodiversity Offsets

As noted in section 5.1, Tables 5.2 and Table 5.3 in the WBOMP should be revised. These tables set out ecological attributes and the scale of measurement in vegetation condition monitoring plots. The attributes & methods outlined in these tables are not consistent with BioBanking methodology for site attributes (OEH 2012).

The baseline surveys for threatened species in offset areas should be conducted in accordance with the department's Survey Guidelines for Australia's Threatened Birds and the Survey Guidelines for Australia's Threatened Bats. The annual monitoring reports should confirm compliance with the two stated methodologies.

Given the sensitivity of the surrounding environment at TCPL, a more formal ground disturbance permitting system would be insurance against excessive clearing, unplanned impacts to flora and fauna or accidental disturbance of archaeological deposits.

Develop and implement a procedure for checking vehicles for weed vectors and cleaning them if necessary prior to entering the biodiversity offset area.

6.1. Bushfire Management

Conduct and document fuel load and fire break inspections annually.

6.2. Heritage

The scar tree preservation procedure in the Heritage Management Plan requires revision to ensure it can be complied with. It is currently too proscriptive and does not allow for flexibility in on ground application.

6.1. Public Safety

Traffic impacts associated with the Goonbri Road intersections and public road impacts were identified. This realignment has not yet occurred and has been postponed due to the sites financial situation. The residual risk to the public should be reassessed to establish if changes are still required to ensure public safety.

6.2. Rehabilitation

Work be conducted to soften the visual impact of the unrehabilitated southern emplacement, reduce risk of impacts to Goonbri Creek and to lessen the levels of fugitive particulate emissions

Site observations indicated that rejects are not placed on the pit floor as required by the MOP but are placed at higher elevations, as part of the backfilling process with suitable cover to prevent spon com. There is nothing unacceptable in this approach provided there is adequate separation for the surface of the final landform of the emplacement. To rectify, TCPL should consider changing the wording of the MOP at the next revision.

The MOP and several other management plans refer to the use of mulching and tree guards to protect tubestock. This is not conducted and does not seem necessary based on the site inspection of rehabilitated areas that showed no excessive grazing of planted tubestock was occurring and the EO noted that there was no history of issues in this area. Reference to the use of these protective methods should be reduced to a contingency should excessive grazing become an issue rather than a requirement.

Seed collection should be conducted in the areas immediately surrounding the mine site to collect seed for rehabilitation.

No rehabilitation works were observed that did not comply with the rehabilitation management plan. However, revegetation quality could be improved, some trials have been conducted, further trials should be implemented to establish the most appropriate vegetation establishment methodologies.

Some topsoil at the site has been stored since the site inception. As soon as there is an opportunity to use this material it should be used. The longer topsoil is stored the less effective it is for vegetation establishment.

Topsoil should be characterised prior to striping to ensure stripping depths are suitable and that similar soil qualities are stored together allowing the application of suitable soil ameliorants when the topsoil is spread.

Topsoils should be characterised prior to spreading to allow the application of suitable ameliorants (predominantly gypsum and lime).

6.3. Water Management

The water in dirty water dams is regularly tested and monitored. When salt levels in these dams exceed acceptable limits; the water will be pumped into PW2 or PW3 and recycled. TCPL does not define "acceptable limits", this should be quantified and included in the Water Management Plan.

Dirty water management needs review in consideration of the water from around the coal loader going into the dirty water system but with a discharge point in this portion of the system. The water from the vicinity of the coal loader should not be able to leave site.

The groundwater model is overdue for an update and review to be based on monitoring to date.

To obtain groundwater samples that are representative of the water within the aquifer being sampled, groundwater wells should be purged (see *Groundwater sampling guidelines*, EPA Victoria 2000).

6.1. Management Plans

As a general comment, the management plans do not include enough information on the background data that was used to formulate them. Future revisions should consider ways to present this information to inform the measures described.

Some of the management plans cross into areas that are not core to the reason for the management plan. They should instead reference other management plans rather than reproducing slabs of information.

In a number of cases, non-compliances have resulted from unnecessary wording in management plans (e.g. water samplers are currently required to be "suitably qualified professionals" when in most instances they are technicians without professional qualifications). TCPL should review the findings of the audit and revise wording to ensure compliance in the future.

The management plans all include requirements for review and it is apparent that these occur. The site however needs to document these reviews in order to demonstrate they have occurred particularly when no changes to the management plan eventuate from the review.

6.1. Reporting

Life of mine monitoring data is reported in the AEMRs but commentary on trends and changes is not adequate to determine whether the site is trending towards compliance or out of compliance with requirements.

Include a description of the effectiveness of waste minimisation actions in the waste section of the AEMR.

APPENDIX A AUDIT TEAM APPROVAL



Development Assessment Systems & Approvals

Mining Projects

Contact: Stephen O'Donoghue

Phone: 0477 345 626

Email: stephen.o'donoghue@planning.nsw.gov.au

Mr Lachlan Johnson Environmental Officer – Tarrawonga Coal Mine Whitehaven Coal Limited PO Box 600 GUNNEDAH NSW 2380

Dear Mr Johnson

Tarrawonga Coal Mine (MP 11_0047) Approval Team for Independent Environmental Audit 2014

I refer to your email of 11 July 2014 seeking approval for SMEC supported by relevant environmental experts to undertake the independent environmental audit (Schedule 5 Condition 10) and biodiversity audit (Schedule 3 Condition 50) of the Tarrawonga Coal Mine in accordance with the project approval. The Acting Secretary has approved the following team to conduct this audit:

- Peter Horn, SMEC Lead Auditor and Rehabilitation
- Catherine Gallery, SMEC Biodiversity
- Glen Homes, SMEC Acoustics
- · Aleks Todoroski, Todoroski Air Sciences Air Quality
- Glen Mounser, SMEC Surface Water
- Yohannes Woldeyohannes, SMEC Groundwater
- Matthew Harland, Joy Duncan Audit Team Members

Should you have any enquiries in relation to the above, please contact Stephen O'Donoghue.

Yours sincerely

Mike Young

Manager

Mining Projects

APPENDIX B CONSULTATION WITH REGULATORY STAKEHOLDERS



74 Hunter Street
Newcastle, NSW 2300, Australia
(PO Box 1346, Newcastle, NSW 2300, Australia)
T +61 2 4925 9600 F +61 2 4925 3888 E newcastle@smec.com
www.smec.com

2 September 2014

Department of Trade and Investment Division of Resources and Energy Environmental Sustainability Unit Maitland NSW 2320

Attention: John Trotter

John,

RE: Tarrawonga Coal - September 2014 Independent Environmental Audit

In accordance with Tarrawonga Coals Project Approval DA 11_0047 for the Tarrawonga Coal Project, an Independent Environmental Audit will be undertaken in mid September 2014. The audit team has been now been endorsed by Department of Planning and Environment (DP&E) and will include experts in biodiversity, surface water, groundwater, rehabilitation, air quality and noise.

The Independent Environmental Audit will assess the environmental performance of Tarrawonga Coal, and its compliance with the requirements of PA 11_0047, Environmental Protection Licence 12365, Mining Leases and management plans. The audit will also involve a review of the adequacy of strategies, plans and programs required under the above mentioned approvals and, where necessary, recommend appropriate measures or actions to improve the environmental performance of the project.

The audit will be comprehensive however, if there are any particular aspects within the items listed above that you would like the audit to take into consideration, please contact the undersigned (02).

Regards,

Peter Horn Lead Auditor

Environmental Manager - Newcastle & Northern NSW

SMEC Australia









From: John Trotter

Sent: Tuesday, 2 September 2014 2:52 PM

To: Horn, Peter

Subject: RE: Tarawonga Audit

Hello Peter,

DRE is currently looking to prepare a standard response to invitations to provide items for consideration during Independent Environmental Audits, however, this has not been completed.

In the meantime, Matthew Newton (Principal Officer, Rehabilitation Standards) has offered the following for your consideration:

Audit Component - Desktop

- Is there a current Mining Operations Plan (MOP) in place and has it been approved by DRE?
- Has the MOP been prepared in consultation with the relevant agencies as outlined in the Project Approval?
- Is the rehabilitation strategy as outlined in the MOP consistent with the Project Approval in terms of progressive rehabilitation schedule; and proposed final land use(s)?
- Has the rehabilitation objectives and completion criteria as outlined in the MOP been developed in accordance with the proposed final land(s) as outlined in the Project Approval?
- Has a rehabilitation monitoring program been developed and implemented to assess performance against the nominated objectives and completion criteria? verified by reviewing monitoring reports and rehabilitation inspection records.
- Has a rehabilitation care and maintenance program been developed and implemented based on the outcomes of monitoring program? verified by reviewing Annual Rehabilitation Programs or similar documentation.

Audit Component - Site Inspection

- Are mining operations being conducted in accordance with the approved MOP (production, mining sequence etc.), including within the designated MOP approval boundary? to be verified by site plans and site inspection.
- Is rehabilitation progress consistent with the approved MOP as verified by site plans and a site inspection? This should include an evaluation against rehabilitation targets and whether the final landform is being developed in accordance with conceptual final landform in Project Approval.
- Based on a visual inspection, are there any rehabilitation areas that appear to have failed or that have incurred an issue that may result in a delay in achieving the successful rehabilitation?

In addition to the above, the audit should note observations where rehabilitation procedures, practices and outcomes represent best industry practice.

John Trotter	
Inspector Environment	
Environmental Sustainability Unit	
Tel:	
Mob:	
Email:	

www.resourcesandenergy.nsw.gov.au

From: Horn, Peter Sent: Tuesday, 2 September 2014 12:07 PM To: john.trotter Subject: Tarrawonga Audit Hi John, Letter attached re the impending audit at Tarrawonga for DP&E Compliance Team. Please respond with any comment or areas of focus. If you are not the correct person in DRE, please let me know who I should be speaking with. Cheers Peter Peter Horn | Environmental Manager - Newcastle & Northern NSW **SMEC Australia** 74 Hunter Street, Newcastle, NSW, 2300, Australia (PO Box 1346, Newcastle, NSW, 2300, Australia) T +61 2 | F +61 2 | M +61 | www.smec.com | LinkedIn **SMEC SNOWY MOUNTAINS ENGINEERING CORPORATION** Local People. Global Experience. Disclaimer: The information contained in this e-mail and any attached file is confidential. It is intended solely for the addressee, and may not be used, reproduced, disclosed or distributed without SMEC's permission. SMEC accepts no liability for loss or damage (whether caused by negligence or not) resulting from the use of any attached files. Please consider the environment before printing this email. This e-mail has been scanned for viruses by Symantec.Cloud. This e-mail has been scanned for viruses by Symantec.Cloud.

This message is intended for the addressee named and may contain confidential information. If you are not the intended recipient, please delete it and notify the sender. Views expressed in this message are those of the individual sender, and are not necessarily the views of their organisation.



74 Hunter Street
Newcastle, NSW 2300, Australia
(PO Box 1346, Newcastle, NSW 2300, Australia)
T+61 2 4925 9600 F+61 2 4925 3888 E newcastle@smec.com
www.smec.com

22 October 2014

Department of the Environment Environment Assessment and Compliance Division Post Approvals

Attention: Dr Dylan Horne

Dylan,

RE: Tarrawonga Coal - September 2014 Independent Environmental Audit

In accordance with Tarrawonga Coals Project Approval DA 11_0047 for the Tarrawonga Coal Project, an Independent Environmental Audit was be undertaken in mid September 2014. The audit team was endorsed by Department of Planning and Environment (DP&E) and included experts in biodiversity, surface water, groundwater, rehabilitation, air quality and noise.

The Independent Environmental Audit assessed the environmental performance of Tarrawonga Coal, and its compliance with the requirements of PA 11_0047, Environmental Protection Licence 12365, Mining Leases and management plans. The audit also involved a review of the adequacy of strategies, plans and programs required under the above mentioned approvals and, where necessary, will recommend appropriate measures or actions to improve the environmental performance of the project.

The audit will be comprehensive however, if there are any particular aspects within the items listed above that you would like the audit to take into consideration, please contact the undersigned so we can consider your thoughts in our analysis of the data and audit focus. Whilst the site aspect of the audit is complete, the audit documentation continues so it is not too late for your input into the process.

Regards,

Peter Horn Lead Auditor

Environmental Manager - Newcastle & Northern NSW

SMEC Australia









Horn, Peter

From: Horne, Dylan

Sent: Thursday, 23 October 2014 8:02 AM

To: Horn, Peter

Cc: Taylor, Alex; Samarakoon, Manel

Subject: RE: Tarawonga Compliance Audit [SEC=UNCLASSIFIED]

Hi Peter

Thanks for the email. I was the contact for this earlier in the year but I've since moved to a different role in the department.

In the first instance please contact Manel Samarakoon (cc'd). I forwarded your other email with the attached letter.

Kind regards Dylan

Dr Dylan Horne

Senior Regulatory Advisor

Regulatory Capability and Performance | Policy Analysis and Implementation Division

Department of the Environment GPO Box 787 CANBERRA ACT 2600

t: (02) e:

From: Horn, Peter

Sent: Wednesday, 22 October 2014 4:22 PM

To: Horne, Dylan

Subject: Tarrawonga Compliance Audit

Hi Dylan,

Your contact details were provided to me by Tarrawonga as the DoE contact for the site.

There is a letter attached re an audit of Tarrawonga Coal for the NSW DP&E Compliance Team.

Please respond with any comment or areas of focus.

If you are not the correct contact person in DoE for the site and this sort of request, please let me know who I should be speaking with.

Kind Regards

Peter

Peter Horn | Environmental Manager - Newcastle & Northern NSW

SMEC - Australia & New Zealand Division

74 Hunter Street, Newcastle, NSW, 2300, Australia (PO Box 1346, Newcastle, NSW, 2300, Australia) T +61 2 | F +61 2 4 | M +61

| www.smec.com | LinkedIn

SMEC SNOWY MOUNTAINS ENGINEERING CORPORATION

Local People. Global Experience.

Disclaimer: The information contained in this e-mail and any attached file is confidential. It is intended solely for the addressee, and may not be used, reproduced, disclosed or distributed without SMEC's permission. SMEC accepts no liability for loss or damage (whether caused by negligence or not) resulting from the use of any attached files.



74 Hunter Street
Newcastle, NSW 2300, Australia
(PO Box 1346, Newcastle, NSW 2300, Australia)
T+61 2 4925 9600 F+61 2 4925 3888 E newcastle@smec.com
www.smec.com

2 September 2014

Tarrawonga Coal Community Consultation Committee

Attention: John Turner

John,

RE: Tarrawonga Coal - September 2014 Independent Environmental Audit

In accordance with Tarrawonga Coals Project Approval DA 11_0047 for the Tarrawonga Coal Project, an Independent Environmental Audit will be undertaken in mid September 2014. The audit team has been now been endorsed by Department of Planning and Environment (DP&E) and will include experts in biodiversity, surface water, groundwater, rehabilitation, air quality and noise.

The Independent Environmental Audit will assess the environmental performance of Tarrawonga Coal, and its compliance with the requirements of PA 11_0047, Environmental Protection Licence 12365, Mining Leases and management plans. The audit will also involve a review of the adequacy of strategies, plans and programs required under the above mentioned approvals and, where necessary, recommend appropriate measures or actions to improve the environmental performance of the project.

The audit will be comprehensive however, if there are any particular aspects within the items listed above that you would like the audit to take into consideration, please contact the undersigned (02).

Regards,

Peter Horn Lead Auditor

Environmental Manager - Newcastle & Northern NSW

SMEC Australia









From: John Turner

Sent: Tuesday, 9 September 2014 5:06 PM

To: Horn, Peter Cc: Jill Johnson

Subject: Re: Tarrawonga Independent Environmental Audit

Peter

I have not received any comments or requests in relation to this matter from CCC members. Regards

John Turner

Sent from my iPhone

On 02/09/2014, at 2:45 PM, "Horn, Peter" wrote:

Thanks very much John, I will still be onsite after the 9th and it would be good to get your comments then (finish up on-site on the 11th) in order to take them into consideration during the site inspection(as best I can in my limited time on-site).

Kind Regards

Peter

Peter Horn | Environmental Manager - Newcastle & Northern NSW

SMEC Australia T +61 2 | M +61

From: John Turner _

Sent: Tuesday, 2 September 2014 2:40 PM

To: Horn, Peter **Cc:** Jill Johnson

Subject: RE: Tarrawonga Independent Environmental Audit

Dear Peter,

Thank you for your letter.

As the committee has not previously discussed the proposed audit, I have sent an email to the committee asking if they have any specific comment in relation to the specifications of the audit they would like to make.

I have asked for feed back by 9/9/14.

I will be in touch with you after that date.

Regards

John Turner

Chair, Tarrawonga CCC

From: To: Subject: Tarrawonga Independent Environmental Audit Date: Tue, 2 Sep 2014 02:35:36 +0000
Hello John,
I'm writing to you to notify you and the CCC of an Independent Environmental Audit of Tarrawonga Coal. The audit is required as a condition of their new Project Approval. Attached is a letter seeking comment and input into the audit focus. Apologies for the short time frame.
Kind Regards Peter Lead Auditor
Peter Horn Environmental Manager - Newcastle & Northern NSW SMEC Australia 74 Hunter Street, Newcastle, NSW, 2300, Australia (PO Box 1346, Newcastle, NSW, 2300, Australia) T +61 2 F +61 2 49 M +61 Pe www.smec.com LinkedIn
SMEC SNOWY MOUNTAINS ENGINEERING CORPORATION Local People. Global Experience.
Disclaimer: The information contained in this e-mail and any attached file is confidential. It is intended solely for the addressee, and may not be used, reproduced, disclosed or distributed without SMEC's permission. SMEC accepts no liability for loss or damage (whether caused by negligence or not) resulting from the use of any attached files.
Please consider the environment before printing this email.
This e-mail has been scanned for viruses by Symantec.Cloud.
This e-mail has been scanned for viruses by Symantec.Cloud.
This e-mail has been scanned for viruses by Symantec.Cloud.



74 Hunter Street
Newcastle, NSW 2300, Australia
(PO Box 1346, Newcastle, NSW 2300, Australia)
T+61 2 4925 9600 F+61 2 4925 3888 E newcastle@smec.com
www.smec.com

2 September 2014

Environment Protection Authority Armidale Office

Attention: Kharl Turnbull

Kharl,

RE: Tarrawonga Coal - September 2014 Independent Environmental Audit

In accordance with Tarrawonga Coals Project Approval DA 11_0047 for the Tarrawonga Coal Project, an Independent Environmental Audit will be undertaken in mid September 2014. The audit team has been now been endorsed by Department of Planning and Environment (DP&E) and will include experts in biodiversity, surface water, groundwater, rehabilitation, air quality and noise.

The Independent Environmental Audit will assess the environmental performance of Tarrawonga Coal, and its compliance with the requirements of PA 11_0047, Environmental Protection Licence 12365, Mining Leases and management plans. The audit will also involve a review of the adequacy of strategies, plans and programs required under the above mentioned approvals and, where necessary, recommend appropriate measures or actions to improve the environmental performance of the project.

The audit will be comprehensive however, if there are any particular aspects within the items listed above that you would like the audit to take into consideration, please contact the undersigned (02).

Regards,

Peter Horn Lead Auditor

Environmental Manager - Newcastle & Northern NSW

SMEC Australia











74 Hunter Street
Newcastle, NSW 2300, Australia
(PO Box 1346, Newcastle, NSW 2300, Australia)
T+61 2 4925 9600 F+61 2 4925 3888 E newcastle@smec.com
www.smec.com

4 September 2014

Office of Environment and Heritage Dubbo Office

Attention: Phil Purcell

Phil,

RE: Tarrawonga Coal - September 2014 Independent Environmental Audit

In accordance with Tarrawonga Coals Project Approval DA 11_0047 for the Tarrawonga Coal Project, an Independent Environmental Audit will be undertaken in mid September 2014. The audit team has been now been endorsed by Department of Planning and Environment (DP&E) and will include experts in biodiversity, surface water, groundwater, rehabilitation, air quality and noise.

The Independent Environmental Audit will assess the environmental performance of Tarrawonga Coal, and its compliance with the requirements of PA 11_0047, Environmental Protection Licence 12365, Mining Leases and management plans. The audit will also involve a review of the adequacy of strategies, plans and programs required under the above mentioned approvals and, where necessary, recommend appropriate measures or actions to improve the environmental performance of the project.

The audit will be comprehensive however, if there are any particular aspects within the items listed above that you would like the audit to take into consideration, please contact the undersigned (02) or by e-mail

Regards,

Peter Horn Lead Auditor

Environmental Manager - Newcastle & Northern NSW

SMEC Australia









APPENDIX C AUDIT PROTOCOL





Consequences

Level Descriptor Consequences

А	Catastrophic	Long term environmental damage (5 years or longer), requiring \$5million to correct or in penalties
В	Major	Medium-term (1-5 years) environmental damage, requiring \$1 to 5 smillion to correct or in penalties
С	Moderate	Short-term (less than 1 year) environmental damage, requiring up to \$1million to correct or in penalties
D	Minor	Environmental damage, requiring up to \$200,000 to correct
E	Insignificant	Negligible environmental impact, managed within operating budgets

Likelihood

 $\label{eq:Level Descriptor} \text{Likelihood of the risk arising and leading to the assessed level of } \\ \text{consequence}$

1	Almost certain	Is expected to occur in most circumstances and has a history of occurrence	Once a year or more frequent
2	Likely	Will probably occur in most circumstances	Once in 1 to 3 years
3	Possible	Could occur at some time	Once in 3 to 10 years
4	Unlikely	Not likely to occur in normal circumstances	Once in 10 to 50 years
5	Rare	May occur only in exceptional circumstances	Once in 100 years or more

		Catastrophic	Major	Moderate	Minor	Insignificant
		Α	В	С	D	E
Almost certain	1	Extreme	Extreme	High	High	Medium
Likely	2	Extreme	Extreme	High	Medium	Low
Possible	3	Extreme	High	Medium	Medium	Low
Unlikely	4	High	Medium	Medium	Low	Low
Rare	5	High	Medium	Low	Low	Low





		Requirement				Risk	
Reference	Condition	nequirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	Schedule 2 - Administ	rative Conditions			Consequence		
	OBLIGATION TO MINI	MISE HARM TO THE ENVIRONMENT			1		
Project Approval 11-0047	1	In addition to meeting the specific performance criteria established under this approval, the Proponent shall implement all reasonable and feasible measures to prevent and/or minimise any material harm to					
	_	the environment that may result from the construction, operation, or rehabilitation of the project.					
	OBLIGATION TO MINI	MISE HARM TO THE ENVIRONMENT			1		
			Generally In compliance though exceptions are noted in various places throughout the audit findings				
		b) statement of commitments; and					
		c) conditions of this approval					
Project Approval 11-0047	2	Notes:		Compliant			
		The E19 layout of the project is shown in Appendix E17; and					
		The statement of commitments is reproduced in Appendix 4					
		If there's is any inconsistency between the above documents, the most recent document shall prevail to	Noted				
Project Approval 11-0047	3	the extent of the inconsistency. However, the conditions of this approval shall prevail to the extent of					
		any inconsistency. The Proponent shall comply with any reasonable requirement/s of the Director-General arising from the	Not in the audit period				
		Department's assessment of:					
Project Approval 11-0047	4	a) any reports, strategies, plans, programs, reviews, audits or correspondence that are submitted in		Not Applicable			
, ,,		accordance with this approval; and b) the implementation of any actions or measures contained in these documents					
	LIMITS ON APPROVAL	<u> </u>					
	Mining Operations	The Proponent may carry out mining operations on the site until the end of December 2030.	15 years approved mining remaining			J	
			0 1 0				
		Note: Under this approval, the Proponent is required to rehabilitate the site and carry out additional					
Project Approval 11-0047	5	undertakings to the satisfaction of both the Director-General and the Executive Director Mineral Resources. Consequently, this approval will continue to apply in all other respects - other than the right		Not Applicable			
		to conduct mining operations - until the rehabilitation of the site and these additional undertakings have					
		been carried out satisfactorily.					
	Coal Extraction			-			
		For the period until up to 3 months after the commissioning of the Boggabri Rail Spur Line and Boggabri	The rail spur is not commissioned yet.				
		CHPP, the Proponent may transport up to: a) 2 million tonnes of ROM coal from the site to the Whitehaven CHPP along the dedicated haulage					
		route in any calendar year;					
Project Approval 11-0047	7	b) and 150,000 tonnes of this ROM coal from the site in any calendar year for direct distribution to		Not Applicable			
,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,,		domestic markets via the dedicated haulage route to the Kamilaroi Highway.					
		Note: For the avoidance of doubt, the total amount of coal permitted to be transported from the site by					
		road in any calendar year is 2 million tonnes.					
		During this period, the Proponent shall only transport coal from the site or receive coal reject from the	Included in induction, controlled by Tarrawonga loader operator, trucks				
			GPS tracked. No observed truck transport outside hours in site				
Project Approval 11-0047	8		inspection.	Compliant			
		b) 7 am to 5.15 pm Saturday; and c) at no time on Sundays or public holidays					
				<u> </u>			
	Coal Transport - After	Commissioning of the Boggabri Rail Spur Line and Boggabri CHPP 9 For the period commencing 3 months after the commissioning of the Boggabri Rail Spur Line &	The rail spur is not commissioned yet.		1	ı	
		Boggabri CHPP, the Proponent:	The Fall Spain is not commissioned yet.				
		a) shall not transport more than 3 million tonnes of this ROM coal from the site in any calendar year for		1			
Project Approval 11-0047	9	direct distribution to domestic markets via the dedicated haulage route to the Kamilaroi Highway; and c) shall transport all other coal from the site via the Boggabri rail spur line.		Not Applicable			
		-, surprise on other countries are size to the object of the sput file.					
		District this could be December that the beautiful to the country of the country	The self-annial and annial and annial and and				
		During this period, the Proponent shall only transport coal from the site by truck (excluding coal transport to the Boggabri coal mine for subsequent despatch via the Boggabri rail spur line) between 7	The rail spur is not commissioned yet.				
Project Approval 11-0047	10	am and 6 pm Monday to Saturday.		Not Applicable			
	Gravel Extraction and	Transport					
			Compliant, n very little gravel extraction in the audit period. (608 Tonnes				
Project Approval 11-0047	11	any calendar year.	in 2013, none 2012)	Compliant			
Project Approval 11-0047	12		Compliant, n very little gravel extraction in the audit period. (608 Tonnes in 2013, none 2012)	Compliant			
rioject Approvai 11-0047	12	year. This gravel is to be transported from the site to the Kamilaroi Highway via the dedicated haulage route	2013, none 2012)	Compilant			
Project Approval 11-0047	13	The proponent shall only transport gravel from the site by truck between 7 am and 6 pm Monday to	Compliant, reviewed eslewhere, trucking is tightly controlled by site.	Compliant			
		Saturday.		Compliant	1		
Project Approval 11-0047	CURRENDER OF SUCE	Saturday. NG DEVELOPMENT CONSENT					





Deference	Constitute	Requirement	E. Maria	A coults at court		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	14	By the end of December 2013, or as otherwise agreed by the Director-General, the Proponent shall surrender the existing development consent (DA-88-4-2005) for the Tarraway Coal Mine in accordance with Section 104A of the EP&A Act. Prior to the surrender of this development consent, the conditions of this approval shall prevail to the extent of any inconsistency with the conditions of the development consent.	Not done work is underway to surrender the DA.	Not Compliant Administrative			
	STRUCTURAL ADEQUA						
Project Approval 11-0047	15	The Proponent shall ensure that all new building and structures, and any alterations or additions to existing building and structures, are constructed in accordance with the relevant requirements of the BCA. Notes: Under Part 4Aof the EP&A Act, the Proponent is required to obtain construction and occupation certificates for the proposed building works; and Part 8 of the EP&A Regulation sets out the requirements for the certification of the project.	Two new buildings on-site, shipping containers with awning. No CC or OCs, buildings are borderline though and could be considered temporary as the infrastructure area is to be moved and the building fits the description of Temporary Buildings in the SEPP (exempt and complying development)	Compliant			
	DEMOLITION						
Project Approval 11-0047	16	The Proponent shall ensure that all demolition work on site is carried out in accordance with Australian Standard AS 2601-2001: The Demolition of Structures, or its latest version.	No demolition work	Compliant			
	PROTECTION OF PUB			1			
Project Approval 11-0047	17	Unless the Proponent and the applicable authority agree otherwise, the Proponent shall: a) repair, or pay the full costs associated with repairing, any public infrastructure that is damaged by the project; and b) relocate, or pay the full costs associated with relocation, any public infrastructure that needs to be relocated as a result of the project.	None noted	Not Applicable			
	OPERATION OF PLAN	T AND EQUIPMENT					
Project Approval 11-0047	18	18 The Proponent shall ensure that all the plant and equipment used on site, or to transport coal from the site, is: a) maintained in a proper and efficient condition; and b) operated in a proper and efficient manner.	Observed maintenance planning system on site, discussed with maintenance planner at interview. No loud or smoky equipment observed on site inspection.	Compliant			
	STAGED SUBMISSION	OF STRATEGIES, PLANS AND PROGRAMS					
Project Approval 11-0047	19	With the approval of the Director-General, the Proponent may submit any strategy, plan or program requirement by this approval on a progressive basis. Notes: While any strategy, plan or program may be submitted on a progressive basis, the Proponent will need to ensure that the existing operations on site are covered by suitable strategies, plans or programs at all times. If the submission of any strategy, plan or program is to be staged, then the relevant strategy, plan or program must clearly describe the specific stage to which the strategy, plan or program applies, the relationship of this stage to any future stage, and the trigger for updating the strategy, plan or program.	Some plans submitted in this manner	Noted			
Project Approval 11-0047	20	Until they are replaced by an equivalent strategy, plan or program approved under this approval, the Proponent shall implement the existing strategies, plans or programs for the site that have been approved under DA-88-4-2005.	Noted and observed on -site where new commitments that provided environmental improvement were made in plans that were not approved, they had been implemented.	Compliant			
	COMMUNITY ENHAN		T	1			
Project Approval 11-0047	21	21 By the end of March 2013, unless otherwise agreed by the Director-General, the Proponent shall enter into a planning agreement with Council in accordance with: a) Division 6 of Part 4 of the EP&A Act; and b) the terms of the Proponent's offer provided in Appendix 3.	No evidence provided	Not Compliant	E	1	Medium
		nental Performance Conditions					
	ACQUISITION ON REC	QUEST					





Defenses	Can division	Requirement	5.11	A coding to disco		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	1	Upon receiving a written request for acquisition from an owner of the privately-owned land listed in Table 1, the Proponent shall acquire the land in accordance with the procedures in conditions 8 and 0 of schedule 4. Table1: Land subject to acquisition upon request Notes: To interpret the locations referred to in Table 1 see the applicable figure(s) in Appendix 5. Properties 43, 44 and 45 also have acquisition rights under the approval for the Boggabri coal mine, and/or the existing consent (DA 88-4-2005) for the Tarrawonga mine. The proponent may acquire these properties on an equitable basis with the owner of the Boggabri mine. For the purposes of acquisition under this condition, parcels of land that are in close proximity and operated as a single agricultural enterprise should be included as part of the land to be acquired. Where the Proponent and the owner(s) cannot agree on whether non-contiguous parcels of fand should be included, either party may refer the matter to the Director-General for resolution. The Director-General's decision as to the lands to be included for acquisition under the procedures in conditions 8 and 9 of Schedule 4 shall be final.	No formal written requests but two acquisitions have occurred in the audit period.	Not Applicable			
Project Approval 11-0047	1	Time East America American quant research August Descript	Noted				
	ADDITIONAL NOISE A	L ND/OR AIR QUALITY MITIGATION ON REQUEST					
Project Approval 11-0047	2	Upon receiving a written request from the owner of any residence on the land listed in Table 1, the Proponent shall implement additional noise and/or quality mitigation measures (such as double glazing, insulation, air filters, a first flush roof water drainage system and/or air conditioning) at the residence in consultation with the owner. These measures must be reasonable and feasible and directed towards reducing the noise and/pr air quality impacts of the project on the residence. If within 3 months of receiving this request from the owner, the Proponent and the owner cannot agree on the measures to be implemented, or there is a dispute about the implementation of these measures, then either party may refer the matter to the Director-General for resolution.	No written requests in the audit period	Not Applicable			
	NOISE AND VIBRATIO	N					
Project Approval 11-0047	Noise Criteria 3	generated by the project does not exceed the criteria in Table 2 at any residence on privately-owned land. Table 2: Noise criteria dB(A) Land Table 3: Noise criteria dB(A) Land Table 4: Noise criteria dB(B) Land Table 5: Noise criteria dB(B) Land Table 6: Noise Land Land	Noise levels at Tarrawonga in excess of criteria but Tarrawonga is listed as noise affected in the PA and was purchased by TCL in the audit period.	Compliant			
	Noise Acquisition Req	uirements - Residences	This has not occurred in the audit period				
Project Approval 11-0047	4	If the owner(s) of a privately-owned residence, that is not listed in Table 1, have reason to believe that operational noise from the project is causing the criteria in Table 2 to be exceeded at the residence, that owner(s) can request an independent noise impact assessment for the residence. The request shall be made in writing to the Director-General. If the Director-General considers that a noise impact assessment is warranted, then the Proponent shall commission the assessment.	This has not occurred in the audit period	Not Applicable			





Deference	Condition	Requirement	Fridamas	A dia Fin din a		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	4	If the noise impact assessment determines that the nouse generated by the project causes sustained exceedancies, or is likely to cause sustained exceedances, of the criteria in Table 2, the owner(s) can make a written request to the Proponent for one of the following: a) mitigation (such as double glazing, insulation and air conditioning) at the residence in consultation with the owner(s). These measures must be reasonable and feasible and directed towards reducing the noise impacts of the project on the residence. If within 3 months of receiving this request from the owner(s), the Proponet and owner(s) cannot agree on the measures to be implemented, or there is a dispute about the implementation of these measures, then either party may refer the matter to the Director-General for resolution; or b) acquisition of the residence and land in accordance with the procedures in conditions 8 and 9 of Schedule 4.	This has not occurred in the audit period	Not Applicable			
Project Approval 11-0047	4	Upon receiving a written request form the owner(s), the Proponent must undertake whichever option has been requested by the owner(s).	This has not occurred in the audit period	Not Applicable			
Project Approval 11-0047	4	However, this condition does not apply if the Proponent has an agreement with the owner(s) of the relevant residence to generate higher noise levels, and the Proponent has advised the Department in writing of the terms of this agreement.	Noted				
	Noise Acquisition Req						
		If the owner(s) of land containing a privately owned residence, which is not listed in Table 1, have reason to believe that operational noise from the project is causing noise levels to exceed 40 dB(A) LAeq(15 min) over more than 25% of that land, the owner(s) can request an independent noise impact assessment for the land. The request shall be made in writing to the Director-General. If the Director-General considers tha a noise impact assessment is warranted, then the Proponent shall commission the assessment.	This has not occurred in the audit period				
Project Approval 11-0047	5	If the noise impact assessment determines that the noise generated by the project causes sustained exceedances, or is likely to cause sustained exceedances, of the 40 dBA criteria, the owner(s) can make a witten request to the Proponent for acquisition of the residence and land in accordance with the provedures in conditions 8 and 9 of Schedule 4. Upon receibing a written request from the owner(s), the Proponent must purchase the residence and land in accordance with the procedures in conditions 8 and 9 of Schedule 4.		Not Applicable			
Project Approval 11-0047	5	relevant residnece to generate higher noise levels, and the Proponent has an agreement with the owner(s) of the relevant residence to genrrate higher noise levels, and the Proponent has advised the Department in writing of the terms of this agreement. Note: The notes to condition 4 of this Schedule also apply to this condtion.	Noted				
	Cumulative Noise Crit						
Project Approval 11-0047	6	Except for the land listed in Table 1, the Proponent shall ensure that the operational noise generated by the project combined with the noise generated by other mines does not exceed the criteria in Table 3 at any residence on privately-owned land. Selfs 3 Considers rose critera 80(4)	in discussion with DP&E re implementation. Not yet implemented.	Not Applicable			
	Cumulative Noise Acc	uisition Requirements	White has a second to also said and the			1	
Project Approval 11-0047	7	It the owner(s) of a privately-ownedresidence, which is not listed in Table 1, reasonably believes that the noise limits in Table 3 are being exceeded at the residence and that the exceedance is caused by operational noise from the project and one of more other mines (including use of private roads or rail spurs), the owner(s) can request an independent noise impact assessment for the residence. The request shall be made in writing to the Director-General. If the Director-General considers that a noise impact assessment is warranted, then the Proponent shall commission the assessment.	Inis has not occurred in the audit period	Not Applicable			





Defenence	Condition	dition	Eddama	Audit Finding		Risk	
Reference	Condition		Evidence		Consequence	Likelihood	Risk
Project Approval 11-0047	7	Where the noise impact assessment determines that the cumulative noise generated by the project combined with the noise from the other mine(s) cause, or is likely to cause, sustained exceedances of the criteriea in Table 3, then the owner(s) can make a written request to the Proponent for one of the following: a) mitigation (such as double glazing, insulation and air conditioning) at the residence in consultation with the owner(s). These measures must be reasonable and feasible and directed towards reducing the noise impacts of the project of the redience. If within 3 month of receiving this request from the owner(s), the Proponent and owner(s) cannot agree on the measures to be omplemented, or there is a dispute about the implementation of these measures, then either party may reger the matter to the Director-General for resolution; or b) acquisition of the residence and land in accordance with the procedures in conditions 8 and 9 of Schedule 4.	This has not occurred in the audit period	Not Applicable			
Project Approval 11-0047	7	Upon receiving a written request from the owner(s), the Proponent must undertake whichever option has been requested by the owner(s).	This has not occurred in the audit period	Not Applicable			
Project Approval 11-0047	7	However, this condition does not apply if the Proponent has an agreement with the owner(s) of the relevant residence to generate higher noise levels, and the Proponent has advised the Department in writing of the terms of this agreement.	Noted				
Project Approval 11-0047	7	The Proponent may seek to recover an equitable share of the costs incurred from the other mines contributing to the cumulative impact. Unless otherwise agree between the mines, the proportional contributions should be based on expert analysis of the monitoring results to asssess relative contribution to the impact. In the event of a dispute between the mines the Proponent, or one of the contributing mines, may submit the matter to the Director-General for resolution. The Director-General's decision shall be final.	Noted				
Project Approval 11-0047	7	Notes: 1 The notes to condition 4 of this Schedule also apply to this condition. 2 The noise impact assessment shall include assessment of the relative contribution of the mines to the impact at the residence.	Noted				
	Road Traffic Noise Cri	teria					
Project Approval 11-0047	8	The Proponent shall ensure that the noise generated by the project on public roads does not exceed the criteria in Table 4 at any existing residence on privately-owned land. Table 4. Road trafic rose criteria dRAI Land All privately-owned residences 60 60 55 However, these criteria do not apply if the Proponent has a written agreement with the relevant landowner to exceed the criteria, and the Proponent has advised the Department in writing of the terms of this agreement.	Spectrum reports indicate no exceedance of this criteria	Compliant			
	Attenuation of Plant	I		T.			
Project Approval 11-0047	9	The Proponent shall: a) ensure that: all trucks, dozers, drills and excavators purchased for used on the site after the date of this approval are commisioned as noise suppressed (or attenuated) units; - All equipment and noise control measures deliver sound power levels that are equal to, or less than, the sound power levels identified in the EA, and correspond to best practive, or the application of best available economically achievable technology; - Improvements are made to existing noise suppression equipment as improved technology becomes available where reasonable and feasible; and b) monitor and report on the implementation of these requirements annually on its website.	No noise attenduated equipment. No new equipment purchased since the approval was granted.	Compliant			
Project Approval 11-0047	10	The Proponent shall: a) conduct an annual testing program of the plant on site; b) restore the effectiveness of any attenuation if it is found to be defective; and c) reprt on the results of any testing and/or attenuation work within the Annual Review.	Spectrum reports sighted for SPL testing of equipment	Compliant			
	Operation Conditions					*	





Reference Condition The Programment was in the second part of the time and second part of the second par	Defen	Condition	Requirement	Full de cons	A codia Finali		Risk	
In general teasurement per circle to material and interest to the contracting of the cont	Keterence	Condition			Audit Finding	Consequence	Likelihood	Risk
Project Approval 11:0017 11 start forcat Mining Procinct, to monitor the cumulative noise impacts of these minies, to the standardiscut of the Director-General Landardiscut on the Director-G	Project Approval 11-0047	11	a) implement best management practice to minimise all operational, low frequency, road and rail traffic noise levels associated with the project; b) operate a comprehensive on-site noise management system that uses a combination of predictive meteorological forecasting and real-time noise monitoring data to guide the day to day planning of mining operations and the implementation of both proactive and reactive noise mitigation measures to ensure compliance with the relevant conditions of this approval; c) maintain the effectiveness of noise suppression equipment on plant at all times and ensure defective plant is not operationally used until fully repaired; d) ensure that noise attenuated plant is deployed preferentially in locations near to sensitivie receivers; e) minimise the noise impacts of the project during meteorogical conditions under which the noise limits in this approval do not apply (see condition 13); f) ensure that project related trains on the Boggabri spur line only use locomotives that are approved to operate on the NSW rail network in accordance with the noise limits in ARTC's EPI (No. 3142); g) use its best endeavours to ensure that project-related of ling stock supplied by service providers on	frequency noise complaints. Since there has been no change in equipment, operational changes to minimise noise impacts appear to have been successful. B) site uses directional real time noise monitoring combined with supplied predictive meteorological data to manage the issue c) reviewed maintenance system, no noisy equipment identified on-site. d) not applicable, no new equipment e) confirmed during interview f) rail spur not constructed at the time of the audit	Compliant			
12. The Proposent shall prepare and implement a Nove Management Starter to the project to the statistation of the Intervised General. This plan must is statistation of the Intervised General for approval by Project Approval 11.0047 12 21 22 23 23 24 24 24 24 24	Project Approval 11-0047	11	Leard Forest Mining Precinct, to minimise the cumulative noise impacts of these mines, to the		Not Applicable			
12. The Proposent shall prepare and implement a Nove Management Starter to the project to the statistation of the Intervised General. This plan must is statistation of the Intervised General for approval by Project Approval 11.0047 12 21 22 23 23 24 24 24 24 24		Noise Management P	an					
d) include a risk/response matrix to codify operational responses to varying levels of risk resulting from weather conditions and specific imining activities; e) include commitments to provide summary reports and specific briefings at CCC meetings on issues airling from noise monitoring; f) include a monitoring program that; uses attended monitoring to evaluate the performance of the project, including a minimum of four days attended monitoring to evaluate the performance of the project, including a minimum of four days attended monitoring to evaluate the performance of the project, including a minimum of four days attended monitoring to evaluate the performance of the project, including a minimum of four days attended monitoring to evaluate the performance of the project including a minimum of four days attended monitoring to evaluate the performance of the project including a minimum of four days attended monitoring to evaluate the monitoring to upport the proactive and reactive noise management system on site; provides for the annual validation of appropriate sampling rate to determine compliance with noise in the committee of the minimal validation of the noise model for the project (including the tenth percentile methodology); and g) includes a Leard Forest Mining Precinct Noise Management Strategy, that has been prepared in consultation with other coal mines in the Precinct, to minimals the cumulative noise impacts of all mines within the Precinct, and includes: a description of the measures that would be implemented to ensure that the noise management of the mines is properly co-ordinated to ensure continuous of the same and the sources, using the noise and meterorological monitoring network for the precinct; protocols for data sharing; and other the operation minimum events of the dependance minimum events of the dependance minimum events of the dependance minimum events of the same and there sources, using the noise and meterorological monitoring network of the saudit. Project Approval 11-0047 12 2	Project Approval 11-0047	12	satisfaction of the Director-General. This plan must: a) be prepared in consultation with the EPA, and be submitted to the Director-General for approval by the end of May 2013; b) describe the measures that would be implemented to ensure: best management practice is being employed; the noise impacts of the project are minimised during meteorological conditions under which the noise limits in the approval do not apply; and	prepared in consultation with the EPA.	Compliant			
Project Approval 11-0047 12 consultation with other coal mines in the Precinct, to minimise the cumulative noise impacts of all mines within the Precinct, and includes: a description of the measures that would be implemented to ensure that the noise management of the mines is properly co-ordinated to ensure comiliance with the relevant noise criteria; a suitable monitoring network for the precinct; protocols for data sharing; and procedures for idenfitying and apportioning the sources, using the noise and meterorological monitoring network and appropriate investigative tools. Note: The Leard Forest Mining Precinct Noise Management Stategy can be developed in stages and will need to be subject to ongoing review dependent upon the determination and commencement of other mining projects in the area. Note Applicable Not Applicable	Project Approval 11-0047	12	d) include a risk/response matrix to codify operational responses to varying levels of risk resulting from weather conditions and specific mining activities; e) include commitments to provide summary reports and specific briefings at CCC meetings on issues arising from noise monitoring; f) include a monitoring program that; uses attended monitoring to evaluate the performance of the project, including a minimum of four days attended monitoring per quarter at locations agree to by the Director-General, or more regularly where required; uses real-time monitoring to support the proactive and reactive noise management system on site; includes monitoring of inverstion strength at an appropriate sampling rate to determine compliance with noise limits; evaluates and reports on the effectiveness of the noise management system on site; provides for the annual validation of the noise model for the project (including the tenth percentile methodology); and		Compliant			
Project Approval 11-0047 12 need to be subject to ongoing review dependent upon the determination and commencement of other mining projects in the area. Not Applicable	Project Approval 11-0047	12	consultation with other coal mines in the Precinct, to minimise the cumulative noise impacts of all mines of within the Precinct, and includes: a description of the measures that would be implemented to ensure that the noise managemenet of the mines is properly co-ordinatd to ensure comiliance with the relevant noise criteria; a suitable monitoring network for the precinct; protocols for data sharing; and procedures for idenfitiying and apportioning the source/s and contribution/s to cumulative noise impacts for the operating mines and other sources, using the noise and meterorological monitoring		Not Applicable			
Noise Management			need to be subject to ongoing review dependent upon the determination and commencement of other	Noted, first version not approved at the time of the audit.	Not Applicable			





- /		Requirement				Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	13	13. Where conditions in this approval refer to measurement of noise inthe context of the NSW Industrial Noise Policy the inversion class to be applied to the project is Class G. However, the Proponent may undertake an investigation to determine whether a proposal for change in this classification could be considered for approval by the Director-General. Any such investigation must be conducted in consultation with the EPA and be conducted by a suitably qualified person whose appointment has been endorsed bythe EPA and approved by the Director-General. The report and recommendation must be submitted to the EPA for endorsement prior to submission to the Director-General is referral is satisfied that the recommendation is reasonable, then the Director-General may amend the inversion class applying to the project under this approval.	ectrum acoustics monitoring reports specify Class G inversions	Compliant			
	BLASTING						
	Blasting Criteria						
Project Approval 11-0047	14	Airplant land	-06-14 Matong Station 122.7dBA, this monitor is on project owned nd. herwise compliant in the audit period	Compliant			
Project Approval 11-0047	Blasting Hours	The Proponent shall only carry out blasting on the site between 9 am and 5 pm Monday to Saturday inclusive. No blasting is allowed on Sundays, public holidays. Or at any time without the written approval from the Director-Gerneral	mpliant in the audit period	Compliant			
Project Approval 11-0047	Blasting Frequency 16	(a) 1 blast a day; unless an additional blast is required following a blast misfire; and not (b) 4 blasts a week, averaged over a calendar year, for the project.	th May 2014, two blasts on the same day, second blast fr safety and tifications made. 4-13 two blasts fired, the second was not notified to neighbours or the A. th blasts were below 0.5mm/s at all monitors.	Compliant			
	Property Inspections	If the Drawnest resolute a written request from the county of any privately award land while 2	no vessived in the sould navied				
Project Approval 11-0047	17	If the Proponent receives a written request from the owner of any privately-owned land within 2 kilometres of the approved open-cut pit on site, for a property inspection to establish the baseline condition of any buildings and/or structures on his/her land, or to have a previous property inspection report updated, then within 2 months of receiving this request the Proponent shall: (a) commission a suitably qualified, experienced and independent person, whose appointment is acceptable to both parties, to: (b) establish the baseline condition of any buildings and/or structures on the land, or update the previous property inspection report; and (c) identify any measures that should be implemented to minimise the potential blasting impacts of the project on these buildings and/or structures; and (d) give the landowner a copy of the new or updated property inspection report. If there is a dispute over the selection of the suitably qualified, experienced and independent person, or the Proponent or landowner disagres with the findings of the idependent property investigation, either party may refer the matter to the Director-General for resolution.	received in the audit period	Not Applicable			
	Property Investigation	15					





Reference	Condition	Requirement	Evidence	Audit Finding		Risk	
keterence	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	18	If any owner of the privately-owned land within 2 kilometres of blasting operations, or any other landowner nominated by the Director-General, claims that the building and/or structures on his/her land have been damaged as a result of blasting on site, then within 2 months of receiving this claim in writing from the landowner, the proponent sall: (a) commission a suitably qualified, experienced and independent person, whose appointment is acceptable to both parties, to investigate the claim; and (b) give the landowner a copy of the property investigation report. If this independent property investigation confirms the landowner's claim, and both parties agree with these findings, then the Proponent shall repair the damages to the satisfaction of the Director-General. If there is a dispute over the selection of the suitably qualified, experienced and independent person, or the Proponent or landowner disagrees with the findings of the independent property investigation, either party may refer the matter to the Director-General.	This has not occurred in the audit period	Not Applicable			
	Property Investigation	ns					
Project Approval 11-0047	19	During mining operations on site, the Proponent shall: (a) implement best practice blasting management to: protect the safety of people and livestock in the surrounding area; protect public or private infrastructure/property in the surrounding area from any damage; minimise the dust and fume emissions of any blasting; and minimise blasting impacts on heritage items in the vicinity of the site; (b) co-ordinate the timing of the blasting on site with the timing of blasting at other mines within the Leard Forest Mining Precinct to minimise the cumulative blasting impacts of the mines; and (c) operate a suitable system to enable the public to get up-to-date information on the proposed balsting schedule on site, to the satisfaction of the Director-General.	Verified in the Blast Management Plan section of the audit.	Compliant			
Project Approval 11-0047	20	The Proponent shall not undertake blasting on-site within 500 metres of: (a) any public road without the approval of Council; or (b) any land outside of the site not owned by the Proponent, unless: - the Proponent has a written agreement with the relevant landowner to allow blasting to be carried out closer to the land, and the Proponent have advised the Department in writing of the terms of this agreement; or the Proponent has: - demonstrated that the blasting can be carried out closer to the land without compromisign the safety of the people or livestock on the land, or damaging the buildings and/or structures on the land; and - updated the Blast management Plan to include the specific measures that would be implemented while blasting is being carried out within 500 metres of the land, to the satisfaction of the Director- General.	No blasting has triggered this condition through being located within the 500m limit	Not Applicable			
	Blast Management Pla	an					
Project Approval 11-0047	21	satisfaction of the Director-General. This plan must: (a) be submitted to the Director-General for approval by the end of May 2013; (b) be prepared in consultation with the EPA and interested members of the local community who would potentially be affected by blasting; (c) propose and justify any alternative ground vibration limits for public infrastructure in the vicinity of the site; (d) describe the measures that would be implemented to ensure:best management practice is being employed; and compliance with the relevant conditions of this approval;	a) Compliant b) Compliant c)None proposed d) Best practise applied, not blast results are generally good indicative of good practise. e) Road Closure protocol not included but has been developed separately and approved. f) Fume Management Plan developed g) this is addressed and monitoring results were reviewed by the audit team as were investigations into blasts that did not trigger the monitors. There was no evidence of issues with fume.	Compliant			
Project Approval 11-0047	21	(h) include a Leard Forest Mining Precinct Blast Management Strategy, that has been prepared in consultation with other mines within the Leard forest Mining Precinct, to minimise cumulative blasting impacts. Note: The Leard Forest Mining Precinct Blast Management Strategy can be developed in stages, and will need to be subject to ongoing review dependent upon the determination of and commencement of other mining projects in the area.	The Leard Forest BMP has not been finalised or approved.	Not Applicable			
	AIR QUALITY AND GRI Odour	EENHOUSE GAS					
Project Approval 11-0047	22	Unless otherwise authorised by an EPL, the Proponent shall ensure that no offensive odours are emitted from the site, as defined under the POEO Act.	No evidenc of offensive odours and no odour complaints	Compliant			
	Greenhouse Gas Emis	sions					





D.C.	Constitution	Requirement	F.11	a dia Et ii		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	23	The Proponent shall implement all reasonable and feasible measures to minimise the release of greenhouse gas emissions from the site to the satisfaction of the Director-General.	Air Quality and Greenhoue Gas Management Plan	Compliant			
	Air Quality Criteria						
Project Approval 11-0047	24	The Proponent shall ensure that all reasonable and feasible avoidance and mitigation measures are implemented so that particulate matter emissions generated by the project do not cause exceedances of the criteria in Table 6, Table 7 and Table 8 at any residence-on privately owned land or on more than 25 percent of any privately-owned land. The assessment acknowledges that it may not be reasonable and feasible to prevent exceedance of yhe PM10 criteria in Table 6 at property 45 and exceedance of the criteria in Table 7 in year 16 at property 49. (To interpret the property locations referred to see the applicable figure(s) in Appendix 5.)	Air Quality and Greenhoue Gas Management Plan	Compliant			
Project Approval 11-0047	24	Politishit Averaging Period "Criterion	PM10 data from the Tarrawonga D2 TEOM monitor from 1 January 2013 to 1 December 2014 were provided to the audit. The 24-hour average PM10 data from the monitor show that the monitor recorded three PM10 24-hour average levels greater than the applicable criteria of SOyg/m3. The annual average PM10 concentration for 2013 is 15.3µg/m3 which is below the applicable annual average criteria of 30µg/m3. Two of the three elevated 24-hour average PM10 levels recorded at the D2 TEOM monitor (which occurred on 19 April 2013 and 11 May 2013) were due to a large spike in the data at a time that the wind was not blowing from the Mine towards the monitor, and thus the elevated levels on these two days are likely to be due to localised sources which were unrelated to Tarrawonga. A detailed analysis was conducted of the third elevated level of 78.7µg.m3 which occurred on 15 November 2014. It was found that elevated PM10 levels were recorded when the wind direction was generally downwind to the monitor. Tarrawonga's contribution to the 24-hour average level recorded at the D2 monitor on 15 November 2014 was calculated as the total dust level recorded when the monitor was downwind of the mine. This analysis conservatively assumes all dust levels recorded during these periods were due to Tarrawonga (and no other sources) and on this basis the contribution was estimated to be up to 32.5% of the total level of 78.7µg/m3, or 28.1µg/m3. HVAS data were provided for two locations from May 2007 to June 2014, however the data at one of the locations are non-compliant as they are being measured on a non-EPA run day. In this case the data show levels above the 24-hr average criterion that are not explained by the Mine.	Compliant			
Project Approval 11-0047	24	In incremental impact (in incremental increase in concentrations due to the project on its own). Deposited that if is the assessed a insoluble solids as defined by Shadettak SUSYRIA, ASSYZS 3580 10 t 2003- Methods for Sampling and Analysis of Ambient Air - Determination of Particulate Matter - Deposited Matter - Graymetric Methods for Sampling and Analysis of Ambient Air - Determination of Particulate Matter - Deposited Matter - Graymetric Methods - State State - S	Noted				
	Mine-Specific Air Qua	The Proponent shall ensure that particulate matter emissions generated by the project do not exceed	The 24-hour average PM10 data from the Templemore and Merriown				
Project Approval 11-0047	25	the criteria listed in Table 9 at any residence on privately-owned land or on more than 25 percent of any privately-owned land, except on property 49 in year 16. Table 9 Short-ferm criteria for particulate multier. Politiciant Averaging Period Criterion Particulate moster < 10 um 24 flour 30 upm² Since As provided by the EPAA Act, the criterion in Table 9 may be amended to a more stringent enterior in an EPL after the livel however of the EPL under auction 18 of the POEO Act	HVAS monitors for January 2013 to June 2014 are presented in Figure 3 and show three PM10 24-hour average levels greater than the applicable criteria of Solgy/m3. The Templemore and Merriown HVAS monitors recorded 2013 annual average PM10 levels of 19.7µg/m3 and 14.7µg/m3 respectively, which are below the applicable criteria of 30µg/m3. Analysis of the meteorological conditions on the three days with elevated HVAS PM10 levels show that the monitors would have been downwind of Tarrawonga approximately 32%, 2% and 11% of the time on 9 January 2013, 15 October 2013 and 15 February 2014 respectively. Based on the relatively small percentage of time the monitors were downwind of the mine, it is unlikely that Tarrawonga would have contributed significantly to the total levels measured on 15 October 2013 and 15 February 2014, and is unlikely to have contributed more than approximately 32% of 24-hour average PM10 result of 52.9µg/m3 recorded on 9 January 2013, or more than approximately 17µg/m3. It should be noted that the analysis assumes a constant dust level over the monitoring period to provide an indicative estimate of Tarrawonga's contribution. A more accurate analysis would subtract the level recorded at Merriown monitor, as it was generally upwind of the mine, from the corresponding downwind level at the Templemore monitor. However as noted above, the monitors do not record data on the same day as required, hampering such an analysis.	Not Compliant Administrative			





Reference	Condition		Re	quirement		Evidence	Audit Finding		Risk	
Kelefelice	Condition						Addit Finding	Consequence	Likelihood	Risk
Project Approval 11-0048	25					Three HVAS results exceeded 24-hr average PM ₁₀ criteria levels between January 2013 and June 2014, but no evidence of any investigation by the mine to determine whether there were extraordinary events. The three machines were not controlled by TCPL. No publically available information with which to assess compliance (or				
Project Approval 11-0049	25					not) for PM_{10} and TSP. The AQMP states that the TEOMs are not used for compliance purposes, but there is no otherwise inferred assessment of compliance available for 2013-2014 for the audit.				
	Air Quality Acquisition							T		
Project Approval 11-0047	26	exceedance of the releve privately-owned land or	vant cumulative criterion on more than 25 perousition from the lando	wner the Proponent shall ac		No acquisition due to air quality in the audit period.	Not Applicable			
		Table 10: Long term land	acquation orders for part	culate matter		Noted, no acquisition in the audit period				
		Polit	utant	Averaging period	"Criterion					
		Total suspended partie	culate (TSP) matter	Annual	*90 µg/m³					
		Particulate matter < 10	0 µm (PM ₁₀)	Annuai	* 30 µg/m ³					
		Table 11: Street form land	acqueition criteria for part	loutate matter						
			utant	Averaging period	^o Criterion					
		Particulate matter < 1	0 pm (PM _{to})	24 hour	*150 µg/m²					
		Particulate matter < 1	0 μm (PM ₁₀)	24 hour	*50 µg/m²					
Project Approval 11-0047	26	Table 12: Long term land	acquisition enterto for dap	outlid dust			Not Applicable			
		Pollutant Averaging period Maximum increase in deposited dust level dust level								
		other sources): 6 incremental impact (iii ii C Deposited dust iii ii 3580-10 1-2003 Melnoda Mattar - Grasmetric Math	entri increase in consente neremental increase in con lo las assault as in for Samulting and Analysi od.	manhations that in the project on multiple solds as defend by a of Ambient An - Defermandor	"A girn ² /month signound concentrations due in all is own; Samularita Australia, ASA225 of Purificularly Matter Deposited sees fog, fire incutents, oc.arry other					
	Mine-ownded Land	The Bronopoet shall one	cure that all reasonable	and feasible avoidance and	mitigation measures are	None of these scenarios have occurred in the audit period.	I			
Project Approval 11-0047	27	implemented so that pa Table 6, table 7 and Tab adjacent mines), unless: (a) the tenant and/or la requirements under sch (b) the tenant on projec giving reasonable notic relocation and sourcing (c) air mitigation measu conditioning are installe another mine other tha (d) particulate matter a potential health risks; a (e) monitoring data is p	articulate matter emissible 8 at any occupied reison and owner has been not needule 4 of this approvinterelated land can terre, and the Proponent Lof alternative accomoures such as air filters, and the Proponent Lof alternative accomoures such as air filters, if not the Proponent Lof alternative accomoures such as air filters, if not the Proponent); ir quality monitoring is not described to the tenant resented to the tenant resented to the tenant and the proponent of the propon	ions generated by the proje esidence on any mine-owned iffied of any health risks in adal; ninate the tenancy agreeme ses its best endeavours to p dation; s first flush roof water drain:	ct do not exceed the criteria in I land (including land owned by ccordance with the notification int without penalty, subject to rovide assistance with age system and/or air landowner (where owned by enant and landowner of or a medical practitioner to	·	Not Applicable			
		property, to the satisfac			ateu with occupying the					





Reference	Condition	Requirement	Evidence	Audit Finding		Risk	
Reference	Condition		Evidence	Addit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	28	The Proponent shall: (a) implement best practice management to minimise the off-site odour, fume and dust emissions of the project; (b) operate a comprehensive air quality management system on site that uses a combination of predictive meteorological forecasting, predictive and real time are dispersion modelling and real-time air quality monitoring data to guide the day-to-day planning of mining operations and implementation of both proactive and reactive air quality mitigation measures to ensure compliance with the relevant conditions of this approval; (c) manage PM2.5 levels in accordance with any requirements of an EPL; (d) minimise the air quality impacts of the project during adverse meteorological conditions and extraordinary events (see note D under table 8); (e) minimise any visible off-site air pollution; (f) minimise the surface disturbance of the site generated by the project; and (g) co-ordinate the air quality management on site with the air quality management at other mines within the Leard Forest Mining Precinct to minimise the cumulative air quality impacts of the mines, to the satisfaction of the Director-General.	B) Mine operates predictive meteorological forecasting, but does not appear to operate any predictive or real-time air dispersion modelling to	Not Compliant	D	2	Medium
	Operating Conditions			1			
Project Approval 11-0047	29	29 The proponent shall prepare and implement an Air Quality and Greenhouse Gas Management Plan for the project to the satisfaction of the Director-General. This plan must: (a) be prepared in consultation with the EPA and be submitted to the Director-General for approval by the end of May 2013; (b) describe the measures that would be implemented to ensure: best practice management is being employed; the air Quality impacts of the project are minimised during adverse meteorological conditions and extraordinary events; and compliance with the relevant conditions of this approval;	a) compliant b) AQMP incorporates such measures, but is not specific about extraordinary events, and does not contain a clear procedure to identify compliance with the 24-hour average PM10 criterion.				
Project Approval 11-0047	29	c) describe the proposed air quality management system; (d) include a risk/response matrix to codify mine operational responses to varying levels of risk resulting from weather conditions and specific mining activities; (e) include commitments to provide summary reports and specific briefings at CCC meetings on issues arising from air quality monitoring; (f) include and air quality monitoring program that: uses a combination of real-time monitors and supplementary monitors to evaluate the performance of the project; adequately supports the proactive and reactive air quality management system; includes PM2.5 monitoring; includes PM2.5 monitoring; includes monitoring of occupied mine-owned residences and residences on the air quality affected land in Table 1, subject to the agreement of the tenant and/or landowner; evaluates and reports on the effectiveness of the air quality management system; includes sufficient random audits of operating responses to real time air quality management systems to determine the ongoing effectiveness of these responses in maintaining the project within the relevant criteria in this Schedule and the requirements of conditions 24 and 25, above; includes a protocol for determining any exceedences of the relevant conditions in this approval; and	c) AQMP outlines the system to manage air quality. D) AQMP outlines risk and response matrix as part of daily dust inspections. E) AQMP includes such commitments. F) AQMP includes most of these aspects, but does not appear to cover; - supplementary monitors; - protocol for determining exceedances of 24-hour PM10 and TSP criterion.	Not Compliant Administrative			





(g) includes a Leard Forest Mining Precinct Air Quality Management Strategy consultation with the other coal mine in the Precinct on minimise the cumular all mines within the Precinct, that includes: systems and processes to ensure that all mines are managed to achieve their a shared environmental monitoring network and data sharing protocol; control monitoring site (g) to provide real time data on background air quality by mining in the Leard Forest Mining Precinct and representative of regional a shared predictive and real time air dispersion model covering the Lear Fore used for assessment of cumulative impacts, optimising locations of the shared network, validation of air predictions and optimising mitigation measures an procedures for identifying and apportioning the source/s and contribution/s for both mines and other sources, using the air quality and meteorological mappropriate investigative tools such as modelling of post incident plume disper monitors and chemical methods of source apportionment (where possible). Notes: the requirement for regionally based control sites can be firther reviewed if it network is implemented and operated by the EPA as recommended in the direct of the site that. The Leard Forest Mining Precinct Air Quality Management Strategy can be deneed to be subject to ongoing review dependent upon the determination of a other mining projects in the area. The management plan should be cinsistent with the EPA's guidance on Best N reporting and Reactive Particulate Management Strategies. METEOROLOGICAL MONITORING For the life of the project, the Proponent shall ensure that there is a meteoro of the site that: (a) complies with the requirements in the Approved Methods for Sampling of South Wales guideline; and (b) is capable of continuous real-time measurement of temperature lapse and the mining projects in the area. The management plan should be cinsistent with the EPA's guidance on Best N reporting and Reactive Particulate Management Strategies. The Project Approval 11-0047	Puldanas	Audit Finding		Risk	
consultation with the other coal mine in the Precinct, that includes: all mines within the Precinct, that includes: systems and processes to ensure that all mines are managed to achieve their a shared environmental monitoring network and data sharing protocol; control monitoring site (s) to provide real time data on background air quality by mining in the leard forest Mining Precinct and representative of regional. a shared predictive and real time air dispersion model covering the Lear Fore used for assessment of cumulative impacts, optimising locations of the shared network, validation of air predictions and optimising mitigation measures; an procedures for identifying and approroinging the source/s and contributions/s for both mines and other sources, using the air quality and meteroological appropriate investigative tools such as modelling of post incident plume disper monitors and chemical methods of source apportionment (where possible). Notes: the requirement for regionally based control sites can be firther reviewed if in network is implemented and operated by the EPA as recommended in the dri Use Plan for New England North West. The Leard Forset Mining Precinct Air Quality Management Strategy can be de need to be subject to ongoing review dependent upon the determination of a other mining projects in the area. The management plan should be cinsistent with the EPA's guidance on Best in reporting and Reactive Particulate Management Strategies. METEOROLOGICAL MONITORING For the life of the project, the Proponent shall ensure that there is a meteorio of the site that: (a) complies with the requirements in the Approved Methods for Sampling of South Wales guideline; and (b) is capable of continuous real-time measurement of temperature lapse rat Note: Under the Water Act 1912 and/or the Water Management Act 2000, th obtain the necessary water licences for the project. Compensatory Water Supply The Proponent shall ensure that it as sufficient water for all stages of the pro adjust the scale of mining o	Evidence	Audit Finding	Consequence	Likelihood	Risk
For the life of the project, the Proponent shall ensure that there is a meteoro of the site that: (a) complies with the requirements in the Approved Methods for Sampling of South Wales guideline; and (b) is capable of continuous real-time measurement of temperature lapse rat NSW Industrial Noise Policy, unless a suitable alternative is approved by the Disconsultation with the EPA. SOIL AND WATER	appear to cover; - a suitable control monitoring site at present (given that the "Will-gai site is close to other mining activity, it is not a suitable control monitoring site at present (given that the "Will-gai site is close to other mining activity, it is not a suitable control monitoring site representative of regional air quality. Tamworth is not in the same shed and is not a suitable control site either.) - AQMS is not clear that the proposed air dispersion model would covall of these elements; - Conceptual strategy proposed in the AQMS to identify impacts may be as effective as described in controlling day to day effects. AQMP includes an AQMS for the BTM complex, but the AQMP/S does appear to cover; - a suitable control monitoring site at present (given that the "Will-gai site is close to other mining activity, it is not a suitable control monitor is ite representative of regional air quality. Tamworth is not in the same shed and is not a suitable control site either.) - AQMS is not clear that the proposed air dispersion model would coval in the same shed and is not a suitable control site either.) - AQMS is not clear that the proposed in the AQMS to identify impacts may in the same shed and is not a suitable control site either.) - AQMS is not clear that the proposed in the AQMS to identify impacts may in the same shed and is not a suitable control site either.)	ng air r ot ot ang air			
For the life of the project, the Proponent shall ensure that there is a meteoro of the site that: (a) complies with the requirements in the Approved Methods for Sampling of South Wales guideline; and (b) is capable of continuous real-time measurement of temperature lapse rat NSW Industrial Noise Policy, unless a suitable alternative is approved by the Disconsultation with the EPA. SOIL AND WATER Note: Under the Water Act 1912 and/or the Water Management Act 2000, the obtain the necessary water licences for the project. SOIL AND WATER The Proponent shall ensure that it as sufficient water for all stages of the proadjust the scale of mining operations on site to match its available water supply the Director-General. Compensatory Water Supply The Proponent shall provide a compensatory water supply to any owner of provide a sufficient water for all stages of the project in consultation with NOW, and to the satisfaction of the Director-General. Project Approval 11-0047 32 If the Compensatory water supply measurer suprivade an alternative long-time equivalent to the loss attributed to the project. Equivalent water supply should an interim basis) within 24 hours of the loss being identified. If the Proponent is unable to provide an alternative long-time dispute about the implementation of these measures, then either party may in Director-General. Surface Water Discharges					
Note: Under the Water Act 1912 and/or the Water Management Act 2000, the obtain the necessary water licences for the project. SOIL AND WATER The Proponent shall ensure that it as sufficient water for all stages of the pro adjust the scale of mining operations on site to match its available water supp the Director-General. Compensatory Water Supply The Proponent shall provide a compensatory water supply to any owner of pr water supply is adversely and directly impacted (other than negligible impact in consultation with NOW, and to the satisfaction of the Director-General. The compensatory water supply measures must provide an alternative long-te equivalent to the loss attributed to the project. Equivalent water supply shoul an interim basis) within 24 hours of the loss being identified. If the Proponent is unable to provide an alternative long-tem supply measures to be imp dispute about the implementation of these measures, then either party may in Director-General for resolution. If the Proponent is unable to provide an alternative long-term supply of water provide alternative compensation to the satisfaction of the Director-General.	Data reviewed was not able to sureity as to compliane with a) and b) accordance with the	Noted			
Project Approval 11-0047 SOIL AND WATER The Proponent shall ensure that it as sufficient water for all stages of the propent adjust the scale of mining operations on site to match its available water supply the Director-General. Compensatory Water Supply The Proponent shall provide a compensatory water supply to any owner of propensatory water supply is adversely and directly impacted (other than negligible impact in consultation with NOW, and to the satisfaction of the Director-General. The compensatory water supply measures must provide an alternative long-tequivalent to the loss attributed to the project. Equivalent water supply shoul an interim basis) within 24 hours of the loss being identified. If the Proponent and the landowner cannot agree on the measures to be imposite about the implementation of these measures, then either party may in Director-General. Surface Water Discharges					
Project Approval 11-0047 The Proponent shall ensure that it as sufficient water for all stages of the pro adjust the scale of mining operations on site to match its available water suppl the Director-General. Compensatory Water Supply The Proponent shall provide a compensatory water supply to any owner of pr water supply is adversely and directly impacted (other than negligible impact in consultation with NOW, and to the satisfaction of the Director-General. The compensatory water supply measures must provide an alternative long-te equivalent to the loss attributed to the project. Equivalent water supply shoul an interim basis) within 24 hours of the loss being identified. If the Proponent and the landowner cannot agree on the measures to be imp dispute about the implementation of these measures, then either party may in Director-General. Surface Water Discharges	roponent is required to Water Licenses provided and are adequate to cover groundwater extraction.	Compliant			
The Proponent shall provide a compensatory water supply to any owner of pr water supply is adversely and directly impacted (other than negligible impact in consultation with NOW, and to the satisfaction of the Director-General. The compensatory water supply measures must provide an alternative long-te equivalent to the loss attributed to the project. Equivalent water supply shoul an interim basily within 24 hours of the loss being identified. If the Proponent and the landowner cannot agree on the measures to be imp dispute about the implementation of these measures, then either party may in Director-General for resolution. If the Proponent is unable to provide an alternative long-term supply of water provide alternative compensation to the satisfaction of the Director-General. Surface Water Discharges		Compliant			
water supply is adversely and directly impacted (other than negligible impact in consultation with NOW, and to the satisfaction of the Director-General. The compensatory water supply measures must provide an alternative long-te equivalent to the loss attributed to the project. Equivalent water supply shoul an interim basis) within 24 hours of the loss being identified. If the Proponent and the landowner cannot agree on the measures to be implicated in the project. Equivalent water supply shoul an interim basis) within 24 hours of the loss being identified. If the Proponent and the landowner cannot agree on the measures to be implicated in the proposed in the provide and alternative long-term supply of water provide alternative compensation to the satisfaction of the Director-General. Surface Water Discharges	Ash, and disable to the bald of the bald o	<u> </u>	· -		
	s a result of the project, n supply of water that is be provided (at least on nented, or there is a er the matter to the	Not Applicable			
The Proponent shall ensure that all surface water discharges from the site cor	Loudshale discharge APAR to discharge to the control of the contro				
Project Approval 11-0047 33 limits (both volume and quality) set for the project in any EPL.	ly with the discharge AEMR indicates that site complies and a review of the monitoring database supports this.	Compliant			





Defenence	Condition		Requirement	Eddones	Audia Finding		Risk	
Reference	Condition			Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	34	in the EA and complies with the performan General. Takin 1.5 Gazanthi Comme and utilizated anything pro-	thas no greater environmental consequences than predicted ce objective in Table 13, to the satisfaction of the Director- formance objective. **Objective** **Objective** **No impres than negligible preservemental consequences to the allowed againty, including: **negligible change in pround-water levels. **negligible impact to director provides and the allowed againty in a registration impact to director promisers uses. **Hydraximizity and generous/prioragically stable (including the low permonality tearner) **Registrate change to difeate flooding Characteristics remaining flood inveils, velocities and flood objects change to appear to the proposition of the propos	Mining is currently not moving towards Goonbri Creek as quickly as predicted in the EA and there will be no impacts in the medium term. When mining returns to plan the Goonbri Creek Management Plan and design of the low permiability barrier will be developed for approval.	Not Applicable			
	Goonbri Creek Divers	ion and Flood Bund Concept Design Plan						
Project Approval 11-0047	35	geomorphic baseline conditions within the (e) set out the construction program for th staged, and integrated with mining operait (f) describe the revegetation program for t species; (g) establish the water quality, ecology, hy and completion criteria for the creek diver and (h) be revised in consultation with NOW, O	or approval by December 2016; It diversion; ter quality, ecology, hydrological (including flooding) and creek; e creek diversion and LPB, describing how the work would be ons; he creek diversion and the use of a range on suitable native drological (including flooding) and geomorphic performance sion and LPB based on the assessment of baseline conditions; EH and the Namoi CMA, and resubmitted for approval by the sto fthe detailed technical design required in condition 36 and	Mining is currently not moving towards Goonbri Creek as quickly as predicted in the EA and there will be no impacts in the medium term. When mining returns to plan the Goonbri Creek Management Plan and design of the low permiability barrier will be developed for approval.	Not Applicable			
	Goonbri Creek Divers	ion and Low Permeability Barrier - Design a						
Project Approval 11-0047	36	"36 The Proponent shall design the Goonb Director-General. The detailed designs mu: (a) be designed by a suitably qualified and (b) be endorsed by NOW and approved by works or construction on the Goonbri Cree (c) be generally in accordance with the cor applicable to Australian Standards (includir (d) include detailed design, construction ar completion criteria; (e) deonstrate that the design would achie (f) demonstrate the LPB design would achie (f) demonstrate the LPB design would remwithstand mining operations, geological arbiological attack. 37 The Proponent shall: (a) construct the Goonbri Creek diversion a metres of the Goonbru Creek alluvium, and and (b) within 2 months of the construction of report to the Director-General and NOW, c and barrier have been constructed:	ri Creek diversion and LPB to the satisfaction of NOW and the st: experienced expert/s; the Director-General prior to the commencement of any k k diversion and LPB; expertial designs in the EA (and depicted in Appendix 6), and ng AS 3798-2007); de dengineering specifications, performance criteria and enve the relevant performance objectives and criteria; and ain effective over an appropriate lifespan and would did weather events, decay and corrosive attack - including and LPB prior to undertaking any mining operations within 200 did at least 5 years prior to the planned mining in the alluvium; the Goonbri Creek diversion and LPB, submit an as-executed retrified by a practising engineer, confirming that the diversion the EA, applicable Australian Standards)including AS 3798- ion 36 above); and		Not Applicable			





D. C.	Company of the Compan	Requirement	5.1	A coding from the		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	36	"Notes: The Goonbri Creek alluvium, diversion, conceptual low permeability barrier and flood bunds are shown in Appendix 6. The diversion and low permeability barrier may be constructed on a staged basis. In this case, the reports required under conditions 36 and 37 shall be submitted for each stage "	Noted				
	Goonbri Creek Diversi	l ion and Low Permeability Barrier - Monitoring and Management Plan					
Project Approval 11-0047	38	The Proponent shall prepare and implement a Goonbri Creek Diversion and Low Permeability Barrier Monitoring and Management Plan to the satisfaction of the NOW and the Director-General. The plan must: a) be prepared by a suitably qualified and experienced expert/s; b) be endorsed by the NOW and approved to the Director-General prior to commencement of any works or construction on the Goonbri Creek diversion and LPB; c) describe the monitoring and maintenance procedures to be implemented and the scheduling of these procedures; d) demonstrate the monitoring system would be capable of timely detection of any failure of deficiency in the LPB and any impacts on Goonbri Creek and its associated alluvium; e) describe the contingency measures that would be implemented in the event of a failure or deficiency in the LPB, or other impoact on te Goonbri Creek and its associated alluvium; and f) identify the entity that would take responsibility for the future liabilities and costs associated with the long-term monitoring and maintenance of the LPB, flood bund, void and pit lake, and demonstrate that this entity's security and finances would be assured in the long term.	Mining is currently not moving towards Goonbri Creek as quickly as predicted in the EA and there will be no impacts in the medium term. When mining returns to plan the Goonbri Creek Management Plan and design of the low permiability barrier will be developed for approval.	Not Applicable			
	Water Management F	Plan	!				
Project Approval 11-0047	39	The Proponent shalli prepare and implement a water Management Plan for the project to the satisfaction of the Director-General. This plan must: a) be prepared in consultation with OEH, NOW and Namoi Cma, by suitably qualified and experienced person/s whose appointment has been approved by the Director-General. b) be submitted to the Director-General for approval by the end of May 2013; and c) in addition to the standard requirements for management plans (see condition 3 of schedule 5), include a:	WMP Issued to DP&E in accordance with project approval. Comments received, and being addressed at the time of the audit.	Compliant			
Project Approval 11-0047	39	i) Site Water Balance, that includes: details of: sources and security of water supply, including contingency for future reporting periods; water use and managemen on site; any off-site water discharges; reporting procedures, including the preparation of a site water balance for each calendar year; a program to validate the surface water model, including monitoring discharge volumes from the site and comparison of monitoring results with modelled predictions; and describes the measures that would be implemented to mininmise clean water use on site;	There is a water balance included. The water balance meets these criteria though some of the criteria are met in the WMP. It is an updated version, created in 2014 in line with the requirement for annual updates but there is no specific reference to an annual update fo the Water Balance.				
Project Approval 11-0047	39	ii) Surface Water Management Plan, that includes: detailed baseline data on surface water flows and quality in the water-bodies that could potentially be affected by the project; detailed baseline data on soils within the irrigation area; detailed baseline data on hydrology across the downstream drainage system of the Namoi River floodplain from the mine site to the Namoi River, including Barbers Lagoon and The Slush Holes; a detailed description of the water management system on site, including the: clean water diversion systems; erosion and sediment controls (mine water system); mine water management systems including irrigation areas; discharge limits in accordance with FPL requirements; and water storages; detailed plans, including design objectives and performance criteria for: design and management of final voids; design and management for the emplacement of reject materials, sodic and dispersible soils and acid or sulphate generating materials; the Goonbri Creek diversion and low permeability barrier; reinstatement of drainage lines on the rehabilitated areas of the site; and control of any potential water pollution from the rehabilitated areas of the site; performance criteria for the following, including trigger levels for investigating any potentially adverse impacts associated with the project: the water management system; soils within the irrigation area; downstream surface water quality; downstream surface water quality; downstream surface water quality;	The WMP developed by URS in 2013 does not contain all of these details. The detailed baseline data on hydrology of the Namoi is not present.	Not Compliant	_	1	Medium





		Requirement				Risk	
Reference	Condition	·	Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	39	stream and riparian vegetation health, including the Namoi River and its tributaries including Barbers Lagoon and The Slush Holes; a program to monitor and assess: the effectiveness of the water management system; soils withing the irrigation area; the effectiveness of the Goonbri Creek diversion and flood bunds (see conditions 34-38); surface water flows and quality in the watercourses that could be affected by the project; and downstream flooding impacts; reporting procedures for the results of the monitoring program; and a plan to respond to any exceedences of the performance criteria, and mitigate and/or offset any adverse surface water impacts of the project;		NOCCOMPRAN	ć	•	wedum
Project Approval 11-0047	39	iii) Groundwater Management Plan, that includes: detailed baseline data of groundwater levels, yield and quality in the region, and privately-owned groundwater bores including a detailed survey/schedule of groundwater dependent ecosystems (including stygo-fauna), that could be affected by the project; detailed plans, including design objectives and performance criteria, for the design and management of; the proposed final void; and coal reject and potential acid forming material emplacement; groundwater assessment criteria including trigger levels for investigating any potentially adverse groundwater impacts; a program to monitor and assess: groundwater inflows to be open cut mining operations; the effectiveness of the LPB; the seepage/leachate from the LPB, water storages, emplacements and the final void; interconnectivity between the alluvial and bedrock aquifers; background changes in groundwater yield/quality against mine-induced changes; the impacts of the project on: regional and local (including alluvial) aquifers; groundwater supply of potentially affected landowners; groundwater supply of potentially affected landowners; groundwater dependent ecosystems (including potential impacts on stygo-fauna) and riparian vegetation; a program to validate the groundwater model for the project, including an independent review of the model every 3 years, and comparison of monitoring results with modelleid predictions; and a plan to respond to any exceedences of the performance criteria; and	The Groundwater Management Plan does not refer to stygo-fauna or groundwater dependent ecosystems.				
Project Approval 11-0047	39	"iv) Leard Forest Mining Precinct Water Management Strategy, that has been prepared in consultation with other mines within the precinct to: minimise the rumulative water quality impacts of the mines; review opportunities for water sharing/water transfers between mines; co-ordinate water quality monitoring programs as far as practicable; undertake joint investigations/studies in relation to complaints/exceedences of trigger levels where cumulative impacts are considered likely; and co-ordinate modelling programs for validation, re-calibration and re-running of the groundwater and surface water models using approved mine operation plans. Note: The Leard Forest Mining Precinct Water Management Strategy can be developed in stages and will need to be subject to ongoing review, dependent upon the determination of and commencement of other mining projects in the area.	The Strategy has been prepared and submitted but is the focus of ongoing improvements and has not been fully implemented.	Not Applicable			
	BIODIVERSITY						
Project Approval 11-0047	Biodiversity Offset Str 40	The Proponent shall implement the biodiversity offset strategy described in the EA, summarised in Table 14 and shown conceptually in Appendix 7, to the satisfaction of the Director-General. The EA of the Common of the Bulletin of the	The BOS has only been recently developed and approved, implementation has commenced but is in it's infancy.	Compliant			
Project Approval 11-0047	40	Note: For the purposes of this approval Box Gum Woodland refers to the EEC listed as White Box Yellow Box Blakely's Red Gum Woodland under the TSC Act, and the CEEC listed as White Box Blakely's Red Gum Grassy Woodland and Derived Native Grassslands under the EPBC Act, or similar EEC as may be updated from time to time.	Noted				
	Leard Forest Mining P	recinct Regional Biodiversity Strategy	•	•			





Deference	Complete	Requirement	F. dd	Audia Findi		Risk	
Reference	Condition	·	Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	41	The Proponent shall contribute to the funding and preparation of the Leard Forest Mining Precinct Regional Biodiversity Strategy, as required under the approvals for the Boggabri coal mine and Maules Creek coal mine, to the satisfaction of the Director-General.	Submitted to Commonwealth but State have not approved	Not Applicable			
Project Approval 11-0047	41	"Notes: The approvals for the Boggabri coal mine and Maules Creek coal mine require the proponents of the mines in the Leard Forest mining precinct to prepare the regional biodiversity strategy in 3 stages, including: Stage 1 Stooping Stage, by the end of January 2013; Stage 2 Stategy Development Stage, by the end of January 2014; and Stage 3 Stategy Review Stage, by the end of December 2018. The strategy is required to be prepared in collaboration with a working group comprising relevant government agencies and the Leard Forest Mining precinct mining, rand charmies, and charied by an independent person. Funding of the strategy should be based on predicted clearing of native vegetation for the three project within the Leard Forest Mining Precinct. Based on the predicted clearing for the projects, the funding spilt would equate to total contributions of 36% from Boggabri (clearing of 1,385 ha), 54% from Maules Creek (clearing of 2,078 ha) and 10% from Torrawonga (clearing of 397 ha). This funding arrangement can be further refined in the Stage 1 Scoping Stage. "	Noted				
	Revised Biodiversity C		The second secon	1			
Project Approval 11-0047	42	"Within 6 months of the approval of Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Strategy the Proponent shall review, and if necessary revise, the biodiversity offset strategy for the project to the satisfaction of the Director-General. The review/revision must: be prepared in consultation with OEH, Namoi CMA, Forests NSW, the CCC, DPI Catchments and Lands and SEWPaC; not reduce the size or quality of the offset area; and be consistent (as far as is possible) with the recommendations and objectives of the Leard Forest Mining Precinct Regional Biodiversity Strategy. "	The strategy is not yet approved and undergoing further negotiation with the regulatory authorities.	Not Applicable			
	Threatened Species						
Project Approval 11-0047	43	For the White Box – Yellow Box – Blakely's Red Gum Grassy Woodland Endangered Ecological Community the Proponent shall: a) ensure that the Biodiversity Offset Strategy and site Rehabilitation Strategy is focused on protection rehabilitation, re-establishment and long-term maintenance of viable stands of this community; b) investigate in consultation with DEH and the Namio CMA, all factors likely to enhance or impede the effective long term restoration of degraded remnants of this EEC in offset areas or regeneration of this EEC on disturbed areas (both offset areas and the site); o) within 24 months of the date of this approval (and if possible in conjunction with Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Strategy), submit a report of this investigation and provide d) incorporate the approved implementation plan into the revised Biodiversity Management Plan, required under condition48.	a) Complies b) Not commenced c) Due 22-1-15 d) The Strategy is not approved and as in c) above, not yet required	Not Applicable			
Project Approval 11-0047	44	For all threatened species on site, the Proponent shall ensure that the Biodiversity Offset Strategy and Rehabilitation Strategy are focused on protection, rehabilitation and long-term maintenance of viable stands of suitable habitat for these species.	The BOS addresses these issues but they are better detailed in the Willerol BOMP	Compliant			
Project Approval 11-0047	45	The Proponent shall: a) investigate, in consultation with OEH and the Namoi CMA, all factors likely to enhance or impede the effective long term provision of suitable habitat(s) for the following species: Speckled Warbler, Brown Treecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittella, Turquoise Parrot, Masked Owl, Yellow-bellied Sheath Tail Bat and Squirrel Glider; b) within 12 months of the date of this approval (and if possible, in conjunction with Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Strategy), submit a report of this investigation and provide an implementation plan to ensure delivery of suitable areas of viable habitat for the species included in (a) above, for approval by the Director-General; and c) incorporate the approved implementation plan into the revised Biodiversity Management Plan, required under condition 48.	See Tarrawonga Coal Mine Threatened Fauna Investigation Report	Compliant			





Reference	Condition	Requirement	Evidence	Audit Finding		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	46	The Proponent shall make suitable arrangements to provide appropriate long-term security for the offset areas: a for the Willeroi Offset Area the long-term security shall be provided by way of: the Proponent entering into a conservation agreement of agreements pursuant to section 69B of the National Parks and Wildlife Act 1974, recording the obligations assumed by the Proponent under the conditions of this approval in relation to these offset, and registering the agreement(s) pursuant to section 69F of the National Parks and Wildlife Act 1974; or a tenure of higher conservation status such as a National Park, or Nature Reserve, under the National Parks and Wildlife Act 1974, the conservation agreement(s) must be registered by the end of December 2013 unless agreed otherwise by the Director-General after consultation b) by the end of December 2030 unless otherwise agreed by the Director-General, for the woodland to the satisfaction of the Director-General. Note: The Deparment acknowledges that the Proponent is investigating the potential to transfer part of all of the Willeroi Offset Area directly to the national park estate, and accepts that interim conservation measures may be implemented prior to this transfer.	negotiating with DP&E for an extension to this date.	Not Compliant Administrative			
	Biodiversity Managen	nent Plan					
Project Approval 11-0047	47	"The Proponent shall prepare and implement a Biodiversity Management Plan for the project to the satisfaction of the Director-General. This plan must: a) be prepared in consultation with OEH, SEWPaC, Forests NSW, the CCC, DPI Catchments and Lands and the Namoi CMA, and be submitted to the Director-General for approval by the end of May 2013; b) describe the short, medium, and long term measures that would be implemented to: manage the remnant vegetation and habitat on the site and in the offset area; and implement the biodiversity offset strategy, including detailed performance and completion criteria for evaluation the performance of the biodiversity offset strategy, and triggering remedial action (if necessary); d) include a detailed description of the measures that would be implemented for: enhancing the quality of existing vegetation and fauna habitat; restoring native vegetation and fauna habitat on the biodiversity offset area and rehabilitation area through focusing on assisted natural regeneration, targeted vegetation establishment and the introduction of naturally scarce fauna habitat features;	The Willeroi BOMP complies with these requirements	Compliant			
Project Approval 11-0047	47	maximising the salvage of resources within the approved disturbance area – including enhancement of the biodiversity offset area or rehabilitation area; collecting and propagating seed; minimising the impacts on fauna on site, including undertaking pre-clearance surveys; managing any potential conflicts between the proposed restoration works in the biodiversity offset area and any Aboriginal heritage values (both cultural and archaeological); managing salinity; controlling weeds and feral pests; controlling erosion; controlling access; and managing bushfire risk;	The Willeroi BOMP complies with these requirements	Compliant			
Project Approval 11-0047	48	"e) include a seasonally-based program to monitor and report on the effectiveness of these measures, and progress against the detailed performance and completion criteria; f) identify the potential risks to the successful implementation of the biodiversity offset strategy, and include a description of the contingency measures that would be implemented to mitigate against these risks; and g) include details of who would be responsible for monitoring, reviewing, and implementing the plan. Note: The Biodiversity Management Plan and Rehabilitation Management Plan need to be substantially integrated for achieving biodiversity objectives for the rehabilitated mine-site.	The Willeroi BOMP complies with these requirements	Compliant			





by the sea of interpretation for the protection and blockwestly desired by the protection of the solidownist place of the	D. Communication	Constitute	Requirement	E. Maria	a colta et colt		Risk	
Project Agenoval 11:007 **Right bordown for Not 2014, the Proposent and Land Seque a Conservation and Educational Seque and Educational Sequence of Conservation Sequence	Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
By the end of June 2014 and every 3 years thereafter, unless both the Director-General and OEH agree to a different interfame, the Proponent shall commission suitably qualified, experiences and independent person/s, whose appointment has been approved by the Director-General, to undertake an audit of the revergetation of the rehabilitation area and management and restoration within the Biodiversity Offset Strategy areas to the satisfaction of the Director-General, this audit must: a) include consultation with DEH, Namior Chang Policy Cand DEE; b) assess the performance of the revegetation in the rehabilitation area completed to date (and the Goothor Greek Diversion, once commenced) against the completion criteria in the Rehabilitation Management Plan; c) assess the performance of management and restoration in the off-site Biodiversity Offset d) identify any additional measures that should be implemented to improve the performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be implemented to improve the performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be applied in the establishment of native vegetation, including riparian vegetation around the realigned Goothir Creek, both before and after the realignment is undertaken; e) if the completion criteria have not been met, or are not adequately trending towards being met, determine the liekly ecological value of the rehabilitation and recommend additional measures to augment the Biodiversity Offset Strategy to accommend additional measures to augment the Biodiversity Offset Strategy, in constitution of additional measures to augment the Biodiversity Offset Strategy, in constitution of additional measures to augment the Director-General.	Project Approval 11-0047	49	Department to ensure that the biodiversity offset strategy is implemented in accordance with the performance and completion criteria of the Biodiversity Management Plan. The sum of the bond shall be determined by: a) calculating the full cost of implementing the biodiversity offset strategy (other than land acquisition costs); and b) employing a suitably qualified quantity surveyor to verify the calculated costs, to the satisfaction of the Director-General. If the offset strategy is completed generally in accordance with the completion criteria in the Biodiversity Management Plan to the satisfaction of the Director-General, the Director-General will release the bond. If offset strategy is not completed generally in accordance with the completion criteria in the Biodiversity Management Plan, the Director-General will call in all, or part of, the conservation bond, and arrange for the satisfactory completion of the relevant works. With the agreement of the Director-General, this bond may be combined with rehabilitation security deposit administered by DRE. Notes: Alternative funding arrangements for long term management of the Biodiversity Offset Strategy, such as provision of capital and management funding as agreed by OEH as part of a Biobanking Agreement or transfer to conservation reserve estate can be used to reduce the liability of the conservation and biodiversity bond. The sum of the bond may be reviewed in conjunction with any revision to the biodiversity offset	amount, not compliant with deadline stated.				
By the end of June 2014 and every 3 years thereafter, unless both the Director-General and OEH agree to a different interfame, the Proponent shall commission suitably qualified, experiences and independent person/s, whose appointment has been approved by the Director-General, to undertake an audit of the revergetation of the rehabilitation area and management and restoration within the Biodiversity Offset Strategy areas to the satisfaction of the Director-General, this audit must: a) include consultation with DEH, Namior Chang Policy Cand DEE; b) assess the performance of the revegetation in the rehabilitation area completed to date (and the Goothor Greek Diversion, once commenced) against the completion criteria in the Rehabilitation Management Plan; c) assess the performance of management and restoration in the off-site Biodiversity Offset d) identify any additional measures that should be implemented to improve the performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be implemented to improve the performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be applied in the establishment of native vegetation, including riparian vegetation around the realigned Goothir Creek, both before and after the realignment is undertaken; e) if the completion criteria have not been met, or are not adequately trending towards being met, determine the liekly ecological value of the rehabilitation and recommend additional measures to augment the Biodiversity Offset Strategy to accommend additional measures to augment the Biodiversity Offset Strategy, in constitution of additional measures to augment the Biodiversity Offset Strategy, in constitution of additional measures to augment the Director-General.		Independent Biodiver	city Δudit					
determine the liekly ecological value of the rehabilitation and restoration once completed, and recommend additional measures to augment the Biodiversity Offset Strategy to ensure that it adequately offsets the project's impacts on biodiversity. Project Approval 11-0047 50 If the audit recommends the implementation of additional measures to augment the Biodiversity Offset Strategy in accordance with (e) above, then within 6 months of the completion of the audit the Proponent shall revise the Biodiversity Offset Strategy, in consultation with the Department, OEH and SEWPaC, and to the satisfaction of the Director-General.	Project Approval 11-0047		By the end of June 2014 and every 3 years thereafter, unless both the Director-General and OEH agree to a different timeframe, the Proponent shall commission suitably qualified, experiences and independent person/s, whose appointment has been approved by the Director-General, to undertake an audit of the revegetation of the rehabilitation area and management and restoration within the Biodiversity Offset Strategy areas to the satisfaction of the Director-General. This audit must: a) include consultation with OEH, Namoi CIMA, DPI Catchments and Lands, SEWPaC, CCC and DRE; b) assess the performance of the revegetation in the rehabilitation area completed to date (and the Goonbri Creek Diversion, once commenced) against the completion criteria in the Rehabilitation Management Plan; c) assess the performance of management and restoration in the off-site Biodiversity Offset d) identify any additional measures that should be implemented to improve the performance of rehabilitation, management and restoration within the rehabilitation and biodiversity offset areas; and d) identify any additional measures that should be applied in the establishment of native vegetation, including riparian vegetation around the realigned Goonbri Creek, both before and after the realignment	This audit, non-compliant with timing requirements				
HERITAGE	Project Approval 11-0047	50	determine the liekly ecological value of the rehabilitation and restoration once completed, and recommend additional measures to augment the Biodiversity offest Strategy to ensure that it adequately offsets the project's impacts on biodiversity. If the audit recommends the implementation of additional measures to augment the Biodiversity Offset Strategy in accordance with (e) above, then within 6 months of the completion of the audit the Proponent shall revise the Biodiversity Offset Strategy, in consultation with the Department, OEH and					

Project Approval 11-0047





Deference	Condition	Requirement	Evidence	Audit Finding		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	51	The Proponent shall prepare and implement an Aboriginal Heritage Conservation Strategy for the project and the Biodiversity Offset Strategy areas to the satisfaction of the Director-General. This strategy must enhance and conserve the Aboriginal cultural heritage values (both cultural and archaelogical) and provide for their long-term protection and management. The Strategy must: a) be prepared by suitably qualified and experiences person/s whose appointment has been endorsed by the Director-General; b) be prepared in consultation with OEH, the local Aboriginal community and other mines within the Leard Forest Mining Precinct, and submitted the Director-General for approval within 12 months from the date of project approval; c) identify the Aboriginal cultural heritage values of the Biodiversity Offset Strategy areas; d) identify areas of high Aboriginal cultural heritage significance within both the site and the Leard Forest Mining Precinct; e) identify a range of options for enhancing and conserving Aboriginal cultural heritage values, with specific consideration of the potential for the long-term protection and management of significant sites within either the site, the Biodiversity Offset Strategy areas or other lands within the LEard Forest Minin Precinct identified as have high cultural heritage significance to the Aboriginal community; and	Submitted but not aproved. The TCL ACHMP ensures Aboriginal heritage is managed correctly in the interim.	Not Applicable			
Project Approval 11-0047	51	f) consider cumulative impacts and potential for developing joint initiatives with other mines within the Leard Forest Mining Precinct for enhancing and conserving Aboriginal cultural heritage values.					
	Heritage Managemen		Account of the Budgeth wellfield (1911) in the		T		
Project Approval 11-0047	52	The proponent shall prepare and implement a Heritage Management Plan for the project to the satisfaction of the Director-General. This plan must: a) be prepared by suitably qualified and experienced person/s whose appointment has been endorsed by the Director-General; b) be prepared in consultation with OEH and local Aboriginal stakeholders (in relation to the management of Aboriginal heritage values); c) be submitted to the Director-General for approval prior to undertaking any activities that may impact heritage items or sites, unless the Director-General agrees otherwise;	a) Approval of the "suitably qualified person/s" by the DG provided as evidence b) consultation appears to be compliant c) Noted but prior to the audit period	Compliant			
Project Approval 11-0047	52	include the following for the management of Aboriginal cultural heritage: a detailed archaeological salvage program for Aboriginal sites/objects within the project disturbance area, including methodolog and procedures/protocols for: sub-surface testing; staged salvage, based on anticipated mine planning; pre-disturbance monitoring; site assessment and reporting; research objectives inform knowledge of Aboriginal occupation; protection, storage and management of salvaged Aboriginal objects; addressing relevant statutory requirements under the National Parks and Wildlife Act 1974; and long term protection of salvaged Aboriginal objects; a description of the measures that would be implemented for: protecting, monitoring and managing reasonable access for Aboriland stakeholders to cultural heritage items on site and in the biodiversity offset area; managing the discovery of any human reamins or previously unidentified Aboriginal objects on site, including (in the case of human remains) stop work provisions and notificiation protocols; ongoing consultation with the local Aboriginal stakeholders in the conservation and management of Aboriginal cultural heritage both on-site and in the biodiversity offset area;	Existing Heritage Management plan contains most of these requirements, others are embedded in site procedures	Compliant			
Project Approval 11-0047	52	ensuring any workers on site receive suitable heritage induction prior to carrying out any activities which may disturb Aboriginal sites, and that suitable records are kept of these inductions; include the following for the management of historic heritage: a description of the measures that would be implemented for: managing the discovery of human remains for previously unidentified historic heritage items at the site, including (in the case of human remains) stoy work provisions and notification protocols; and ensuring workers on site receive suitable heritage inductions prior to carrying out any development on site, and ensure that suitable records of these inductions are kept	This is addressed in the HMP	Compliant			
Project Approval 11-0047	52	Note: The Department acknowledges that the initial Heritage Management Plan may not include a detailed plan for the implementation of the Aboriginal Heritage Conservation Strategy. If this occurs, the Proponent will be required to update the plan as soon as practicable following the Director-General's approval of the Aboriginal Heritage Conservaion Strategy.	Noted				
	TRANSPORT Roadworks						





Deference	Condition	Requirement	Cuidenes	Audit Finalia		Risk	
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk
Project Approval 11-0047	53	The Proponent shall: a) construct the Goonbri Road realignments and associated mine access road intersection, Goonbri Road/private coal haulage road intersection and the Goonbri Road/Dripping Rock Road/Blair Athol Lane intersection as shown conceptually in the EA; and b) install appropriate advance warning signs lighting on Goonbri Road, the private coal haulage road at the intersection of the Northern Site Access Road, to the satisfaction of Council. The road realignments and associated intersection upgrades shall be undertaken: to a bitument sealed standard, unless otherwise agreed by Council; and prior to any project works occuring within 25 metres of the existing Goonbri Road alignment, or on the southern/eastern side of the existing road alignment.	Not yet occurred due to scaling back in activities	Not Applicable			
Project Approval 11-0047	53	Note: The road upgrade works may be udnertaken in stages, with the agreement of Council.	Noted				
7 1	Coal Transport			·			
Project Approval 11-0047	54	Whilst coal transport by road is permitted under this approval, the Proponent shall ensure that: a) trucks travelling to and from the site do no exceed 40 kilometres per hour in the visinity of the school bus when it is operating on Hoad Lane, unless an alternative protocol is agreed by the Director-General; and b) spillage from coal haulage vehivles is minimised and promptly managed.	Radio comms, call up sight, notification within the site. All coal haulage is tarped to prevent spillage.	Compliant			
	Road Maintenance						
Project Approval 11-0047	55	The Proponent shall: a) keep records of the: amount of coal and gravel transported from the site (on a monthly basis); and date and time of each train movement on the Boggabri rail spur line generated by the project; and b) make these records available on its website at the end of each calendar year.	Website contains this information and is up to date till end 2013 which complies, Note the Boggabri rail spur is not yet in operation.	Compliant			
	VISUAL						
	Operating Conditions	The Proponent shall:	Annual inspections of the site by DP&E, lighting management procedure		T		
Project Approval 11-0047	57	a) implement all reasonable and feasible measures to minimise the visual and off-site lighting impacts of the project; b) ensure no outdoor lights shine above the horizontal; c) whereever possible, ensure that mobile equipment is appripriately designed and/or retroffitted to prevent light being directed above the horizontal; d) ensure that all external lighting associated with the project complies with Australian Standard AS4282 (INT) 1997 – Control of Obtrusive Effects of Outdoor lighting or its latest version; e) provide for the establishment of trees and shrubs and/or the construction of mounding or bunding: along the realigned Goonbri Road and access road to the mine site; along the services corridor to the Boggabri Coal Mine; around the water storage dams; and at other areas identified as necessary for the maintenance of satisfactory visual amenity; ans f) ensure that the visual appearance of all buildings, structurs, facilities or works (including paint colours and specifications) is aimed at blending as far as possible with the surrounding landscape, to the satisfaction of the Director-General.	and OCE inspections at afternoon shift startup. Note no lighting related complaints.	Compliant			
	Additional Visual Impa						
Project Approval 11-0047	58	Upon receiving a written request from the owner of any residence on privately-owned land which has, or would have, significant direct views of the mining operations and infrastructure on-site during the project, the Proponent shall implement additional visual impact mitigation measures (such as landscaping treatments or vegetation screens) to reduce the visibility of the mining operations and infrastructure from the residences on the privately-owned land. These mitigation measures must be reasonable and feasible, and must be implemented within a reasonable timeframe. If the Proponent and the owner cannot agree on the measures to be implemented, ir there is a dispute abdout the implementation of these measures, then either party may refer the matter to the Director-General for resolution.	No requests received in the audit period	Not Applicable			
Project Approval 11-0047	58	Notes: The additional visual impact mitigation measures must be aimed at reducing the visibility of the mining operations on site from affected residences, and do not require measures to reduce the visibility of the mining operations on site from affected residences, and do not require measures to reduce the visibility of the mining operations from other locations on the affected properties. The additional visual impact mitigation measures do not necessarily have to include the implementation of measures on the affected property itself (ie. The additional measures could involve the implementation of measures outside the affected property boundary that provide an effective reduction in visual impacts). Except in exceptional circumstances, the Director-General will not require additional visual impact mitigation to be undertaken for residences that are more than 5 kilometres form the mining operations.	Noted				





Reference	Condition	Requirement	Evidence	Audit Finding	Risk			
Hererenee			Lvidence	Additimanig	Consequence	Likelihood	Risk	
Project Approval 11-0047	BUSHFIRE MANAGEN	The Proponent shall: a) implement all reasonable and feasible measures to manage bushfire risks, including the suspension of activities that may have the potential to ignite a fire, during adverse conditions; b) ensure that the project is suitably equipped to respond to any fires on site; and c) assist the Rural Fire Service, Forests NSW, emergency services and National Parks and Wildlife Service as much as possible if there is a fire in the surrounding area.	audit team don't have the sites hot works permiting documentation and most of these activities would be covered by this permitting system.	Compliant				
	WASTE				<u> </u>			
Project Approval 11-0047	60	The Proponent shall: a) implement all reasonable and feasible measures to reduce waste, (including coal reject) generated by the project; b) ensure that the waste generated by the porject is appropriately stored, handled and disposed of; and c) monitor and report on the effectiveness of waste minimisation and management measures in the Annual Review.	and is reported in the AEMR	Compliant				
	REHABILITATION							
Project Approval 11-0047	61	The Proponent shall rehabilitate the site to the satisfaction of the Executive Director Mineral Resources This rehabilitation must be generally consistent with the proposed Rehabilitation Strategy describes in the EA (and depicted conceptually in Appendix 8) and comply with objectives in Table 15. Falls to disfusione	Site is not yet at the point where relinquishment is possible. The AEMR reports against rehabilitation objectives and annual inspections by DRE help guide the site towards these requirements. There are challenges for the saite associated with poor soil quality and weed seed bank in topsoils that are being addressed with varying success. As above	Compliant				
Project Approval 11-0047	61	Feature All land – excluding the 210 his of agricultural land and the final void agricultural land and the final void estatement plant good group consistency or continued ECs, and con	AS above	Compliant				
	Operating Conditions							
Project Approval 11-0047	62 Progressive Rehabiliti	The Proponent shall, in consultation with the Namoi CMA: a) develop a detailed soil management protocol that identifies procedures for: comprehensive soil surveys prior to soil stripping; assessment of top-soil and sub-soil suitability for mine rehabilitation; and annual soil balances to manage soil handling including direct respreading and stockpiling; b) maximise the salvage of suitable top-soils and sub-soils and biodiversity habitat components such as bush rocks, tree hollows and fallen timber for rehabilitation of disturbed areas within the site and for enhancement of biodiversity offset areas; and c) ensure that coal reject, or any potentially acid forming interburden materials, are not emplaced at elevations in the pit shell where they may promote acid or sulphate species generation and migration beyond the pit shell.	The requirments detailed here have been met in the Tarawonga Rehabilitation Management Plan and the Clearing and Pre-strip Procedure apart from the comprehensive soil surveys prior to topsoil stripping	Not Compliant Administrative				





Reference Condition Evidence Audit Finding	D. f.	6	Requirement		A dia et di	Risk			
Introduced contractions of the contraction of the c	Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk	
The Progenit Approval 11-0017 Project Approval 11-0017 Fig. 1. Approval 11-0017 Fig. 1. Approval 11-0017 Fig. 2. Approval 11-0017 Fig. 2. Approval 11-0017 Fig. 2. Approval 11-0017 Fig. 3. Approval 11-0017 Fig. 3. Approval 11-0017 Fig. 4. Approval 11-0017 Fig. 3. Approval 11-0017 Fig. 4. Approval 11-0017 Fig. 3. Approval 11-0017 Fig. 4. Approval 11-0017 Fig. 4. Approval 11-0017 Fig. 5. Approval 11-0017 Fig. 6. Approval 11-0017 Fig. 7. Approval 11-0017 Fig. 6. Approval 11-0017 Fig. 6. Approval 11-0017 Fig. 6. Approval 11-0017 Fig. 7. Approval 11-0017 Fig. 8. Appro	Project Approval 11-0047	63	following disturbance. All reasonable and feasible meaures must be taken to minimise the total area exposed for dust generation at any time. Interim rehabilitation strategies shall be employed when areas prone to dust generation cannot yet be permanently rehabilitated. Note: It is accepted that the parts of the site that are progressively rehabilitated may be subject to	the exception of the southern face of out of pit emplacement area to the south of the site which is exposed to the publics view and is not planned of the rehabilitated for some time. The site discussed the potential to aerially seed the slope which would assist with dust suppression but not visual impact. Recommendation - work be conducted to soften the visual impact of the unrehabilitated southern emplacement and to lessen the levels of	Compliant				
the Fescive Director, Monra Resources, This plan must: a be perspect in complation with the Separative Director, Without Biology and the Separative Separative Director, Without Biology and the Separative Separative Director, Without Biology and University Separative Director, Without Biology and University Separative Director, Without Biology and University Separative Director, Without Separative Director, Without Separative Director, Without Separative Sep		Rehabilitation Manag							
Final Void Design and Closure Final Void Design and Closure The Proponent shall prepare and implement an updated Final Void and Mine Closure Plan (as a component of the Coveral Rehabilitation Management Plan required under condition 64 of schedule 3) to the satisfaction of the Executive Director Mineral Resources (following consultation with the Director-General A draft plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2024. Each version of help and must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Sciences and independent persons, (including a groundwater expert) whose appointment has been approved by the Director Continued mining beyond current project life; interactions with the final funding and groundwater expert) whose appointment has been approved by the Director Continued mining beyond current project life; interactions with the final funding and groundwater expert with the final funding and produced mining beyond current project life; interactions with the final funding and produced mining beyond current project life; interactions with the final funding and produced mining beyond current project life; interacti	Project Approval 11-0047	64	the Executive Director, Mineral Resources. This plan must: a) be prepared in consultation with the Department, Forests NSW, NOW, OEH, Namoi CMA and Council; b) be submitted to the Executive Director, Mineral Resources for approval by the end of May 2013; c) be prepared in accordance with any relevant DRE guideline; d) describe how the rehabilitation of the site would be integrated with: the implementation of the biodiversity offset strategy; and the final landform for the Boggabri coal mine; e) include detailed performance and completion criteria for evaluating the performance of the rehabilitation fo the site, and triggering remedial action (if necessary); f) describe the measures that would be implemented to ensure compliance with the relevant conditions of this approval, and address all aspects of rehabilitation including mine closure, final landform and final land use; g) include interim rehabilitation where necessary to minimise the area exposed for dust generation; h) include a program to monitor, independently audit and report on the effectiveness of the rehabilitation measures, and progress against the detailed performance and completion criteria; and i) build to the maximum extent practicable on the other management plans required under the	Submitted as a draft to DRE and DP&E in May 2013 Is in accordance with DRE guidelines The integration with Boggabri Coals landform to the north could be dealt with in more detail though it is in the MOP Completion Citeria are included, most are measurable. Closure is addressed as is compliance with other documentation Interim rehabilitation techniques and triggers are not described but are described in the Biodiversity MP. Monitoring and auditing is included					
Final Void Design and Closure The Proponent shall prepare and implement an updated Final Void and Mine Closure Plan (as a component of the overall Rehabilitation Management Plan required under condition 64 of schedule 3) to the satisfaction of the overall Rehabilitation Management Plan required under condition 64 of schedule 3) to the satisfaction of the Executive Director Mineral Resources, following consultation with the Director-General. A draft plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources of December 2019, and a final plan must be prepared and submitted to th	Project Approval 11-0047	64		Noted					
The Proponent shall prepare and implement an updated Final Void and Mine Closure Plan (as a component of the overall Rehabilisation Management Plan required condition 54 of Security Component of the overall Rehabilisation Management Plan required condition 54 of Security Component of the overall Rehabilisation Management Plan required condition 54 of Security Component of the Very Review of the Security Component of the University of the Security Component		Final Void Design and							
d) demonstrate that the proposed final landform: satisfies the relevant objecties in Table 15! Project Approval 11-0047 65 minimises the extent of any resulting pit lake; avoids salt scalding; maximises the capacity of emplaced spoil to drain to the natural environment; and			The Proponent shall prepare and implement an updated Final Void and Mine Closure Plan (as a component of the overall Rehabilitation Management Plan required under condition 64 of schedule 3) to the satisfaction of the Executive Director Mineral Resources, following consultation with the Director-General. A draft plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2019, and a final plan must be prepared and submitted to the Executive Director Mineral Resources by the end of December 2024. Each version of the plan must: a) be subject to independent review and verification by suitably qualified, experienced and independent person/s (including a groundwater expert) whose appointment has been approved by the Director-General; b) identify and consider: options for continued mining beyond current project life; interactions with the final landform of adjoining mines (including any direct or indirect interaction between final voids); opportunities for integrated mine planning with adjoining mines to minimise environmental impacts of the mines' final landforms; all reasonable and feasible landforms options for the final void (including filling); predicted stability of the proposed landforms; and predicted hydrochemistry and hydrogeology (including long-term groundwater recovery and void		Not Applicable				
AGRICULTURE			d) demonstrate that the proposed final landform: satisfies the relevant objecties in Table 151 minimises the extent of any resulting pit lake; avoids salt scalding; maximises the capacity of emplaced spoil to drain to the natural environment; and	Not yet required	Not Applicable				





Reference	Candition	Requirement	Fridance	Audia Finalis	Risk			
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk	
Project Approval 11-0047	66	The Proponent shall use its best endeavours to ensure that the agricultural productivity and production of non-operational project-related land is maintained or enhanced. Note: This includes properties primarily used for agriculture that are acquired by the Proponent due to noise and/or air quality impacts. However, it does not include land where disturbance is permitted udner the conditions of this approval or land that forms part of the biodiversity offset area.	Land is either leased back to origjinal owners or is let to lease with these conditions backed up in the lease agreement.	Compliant				
	SCHEDULE 4							
	ADDITIONAL PROCED	URES IDOWNERS/TENANTS						
Project Approval 11-0047	1	Within 3 months of the date of this approval, the Proponent shall: a) notify in writing the owners of: the land listed in Table 1 of schedule 3 tha they have the right to require the Proponent to acquire their land in accordance with the procedures in conditions 8 and 9 below at any stage during the project; any residence on the land listed in Table 1 of schedule 3 that they have the right to request the Proponent to ask for additional noise and/or air quality mitigation measures to be installed at their residence at any stage during the prokect; and any privately-owned land within 2 kilometres of the approach open-cut mining pit/s that they are entitles to ask for a property inspection, to establish the baseline condition of any buildings or structures on their land, or to have a previous property inspection report updated; b) notify the tenants of any mine-owned land of their rights under this approval; and c) send a copy of the NSW Health fact sheet entitled ""Mine Dust and You"" as may be updated from time to time) to the owners and/or existing tenants of any land (including mine-owned land) where the predictions in the EA identify that dist emissions generated by the project are likely to be greater than the relevant air quality criteria in schedule 3 at any time during the life of the project.	Letters to Callandar, Crosby, Laird, Suey, Picton, Wellwood, McCormack, Gabriel, Mainey, Angwin and Griffiths sighted.	Compliant				
Project Approval 11-0047	2	Prior to entering into any tenancy agreement for any land owned by the Proponent that is predicted to experience exceedences of the recommended dust and/or noise criteria, or for any of the land listed in Table 1 that is subsequently purchased by the Proponent, the Proponent shall: a) advise the prospective tenants of the potential health and amenity impacts associated with living on the land, and give them a copy of the NSW Health fact sheet entitled ""Mine Dust and You"" (as may be updated from time to time); b) advise the prospective tenants of the rights they would have under this approval; and c) request the prospective tenants consult their medical practitioner to discuss the air quality to the satisfavtion of the Director-General	Copies sighted.	Compliant				
Project Approval 11-0047	3 INDEPENDENT REVIEW	As soon as practicable after obtaining monitoring results showing: a) an exceedence of the relevant criteria in schedule 3, the Proponent shall notify the affected landowner in writing of the exceedence, and provide regular monitoring results to the landowner until the project in complying with the relevant criteria again; b) an exceedende of the relevant air quality criteria schedule 3, the Proponent shall send to the affected landowners and/or existing tenants of the land (including the tenants of any mine-owned land) a copy of: the NSW Health fact shee entititled ""Mine Dust and You" as may be updated from time to time); and the monitoring data, in an appropriate format so that a medical practitioner can assist the resident in making an informed decision on the health risks associated with occupation of the property.	Not applicable as there have been no exceedences at privately owned properties.	Not Applicable				
	INDEPENDENT REVIEV	T. C.	No such request received during the audit period	I	I	T T		
Project Approval 11-0047	4	If an owner or privately-owned and considers the project to be exceeding the criteria in schedule 3 at his/her land, then he/she may ask the Director-General in writing for an independent review of the impacts of the project on his/her land. If the Director-General is satisfied that an independent review is warranted, then within 2 months of the Director-General's decision, the Proponent shall: a) commission a suitably qualified, experienced and independent expert, whose appointment has been approved by the Director-General, to: consult with the landowner to determine his/her concerns; conduct monitoring to determine whether the project is complying with the relevant impact assessment criteria in schedule 3; and if the project is not complying with the relevant criteria, then: determine if the more than one mine is responsible for the exceedence, and if so the relative share of each mine towards the impact on the land; identify the measures that could be implemented to ensure compliance with the relevant criteria; and b) give the Director-General and landowner a copy of the independent review.		Not Applicable				





Reference	Condition	Requirement	Evidence	Audit Finding	Risk			
Reference	Condition				Consequence	Likelihood	Risk	
Project Approval 11-0047	5	If the independent review determines that the project is complying with the relevant criteria in schedule 3, then the Proponent may discontinue the independent review with the approval of the Director-General. If the independent review determines that the project is not complying with the relevant criteria, and that the project is primarily responsible for this non-compliance, then the Proponent shall: a) implement all reasonable and feasible mitigation measures, in consultation with the landowner and appointed independent expert, and conduct further monitoring until the project complies with the relevant criteria; or b) secure a written agreement with the landowner to allow exceedances of the relevant criteria, to the satisfaction of the Director-General. If the independent review determines that the project is not complying with the relevant acquisition criteria, and that the project is primarily responsible for this non-compliance, then upon receiving a written request from the landowner, the Proponent shall acquire all or part of the landowner's land in accordance with the procedures in conditions 8 and 9 below.	No such request received during the audit period	Not Applicable				
Project Approval 11-0047	6	If the independent review determines that the relevant criteria are being exceeded, but that more than one mine is responsible for this exceedance, then together with the relevant mine/s the Proponet shall: a) implement all reasonable and feasible mitigation measures, in consultation with the landowner and appointed independent expert, and conduct further monitoring until there is compliance with the relevant critetia; or b) secure a written agreement with the landowner and other relevant mine/s to allow exceedances of to the satisfaction of the Director-General. If the indepenent review determines that the procet is not complying with the relevant acquisition criteria in schedule 3, but that more than one mine is responsible for this non-compliance, then upon receiving a written request from the landowner, the Proponent shall acquire all or part of the landowner's land on as equitable a basis as possible with the relevant mine/s, in accordance with the procedures in conditions 8 and 9 below	No such request received during the audit period	Not Applicable				
	Biodiversity & Heritag	e e		II				
Project Approval 11-0047	7	If a person had good reason to believe the Proponent is not implementing the biodiversity and/or heritage conditions in schedule 3 satisfactorily, then he/she may ask the Director-General in writing for an independent review of the matter. If the Director-General is satisfied that an independent review is warranted, then within 2 months of the Director-General is satisfied that an independent review is warranted, then within 2 months of the Director-General is activately qualified, experienced and independent person, whose appointment has been approved by the Director-General, to: consult with the person and/or any relevant agencies; investigate the person's complaints/claims; review the environmental performance of the Proponent; determine whether the Proponent's performance is satisfactory or not; and if necessary recommend measures to improve the Proponent's performance; and b) give the Director-General and complainant a copy of the indepenent review.	No such request received during the audit period	Not Applicable				
	LAND ACQUISITION	1						
Project Approval 11-0047	8	Within 3 months of receiving a written request from a landowner with acquisition rights, the Proponent shall make a binding written offer to the landowner based on: a) the current market value of the landowner's interest in the land at the date of the written request, as if the land was unaffected by the project, having regard to the: existing and permissible use of the land, in accordance with the applicable planning instruments at the date of the written request; and presence of improvements on the land and/or any approved building or structure which has been physically commenced at the date of the landowner's written request, and is due to be completed subsequent to that date, by excluding any improvements that have resulted from the implementation of the additional mitigation measures required under condition 2 of schedule 3; b) the reasonable costs associated with: relocating within the Tamworth, Narrabri, Gunnedah or Moree local government areas, or to any other local government area as agreed by the Director-General; and obtaining legal advice and expert advice for determining the acquisition price of the land, and the terms upon which it is to be acquired; and	No such request received during the audit period	Not Applicable				





Deference	Condition	Requirement	Eddone	Audia Findina	Risk			
Reference	Condition		Evidence	Audit Finding	Consequence	Likelihood	Risk	
Project Approval 11-0047	8	c) reasonable compensation for any disturbance cause by the land acquisition process. However, if the Proponent and landowner cannot agree on the acquisition price of the land and/or the terms upon which the land is to be acquired, then either party may refer the matter to the Director-General for resolution. Upon receiving such a request, the Director-General shall request the President of the NSW Division of the Australian Property Institute to appoint a qualified independent valuer to: consider submissionions from both parties; determine a fair and reasonable acquisition price for the land and/or the terms upon which the land is to be acquired, having regard to the matters referred to in paragraphs (a)-(c) above; prepare a detailed report setting out the reasons for any determination; and provide a copy of the report to both parties Within 14 days of receiving the independent valuer's report, the Proponent shall make a binding written offer to the landowner to purchase the land at a price not less than the independent valuer's determination. However, if either party disputes the independent valuer's determination, then within 14 days of receiving the independent valuer's report, they may refer the matter to the Director-General for review. Any request for a review must be accompanies by a detailed report setting out the reasons why the party disputes the independent valuer's determination. Following consultation with the independent valuer and both parties, the Director-General will determine a fair and reasonable acquisition price for the land, having regard to the matters referred to in paragraphs (a)-(c) above, the independent valuer's determination and any other relevant submissions. Within 14 days of this determination, the Proponent shall make a binding written offer under this condition within 6 months of the offer being made, then the Proponent's obligations to acquire the land shall cease, unless the Director-General determines otherwise.	No such request received during the audit period	Not Applicable				
Project Approval 11-0047	9	The Proponent shall pay all reasonable costs associated with the land acquisition process described in condition 5 above, including the costs associated with obtaining Council approval for any plan of subdiviion (where permissible), and registration of this plan at the Office of the Registrar-General.	No such request received during the audit period	Not Applicable				
	SCHEDULE 5 ENVIRONMENTAL MA	NAGEMENT, REPORTING AND AUDITING						
	ENVIRONMENTAL MA	NAGEMENT						
	Environmental Manag			T	1			
Project Approval 11-0047	1	the satisfaction of the Director-General. The strategy must: a) be submitted to the Director-General for approval by the end of May 2013; b) provide the strategic framework for environmental management of the project; c) identify the statutory approvals that apply to the project; d) describe the role, responsibility, authority and accountability of all key peronnel involved in the environmental management of the project; e) describe the procedures that would be implemented to: keep the local community and relevant agencies informed about the operation and environmental performance of the project; revelve, record, handle, and respond to complaints; resolve any disputes that may arise during the course of the project;	The copy provided was current and met all requirements	Compliant				
		respond to any non-compliance; respond to emergencies; and f) include: f) include: copies of any strategies, plan and programs approved under the conditions of this approval; and a clear plan depicting all the monitoring to be carried out in relation to the project.						
	Adaptive Managemen	respond to emergencies; and f) include: copies of any strategies, plan and programs approved under the conditions of this approval; and a clear plan depicting all the monitoring to be carried out in relation to the project.	No such avendances					
Project Approval 11-0047	Adaptive Managemen	respond to emergencies; and f) include: copies of any strategies, plan and programs approved under the conditions of this approval; and a clear plan depicting all the monitoring to be carried out in relation to the project.	No such exceedences	Compliant				





Reference	Condition	Requirement	Evidence	Audit Finding	Risk			
Reference	Condition			Audit Finding	Consequence	Likelihood	Risk	
Project Approval 11-0047	3	The Proponent shall ensure that the management plans required under this approval are prepared in accordance with any relevant guideleines, and include: a) detailed baseline data; b) a description of: the relevant statutory requirements (including any relevant approval, licence or lease conditions); any relevant limits or performance measures/criteria; the specific performance indicators that are proposed to be used to judge the performance of, or guide the implementation of, the development or any management measures c) a description of the measures that would be implemented to comply with the relevant statutory requirements, limits, or performance measures/criteria; d) a program to monitor and report on the: impacts and environmental performance of the project; effectiveness of any management measures (see c above); e) a contingency plan to manage any unpredicted impacts and their consequences; f) a program to investigate and implement ways to improve the environmental performance of the project over time; g) a protocol for managing and reporting any: incidents; compliants; non-compliances with statutory requirements; and exceedences of the impact assessment criteria and/or performance criteria; and h) a protocol for periodic review of the plan.	a) Detailed baseline data not included in some of the draft plans. The only approived plan under the new EA is the CHMP which has adequate background info. Recommendation to revise draft plans to include adequate background information.	Compliant				
	Annual Review			ļ				
Project Approval 11-0047	4	By the end of June each year (or as otherwise agreed by the Director-General), the Proponent shall review the environmental performance of the project to the previous calendar year to the satisfaction of the Director-General. This review must: a) describe the development (including any rehabilitation) that was carried out in the past calendar year, and the development that is proposed to be carried out over the current calendar year; b) include a comprehensive review of the monitoring results and complaints records of the project over the past year, which includes a comparison of these results against the: relevant statutory requirements, limits or performance measures/criteria; monitoring results of previous years; and relevant predictions in the EA; c) identify any non-compliance over the last year, and describe what actions were (or are being) taken to ensure compliance; d) identify any trends in the monitoring data over the life of the project; e) identify any discrepancies betweem the predicted and actual impacts of the project, and analyse the potential cause of any significant discrepancies; and f) describe what measures will be implemented over the next year to improve the environmental performance of the project.	The 20012-13 AEMR contains all the requirements listed and addresses the requirements adequately.	Compliant				
	Revision of Strategies			1				
Project Approval 11-0047	5	Within 3 months of the submission of an: a) annual review under condition 4 above; b) incident report under condition 8 below; c) audit under condition 10 below; or d) any modification to the conditions of this approval, the Proponent shall review, and if necessary revise, the strategies, plans, and programs required under this approval to the satisfaction of the Director-General. Note: This is to ensure the strategies, plans and programs are updated on a regular basis, and incorporate any recommended measures to improve the environemental performance of the project.	The EMPs and Strategies have been revised on a regular basis and within the short audit period. Thus it is difficult to determine whether they were revised as a result of the above circumstances or for other reasons. Recommendation - Tarawonga should develop a method for documenting reviews so that if there are no changes resulting from the review there is evidence of the review being conducted.	Compliant				
	Management of Cumi		None of the Least Ctate Forest strategies that formulate cumulative					
Project Approval 11-0047	6	In conjunction with the owners of the nearby mines in the Leard Forest Mining Precinct, the Proponent shall use its best endeavours to minimise the cumulative impacts of the project on the surrounding area, to the satisfaction of the Director-General.	None of the Leard State Forest strategies that formulate cumulative impact mitigation have been approved. They are all in draft.	Not Applicable				
	Community Consultat	ive Committee The Proponent shall establish and operate a Community Consultative Committee (CCC) for the project to	Minutes on website					
Project Approval 11-0047	7	the satisfaction of the Director-General. This CCC must be operated in general accordance with the Guidelines for Establishing and Operating Community Consultative Committees for Mining Projects (Department of Planning, 2007, or its latest version), and be operating by the end of May 2013. The CCC must seek to include joint membership with CCCs for other operating coal mines within the Leard Forest Mining Precinct, unless otherwise agreed by the Director-General. Notes: The CCC is an advisory committee. The Department and other relevant agencies are responsible for ensuring that the Proponent complies with this approval. In accordance with the Department's guideline, the CCC should be comprised on an independent chair and appropriate representation from the Proponent, Council and the local community		Compliant				
	REPORTING							





Reference	Condition	Requirement Evidence		Audit Finding	Risk				
Reference			Evidence	Addit Finding	Consequence	Likelihood	Risk		
	Incident Reporting								
		The Proponent shall notify, at the earliest opportunity, the Director-General and any other relevant	No incidents reported						
		agencies of any incident that has cause, or threatens to cause, material harm to the environment. For any other incident associated with thte project, the Proponent shall notify the Director-General and any							
Project Approval 11-0047	8	other relevant agencies as soon as practible after the Proponent becomes aware of the incident. Within		Not Applicable					
		7 days of the date of the incident, the Proponent shall provide the Director-general and any relevant							
		agencies with a detailed report on the incident, and such further reports as may be requested.							
	Regular Reporting			T					
Project Approval 11-0047	9	The Proponent shall provide regular reporting on the environmental performance of the project on its website, in accordance with the reporting arrangements in any plan or programs approved under the	Noted						
rioject Approvai 11-0047	,	conditions of this approval.							
	AUDITING								
	Independent Environ		I	T	1		Г		
		"10 By the end of June 2014 and every 3 years thereafter, unless the Director-General directs otherwise,	This audit, non-compliant with timing requirements						
		the Proponent shall commission and pay the full cost of an indepenent Environmental Audit of the project. This audit must:							
		a) be conducted by a suitable qualified, experiened and independent team of experts whose							
		appointment has been endorsed by the Director-General;							
		b) include consultation with the relevant agencies;							
		c) assess the environmental performance of the project and assess whether it is complying with the							
		requirements in this approval, and any other relevant approvals, relevant EPL/s and/or Mining Lease/s							
		(including any assessment, plan or program required under these approvals);							
		d) assess whether the Proponent is implementing best noise, blasting and air quality management practice:		1					
Project Approval 11-0047	10	e) investigate and report on the measures taken to minimise the noise and air quality impacts of the		Not Compliant					
7 ,,		project during meterological conditions and/or extraordinary events when the relevant noise and air		Administrative					
		quality limits in this approval do not apply, including:							
		the effectiveness of these measures in maintaining impacts within the relevant criteria in this approval							
		and/or the limits in the relevant EPL; and							
		any additional measures available to mitigate impacts under such conditions;							
		f) review the adequacy of any approved strategy, plan or program required under the abovementioned							
		approvals; and g) recommend measures or actions to imprive the environmental performance of the project and/or any							
		strategy, plan or program required under these approvals.							
		"							
		Note: This audit team must be led by a suitably qualified auditor, and include experts in noise, air quality, water,	Noted						
Project Approval 11-0047	10	ecology, and any other fields specified by the Director-General.							
		Within 3 months of commissioning this audit, or as otherwise agreed by the Director-General, the	This audit, non-compliant with timing requirements						
Project Approval 11-0047	11	Proponent shall submit a copy of the audit report to the Director-General, together with its response to		Not Compliant					
		any recommendations contained in the audit report.		Administrative					
	ASSESS TO INFORMAT			,			,		
		The Proponent shall:	see website, all this information is available						
		a) within 3 months of the date of this approval, make the following information publicly available on its website:							
		website: the EA;		1					
		all current statutory approvals for the project;		1					
		approved strategies, plans and programs required under the conditions of this approval;		1					
		a comprehensive summary of the monitoring results of the project, which have been reported in		1					
Broject Approval 11 0047	12	accordance with the various plans and programs approved under the conditions of this approval;		Compliant					
Project Approval 11-0047	12	a complaints register, which is to be updated on a monthly basis;		Compliant					
		minutes of CCC meetings;							
		the last five annual reviews; any independent environmental audit, and the Proponent's response to the recommendations in any							
		audit;							
		any other matter required by the Director-General; and							
		b) keep this information up to date, to the satisfaction of the Director-General.							
	On-line Communicati	on of Onsite Activities and Monitoring of Noise and Air Quality							
		The Proponent shall, within 3 months of the date of this approval:	Checked website, all present but a longer term daily check by the AQ						
		a) make the following information for the project publicly available on its website, on a daily basis and in		1					
		a clearly understandable form:	every day.	1					
	1	daily weather forecasts for the coming week;		1					
		proposed operational responses to these weather forecasts;		Not Compliant					
Project Approval 11-0047	12	real-time noise and air quality monitoring data (subject to any necessary cayeats), and		1	1		l .		
Project Approval 11-0047	13	real-time noise and air quality monitoring data (subject to any necessary caveats); and any operational responses that were taken in response to the noise and air quality monitoring data, and		Administrative					
Project Approval 11-0047	13	real-time noise and air quality monitoring data (subject to any necessary caveats); and any operational responses that were taken in response to the noise and air quality monitoring data, and b) make provision on its website for the provision of on-line and/or email comments by members of the		Administrative					
Project Approval 11-0047	13	any operational responses that were taken in response to the noise and air quality monitoring data, and		Administrative					
Project Approval 11-0047	13	any operational responses that were taken in response to the noise and air quality monitoring data, and b) make provision on its website for the provision of on-line and/or email comments by members of the		Administrative					





Reference	Clause	Requirement	Evidence	Audit Finding	Risk						
Kelefelice	Clause			Addit Finding	Consequence	Likelihood	Risk				
		ect Environmental Assessment - Statement of Commitments									
	SOC1.1 Proposed Project Environmental Management, Monitoring and Reporting										
TC EA Statement of Commitments	1.1	TCPL will review and revise the existing Tarrawonga Coal Mine management and monitoring plans listed in Table SOC-1. Table SOC-1 also lists new management and monitoring plans that are proposed to be prepared for the Project.	SOC-1 includes: Management and Monitoring Environmental Management Strategy Water Management Plan Site Water Balance Erosion and Sediment Control Plan Surface Water Monitoring Program Groundwater Monitoring Program Surface and Groundwater Response Plan Goonbri Creek Management Plan# Noise Management Plan Blast Management Plan# Noise Management Plan Blast Management Plan Blodiversity Offset Strategy Biodiversity Management Plan# (new management plan) Offset Area Management Plan# (new management plan) Farm Management Plan# (new management plan) Aboriginal Heritage Management Plan Waste Management Plan Behabilitation Strategy Rehabilitation Management Plan Reporting Requirements Annual Environmental Management Report and Mining Operations Plan or Rehabilitation and Environmental Management Plan Uicences and Approvals Greenhouse Gas Reporting Note most of these plans have been revised but are not yet approved by the DG.	Compliant							
TC EA Statement of Commitments	1.1	The existing monitoring program at the Tarrawonga Coal Mine will be augmented to address additional Project disturbance areas and the open cut extensions.	Noted	Noted							
TC EA Statement of Commitments	1.1	Environmental management, monitoring and reporting will be conducted in accordance with finalised Project Approval conditions, with the final monitoring details (locations, parameters and frequencies) to be provided in the relevant management plans/monitoring programs.	Noted	Noted							
	SOC1.2 Specific Enviro	onmental Commitments									
TC EA Statement of Commitments	1.2	Environmental management and offset measures to be implemented for the Project are described in Section 4. Key commitments include: • design and construction of an engineered low permeability barrier to the east and south-east of the open cut;	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. A low permeability barrier would be constructed in the alluvium, approximately 50-100 m to the east and south-east of the final open cut extent. Construction of the low permeability barrier would be completed before the open cut intersects the alluvium that feeds Goonbri Creek (approximately Year 12).	Not Applicable							





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2		Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. In approximately Year 15 of the approved works, open cut mining would progress through a 3 km section of Goonbri Creek. Prior to the open cut advancing into this section of the creek, the permanent Goonbri Creek alignment would be established to the east of the open cut. The open cut has not advanced towards Goonbri Crk enough to trigger this requirment	Not Applicable			
TC EA Statement of Commitments	1.2	integration of key aspects of the Project with the adjoining Boggabri Coal Mine (i.e. Northern Emplacement, coal processing and loading of Project product coal onto trains);	MOP and Project Approval PA 11_0047	Noted			
TC EA Statement of Commitments	1.2	Whitehaven Coal Handling and Preparation Plant (once suitable approvals and upgrades are in place);	MOP In accordance with Project Approval PA 11_0047 Schedule 2 Condition 7 road transport of ROM coal to the Whitehaven CHPP may continue for up to three months after the commissioning of the Boggabri Infrastructure Facility upgrades. This is not tiggered as the rail loop is not completed.	Not Applicable			
TC EA Statement of Commitments	1.2		Noise Management Plan (June 2014) includes management and mitigation of operational noise	Compliant			
TC EA Statement of Commitments	1.2	rehabilitation of Project disturbance areas, including the reinstatement of key agricultural and ecological values;	Rehabilitation Management Plan (2013)	Compliant			
TC EA Statement of Commitments	1.2	provision of biodiversity offset measures for the Project;	BMP (2013)	Compliant			
TC EA Statement of Commitments	1.2	management of the Project final void to minimise potential long- term impacts on water resources; and	Rehabilitation Management Plan (2013)	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	participation in joint air quality, operational noise and regional groundwater monitoring schemes with the adjoining Boggabri Coal Mine and the Maules Creek Coal Project.	AQMP (2013) states: TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in relation to developing an Air Quality Management Strategy for the Boggabri – Tarrawonga – Maules Creek (BTM) Complex incorporating cumulative air quality monitoring and management, as required under the conditions of consent for the Tarrawonga Coal Mine. The Strategy has been submitted to the Department of Planning and Environment (DPE) for approval. Once approved, the Strategy will be appended to this Plan. The Strategy has been drafted but is not yet approved.	Not Applicable			
	Low Permeability Barr						
TC EA Statement of Commitments	1.2	A low permeability barrier will be constructed in the alluvium to the east and south-east of the open cut. Construction of the low permeability barrier will be completed before the Project open cut intersects the alluvium (approximately Year 12).	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. A low permeability barrier would be constructed in the alluvium, approximately 50-100 m to the east and south-east of the final open cut extent. Construction of the low permeability barrier would be completed before the open cut intersects the alluvium that feeds Goonbri Creek (approximately Year 12). Not triggered yet	Not Applicable			
TC EA Statement of Commitments	1.2	The design objectives of the low permeability barrier include minimising the potential for drainage of alluvial groundwater into the open cut during operations and post-mining, and maintaining the hydraulic character of Goonbri Creek.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure.	Not Applicable			
TC EA Statement of Commitments	1.2	TCPL commits to construction of the low permeability barrier to meet the following design objectives: • minimise the potential for local drainage of alluvial groundwater into the open cut during operations and post-mining; • minimise the potential for future instability of the open cut batters formed in the alluvium; • maintain the hydraulic character of Goonbri Creek by minimising the potential loss of baseflow; and • maintain the value of alluvial groundwater, by minimising potential interactions with the mine final void, post-mining.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure.	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	In addition, TCPL will augment the existing piezometer network with additional sites to validate the performance of the low permeability barrier.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure: The existing TCM network of piezometers will be augmented, particularly prior to and during Years 12 to 17 of the extension and remain in place for 2 years post-mining.	Not Applicable			
	Permanent Goonbri	Creek Alignment and Associated Flood Bund					
TC EA Statement of Commitments	1.2	In approximately Year 15, open cut mining would remove a 3 kilometre (km) section of Goonbri Creek. Prior to the open cut advancing into this section of the creek, the permanent Goonbri Creek alignment will be established.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	A permanent flood bund will also be constructed to prevent inundation of the open cut during operations and postmining. The permanent flood bund will generally coincide with the alignment of the low permeability barrier.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	TCPL commits to the design, construction and implementation of the permanent Goonbri Creek alignment to meet the following design objectives: • construct a low flow channel that approximates the existing section of Goonbri Creek upstream of the Project in terms of stream geometry, hydrology and geomorphology;	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	mimic the meandering path of the existing alignment of Goonbri Creek, such that the length of the permanent Goonbri Creek alignment is approximately the same length as the section of Goonbri Creek being removed;	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	minimise the disturbance to the reaches of Goonbri Creek upstream of the permanent Goonbri Creek alignment; and	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	provide a stable transition back to the existing Goonbri Creek alignment which results in no detectable change to the hydraulic conditions in the reaches of Goonbri Creek or the Bollol Creek floodplain area downstream.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	In addition, TCPL commits to the design and construction of the permanent flood bund to a height that will provide protection against the peak flood height associated with a Probable Maximum Precipitation rainfall event.	Tarrawonga Water Management Plan (May 2013, Draft) includes this measure. Not triggered	Not Applicable			





Deference	Clause	Requirement	Fridance	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	TCPL will develop and implement a Goonbri Creek Management Plan prior to the commencement of construction activities associated with the low permeability barrier, permanent Goonbri Creek alignment and flood bund.	MOP 2013 Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	The Goonbri Creek Management Plan will describe: • the design and construction details of the permanent Goonbri Creek alignment and flood bund;	MOP 2013 Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	revegetation objectives and activities;	MOP 2013 Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	water quality, ecological, hydrological and geomorphic performance and completion criteria for the permanent Goonbri Creek alignment based on baseline conditions; and	MOP 2013 Not triggered	Not Applicable			
TC EA Statement of Commitments	1.2	a monitoring/maintenance program for water quality, ecological, hydrological and geomorphic integrity of the permanent Goonbri Creek alignment.	MOP 2013 Not triggered	Not Applicable			
	Management of Oper	ational Noise					
TC EA Statement of Commitments	1.2	TCPL will implement the following noise management and mitigation measures to appreciably reduce noise emissions associated with the Project: • installation of an earth bund on the southern side of exposed sections of the services corridor (i.e. ROM coal haul road to the Boggabri Coal Mine);	Noise Management Plan 2014 states: Installation of an earth bund on the southern side of exposed sections of the services corridor (ROM coal haul road to the Boggabri Coal Mine). This will not occur over the next 2 year term as it is reliant on construction of infrastructure at Boggabri Coal.	Not Applicable			
TC EA Statement of Commitments	1.2	modified alignment of haul routes to reduce their exposure relative to nearby receivers; and	Noise Management Plan 2014 states: The key component of this strategy is to reduce exposure from the dump trucks hauling to the northern emplacement via a haul road located at the southern end of the emplacement area. A northern haul road is now also available, so in circumstances where noise levels approach compliance criteria, an alternate haul road is available. This northern haul road will also become more relevant as the northern extension area is developed.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Addit Filldling	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	a reduction in the number of mobile fleet items operating during the evening and night-time periods.	Noise Management Plan 2014 states: The reduction in operating fleet would be based on actual noise measurement, preliminary review of site operations to determine if alternate operating locations are available, followed by stand down of equipment if necessary. This was verified duringthe site inspection and in interview with several employees including the Env Officer and Mine Manager.	Compliant			
	Rehabilitation Object	tives and Final Landform					
TC EA Statement of Commitments	1.2	The Project final landform and revegetation program will provide for a combination of approximately 752 ha of native woodland/forest and some 210 ha of Class 3 agricultural suitability land.	Rehabilitation Management Plan (2013) includes this Medium to Long Term Rehabilitation Objective	Noted			
TC EA Statement of Commitments	1.2	The agricultural land will be capable of being used for pasture production for grazing and occasional cropping. Revegetation of woodland/forest areas will include the planting of species characteristic of the local vegetation communities, including species from the Box-Gum Woodland endangered ecological community.	Rehabilitation Management Plan (2013) Not yet established, landform has not progressed to the point where the propsed areas area ready for rehab	Not Applicable			
TC EA Statement of Commitments	1.2	In addition, TCPL commits to a riparian vegetation enhancement program on a 3.2 km section of Goonbri Creek downstream of the Project open cut, through measures such as revegetation and stock exclusion.	Rehabilitation Management Plan (2013) does not mention commitment on a 3.2 km. It states: Biodiversity enhancement will be accomplished by native vegetation regeneration along riparian corridors and within remnant native vegetation patches. The work has not commenced, confirmed during site interview. Given the life of the current approval is throught to 2030, there is still time to complete this work.	Not Applicable			
TC EA Statement of Commitments	1.2	A Rehabilitation Management Plan will be developed and implemented for the Project, including a rehabilitation monitoring program designed to track the progress of rehabilitation and revegetation.	Rehabilitation Management Plan (2013)	Compliant			
	Biodiversity Offset N						
TC EA Statement of Commitments	1.2	TCPL commits to the provision of an area to offset the residual impacts of the Project on flora and fauna and maintain or improve the biodiversity values of the region in the medium to long-term.	BMP (2013)	Compliant			
TC EA Statement of Commitments	1.2	The biodiversity offset for the Project comprises approximately 1,600 ha of freehold land that has been purchased by Whitehaven.	BMP (2013)	Compliant			
TC EA Statement of Commitments	1.2	The offset is situated approximately 20 km to the north-east of the Project and adjoins Mount Kaputar National Park (Figure SOC-2). Prior to its recent purchase by Whitehaven the offset area was part of a larger agricultural property.	, ,	Compliant			





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	Ecological gains from the biodiversity offset include: • Similar vegetation communities/fauna habitats, compared to the Project area, will be conserved/enhanced in the biodiversity offset area.	Noted	Noted			
TC EA Statement of Commitments	1.2	The biodiversity offset area is suitably located to benefit flora and fauna populations (biodiversity values) potentially impacted by the Project.		Noted			
TC EA Statement of Commitments	1.2	The biodiversity offset area is located adjacent to Mount Kaputar National Park.	Noted	Noted			
TC EA Statement of Commitments	1.2	Ephemeral creeks occur within the biodiversity offset area, providing a diversity of habitats.	Noted	Noted			
TC EA Statement of Commitments	1.2	Substantial areas of Box-Gum Woodland (232 ha) occur in the biodiversity offset area.	Noted	Noted			
TC EA Statement of Commitments	1.2	Through active management, particularly of areas previously cleared for agriculture, the ecological values of the biodiversity offset area can be further improved. TCPL commits to a number of management measures to enhance the offset area's flora and fauna values. These measures will be detailed in the Offset Area Management Plan to be prepared for the Project.	Offset Area Management Plan as a separate doc	Compliant			
TC EA Statement of Commitments	1.2	The Offset Area Management Plan will also include a program to monitor and audit the effectiveness of the management measures and to evaluate performance against specified completion criteria.	Biodiversity Management Plan (May 2013) states: The monitoring results will be assessed and utilised in the continual improvement and refinement of rehabilitation/revegetation techniques and will be documented as part of the Annual Environmental Review Report. Tarrawonga Biodiversity Moniotoring Report Vols 1 & 2 - 2014 - EcoLogical	Compliant			
TC EA Statement of Commitments	1.2	TCPL intends to reach an agreement with the New South Wales (NSW) Government so that the biodiversity offset area can be permanently added to the adjoining Mount Kaputar National Park.	Biodiversity Management Plan (May 2013) - no information on this issue included in the BMP 2013.	Not Applicable			
TC EA Statement of Commitments	1.2	In the interim, TCPL will enter into a conservation arrangement with the NSW Government to ensure the protection and management of the offset area (e.g. a voluntary conservation agreement with the NSW Minister for the Environment).	Biodiversity Management Plan (May 2013) - no information on this issue included in the BMP 2013.	Not Applicable			
	Management of the	Project Final Void					
TC EA Statement of Commitments	1.2	TCPL commits to installing permanent perimeter bunds and/or diversion channels to limit the catchment area of the final void.	Rehabilitation Management Plan 2013 Not yet required	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Findin		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
TC EA Statement of Commitments	1.2	· · · · · · · · · · · · · · · · · · ·	Rehabilitation Management Plan 2013 Not yet required	Not Applicable			
TC EA Statement of Commitments	1.2	, , , , , , , , , , , , , , , , , , , ,	Rehabilitation Management Plan 2013 Not yet required	Not Applicable			
	Participation in Joint A	ir Quality, Operational Noise and Regional Groundwater Monitoring					
TC EA Statement of Commitments	1.2	Noise Management Plan and Water Management Plan.	TCM has been in consultation with the nearby Boggabri Coal mine and Maules Creek Project in developing a Leard	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk		
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk	
		nd Water and Applications to Land						
	P1 Location of monito	pring/discharge points and areas			ı			
EPL 12364	Preamble	Fee Based Activity Scale Coal works 0-2,000,000 T handled Mining for coal > 500,000-2,000,000 T produced	2013-14 AEMR identified 2,136,045t sent to Narrabri CHPP.	Not Compliant Administrative				
EPL 12365	P1.1	The following points referred to in the table below are identified in this licence for the purposes of monitoring and/or the setting of limits for the emission of pollutants to the air from the point. Air	AEMR/Annual Review 2012/2013, Section 3.1	Compliant				
EPL 12365	P1.2	The following points referred to in the table are identified in this licence for the purposes of the monitoring and/or the setting of limits for discharges of pollutants to water from the point.	Annual Return	Compliant				
EPL 12365	P1.3	The following utilisation areas referred to in the table below are identified in this licence for the purposes of the monitoring and/or the setting of limits for any application of solids or liquids to the utilisation area. **Water and land** EPA Identification no. 1	Annual Return	Compliant				





Reference	Clause	Requireme	nt	Evidence	Audit Finding		Risk		
Reference	Clause				Addit Fillding	Consequence	Likelihood	Risk	
		5 Ambient water quality monitoring	Bollol Creek upstream of discharge from premises labelled "BC-U" on Figure 4: Proposed Water Discharge Points and Weather Station submitted with licence variation application form 13 October 2008.	Annual Return	Annual Return				
		6 Ambient water quality monitoring	Bollol Creek downstream of discharge from premises labelled "BC-D" on Figure 4: Proposed Water Discharge Points and Weather Station submitted with licence variation application form 13 October 2008.						
		7 Ambient water quality monitoring	Nagero Creek upstream of discharge from premises labelled "NC-U "on Figure 4: Proposed Water Discharge Points and Weather Station submitted with licence variation application form 13 October 2008.						
EPL 12365	P1.3	8 Ambient water quality monitoring P1.3	Nagero Creek downstream of discharge from premises labelled "NG-D "on Figure 4: Proposed Water Discharge Points and Weather Station submitted with licence variation application form 13 October 2008.		Compliant				
		9 Groundwater monitoring	Groundwater monitoring bore located on property "Thuin" labelled MW-1 on Figure 5 Water Monitoring Locations in the Site Water Management Plan submitted with licence application 17 November 2005						
		10 Groundwater monitoring	Groundwater monitoring bore located on property "Bollol Ck Station" labelled MW-2 on Figure 5 Water Monitoring Locations in the Site Water Management Plan submitted with licence application 17 November 2005						
		11 Groundwater monitoring	Groundwater monitoring bore located on property "Nagero" labelled MW-3 on Figure 5 Water Monitoring Locations in the Site Water Management Plan submitted with licence application 17 November 2005						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
		12 Groundwater monitoring Groundwater monitoring bore located on property "Turnwonga" labelled MW-4 on Figure 5 Water Monitoring Locations in the Site Water Management Plan submitted with licence application 17 November 2005	Annual Return		-		
		13 Surface water quality Mining void (variable location) labelled MV1 on Figure 5 Water Monitoring Ucations in the Site Water Monitoring Locations in the Site Water Management Plan submitted with licence application 17 November 2005					
		24 Wet weather discharge Wet weather discharge Spillway on Storage Dam 16 located on the southern side of the					
		Discharge water quality premises labelled "SD16" on monitoring monitoring premises labelled "SD16" on "Figure 1: Existing and Proposed Water Discharge Monitoring Points" submitted with licence variation application form					
EPL 12365	P1.3	25 Wet weather discharge Wet weather discharge Spillway on Sediment Basin 22 Dischagre water quality Dischagre water quality monitoring monitoring water quality monitoring stabelled "5822" in Floure titled "Tarawonga Mine Site. Licensed Discharge Points" submitted with licence variation application form dated 19 August 2011.		Compliant			
		26 Wet weather discharge Wet weather discharge Spillway on Sediment Basin 23 loscharge water quality Discharge water quality monitoring monitoring monitoring water quality monitoring water quality monitoring water quality monitoring water quality water quality water wat					
		27 Wet weather discharge Discharge water quality monitoring Wet weather discharge Spillway on Sediment Basin 18 located on northern side of primary monitoring monitoring premises labelled "S818" in Figure titled "Tarrawonge Mine Site. Licensed Discharge Points" submitted with licence variation application form dated 19 August 2011.					
			1515/1				
		The following point(s) in the table are identified in this licence for the purpose of the monitoring of weather parameters at the point.	AEMR/Annual Review 2012/2013, Section 3.22				
		EPA Identification number Type of Monitoring Point Description of Location	5.22				
		EPA Identification number Type of Monitoring Point Description of Location					
EPL 12365	P1.4	W1 Weather analysis Vesitive tation located on "Templemore" labelled "Templemore" labelled "Templemore labelled "Templemore labelled" on Figure submitted with licence variation application form received by EPA on 12 June 2012 (DOC12/23618).		Compliant			
	3. Limit Conditions						
	L1 Pollution of waters	le un un barron de la dela constante de la con	AFAM /A I D				
EPL 12365	L1.1	Except as may be expressly provided in any other condition of this licence, the licensee must comply with section 120 of the Protection of the Environment Operations Act 1997.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	L2 Concentration Limi						
EPL 12365	L2.1	For each monitoring/discharge point or utilisation area specified in the table\s below (by a point number), the concentration of a pollutant discharged at that point, or applied to that area, must not exceed the concentration limits specified for that pollutant in the table.		Noted			
EPL 12365	L2.2	Where a pH quality limit is specified in the table, the specified percentage of samples must be within the specified ranges.		Noted			
EPL 12365	L2.3	To avoid any doubt, this condition does not authorise the pollution of waters by any pollutant other than those specified in the table\s.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
EPL 12365	L2.4	Water and/or Land Concentration Limits POINT 1,2,3,24,25,26,27 Pollutant Units of Measure 50 percentile concentration limit 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L2.5	The Total Suspended Solids concentration limits specified for Points 1, 2, 3, 24, 25, 26 and 27 may be exceeded for water discharged provided that: (a) the discharge occurs solely as a result of rainfall measured at the premises that exceed 38.4 millimetres over any consecutive 5 day period immediately prior to the discharge occurring; and (b) all practical measures have been implemented to dewater all sediment dams within 5 days of rainfall such that they have sufficient capacity to store run off from a 38.4 millimetre, 5 day rainfall event. Note: 38.4 mm equates to the 5 day 90%ile rainfall depth for Gunnedah sourced from Table 6.3a Managing Urban Stormwater: Soils and Construction Volume 1: 4th edition, March 2004.	ls	Noted			
	L3 Waste						
EPL 12365	L3.1	The licensee must not cause, permit or allow any waste generated outside the premises to be received at the premises for storage, treatment, processing, reprocessing or disposal or any waste generated at the premises to be disposed of at the premises, except as expresipermitted by the licence.	r TABLE A3.2 Compliance Review –	Compliant			
EPL 12365	L3.2	This condition only applies to the storage, treatment, processing, reprocessing or dispose of waste at the premises if those activities require an environment protection licence.	I AEMR/ 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L3.3	Reject material from the Whitehaven CHPP can be disposed of at the premises in accordance with the disposal method outlined in the Environmental Impact Statement titled "East Boggabri Joint Venture, Environmental Impact Statement for the Proposed Ea Boggabri Coal Mine, May 2005" prepared by R.W. Corkery & Co. Pty. Limited dated May 2005, or as otherwise approved by the EPA.	Noted	Noted			
	L4 Noise Limits	Main annual at the annual act the second and a second the second fine in the table below.	AFAAD 2012/2012 Amaaadii: 2				
EPL 12365	L4.1	Noise generated at the premises must not exceed the noise limits in the table below. Locality and Day- LAeq (15 Evening- LAeq (15 Night- Night	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. AEMR/Annual Review 2012/2013, Section 3.10.3.2 J une 2012 Noise Monitoring - Attended. AEMR/Annual Review 2012/2013, Section 3.10.3.4 September 2012 Noise Monitoring – Attended	Not Compliant	D	3	Medium
EPL 12365	L4.2	For the purpose of the table above: a) Day is defined as the period from 7am to 6pm Monday to Saturday and 8am to 6pm Sundays and Public Holidays; b) Evening is defined as the period from 6pm to 10pm; c) Night is defined as the period from 10pm to 7am Monday to Saturday and 10pm to 8al Sundays and Public Holidays.	Noted n	Noted			





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	L4.3	To determine compliance: a) with the Leq(15 minute) noise limits in the Noise Limits table, the noise measurement equipment must be located: i) approximately on the property boundary, where any dwelling is situated 30 metres or less from the property boundary closest to the premises; or ii) within 30 metres of a dwelling façade, but not closer than 3m, where any dwelling on the property is situated more than 30 metres from the property boundary closest to the premises; or, where applicable iii) within approximately 50 metres of the boundary of a National Park or a Nature Reserve. b) with the LA1(1 minute) noise limits in the Noise Limits table, the noise measurement equipment must be located within 1 metre of a dwelling façade. c) with the noise limits in the Noise Limits table, the noise measurement equipment must be located: i) at the most affected point at a location where there is no dwelling at the location; or ii) at the most affected point within an area at a location prescribed by part (a) or part (b) of this condition.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L4.4	The noise limits set out in the Noise Limits table apply under all meteorological conditions except for the following: a) Wind speeds greater than 3 metres/second at 10 metres above ground level; or b) Stability category F temperature inversion conditions and wind speeds greater than 2 metres/second at 10 metres above ground level; or c) Stability category G temperature inversion conditions. For the purposes of this condition: a) Data recorded by the meteorological station identified as EPA Identification Point(s) W1 must be used to determine meteorological conditions; and b) Temperature inversion conditions (stability category) are to be determined by the sigmatheta method referred to in Part E4 of Appendix E to the NSW Industrial Noise Policy.	Noted	Noted			
EPL 12365	L4.5	For the purposes of determining the noise generated at the premises the modification	Noted	Noted			
		factors in Section 4 of the NSW Industrial Noise Policy must be applied, as appropriate, to the noise levels measured by the noise monitoring equipment.		Noted			
	L5 Blasting			1			
EPL 12365	L5.1	residential building or other noise sensitive location. Error margins associated with any monitoring equipment used to measure this are not to be taken into account in determining whether or not the limit has been exceeded.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. 30-06-14 Matong Station 122.7dBA, located on TCL owned land, all other results compliant	Compliant			
EPL 12365	L5.2	The overpressure level from blasting operations at the premises must not exceed 115dB (Lin Peak) for more than five per cent of the total number of blasts over each reporting period at any time and at any point within 30 metres of any non-project related residential building or other noise sensitive location. Error margins associated with any monitoring equipment used to measure this are not to be taken into account in determining whether or not the limit has been exceeded.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L5.3		AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	L5.4	Ground vibration peak particle velocity from the blasting operations at the premises must not exceed 5mm/sec for more than five per cent of the total number of blasts over each reporting period at any point within 3.5 metres of any non project related residential building or other noise sensitive location. Error margins associated with any monitoring equipment used to measure this are not to be taken into account in determining whether or not the limit has been exceeded.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L5.5	Blasting operations on the premises must only be carried out between the hours 9am to 5pm, Monday to Saturday, inclusive.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	L5.6	The hours of operation for blasting operations specified in condition L7.3 may be varied if the EPA, having regard to the effect that the proposed variation would have on the amenity of the residents in the locality, gives written consent to the variation.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	L5.7	Blasting at the premises is limited to 1 blast on each day on which blasting is permitted. Note: Additional blasts are permitted where it is demonstrated to be necessary for safety reasons and the EPA and neighbours have been notified of the intended blast prior to the additional blast being fired.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	4. Operating Condition	ons					
		caried out in a competent manner					
EPL 12365	01.1	Licensed activities must be carried out in a competent manner. This includes: a) the processing, handling, movement and storage of materials and substances used to carry out the activity; and b) the treatment, storage, processing, reprocessing, transport and disposal of waste generated by the activity.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review — Environment Protection Licence 12365.	Compliant			
	O3 Dust	<u> </u>		1			
EPL 12365	O3.1	All operations and activities occurring at the premises must be carried out in a manner that will minimise the emission of dust from the premises.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	03.2	Trucks transporting coal from the premises must be covered immediately after loading to prevent wind blown emissions and spillage. The covering must be maintained until immediately before unloading the trucks.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	5. Monitoring and Re						
	M2 Requirement to r	nonitor concentration of pollutants discharged					
EPL 12365	M2.1	For each monitoring/discharge point or utilisation area specified below (by a point number), the licensee must monitor (by sampling and obtaining results by analysis) the concentration of each pollutant specified in Column 1. The licensee must use the sampling method, units of measure, and sample at the frequency, specified opposite in the other columns:	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M2.2	Air Monitoring Requirements POINT 28 Pollutant Units of measure Frequency Sampling Method PM10 micrograms per cubic metre Continuous AM-22	Noted	Noted			





Reference	Clause		Red	quirement		Evidence	Audit Finding		Risk	
Reference	Clause						Audit Finding	Consequence	Likelihood	Risk
		Water and/ or Land N	Monitoring Requiren	nents		Annual Return. All the land and water monitoring points were listed.				<u> </u>
		Pollutant Conductivity Oil and Grease pH Total suspended solids	Units of measure microsiemens per centimetre milligrams per litre pH milligrams per litre	Frequency Special Frequency 1 Special Frequency 1 Special Frequency 1 Special Frequency 1	Sampling Method Grab sample Grab sample Grab sample Grab sample					
		POINT 5,6,7,8								
		Pollutant	Units of measure	Frequency	Sampling Method					
EPL 12365	M2.3	Conductivity Oil and Grease pH Total suspended solids	microsiemens per centimetre milligrams per litre pH milligrams per litre	Special Frequency 1 Special Frequency 1 Special Frequency 1 Special Frequency 1	Grab sample Grab sample Grab sample Grab sample		Compliant			
		POINT 9,10,11,12 Pollutant	Units of measure	Frequency	Sampling Method					
		Conductivity	microsiemens per	Every 6 months	Grab sample					
		Lead	milligrams per litre	Every 6 months	Grab sample					
		pH Standing Water	pH metres	Every 6 months Every 6 months	Grab sample In situ					
		Level								
		POINT 13								
		Pollutant Conductivity	Units of measure microsiemens per	Frequency Quarterly	Sampling Method Grab sample					
		Oil and Grease	centimetre milligrams per litre	Quarterly	Grab sample					
		pH Total suspended	pH milligrams per litre	Quarterly Quarterly	Grab sample Grab sample					
		solids								
ED. 1005E					eans the collection of sample					
EPL 12365	M2.4	and in any case not n	-		1, 25, 26, and 27 commences		Noted			
					es that points 15, 16, 20 and	Noted				
					who holds Environment					
				older of environmen	t protection licence 12365					
EPL 12365	M2.5	obtains monitoring d					Noted			
			•	•	der this licence. The licensee where the licensee is unable					
		to obtain the monito			where the licensee is unable					
	M3 Testing methods	- concentration limits	mig data to meet th	iis condicioni						
		Monitoring for the co	oncentration of a po	llutant emitted to th	e air required to be	AEMR 2012/2013, Appendix 3				
		conducted by this lice				TABLE A3.2 Compliance Review –				
				or under the Act to	be used for the testing of the	Environment Protection Licence 12365.				
EDI 12265	M3.1	concentration of the		arundartha Act	, mathadalagu which -		Compliant			
EPL 12365	IVI3.1	condition of this licer			methodology which a		Compliant			
			•	-	y a condition of this licence,					
					ooses of that testing prior to					
		the testing taking pla			3,					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Ciause			Addit rinding	Consequence	Likelihood	Risk
EPL 12365	M3.2	Subject to any express provision to the contrary in this licence, monitoring for the concentration of a pollutant discharged to waters or applied to a utilisation area must be done in accordance with the Approved Methods Publication unless another method has been approved by the EPA in writing before any tests are conducted. Note: The Protection of the Environment Operations (Clean Air) Regulation 2010 requires testing for certain purposes to be conducted in accordance with test methods contained in the publication "Approved Methods for the Sampling and Analysis of Air Pollutants in NSW".	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	M4 Weather Monitor	ing					
EPL 12365	M4.1	For each monitoring point specified in the table below the licensee must monitor (by sampling and obtaining results by analysis) the parameters specified in Column 1. The licensee must use the sampling method, units of measure, averaging period and sample at the frequency, specified opposite in the other columns. Point W1 Parameter	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M4.2	The meteorological weather station must be maintained so as to be capable of continuously monitoring the parameters specified in this section.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	M5 Recording of pollu						
EPL 12365	M5.1	The licensee must keep a legible record of all complaints made to the licensee or any employee or agent of the licensee in relation to pollution arising from any activity to which this licence applies.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M5.2	The record must include details of the following: a) the date and time of the complaint; b) the method by which the complaint was made; c) any personal details of the complainant which were provided by the complainant or, if no such details were provided, a note to that effect; d) the nature of the complaint; e) the action taken by the licensee in relation to the complaint, including any follow-up contact with the complainant; and f) if no action was taken by the licensee, the reasons why no action was taken.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M5.3	The record of a complaint must be kept for at least 4 years after the complaint was made.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M5.4	The record must be produced to any authorised officer of the EPA who asks to see them.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	M6 Telephone compl	I aints line					





Reference	Clause	Requirement	Evidence	Audit Finding	Risk		
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	M6.1	The licensee must operate during its operating hours a telephone complaints line for the purpose of receiving any complaints from members of the public in relation to activities conducted at the premises or by the vehicle or mobile plant, unless otherwise specified in the licence.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M6.2	The licensee must notify the public of the complaints line telephone number and the fact that it is a complaints line so that the impacted community knows how to make a complaint.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M6.3	The preceding two conditions do not apply until 3 months after: a) the date of the issue of this licence or b) if this licence is a replacement licence within the meaning of the Protection of the Environment Operations (Savings and Transitional) Regulation 1998, the date on which a copy of the licence was served on the licensee under clause 10 of that regulation.	Noted	Noted			
	M7 Other monitoring	g and recording conditions					
EPL 12365	M7.1	For each monitoring point specified below, the Licensee must monitor the noise or vibration parameter specified in Column 1. The Licensee must use the sampling method, units of measure, and sample at the frequency, specified opposite in the other columns. Points: N1 Parameter Units of Measure Frequency Sampling Method Ambient Noise LAeq (15 minute) Frequency of monitoring as detailed in the most recently approved Noise Approved Noise LAP Noise LAP Noise LAP Noise Management Farr and LAP Noise Management Farr and Plant for the premises.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M7.2	Point: N3 Parameter Units of Measure Frequency Sampling Method Ambient Noise Leq (15 minute) Continuous real time noise monitoring as detailed in the most recently approved "Noise monitoring as detailed in the most recently approved "Noise Management Flam" (140 Measurement Flam") (140 Measurement Flam	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M7.3	Points: N1 Parameter Units of measure Frequency Sample Method Blast Noise DB(Lin Peak) Every Blast Type 1 Noise Blast Logger Blast Vibration mm/s Every Blast Geophone Logger or Similar	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review — Environment Protection Licence 12365.	Compliant			
EPL 12365	M7.4	For the purpose of conditions M7.1 and M7.2, the noise monitoring locations are described as: EPA Identification No. Description of Location	AEMR 2012/2013, Appendix 3 d TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M7.5	Note: N3 is a portable monitor enabling the monitor to be relocated to areas of potential greatest impact. The licensee is responsible to ensure that it is located at the most suitable location.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Filluling	Consequence	Likelihood	Risk
EPL 12365	M7.6	Note: The location, frequency of monitoring and the parameters to be monitored may be varied by the EPA once the variability of the noise impact is established.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	M7.7	To assess compliance with the noise limits presented in the Noise Limits table, attended noise monitoring must be undertaken in accordance with the condition titled Determining Compliance, outlined above, and: a) at each one of the locations listed in the Noise Limits table; b) occur Quarterly in a reporting period; c) occur during each day, evening and night period as defined in the NSW Industrial Noise Policy for a minimum of: i) 1.5 hours during the day; ii) 30 minutes during the evening; and iii) 1 hour during the night. d) occur for three consecutive operating days.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	6 Reporting Conditio	ns		•			
	R1 Annual return doo	cuments					
EPL 12365	R1.1	The licensee must complete and supply to the EPA an Annual Return in the approved form comprising: a) a Statement of Compliance; and b) a Monitoring and Complaints Summary. At the end of each reporting period, the EPA will provide to the licensee a copy of the form that must be completed and returned to the EPA.	TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	R1.2	An Annual Return must be prepared in respect of each reporting period, except as provided below.	Annual returns sighted in website from periods Jan 2007-Jan2008 to Jan 2013-Jan 2014	Compliant			
EPL 12365	R1.3	Where this licence is transferred from the licensee to a new licensee: a) the transferring licensee must prepare an Annual Return for the period commencing on the first day of the reporting period and ending on the date the application for the transfer of the licence to the new licensee is granted; and b) the new licensee must prepare an Annual Return for the period commencing on the date the application for the transfer of the licence is granted and ending on the last day of the reporting period.	Noted	Noted			
EPL 12365	R1.4	Where this licence is surrendered by the licensee or revoked by the EPA or Minister, the licensee must prepare an Annual Return in respect of the period commencing on the first day of the reporting period and ending on: a) in relation to the surrender of a licence - the date when notice in writing of approval of the surrender is given; or b) in relation to the revocation of the licence - the date from which notice revoking the licence operates.	Noted	Noted			
EPL 12365	R1.5	The Annual Return for the reporting period must be supplied to the EPA by registered post not later than 60 days after the end of each reporting period or in the case of a transferring licence not later than 60 days after the date the transfer was granted (the 'due date').		Compliant			
EPL 12365	R1.6	The licensee must retain a copy of the Annual Return supplied to the EPA for a period of at least 4 years after the Annual Return was due to be supplied to the EPA.	Noted	Noted			
EPL 12365	R1.7	Within the Annual Return, the Statement of Compliance must be certified and the Monitoring and Complaints Summary must be signed by: a) the licence holder; or b) by a person approved in writing by the EPA to sign on behalf of the licence holder.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	R1.8	A person who has been given written approval to certify a certificate of compliance under a licence issued under the Pollution Control Act 1970 is taken to be approved for the purpose of this condition until the date of first review of this licence. Note: The term "reporting period" is defined in the dictionary at the end of this licence. Do not complete the Annual Return until after the end of the reporting period. Note: An application to transfer a licence must be made in the approved form for this purpose.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	R2 Notification of env						
EPL 12365	R2.1	Notifications must be made by telephoning the Environment Line service on 131 555.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	R2.2	The licensee must provide written details of the notification to the EPA within 7 days of the date on which the incident occurred. Note: The licensee or its employees must notify all relevant authorities of incidents causing or threatening material harm to the environment immediately after the person becomes aware of the incident in accordance with the requirements of Part 5.7 of the Act.	TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	R3 Written report						
EPL 12365	R3.1	Where an authorised officer of the EPA suspects on reasonable grounds that: a) where this licence applies to premises, an event has occurred at the premises; or b) where this licence applies to vehicles or mobile plant, an event has occurred in connection with the carrying out of the activities authorised by this licence, and the event has caused, is causing or is likely to cause material harm to the environment (whether the harm occurs on or off premises to which the licence applies), the authorised officer may request a written report of the event.	No requests received by Tarrawonga Coal during reporting period. AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	R3.2	report to the EPA within such time as may be specified in the request.	No requests received by Tarrawonga Coal during reporting period. AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit rinding	Consequence	Likelihood	Risk
EPL 12365	R3.3	b) the type, volume and concentration of every pollutant discharged as a result of the event; c) the name, address and business hours telephone number of employees or agents of the licensee, or a specified class of them, who witnessed the event;	TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	R3.4	provide such further details to the EPA within the time specified in the request.	No requests received by Tarrawonga Coal during reporting period. AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
	R4 Other reporting co	nditions					
EPL 12365	R4.1	days of the completion of the quarterly noise monitoring. The assessment must be	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
	7 General Conditions						
EPL 12365	G1 Copy of licence ke	ot at premises or plant A copy of this licence must be kept at the premises to which the licence applies.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
EPL 12365	G1.2		AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	G1.3	- · ·	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Compliant			
		d Reduction Programs					
	U1 Particulate Matter	Control Best Practice Implementation - Wheel Generated Dust					





Reference	Clause	Clause	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	U1.1	The Licensee must achieve and maintain a dust control efficiency of 80% or more on all active haul roads by 17 May 2013. Control efficiency is calculated as: CE = E (uncontrolled) - E (controlled) × 100 E (uncontrolled) Where E = the emission rate of the activity	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	U1.2	To assess compliance with Condition U1.1, the Licensee must: - measure uncontrolled and controlled haul road emissions on at least 2 occasions using a mobile dust monitor; - continuously measure and record 'additional site data' including: - vehicle kilometres travelled (VKT), - meteorological conditions, - water use for dust suppression. - undertake silt content and soil moisture sampling during sampling events; and - determine if a site specific relationship can be derived between the measured control efficiency, additional site data, water use, meteorological data; and silt content and soil moisture levels. The measurement of uncontrolled and controlled haul road PM10 emissions must be undertaken under varying meteorological conditions, including at those times when analysis of meteorological data indicates that elevated levels of dust are most likely at the Premises. Note: The EPA acknowledges that in order to determine uncontrolled PM10 emissions, the section of haul road to be sampled will need to be left untreated for a period of up to 12 hours prior to the sampling taking place.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. Mechanism to test dust control efficiency attached to monitoring program currently pending final approval by EPA.	Not Applicable			
EPL 12365	U1.3	The Licensee must submit a report to the EPA which documents the results of the assessment undertaken in accordance with Condition U1.1. The report must include an assessment of: the dust control effectiveness, the dust levels recorded, and any relationship established between control effectiveness and the additional site data. The report must be submitted by the Licensee to the Environment Protection Authority Regional Manager Armidale, at PO Box 494, ARMIDALE by 15 August 2014.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	U1.4	The report required by condition U1.3 must be made publicly available by the Licensee on the Licensee's website by (two weeks from submission date nominated in U1.3).	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. No identification of adverse weather conditions impacting on dust generation on overburden during reporting period.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
EPL 12365	U2.1	dumping of overburden during adverse weather conditions to minimise the generation of particulate matter from 22 March 2013.	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. No identification of adverse weather conditions impacting on dust generation on overburden during reporting period.	Compliant			
EPL 12365	U2.2	- undertake daily visual dust level assessments, continuously record real-time PM10 levels and continuously measure and record real-time meteorological conditions; and	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365. Pending approval of Monitoring Program	Not Applicable			
EPL 12365	U2.3		AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
EPL 12365	U2.4	the Licensee's website by (two weeks from submission date in 2.3 above).	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			
	U3 Particulate Matter	Control Best Practice Implementation - Trial of Best Practice Measures for Disturbing and Ha	ndling Overbuden		Į.		
EPL 12365	U3.1	measures for the control of particulate matter from the use of equipment on overburden	AEMR 2012/2013, Appendix 3 TABLE A3.2 Compliance Review – Environment Protection Licence 12365.	Not Applicable			





Deference	Clause	Dogwinsmont	Fuidones	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	Mining Lease - 1579						
	Mining, Rehabilitatio	n, Environmental Management Process (MREMP) Mining Operation					
ML 1579	2.1	Mining operations, including mining purposes, must be conducted in accordance with a Mining Operations Plan (the Plan) satisfactory to the Director-General. The Plan together with environmental conditions of development consent and other approvals will form the basis for: (a) ongoing mining operations and environmental management; and (b) ongoing monitoring of the project.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			
ML 1579	2.2	The Plan must be prepared in accordance with the Director- Generals guidelines current at the time of lodgement.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and	Compliant			
ML 1579	2.3	A Plan must be lodged with the Director-General: (a) prior to the commencement of mining operations (including mining purposes); (b) subsequently as appropriate prior to the expiry of any current Plan; and (c) in accordance with any direction issued by the Director-General.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			





Deference	Clause	Dominomont	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
ML 1579	2.4	The Plan must present a schedule of proposed mine development for a period of up to seven years and contain diagrams and documentation which identify: a. Area(s) proposed to be disturbed under the plan; b. mining and rehabilitation methods to be used and their sequence c. areas to be used for disposal of tailings/waste d. existing and proposed surface infrastructure; e. existing flora and fauna on the site; f. progressive rehabilitation schedules; g. areas of particular environmental, ecological and cultural sensitivity and measures to protect these areas; h. water management systems (including erosion and sediment controls); i. proposed resource recovery; and j. where the mine will cease extraction during the term of the plan, a closure plan including final rehabilitation objectives/methods and post mining landuse/vegetation	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			
ML 1579	2.5	The Plan when lodged will be reviewed by the Department	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			
ML 1579	2.6	The Director-General may within two months of the lodgement of a plan, require modification and re-lodgement	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			





Reference	Clause	Paguirament	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
ML 1579	2.7	If a requirement in accordance with clause (6) is not issued within two months of the lodgement of a Plan, the lease holder may proceed with implementation of the Plan	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			
ML 1579	2.8	During the life of the Mining operations Plan, proposed modifications to the Plan must be lodged with the Director-General and will be subject to the review process outlined in clauses (5) - (7) above.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Initial MOP lodged with DMR and accepted on the 9th May 2006. MOP amendment for Section 75W extension approved in October 2010. Revised MOP for Tarrawonga Extension area lodged May 2013.	Compliant			
	Annual Environmenta	I Management Report (AEMR)					
ML 1579	3.1	Within 12 months of the commencement of mining operations and thereafter annually or, at such other times as may be allowed by the Director-General, the lease holder must lodge an Annual Environmental Management Report (AEMR) with the Director-General.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 AEMR was prepared in satisfaction with this condition. Content of AEMR follows guidelines.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Finding	Consequence	Likelihood	Risk
ML 1579	3.2	The AEMR must be prepared in accordance with the Director-General's guidelines current at the time of reporting and contain a review and forecast of performance for the preceding and ensuing twelve months in terms of: (a) the accepted Mining Operations Plan; (b) development consent requirements and conditions; (c) Department of Environment and Conservation and Department of Planning licences and approvals; (d) any other statutory environmental requirements; (e) details of any variations to environmental approvals applicable to the lease area; and (f) where relevant, progress towards final rehabilitation objectives.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 AEMR was prepared in satisfaction with this condition. Content of AEMR follows guidelines.	Compliant			
ML 1579	3.3	After considering an AEMR the Director-General may, by notice in writing, direct the lease holder to undertake operations, remedial actions or supplementary studies in the manner and within the period specified in the notice to ensure that operations on the lease area are conducted in accordance with sound mining and environmental practice.	AEMR/Annual Review 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 AEMR was prepared	Compliant			
	Subsidence Manage	ement					
ML 1579	4.1	(a). The lease holder shall, as and when directed by the Minister, co-operate with the Director-General to conduct and facilitate review of the AEMR involving other government agencies and the local council	Noted	Noted			
ML 1579	4.1	(b) Underground mining operations which will potentially lead to subsidence include secondary extraction panels such as longwells or miniwalls, associated liest workings (gateroads, installation roads and associated main headings, atc), and pillar extractions, and are otherwise defined by the Applications for Subsidence Management Approvals guidelines (EDG17).		Not Applicable			
ML 1579	4.1	The lease holder must not commence or undertake underground mining operations that will potentially lead to subsidence other than in accordance with a Subsidence Management Plan approved by the Director-General, an approval under the Coal Miniss Regulation Act 1982, or the document New Subsidence Management Plan Approval Process—Transitional Provisions (EDP09).		Not Applicable			
ML 1579	4.1	Subsidence Management Plans are to be prepared in accordance with the Guideline for Applications for Subsidence Management Approvals		Not Applicable			
ML 1579	4.1	(6) Subsidence Management Plans as approved shall form part of the Mining Operations Plan required, under Condition 2 and will be subject to the Annual Environmental Management Report process as set out under Condition 3. The SMP is also subject to the requirements for subsidence monitoring and reporting set out in the document New Approval Process for Management of Coal Mining Subsidence - Policy.		Not Applicable			
		Hit management of Ordi mining Sources					





Reference	Clause	Poguiroment	Evidones	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
ML 1579	11	(a) Ground Vibration The lease holder must ensure that the ground vibration peak particle; velocity generated by any blasting within the lease area does not exceed 10 mm/second and does not exceed 5 mm/second in more than 5% of the total number of blasts over a ported of 12 months at any dwelling or escaped premises as the case may be, unless determined otherwise by the Department of Environment and Conservation.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 No exceedances recorded during the reporting period	Compliant			
ML 1579	11	(b) Biast Overprossure The lease holder must ensure that the blast overpressure noise level generated by any blasting within the lease area does not exceed 120 dB (linear) and does not exceed 115 dB (linear) in more than 5% of the total number of blasts over a period of 12 months, at any dwelling or occupied premises, as the case may be, unless determined otherwise by the Department of Environment and Conservation.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 No exceedances recorded during the reporting period	Compliant			
	Rehabilitation						
ML 1579	13	(a) Land disturbed must be rehabilitated to a stable and permanent form suitable for a subsequent land use occeptable to the Director-General and in accordance with the Mining Operations Plan so that. • there is no adverse environmental effect outside the disturbed area and that line land is properly drained and protected from soil eroston. • the state of the tend is compatible with the surrounding fand and land use requirements. • the tendforms, soils, hydrology and flora require no greater maintenance than that in the surrounding land. • in cases where revegetation is required and native vegetation has been removed or damaged, the original species must be restablished with close reference to the flora survey included in the Mining Operations Plan. If the original vegetation was not native, any re-established vide density. • the land does not pose a threat to public safety.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Current rehabilitation activities are in accordance with commitments identified in the MOP Tarrawonga Coal Mine, Rehabilitation Management Plan, May 2013, Section 3 and 6.5	Compliant			
ML 1579	13	b. Any topsoil that is removed must be stored and maintained in a manner acceptable to the Director-General.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 Topsoil is stockpiled and seeded in accordance with commitments identified in the MOP. Tarrawonga Coal Mine, Rehabilitation Management Plan, May 2013, Section 6.3.2	Compliant			
ML 1579	14	The lease holder must comply with any direction given by the Director-General regarding the stabilisation and revegetation of any mine residues, tailings or overburden dumps situated on the lease area.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579 No directions issued.	Not triggered			





Deference	Clause	Dogwinsmont	Fuidones	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	Exploratory Drilling			1			
ML 1579	15.1	At least twenty eight days prior to commencement of drilling operations the lease holder must notify the relevant Department of Natural Resources regional hydrogeologist of the intention to drill exploratory drill holes together with information on the location of the proposed holes.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
ML 1579	15.2	If the lease holder drills exploratory drill holes he must satisfy the Director-General that: (a) all cored holes are accurately surveyed and permanently marked in accordance with Departmental guidelines so that their location can be easily established; b) all holes cored or otherwise are sealed to prevent the collapse of the surrounding surface; (c) all drill holes are permanently sealed with cement plugs to prevent surface discharge of groundwaters; (d) if any drill hole meets natural or noxious gases it is plugged or sealed to prevent their ascape; (e) if any drill hole meets an artesian or sub-artesian flow it is effectively sealed to prevent contamination of aquifiers. (f) once any drill hole coases to be used the hole must be sealed in accordance with Departmental guidelines. Alternatively, the hole must be sealed as instructed by the Director-General. (g) once any drill hole ceases to be used the land and its immediate vicinity is left in a close, tidy and stable condition.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
	Prevention of Soil Er	Losion and Pollution					
ML 1579	16	Operations must be carried out in a manner that does not cause or aggravate air pollution, water pollution (including sedimentation) or soil contamination or erosion, unless otherwise authorised by a relevant approval, and in accordance with an accepted Mining Operations Plan. For the purpose of this condition, water shall be taken to include any watercourse, waterbody or groundwaters. The lease holder must observe and perform any instructions given by the Director-General in this regard.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
	Transmission lines (Communication lines and Pipelines					
ML 1579	17	Operations must not interfere with or impair the stability or efficiency of any transmission line, communication line, pipeline or any other utility on the lease area without the prior written approval of the Director-General and subject to any conditions he may stipulate.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
	1			1	1		





Doforonce	Clause	Poguiroment	Evidonas	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
ML 1579	18	(a) Activities on the lesse must not interfere with or damage fences without the prior written approval of the owner thereof or the Minister and subject to any conditions the Minister may stipulate. (b) Gates within the lease area must be closed or left open in accordance with the requirements of the landholder.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
ML 1579	19	(a) Operations must not affect any road unless in accordance with an accepted Mining Operations Plan or with the prior written approval of the Director-General and subject to any conditions he may stipulate. (b) The lease holder must pay to the designated authority in control of the road (generally the local council or the Roads and Traffic Authority) the cost incurred in fixing any damage to roads caused by operations carried out under the lease, less any amount paid or payable from the Mine Subsidence Compensation Fund.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
ML 1579	20	Access tracks must be kept to a minimum and be positioned so that they do not cause any unnecessary damage to the land. Temporary access tracks must be ripped, topsolicd and revegetated as soon as possible after they are no longer required for mining operations. The design and construction of access tracks must be in accordance with specifications fixed by the Department of Natural Resources.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
	Trees and Timber						
ML 1579	21	(a) The lease holder must not fell trees, strip bark or cut timber on the lease without the consent of the landholder who is entitled to the use of the timber, or if such a landholder refuses consent or attaches unreasonable conditions to the consent, without the approval of a warden.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
ML 1579	21	(b) The lease holder must not cut, destroy, ringbark or remove any timber or other vegetative cover on the lease area except such as directly obstructs or prevents the carrying on of operations. Any clearing not authorised under the Mining Act 1992 must comply with the provisions of the Native Vegetation Act 2003.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			
ML 1579	21	The lease holder must obtain all necessary approvals or licences before using timber from any Crown land within the lease area.	AEMR 2012/2013, Appendix 3 TABLE A3.3 Compliance Review – ML1579	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Kequirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
	Mining Lease - 1685						
	Rehabilitation	<u> </u>	T 0.1.175 11 A4	I	T -		
ML 1685	2	Any disturbance resulting from the activities carried out under thi mining lease must be rehabilitated to the satisfaction of the Minister	Tarrawonga Rehabilitation Management Plan, May 2013 (SLR Consulting), Section 1.2 states that the Rehabilitation Management Plan has been prepared in accordance with all relevant approval consent conditions and commitments, relevant regulatory and industry policies and guidelines.	Compliant			
	Mining Operations P	Plan and Annual Rehabilitation Report					
ML 1685	3	(a) The lease holder must comply with an approved Mining Operations Plan (MOP) in carrying out any significant surface disturbing activities, including mining operations, mining purposes and prospecting. The lease holder must apply to the Minister for approval of a MOP. An approved MOP must be in place prior to commencing any significant surface disturbing activities, including mining operations, mining purposes and prospecting.	Sighted MOP Tarrawonga MOP Amendment, Dec 2013 (SLR Consulting)	Compliant			
ML 1685	3	(b) The MOP must identify the post mining land use and set out a detailed rehabilitation strategy which (i) identifies areas that will be disturbed: (ii) identifies how staging of specific mining operations, mining purposes and prospecting. (iii) Identifies how the mine will be managed and rehabilitated to achieve the post mining land use. (iv) Identifies how mining operations, mining purposes and prospecting will be carried out it order to prevent and or minimise harm to the environment, and (iv) reflects the conditions of approval under the Environmental Planning and Assessment Act 1979; the Protection of the Environment Operations Act 1997; and any other approvas relevant to the development including the conditions of this mining lesse.	Sighted MOP Tarrawonga MOP Amendment, Dec 2013 (SLR Consulting) Tarrawonga Rehabilitation Management Plan, May 2013 (SLR Consulting), Section 1.2 states that the Rehabilitation Management Plan has been prepared in accordance with all relevant approval consent conditions and commitments, relevant regulatory and industry policies and guidelines.	Compliant			
ML 1685	3	(c) The MOP must be prepared in accordance with the ESG3. Mining Operations Plan (MOP). Guidelines September 2013 published on the Department's website at: www.resourbes.nsw.gbv.aulienvironment.	Sighted MOP and mention of guidelines within MOP. Tarrawonga Rehabilitation Management Plan, May 2013 (SLR Consulting), Section 1.2 states that the Rehabilitation Management Plan has been prepared in accordance with all relevant approval consent conditions and commitments, relevant regulatory and industry policies and guidelines.	Compliant			
ML 1685	3	d. The lease holder may apply to the Minister to amend an approved MOP at any time.	Noted	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
ML 1685	3	(e) It is not a breach of this condition if (i) the operations which, but for this condition 3(e) would be a breach of condition 3(a), were necessary to comply with a lawful order or direction given under the Environment Planning and Assessment Act 1979, the Protection of the Environment Operations Act 1997, the Mine Health and Safety Act 2004 / Coall Mine Health and Safety Act 2002 and Mine Health and Safety Act 2002 and Mine Health and Safety Act 2011; and (ii) line Minister had been notified in writing of the terms of the order or direction prior to the operations constituting the breach being carried out.	Noted	Compliant			
ML 1685	3	(f) The lease holder must prepare a Rehabitation Report to the eatisfaction of the Minister. The report mist: (i) provide a detailed review of the progress of rehabitation against the performance measures and criteria established in the approved MOP: (ii) be submitted annually on the grant anniversary date (or at such other times as agreed by the Minister) and (iii) be prepared in accordance with any relevant annual reporting guidelines published on the Department's veboats at years resource annual reporting submission. Note: The Penhabitation Report replaces the Annual Environmental Management Report.	Sighted MOP and Rehabilitation section within report. Tarrawonga Rehabilitation Management Plan, May 2013 (SLR Consulting), Section 1.2 states that the Rehabilitation Management Plan has been prepared in accordance with all relevant approval consent conditions and commitments, relevant regulatory and industry policies and guidelines.	Compliant			
	5. Environmental Inc	ident Report					
ML 1685	5	(a) The lease holder must notify the Department of all. (ii) breaches of the conditions of this mining lease or breaches of the Act causing on threatening material harm to the environment, and (iii) breaches of environmental protection legislation causing or threatening material harm to the environment (as defined in the Protection of the Environment Administration Act 1991), anising in connection with significant surface disturbing activities, including mining operations, mining purposes and protecting operations, under the mining lease. The notification must be given immediately after the lease holder becomes aware of the treach. Alsh. Refer to green resources user our authorizonated for notification contact details.	No notifications made in the audit period	Not Applicable			
ML 1685	5	(b) The lease holder must submit an Environmental Incident Report to the Department within seven (7) days of all breaches referred to in condition 5(a)(i) and (ii). The Environmental incident Report must include: (i) the details of the mining lease; (ii) contact details for the issue holder; (iii) a map identifying the location of the incident and where material harm to the environment has or is likely to occur; (iv) a description of the nature of the incident or breach, likely causes and consequences. (v) a description of the nature of the incident or breach, likely causes and consequences. (v) a simetable showing actions taken or planned to address the incident and to prevent have incident or breaches referred to this(a). (vi) a summitty of all previous incidents or breaches which have occurred in the provious 15 months relating to supriscales station departicing actives, including mixing operations, mining purposes and prospecting operations unlike this mixing leases. Anni. The lease holder should have regard to any relevant Devictor General's guidelines at the proparation of an Environmental lanction of Report Plater to environmental lanction of the provious cause of the programment of an Environmental lanction of the province of the environmental lanction of the environmental and the environmental lanction of the environmental lanction of the environmental state of the environmental lanction of the environmental lanction of the environmental lanction.	No notifications made in the audit period	Not Applicable			





Reference	Clause	Paguiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
ML 1685	5	(c) In addition to the requirements set out in conditions 5(a) and (b), the lease holder must immediately advise the Department of any notification made under section 148 of the Protection of the Environment Operations Act 1997 arising in connection with significant surface disturbing activities including mining operations, mining purposes and prospecting operations, under this mining lease.	No notifications made in the audit period	Not Applicable			
	Mining Lease - 1693	3					
	Environmental Harn	n .					
ML 1685	2	(a) The lease holder must implement all practicable measures to prevent and/or minimise any harm to the environment that may result from the construction, operation or rehabilitation of any activities under this lease.	Noted	Noted			
ML 1685	2	(b) For the purposes of this condition (i) environment means components of the earth, including. (A) land, air and water, and (B) airly layer of the atmosphere, and (C) any organic or inorganic matter and any living organism, and (D) fluman-mode or modified structures and areas, and includes interacting natural ecosystems that include components referred to in paragraphs (A)–(C). (ii) harm to the environment includes any direct or indirect elteration of the environment that has the effect of degrading the environment and, without limiting the generality of the above, includes any act or omission that results in pollution, contributes to the extinction or degradation of any threatened species, populations or ecological communities and their habitats and causes impacts to places, objects and features of significance to Aboriginal people.	Noted	Noted			
	3. Mining Operation	ns Plan					
ML 1685	3	(a) Mining operations must not be carried out otherwise than in accordance with a Mining Operations Plan (MOP) which has been approved by the Director-General.	Noted	Compliant			





Reference	Clause	e Requirement	Evidence	Audit Finding		Risk	
vererence	Clause	Requirement	Evidence	Addit Finding	Consequence	Likelihood	Risk
ML 1685	3	(b) The MOP must: (ii) identify areas that will be disturbed by mining operations. (iii) identify how the mine will be managed to allow mino closure; (iv) identify how the mine will be managed to allow mino closure; (iv) identify how mining operations will be carried out in order to prevent and or minimise harm to the environment. (v) reflect the conditions of approval under: • the Environmental Planning and Assessment Act 1979 • the Protection of the Environment Operations Act 1997 • and any other approvals relevant to the development including the conditions of this lease; and • have regard to any relevant guidelines adopted by the Director-General	Noted	Compliant			
ML 1685	3	c. The leaseholder may apply to the Director-General to amend an approved MOP at any time.	Noted	Noted			
ML 1685	3	(ii) the operations constituting the breach were necessary to comply with a learly order or direction given under the Mining Act 1992, the Environmental Planning and Assessment Act 1979, Protection of the Environment Operations Act 1997, Mine Health and Safety Act 2004 / Coal Mine Health and Safety Act 2004 ord Mine Health and Safety Regulation 2007 / Coal Mine Health and Safety Regulation 2006 or the Occupational Health and Safety Act 2000, and (ii) the Director-General had been notified in writing of the terms of the order or direction prior to the operations constituting the breach being carried out.	AEMR's for 2009 - 2010 to 201202013 sighted	Compliant			
ML 1685	3	A MOP ceases to have effect 7 years after date of approval or other such period as identified by the Director-General.	Noted	Noted			
	4. Environment Man	agement Report		•			
ML 1685	4	(a) The lease holder must lodge Environmental Management Reports (EMR) with the Director-General annually or at dates otherwise directed by the Director-General.	AEMR's for 2009 - 2010 to 201202013 sighted	Compliant			
ML 1685	4	(b) The EMR must: (i) report against compliance with the MOP. (ii) report on progress in respect of rehabilitation completion criteria. (iii) report on the extent of compliance with regulatory requirements, and (iv) have regard to any relevant guidelines adopted by the Director-General.	AEMR's for 2009 - 2010 to 201202013 sighted	Compliant			
	5. Environmental Inc	I I I I I I I I I I I I I I I I I I I					
ML 1685	5		No notifications made in the audit period	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
ML 1685	5	(b) For the purposes of this condition, environmental incident includes: (i) any incident causing or threatening material harm to the environment (ii) any breach of Conditions 1 to 9 and 11 to 24; (iii) any breach of environment protection legislation; or, (iv) a serious complaint from landholders or the public.	No notifications made in the audit period	Not Applicable			
ML 1685	5	(c) For the purposes of this condition, harm to the environment is material if: (i) it involves actual or potential harm to the health or safety of human beings or to ecosystems that e. not trivial, or (ii) it results in actual or potential loss or property damage of an amount, or amounts in aggregate, exceeding \$10,000, where loss includes the reasonable costs and expenses that would be incurred in taking all reasonable and practicable measures to prevent, mitigate or make good harm to the environment.	No notifications made in the audit period	Not Applicable			
	7. Rehabilitation						
ML 1685	7	Any disturbance as a result of activities under this lease must be rehabilitated to the satisfaction of the Director-General.	Noted	Noted			
	8. Subsidence Mana						
ML 1685	8	(a) The lease holder shall prepare a Subsidence Management Plan prior to commencing any underground mining operations which will potentially lead to subsidence of the land surface.		Not Applicable			
ML 1685	8	(b) Underground mining operations which will potentially lead to subsidence include secondary extraction panels such as longwalls or miniwalls, associated first workings (gateroads, installation roads and associated main headings, eto), and pillar extractions, and are otherwise defined by the Applications for Subsidence Management Approvais guidelines (EDG17)		Not Applicable			
ML 1685	8	(c) The lease holder must not commence or undertake underground mining operations that will potentially lead to subsidence other than in accordance with a Subsidence Management Plan approved by the Director-General, an approval under the Coal Mine Health & Safety Act 2002, or the document New Subsidence Management Plan Approval Process – Transitional Provisions (EDP09).		Not Applicable			
ML 1685	8	Subsidence Management Plans are to be prepared in accordance with the Guideline for Applications for Subsidence Management Approvals.		Not Applicable			
ML 1685	8	(e) Subsidence Management Plans as approved shall form part of the Mining Operations Plan required under Condition 3 and will be subject to the Environmental Management Report process as set out under Condition 4. The SMP is also subject to the requirements for subsidence monitoring and reporting set out in the document New Approval Process for Management of Coal Mining Subsidence - Policy.		Not Applicable			





Reference	Clause	Dogwigomont	Evidence	Audit Finding		Risk	
Kererence	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	10. Blasting						
ML 1685	10	(a) Ground Vibration The lease holder must ensure that the ground vibration peak particle velocity generated by any blasting within the lease area does not exceed 10 mm/second and does not exceed 5 mm/second in more than 5% of the total number of blasts over a period of 12 months at any dwelling or occupied premises as the case may be, unless determined otherwise by the Department of Environment, Climate Change and Water.	This is also in line with PA 11_0047	Compliant			
ML 1685	10	(b) Blast Overpressure The lease holder must ensure that the blast overpressure noise level generated by any blasting within the lease area does not exceed 120 dB (linear) and does not exceed 115 dB (linear) in more than 5% of the total number of blasts over a period of 12 months, at any dwelling or occupied premises, as the case may be, unless determined otherwise by the Department of Environment, Climate Change and Water	AEMR 2012-2013, Section 3.9.2 This is also in line with PA 11_0047	Compliant			
	12. Prevention of so	il erosion and pollution					
ML 1685	12	Prospecting operations must be carried out in a manner that does not cause or aggravate air pollution, water (including groundwater) pollution, soil contamination or erosion, unless otherwise authorised by a relevant approval, and in accordance with an accepted Mining Operations Plan.	AEMR 2013-2013, Section 3.2.2 During late January 2013, after an extensive dry weather period, approximately 120mm of rain fell over a 24 hour period. This rainfall caused extensive pooling of water on top of the southern emplacement which ultimately overflowed down the rehabilitated area causing extensive washouts and silting of contour drains. It is noted that no wet weather discharge occurred during this event. Additional sediment control structures were installed during the reporting period. URS (2013), WATER MANAGEMENT PLAN, Section 3.2 (Erosion Management) reported that improved water balance modelling is currently under way and this will allow a more proactive approach to water management.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
ML 1685	15	(a) The lease holder must not fell trees, strip bark or cut timber on any land subject of this lease without the consent of the landholder who is entitled to the use of the timber.	MOP, Section 2.4.1	Noted			
ML 1685	15	(b) The lease holder must contact Forests NSW and obtain any required permit, licence or approval before taking fimber from any Crown land within the lease area. Note: Any clearing not authorised under the Act must comply with the requirements of the Native Vegetation Act 2003. Any cleaning or taking of timber on Crown land is subject to the requirements of the Forestry. Act 1916.	MOP, Section 8.2.1	Noted			





Deference	Clause	Requirement	Evidence	A dit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	• '	an - November 2013 to October 2015					
	3.2.1 Air Quality			T	1 1	T	
MOP (Nov 2013 to Oct 2015)	3.2.1	Air quality at Tarrawonga is managed and monitored through the Air Quality and Greenhouse Gas Management Plan, which will be revised and approved during the MOP period.	AQ and GHG Management Plan dated June 2014 (post MOP amendment - Oct 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.1	Air quality is monitored in accordance with the Air Quality and Greenhouse Gas Management Plan, with meteorology and air quality being continuously monitored. Results are reported on a daily basis online and on an annual basis in the Annual Review and the EPL Annual Return.	AQ and GHG Management Plan dated June 2014 (post MOP amendment - Oct 2013), Section 4.2.3 Continuous monitoring for deposited dust, PM2.5 and PM10. -Meteorology is currently being monitored continuously (Section 4.1) -Results are being reported on a daily basis online (Section 4.3.2). -Results are being reported on a annual basis in the AEMR. -Annual return Jan2013-Jan2014 only included real time data of monitoring point 28 for PM10	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.1	A Particulate Matter Control Best Practice Pollution Reduction Program has also been prepared for Tarrawonga in accordance with condition U1 of EPL 12365. This report provides best practice recommendations for the ongoing reduction of particulate matter emissions and includes options for the Tarrawonga Coal Project which have been incorporated into the Air Quality and Greenhouse Gas Management Plan.	AQ and GHG Management Plan dated June 2014 (post MOP amendment - Oct 2013), Section 3	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.1	Tarrawonga proposes to establish, in co-operation with the adjoining Boggabri and Maules Creek Coal Project, a cumulative particulate matter monitoring program. While this program is currently under development, a real time PM10 monitor (TEOM) has been installed to the south east of Tarrawonga, with additional monitoring equipment, including predictive meteorological forecasting and dispersion modelling, to be established in consultation with the Boggabri and Maules Creek operations.	AQ and GHG Management Plan dated June 2014 (post MOP amendment - Oct 2013), Section 1 states TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in relation to developing an Air Quality Management Strategy for the Boggabri – Tarrawonga – Maules Creek (BTM) Complex incorporating cumulative air quality monitoring and management, as required under the conditions of consent for the Tarrawonga Coal Mine. The Strategy has been submitted to the Department of Planning and Environment (DPE) for approval. Once approved, the Strategy will be appended to this Plan.	Compliant			
	3.2.2 Surface Water a	ınd Groundwater					
MOP (Nov 2013 to Oct 2015)	3.2.2	Surface and groundwater at Tarrawonga is managed in accordance with the Tarrawonga Water Management Plan. In accordance with Project Approval PA 11_0047, a Water Management Plan (including Site Water Balance, Surface Water Management Plan, Groundwater Management Plan and a Leard Forest Mining Precinct Water Management Strategy) will be prepared during the MOP period.	Tarrawonga Water Management Plan (revision date: May 2013 - draft). The new plan has been drafted but was deemed inadequate and consultants have been engaged to develop another plan to satisfy this requirement A new plan is required for the new MOP period (Nov 2013 to Oct 2015)	Not Applicable			





Deference	Clause	Requirement	Evidence	A dia Fin din a		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.2	The EPBC Approval 2011/5923 (see Table 2) also requires surface and groundwater management plans to be prepared during this MOP term to meet criteria specified in that Approval, including cumulative groundwater impacts from mining activities at Tarrawonga, Boggabri and Maules Creek Coal Project.	Tarrawonga Water Management Plan (revision date: May 2013 - draft). The new plan has been drafted but was deemed inadequate and consultants have been engaged to develop another plan to satisfy this requirement A new plan is required for the new MOP period (Nov 2013 to Oct 2015)	Not Applicable			
MOP (Nov 2013 to Oct 2015)	3.2.2	Project Approval PA 11_0047 and EPL 12365 impose requirements for managing surface and groundwater quality and quantity, monitoring requirements and water quality discharge criteria. EPL 12365 includes licenced discharge points (LDPs).		Noted			
MOP (Nov 2013 to Oct 2015)	3.2.2	The primary water use at Tarrawonga is dust suppression on internal haul roads and at the coal crushing and screening facility (pending the Boggabri Infrastructure Facilities).	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.2	Surface water and groundwater quality monitoring and water discharge records are reported in the Annual Review and the EPL Annual Return.	AEMR/Annual Review 2012/2013 sighted EPL Annual Return Jan2013-Jan2014 sighted	Compliant			
	3.2.3 Erosion and Se						
MOP (Nov 2013 to Oct 2015)	3.2.3	Erosion and sediment control (ESC) at Tarrawonga is managed in accordance with the Tarrawonga Site Water Management Plan (including an Erosion and Sediment Control Plan), relevant regulatory requirements and the guideline documents titled Managing Urban Stormwater Volume 1: Soils and Construction (Landcom 2004) and Managing Urban Stormwater Volume 2E: Mines and Quarries (DECC 2008).	Tarrawonga Water Management Plan (revision date: May 2013 - draft). This seems to be a draft document. It needs to be confirmed if this plan has been formally approved and if ESC on site is indeed managed according to this plan. Erosion and sed control managed to this plan, but other water management actions not to this plan, currently under revision.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.3	Key sources of erosion and sedimentation are generally related to surface water runoff from exposed surfaces, including cleared areas, stockpiles (coal, soil and waste rock) and unsealed roads, and to a lesser degree caused by wind erosion from emplacement areas and stockpiles. Key potential erosion and sediment risks are managed by: • Diverting clean water runoff around disturbed areas and detaining runoff from disturbance areas (dirty water); • Treating detained dirty water (for example, settling, flocculation) and discharging from site when water quality is within prescribed limits; • Minimising the volume of dirty water discharged from the site by maximising re-use (for example, dust suppression); • Providing engineering solutions to restrict runoff to sub-erosive velocities (for example, channels, dams, banks and drop structures); • Monitoring weather and undertaking mining activities in appropriate weather conditions; • Minimising disturbance areas; • Stabilising disturbance areas and stockpiles left dormant for extended periods (for example, establishing cover crops); and • Progressively rehabilitating disturbance areas.	May 2013 - draft), Section 3.2 includes the erosion and sediment control measures listed in the previous column. There is a dirty water segregation issue but that is best dealt with elsewhere in this audit.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.3	Erosion and sedimentation is monitored and reported in the Annual Review.	AEMR/Annual Review 2012/2013 Section 3.2 sighted	Compliant			
	3.2.4 Contaminated L						
MOP (Nov 2013 to Oct 2015)	3.2.4	Preliminary and detailed land contamination investigations were undertaken for the Project EA in accordance with Managing Land Contamination – Planning Guidelines SEPP 55 – Remediation of Land (Department of Urban Affairs and Planning and Environment Protection Authority 1998). Land contamination associated with historic landfilling and derelict farm machinery was identified in the project area. All instances of land contamination will be remediated before the land is used for mining.	Noted	Not Applicable			
MOP (Nov 2013 to Oct 2015)	3.2.4	Remediation activities and outcomes will be reported in the Annual Review.	AEMR/Annual Review 2012/2013 Section 3.17.2 stated that: No surface or groundwater contamination evident or reported by landowners; and No requirement for off-site disposal of contaminated materials. Therefore no reporting on remediation activities.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.4	A number of hazard control and mitigation measures for potential mine-related land contamination are described in the following management plans/systems in place at Tarrawonga: • Emergency Management System; • Pollution Incident Response Management Plan; • Contractor Management Standard; • Blast Monitoring Program; • Bushfire Management Plan (Appendix 12); • Surface Water and Groundwater Response Plan (part of the Water Management Plan) (Appendix 7); and • Waste Management Plan (Appendix 13).	Documents sighted: -PIRMP -Blast monitoring Program -Bushfire Management Plan (Appendix 12); -Surface Water and Groundwater Response Plan (part of the Water Management Plan) (Appendix 7); and -Waste Management Plan (Appendix 13)Emergency Management System -Contractor Mangement Standard	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.4	The following controls will be implemented at Tarrawonga to prevent contamination: • All water from wash-down areas and workshops will be directed to an oil separator and containment system; • All storage tanks will either be self bunded or partitioned within an impermeable bund with a capacity to contain a minimum 110% of the largest storage tank capacity; • All hydrocarbon products will be securely stored; and • With the exception of less mobile mining equipment which will be refuelled within the open cut area, designated areas will be allocated for refuelling and minor maintenance work.		Noted			
MOP (Nov 2013 to Oct 2015)	3.2.4	Where land contamination is identified arising from mining activities, the affected area will be regularly inspected to ensure there is no on-going effect on the land that will prevent it from being successfully rehabilitated. A register of known and potential contaminated sites will be maintained by TCPL for remediation at conclusion of mining activities in relevant areas.	No register of known and potential contaminated sites was sighted. The EO notes at interview that there were no known contaminated sites at the time of the audit and therefore no need for the register. Note also tht the site is still nominally over 15 years from closure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.4	All contamination monitoring activities and mitigation measures will be reported in the Annual Review.	AEMR/Annual Review 2012/2013 Section 3.17.2 stated that: No surface or groundwater contamination evident or reported by landowners; and No requirement for off-site disposal of contaminated materials.	Compliant			
	3.2.5 Acid Rock Drain		In a second				
MOP (Nov 2013 to Oct 2015)	3.2.5	Overburden and interburden at the northern and eastern open cut extension areas was assessed as part of the Project EA and it is generally expected to be non-acid forming (NAF). Consequently, the current management practice of blending ROM overburden and interburden for emplacement will continue.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.5	Acid generating materials will be managed in accordance with relevant requirements of Project Approval PA 11_0047 and EPL 12365, particularly in relation to surface water and groundwater management.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.5	Project Approval PA 11_0047 requires TCPL to develop a soil management protocol that includes management of coal reject or potentially acid forming interburden materials by emplacement in the pit shell such that there is no out of pit migration of acid or sulphate species. The soil management protocol will be developed in consultation with Namoi Catchment Management Authority and appended to the Biodiversity Management Plan during this MOP term. The BMP requires revision to address matters relating to the Commonwealth EPBC approval by the end of January 2014, which provides an appropriate timing for inclusion of the soil management protocol.		Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.5	Tarrawonga will conduct further material testing during soil stripping campaigns and will manage any occurrences of PAF-LC material by selectively handling and emplacing the material so it is covered with at least 15 m of NAF material.	No records on file of further material testing following that done for the EA for management of occurrence of PAF_LC material	Not Compliant	D	2	Medium
MOP (Nov 2013 to Oct 2015)	3.2.5	Tarrawonga will continue to undertake water quality monitoring within on-site water storages during the life of the mine in accordance with the Surface Water Monitoring Program. If in the event acid rock drainage (ARD) is identified, a works program for the identification of any PAF-LC material would be undertaken and specific controls implemented. All monitoring activities and the management of any PAF materials will be reported in the Annual Review.	AEMR/Annual Review 2012/2013, no mention of PAF LC material in this report. The water quality indicators for pH do not provide evidence of acid producing materials being exposed to the atmosphere.	Not Applicable			
	3.2.6 Flora and Faur	na					
MOP (Nov 2013 to Oct 2015)	3.2.6	Flora and fauna is currently managed at Tarrawonga in accordance with the Tarrawonga Flora and Fauna Management Plan. Project Approval PA 11_0047 requires the preparation of a Biodiversity Management Plan, which will supersede the current Tarrawonga Flora and Fauna Management Plan, to manage impacts to native flora and fauna in remnant native vegetation communities within the mine boundary.	Tarrawonga Mine Biodiversity Management Plan, May 2013, sighted doc	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.6	Under the EPBC Approval 2011/5923, Tarrawonga is permitted to clear: • Up to 13 ha of White Box – Yellow Box – Blakely's Red Gum Grassy Woodland and Derived Native Grassland critically endangered ecological community; • Up to 279 ha habitat for the Regent Honeyeater; • Up to 54 ha of habitat for the Swift Parrot, and Up to 334 ha of habitat for the Greater Long-Eared Bat.	EPBC Approval 2011/5923 sighted	Compliant			





Reference	Clause	Requirement	Evidence	Acadia Finalina		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.6	The impacts of clearing and loss of habitat will be minimised by: • Undertaking clearing on a progressive basis over the 17 years of mining, which will include: - Adopting clearing strategies to minimise impacts on fauna (timing, seasons); - Delineating areas to be cleared; and - Engaging a suitably qualified ecologist to be present during the clearing process. • Progressive rehabilitation of the post mine landforms, resulting in re-establishment of woodland/forest; • Salvaging and re-using material from the site where practical for habitat establishment; • Implementing a nest box program; • Enhancing farm dams; and • Controlling feral animals.	EPBC Approval 2011/5923, Section 1.1 sighted. All of this occurs with the exception of the farm dams to date, given there is no time limit there is still time left on the consent for this to occur.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.6	Other key management and mitigation measures are: • The rehabilitation of 752 ha of native woodland community focused on Box Gum Woodland EEC. • Compensatory habitat offset (legally binding conservation covenant) comprising a minimum of 1,660 ha of existing native vegetation to be enhanced and restored, including 193 ha of Box Gum Woodland EEC at the Willeroi Offset Area. In addition, this area will require equivalent or better quality habitat of no less than: - 1,055 ha for the Regent Honeyeater; - 397 ha habitat for the Swift Parrot; - 1,355 ha habitat for the Greater Long-Eared Bat; and - 232 ha White Box – Yellow Box – Blakely's Red Gum Grassy Woodland and Derived Native Grassland EEC.	EPBC Approval 2011/5923, Conditions of approval sighted. Wileroi has been established though only recently so little work has been done to date	Compliant			
	3.2.7 Weeds and Pes						
MOP (Nov 2013 to Oct 2015)	3.2.7	, , ,	BMP 2013, Section 4.7 sighted, it includes weeds and pests management measures.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.7	The objective of weed control activities at Tarrawonga is to limit the spread and colonisation of noxious and environmental weeds. Management measures include: • Regular inspections (minimum monthly) of proposed disturbance areas and adjacent land, topsoil stockpiles and rehabilitation areas are conducted for early identification of weed infestations. • Mechanical removal and/or the application of approved herbicides in areas identified as being affected by weeds (in accordance with the Pesticides Act 1999); • Follow-up site inspections to evaluate the effectiveness of weed control programs; • Follow-up weed control in previously treated areas where weed management has been suboptimal; and • Minimisation of potential seed transport to or from the site through the inspection of vehicles and use of the site's vehicle wash bay.		Noted			
MOP (Nov 2013 to Oct 2015)	3.2.7	TCPL will also liaise with local landholders and relevant government agencies as required to monitor the spread and management of weeds within the local area.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.7	Vertebrate pests are not considered a significant risk and control programs are currently only undertaken on an as-needs basis. Management activities include exclusion fencing, trapping and/or baiting programs, and follow-up monitoring programs.	Noted Also included in the BMP	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.7	Vertebrate pest management procedures will be formalised in this MOP period and documented in the Biodiversity Management Plan. TCPL will participate in regional pest control programs undertaken by neighbouring landholders, coordinated by the North West Livestock Health and Pest Authority (LHPA). Tarrawonga recently participated in an aerial control program for pigs and goats with the LHPA and surrounding landholders, and this form of control will be considered on an as needs basis in consultation with these parties.	Described in MOP and Biodiversity Management Plan.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.7	Weeds and vertebrate pests will be monitored in accordance with the Project Approval PA 11_0047 and the approved Biodiversity Management Plan. Monitoring results and any control activities will be reported in the Annual Review.	PA 11_0047,Biodiversity Management Plan and AEMR/Annual Review 2012/2013 sighted	Compliant			
	3.2.8 Blasting						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.8	Blasting is currently managed in accordance with the Blast Management Plan. Project Approval PA 11_0047 requires the preparation of: • A new Blast Management Plan (including Leard Forest Precinct Blast Management Strategy); and • A new Noise Management Plan (including Leard Forest Precinct Noise Management Strategy).	Tarrawonga Mine, Blast Management Plan (Last revision May 2013- Draft) is a draft doc and the Leard Forest Mining Precinct Blast Management Plan Strategy is reported to be awaiting approval states that: TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in developingand has developed a Leard Forest Mining Precinct Blast Management Plan Strategy is in draft format awaiting approval. Noise Management Plan TCM have been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in relation to developing a Leard Forest Mining Precinct Noise Management Strategy incorporating cumulative noise impacts and management, as required under the conditions of PA 11_0047 for the Tarrawonga Coal Mine. The Strategy has been submitted to the Department of Planning and Environment (DP&E) for approval. Once approved, the Strategy will be appended to this Plan.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.8	Measures to minimise potential impacts of blasting include: • Managing operations to comply with blast frequency limits; • Co-ordinating blasting days and times with Boggabri (and other mines within the Leard Forest Mining Precinct) to minimise disruptions and impacts on neighbouring landholders; and • Engaging with Narrabri Shire Council and private land owners if proposing to blast within 500 m of property not owned by TCPL/Whitehaven.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.8	Airblast overpressure and ground vibration will continue to be monitored at a selection of surrounding residences as provided in the Blast Management Plan and in accordance with the specifications presented in Table 10 in Section 4.0.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.8	In the event a landowner or resident claims property or infrastructure has been damaged as a result of blasting, TCPL will, within 3 months of receiving a written request, commission an investigation into the claim by a suitably qualified person whose appointment is approved by the Director-General of the DP&I. A copy of the investigation report will be provided to the landowner or resident and any damages caused by blasting repaired at the cost of TCPL.	Noted Also included in the existing noise management plan (Last revision 23/06/2014)	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.8	Blasting will be monitored in accordance with Project Approval PA 11_0047 and the approved Blast Management Plan and Noise Management Plan. Results will be reported in the Annual Review.	blasting results	Compliant			
	3.2.9 Noise	1					
MOP (Nov 2013 to Oct 2015)	3.2.9	Noise is managed in accordance with the Tarrawonga Road Noise Management Plan and Tarrawonga Noise Monitoring Program. Project Approval PA 11_0047 and EPL 12365 prescribe operational noise limits and management requirements.	Tarrawonga Road Noise Management Plan not sighted, but a document prepared earlier in the site history when it was run under a different name was provided and is still followed.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.9	As outlined in Section 3.2.8, TCPL will prepare and submit a Blast Management Plan and Noise Management Plan (including a Leard Forest Precinct Noise Management Strategy) to manage potential noise impacts in accordance with Project Approval PA 11_0047 during the MOP term. Road noise will continue to be managed under the Tarrawonga Road Noise Management Plan until coal ceases to be transported via public roads to the Whitehaven CHPP.	Tarrawonga Mine, <u>Blast Management Plan</u> (Last revision May 2013- Draft) is a draft doc and the Leard Forest Mining Precinct Blast Management Plan Strategy is reported to be awaiting approval. TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in developing and has developed a Leard Forest Mining Precinct Blast Management Plan Strategy that is in draft format awaiting approval. Noise Management Plan TCM have been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in relation to developing a Leard Forest Mining Precinct Noise Management Strategy incorporating cumulative noise impacts and management, as required under the conditions of PA 11_0047 for the Tarrawonga Coal Mine. The Strategy has been submitted to the Department of Planning and Environment (DPE) for approval. Once approved, the Strategy will be appended to the NMP. None of these plans are approved.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.9	Noise mitigation measures for the mine within the MOP period will include: • Modification of the alignment of haul routes to reduce their exposure relative to nearby receivers; • A reduction in the number of mobile fleet items operating during the evening and night-time periods and/or consideration for relocation of equipment to other areas of the mine where noise monitoring indicates systemic noise impacts; and • Implementation of reasonable and feasible acoustical mitigation measures at impacted receivers in consultation with the affected landholder.	· ·	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Filluling	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.9	Implementation of the above mitigation measures will be guided by the operation of the real time noise management system and ongoing attended noise monitoring to verify noise performance.	Tarrawonga Noise Management Plan (June 2014) Section 1, states that noise impacts are continuously monitored via a mobile real time noise monitor under all meteorological conditions, with measures to reduce noise impacts undertaken when required. Evidence of this was observed on-site.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.9	Tarrawonga is approved to operate 24 hours a day seven days per week, however this is not expected to occur during this MOP period. Until the commissioning of the Boggabri Rail Spur Line and Boggabri CHPP, the transport of the coal from Tarrawonga to the Whitehaven CHPP is restricted to: • 7:00 am to 9:15 pm Monday to Friday; and • 7:00 am to 5:15 pm Saturdays.	Noted Tarrawonga Noise Management Plan (June 2014), information on restricting times not sighted in this document	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.9	There will be no transport of coal between Tarrawonga and Whitehaven CHPP on Sundays or public holidays.	Noted Tarrawonga Noise Management Plan (June 2014), information on no transport of coal not sighted in this document	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.9	The transport of gravel from Tarrawonga is limited to truck transport hours of 7:00 am to 6:00 pm Monday to Saturday.	Noted Tarrawonga Noise Management Plan (June 2014), information on transport limited to trucks not sighted in this document	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.9	Project Approval PA 11_0047 requires TCPL to acquire up to five properties, which are identified as properties 43, 44, 45, 47 and 49 (see Appendix 5 of the Project Approval PA 11_0047), upon written request from the owners.	Tarrawonga Noise Management Plan (June 2014), Section 2.2 stated that: Of the properties identified in Table 1 from the Project Approval, properties 43, 44, 45 and 47 have now been acquired by either Tarrawonga or Boggabri Coal. Preliminary discussions have been held with property 49 in relation to their rights for acquisition.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.9	Assessment of cumulative noise from the concurrent operation of Tarrawonga, Boggabri and the Maules Creek Coal Project as part of the Tarrawonga Coal Project EA indicated that cumulative noise levels will comply with the specified night-time maximum amenity criteria at all receivers and the specified night-time acceptable amenity criteria for all but two privately-owned receivers. Tarrawonga proposes to establish, in co-operation with Boggabri and the Maules Creek Coal Project, a cumulative noise monitoring program.	Tarrawonga Noise Management Plan (June 2014), foreword section, stated that: TCM have been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in relation to developing a Leard Forest Mining Precinct Noise Management Strategy incorporating cumulative noise impacts and management, as required under the conditions of PA 11_0047 for the Tarrawonga Coal Mine. The Strategy has been submitted to the Department of Planning and Environment (DPE) for approval. Once approved, the Strategy will be appended to this Plan.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.9	Noise will be monitored in accordance with Project Approval PA 11_0047 and the approved Blast Management Plan and Noise Management Plan, and results will be reported in the Annual Review.	This occurs (AEMRS) and is reported.	Compliant			
	3.2.10 Visual and Lig	hting					
MOP (Nov 2013 to Oct 2015)	3.2.10	Whilst a specific visual management plan is not a requirement of Project Approval PA 11_0047, the PA does impose conditions on minimising visual impacts from operations and a response process to private landholders concerned about visual impacts when viewed from private residences.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.10	Visual impacts associated with the mining operations will be managed in accordance with Project Approval PA 11_0047 and the required, • Biodiversity Management Plan, and • Rehabilitation Strategy and the associated Rehabilitation Management Plan	Noted RMP and PA include visual impacts No visual impacts included in the BMP 2014 but it does address planting and rehab which address visual impact	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.10	TCPL commits to preparing these plans and strategy during the MOP term	Pland and strategy in place	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.10	The major aspects of Tarrawonga considered to have the potential to impact on the visual landscape include: • Modification of topographic features including: - the extension of the open cut; - the extension of the Northern and Southern Emplacements; - a temporary increase in height of the Southern Emplacement prior to a reduction in final height during rehabilitation; - construction of the permanent flood bund; - construction of the noise control earth bund; and - establishment of the permanent Goonbri Creek alignment; • Re-alignment of sections of Goonbri Road; and • Extension of lighting associated with extended night-time mining operations.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.10	The final elevation of the Northern Emplacement would remain unchanged at a height of 370 m AHD. The height of the Southern Emplacement would temporarily increase to a maximum height of 360 m AHD (i.e.: an increase of 20 m) during the operational life of the mine. The Southern Emplacement will be reduced to a final height of approximately 340 m AHD. The additional material will be rehandled and emplaced in-pit when space becomes available.	Noted Info not included in the Rehab Management Plan	Noted			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.10	Cumulative visual impacts of Tarrawonga and the proposed Continuation of Boggabri Coal Mine are anticipated to be limited to viewpoints from the southern and western sides of Tarrawonga associated with elevated areas where no vegetation screening is present (e.g.: from portions of cleared paddocks and private roads).	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.10	The mitigation and management measures that would be implemented for the maintenance of visual amenity at Tarrawonga include: • Progressive rehabilitation of the Northern Emplacement, Southern Emplacement, open cut and mine infrastructure areas will be undertaken in order to reduce the contrast between the landforms at Tarrawonga and the surrounding environment. • Rehabilitation will be conducted in accordance with the Rehabilitation Strategy and Rehabilitation Management Plan (and as updated as required by the PA). • Visual screening (e.g.: a vegetation screen consisting of native plants that are compatible with the existing surrounding vegetation) will be used to reduce potential visual impacts from local sensitive viewpoints. Vegetation screens will be established on the noise control earth bund and along the re-aligned sections of Goonbri Road to reduce potential views of the Tarrawonga landforms. For the re-aligned sections of Gonbri Road, the vegetation screens would be planted in advance of the re-alignment works, in order to reduce direct views of the mine once the re-alignments are completed. • Responding to private landholder concerns about visual impacts, by investigating and if appropriate vegetative screening or other measures in consultation with the affected landholder.		Noted			
MOP (Nov 2013 to Oct 2015)	3.2.10	The potential for fugitive night-lighting emissions to be visible at additional locations may change during the MOP term due to the: • Increased elevation of light sources on the Southern Emplacement; • The increased extent of the mine waste rock emplacements; and • Potential increases in the mine fleet and operational hours and consequently there would be an increase in mobile vehiclemounted night-lighting effects.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.10	Tarrawonga will seek to minimise light emissions by carefully selecting the sites where lighting plants or permanent lighting installations would be placed. Measures that would be employed to mitigate potential impacts from night-lighting would include one or more of the following, where practicable: • All external lighting associated would comply with AS 4282: 1997 – Control of the Obtrusive Effects of Outdoor Lighting; • Night-lighting would be restricted to the minimum required for operations and safety requirements; • Directional lighting techniques would be used; • Light shrouds and reflectors would be used to limit the spill of lighting; • The permanent flood bund and the noise control earth bund would be revegetated to establish visual screens; Note: Not required within this MOP period. • In consultation with the landholder, trees would be planted at nearby private dwellings to help screen identified adverse night-lighting impacts in the event that significant direct night-lighting views are available; and. • In consultation with the landholder, curtains, cladding and/or screens would be provided at nearby private dwellings to help screen identified adverse night-lighting impacts in the event that significant direct night-lighting views are available.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.10	Visual and lighting responses will be monitored in accordance with the Project Approval and results will be reported in the Annual Review.	Noted	Noted			
	3.2.11 Heritage (abor		Townsummer Heritage Management Star (Last	1			
MOP (Nov 2013 to Oct 2015)	3.2.11	Aboriginal heritage is currently managed in accordance with the Aboriginal Heritage Management Plan. This Plan will be revised and updated to account for the proposed expansion of Tarrawonga in accordance with the PA to prepare a new Heritage Management Plan. This Plan will be developed in consultation with Aboriginal stakeholders and OEH.	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014) mentions that the mine site operates under Project Approval (PA) 11_0047 (granted 22 January 2013).	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	The following general approach will be taken to manage Aboriginal cultural heritage during the life of the mine and be incorporated into the revised Heritage Management Plan:	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.11	A record of known Aboriginal sites, their status and location would be maintained by Tarrawonga and relevant site personnel aware of the site locations.	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014), Appendix 2 Aboriginal Heritage Sites Recorded at the Tarrawonga Coal Mine	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.11	Ongoing consultation would be undertaken with the Aboriginal community over the life of the mine. Appropriate Aboriginal representation would be facilitated during archaeological fieldwork (e.g. salvage of artefacts prior to disturbance).	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014), Section 6.1. Maintain ongoing consultation with the RAP's over the life of the mine, including appropriate RAP representation during topsoil stripping, disturbance activities and archaeological fieldwork (e.g. collection of artefacts);	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	Tarrawonga would provide opportunities for Aboriginal community members to access known Aboriginal sites located on Whitehaven-owned land (e.g. for cultural reasons or as part of scheduled field activities). Such access would be subject to Occupational Health and Safety requirements.	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014), Section 4. If requested by the RAP's, TCPL will facilitate access to known sites within the mining lease areas, along the Haul Route and in the Biodiversity Offset Area, subject to satisfaction of required safety protocols.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	Erosion and sediment control works would be undertaken in accordance with the requirements of the Erosion and Sediment Control Plan and in consideration of Aboriginal cultural heritage management measures.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.11	Any additional Aboriginal heritage sites which may be identified during the development of the mine would be recorded and registered with the OEH in consultation with Aboriginal stakeholders. Should additional Aboriginal heritage sites be identified, they would be managed in accordance with the measures described in the Heritage Management Plan.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.11	Where avoidance of known Aboriginal heritage sites is not practicable, site(s) would be subject to baseline recording, in consultation with representatives of the Aboriginal community, prior to disturbance and artefacts would be salvaged for safekeeping in accordance with the wishes of the Aboriginal community and OEH.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.11	Scar trees located in disturbance areas would be considered for salvage, if appropriate in the context of the tree's condition. A suitable location for the storage and/or display of the salvaged sections would be identified and managed in consultation with the Aboriginal community and OEH.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.11	Culturally modified trees located outside of (but in close proximity to) the mine disturbance areas would be suitably demarcated and signed to reduce the risk of accidental damage if considered appropriate by the Aboriginal community.	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014), Section 4.2 states: Culturally modified trees will be fenced at a distance of 5m from the tree where practicable. The fence line location will be subject to physical constraints such as other vegetation and topography Fence posts will be marked white and set vertically apart at approximate 5m intervals; Observed.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	It is anticipated that the Aboriginal community would also provide advice on the management of salvaged artefacts at the completion of mine activities (e.g.: artefact replacement onto the post mining landscape).	Revision: Jan 2014), Table 1 Roles and	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	In the event of inadvertent damage to any Aboriginal site or place, the activities causing the damage would cease immediately and procedures identified in the revised Heritage Management Plan followed. Subject to the nature of the damage, appropriate professional advice in addition to that provided by Tarrawonga's consultant archaeologist may be sought.	Tarrawonga Heritage Management Plan (Last Revision: Jan 2014), Section 3.1 In the event that other items of potential heritage significance are uncovered during mining operations, the following protocol will be followed: 1. Work will cease in the area of discovery immediately; 2. The person discovering the item will notify their supervisor immediately who will ensure that work has ceased and area(s) is (are) cordoned off with tape; 3. The supervisor will notify the Operations Manager or senior TCPL person on site; 4. The Operations Manager/Environmental Manager will: a. arrange a qualified heritage management consultant to attend the site and advise on its heritage significance at their earliest opportunity; and b. if the find is determined to be of heritage significance, notify OEH Heritage Branch with the advice from the heritage consultant and seek relevant advice as to appropriate protocols in terms of relocation and notification requirements. There has been no inadvertant damage to aboriginal heritage in the audit period.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	Aboriginal cultural heritage will be monitored in accordance with the revised Heritage Management Plan and the results summarised in the Annual Review.	AEMR 2012-2013 includes results summarised	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.11	There are no items of state or regional non-Aboriginal heritage significance in the vicinity of Tarrawonga.	Noted No info included in the AHMP	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.11	While of no heritage significance, a rubbish dump identified within the proposed mine contains objects that may be of interest to local historical collectors (e.g. old car and truck bodies, farm equipment, an oil engine and building materials). Prior to mine disturbance of the rubbish dump, these objects will be offered to the Boggabri Historical Society and the Gunnedah Museum.	Noted Information also included in the AHMP	Noted			
	3.2.12 Spontaneous	s Combustion					
MOP (Nov 2013 to Oct 2015)	3.2.12	The EMS details the roles and responsibilities of site personnel and the procedures to be undertaken in the event of an environmental incident, such as spontaneous combustion. This includes notification requirements and contingency measures.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Testing was conducted on each coal seam to be exposed by mining at Tarrawonga, with 0.44% sulphur content being the highest recorded value (Velyama Seam). The low percentage of inorganic sulphur is indicative of a low potential for exothermic oxidation reactions. This has been demonstrated over the past seven years of operation, with only one instance of spontaneous combustion occurring over that time.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Protocols for managing potential spontaneous combustion on coal stockpiles include: • Regular inspection of ROM coal stockpiles as part of the site supervisor's check areas for evidence of combustion (visual and/or smell) and general stockpile maintenance.	Noted, no evidence of current or past spon com noted in the site inspection conducted for the audit	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	General short residence time of ROM coal stockpiles. Tarrawonga maintains three main stockpiles (TW07r, TW12r and TWxr) based on product type. General residence time of TW07r and TW12r is maintained at approximately 2 weeks. While the TWxr stockpile can have a residence time of up to 2 months, it is managed to minimise the potential for spontaneous combustion by the dumping of fresh coal on the northern side of the stockpile and coal being taken for loading from the southern side of the stockpile. This ensures continual cycling of coal through the stockpile.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Stockpile heights are retained at less than 15 m to assist in limiting residence time.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Stockpiles are subject to weekly survey to confirm height and volume.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	In longer term stockpiles (for example, TWxr) shallow stockpile batter faces are created to the direction of the prevailing wind.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.12	Mobile equipment is used to compact the stockpile area.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Tarrawonga Coal personnel and contractors working in proximity to the ROM coal stockpiles have received appropriate training to be alert for, and respond to, indications of spontaneous combustion. Any incident would be followed by excavation of the stockpile to locate the source, and extinguishment by water saturation.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Overburden and interburden sampling by Geo-Environmental Management in 2011 identified an average total sulphur content of 0.03% across both the eastern and northern extension areas and concluded overburden and interburden is unlikely to be reactive; Regardless, the site implements operational protocols to minimise the risk of spontaneous combustion.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Protocols for overburden and rejects emplacement include: • Selectively handling partings, interleaves and other carbonaceous waste rock for in-pit emplacement in the pit floor to maximise cover with inert material.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.12	Emplacing coarse rejects (back hauled from the Whitehaven CHPP) in the pit floor to maximise cover with inert material. Sampling of coarse reject by Geo Environmental Management in 2011 identified an average total sulphur content of 0.4%. The material is managed by emplacement in the pit floor and subsequent coverage by the advancing overburden emplacement. This is in accordance with the existing Section 100 approval at the Tarrawonga site for reject emplacement.	Noted Site observations indicate that the rejects are not placed on the pit floor as required by the MOP but are placed at higher elevations as part of the backfilling process with suitable cover to prevent spon com. There was no evidence of AMD. Recommendation - the wording of the MOP should change at the next revision.	Not Compliant Administrative			
MOP (Nov 2013 to Oct 2015)	3.2.12	Spontaneous combustion events (if any) will be reported in the Annual Review.	AEMR 2012-2013, Section 3.14 includes information on spontaneous combustion	Compliant			
	3.2.13 Bushfire						
MOP (Nov 2013 to Oct 2015)	3.2.13	Bushfire is currently managed under the Tarrawonga Environmental Management Strategy (EMS) and the Tarrawonga Bushfire Management Plan. The Biodiversity Management Plan will also address bushfire prevention and mitigation of bushfire impacts. TCPL commits to preparing the Biodiversity Management Plan and updating the Bushfire Management Plan during the MOP term.	Biodiversity Management Plan (May 2013) and updating the Bushfire Management Plan (Mar 2013) MOP term started in Nov 2013 (after the above plans were compiled). Therefore the above MP are to be updated	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	A range of management measures are in place to manage the behaviour of people at Tarrawonga and the overall risk of increased bushfire frequency due to mine activities is likely to be low.	Bushfire Management Plan 2013 - includes management measures. Environmental Risk Assessment 2011 (Appendix 4 of MOP (Nov 2013 to Oct 2015) confirms that increased bushfire risk is low.	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.13	The bushfire management measures in place at Tarrawonga include: • Clearing restrictions – clearing will not be undertaken during periods of extreme fire danger as defined by the Bureau of Meteorology.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Controlled grazing – controlled high intensity short-term grazing on vegetated areas where stock fencing is available may be employed (if appropriate).	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Operational activities (eg. vehicle movements, welding, refuelling, blasting) – are managed to minimise the risk of ignition or spread of fire.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Fire breaks – fire breaks will be maintained around the perimeter of the mining leases.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	No smoking areas – smoking will be prohibited in fire prone areas.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Fire fighting equipment – on-site fire fighting equipment will be provided, including a fully equipped fire tender to provide immediate response to a bushfire.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Training – all relevant mine personnel will receive basic fire control training.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Assistance would be sought from the NSW Rural Fire Service in the event of significant bushfire incident or offsite bushfire potentially impacting upon the mine site.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
MOP (Nov 2013 to Oct 2015)	3.2.13	Bushfire events and mitigation measures will be reported in the Annual Review.	Bushfire Management Plan 2013 - includes this management measure.	Compliant			
	3.2.14 Other Issues	identified in Risk Assessment					
MOP (Nov 2013 to Oct 2015)	3.2.14	Traffic impacts associated with the Goonbri Road intersections and public road impacts were identified in the ERA. These impacts are being alleviated by the reconstruction and realignment of Goonbri Rd and associated intersections.	This reallignment has not yet occurred and has been postponed due to the sites financial situation. Recommendation - TCL should reassess the risk to the public and put in place other mitigation measures if the road reallignment is not taking place.	Not Applicable			
MOP (Nov 2013 to Oct 2015)	3.2.14	Whilst identified in the ERA, rehabilitation is the key focus of this MOP and is covered in detail throughout this document.	Noted	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.14	Mining activities at Tarrawonga will reduce the area of Class 4 agricultural suitability land at the site by approximately 125 ha in the long-term. The proposed rehabilitation of Class 3 agricultural suitability lands will result in no long-term change in the area of Class 3 agricultural suitability lands on the site.	Noted	Noted			





Reference	Clauses	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	3.2.14	Agricultural land resource management at Tarrawonga will include the following key components: • Minimisation of disturbance to agricultural lands, where practicable; • Continued use of adjoining Whitehaven-owned land for agricultural uses; • Management of soil resources at the site so that they can be used for rehabilitation; and • Inclusion of agricultural lands in the Tarrawonga rehabilitation strategy (i.e. some 210 ha of Class 3 agricultural suitability land).	Noted, the area noted in the MOP as agricultural land will not be rehabilitated for a number of years from the time of the audit.	Noted			
MOP (Nov 2013 to Oct 2015)	3.2.14	Rehabilitation is a key focus of this MOP and the outcomes will be included in the Annual Review and during MOP reviews.	MOP reviews have not taken place but the MOP was not 12 months old when the audit took place AEMR 2012-2013 includes information on rehab performance	Not Applicable (Rehab information not included in MOP reviews) Compliant (Rehab information included in the AEMR)			
	3.3.3 Geology and G	eochemistry					
MOP (Nov 2013 to Oct 2015)	3.3.3	Tarrawonga will undertake testing to identify any PAF materials. PAF material will be selectively emplaced under at least 15 m of NAF material to minimise the potential for ARD. Volumes of PAF material identified and selectively emplaced will be reported in the Annual Review.	AEMR 2012-2013 does not include information on PAF materials	Not Compliant	D	2	Medium
MOP (Nov 2013 to Oct 2015)	3.3.3	To minimise erosion, the final outer surfaces of the overburden emplacements will be constructed with suitable non-sodic or low sodicity material and/or will be treated with gypsum.	Noted, no evidence of sodicity in the rehabilitated faces of the emplacements.	Noted			
MOP (Nov 2013 to Oct 2015)	3.3.3	Sodic material identified in the final void highwalls and lowwalls will be covered with backfill and/or ameliorated in accordance with the Final Void and Mine Closure Plan incorporated in the Rehabilitation Management Plan.	Detailed in the Rehabilitation Management Plan (May 2013) though no sodic material identified to date, note the lack of testing may influence this outcome	Not Applicable			
		nning and Management					
	5.4.2 Decommission	ing					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	5.4.2	Tarrawonga will decommission fixed plant, built infrastructure and services progressively when infrastructure items and plant become redundant. All mining related infrastructure will be removed at mine closure. Decommissioning activities include: • Disconnection of all above ground and buried services and removal of associated infrastructure; • Removal of all built infrastructure and plant; • Removal of all wastes and hazardous materials; and • Removal (or on-site remediation) of any contaminated soils in accordance with a contaminated land assessment (where required).	Rehabilitation Management Plan (May 2013) Note no infrastructure decommissioned yet.	Not Applicable			
	5.4.3 Landform Estab	lishment and Surface Preparation					
MOP (Nov 2013 to Oct 2015)	5.4.3	The shape of the post mining landform is critical to achieving the rehabilitation objective of a safe, structurally stable, landform that complements the surrounding landscape. Final landforms will have acceptable slopes and unimpeded drainage lines. Slopes are generally to be graded to no more than 10 degrees and no more than 18 degrees without DRE's approval, as per DRE's guidelines.	Rehabilitation Management Plan (May 2013) No steep slopes noted in the site inspection	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.3	The Northern Emplacement will be shaped to integrate with the Boggabri Coal Mine waste rock emplacement to the north and the Leard State Forest to the east, providing continuity between the Tarrawonga and Boggabri Coal Mine post mining landforms, and habitat connectivity with undisturbed vegetation communities at Leard State Forest. Both the Northern Emplacement and Southern Emplacement will be shaped to integrate with the open cut infill area.	Rehabilitation Management Plan (May 2013)	Noted			
MOP (Nov 2013 to Oct 2015)	5.4.3	Waste rock will be selectively handled and blended to avoid emplacement of potentially acid forming material in concentrated areas. As outlined in Section 3.2.5 where PAF-LC material is identified it will be encapsulated with at least 15 m of NAF to minimise potential for acid leachate.	Rehabilitation Management Plan (May 2013) no PAL mateials identifed to date.	Not Applicable			
MOP (Nov 2013 to Oct 2015)	5.4.3	Dispersive materials will be avoided for material layers at the final landform surface where practical to minimise potential for significant scouring or land slumping. Non-sodic and low sodicity materials will be selectively handled and preferentially emplaced at or near the surface. Where there are potentially dispersive materials emplaced at or near the surface, appropriate amelioration with lime or gypsum will be undertaken to stabilise soils, particularly foundation materials for earthworks structures such as contour banks.	Rehabilitation Management Plan (May 2013)	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding	Risk		
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	5.4.3	By definition Agricultural Class 3 land requires soil conservation practices to support a cropping landuse. Lower terraces to be rehabilitated to a Class 3 standard will be graded at no more than 2 per cent grades, and incorporate contour banks to: • Produce free draining landforms that minimise soils at depth becoming water logged; and • Impede runoff velocities to sub-erosive levels.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.3	To optimise growing conditions for deeper rooted native grasses and crops, the surface layers of spoil will be: • Selectively emplaced with non-sodic or low sodicity spoils and subsoils; • Rock-raked to remove any large rocks that may impede ripping tynes and agricultural equipment; and • Deep ripped along the contour and ameliorated with coarse gypsum prior to soil re-spreading.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.3	General principles that may be adopted to make the final void safe and stable include: • Battering back the lowwalls and highwall to minimise potential for failures and mass movement; • Capping (or excavating) exposed coaliferous material with inert material to prevent ignition from spontaneous combustion, bushfires or human interference; • Constructing a physical barrier to isolate the perimeter of the void to prevent human access. The highwall areas will be secured by the construction of a trench and a safety berm, as well as a security fence along the entire length of the eastern and northern highwalls; • Suitable signs, clearly stating the risk to public safety and prohibiting public access will be erected at intervals along the entire length of the fence; and • Surface runoff from land surrounding the void will be diverted so as to prevent any potential development of instability of the void walls.		Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.3	Elements such as drainage paths, contour drains, ridgelines, and emplacements will be shaped, as much as practical, to undulating profiles in keeping with natural landforms of the surrounding environment. Contour and catch drains are designed to collect surface runoff from rehabilitation or disturbed areas. Sedimentation dams will be constructed and retained during rehabilitation to collect runoff from rehabilitated areas until discharge water quality meets regulatory criteria and dams can be decommissioned.	Rehabilitation Management Plan (May 2013)	Compliant			
	5.4.4 Growth Media	a Develonment					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Filluling	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	5.4.4	In the context of this MOP, growth media development encompasses activities to reinstate soils with the required initial physical, chemical and biological characteristics; ongoing management actions to develop self-sustaining soil profiles are discussed in the Ecosystem Development Rehabilitation Phase (Section 5.4.5)	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.4	Topsoil management procedures are described in Section 2.4.2. During this MOP term soil resources suitable for Woodland Rehabilitation Areas will be re-spread in rehabilitation areas at the Northern and Southern Emplacements (refer to Maps 3a to 3c).	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.4	Installation of fauna refuge structures (habitat augmentation) will be undertaken during this MOP term in Woodland Rehabilitation Areas to encourage desired fauna species into the rehabilitation areas. Felled timber that is not mulched and salvaged during land clearing will be replaced on topsoiled areas and provides habitat for native reptiles. Hollow bearing timber will be reinstated in rehabilitation areas, and is supplemented with nest boxes and bat boxes (where appropriate) to improve habitat opportunities for native reptiles and mammals, birds and microbats.	timber noted in stockpiles and on rehab areas during site inspection	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.4	Dams retained in the final landform will include habitat enhancements including woody debris placed adjacent to the wet edge of dams (where appropriate), submerged logs, and plain wire fencing to enclose portions of dams for habitat refuge and exclude vertebrate pest species. Final landform design may also include undulations to produce small drainage depressions (where appropriate) to provide habitat for frogs and water-borne invertebrates.	Rehabilitation Management Plan (May 2013), no dams in final rehab as yet	Not Applicable			
	5.4.5 Ecosystem Est	ablishment					
MOP (Nov 2013 to Oct 2015)	5.4.5	In the context of this MOP, ecosystem establishment includes activities to establish the desired floristic composition including: • Seeding, tubestock planting and transplanting; and • Activities to manage risks to vegetation establishment such as weed and vertebrate pest management, and bushfire risk mitigation works.	Rehabilitation Management Plan (May 2013)	Noted			
MOP (Nov 2013 to Oct 2015)	5.4.5	Native vegetation areas are planted with native tubestock species, selected on a site by site basis considering factors such as adjacent remnant vegetation communities, aspect, slope, and soil types.		Noted			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	5.4.5	Woodland Rehabilitation Areas will include areas planted with species commensurate with the BoxGum Woodland EEC. Retained dams and watercourses including the Goonbri Creek Alignment will be planted with riparian species. The Final Void domain will be planted with native vegetation tubestock above the predicted permanent water line.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.5	Tubestock planting is preferentially undertaken in Autumn and early Winter when weather conditions are generally suited to avoid heat stress on newly planted seedlings; however opportunistic plantings are undertaken in spring and late summer following suitable rainfall. Tubestock seedlings are generally propagated from seed collected on-site and in adjacent areas or sourced from suppliers in the region. Tubestock is typically planted at approximately 400 trees per hectare in native vegetation areas. In areas of planting within the Leard State Forest and where the rehabilitation blends with the Boggabri Coal final landform, planting densities will increase consistent with the approach undertaken by Boggabri Coal		Noted			
MOP (Nov 2013 to Oct 2015)	5.4.5	Tubestock plantings are mulched, watered and protected from grazing by tubestock bags and/or fencing where required. Tubestock planting methodologies are described in the Rehabilitation Management Plan and Rehabilitation Monitoring Report (2012 Annual Review, Appendix 7). Further specific rehabilitation methodologies for establishing the woodland and agricultural final land-uses are under development and will be documented in the Rehabilitation Management Plan.	Rehabilitation Management Plan (May 2013) Rehabilitation Monitoring Report (2012 Annual Review, Appendix 7), provided Tubestock is not mulched at the site but is incorporated into the topsoil, no evidence of excessive grazing of tubestock in the site inspection. Recommendation - remove reference to guards and mulching from MOP.	Not Compliant Administrative			
MOP (Nov 2013 to Oct 2015)	5.4.5	TCPL will also trial direct seeding of eucalypts and some understorey species for woodland areas. If these trials are successful consideration will be given to utilising direct seeding to rehabilitate open woodland rehabilitation areas. Direct seeding research trials are discussed further in Section 8.2.2.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.5	Preliminary species lists currently used for native vegetation planting at Tarrawonga are shown in Table 18. TCPL will consult relevant government agencies and stakeholders in this MOP period to finalise suitable native species mixes for all rehabilitation areas, and document the species lists in the Rehabilitation Management Plan. TCPL will consult Forests NSW regarding inclusion of commercially desirable species for future timber harvesting for rehabilitation areas adjacent to the Leard State Forest (Northern Emplacement Area).		Compliant			
\	Weed and Vertebra	te Pest Control					





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	5.4.5	Management measures include: • Scalping any weeds from topsoil stockpiles prior to re-spreading in rehabilitation areas; • Ensuring all plant and equipment are weed free prior to mobilisation to rehabilitation areas; • Maximising the retention of ground cover (cover crop stubble) when planting tubestock to minimise opportunities for weed activity; and • Installation of fauna exclusion fencing and/or tree guards for newly planted tubestock where predation by grazing herbivores represents a risk to establishment.	Rehabilitation Management Plan (May 2013) Tree guards and fauna exclusion fencing not used.	Not Compliant	E	2	Low
	5.4.6 Ecosystem Dev						
MOP (Nov 2013 to Oct 2015)	5.4.6	In the context of this MOP, the Ecosystem Development phase constitutes the medium to long term management of rehabilitation areas toward achieving the desired final land-use. This phase includes: Ongoing environmental management to minimise risks to rehabilitation; Comparing specific ecosystem characteristics such as soil profile development, floristic composition and structure and faunal diversity and abundance with the characteristics of appropriate analogue sites; and Undertaking adaptive management and remedial works where characteristics of the rehabilitation are not trending toward desired outcomes	Rehabilitation Management Plan (May 2013), still early in this phase in most areas of the site that have been rehabilitated though the western face of the norther e emplacmeent is now well established	Compliant			
	Rehabilitation Moni	toring					
MOP (Nov 2013 to Oct 2015)	5.4.6	Ecosystem Development phase incorporates rehabilitation monitoring programs to track progress toward the quantitative and qualitative completion criteria. Monitoring programs in suitable analogue sites is undertaken to refine rehabilitation benchmarks that constitute the indicators and completion criteria for Ecosystem Development phase.	Rehabilitation Management Plan (May 2013)	Compliant			
	Intervention and Ad	aptive Management		T	T T		
MOP (Nov 2013 to Oct 2015)	5.4.6	Where monitoring results reveal that indicators are not trending towards the completion criteria as predicted, plans should be executed to promptly modify management practices to achieve the desired rehabilitation results.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	5.4.6	Early intervention will be actioned in the form of a Trigger Action Response Plan (TARP), to be developed for each rehabilitation domain, to minimise the risk of failure to achieve the desired rehabilitation outcomes for each domain	Rehabilitation Management Plan (May 2013)	Compliant			
	7 Rehabilitation Imp						
	7.2 Proposed Rehab	ilitation and Disturbance Activities this MOP term					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	7.2	During this MOP term rehabilitation activities will continue to be associated with progressive rehabilitation works at the Northern Emplacement and Southern Emplacement, as well as progressive back filling of the open cut pit as mining advances to the east.	Noted	Noted			
	8 Rehabilitation Mon	itoring, Research and Reporting					
	8.1 Rehabilitation Mo	•					
MOP (Nov 2013 to Oct 2015)	8.1	TCPL undertakes annual rehabilitation monitoring to provide quantitative and qualitative data to: • Assess rehabilitation progress against completion criteria and/or triggers for re-work and adaptive management; and • Assist in refining rehabilitation methods.	Rehabilitation Monitoring Reports sighted	Compliant			
	8.1.1 Active Mining R	ecords					
MOP (Nov 2013 to Oct 2015)	8.1.1	TCPL maintains active mining records related to processes that may impact on future rehabilitation quality. These records contribute to future assessment of rehabilitation outcomes and inform the continual improvement process.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.1	Records maintained include the following: Detailed rehabilitation procedures; Register of hazardous and explosive materials; Register of contaminated sites; Records of production wastes and other waste streams and where they are located on site; Environmental monitoring records, including surface and groundwater quality and detailed annual rehabilitation monitoring reports from qualified ecologists; A topsoil and subsoil stockpile register which includes the date stockpiles were formed and maintenance works undertaken (e.g. weed control, planting with cover crops etc.), and	Rehabilitation Management Plan (May 2013) No register of known and potential contaminated sites was sighted. The EO notes at interview that there were no known contaminated sites at the time of the audit and therefore no need for the register. Note also tht the site is still nominally over 15 years from closure.	Compliant			
	8.1.2 Rehabilitation N	Methodology Records					
MOP (Nov 2013 to Oct 2015)	8.1.2	TCPL records the details of each rehabilitation campaign so that they are available to provide context for rehabilitation monitoring results and assist the continuous improvement process.	Rehabilitation Management Plan (May 2013)	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	8.1.2	The key monitoring parameters to be included in the program include: • Landform and drainage design details; • Substrate (spoil) characterisation; • Site preparation techniques (for example, topsoil source and depth, soil ameliorants used etc.); • Revegetation methodologies (for example, rate and type of fertiliser, cover crop and rate, time of sowing /planting); • Weather conditions; • Photographic records; and • Initial follow-up care and maintenance works.	Rehabilitation Management Plan (May 2013)	Compliant			
	8.1.3 Annual Rehabili						
MOP (Nov 2013 to Oct 2015)	8.1.3	TCPL engage ecologists to undertake spring monitoring campaigns at: • Existing Woodland Rehabilitation Area sites at the Northern Emplacement and Southern Emplacement; • Regenerating native vegetation in the north east corner of ML 1579; and • Analogue sites located at the southern extent of ML 1579 (agricultural land control sites) and in undisturbed native vegetation to the east of ML 1579 (Woodland analogue site).	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.3	Monitoring at rehabilitation areas and analogue sites includes survey programs tailored for both Agricultural and Woodland Rehabilitation Areas Secondary Domains.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.3	Vegetation and fauna surveys in Woodland sites monitor vegetation diversity and abundance in each stratum and presence of arboreal and terrestrial fauna species. Woodland surveys also include soil profile characterisation and measure the accumulation of leaf litter, bare patches and presence of weeds.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.3	Surveys in pastoral control sites monitor the pastoral vegetation species, total biomass per hectare and soil physical-chemical data. Rehabilitation monitoring outcomes are reported annually in the Annual Review.	Rehabilitation Management Plan (May 2013) AEMR 2012-2013 mentioned that as rehabilitation of pasture areas had only recently been undertaken, monitoring of the rehabilitated pasture will take place and be reported to during the next reporting period.	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.3	Monitoring at the analogue sites will be used for further development of quantitative indicators and completion criteria for Agricultural Rehabilitation Areas and Woodland / Forest Rehabilitation Areas (refer to Section 8.2 below).	Rehabilitation Management Plan (May 2013)	Not Applicable			





Deference	Clause	Requirement	Evidence	Andie Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	8.1.3	TCPL is trialling multi-spectral remote sensing monitoring techniques including the Normalised Difference Vegetation Index (NDVI) technique to detect any deterioration in vegetation health in rehabilitation areas. Should NDVI data indicate a deterioration in vegetation health, additional 'on the ground' monitoring will be undertaken to determine if any remedial actions are required in accordance with the rehabilitation TARP (refer to Section 9.2).	Rehabilitation Management Plan (May 2013)	Noted			
MOP (Nov 2013 to Oct 2015)	8.1.3	In addition to the scheduled monitoring events, Tarrawonga will be independently audited in June 2014 (and every 3 years thereafter) to assess compliance with the requirements of Project Approval PA 11_0047, EPL 12365, ML 1579 and any assessment, EMP, strategy or program required under those approvals.	Rehabilitation Management Plan (May 2013)	Compliant			
	8.1.4 Independent Bio	odiversity Audit					
MOP (Nov 2013 to Oct 2015)	8.1.4	In addition to the scheduled monitoring events, Tarrawonga will be independently audited in June 2014 (and every 3 years thereafter) to assess compliance with the requirements of Project Approval PA 11_0047, EPL 12365 and ML 1579, along with any assessment, management plan, strategy or program required under those approvals.	This Audit	Compliant			
MOP (Nov 2013 to Oct 2015)	8.1.4	All rehabilitation areas will be audited in the Independent Biodiversity Audit to verify rehabilitation progress documented in the MOP and Annual Review.	This audit	Compliant			
	8.2 Research and Reh	nabilitation Trials and Use of Analogue Sites					
	8.2.1 Use of Analogue						
MOP (Nov 2013 to Oct 2015)	8.2.1	Monitoring at the pastoral control sites will be used to develop baseline data for pasture species densities and diversity, total biomass per hectare and agricultural soil physical-chemical data at representative productive pastoral control sites. Baseline data will be used to quantify indicators and completion criteria for Agricultural Rehabilitation Areas to be documented in the Rehabilitation Management Plan.	Rehabilitation Management Plan (May 2013)	Compliant			
MOP (Nov 2013 to Oct 2015)	8.2.1	Similarly, surveys in the Woodland/Forest analogue site is undertaken to build baseline data for soil profiles, vegetation features and presence of fauna and habitat features.	Rehabilitation Management Plan (May 2013)	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kelerence	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	8.2.1	The project will disturb the existing analogue woodland site when mining advances into ML 1693. TCPL will engage in consultation with Forests NSW to secure access to alternative appropriate analogue sites in this MOP period.	Rehabilitation Management Plan (May 2013) states that: The project will disturb the existing analogue woodland site when mining advances into MLA 2. TCPL will secure access to an alternative appropriate analogue site prior to disturbance at the existing site. Negotiation not yet commenced.	Not Applicable			
	8.2.2 Research and Re	ehabilitation Trials					
	Continuous Improver						
MOP (Nov 2013 to Oct 2015)	8.2.2	Tarrawonga adopts a continuous improvement approach to rehabilitation; results from rehabilitation monitoring surveys and opportunistic monitoring observations are used to refine rehabilitation methodologies on an on-going basis. For example, the success rate for cool and warm climate seed mix variations are monitored to determine the most effective cover crop seed mixes.	Rehabilitation Management Plan (May 2013) Discussed in site interviews and some observations support this during the site inspection	Compliant			
	Direct Seeding Trials						
MOP (Nov 2013 to Oct 2015)	8.2.2	As outlined in Section 5.4.5.1 TCPL will trial direct seeding techniques in small areas to determine the effectiveness of direct seeding to establish native vegetation. TCPL will engage contractors to undertake direct seeding works. Rehabilitation progress will be monitored to identify potential benefits to supplementing tubestock planting with direct seeding. Trial results will be reported in the Annual Review and if successful, direct seeding methodologies will be formalised in the Rehabilitation Management Plan.	Rehabilitation Management Plan (May 2013)	Compliant			
	Native Vegetation Tu	l bestock Trials					
MOP (Nov 2013 to Oct 2015)	8.2.2	TCPL and Boggabri share growth rate data for tubestock sourced from different areas to verify if stock from different sources produces better growth.	Rehabilitation Management Plan (May 2013)	Compliant			
	8.3 Reporting						
MOP (Nov 2013 to Oct 2015)	8.3	Results of rehabilitation monitoring will be reported in the Annual Review. The Annual Review will record monitoring results and discuss rehabilitation performance against key performance measures/indicators; compliance with regulatory requirements and TCPL commitments. The Annual Review will also discuss identified trends and instances where potential rehabilitation failure has been identified triggering intervention in accordance with a Rehabilitation TARP (Section 9.2).	Rehabilitation Management Plan (May 2013) AEMR 2012-2013 includes rehab monitoring results but does not assess performance against key performance measures/indicators; compliance with regulatory requirements and TCPL commitments. AEMR 2012-2013 does not discuss identified trends and instances where potential rehabilitation failure has been identified triggering intervention in accordance with a Rehabilitation TARP.	Not Compliant Administrative			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)	8.3	Annual Reviews and independent audits will be submitted to relevant government agencies and made publically available via the Whitehaven/Tarrawonga website: http://whitehavencoal.com.au/operations/tarrawonga_mine.cfm	The relevant docs are on the website	Compliant			
	9 Intervention and Ad	,					
	9.2 Trigger Action Res						
MOP (Nov 2013 to Oct 2015)	9.2	A high-level Trigger Action Response Plan (TARP) has been developed in order to provide an initial framework to manage potential key risks to rehabilitation. It has been prepared in line with the TARP in Tarrawonga's Biodiversity Management Plan and includes: • Identification of the principal contributing factors and impacts for each major risk to rehabilitation; • Identification of upper limits (trigger values) for causes and impacts that are considered to represent an unacceptable level of risk; and • Identification of appropriate responses to mitigate or remediate the causes and impacts, including a notification protocol.		Compliant			
	•	mentation of the MOP					
	10.1 Periodic Review						
MOP (Nov 2013 to Oct 2015)	10.1	Periodic reviews will be conducted to assess the effectiveness of this plan. This plan will be reviewed, and if necessary revised, annually or within three months of the submission of an: • Updated or additional management plan as required by conditions 12, 21, 29, 39, 47, and 64 of Schedule 3 of Project Approval PA 11_0047; • Annual Review undertaken in accordance with condition 4 of Schedule 5 of Project Approval PA 11_0047; • Incident report undertaken as per condition 8 of Schedule 5 of Project Approval PA 11_0047; • Audit undertaken to satisfy Schedule 10 of Project Approval PA 11_0047; and • Approval to modify Project Approval PA 11_0047.	Rehabilitation Management Plan (May 2013) Recommendation - The review process needs to be documented.	Compliant			
MOP (Nov 2013 to Oct 2015)	10.1 10.2 Implementation	This plan may also be revised due to: • Deficiencies being identified; • Recommendations arising from the Rehabilitation Monitoring Programs; • Changes to environmental requirements due to (for example) changed legislation or regulatory requirements; and • Changes in the activities or mining operations described in this MOP.	Rehabilitation Management Plan (May 2013) Not required in the audit period	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
MOP (Nov 2013 to Oct 2015)		Tarrawonga personnel responsible for the monitoring, review and implementation of this plan are defined in Table 27.	Rehabilitation Management Plan (May 2013)	Noted			





Reference	attached to th	Denvisement	Evidence	Audit Finding		Risk		Doon on a ibility
Keterence			Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
	Tarrawonga Coal Mine Conditions attached to	Extension (EPBC 2011/5923)						
	Conditions attached to							
EPBC 2011/2013		The person taking the actions must not clear more than 13 hectares (ha) of the EPBC listed Wite Box-Yellw Box-Blakely's Red Gum Grassy Woodland and Derived Native Grassland critically endangered ecological community within the Tarrawonga Coal Extension project area, as identified in Attachment A of these conditions.	There are only 13ha within the project area, to clear more TCL would have to clear beyond the project boundary, there is no evidence of this occurring.	Compliant				
EPBC 2011/2013	2	The person taking the action must not clear more than: a. 279 ha of habitat for the regent honeyeater (Anthochaera phrygia: formerly Xanthomyza phrygia); b. 54 ha of habitat for the swift parrot (Lathamus discolor); and c. 334 ha of habitat for the greater long-eared bat (Nyctophilus corbeni), within the Tarrawonga Coal Extension project area.	The regent honeyeater habitat has only had minor areas clear to date, around 30ha.	Compliant				
EPBC 2011/2013	3	The person taking the action must submit to the Minister for approval within three months of commencement of construction, an approach that: a. limits the maximum disturbance (in hectares) specified for each of the years 5, 10, 15 and 17 from the date of this approval of the White Box-Yellow Box-Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community and the habitat or potential habitat for the regent honeyeater, swift parrot and greater long-eared bat; b. incorporates an analysis, undertaken by independent ecological experts approved by the Department, that deomonstrates the maximum disturbance limits which will minimise any impacts on relevant matters of national environmental significance; c. demonstrates collaboration with the person taking the action to develop and operate the Maules Creek Coal Project (EPBC 2009/5256) and the person taking the action to develop and operate the Maules Creek Coal Project (EPBC 2010/5566), in order to minimise progressive disturbance limits are to be reflected in the development of the Leard Forest Mining Precinct Biodiversity Strategy.		Not Applicable				
EPBC 2011/2013	4	The person taking the action must not clear more than the maximum project area disturbance limits specified for each of the years 5, 10, 15 and 17 as described in condition 3, unless otherwise approved by the Minister.	We are not yet at year 5.	Not Applicable				
EPBC 2011/2013	5	The person taking the action must publish the analysis under condition 3 on their website.	We are not yet at year 5.	Not Applicable				
	Direct Offsets							
EPBC 2011/2013	6	The person taking the action must register a leally binding conservation covenant over offset areas of no less than: a. 1055 ha of an equivalent or better quality of habitat for the regent honeyeater; b. 397 ha of an equivalent or better quality of habitat for swift parrot; c. 1355 ha of an equivalent or better quality of habitat for swift parrot; c. 1355 ha of an equivalent or better quality of habitat for the greater long-eared bat; and d. 232 ha of an equivalent or better quality of the White Box-Yellow Box-Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community. Note: Offset areas described in condition 6 do not necessarily need to be separate if the same areas can meet the listing criteria for the EPBC listed threatened species or community and meet the requirements of condition 6	Noted					





						Risk		_
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2013		The person taking the action must verify through independent review the quantity and condition class of White Box-Yellow Box-Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community and the quantity and quality of habitat for the regent honey-eater, swift parrot and greater long-eared bat within all proposed offset areas including those proposed in the Environmental Assessment and any additional offsets as required at condition 8. Details of all independently verified offset areas must be submitted to the Minister for approval by 31 January 2014. The findings of the independent review must be published on the proponent's website.	Condition class report submitted	Compliant				
EPBC 2011/2013	8	If the independent review finds that the offset areas do not meet the requirements of conditions 6, 7 and 9then additional areas must be included in the offset areas until all relevant criteria under these conditions are met.	The report found the areas were suitable in size and vege class.	Compliant				
EPBC 2011/2013	9	The offset areas must be of an overall equivalent or better quality than the areas being cleared. This means: a, for White Box – Yellow Box – Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community, offset areas must meet the definition of the ecological community described in the listing advice, and must be of an overall equivalent or better condition class than the areas being cleared, based on the proportion of each condition class represented and other relevant ecological attributes;" b, for the threatened species, the quality of the habitat for the species, taking account of its ecological requirements, must be equivalent to or better than the areas being cleared.	Noted, see classification assessment	Noted				
	Direct Offsets							
EPBC 2011/2013	10	The mechanism/s for registering a legally binding covenant must provide protection for the offset areas in perpetuity and be registered within 5 years of the date of this approval.	The approval is not yet 5 years old, there is no legally binding covenant in place	Not Applicable				
EPBC 2011/2013	11	If the person taking the action proposes to undertake any action within areas secured under conditions 6, other than thos management activites related to managing the offset areas or as set out in the conditions of approval, then approval to undertake that action must be obtained in writing from the Minister. In seeking the Minister's approval, the person undertaking the action must provide a detailed assessment of the area where the action is proposed to take place and an assessment of all associated adverse impacts on matters of national environmental significance. If the Minister agrees to the action within the offset areas, the area identified for the action must be excised from the offset area and alternative offsets secured by the person taking the action at a ratio of at least 20:1 in relation to the impact on matters of national environmental significance.	Noted, no such actions considered to date	Not Applicable				
		Offset management plan						<u> </u>
EPBC 2011/2013	12	12. The person taking the action must submit to the Minister for approval an <i>Offset management plan</i> for all of the offset areas , specified in condition 6, within 12 months of the date of this approval. The approved <i>Offset management plan</i> must be implemented.	Willeroi Offset Management Plan letter with submission details sighted, includes response from DSEWPAC.	Compliant				
EPBC 2011/2013	12	Note: for consistency, the proponent may develop a Biodiversity Management plan that includes the requirements set for managing offsets and set out in these conditions, to align with the requirements of the NSW state government Project Approval dated 22 January 2013 (application number 11_0047) and this approval.	Noted					
EPBC 2011/2014	13	The Offset management plan must include:						





Reference	attached to th	Danninamant	Evidence	Audit Finding		Risk		Responsibility
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2015	13a	 a) a text description and map which clearly defines the location and boundaries of the offset areas. This must be accompanied by the offset attributes and shapefiles; 	WBOMP 2.1, Figure 2.1, 2.2. Table 5.2 (Ecological attributes measured in condition plots). Appendix A, GIS Shape files	Compliant				
EPBC 2011/2016	13b	baseline ecological conditions in the offset areas. This must be consistent with the State and Transition Model and include but not be limited to: I. the extent and condition of all vegetation communities, including a description of the structure, floristics and tree age class representation of each community; ii. the extent and condition class of all areas of the White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community; iii. surveys targeting the regent honeyeater, swift parrot and greater longeared bat;	Methods for baseline vegetation condition in BOA (WBOMP, Section 2.6 cited FloraSearch 2011 and independent review ELA 2013b). Methods for surveys of terrestrial fauna in the BOA (WBOMP Section 2.7 cited Cenwest Environmental Services 2011). Results of baseline surveys reported in WBOMP: i. & ii. Extent and condition of each of the vegetation types within the BOA (WBOMP Table 2.2, Figure 2.2). State and Transition Model for BGW has been applied to the EEC located within the BOA to provide a condition class (WBOMP Figure 2.3). iii. Methods for baseline surveys targeting regent honeyeater, swift parrot and greater longeared bat in BOA (Cenwest Environmental Services 2011, Section 2.3.4 and 6.2). Methods for monitoring surveys targeting regent honeyeater, swift parrot and greater longeared bat in BOA (MBOMP, Section 5.3) iv. The extent and quality of all areas of habitat for the regent honeyeater, swift parrot and greater long-eared bat in BOA (WBOMP Table 2.4) v. Location of survey sites including coordinates in BOA (MBOMP, Flora survey sites: FloraSearch 2011, Attachment C Figure 2.7 shauns survey sites: Figure 2.5) vi. Photo reference points in BOA: FloraSearch 2011, Offset Strategy, Vegetation of the 'Willeroi' Offset Area, Figure 2.	Compliant				
EPBC 2011/2017	13c	c) clearly defined ecological management objectives for the offset areas;	WBOMP Section 3 outlines strategic ecological management objectives relevant to each zone.	Compliant				
EPBC 2011/2018	13d	 d) detailed description of all ecological management activities proposed to be undertaken, including maps and/or diagrams showing areas to be managed and the timing of the proposed activities; 	Maps of biodiversity management zones are shown in WBOMP Section 3. Details of management actions are set out in WBOMP Section 4. Table 7.1 outines the timing of management activities.	Compliant				
EPBC 2011/2019	13e	e) details of ongoing ecological monitoring programs, performance criteria, targets and provisions for adaptive management, including but not limited to: i. a set of measurable ecological indicators for detecting changes to the White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community, including those that may be ascribed to ongoing water stress;	13e. details of ongoing ecological monitoring programs, performance criteria, targets and provisions for adaptive management have been presented in the WBBMP Section 5 – Ecological Monitoring, and Section 6 – Contingency Planning, Condition monitoring plots will be placed in control (reference) sites (one in each of the eight vegetation communities in benchmark condition) and intervention sites (two in each vegetation type/management zone combination). i. Tables 5.2 and Table 5.3 set out ecological attributes and the scale of measurement in vegetation condition monitoring plots. The attributes & methods outlined in these tables are not consistent with BioBanking methodology for site attributes (OEH 2012). It is recommended that these tables be revised. For example, transects should be 50 m long (not 100 m), cover recorded along transects should be measured as percent foliage cover (not projected crown cover), groundcover should be recorded for grasses, sub-shrubs and other vascular plants <1m high, coarse woody debris should be recorded to make comparisons against ten Biometric benchmark attributes. Changes in site attributes over time provide an indication of trends in condition of threatened ecological communities threatened fauna species habitat. Impacts of climate change on grassy woodlands include death of trees from moisture stress, altered shrub-grass balance potentially in favour of shrubs, and changes in groundcover composition (Prober et al. 2010). It is recommended that 2 additional attributes (tree health, and density) should be monitored in Box-gum Woodland vegetation types on Willeroi (White Box grassy woodland of the Nandewar and Brigalow Belt South Bioregions & Yellow Box - Blakely's Red Gum grassy woodland of the Nandewar and Brigalow Belt South Bioregions & Yellow Box - Blakely's Red Gum grassy woodland of the Nandewar and Brigalow Belt South Bioregions & Yellow Box - Blakely's Red Gum grassy woodland of the Nandewar and Brigalow Belt South Bioregions & Yellow Box - Blakely's Red Gum grassy	Compliant				





Deferre	attack alta d	Demuir	Fulderess	Audia Finalia		Risk		Doon our vile ilit
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2020	13e	e) ii. a monitoring plan to assess the success of the management activities measured against the baseline condition. The monitoring must be statistically robust and able to quantify change in the condition of the White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community and habitat for the regent honeyeater, swift parrot and greater long-eared bat. This should include the use of control sites and periodic ecological surveys to be undertaken by a qualified ecologist. iii. a list of performance criteria based on the ecological management objectives for the White Box—Vellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community and habitat for the regent honeyeater, swift parrot and greater long-eared bat; iv. measures to exclude weeds from all offset areas for the period covered by this approval; v. a description of the potential risks to successful management against the performance criteria, and a description of the contingency measures that would be implemented to mitigate against these risks; vi. a process by which to report to the department the progress of management activities undertaken in the offset areas and the outcome of those activities, including identifying any need for improved management and activities to undertake such improvement.	e) ii. a monitoring plan to assess the success of the management activities measured against the baseline condition (Appendix C Willeroi BOA Works Program). The flora and fauna monitoring program is outlined in Tables 5.1, 5.2, 5.3 & 5.4 (see comments above regarding suggested changes to Tables 5.2 & 5.3). iii. Section 4 sets out management activities which will be implemented on Willeroi in order to counterbalance the loss in biodiversity value on the project site. Table 5.1 sets out performance criteria for biodiversity management activities in all management zones which are consistent with the objectives for White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community and habitat for the regent honeyeater, swift parrot and greater long-eared bat on Willeroi; iv. Section 4.8 details weed control measures in all offset areas for the period covered by this approval; v. Section 6 outlines potential risks to successful management against the performance criteria, and a description of the contingency measures that would be implemented to mitigate against these risks; Table 6.1 provides a risk assessment of potential incidents / fallures, with potential consequences, impacts and proposed controls. Table 6.2 outlines planning to deal with contingencies identified in the risk assessment and linking response action, responsibility and reporting requirements. vi. Table 9.1, Section 9 outlines reporting requirements for progress of management activities undertaken in the offset areas and the outcome of those activities, including identifying any need for improved management and activities to undertake such improvement. Sections 9.4 & 9.5 detail review and auditing requirements for the management of the Willeroi BOA.					
EPBC 2011/2021	13f	 f) details of all parties responsible for management, monitoring and implementing the management activities, including their position or status as a separate contractor. 	Roles & responsibilities are summarised in WBOMP, Table 1.3					
EPBC 2011/2022	13g	 g) details of the funding requirements for the ongoing management activities, including an estimate of the costs of the activities and details of the parties responsible for funding the activities 	TCPL will be responsible for funding the offset management activities on the Willeroi BOA. Funding requirements are detailed in Table 7. 1. Funding requirements for each management period are shown in Table 7. 2	Compliant				
EPBC 2011/2013	14	Unless otherwise agreed to in writing by the department, the baseline surveys for threatened species must be conducted in accordance with the department's Survey Guidelines for Australia's Threatened Brids and the Survey Guidelines for Australia's Threatened Brids. Subsequent monitoring must be carried out annually at the same time of year as the baseline surveys, unless otherise agreed to in writing by the department.	EcoLogical report does not verify compliance with these requirements. Recommendation - The consultant should confirm compliance with the two stated methodologies in their annual monitoring reports.	Compliant				
	Surface and groundwa	iter management plans		l.				-
EPBC 2011/2013	15	The person taking the action must provide by the 31 May 2013, to the Minister for approval, the surface and groundwater management plans as identified in condition 39 of the NSW state government Project Approval dated 22 January 2013 (application number 11_0047).	This was submitted on time, the comments from the agencies are being addressed at present.	Compliant				
EPBC 2011/2013	16	The surface and groundwater management plans must be consistent with the National Water Quality Management Strategy.	The existing Water Management Plan is not approved but does not reference EPBC requirements or the National Water Quality Management Strategy.	Not Compliant Administrative				
EPBC 2011/2013	17	The person taking the action must, within six months of this approval, in collaboration with the proponent to develop and operate the Maules Creek Mine (EPBC 2009/5566), provide written advice to the Minister demonstrating how the NSW government approved surface and groundwater management plans, address the cumulative impact of groundwater drawdown as a result of mining and how this may impact on the consequent health of the remnant native vegetation in the Leard State Forest, the Leard State Conservation Area and surrounding areas. In particular the advice must address the following matters:	The Leard Forest WMS (6-03-13) addresses this requirement. Note however, the strategy does not acknowledge the TCL EPBC requirements as the TCL EPBC approval is date 11-03-13.	Compliant				





5.6			z .,			Risk		
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2013	17	a. maximum amount of allowable drawdown in the alluvial aquifer b. drawdown in hard rock c. trigger levels pertaining to drawdown in the alluvial aquifer when corrective actions will be required to be undertaken d. identify the depth of root zone of the native vegetation e. monitoring to assess the ongoing quality and quantity of both surface and groundwater to identify impacts on the native vegetation	a) not stated in WMP, risk of not identifying GW impacts b) not stated in WMP, risk of not identifying GW impacts c) Stated in WMP d) Stated in Leard Forest Strategy e) This is adequately covered and the GDEs are addressed in the Leard Forest Strategy	Not Compliant	D	2	Medium	
EPBC 2011/2013	18	The person taking the action must within 6 months of the date of this approval, or such other timeframe specified by the Minister, provide to the Minister a report on: a. any updated modelling of surface and groundwater impacts that has been undertaken in preparing the surface and groundwater management plans b. how the surface and groundwater management plans addressed groundwater and surface water impacts on matters of national environemental significance	No evidence provided.	Not Compliant Administrative				
	Goonbri Creek Diversio	n and Low Permeability Barrier						
EPBC 2011/2013	19	The person taking the action must provide to the Minister for approval, before commencement of the construction of the permanent Goonbri Creek alignment, permanent flood bund and low permeability barrier, a Goonbri Creek Diversiion and Flood Bund Concept Design Plan. This approved Goonbri Creek Diversion and Flood Bund Concept Design Plan must be implemented.	This has not yet occurred, and is not planned for some time yet.	Not Applicable				
EPBC 2011/2013	20	The Goonbri Creek Diversion and Flood Bund Concept Design Plan must include: a. an assessment of the surface water and groundwater quality, ecology, hydrological and geomorphic baseline conditions within the creek; b. a description of how restoration of the re-aligned riparian zone will be undertaken to best replicate the habitat of the existing creek, including plant species and fauna habitat features; c. water quality, ecology, hydrological and geomorphic performance and completion criteria for the creek diversion and low permeability barrier based on the condition 20 (a); and d. a risk assessment of the environmental consequences of the proposed low permeability barrier and the proposed Goonbri Creek realignment including the potential for impacts on groundwater and surface discharge. The risk assessment must be peer-reviewed. e. details for ongoing monitoring and management of downstream impacts on the adjacent floodplains and Namoi River floodplain.	This has not yet occurred, and is not planned for some time yet.	Not Applicable				
EPBC 2011/2013	21	The person taking the action must ensure that dispersed waters downstream of the Goonbri Creek re-alignment do not adversely affect the downstream environment and avoid any impacts on matters of national environmental significance.	This has not yet occurred, and is not planned for some time yet.	Not Applicable				
	Leard Forest Mining Pr	ecinct Regional Biodiversity Strategy The person taking the action must implement the regional biodiversity	The Regional Biodivesity Plan has been developed but is not approved nor fully implemented.		,			
EPBC 2011/2013	22	strategy as required under condition 41 of the NSW state government project approval dated 22 January 2013 (application number 11_0047). The required scoping report for the development of the strategy must be submitted to the Minister for approval on or before 31 July 2013. The approved strategy must be implemented.	Evidence of the submission of the scoping strategy has not been provided.	Not Compliant Administrative				
	Mine site rehabilitation	1						





THE COLUMN TO THE COMPANY OF THE COLUMN TO THE COMPANY OF THE COLUMN TO	Doforance	attached to th	Domuiroment	Evidence	Audit Finding		Risk		Posnovsihilit.
Service control and better state of control and better state of the control and better state o	Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPIC 2011/2013 24 Inthis proposed inhabilitations strokings as provided in the Compliant Community institute of the Compl	EPBC 2011/2013	23	Grassy Woodland and Derived Native Grassland and the habitat of the regent honeyeater, swift parrot and greater long-eared bat, the person taking the action must, within 12 months of the commencement of construction, submit to the Minister for approval a mine site rehabilitation plan for the progressive rehabilitation and revegetation of no less than 752 ha of native forest and woodland in the project area including 13 ha using species consistent with a White Box – Yellow Box – Blakely's Red Gum Grassy Woodland and Derived Native Grassland Ecological Community. This approved mine site rehabilitation plan must		Compliant				
following information. 1. target and apprimanue indicators to achieve effective restorate of alternatival builds for the region haveyout, will part an adjournate of alternatival builds for the region haveyout, will part an adjournate of alternatival builds for the region haveyout, will part an adjournate of the restoration of alternatival builds for the region of the restoration of the res	EPBC 2011/2013	24	with the proposed rehabilitation streategy as provided in the Environmental Assessment and, as required under the NSW state	The RMP reflects the commitments in the EA	Compliant				
management actions undertaken and the outcome of those actions, and the mechanisms to be used to identify the need for improved management; f. a description of the potential risks to successful management and rehabilitation on the project site, including weed invasion, and a description of the contingency measures that would be implemented to mitigate these risks; g. details of long-tem management and protection of the mine site, including details of the commitment of funds to achieve this The mine site rehabilitation plan must be subject to an independent review by a qualified ecologist prior to being submitted to the Minister for approval. The findings of the independent review must be published EPBC 2011/2013 26 The mine site rehabilitation plan must be subject to an independent review must be published The Rehab Mngt Plan has not been approved and independent review has therefore not been conducted. Not Applicable	EPBC 2011/2013	25	following information: a. targets and performance indicators to achieve effective restoraton of potential habitat for the regent honeyeater, swift parrot and greater long-eared bat and White Box—Yellow Box—Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community, including weed management; b. details of the vegetation communities to be rehabilitatied and the timing of progressive rehabilitation (commencing as soon as practivable following disturbance); c. detailed soil depth surveys and analysis to inform the effective placement and restoration of soild unerlying the proposed rehabilitation sites; including mapping of soils across the disturbance sites and soil sampling at no less than one sample point per 20 ha of each soil type identified. Sampling must identified. Sampling must identified, Sampling must identified, stepth, water holding capacity, structure and physio-chemical properties of each of the soil and subsoil layers; d. processes and methodologies for the removal, storage and re-layering of the top soil and sub soil layer underlying the disturbed sites being prepared for rehabilitation. These processes and methodologies must ensure the replacement of top and sub soil layers as provided in the Environment Assessment.	b) in the RMP, BMP c) This has not been done, some soil assessment was conducted in the EA. D) There are procedures and management controls to ensure this happens	Not Compliant	D	2	Medium	
review by a qualified ecologist prior to being submitted to the Minister EPBC 2011/2013 26 for approval. The findings of the independent review must be published Not Applicable	EPBC 2011/2013	25	management actions undertaken and the outcome of those actions, and the mechanisms to be used to identify the need for improved management; f. a description of the potential risks to successful management and rehabilitation on the project site, including weed invasion, and a description of the contingency measures that would be implemented to mitigate these risks; g. details of long-tem management and protection of the mine site,	f) Includes a risk assessment and analysis g) Includes closure management, final void management and information on financial support					
Conservation Bond Conservation	EPBC 2011/2013		review by a qualified ecologist prior to being submitted to the Minister for approval. The findings of the independent review must be published		Not Applicable				





Doforonco	attached to th	Paguirament	Evidence	Audit Finding		Risk		Posnonsihilit.
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2013	27	The person taking the action is required to submit a Conservation and Biodiversity Bond under condition 49 of the New South Wales state government project approval dated 22 January 2013 (application 11_0047). It is noted that this bond may be combined with the rehabilitation security deposit as required under the New South Wales Mining Act 1992. The person taking the action must submit details of this bond and the rehabilitation security deposit, to the Minister, within six months of this approval. If the Minister is not satisfied that the bond and rehabilitation security deposit lodged by the person taking the action is adequate to provide for the requirements referred to under conditions 19, 20, 22, 23 and 24, the Minister may require the person taking the action establish an additional bond or equivalent financial instrument in trust, under conditions approved in writing by the Minister.		Not Compliant Administrative				
		Final Landform						
EPBC 2011/2013		Note: for consistency, the person taking the action may develop a single mine rehabilitation plan to align with the requirements, including timing of reporting, of the NSW State Government approval dated 22 january 2013 (Application 11_0047_ and this approval. The Offset Management Plan and the Rehabilitation management plan need to be substantially integrated for achieving biodiversity objectives for the rehabilitated minesite.	Noted					
EPBC 2011/2013	28	The person taking the action must undertake rehabilitation to ensure the final landform provides the optimum opportunity for the successful restoration of native forest and woodland including the critically endagered White Box – Yellow Box – Blakely's Red Gum Grassy Woodland and Derived Native Grassland ecological community.	Noted, not yet able to be verified but mineplanning reflects this rquirement	Not Applicable				
EPBC 2011/2013	28	Note: for consistency, the proponent may develop a single mine rehabilitation plan to align with the requirements of the NSW Governmeent and this approval. The Offset Management Plan and the Rahbilitation managemente Plan need to be substatially integrated for achieving biodiversity objectives for the rehabilitated mine-site.	Noted					
EPBC 2011/2013	29	The person taking the action must undertake rehabilitation to ensure the final void and landform minimises the extent od any resulting pit lake, avoids salt scalding and ensures that drained waters do not adversely affect the downstream environment and avoids any impacts on matters of national environmental significance.	Noted. Final void management plan not triggered yet					
EPBC 2011/2013	29 Survey data	Note: the State approval conditions for project 11_0047 require the preparation and implementation of a Final Void and Mine Closure Plan that considers interactions with the adjoining mines, including interaction between final voids, opportunities for integrated mine planning with adjoining mines to minimise environmental impacts, all reasonable and feasible landform options for the final void (including filling) and predicted hydrochemistry and hydrogeology (including long-term groundwater recovery and void groundwater quality).	Noted. Final void management plan not triggered yet					





D. C		<u> </u>	- 11	a 10 m 10 m	Risk			5
Reference	attached to th	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2013	30	All survey data collected for the project must be recorded so as to conform to data standards notified from time to time by the department . When requested by the department , the proponent must provide to the department all species and ecological survey data and related survey information from ecological surveys undertaken for matters of national environmental significance. This survey data must be provided within 30 business days of request, or in a timeframe agreed to by the department in writing, the department may use the survey data for other purposes.	No evidence of compliance with the required survey standards	Not Compliant Administrative				
EPBC 2011/2013	30	In the event that any additional matters of national environmental significance are recorded within the project area and a significant impact on the matter/s is likely, the department must be notified in writing within 14 days of the matter/s being recorded. In accordance with condition 36, the Minister may request that the person taking the action revise any relevant plans to ensure better protection of the relevant matter/s.	Noted, no such incidents to date	Not Applicable				
	Reporting and auditing		h ii - 1 - 1 - 2 - 2 - 4 - 1 - 1 - 1 - 2 - 4 - 1 - 2 - 4 - 1 - 2 - 4 - 2 - 2	1	<u> </u>		1	1
EPBC 2011/2013	31	Within 14 days after the commencement of construction, the person taking the action must advise the department in writing of the actual date of commencement of construction.	Letter dated 26-03-14, construction commenced 12-3-14	Compliant				
EPBC 2011/2013	32	By the end of March of each year after the commencement of the action, the person taking the action must publish a report on their website addressing compliance with the conditions of this approval over the previous 12 months, including implementation of any management plans as specidied int he conditions. Non-compliance with any of the conditions of this approval must be reported to the department at the sane time as the compliance report is published.	Not yet due	Not Applicable				
EPBC 2011/2013	33	Upon the direction of the Minister, the person taking the action must ensure that an independent audit of compliance with the conditions of approval is conducted and a report submitted to the Minister. The independent auditor must be approved by the Minister prior to the commencement of the audit. Audit criteria must be agreed to by the Minister and the audit report must address the criteria to the satisfaction of the Minister.	No such direction to date	Not Applicable				
EPBC 2011/2013	34	If the person taking the action wishes to carry out any activity otherwise than in accordance with the plans, as specified in the conditions, the person taking the action must submit to the department for the Minister's written approval a revised version of that plan. The varied activity shall not commence until the Minister has approved the revised plan in writing. The Minister will not approve a revised plan, unless the revised plan would result in an equivalent or improved environmental outcome. If the Minister approves the revised plan that plan must be implemented in place of the plan originally approved.	No such request submitted in the audit period	Not Applicable				
EPBC 2011/2013	35	If the Minister believes that it is necessary or convenient for the better protection of listed threatened species and communities or listed mogratory species to do so, the Minister may request that the person taking the action make specified revisions to the management plan specified in the conditions and submit the revised plan for the Minister's written approval. The person taking the action must comply with any such request. The revised approved plan must be implemented. Unless the Minister has approved the revised plan then the person taking the action must continure to implement the originally approved plan, as specidied in the conditions.	This has not occurred in the audit period	Not Applicable				





Reference	attached to th	Requirement	Evidence	Audit Finding		Risk		Responsibility
Reference	attached to th	Kequirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk	Responsibility
EPBC 2011/2013	36	If, at any time after 5 years from the date of this approval, the person taking the action has not substantially commenced the action, then the person taking the action must not substantially commence the action without the written agreement of the Minister.	Not 5 years since the approval.	Not Applicable				
	Publication of plans							
EPBC 2011/2013	37	The person taking the action must maintain accurate records substantiating all activities and outcomes associated with or relevant to the above conditions of approval, including measures taken to implement the management plans required by this approval, and make them available upon request to the department. Such records maybe subject to audit by the department or an independent auditor in accordance with section 458 of the Environment Protection and Biodiversity Conservation Act 1999, or used to verify compliance with the conditions of approval. Summaries of audits will be posted on the department's website. The results of audits may also be publicised through the general media.	Noted no such request or audit occurred in the audit period.	Not Applicable				
EPBC 2011/2013	38	Unless otherwise agreed to in writing by the Minister, the person taking the action must publish all management plans referred to in these conditions of approval on their website. Each management plan must be published on the website within 1 month of being approved.	The Willeroi BOMP is not approved	Not Applicable				





2.6	C)	2		a 111 mm 11		Risk		
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
		ronmental Management Strategy						
	4.2 Training and Com							1
EMS	4.2	All training, competency assessment and maintenance of training records in relation to this EMS must be conducted in accordance with the requirements of the Tarrawonga Coal Mine's Training and Competency Management Plan.	Noted, see induction	Compliant				
EMS	4.2	The Training and Competency Management Plan defines the induction and training requirements for each classification of employee based on the type of work and the work environments that each classification is exposed to.	Noted, see induction	Compliant				
	4.3 Tarrawonga Coal	Mine Internal Communication						
EMS	4.3	Communication of information related to this EMS must be conducted and recorded in accordance with the requirements of the Whitehaven's Information and Communication Arrangements Standard.	Noted, see induction	Compliant				
	4.4 EMS Monitoring a	and Corrective Action						
	4.4.1 Audits							
EMS	4.4.1	The audit process and the completion and retention of audit records will be conducted in accordance with the requirements of the Whitehaven Audit and Review Standard.	Noted, see induction	Compliant				
EMS	4.4.1	The Environmental Officer will ensure that an internal audit of this EMS is conducted every 2 years and that an external audit is carried out at intervals outlined in PA 11_0047.	The EMS has not been audited lately but a new EMS has been developed in accordance with the latest approval for the site	Compliant				
	4.4.2 Reviews							
EMS	4.4.2	The review process and the retention of review records will be conducted in accordance with the requirements of the Audit and Review Standard. The Environmental Manager will ensure that a review of this EMS is conducted every 2 years, and following any event based trigger identified in the Audit and Review Standard that is relevant to the implementation of this EMS.	See above, there is no process for recording reviews so ther eis no evidenc eof a review being conducted, the document has been regularly revised	Compliant				
	4.4.3 Plan Monitoring	3						
EMS	4.4.3	In addition to the formal means of monitoring the performance of this EMS by scheduled audit and review, every person who performs a function defined by this EMS also has a responsibility to monitor the performance of this EMS and to bring to the attention of the Environmental Manager any non conformance, deviation or potential improvement of which they may become aware.		Compliant				
	4.4.4 Corrective Action	on						
EMS	4.4.4	Where monitoring of this EMS identifies the need for corrective action to be taken, the Environmental Manager will prepare a corrective action plan. Preparation of the plan may require the assistance of internal and external specialists. Details on the preparation of the corrective action plan will be included in the relevant AEMR/Annual Review and EPL Annual Return where relevant.	Noted, no requirement for this in the audit period	Compliant				
	4.5 Document Contro		l .	l	·			





Poforonce	Clause	Paguiroment	Evidonco	Audit Einding		Risk		Posnonsihilit
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EMS	4.5	This EMS will be maintained in accordance with the mine's document control system: • The master copy will be maintained electronically on the mine server; • Where alterations to the Plan are required, such as following formal review, the Document Controller will update the master copy and notify the Document Owner of the alteration; and • The Document Controller will be the only person with access to the master copy for the purpose of making alterations.	Noted					
EMS	4.5	Copies of the Plan printed from the mine server must be considered to be uncontrolled copies.	Noted					
	4.6 Retention of Rec	ords						
EMS	4.6	The assigned persons will ensure this EMS and associated records are retained on file at the mine for a period of at least 5 years.	Noted, EMS is over 5 years old and a copy is on the website					
		emination, Complaints and Incident Management and Dispute						
	4.7.1 Information Dis		T		1		1	
EMS	4.7.1	TCPL is committed to a policy of community membership and a sense of mine ownership by employees and local community members, and will undertake a program of regular liaison/contact with local residents, landowners and the broader community to inform them of the mine's progress. Such a program will also provide an opportunity to discuss issues of "concern" which residents are reticent to register as complaints.	This occurs see consultation comments elsewhere in this audit	Compliant				
EMS	4.7.1	All such liaison/contacts/comments will be documented.	This occurs see consultation comments elsewhere in this audit	Compliant				
EMS	4.7.1	Dissemination of information to the local community and relevant agencies regarding the mining operation, its progress and environmental management performance, will be achieved by both formal and informal means including the following.	Noted					
EMS	4.7.1	The CCC will comprise an independent chair and appropriate representation from Tarrawonga Coal, Narrabri Shire Council, Gunnedah Shire Council and the local community.	This occurs see consultation comments elsewhere in this audit	Compliant				
EMS	4.7.1	The committee will act as local focal points for the provision of information to, and receipt of comments from, community members. CCC meetings will be held at least four times per year. TCPL representatives will provide advice on the status of construction activities, the mine's progress, environmental performance and monitoring results and complaints. The CCC meetings will also act as a forum for discussion of each of the above aspects or any other issue brought up by members of the community through the CCC representatives, or directly with the Tarrawonga Coal Mine.	CCC Minutes on web indicate site consistently hits this target	Compliant				
EMS	4.7.1	The minutes of the CCC meetings will be available on the Whitehaven website (www.whitehavencoal.com.au). In addition to their tabling at CCC meetings, relevant environmental monitoring results pertaining to individual landholders will be provided on request, and all results will be available for public examination on the Whitehaven website.	This is the case	Compliant				





Reference	Clause	Dogwisement	Evidence	Audit Finding		Risk		Dosnonsihili
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibili
EMS	4.7.1	Tarrawonga Coal Mine will provide the opportunity for the CCC and local residents, landholders, schools and community groups to visit the mine, as well as maintaining an open door policy for interested local residents, where practically possible. The extent of visitation to the site will ultimately be governed by ensuring site standards for safety are maintained at all times.		Compliant				
EMS	4.7.1	Copies of all management plans/strategies or monitoring programs together with the results of independent audits undertaken in accordance with PA 11_0047 will also be provided to the CCC and the Councils, and made publicly available on the Whitehaven website.	These are on the website and provided to the CCC. No evidence that they are provided to Councils	Not Compliant Administrative				
EMS	4.7.1	Each year, TCPL will prepare an AEMR/Annual Review which will address both the ML and Project Approval conditions as follows:	This occurs	Compliant				
EMS	4.7.1	Contain a review and forecast of performance for the preceding and ensuing twelve months in terms of: o the accepted Mining Operations Plan; o development consent requirements and conditions; o Department of Environment and Conservation (now EPA) and Department of Planning (now DOPI) licences and approvals; o any other statutory environmental requirements; o details of any variations to environmental approvals applicable to the lease area; and o where relevant, progress towards final rehabilitation objectives.	The AEMR contains these items	Compliant				
EMS	4.7.1	Describe the development (including any rehabilitation) that was carried out in the past year, and the development that is proposed to be carried out over the next year;	The AEMR adresses this	Compliant				
EMS	4.7.1	 Include a comprehensive review of the monitoring results and complaints records of the development over the past year, which includes a comparison of these results against: o the relevant statutory requirements, limits or performances measures/criteria; o the monitoring results of previous years; and o the relevant predictions in any EIS or EA of the development; 	The AEMR adresses this	Compliant				
EMS	4.7.1	 Identify any non-compliance over the last year, and describe what actions were (or are being) taken to ensure compliance; 	The AEMR adresses this	Compliant				
EMS	4.7.1	Identify any trends in the monitoring data over the life of the development;	Life of mine data is presented but comentary is negligable. Recommendation - more detailed analysis of the life of mine environmental monitoring data trends would help the public and other stakeholders understand the sites impacts.	Compliant				
EMS	4.7.1	 Identify any discrepancies between the predicted and actual impacts of the development, and analyse the potential cause of any significant discrepancies; and 	Comparisons with EA in the AEMR	Compliant				
EMS	4.7.1	 Describe what measures will be implemented over the next year to improve the environmental performance of the development. 	This is included	Compliant				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk		Responsibility
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EMS	4.7.1	The AEMR/Annual Review will also address the matters identified in the DRE document entitled "Guidelines to the Mining, Rehabilitation and Environmental Management Process".	Noted, the AEMR is reviewed by DRE	Compliant				
EMS	4.7.1	The AEMR/Annual Review will be provided to both NSC and GSC, relevant agencies and to the CCC members and also made available to the public on Whitehaven's website.	No evidence of provision of these reports to Councils	Not Compliant Administrative				
EMS	4.7.1	Visits by relevant government agencies, e.g. EPA, DRE and DoPI, to inspect the mine site and verify Tarrawonga Coal's performance will be documented together with the provision of reports or information as requested.	This occurs, evidence sighted	Compliant				
EMS	4.7.1	TCPL will maintain regular formal and informal contact with relevant government agencies.	This occurs	Compliant				
EMS	4.7.1	TCPL will provide reports to relevant government agencies in the event of non-compliance or a potential non-compliance with respect to statutory criteria or guidelines.	Only one identified non-compliance in the audit period, evidence of communication with EPA provided along with hanged report and revised EPL	Compliant				
EMS	4.7.1	In addition to the above, Tarrawonga Coal may also utilise the local press to present feature articles on the mine's progress. Other forms of communication with the community will include the provision of newsletters to the landholders around the mine site, with copies made available at the Council offices and on the Whitehaven website. These newsletters will provide updates as to site developments and will be issued generally every 6 months or following specific events at the site for which community updates are warranted.	On website Newsletters not sent to to Council	Not Compliant Administrative				
	4.7.2 Complaints Ma	anagement						
EMS	4.7.2	A complaints management protocol has been developed to ensure an appropriate and consistent level of reporting, response and follow-up is adopted by Tarrawonga Coal. At a minimum, the following complaints management protocol will be	See complaints elsewhere	Compliant				
EMS	4.7.2	Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the complaint. o The action taken by TCPL in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCPL, the reason why no action was taken.	See complaints elsewhere	Compliant				
EMS	4.7.2	The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a	See complaints elsewhere	Compliant				
ENIS	2	complaint (except in the event of complaints recorded when the mine is not operational).						





Deference	Classes	Downing word	Estdonos	A coding file alice of		Risk		Daniel de la constitution
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EMS	4.7.2	 Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site. 	See complaints elsewhere	Compliant				
EMS	4.7.2	Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register.	See complaints elsewhere	Compliant				
EMS	4.7.2	 If necessary, the Environmental Manager will follow-up to confirm the source of the complaint is adequately mitigated. 	See complaints elsewhere	Compliant				
EMS	4.7.2	 A copy of the Complaints Register will be kept by TCPL and made available to the CCC and the complainant (on request). A summary of complaints received every 12 months will be provided to DoPI, GSC, NSC, EPA, DRE and the CCC through the AEMR. 		Compliant				
	4.7.3 Incident Manag	gement						
EMS	4.7.3	In addition to the protocol for complaints management, environmental incidents at the mine will be managed in accordance with the site's Emergency Management System and categories of environmental incidents will be determined using Whitehaven's Risk Matrix. Environmental incidents will be reported using Whitehaven's Incident Report Form and forwarded immediately to the Environmental Manager. Any corrective/preventative actions identified as a result of the incident will be implemented by the Environmental Manager in accordance with Section 4.4.4. The Tarrawonga site also operates in accordance with the requirements of its Pollution Incident Response Management Plan as required under the provisions of the Protection of the Environment Operations Act.	See incident management elsewhere	Compliant				
	4.7.4 Dispute Resolu	tion						
EMS	4.7.4	In the event that any complainant does not consider Tarrawonga Coal's response or reactions adequately address their concerns, the following procedure will be adopted: 1. A meeting will be convened with the Environmental Manager, Manager Community Relations and/or Operations Manager to seek resolution of the matter. The complainant will be provided with a written response from Tarrawonga Coal detailing the results of investigations to date and the agreed actions to be taken in respect of the measures to be implemented.	This has not occurred in the audit period	Not Applicable				
EMS	4.7.4	Upon implementation of the nominated measures, a further meeting will be convened to seek advice of satisfaction or otherwise as to the outcomes.	This has not occurred in the audit period	Not Applicable				
EMS	4.7.4	If, after 21 days following Steps 1 and 2, the complainant believes the matter remains unresolved and no further agreement can be reached as to additional measures to be undertaken, the matter will be referred to DoPI (and the relevant authority) for investigation and action as appropriate.	This has not occurred in the audit period	Not Applicable				
	4.8 Response to Non							
EMS	4.8	Compliance with all approvals, plans and procedures will be the responsibility of all personnel (staff and contractors) employed on or in association with the mine, and will be developed through promotion of mine ownership under the direction of the Operations Manager and Environmental Manager.	Noted					





Deference	Clause	Dogwiyamant	Evidones	Audit Finding		Risk		Dosnonsihility
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EMS	4.8	The Environmental Manager or Environmental Officer and/or Operations Manager will undertake regular inspections, internal audits and initiate directions identifying any remediation/rectification work required, and areas of actual or potential non-compliance, with all directions provided to the relevant party in writing and/or diarised.	EO inspection checklist sighted, completed copies provided	Compliant				
EMS	4.8	Any non-compliance with regulations, licences or approvals will be reported to the relevant authority, together with details of the corrective actions taken to avoid future occurrences. Areas of potential non-compliance which have the potential to cause environmental harm or result in complaints will also be reported to the relevant authority	This has occurred.	Compliant				
EMS	4.8	Non-compliances with the requirements of the mine's EPL will also be reported in each EPL Annual Return.	these are reported in the annual return	Compliant				
EMS	4.8	A review of the mine's compliance with all conditions of PA 11_0047, ML 1579, EPL 12365 and all other approvals and licences will be undertaken prior to (and included within) each AEMR/Annual Review submitted to the Director-General and DRE. The AEMR will also be provided to Council, relevant agencies, the Community Consultative Committee (CCC) and to the public on Whitehaven's website.	This is done, but not for all approvals and licenses, water licences and EPBC approval not covered.	Not Compliant Administrative				
EMS	4.8	Additionally, an independent environmental audit will be undertaken a minimum of once every three years (commencing in June 2014) and the report submitted to the DirectorGeneral, Narrabri and Gunnedah Shire Councils, all relevant authorities and made available to the public on Whitehaven's website. The independent audit will be undertaken by an appropriately certified auditor in accordance with AS/NZS ISO 19011:2003 "Guidelines for Quality and/or Environmental Management Systems Auditing" or equivalent updated versions of these guidelines. A copy of all publicly available documents will be placed on Whitehaven's website.	This audit	Compliant				
	4.9 Response to Em							L
EMS	4.9	As part of routine mine operations, TCPL will undertake risk assessments to identify the risk probability and consequences of the proposed activities and aspects of the operation, the adequacy of existing controls to contain the hazards and, where identified as deficient, propose additional controls to further manage or eliminate hazards.						
EMS	4.9	The existing Emergency Management System developed for the mine which, though primarily prepared in accordance with OH&S requirements, extends to environmental emergencies.	Noted					
EMS	4.9	Although there are specific procedures for individual situations, all incorporate three basic steps: 1. Notification of the emergency (internal and/or external); 2. Protection of personnel as a first priority; and 3. Protection of the environment, plant and equipment.	No environmental emergency notifications .	Not Applicable				
EMS	4.9	Any emergency situations or incidents, which do or could potentially have caused environmental harm, will be reported to the EPA and other relevant authorities in accordance with the requirements of the sites Pollution Incident Response Management Plan	Only one identified non-compliance in the audit period, evidence of communication with EPA provided along with changed report and revised EPL. No risk of harm to the environment	Compliant				





D (61		1	A Dr. Br. Dr		Risk		
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	Responsibility
EMS	4.9	All site personnel will be trained in situation recognition and emergency response procedures, with regular updates through tool box talks.	This is part of the Pegasus induction competed offsite. It was not possible to verfiy this information during the audit.	Not Applicable				
	5 Environmental Perf							
EMS	5	The success of TCPL's environmental performance will be assessed by way of its satisfaction of the conditions of PA 11_0047, Conditions of Authority appended to the Mining Lease, and conditions within the EPL, other approvals, licences or leases. This, in turn, will be measured by way of achievement of key performance outcomes and satisfaction of TCPL's objectives with respect to the management of: • Aboriginal cultural heritage; • Biodiversity; • Surface and groundwater; • Soil resources; • Bushfire; • Air quality; • Noise and blasting; • Visibility; • Waste; • Rehabilitation; and • Socio-economic issues.	Noted, assessed throughout this document	Not Applicable				
EMS	5	The assessment of performance, which may be quantitative and/or qualitative, will be reported in each relevant AEMR/Annual Review.		Compliant				
EMS	5	TCPL's objectives and key performance outcomes with respect to each of the above areas, with the key performance outcomes being used as a means by which Tarrawonga Coal and others can assess progress towards the achievement of the nominated objectives. Specific monitoring programs, either those developed in satisfaction of the development consent or other leases, licences or approvals (or others developed independently by Tarrawonga Coal), will be progressively appended to this document and updated as warranted.	Noted					
	5.1 Continuous Impro							
EMS	5.1	TCPL will investigate and implement ways to improve the environmental performance of the project over time. This will be achieved by keeping abreast of best environmental management practices in the industry and reporting on environmental performance annually in the AEMR/Annual Review. Stakeholder feedback will form an integral part of assessing environmental performance and assist in outlining measures for continuous improvement.	Noted					





Deference	Clause	Doggiyamant	Cuidones	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	Environmental Monit	oring Program					
	2 Consent Requireme	nts					
Environmental Monitoring Program	2	Consultation, with relevant government agencies as required by Condition 6(3), was undertaken as part of the preparation of the monitoring programs from which the EMonP provides a consolidated summary. These monitoring programs have been approved by the Director-General of the Department of Planning.	Noted				
	3 Environmental Mon	itoring Program			L		
Environmental Monitoring Program	3	Presented in tabular form, the following program consolidates monitoring commitments made in the following documents;	Noted				
Environmental Monitoring Program	3	Air Quality Management Program	Noted				
Environmental Monitoring Program	3	Archaeology and Cultural Heritage Management Plan	Noted				
Environmental Monitoring Program	3	Blast Monitoring Program	Noted				
Environmental Monitoring Program	3	Flora and Fauna Management Plan	Noted				
Environmental Monitoring Program	3	Noise Monitoring Program	Noted				
Environmental Monitoring Program	3	Road Noise Management Plan	Noted				
Environmental Monitoring Program	3	Site Water Management Plan	Noted				
Environmental Monitoring Program	3	Transport Route Management Plan	Noted				





Deference	Clause	Domuiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
	WHC_PLN_TAR_Noise						
Noise Management Plan	2 Noise Impact Assess		AEMR 2012-2013 included the following information: 3.10.3.2 June 2012 Noise Monitoring - Attended Noise monitoring was undertaken from the 25th to 27th June 2012. Monitoring was undertaken at "Bollol Creek Station" (N2), "Pine Grove", "Tarrawonga" (N3) and "Ambardo" (N4) properties. TCM did not exceed the operational noise criterion at either of the "Bollol Creek / Templemore", "Ambardo" or "Pine Grove" monitoring locations during the survey. However, TCM were in excess of the operational noise criterion at the Tarrawonga monitoring location during the evening and night of both June 26th and June 27th. THIS IS OUTSIDE THE AUDIT PERIOD	Compliant			
	2.2 Noise Acquisition	 Criteria					
Noise Management Plan	2.2	Upon receiving a written request for acquisition from an owner of the land listed in Table 1, the Applicant shall acquire the land in accordance with procedures in Conditions 8 – 9 of Schedule 4. • Properties 43, 44 and 45 also have acquisition rights under the approval for the Boggabri Coal mine, and/or the existing consent (DA 88-4-2005) for the Tarrawonga Mine. The Proponent may acquire these properties on an equitable basis with the owner of Boggabri mine • For the single purpose of acquisition under this condition, parcels of land that are in close proximity and operated as a single agricultural enterprise should be included as part of the land to be acquired. Where the Proponent and the owners(s) cannot agree on whether non contiguous parcels of land should be included, either party may refer the matter to the Director-General for resolution. The Director-General's decision as to the lands to be included for cquisition under the procedures in conditions 8 and 9 of Schedule 4 shall be final.	AEMR 2012-2013 included the following information: Of the properties identified in Table 1 from the Project Approval, properties 43, 44, 45 and 47 have now been acquired by either Tarrawonga or Boggabri Coal. Preliminary discussions have been held with property 49 in relation to their rights for acquisition.	Compliant			
Noise Management Plan	2.2	If the owner(s) of a privately owned residence, that is not listed in Table1, have reason to believe that operational noise from the project is causing the criteria in Table 2 to be exceeded at the residence, the owner(s) can request an independent noise impact assessment for the residence. The request shall be made in writing to the Director General. If the Director General considers that a noise impact assessment is warranted, then the Proponent shall commission the assessment.	Noted	Noted			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	LVIdence	Addit Fillding	Consequence	Likelihood	Risk
Noise Management Plan	3	Management measures to be implemented to manage noise impacts on surrounding residents will include: Noise monitoring on-site and within the community;	Attended noise Monitoring reports dated Jul 2013, Oct 2013, December 2013 and Mar 2014 by Spectrum Acoustics	Compliant			
Noise Management Plan	3	Prompt response to any community issues of concern or complaints;	Complaints register includes noise related complaints lodged on 9/04/2014, 7:48am; 22/04/2014, 6:10pm; 8/04/2014, 4:15pm. Environmental Superintendent responded via email advising that the site Environmental Officer would contact the complainant to arrange a meeting to discuss the issues. Register does not include any information on follow up action taken to resolve the issue.	Compliant			
Noise Management Plan	3	Refinement of on-site noise mitigation measures and mine operating procedures, where it is identified through monitoring that prevailing weather conditions are impacting on capacity to comply with noise criteria;	AEMR 2012-2013 reported the following complaint: Phone call to Environmental Manager 10/07/2012, 12:05pm Complaint received in relation to a blast held at Tarrawonga at 12:02pm. Blast shook the complainant's house. Written advice was issued to the complainant advising that the blast was complainant. An offer was made to place a blast monitor at the complainant's property when similar weather conditions are present.	Compliant			
Noise Management Plan	3	Discussions with relevant landowners to assess concerns;	No discussions to date - noise complaints generally from one complainant how historically has had discussions and is seeking acquisition.	Compliant			
Noise Management Plan	3	 Implementation of private noise agreements, where acceptable to the impacted landholder, with the terms of the agreement provided to the DPE and EPA; 	Noted	Not triggered			
Noise Management Plan	3	Implementation of feasible and reasonable acoustical mitigation at receivers (which may include window glazing, insulation and/or air conditioning), where an independent noise impact assessment conducted under the terms of the Project Approval identifies systemic exceedances of the noise criteria, and the landowner requests such measures; and	None done to date. No knowledge of any requests to site for mitigation apart from acquisition.	Not triggered			





Poforonco	Clause	Poguiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	3	Acquisition of the property where an independent noise impact assessment conducted under the terms of the Project Approval identifies systemic exceedances of the noise criteria, and the landowner requests acquisition. Acquisition would be undertaken in accordance with the terms of the Project Approval.	Not triggered, all acquisitions have been my agreement not driven by exceedences	Not Applicable			
Noise Management Plan	3	A continuous noise monitor will be installed at relevant reference locations to assist with noise management and enable real-time noise controls to be implemented. The continuous noise monitor is trailer mounted and can subsequently be relocated to appropriate positions depending upon prevailing conditions at the time. Preference will be given to locating the continuous noise monitor at or near locations most likely to be impacted by noise from the operation that are not subject to acquisition rights in the Project Approval.	AEMR 2012-2013 included the following information: In accordance with the requirements of PA 11_0047 and EPL 12365, TCPL undertook real time noise monitoring during the reporting period. Monitoring was confined to the "Northam" and "Kyalla" properties as well as project related properties "Templemore" and 'Blair Athol". During this period, Tarrawonga Coal has refined its procedures in relation to the real time noise monitoring.	Compliant			
	3.1 General and Speci	ific Noise Mitigation measures					
Noise Management Plan	3.1	A number of general noise management measures will continue to be undertaken, including: • Mining operations are permitted to be undertaken 24 hours 7 days per week. Actual operating hours may vary from time to time depending on production requirements. At all times, operational activities will give appropriate consideration to prevailing conditions to minimise potential for noise levels to exceed the relevant criteria. This will be managed by the use of predictive forecasts in mine planning and the use of real time noise monitoring to enable reactive management to noise levels which may include relocation of equipment in pit to avoid impacts as well as stand down of equipment where noise limits associated with mining activities cannot be met.	Predicitive forecasts not yet as good as they could be but are being implemented and used.	Compliant			
Noise Management Plan	3.1	Tarrawonga maintains a real time weather monitoring system which provides relevant meteorological conditions that may impact on noise propagation. The weather station monitors all parameters consistent with EPL requirements, including sigma theta. In addition to the Tarrawonga weather station, site can access temperature data from the Boggabri Coal communications tower (sensor at 50m), which provides additional relevant detail for site in terms of inversion impacts.	Noted	Noted			





Reference	Clause	Dominomont	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	3.1	Contractors, including all personnel and sub-contractors, will undergo environmental training on noise control and awareness via the generic induction process. Any contractor or subcontractor whose work is likely to create loud noise will be given more detailed guidance on the site's noise criteria and noise management requirements.	Inductions checklist for contractors indictes that this is done.	Compliant			
Noise Management Plan	3.1	The Sound Power Levels of mobile mining equipment will be tested annually in accordance with International Standards Organisation (ISO) 6395:1988 "Acoustics – Measurement of exterior noise emitted by earth-moving machinery – Dynamic test conditions". Equipment will be required to have noise levels that do not exceed the Sound Power Levels listed in Table 6-2 of the Noise Impact Assessment (see Appendix 2) from the Tarrawonga Extension Project EA. Any equipment found defective or not meeting the required sound power level will not be permitted to operate on site.	Spectrum reports, this work is done, there was no evidence of exceedance of the Noise Impact Assessment SPLs and so no way to determine whether any equipment had been prevented form operating.	Compliant			
Noise Management Plan	3.1	Any new trucks, dozers, drills or excavators purchased for the Tarrawonga operation will be acquired as sound suppressed equipment.	No new purchases	Not Applicable			
Noise Management Plan	3.1	Site equipment selection will require equipment to achieve sound power levels specified in Table 6-2 of the Noise Impact Assessment for the Tarrawonga Extension Project EA, and equipment will be maintained in good order.	No new purchases	Noted			
Noise Management Plan	3.1	Personnel and contractors will be required to pay due attention to adverse weather conditions and make modifications to the work program where necessary.	Noted, see adverse conditions procedure	Compliant			





Deference	Clause	Pomuiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	3.1	All complaints will be managed as outlined in Section 5.2.	Section 5.2 includes: • Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint, including any follow-up contact with the complaint, including any follow-up contact with the complaint to If no action was taken by TCPL, the reason why no action was taken by TCPL, the reason why no action was taken. Complaints register includes noise related complaints lodged on 9/04/2014, 7:48am; 22/04/2014, 6:10pm; 8/04/2014, 4:15pm. Environmental Superintendent responded via email advising that the site Environmental Officer would contact the complainant to arrange a meeting to discuss the issues. Register does not include any information on follow up action taken to resolve the issue but evidence of the follow up was provided in the form of an email to the complainant and a response in return agreeing to a meeting.				
Noise Management Plan	3.1	Monitoring of emitted noise levels will be undertaken during mining operations to verify compliance with noise criteria and to assess the need, if any, for additional noise attenuation measures.	Noted, real time unit tin place to support acoustics management	Noted			
Noise Management Plan	3.1	Where feasible and reasonable, operations have been modified to reduce noise emissions and include: Installation of an earth bund on the southern side of exposed sections of the services corridor (ROM coal haul road to the Boggabri Coal Mine). This will not occur over the next 2 year term as it is reliant on construction of infrastructure at Boggabri Coal;	Noted, the two year period is not yet complete	Not Applicable			
Noise Management Plan	3.1	A reduction in the number of mobile fleet items operating during the evening and night time periods where measured noise levels (real time monitoring) indicate likely exceedance of noise criteria. Based on the noise assessment for the Tarrawonga site and review of real time noise data, the key items of plant contributing to the potential for noise exceedance comprise dump trucks operating on the southern emplacement. The reduction in operating fleet would be based on actual noise measurement, preliminary review of site operations to determine if alternate operating locations are available, followed by stand down of equipment if necessary. Over the next 3 years, the key development area comprises the northern extension, which in effect is further removed from the Tarrawonga sensitive receivers located to the south. When operating on the southern emplacement during night time or adverse weather conditions, specific review of noise levels will be undertaken to ensure compliance is maintained; and	Confirmed in interviews	Compliant			





Reference	Clause	Doguiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	3.1	Modified alignment of haul routes to reduce their exposure relative to nearby receivers. The key component of this strategy is to reduce exposure from the dump trucks hauling to the northern emplacement via a haul road located at the southern end of the emplacement area. A northern haul road is now also available, so in circumstances where noise levels approach compliance criteria, an alternate haul road is available. This northern haul road will also become more relevant as the northern extension area is developed.	Confirmed in interviews	Compliant			
Noise Management Plan	3.1	The controls and management procedures will be reviewed in response to the results of noise monitoring, complaints or comments identified through TCPL's consultation effort. Any changes made will be noted as part of annual environmental reporting in the AEMR/Annual Review.	Noted, no changes noted in the AEMRs reviewed for this audit. No evidence of reviews, as mentioned elsewhere, reviews require better documentation. (Recommendation made elsewhere)	Not Applicable			
	3.2 Cumulative Noise	Mitigation Measures					
Noise Management Plan	3.2	TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in developing a Leard Forest Mining Precinct Noise Management Strategy in order to minimise cumulative impacts of mine generated noise on the surrounding community. The Strategy has been submitted to the Department of Planning and Environment (DPE) for approval. Once approved, the Strategy will be appended to this Plan.	Noted, the strategy is waiting approval	Not Applicable			
Noise Management Plan	3.2	Currently, TCM, Boggabri Coal and the Maules Creek Project participate in monthly meetings to discuss cumulative noise impact management measures.	Noted	Noted			
Noise Management Plan	3.2	The intention of the cumulative noise mitigation measures is to utilise an array of noise monitoring equipment utilised by each site to measure noise levels, with a process to apportion noise source for appropriate response. This plan will include agreed access to real time data, use of predictive forecast information to inform development of mine plans and a response protocol in the event of a triggered exceedance and/or community complaint.	Noted	Noted			
Noise Management Plan	3.2	As recommended in the BTM Complex Air Quality Management Strategy, a predictive forecast meteorology system is proposed based on the Weather Research & Forecasting (WRF) model and CALMET, specifically for the BTM complex, and a website will be developed to make data immediately available for sites, with half hourly forecasts up to 48 hours in advance. This system will download global meteorological data and forecasts on a daily basis and will indicate which receiver groups may be subjected to noise-enhancing meteorological conditions	A predicitve forecast system used but it could be from better data sources, the Strategy is not yet approved and implemented.	Not Applicable			





Deference	Clause	Di	Fuldana	Audia Findina		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	3.2	Once the meteorological system is configured and operating, the outcomes will be evaluated by a competent meteorologist or atmospheric science professional against actual meteorological measurements and the meteorological system will be validated and improved, where possible. It should be noted at the time of development of the Tarrawonga Noise Management Plan, the BTM complex strategy had not yet been approved, and consequently, the specific detail of the predictive forecast system may yet be subject to change. Under the BTM complex strategy, implementation of this system would be staged, with the predictive forecast component to be implemented post set-up of required noise monitoring equipment.	Noted. A predicitve forecast system used but it could be from better data sources, the Strategy is not yet approved and implemented.	Not Applicable			
	3.3 Noise Manageme						
Noise Management Plan	3.2	A risk/response matrix has been developed for the site to identify risk associated with weather conditions that require specific action to mitigate potential noise impacts. The intent behind the risk/response matrix is to understand the prevailing wind conditions and stability strength of temperature inversions to then implement management practices accordingly to avoid subsequent noise impacts. The risk/response matrix has been developed based on colour coding and key operational and weather conditions. The Environmental Officer will be responsible for implementation of the risk/response matrix in consultation with the Operations Manager. Refer to Appendix 3 for further information regarding the noise impact risk matrix.	Noted	Noted			
Noise Management Plan	3.2	Whilst it is understood that under certain meteorological conditions, the noise compliance criteria do not apply, it is also acknowledged under the requirements of condition 12(b) that noise impacts from the project are minimised during these adverse conditions. The implementation of the real time monitoring network ensure site responsiveness to adverse conditions, with first response always targeting reduced noise levels by modified site operations. As previously mentioned, this occurs regardless of the meteorological conditions at the time.	Noted	Noted			
	4 Monitoring Program	n				l	
	4.1 Monitoring Activit						





Deference	Clause	Descripement	Evidence	A dit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	4.1	Attended noise monitoring will be undertaken on a quarterly basis by an independent acoustic consultant for compliance purposes. All attended noise monitoring will be conducted in accordance with the NSW Industrial Noise Policy, AS 1055.1-1997 'Acoustics — Description and Measurement of environmental noise — General Procedures', and will expressly monitor the modification factors referred to in the NSW Industrial Noise Policy. Attended noise monitoring will be undertaken during day, evening and night time periods. It is noted that the inversion class to be applied to the project is Class G.	Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal Mine Boggabri, NSW stated: To determine compliance with the Leq (15 min) operational noise criteria the modification factors detailed in Section 4 of the NSW industrial Noise policy must be applied, as appropriate, to the measured noise levels.	Compliant			
Noise Management Plan	4.1	In order to actively manage noise emissions onsite and to adequately comply with the requirement for reactive noise control measures, TCPL will maintain, on site, a real time noise monitor equivalent to a Sentinex type, continuous noise management system. The real time system is not for compliance purposes but for operational management/proactive response to potential noise issues. The key features of a Sentinex type monitor are as follows: Real time access to monitoring information; Universal user interface platform (web browser); Customised alarm settings; Automatic daily reporting; Streaming audio to PC; and Continuous audio recording (.mp3).	Noted	Noted			
Noise Management Plan	4.1	This system will provide real time access to noise data, and provide the capacity to set the unit to a target noise goal. Upon noise emissions reaching the identified target level, an automated SMS message will be delivered to operational personnel on site which will trigger an investigation into the noise source, both through review of audio files, and on site knowledge of surface operations. Upon determination that the noise source is mine site related, active measures can be put in place to modify operations, or stand down the noise source to ensure compliance with noise criteria is maintained.	Noted, observed Barn-Owl interface	Noted			





Clause	Requirement			Risk			
		Evidence	Audit Finding	Consequence	Likelihood	Risk	
4.1	The real time continuous noise management unit will be set to trigger an SMS alert to site personnel upon recording of a continuous noise source at 33dBA (low frequency filtered) for a period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source is identified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below threshold. In the event that a second SMS alert triggers within a 2 hour timeframe, is confirmed as mine related, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 35dB(A). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump.	Noted, discussed with EO, reviewed real time outputs on computer screen	Noted				
4.1	Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis.	Noted Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal Mine Boggabri, NSW stated: Field calibration is carried out at the start and end of each monitoring period.	Noted				
4.1	be completed on an annual basis by the provision of a report from an appropriately qualified acoustical consultant, utilising the data obtained over a 12 month period from the monitoring program and providing a comparative assessment against the	percentile methodology used in the noise modelling for Tarrawonga. This work was ongoing at the time of the Annual Report being released. The results will issue	Compliant				
	4.1	period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source is identified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below threshold. In the event that a second SMS alert triggers within a 2 hour timeframe, is confirmed as mine related, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 35dB(A). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. 4.1 In accordance with the requirements of the consent, TCM will seek validation of the tenth percentile methodology used in the noise impact assessment for the Tarrawonga Coal Project. This will be completed on an annual basis by the provision of a report from an appropriately qualified acoustical consultant, utilising the data obtained over a 12 month period from the monitoring program and providing a comparative assessment against the modelled predictions from the tenth percentile methodology. The outcomes of this review will be reported in the AEMR/Annual	period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source is identified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below abcalled, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 356(M.). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the monitoring outcomes of the stream of the constance of the constance of the monitoring outcomes of the stream of the constance of the c	period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source is dentified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below threshold. In the event that a second SMS alert triggers within a 2 hour timeframe, is confirmed as mine related, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 35dB(A). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. 4.1 In accordance with the requirements of the consent, TCM will seek validation of the tenth percentile methodology used in the noise impact assessment for the Tarrawonga Coal Project. This will be completed on an annual basis by the provision of a report from an appropriately qualified acoustical consultant, utilising the data obtained over a 12 month period from the monitoring program. 4.1 In accordance with the requirements of the consent, TCM will be completed on an annual basis by the provision of a report from an appropriately qualified acoustical consultant, utilising the data obtained over a 12 month period from the monitoring program of the period month emonitoring program of the period month emonitoring program of the period month emonitoring program of the period from the monitoring program of the period f	period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source is identified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below threshold. In the event that a second SMS alert triggers within a 2 hour timeframe, is confirmed as mine related, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 35dB(A). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. 4.1 Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes on a quarterly basis. Attended monitoring will also be used to ensure calibration of the good of the standard of each monitoring program in a coordance with the requirements of the consent, TCM will seek validation of the tenth percentile methodology used in the noise impact assessment for the Tarrawonga Coal Project. This will be completed on an annual basis by the provision of a report from an appropriately qualified acoustical consultant, utilising the data obtained over a 12 month period from the monitoring program and providing a comparative assessment against the modeled predictions from the tenth percentile methodology. The outcomes of this review will be reported in the AEMR/Annual Review. Compliant	period of 15 minutes. Upon the issue of the alert, site personnel will access the web interface of the real time noise management unit to identify the noise source and determine if it is project related. In the event that the noise source and determine if it is project related in the event that the noise source is identified as a low frequency, mine related noise, site personnel will closely monitor noise levels to verify if noise levels remain below threshold. In the event that a second SMS alert triggers within a 2 hour timeframe, is confirmed as mine related, and is indicative of noise levels increasing or trending to the compliance limit, the Environmental Officer and Operations Manager will investigate operational measures to ensure noise levels remain below 3ddl(4). These operational measures may include, but will not be limited to, standing down of specific items of equipment, and or relocating equipment to alternate areas of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes of the site, ie. discontinuing operations on the southern dump. Attended monitoring will also be used to ensure calibration of the monitor and to provide specialist advice on monitoring outcomes of the analysis of the consent, TCM will see validation of the tenth percentile methodology used in the noise impact assessment for the Tarrawong Coal Project. This will use evaluation of the tenth percentile methodology used in the noise noise impact assessment against the working the data obtained over a 12 month period from the monitoring program and providing a comparative assessment against the modelled predictions from the tenth percentile methodology. The modelled predictions from the tenth percentile	





Reference	Clause	Poquirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	4.2	The location of the real time monitor will be established in consultation with EPA, on the basis that it can be relocated on an as needs basis, based on prevailing weather conditions, community complaints/concerns, operations at site so as TCM has capacity to effectively measure noise levels at the most likely affected receiver. As part of the Leard Forest Precinct Noise Management Strategy, further discussions around the noise monitoring network will be initiated with Boggabri and Maules Creek. This will be done on the basis of identifying any potential monitoring measures that can address cumulative impacts and provide for streamlined monitoring and reporting for each operation.	Occurred prior to the audit period. Has been moved during the audit period, no auditable trail for consultation with EPA, done by phone (Lachlan Johnson).	Compliant			
	4.3 Monitoring Frequ	lency					
Noise Management Plan	4.3	Noise monitoring frequencies are nominated in Table 2. If conditions on the day of monitoring are not suitable (ie. high winds, rain etc) then the monitoring event must be rescheduled to ensure ongoing compliance with monitoring frequency. Table 2 Noise Monitoring Summary with Roles and Responsibilities Type Frequency Responsibility Comments Operational Noise Monitoring Attended noise Countries Continuous Environmental Officer Real time noise monitoring requiring Continuous Environmental Officer Real time noise monitoring requiring Continuous Environmental Officer Real time noise monitoring requiring Continuous Continuous Environmental Officer Monitoring requiring Continuous Continuous Environmental Officer Monitoring requiring Continuous C	Noted	Noted			
	4.4 Monitoring Proce	edures					
Noise Management Plan	4.4	Attended noise surveys will be conducted as follows: • All noise investigations will be carried out in accordance with NSW EPA's Industrial Noise Policy, 2000 (INP), Environmental Noise Control Manual (ENCM) and applicable Australian Standards;	Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal Mine Boggabri only makes reference to the Industrial Noise policy Sprectrum are very experienced and this is				
Noise Management Plan	4.4	Noise levels will be measured in one-third octave bands using an instrument with IEC Type 1 characteristics as defined in Australian Standard AS IEC 61672.1 – 2004 "Electroacoustics – Sound Level Meters – Specifications". The instrument will have current calibration as per manufacturer's instructions and field calibration will be confirmed before and after measurements with a sound level calibrator;	Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal Mine Boggabri does not make reference to this standard Sprectrum are very experienced and this is likely to be an administrative oversight in the report	Not Compliant Administrative			
Noise Management Plan	4.4	The instrument will be set to A-weighting, "fast" response and measurements of LAeq(15 minute) will be taken at the location in Each measurement will be stored at a sampling rate of no greater than 5 seconds for further analysis;	Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal MineBoggabri stated: A-weighted noise levels were measured over the appropriate monitoring periods (90 minutes/day, 30 minutes/evening and 60 minutes/night) with data acquired at 1 or 2 second statistical intervals and the meter set to "fast" response.	Compliant			





Reference	Clause	Poquiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	4.4	Attended surveys will be conducted during the approved hours of operation with at least three measurements taken during each day over four consecutive days (as required by PA 11_0047) at each location in Figure 2, so that noise levels during the full range of operating times (day, evening and night) are monitored. Monitoring timeframes will be consistent with requirements of the EPL which requires 1.5 hours during the day, 30 minutes during the evening and 1 hour during the night period;	Spectrum Acoustics, ATTENDED NOISE MONITORING – MARCH 2014 Tarrawonga Coal MineBoggabri	Compliant			
Noise Management Plan	4.4	• Field notes will be taken during each measurement recording the time and duration of noise events, noise sources, instantaneous noise levels and the frequency range of identified site noise sources. Where an obvious noise exceedance is detected, the noise monitor must notify the Group Environment Manager of the exceedance and obtain relevant information as to the possible source of the exceedance (ie. malfunctioning equipment, additional activity contributing to noise levels) to ensure appropriate reporting and action on the exceedance;	Noted	Noted			
Noise Management Plan	4.4	Extraneous noise sources will be filtered from the measured signal using robust methods approved by EPA and DPE and the LAeq(15-minute) level attributable to TCPL activities will be identified and compared with the relevant criteria; and	This occurs, see Spectrum monitoring reports	Compliant			
Noise Management Plan	4.4	Details regarding plant configuration, survey interval, weather conditions, extraneous noise sources, monitoring locations and times of measurement will be recorded for inclusion in the noise monitoring report.	Plant configuration is not recorded	Not Compliant Administrative			
Noise Management Plan	4.4	Real-time noise monitoring will be conducted as follows: • A mobile real-time noise monitoring system utilising IEC Type 1 sound level meter with statistical and third-octave band capabilities will be located at properties most likely to be impacted by mine noise, as assessed in noise modelling for the Extension Project, or at properties where a complaint has been made in relation to mine noise. Site selection will be based on location of operational activity most likely to impact on that receiver;	Noted, the real time unit is deployed.	Noted			
Noise Management Plan	4.4	The unit will calculate, as a minimum, 15-minute statistics comprising LA90, LAeq and LA10 and low-frequency filtered LAeq (nominal threshold 800Hz but able to be changed);	Sighted at EOs desk computer	Compliant			
Noise Management Plan	4.4	Each 15-minute statistic will have an accompanying third-octave band spectrum;	Sighted at EOs desk computer	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
Noise Management Plan	4.4	Statistical data and digital audio files (minimum duration one minute per 15 minute interval) will be continuously uploaded to the internet using an appropriate communications format;	Sighted at EOs desk computer	Compliant			
Noise Management Plan	4.4	 A computer accessible to site personnel will display statistical results in real time via an internet browser and daily charts will be stored for later reference or printing; 	Sighted at EOs desk computer	Compliant			
Noise Management Plan	4.4	Noise levels nominally at site noise criteria will be set as trigger points in the real-time system to send pre-programmed SMS messages to relevant personnel;	Messages are sent, confirmed at interview	Compliant			
Noise Management Plan	4.4	On receipt of an SMS alert, audio files will be reviewed to determine the cause of the noise and, if necessary, the management safeguards and ameliorative actions in Section 3 and 4.1 will be initiated; and	Sighted procedure that drives this action	Complaint			
Noise Management Plan	4.4	If the real-time noise monitor is situated at an attended monitoring location, both the attended and real-time results will be included in the attended monitoring report.	Noted, the real time monitor was not at any of the attended sites in the reports that were reviewed for the audit.				
	5 Management of Exc	eedances, Complaints and Non-Compliance					
	5.1 Noise Compliance	Criteria Exceedance					
Noise Management Plan	5.1	If noise levels exceed the levels outlined in Section 2, advice will be sought from an appropriate acoustic consultant to verify the source of the elevated noise and identify options to address noise related impacts. Such actions will include: • Additional testing to confirm the elevated noise is sustained in nature. Elevated noise levels will be considered sustained in nature where two consecutive quarterly noise monitoring events identify an exceedance in noise criteria; • Consideration to changes to operational procedure or equipment type; and • The installation of sound attenuation measures to plant and equipment, where necessary.	Noted, to date, this has not been required (audit period)	Not Applicable			
Noise Management Plan	5.1	Where it is identified that the above options cannot achieve compliance with noise criteria identified through attended noise monitoring, TCPL will undertake negotiations with the affected landowners with a view to entering into private agreements. Such negotiations would include options with regard to provision of insulation, double-glazing of windows, airconditioning, or other measures designed to reduce noise impact at the affected property in accordance with conditional requirements.	Noted, this has not been required in the audit period.	Not Applicable			
	5.1.2 Independent Re	view and Land Acquisition					_
Noise Management Plan	5.1.2	If an owner of privately-owned land considers the mine to be exceeding noise compliance criteria, they may ask the Director-General in writing for an independent review of the impacts on their land. Conditions 4, 5 and 6 of Schedule 4 of PA 11_0047 specify the independent review process.	Noted, this has not been required in the audit period.	Not Applicable			





Reference	Clause	Doguiroment	Evidence	Audit Finding		Risk	
Kererence	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	5.1.2	Within 3 months of receiving a written request from a landholder with acquisition rights, TCPL shall make a binding written offer as detailed in Condition 4(8) of PA 11_0047. Noise acquisition criteria are specified in Condition 3(4) while Condition 4(9) requires TCPL to pay all reasonable costs associated with the land acquisition process.	Noted, this has not been required in the audit period.	Not Applicable			
	5.2 Complaints						
Noise Management Plan	5.2	from time to time such impacts may occur. In order to ensure an appropriate and consistent level of reporting, response and follow-up to any complaints is adopted by TCPL, the following complaints management protocol will be followed:	http://www.whitehavencoal.com.au/enviro nment/tarrawonga_mine_environmental_	Compliant			
Noise Management Plan	5.2	Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint. o The action taken by TCPL in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCPL, the reason why no action was taken.	Section 5.2 includes: -If no action was taken by TCPL, the reason why no action was taken. Complaints register includes noise related complaints lodged on 9/04/2014, 7:48am; 22/04/2014, 6:10pm; 8/04/2014, 4:15pm. Environmental Superintendent responded via email advising that the site Environmental Officer would contact the complainant to arrange a meeting to discuss the issues. The meeting was arranged by e-mail.	Compliant			





Reference	Clause	Paguirament	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	5.2	The Group Environment Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational). Data from the site weather station and the real time noise monitoring unit will be obtained for the time applicable to the complaint for use in determination of cause and identification of future remedial actions. Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site. Once the identified measures are undertaken, the Group Environment Manager will sign off on the relevant complaint within the Complaints Register. If necessary, follow-up monitoring will take place to confirm the source of the complaint is adequately mitigated. A copy of the Complaints Register will be kept by TCPL and made available to the CCC and the complainant (on request). A summary of complaints received every 12 months will be provided to DPE, GSC, NSC, EPA, DRE and the CCC through the AEMR/Annual Review.	Section 5.2 includes: Once the identified measures are undertaken, the Group Environment Manager will sign off on the relevant complaint within the Complaints Register. Complaints Register It seems measures were not identified in the complaint below: Complaint lodged on 9/04/2014, 7:48am; 22/04/2014, 6:10pm; 8/04/2014, 4:15pm. Environmental Superintendent responded via email advising that the site Environmental Officer would contact the complainant to arrange a meeting to discuss the issues. The meeting was arranged by e-mail.	Compliant			
Noise Management Plan	5.2	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Group Environment Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately.	Noted				
	5.3 Non-Compliance						
Noise Management Plan	5.2	With the exception of noise compliance criteria exceedances (as discussed in Section 5.1), non-compliances relating to noise would most likely relate to not achieving the required quarterly attended monitoring events. If this was to occur, appropriate advice would issue to the DPE and EPA as soon as practicable, and arrangements made to reschedule the monitoring event as soon as practically possible. Any other non-compliances of this nature will be detailed in the EPL Annual Return and/or AEMR/Annual Review, and include details as to why the non-compliance occurred.	EPL Annual Return (Jan 2013 to Jan 2014) AEMR 2012-2013	Not Applicable			
	6 Record Keeping and	Reporting Requirements					
	6.1 Record Keeping						
Noise Management Plan	6.1	Attended noise monitoring reports issued by the acoustic consultant will include details on the date and time of monitoring, location of monitoring and monitoring personnel (as required by Condition M1.3 of the EPL).	ATTENDED NOISE MONITORING – MARCH 2014 Monitoring personnel were not included in the report. Only the names of author of the report and reviewer.	Not Compliant Administrative			





Deference	Clause	Dominoment	Fridance	Adit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Plan	6.1	Data obtained from the real-time noise monitor will be available in real time, as well as through daily reporting from the monitoring unit to selected personnel. Key personnel for the provision of daily reports will be the Group Environment Manager, the Operations Manager for the Tarrawonga site and the Environmental Officer. In addition to daily reports, the monitor will be downloaded at regular intervals with data available on an archival basis.	Availability observed on-site	Compliant			
Noise Management Plan	6.1	Noise monitoring results, both from attended and real time monitoring, will be maintained by the Environmental Officer. The results will be reviewed to determine any mining related exceedances in order to instigate an appropriate response.	Observed on-site	Compliant			
	6.2 Reporting Require			,			
Noise Management Plan	6.2	A summary of noise monitoring results will be reported internally on a monthly basis as well as on a quarterly basis to the CCC via the Environment Monitoring Report. This report will be periodically uploaded onto the company's website (www.whitehavencoal.com.au).	Company website CCC Environment Monitoring Report uploaded in company website	Compliant			
Noise Management Plan	6.2	Each year, the results of the noise monitoring program will be summarised and presented in the AEMR/Annual Review together with reference to prevailing meteorological data and site activities during the measurement period(s), where relevant. Reporting will also include an analysis of the monitoring results against the exceedance criteria, previous monitoring results and predictions made in the EA. Where attenuation of plant is required, and testing of attenuation and/or sound power level tests are completed, the results of these tests will also be made available in the AEMR/Annual Review.	AEMR 2012-2013	Compliant			
Noise Management Plan	6.2	The extent of notification and reporting requirements depends on the severity of the issue, and whether the noise level is within modelled noise exceedance limits, but generally includes notification to DPE and EPA and/or the affected landholder as well as discussion in CCC Environment Monitoring Reports and the AEMR/Annual Review.					
Noise Management Plan	6.2	In addition to the reporting requirements listed above, noise quality monitoring data will be made available to the public upon request. Relevant real time data and operational responses will also be made available to the public on the Whitehaven website.	Company website contains this data, no requests for residents for data in the audit period.	Compliant			
	7 Document Review a	and Continuous Improvement					
Noise Management Plan	7	This document will be reviewed internally on an annual basis following completion of the AEMR/Annual Review, and/or following a noise related incident, audit, or modification to the conditions of approval. In addition, every 2 years, the plan will be subject to review with consultation with the relevant agencies.	As noted eslewhere in this audit, the internal review cannot be verified as there is no system for recording such reviews, the document has been regularly updated though. Recommendation made elsewhere	Compliant			





Reference	Clause	Roguiroment	Evidence	Audit Finding		Risk	
Kererence	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
		trategy - For Boggabri - Tarrawonga - Maules Creek Complex					
	3 Noise Criteria						
	3.1 Cumulative noise	Cumulative noise assessment criteria are contained in the most	T	<u> </u>	T T		
Noise Management Strategy - March 2014	3.1	recent Boggabri Coal Project Approval (09_0182, 18 July 2012), the Maules Creek Coal Project Approval (10_0138, 23 October 2012) and the Tarrawonga Coal Project Approval (11_0047, 22 January 2013). Except for the noise affected land identified in Condition 3 of Project Approval 09_0182, Table 1 of Project Approval 10_0138, and Table 1 of Project Approval 10_0138, and Table 1 of Project Approval 10_047, each mine is required to ensure that the operational noise generated by their respective project, combined with the noise generated by other mines in the BTM Complex, does not exceed the criteria in Table 3.1 at any residence on privately owned land. Table 3.1 Cumulative noise assessment criteria Location Day (Lac purch) Evening (Lac purch) Night (Lac purch) All privately-owned land 40 40 40 Note: Cumululative noise assessment or the relevant requirement, and exemptions (including ordain methological condition), of the NSW locatival Noise Policy. Operational noise includes noise from the mining operations and use of private half reads and rall spure.	Noted				
Noise Management Strategy - March 2014	3.1	Cumulative noise assessment and acquisition criteria applicable to each mine in the BTM Complex will be addressed in each mine's individual NMP.	Noted this is addressed	Compliant			
	4 Monitoring						
Noise Management Strategy - March 2014	4	The mines of the BTM Complex already have comprehensive noise management systems in place. The existing noise monitoring network will be upgraded to reflect the implementation of the BTM Complex cumulative noise monitoring network.	Not yet implemented as the strategy is still in draft.	Not Applicable			
	4.1 Existing monitorin			ı	T T		
Noise Management Strategy - March 2014	4.1	Both Boggabri Coal Mine (BCM) and Tarrawonga Coal Mine (TCM) have noise monitoring programs in place with attended and unattended (in the case of BCM) noise monitoring conducted on a quarterly basis. A noise monitoring program has been developed for MCC and is outlined in its NMP.	Noted				
Noise Management Strategy - March 2014	4.1 4.2 Cumulative noise	In addition to the attended monitoring program, BCM and TCM currently utilise separate a mobile real-time noise monitors to investigate noise complaints or issues as they may arise. BCM's real-time noise monitor is currently located at the Cooboobinbi residence.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.2	The requirements of the cumulative monitoring network at the BTM Complex are to: - facilitate compliance with existing and likely future consent conditions - allow proactive management and real-time noise monitoring to assist in day to day operations of each mine site - develop an integrated and coordinated approach to noise management of the BTM Complex - potentially consolidate existing monitoring allow for predictive meteorological forecasting to guide operations - include procedures for identifying the source(s) and contribution(s) to cumulative noise impacts for mines and other sources include appropriate investigative tools such as noise modelling.	Noted				
	4.2.1 Real-time monit						
Noise Management Strategy - March 2014	4.2.1	A real-time monitoring network will be sourced and installed. The chosen system will have capabilities of sufficient resolution to allow noise emissions from each mine to be accurately determined. Four permanent real-time noise monitors will be installed as part of the Strategy, this will include a combination of the following: - one at the Wongalea (Morris) property in the W Zone (predominantly MCC and BCM noise) - one at Warriahdool (Younger) property north of the MCC project, - one unit at Olivedene (Bastardo) west of BCM and MCC - one at either the Sylvania, Goonbri or Matong property in the E Zone (predominantly BCM and TCM noise).	Not yet implemented as the strategy is still in draft.	Not Applicable			
Noise Management Strategy - March 2014	4.2.1	As part of the proposed real-time monitoring network, two additional mobile real-time units may also be employed to variously monitor cumulative coal haulage noise impacting on SW Zone receivers, noise impacts at the nearest privately owned receivers to the MCC in the NE, NW and SW zones, at a receiver in the S zone during the prevailing winter NW winds, or to investigate noise complaints from any receiver.	Not yet implemented as the strategy is still in draft.	Not Applicable			
Noise Management Strategy - March 2014	4.2.1	The BTM Complex is working with noise specialists to determine the best configuration of monitors for the combined monitoring network.	Noted				
	4.2.2 Attended monit	oring					
Noise Management Strategy - March 2014	4.2.2	Attended noise monitoring has been conducted on a quarterly basis since inception of the TCM and BCM. Continued attended monitoring will be required to assess ongoing compliance with individual and cumulative noise criteria. Recommended monitoring locations sufficient to cover all potentially affected properties are listed in Table 4.2.	Noted				
		l-time noise management					
	4.5.1 Overview of req	uirements					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.5.1	A key method to ensure that noise management systems maintain standards of best available technology is to incorporate predictive and real-time reactive capability. It is proposed that a predictive and reactive noise management system be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and subsequently in real-time - develop a history/library of community impacts and noise incidents and events - evaluate community complaints and determine if BTM Complex activities may have caused an impact - accept information and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide recommendations with respect to abatement or avoidance of potential issues and operational requirements based on outputs of the system	Not yet implemented as the strategy is still in draft.	Not Applicable			
	4.5.2 Components						
Noise Management Strategy - March 2014	4.5.2	It is proposed that the predictive and reactive noise management system will include: - a predictive component using forecast weather data - a reactive component using real-time meteorology and noise monitoring - short term tiered trigger levels and notifications for managing potential impacts - a daily forecast report providing information on temperature inversions, wind conditions at various heights, noise risk, and recommended control actions.	Not yet implemented as the strategy is still in draft.	Not Applicable			
Noise Management Strategy - March 2014	4.5.2	It is extremely important to maintain periodic review of any real- time noise management system to ensure that the system is operating using: - validated meteorological forecasts - data from calibrated monitoring equipment - accurate noise emission levels, informed by routine attended monitoring. Periodic reviews of the model will be quarterly initially, extending	Noted				
Noise Management Strategy - March 2014	4.5.2	to annually once performance is viewed to be satisfactory. Data inputs to the system will be updated quarterly to account for any changes to mine plans or other parameters that have bearing on model performance.	Noted				
	4.5.3 Predictive Forec				T		
Noise Management Strategy - March 2014	4.5.3	It is proposed that a predictive forecast meteorology system be implemented based on the Weather Research & Forecasting (WRF) model, specifically for the BTM Complex, and a website be developed to make data immediately available for each of the mine sites, with half hourly forecasts up to 48 hours in advance. This system will download global meteorological data and forecasts on a daily basis and process and run the WRF model to produce the information required for input into a realtime 3D dispersion model.	Not yet implemented as the strategy is still in draft. The site uses predictive meteorology but the modelling is not specific to the complex, real time particulate results are used in concert with the weather data to manage operations.	Not Applicable			





Reference	Clause	Bouvinement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.5.3	Once this meteorological system is configured and operating, the outcomes will be evaluated by a competent meteorologist or atmospheric science professional quarterly against actual meteorological measurements and the meteorological system will be validated and improved, where possible.	Noted				
	4.5.4 Local observed r	meteorology					
Noise Management Strategy - March 2014	4.5.4	Data from local automatic weather stations will be used to validate the predictive meteorological forecast data as time elapses.	Noted				
Noise Management Strategy - March 2014	4.5.4	Meteorological instrumentation or data communications equipment will be reviewed to confirm that the right quality of data is available to the system.	Noted				
	4.5.5 Integrated real-						
Noise Management Strategy - March 2014	4.5.5	To enable real-time reactive feedback from the system, a connection will be established to receive a data feed from weather stations in the BTM network. A connection will also be established to monitoring equipment located upwind and downwind of noise sources. These data feeds will be connected to the system from a central data repository or directly to loggers on permanent in-field equipment.	Not yet implemented as the strategy is still in draft.	Not Applicable			
Noise Management Strategy - March 2014	4.5.5	As required, the system will be connected with operational and other environmental data and management information systems. Real-time noise management capability builds on the information gained from predictive systems to proactively manage noise. The system will be improved further by incorporating real-time sound recording to identify the instantaneous main source of high emissions. Protocols will be put in place to immediately react to rising noise levels, e.g. automated notices sent to nominated mining personnel to alert the need to respond with control/mitigation, and focus can be given to the most significant identified noise source.	Noted				
Noise Management Strategy - March 2014	4.5.5	Real-time noise monitoring data will be used primarily to determine when noise emissions are approaching compliance and to allow sufficient time to manage noise generating activities from the operations so that the criteria are not breached.					
	4.5.6 System Outputs						





D. Commercial	Cl		F 11	A drawn draw		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.5.6	Once the predictive and reactive system is implemented and configured as described above, a range of user interfaces or simple reports or templates can be generated and used as part of standard operating procedure. Some system outputs that will be required include: - Daily forecast reports providing information on temperature inversions, wind conditions, noise risk, and recommended control actions. - Graphical representation of the forecasted meteorology and realtime monitoring data via the system's web interface. - Capability to analyse and confirm the likely source(s) of noise. This functionality is critical in apportioning responsibility to operations for mitigating noise emissions. - Automated alerts for relevant operations personnel so that the agreed protocol for reacting to a potential noise issue can be activated. These alerts may be generated as SMS or email messages, or by other systems integrated into operating processes (depending on needs). Alerts will be stored in the system for analysis, which will assist in refining trigger criteria over time.	Noted				
	4.6 Mitigation						
Noise Management Strategy - March 2014	4.6.1 Trigger levels 4.6.1	Trigger levels are proposed to be communicated via SMS, email and/or other systems integrated into operating processes (depending on needs). The following trigger levels are currently proposed for the system: 'Investigation' level criteria will be triggered initially when any of the real time monitors reaches a level 3 dB below the cumulative noise criteria. Investigation into the trend of increased noise levels will be conducted upon reaching this trigger level. This investigation will involve: - comparing the low-pass LAeq level with the total LAeq level to determine whether mine noise (predominantly lower-frequency) is a likely cause of elevated noise levels - listening to the most recent audio files to see if the source(s) of the increasing noise can be identified - reviewing meteorological data to determine whether increasing noise levels may be due to wind or temperature inversions. 'High' level criteria will be triggered initially when any of the real time monitors reaches a level 1 dB below the cumulative noise criteria. Implementation of individual mine noise mitigation actions will be instigated upon reaching this trigger level.	Noted, Tarrawonga had trigger levels and alarms in place at the time of the audit but does not coordinate the results with the surrounding mines				





Deference	Clause	Doguiromont	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.6.1	Once the real-time noise management system is operational, trigger levels will be reviewed, updated and refined following a review of the data and calibration of the system. If the trigger levels are not appropriate to site operations (too many or too few investigation or action responses) they will be reviewed and updated. Different trigger levels may be set for each monitoring location within the cumulative network, for example be set lower for monitoring locations closer to noise sources. Trigger levels will also be regularly assessed as part of the ongoing review of this plan.	Noted				
	4.6.2 Processes to mit	igate outcomes					
Noise Management Strategy - March 2014	4.6.2	The noise investigation instigated by the cumulative noise trigger levels will highlight which mine(s) is the cause of the increasing noise levels and the relevant mine(s) will also have individual trigger levels, based on their own noise criteria, which will be significantly lower than the cumulative noise triggers.	Noted				
Noise Management Strategy - March 2014	4.6.2	Processes to mitigate noise outcomes associated with operations are addressed in each sites individual NMPs. Examples of best practice mitigation options are: - moving identified sources to locations that are more distant from, or geographically shielded from, the receiver - stopping the activity/plant - installing temporary noise barriers - re-aligning directional sources (e.g., drills) so that the quietest side faces the noise-affected receiver(s).	Noted				
Noise Management Strategy - March 2014	4.6.2	Operational activities will be ranked based on noise generation potential. Of particular importance is the consideration of wind direction information provided by the predictive meteorology forecast and the local weather data. Receivers which are generally NW of the BTM Complex (nearest to the MCC) will be the potentially most affected during the prevailing S-SE winds during the warmer months. Conversely, the prevailing N-NW winds during the cooler months will reduce noise for receivers NW of the Complex and increase noise levels at receivers generally south of the Complex.	Noted				
Noise Management Strategy - March 2014	4.6.2	The "impact potential" rankings will be used as the basis for proactively scheduling operational activities when noise generation is predicted to reach trigger levels. Examples of pro-active mitigation measures are: - ensuring plant achieves required noise specification - making high-level and low-level OEAs available simultaneously. The low-level (often in-pit) emplacements can be used under inversion and adverse wind conditions - suspending the use of identified major noise sources (e.g., dozers on top of emplacement areas) until after nocturnal inversions have lifted	Noted				
Noise Management Strategy - March 2014	4.6.2	Noise generation assessment will be undertaken by experienced site personnel with the assistance of various specialists as required (e.g. environmental and acoustic specialists).	Noted, as is currently the position.	Compliant			
	4.7 Communication						





Deference	Clause	Doguiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Noise Management Strategy - March 2014	4.7	Regular meetings are being held (and will continue to be held at least quarterly) between BTM personnel to discuss various cumulative impacts. This includes discussing real-time and attended monitoring results and future operational events. Meeting minutes will continue to be documented and retained at each site.	These meetings occur monthly and all environmental issues are discussed.	Compliant	·		
Noise Management Strategy - March 2014	4.7	Trigger levels will initiate internal communication between the BTM mines to allow the BTM Complex to implement individual management measures in order to minimise noise generation.	Noted				
Noise Management Strategy - March 2014	4.7	When noise criteria are identified as exceeded, discussions will be held between BTM and the agencies and affected landholders (where an exceedance occurs on privately-owned land) advised.	Noted				
	4.8 Reporting						
Noise Management Strategy - March 2014	4.8	Internal management reports will be prepared regularly, noting performance against triggers and criteria.	Noted				
Noise Management Strategy - March 2014	4.8	External reporting will be undertaken by all mines within the BTM Complex and include: - Updates on individual Company websites - Presentations to Community Consultative Committees (CCCs) - Annual Environmental Management Reports (AEMRs)/Annual Reviews - exceedance reporting.	As per other documents, this is gennerally complied with, exceptions noted elsewhere in the audit.				
	5 Implementation	, , ,					
	5.1 Staged approach						
Noise Management Strategy - March 2014	5.1	A staged approach will be taken to install the equipment and systems which are additional to individual mine's existing noise monitoring systems. This staged approach of implementation is detailed in Figure 5.1 and described in the following sections. Stage 1 Equipment acquisition, installation and commissioning Stage 2 Review equipment and processes Figure 5.1 Staged approach to implementation	Noted				
	5.1.1 Stage 1 - Equip	ment acquisition, installation and commissioning					
Noise Management Strategy - March 2014	5.1.1	Stage 1 will include the acquisition, installation and commissioning of the proposed permanent realtime directional noise monitors. This equipment will complement the existing mobile real-time noise monitor owned by BCPL and TCPL.	Noted				





Defenses	Clause	Barrian and	Fridance	Audia Findina		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
		equipment and processes					
Noise Management Strategy - March 2014	5.1.2	Within three months of the installation and commissioning of Stage 1 equipment the following will be reviewed: - performance and reliability of the cumulative noise monitoring equipment - triggers proposed in this Strategy - central data repository and data interface.	Noted				
	5.1.3 Stage 3 - Publish	webpages					
Noise Management Strategy - March 2014	5.1.3	Each mine site will establish or update an existing Company webpage. The webpage will present the summarised and validated results of the real-time noise monitoring on a monthly basis.	Noted				
	6 Document Control						
	6.1 Review and revision						
Noise Management Strategy - March 2014	6.1	This NMS, its operation and implementation, will be reviewed and revised at least every two years or on an 'as required' basis to incorporate improvements identified by the BTM Complex or appropriate requirements of government agencies. The NMS will be reviewed and updated at the end of each stage of the project rollout	Noted	Not Applicable			
Noise Management Strategy - March 2014	6.1	In accordance with the project approvals, the NMS will also be revised within three months of: - an annual review - incident threatening material harm, requiring notification of the Director-General / relevant agencies - statutory audit - modification of project approval.	Noted	Not Applicable			





			Evidence			Risk		
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
	WHC_PLN_TAR_Blast	t Management Plan					111011	<u> </u>
	2 Statutory Requirem	nents						
Blast Management Plan	2.1	The Applicant shall prepare and implement a Blast Management Plan for the development to the satisfaction of the Director-General. The program must: (a) be submitted to the Director-General for approval by the end of May 2013;	Submitted prior to the required date but not yet approved.	Compliant				
Blast Management Plan	2.1	(b) be prepared in consultation with the EPA and interested members of the local community who would potentially be affected by blasting;	email is at back of mngt pln	Compliant				
Blast Management Plan	2.1	(c) propose and justify any alternative ground vibration limits for public infrastructure in the vicinity of the site;	Noted, no changes as yet	Not Applicable				
Blast Management Plan	2.1	(d) describe the measures that would be implemented to ensure: • best management practice is being employed; and • compliance with the relevant conditions of this approval;	Included in the BMP, FMP and RCMP	Compliant				
Blast Management Plan	2.1	(e) include a road closure protocol for blasting within 500 metres of a public road, that has been prepared in consultation with council;	Road Closure Management Plan	Compliant				
Blast Management Plan	2.1	(f) include a specific blast fume management protocol to demonstrate how emissions will be minimised, including risk management strategies if blast fumes are generated;	Blast Fume Management Protocol provided.	Compliant				
Blast Management Plan	2.1	(g) include a monitoring program for evaluating blasting performance, which includes • compliance with the applicable criteria; and • minimising blast fume emissions; and	There is a separate Blast Monitoring Program that addresses this requirement	Compliant				
Blast Management Plan	2.1	(h) include a Leard Forest Mining Precinct Blast Management Strategy, that has been prepared in consultation with other mines within the Leard Forest Mining Precinct, to minimise cumulative blasting impacts.	This is included though not yet approved. The sites share blasting timetables and tend to blast at different times of the day which helps prevent cumulative impacts.	Compliant				
Blast Management Plan	2.1	In accordance with Condition 3(14) of PA 11_0047, Conditions 11(a) and (b) of Mining Lease (ML) 1579 and Conditions 15.1 – 15.7 of Environment Protection Licence (EPL) 12365 the airblast overpressure and ground vibration blasting criteria is as follows: Table 1 Blasting Criteria Blasting criteria is as follows: Table 1 Blasting Criteria Blasting Crit	Noted					
Blast Management Plan	2.1	PA 11_0047 and EPL 12365 restrict blasting operations to 1 blast per day, unless an additional blast is required following a blast misfire. Furthermore, PA 11_0047 restricts blasting operations to 4 blasts a week, averaged over a calendar year. Blasting operations will be carried out between 9am and 5pm Monday to Saturday inclusive.	15-05-14 and 17-04-13, two blasts on the same day but second one was for safety reasons.	Compliant				
Blast Management Plan	2.1	All aspects of blast management will be undertaken in accordance with AS 2187.2-2006 - Storage and Use – Use of Explosives, a copy of which will be retained at the site office.		Compliant				
	3 Blasting Controls ar	nd Management Procedures						





	-		Evidence			Risk		
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	3	Tarrawonga Coal Mine seeks to minimise air blast overpressure, ground vibration levels, flyrock, fume, dust and odour from blasting activities. Control of ground vibration, overpressure and flyrock impacts will be achieved by implementing the procedures and safe guards indicated as follows: • Comply with the relevant procedure prior to the commencement of any blast by referring to the internal documents WHC_STD_OC_EXPLOSIVES, WHC_PLN_TAR_EXPLOSIVES MANAGEMENT PLAN, and WHC_PLN_TAR_SITE SECURITY PLAN;	Noted, plans were provided for review and are the management documentation for these issues along with the BMP	Compliant				
Blast Management Plan	3	Undertake a pre-blast environmental assessment with consideration given to meteorological conditions such as cloud cover, wind speed and direction and the strength of temperature inversions prior to each blast;	Observed on site	Compliant				
Blast Management Plan	3	Comply with blast loading and pre blast designs, unless risks are determined by the shotfirer at the time of loading that may be mitigated through changes to design;	Confirmed at interview (D&B Eng)	Compliant				
Blast Management Plan	3	Use of suitable quality stemming material and the use of adequate stemming lengths to ensure maximum internment of explosive charges, therefore minimising overpressure; and	25mm gravel	Compliant				
Blast Management Plan	3	Use of monitoring data to establish and estimate the likely overpressure and vibration levels during the design process of subsequent blasts.	Yes this occurs, confirmed at interview (D&B Eng)	Compliant				
	3.1 Structural and Hu							
Blast Management Plan	3.1	The Tarrawonga Coal Project Environmental Assessment assessed the potential for structural damage from blasting at surrounding project related and privately owned properties. The assessment identified that the potential for a marginal exceedance of structural damage vibration criteria was limited to the project related "Blair Athol" residence, located to the south-east of the project site, later in the project life. The residence will be vacated when structural damage criteria exceedances are predicted and a structural inspection will be undertaken prior to reoccupation if monitoring confirms an exceedancethat an exceedance in blast criteria occurred.	OC progress is not to the point where the Blair Athol residence is impacted by higher level vibration as yet.	Not Applicable				
Blast Management Plan	3.1	Predicted overpressure and ground vibration levels at all other receivers are such that the potential for structural damage or impacts on human comfort resulting from blasting is negligible	Noted					
Blast Management Plan	3.1	The blasting assessment also identified that there is no expected impact from blasting on the proposed low permeability barrier to be constructed on the eastern perimeter of the final extent of the open cut pit. It is proposed that the final design of the low permeability barrier would consider the potential impacts from blasting on the consolidation of the soil-bentonite mixture and the subsequent differential settlement to the adjacent geological sequences and keying in of the cut-off barrier. Upon completion of the design and construction of the low permeability barrier, this management plan will be updated to reflect any additional requirements identified for protection of the barrier from blast related impacts.		Not Applicable				
	3.2 Aboriginal Heritag	ge						
Blast Management Plan	3.2	The nature of the known artefacts (isolated finds, artefact scatters and culturally modified trees) suggests that blasting is likely to have minimal impact on these features, with no predicted impacts from blast vibration and flyrock. Known artefact sites will be subject to regular inspections, as part of the site monthly inspection regime, to verify blasting activities.	unlikely due ot the nature of the aboriginal	Not Applicable				
	3.3 Livestock							





			Evidence			Risk		
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	3.3	Avoidance of impacts on livestock from flyrock is managed via a procedure for blast clearance and firing which requires consideration to be given to the location of livestock when developing the Blast Clearance Plan. Where livestock are identified as being within the blast clearance zone, appropriate measures will be taken to relocate livestock from this location prior to blasts proceeding. At no stage throughout the life of the mine will blasting activities be within 500 metres of privately owned land.	No blasting has occurred within 500m of cattle.	Not Applicable				
	3.4 Management of F	Road Closures						
Blast Management Plan	3.4	A Tarrawonga Coal Mine Road Closure Management Plan, refer to Appendix 3 has been developed in consultation with Narrabri Shire Council (NSC), for approval by the Director General, to put in place protocols and procedures to minimise impacts on the local community. The main objectives are to: • Ensure safety and protection of the public, residents, property and livestock;	RCMP provided. The plan includes the	Compliant				
Blast Management Plan	3.4	Coordinating blast schedules with neighbouring mines to minimise cumulative impacts of blasting;	24 hour notification to other sites .	Compliant				
Blast Management Plan	3.4	Notify in advance relevant stakeholders, including the public and nearby properties, of blasts that will temporarily close local roads;		Not Applicable				
Blast Management Plan	3.4	Minimise road closures and the potential impacts on road users, local residents and businesses, through avoiding peak traffic periods;		Not Applicable				
Blast Management Plan	3.4	Blasting is planned to be undertaken within 500 metres of Goonbri Road within the next 3 years. Prior to submission of this BMP for approval by the Director-General, Whitehaven contacted landholders and tenants of properties closest to the proposed road closures who will likely be most affected by the closures. The local school bus driver was also contacted. No further comments were received from those contacted.	No local roads closed yet, no blasts within 500m.	Not Applicable				
	3.5 Air Vibrations (Ov	verpressure)						
Blast Management Plan	3.5	Noise (the audible part of the air vibration spectrum) and airblast (the remaining sub-audible part of the air vibration spectrum) generation can be controlled by ensuring that all, or nearly all, of the explosion energy is consumed in fragmenting and displacing the overburden by the time the gases vent (via the broken burden rock and/or ejected stemming material) into the atmosphere.	Noted					
Blast Management Plan	3.5	This objective would be met by ensuring that: • Where practicable, the blast face is orientated away from or at an oblique angle to nearby residences; • Blasthole spacing is implemented in accordance with blast design; • The burden distance and stemming length are carefully selected and then implemented precisely; • Appropriate materials (eg. 20mm aggregates) are used for stemming; • Charges detonate in the correct sequence and with inter-row delays that provide good progressive release of burden; and either and the maximum weight of explosive detonated in a given delay period (the MIC) is limited to conservative and proven levels. • Conducting blasting both before the establishment, and after the break-up of low-level atmospheric temperature inversions which typically occur after 4:00pm and can remain until 8.00am.	D&B Engineer - all confirmed at interview.	Compliant				
	3.6 Ground Vibration:	s						





			Evidence			Risk		
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	3.6	Ground vibration would be controlled by ensuring: • The minimum practicable weight of explosive detonates at an instant (ie. minimising the MIC) by using the maximum number of delay periods in each blast; and • Most of the energy liberated by the charge(s) on a given delay number is consumed in providing good fragmentation, adequate displacement and/or a loose, highly dig-able muckpile, rather than in creating ground vibrations (ie. by ensuring that the burden distance and effective sub-drilling are not too large).	Covered above, note also that there have been overpressure exceedances at company owned properties but vibration is well within limits.	Compliant				
	3.7 Dust and Other P			•	1			•
Blast Management Plan	3.7	Control of blast associated dust generation, odour and fumes will be achieved by the following: • Undertaking a pre-blast environmental assessment with consideration given to meteorological conditions such as cloud cover, wind speed and direction and the strength of temperature inversions prior to each blast;	Checklist sighted at site, also ran through checklist with EO for a shot.	Compliant				
Blast Management Plan	3.7	Minimising the potential for the delay in firing of shots which have been loaded, especially into wet holes, considering the restrictions of existing weather conditions;	Confirmed at interview (D&B Eng)	Compliant				
Blast Management Plan	3.7		Blasting checklist	Compliant				
Blast Management Plan	3.7	If inversion conditions are detected or wind speeds are greater than 7m/s emanating from the West or North West, blasting may be delayed if it is considered these conditions pose a risk to environmental compliance, until more favourable weather conditions are encountered;	Noted, checklist considers these issues	Compliant				
Blast Management Plan	3.7	After the blast initiation, monitor and photograph any blast fume to record the direction of travel and dispersion of the fume cloud; and	Yes including offsite monitoring, also sat with the offiste dust plume observer and observed the actions for a shot.	Compliant				
Blast Management Plan	3.7	Fume Management Plan						
Blast Management Plan	3.7	 Pollution Incident Response Management Plan (PIRMP) is applicable across all Tarrawonga Coal Mine operations subject to EPL 12365 and is implemented in the event that of a pollution incident. 	Noted					
Blast Management Plan	3.7	 In the event a blast fume cloud is created and is travelling in the direction of local residences, the fume cloud will by continually monitored by the Environmental Officer, Blast Controller and Shot Firer as well as blast sentries and every endeavour will be made to contact nearby personnel and residents to inform them of any impending fume with advice to remain indoors until the fume has passed. 	Noted, no incidents to date, see noted above re blast observer	Not Applicable				
Blast Management Plan	3.7	Blast-generated dust would be minimised by ensuring that stemming columns are not ejected for considerable distances into the atmosphere. Stemming column lengths would be such that their ejection velocities are low.		Compliant				
Blast Management Plan	3.7	Additionally, the blasting contractor would be required to use aggregates for blasthole stemming and to use nonel delay-type or electronic detonators to initiate charges. The use of nonel-type delay or electronic detonators would avoid the requirement for detonating cord downlines and, with the absence of detonating cord trunklines (ie. surface lines), prevents the dust cloud that is formed when such trunklines detonate on a dry dusty surface.	Confirmed at interview (D&B Eng)	Compliant				





		Par i mara	Evidence			Risk		
Reference	Clause	Requirement	Lindende	Audit Finding	Consequence	Likelihood	Risk	Responsibility
	3.8 Impacts on the Pr	roposed Low Permeability Barrier						
Blast Management Plan	3.8	Blast controls or measures that shall be implemented to reduce the potential impacts of blasting on the proposed low permeability barrier will be addressed and appended to this management plan upon completion of the final design of the low permeability barrier.	Design of permiability barrier not completed and postponed.	Not Applicable				
	3.9 Cumulative Blast	Management Strategy						
Blast Management Plan	3.9	TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project and has developed a Leard Forest Mining Precinct Blast Management Strategy in order to minimise cumulative blasting impacts. The Leard Forest Mining Precinct Blast Management Strategy is currently under review by the Department of Planning and Infrastructure. Upon confirmation of its acceptance, it will be appended to this Management Plan and implemented accordingly.						
	4 Inspections and No							
	4.1 Pre-Blasting Inspe		It attend and it is a set of					
Blast Management Plan	4.1	As a requirement of Condition 4(1), within 3 months of Project Approval, TCPL notified in writing the owners of any privately owned land within 2 kilometres of the approved open cut mining pit/s that they are entitled to ask for a property inspection, to establish the baseline condition of any buildings or structures on their land, or to have a previous property inspection report updated.	Letters provided as evidence	Compliant				
Blast Management Plan	4.1 4.2 Pre-Blasting Notif	In accordance with Condition 3(17) of PA 11_0047, if TCPL receives a written request from the owner of any privately owned land within 2 kilometres of the approved open cut mining pit on site for a property inspection to establish the baseline condition of any buildings and/or structures on their land, or to have a previous property inspection report updated, then within 2 months of receiving this request TCPL will: (a) Commission a suitably qualified, experienced and independent person, whose appointment is acceptable to both parties, to: (b) Establish the baseline condition of the buildings and/or structures on the land or update the previous property inspection report; (c) Identify any measures that should be implemented to minimise the potential blasting impacts of the development on these buildings and/or structures; and (d) Give the landowner a copy of the new or updated property inspection report.		Not Applicable				
	4.2 Pre-Blasting Notificati 4.2.1 Public Notificati							
Blast Management Plan	4.2.1	Prior to TCM commencing, surrounding landholders were contacted to determine if they required notification prior to each blast. TCM provides notification to any person who registered an interest in being notified about the blasting schedule at the mine. Notification consists of an email or phone call (depending on the individual's preference) 24 hours prior to the blast and/or on the day of the blast. This notification procedure has been successfully implemented for approximately 8 years.	Surveyor sends notifications, adminstration makes calls and EO records contacts.	Compliant				
Blast Management Plan	4.2.1	Notification to the general public about proposed blasting dates and times is provided via the blast notification sign at the entrance to the mine site. In addition, details about each upcoming blast (including date, time and road closure information) are provided under the "Community" tab on the Whitehaven Coal website (www.whitehavencoal.com.au).	Observed on the web and on the gatepost sign	Compliant				
	4.2.2 Liaison with Adj	acent Mines						





			Evidence			Risk		_
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	4.2.2	TCM and the adjacent Boggabri Coal Mine have developed a procedure whereby 24 hours' notice is provided prior to a proposed blast. Notification is in the form of an email to all relevant mine personnel. TCM schedule blasts to initiate at 12:00pm & 2pm whereas Boggabri Coal initiates blasts at either 10.30am or 3pm (where possible). This ensures cumulative impacts are avoided if blasts are scheduled on the same day at both mines. Personnel from both mines liaise accordingly (eg. via email or phone) if blast times need to be altered.	This now also includes MC OC as they are progressing towards production.	Compliant				
Blast Management Plan	4.2.2	Cumulative impacts that may be generated between the TCM and Whitehaven's Rocglen Coal Mine are managed by the Environmental Officer's for both Tarrawonga and Rocglen mines ensuring blasts are not scheduled for the same time.	Noted see responses above.	Compliant				
	4.3 Property Investiga	ations						
Blast Management Plan	4.3	In accordance with Schedule 3, Condition 18 of PA 11_0047, if any owner of privately-owned land within 2 kilometres of blasting operations, or any other landowner nominated by the Director-General, claims that the buildings and/or structures on his/her land have been damaged as a result of blasting on site, then within 2 months of receiving this claim in writing from the landowner, the Proponent shall: (a) commission a suitably qualified, experienced and independent person, whose appointment is acceptable to both parties, to investigate the claim; and (b) give the landowner a copy of the property investigation report.	No such claims	Not Applicable				
Blast Management Plan	4.3	If this independent property investigation confirms the landowner's claim, and both parties agree with these findings, then the Proponent shall repair the damages to the satisfaction of the Director-General.	Noted					
Blast Management Plan	4.3	If there is a dispute over the selection of the suitably qualified, experienced and independent person, or the Proponent or landowner disagrees with the findings of the independent property investigation, either party may refer the matter to the Director-General for resolution.	Noted					
	5 Monitoring and Rep	orting						
	5.1 Monitoring Progra							
	5.1.1 Parameters Mea	asured and Monitoring Frequency	I	ı	1			1
Blast Management Plan	5.1.1	EFC 123G5 specifies the following monitoring parameters to be monitored. Monitoring must be undertaken for each blast as specified in Table 2. Table 2	Some events don't comply due to missed monitoring data - blast results records show a number of missed results due to lack of triggering. Interview with EO indicated that these results were from a period where a contractor managed blasting including monitoring. Currently, blasting and monitoring is managed by TCL and there have been no missed results since the contractor left site. No action required by TCL as the issue has been resolved.	Not Compliant	E	3	Low	
Blast Management Plan	5.1.1	In addition to blast monitoring at the nominated sites, TCM also completes an Environmental Blast Checklist which includes a notification checklist, pre-blast weather conditions assessments every hour prior to the blast (commencing at 5 hours prior to the blast) and at the time of the blast and a post-blast assessment which includes fume rating (Appendix 4).	Sighted on-site	Compliant				
	5.1.2 Monitoring Loca	ILIUIIS						





D. Communication of the Commun	Cl	Para tananan	Evidence	A drawn draw		Risk		5
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	5.1.2	EPL 12365 currently requires monitoring to be conducted at the residence listed in Table 3 and shown in Figure 2. In addition to point N1, monitoring is also conducted at the project related "Matong" property for TCM purposes to account for any identified impacts to the South East of the mine (Table 3 and Figure 2). Table 3 Monitoring Locations EPA Identification No. Description of Location N1 Within 30m of the residence on properly "Tatawongs"	These are the sites that are monitored regularly.	Compliant				
Blast Management Plan	5.1.2	The 'Tarrawonga' property has been acquired by Whitehaven and is now project related; therefore the blast monitor will be relocated to the 'Braymont' property located to the South-Southwest of the mine. This relocation will take place once EPL 12365 variation has been approved by the EPA. In the meantime, blast monitoring will continue at the "Tarrawonga" residence as it will continue to provide indicative blasting impacts given its close proximity to the Tarrawonga site.	Noted, License still specifies Tarrawonga as license monitoring point N1	Compliant				
Blast Management Plan	5.1.2	In addition to monitoring of blast overpressure and vibration, each blast is monitored visually for indications of fume or any ejection from the shot face. All blasts are also captured via video for playback in the event of a blast not performing to expectations. This allows for subsequent review and identification of possible contributors to a blast outcome.	Complies, observed whilst onsite	Compliant				
	5.1.3 Permanent Blas		In the agency	T	I			
Blast Management Plan	5.1.3	Permanent Dynamate blast monitors are currently established at the project related "Matong" property and privately owned Tarrawonga property (refer to section 5.1.2 for update) to monitor air-blast overpressure (dBL) and peak particle velocity in a radial, vertical and transverse direction (mm/s), i.e. ground vibration. After each blast, information gathered on a blast can be downloaded via the internet or wireless download. These are then analysed by the blast technician and the environmental officer to check compliance with blasting criteria. The monitors are subject to quarterly maintenance inspections by the supplier, as well as annual calibration.	As noted in AEMR Maintenance and calibration records provided as evidence	Compliant				
Blast Management Plan	5.1.3	The Dynamate blast monitors do not require any pre-blast set up and operate continuously. The monitors do not require any post-blast download, with data available from the host platform immediately following the blast. The blast results will be delivered to the key personnel (Operations Manager, Blast Engineer and Environmental Officer) via text message and email immediately following the blast. Further analysis of blast results can then be undertaken by accessing the web platform. The monitors are compliant with Australian Standard AS2187.2-2006.	Noted					
	5.2 Reporting	Disct manifest regults are available to relevant White-	Observed on site post bloot					
Blast Management Plan	5.2	Blast monitor results are available to relevant Whitehaven personnel, including the Environmental Department, immediately following the blast via SMS, Email or online. The Environmental Blast Checklist is filed onsite once completed.	Observed on-site post blast	Compliant				
Blast Management Plan	5.2	Blast monitoring results will be reported via quarterly Community Consultative Committee (CCC) meetings and annually to relevant government agencies through the AEMR/Annual Review.	AEMRs and CCC minutes support this	Compliant				
Blast Management Plan	5.2	Blast results are reported in accordance with EPL 12365 in the monthly and annual returns.	Exceedances reported in annual returns, and in monthly reporting that is on the website.	Compliant		_		





Deference	Clause	Dt	Evidence	A codity File disease		Risk		Daniel and the little
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
		idents, Exceedances, Non Compliances and Complaints						
	6.1.1 Blasting Related	Incidents	Ta 6		ı		1	1
Blast Management Plan	6.1.1	Safety related incidents (such as misfires) will be recorded and managed via the Whitehaven incident management process. Such incidents are managed in accordance with the Coal Mine Health and Safety Act 2002 and Coal Mine Health and Safety Act Regulation 2006, including notification to DRE NSW and the District Check Inspector by the site's Manager Mining Engineering.	Confirmed at interview (D&B Eng)	Compliant				
	6.1.2 Blasting Criteria							
	6.1.2.1 Agency Notific				1		1	1
Blast Management Plan	6.1.2.1	In the event that the monitoring results of a blast identify an exceedance of: • Peak vector sum velocity (ground vibration) – 5mm/s (ppv); and/or • Peak overpressure – 115dBL, TCM will initiate investigation with Orica Mining Services as to the cause of the exceedance.	Noted see blast register, Orica are no longer the drill and blast contractor. Sighted investigation documents whilst onsite.	Compliant				
Blast Management Plan	6.1.2.1	It is noted that the above criteria are able to be exceeded for up to 5% of the blasts in any one year but not to exceed a: • Peak vector sum velocity (ground vibration) - 10mm/s (ppv); and/or • Peak overpressure – 120dBL.	Blast register indicates that the 5% limit was not exceeded in the audit period. AEMRs support this.	Compliant				
Blast Management Plan	6.1.2.1	As required by Conditions 5(8) and 5(9) of PA 11_0047 and EPL 12365 Condition R2, TCM must notify DoPl and EPA of any blasting related non-compliances or exceedances as soon as practicable after TCM becomes aware of the issue. Additionally, within 7 days of the date of the incident, TCM must provide DoPl and EPA with a detailed written report on the incident. The report would include details such as the date, time and nature of the non-compliance or exceedance, the cause or likely cause, the action taken to date and proposed measures to minimise potential for non-compliances or exceedances with future blasts.	No exceedence of blast criteria in the audit period	Not Applicable				
Blast Management Plan	6.1.2.1	Exceedances will also be reported in the blasting discussion in the AEMR/Annual Review.	No exceedence of blast criteria in the audit period	Not Applicable				
	6.1.2.2 Landholder No	otification						
Blast Management Plan	6.1.2.2	As required by Condition 3 of Schedule 4 of PA 11_0047, results obtained showing an exceedance of blasting criteria, TCM will notify the affected landowners and tenants in writing of the exceedance as soon as practicable, and provide regular monitoring results to each of these parties until the mine is complying with the relevant criteria.	No exceedence of blast criteria	Not Applicable				
	6.1.3 Complaints				·			





			Evidence			Risk		
Reference	Clause	Requirement		Audit Finding	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan	6.1.3	Any complaints received will be managed in accordance with complaints management protocol described as follows: • A publicly advertised telephone complaints line will be in place to receive complaints during operating hours and record complaints at other times. • Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint. o The action taken by TCM in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCM, the reason why no action was taken.	See other complaints results elsewhere in the audit	Compliant				
Blast Management Plan	6.1.3	 The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational). 	See other complaints results elsewhere in the audit	Compliant				
Blast Management Plan	6.1.3	Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site.	See other complaints results elsewhere in the audit	Compliant				
Blast Management Plan	6.1.3	Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register.	See other complaints results elsewhere in the audit	Compliant				
Blast Management Plan	6.1.3	If necessary, follow-up monitoring or will take place to confirm the source of the complaint is adequately mitigated.	See other complaints results elsewhere in the audit	Compliant				
Blast Management Plan	6.1.3	• A copy of the Complaints Register will be kept by TCM and made available to the Tarrawonga Coal Mine Community Consultative Committee (CCC) and the complainant (on request). Complaints are also listed on the Whitehaven website (www.whitehavencoal.com.au) and a summary of complaints received every 12 months will be provided to DoPI, NSC, GSC, EPA, DRE and the CCC through the AEMR/Annual Review.		Compliant				
Blast Management Plan	6.1.3	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Environment Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately. Continuous Improvement	Noted					
	/ Documentation and	Continuous improvement	Reviews occur, from interview with EO but	<u> </u>				
Blast Management Plan	7	This document will be reviewed following any significant changes (ie. changes to monitoring locations) and in accordance with Schedule 5, Condition 5 of PA 11_0047 which states: 5. Within 3 months of the submission of an: (a) annual review under condition 4 above; (b) incident report under condition 8 below; (c) audit under condition 10 below; or (d) any modification to the conditions of this approval, the Proponent shall review, and if necessary revise, the strategies, plans and programs required under this approval to the satisfaction of the Director-General.	there is no documentation to show the review has occurred. For example, there has been an Annual Review in the audit period but there is no evidence of review of the BMP following the Annual Review. Recommendation made	Compliant				
Blast Management Plan	7	Each review will be undertaken in consultation with relevant stakeholders and will be submitted to the Director-General for approval.	There is no evidence of consultation and approval by the DG of reviews of the BMP but the plan is not twelve months old and hs not yet been approved.	Not Applicable				





Reference	Clause	Requirement	Evidence	Evidence Risk Audit Finding	Risk			Responsibility
Reference	Clause	Requirement		Addit Filluling	Consequence	Likelihood	Risk	Responsibility
Blast Management Plan		TCM will investigate and implement ways to improve the environmental performance of the project over time. This will be achieved by keeping abreast of best practice in the industry for blast management and monitoring options and reporting on outcomes of blasting in the AEMR/Annual Review.	Networking - D&B Eng	Compliant				





Deference	Clause	Requirement	Evidence	A reality Fine alterna		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	Blast Management St	rategy - January 2013					
	3 Regional Strategies						
Blast Management Strategy - January 2013	3	The conditions of approval for the Boggabri Coal Mine and Maules Creek Coal specifically require the three mines of the BTM Complex produce joint strategies for: - noise management - blast management - air quality management - water management - regional biodiversity (developed over 3 stages) - biodiversity offsets.	Noted				
Blast Management Strategy - January 2013	3	Additionally, the conditions require cooperation and consultation between the mines with respect to: - Aboriginal heritage conservation - operational noise and air quality management, including online communications of onsite activities and monitoring; operating conditions and reactive dust management; and air quality and Greenhouse Gas (GHG) management - transport, specifically options for transporting workers - management of social impacts - membership of Community Consultative Committees (CCC).	Noted				
	4 Blasting criteria						
	4.1 Boggabri Coal Min	ne					
Blast Management Strategy - January 2013	4.1	The relevant blasting criteria have been extracted from the most recent BCPL Project Approval, and are summarised in Table 4.1. Table 4.1 Boggabri Coal Mine blasting assessment criteria Location Arbitast overpressure Ground vibration (minis) Allowable exceedance (doll, in Peak)) Residence on privately owned land 120 10 0% privately owned land 115 5 5% of the total number of blasts over a period of 12 months of 12 months over a period of 12 months of 13 months of 13 months of 14 months of 14 months of 15 months of 1	Noted, these are the same criteria at each site.				
Blast Management Strategy - January 2013	4.1	The process for day to day management of compliance with respect to these conditions is outlined in the BCPL Blast Management Plan (BLMP).	Noted				
	4.4 BTM Complex						
Blast Management Strategy - January 2013	4.4.	Table 4.4 summarises the current assessment criteria for the three mines of the BTM Complex. Table 4.4 BTM Complex blasting assessment criteria Criteria Brogabri Terremonga Maudes Creek	Noted				
	5 Blast monitoring						
	5.1 Existing monitorin	g network					
		0					





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Blast Management Strategy - January 2013	5.1	The mines of the BTM Complex already have comprehensive blast management systems in place. The existing blast monitoring network will continue to be used.	Noted				
Blast Management Strategy - January 2013	5.1	It is expected that little change will be required to the existing blast monitoring network to continue to ensure compliance with respect to blasting in the BTM Complex. However, there will need to be additional cooperation between mines of the BTM Complex, to minimise the potential for cumulative impacts. Protocols described in this BLMS will be used with the existing monitoring programs to ensure that blasting schedules are coordinated to avoid cumulative impacts on sensitive receivers.	Noted				
	5.2 Predictive forecas						
Blast Management Strategy - January 2013	5.2	It is proposed that more extensive changes will be made to the air quality networks within the BTM Complex in order to manage compliance. The changes, which are detailed in the BTM Complex Air Quality Strategy, include a proposed predictive forecast meteorology system, with half hourly forecasts up to 48 hours in advance. This system will download global meteorological data and forecasts on a daily basis that will be used to guide the planning of blasting activities.	Not yet implemented as the strategies are still draft.	Not Applicable			
Blast Management Strategy - January 2013	5.2	Once the proposed meteorological system is configured and operating, the outcomes will be evaluated by a competent meteorologist or atmospheric science professional against actual meteorological data and the meteorological system will be validated and improved, where possible.	Not yet implemented as the strategies are still draft so no peer review at this point.	Not Applicable			
	6 Cumulative blast ma	-					
Blast Management Strategy - January 2013	6.1 Mitigation of cum	The key management measure for the mitigation of cumulative blast impacts The key management measure for the mitigation of cumulative blast impacts will be scheduling of blasts to ensure each mine fires their blast at a separate times. Processes to mitigate blasting impacts associated with operations will be addressed in each mines' individual BLMPs. Each mine has or will develop a BLMP that outlines a consistent approach for the scheduling of blasts in consultation with other mines in the BTM Complex.	See BMP, sites share blasting schedules and blast at different times of the day reducing the risk of cumulative impact.				
Blast Management Strategy - January 2013	6.1	At least 24 hours' notice will be provided prior to a proposed blast. If there is no conflict regarding the scheduled blast times, there will be no further correspondence. If there are conflicting blast times between the mines, a revised schedule for firing the blasts will be agreed upon. The schedule will be developed to ensure blasts are fired with a considerable time gap between them to reduce any potential cumulative impacts.	This occurs see Blast MP	Compliant			
	6.2 Communication	To 1 11 11 11 11 11 11 11 11 11 11 11 11	Lucia de la compansión de		, , , , , , , , , , , , , , , , , , ,		
Blast Management Strategy - January 2013	6.2	Regular meetings will be held by the BTM Complex to discuss monitoring results and future operational events. Meeting minutes will be documented and distributed to each site.	this occurs, confirmed in inertview with EO	Compliant			
Blast Management Strategy - January 2013	6.2	When blasting criteria are identified as exceeded, discussions will be held within the BTM Complex and the agencies and affected landholders (where an exceedance occurs on privately-owned land).	No exceedences have occurred so this cannot yet be tested.	Not Applicable			
Blast Management Strategy - January 2013	6.2	The mines of the BTM Complex will also, if required, share baseline property inspection reports that are completed at the request of neighbouring landholders, in accordance with each site's Project Approval. Process to identify main source of blasting impacts	No inspection reports have been completed (or requested)	Not Applicable			





D. C		Requirement	Evidence	a drama drama		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Blast Management Strategy - January 2013	6.2	If there is uncertainty around the source of a blasting related incident (e.g. exceedance of assessment criteria or damage to a neighbouring building or other infrastructure), a meeting will be held by the BTM Complex representatives to review relevant data and investigate the cause of the incident. If the cause cannot be determined, then the BTM Complex will engage a suitably qualified expert to undertake an independent blast impact investigation. The outcomes of the investigation will help determine the responsibility of the mines for any corrective actions.	Noted				
	6.3 Blasting related in	cidents					
Blast Management Strategy - January 2013	6.3	Blasting related incidents such as misfires or exceedances of assessment criteria will be reported and managed in accordance with each mines' BLMP and incident management process. Incidents will be managed in accordance with the requirements of the Protection of the Environment Operations Act 1997, Coal Mine Health and Safety Act 2002 and Coal Mine Health and Safety Act Regulation 2006.	Noted				
	6.4 Reporting						
Blast Management Strategy - January 2013	6.4	Management reports will be prepared regularly, noting performance against criteria. External reporting will include: - individual Company websites - Community Consultative Committees (CCCs) - Annual Environmental Management Reports (AEMRs) - annual returns - exceedance reporting.	See BMP	Compliant			
	7 Corrective and preven						
	7.1 Blasting criteria ex						_
Blast Management Strategy - January 2013	7.1	If the monitoring results of a blast identify an exceedance of the mines relevant criteria, written notification of the exceedance will be provided to the other mines within the BTM complex, in addition to any investigation undertaken according to the respective mine's BLMP.	Noted				
	8 Document control						
Blast Management Strategy - January 2013	8	The BLMS has been developed with the input of representatives of BCPL, TCM and MCC.	Noted				
	8.1 Review and revision						
Blast Management Strategy - January 2013	8.1	The BLMS will be reviewed and revised at least every two years or on an 'as required' basis to incorporate improvements identified by the BTM Complex or appropriate requirements of government agencies. It will be the collective responsibility of the BTM Complex to review the BLMS.	Plan is not 2 years old yet, not required	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference		·	Evidence	Addit rillaing	Consequence	Likelihood	Risk
	WHC_PLN_TAR_Blast 2 Pre-blasting Propert	<u> </u>					
Blast Monitoring Program	2	As a requirement of Conditions 4(20) and 4(21) of DA 88-4-2005, TCPL advised all landowners within a 2km radius of the planned blasting operations on the mine site, and additional landowners as advised by the Director-General, that they were entitled to a property inspection, at least 2 months prior to any blasting. Figure 2 depicts land ownership and residences around the Project Site.	Noted, the property inspections occurred prior to the audit period and there have been no new requests or direction from the DG in the audit period.	Not Applicable			
Blast Monitoring Program	2	In the event a written request is made by any of the private landowners / residents within 2km, or any other landowner nominated b the Director-General, TCPL will, within 2 months of receiving the request: • commission a suitably qualified person, whose appointment has been approved by the Director-General, to inspect the condition of any building or structure on the land, and recommended measures to mitigate any potential blasting impacts; and • give the landowner a copy of this property inspection report.	No such requests	Not Applicable			
	3 Pre-Blasting Notifica						
Blast Monitoring Program	3	In addition to notification to each landowner regarding their entitlement to a property inspection, TCPL will provide written notification, at least 2 weeks prior to the commencement of blasting, on the following. • The company's proposed blasting timetable and duration. • A plan showing Blast Monitoring Locations. • The procedure for reporting possible blasting related damage to a residence or other vibration infrastructure (such as water supply or underground irrigation mains), and the entitlement of any landowner within a 2km of proposed blasting activities, to an investigation of the damage claims. • The disputes resolution procedure of TCPL.	Noted - This occurred prior to the audit period.	Not Applicable			
Blast Monitoring Program	3	In addition, TCPL will provide specific notification of individual blasts to any person who registers an interest in being notified about the blasting schedule at the mine. Notification will involve: • a letter, nominating a planned blast date, will be hand delivered on the day before each blast. The letter would provide the indicative time for the blast; and • a telephone call during the morning of the blast to confirm the blast will proceed at the nominated (or varied) time.	Sighted phonecall list plus e-mail list, these are recorded with the preblast enviro checklist on the network.	Compliant			
Blast Monitoring Program	3	TCPL will also provide the adjacent Boggabri Coal Mine with a copy of its planned blasting schedule on a weekly basis to provide at least one weeks notice of blasts planned within the open cut area.	Done and they are on the blast e-mail list.	Compliant			
	4 Blast Monitoring Lo	Air-blast overpressure and ground vibration will be measured at	Orica are no the Blast contractor, TCL now		T T		
Blast Monitoring Program	4	these monitoring locations for all blasts throughout the remaining life of the mine by Orica as the Blasting Contractor. Results of the monitoring will be forwarded to TCPL on the day of each blast. The three nominated blast monitoring locations (referred to as TB-1 and TB-2) are shown on Figure 2.	use there own drill and blast team. Results are recieived by SMS	Compliant			





Deference	Clause	Poguiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Blast Monitoring Program	4	It should be noted that monitoring is not proposed on project related residences, except for "Templemore" (TB-2). However, in the event that the occupier of another project related resident lodges a complaint in relation to blasting, or requests monitoring at the residence, an additional monitor will be deployed.	Noted, the mobile monitor was located at Callendar for two blasts and did not trigger on both occasions	Compliant			
Blast Monitoring Program	4	In addition to the nominated sites, TCPL will also: • observe flyrock/blast rock distribution for each blast; and • video each blast to assist in the interpretation of results.	This occurs	Compliant			
	5 Blast Monitors						
Blast Monitoring Program	5	Only calibrated monitors will be used for blast monitoring with copies of calibration certificates or other means of verification available at the Tarrawonga Coal Mine. The date of last calibration is automatically printed on each monitor print-out.	Certificates provided as evidence	Compliant			
	6 Procedures	is determined by each member print out.					
	6.1 Noise Monitoring	g Procedures					
Blast Monitoring Program	6.1	All aspects of blast monitoring will be undertaken by Orica Explosives in accordance with AS 2187.2-1993- Storage, Transport and Use – Use of Explosives, a copy of which will be retained at the Tarrawonga Coal Mine Site Office.	See the Explosives Management plan and Explsives Security plan.	Compliant			
	6.2 Blast Impact Asse						
Blast Monitoring Program	6.2	In accordance with Condition 4(13) and 4(14) of DA 88-4-2005, the air-blast overpressure and ground vibration impact assessment criteria will be as presented in Tables 2 and 3. Table 2 Air-blast Overpressure Level (dB(Lin Peak)) All-blast Overpressure Level (dB(Lin Peak)) All of the total number of blasts in a 12 month period 15 between cal-off frequency of 21 kc or less if the instrumentation has a higher cal-off frequency according to the measurements are performed with equipment having a lower cal-off frequency of 21 kc or less if the instrumentation has a higher cal-off frequency according to the measurements are performed with equipment having a lower cal-off frequency of 21 kc or less if the instrumentation has a higher cal-off frequency according to the should not be used. Table 3 Ground Vibration Impact Assessment Criteria Peak particle velocity (mm/s) Allowable exceedance 5 5% of the total number of blasts in a 12 month period 10 0%	Noted, covered elsewhere	Compliant			
	6.3 Trigger Levels		<u> </u>		<u> </u>		
Blast Monitoring Program	6.3	In recognition of the distance between the proposed blast locations and local residences to be monitored, far field trigger levels of between 0.3mm/s and 0.4mm/s (for ground vibration) and 111.7dBL to 111.9dBL (for air-blast overpressure) have been adopted for the Tarrawonga Coal Mine.	noted				
	6.4 Pre-Check						





Reference	Clause	Poquiroment	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Finding	Consequence	Likelihood	Risk
Blast Monitoring Program	6.4	Prior to monitors being placed in the field, the following aspects will be verified for each instrument. (i) Battery is charged. Note: batteries will be placed on charge immediately following data downloading from each blast. (ii) Date and time are correct. (iii) Location (Site) for each monitor is marked on the carry case. (iv) Instrument "fields" are correctly set to reflect the distance from the blast site to identified monitor location. The instrument fields, i.e. near, medium and far, determine the period of recording for air-blast once the monitor is triggered on either ground vibration or air-blast itself. (v) Instrument trigger levels are set to minimize the potential for false initiation of the recording sequence by, for example, wind, but will record air-blast or ground vibration events approaching or greater than the standard vibration events approaching or greater than the standard vibration criteria of 115dBL and 5mm/s respectively. For the far field setting, triggers would typically be set between 0.3 and 0.4mm/s and 111.7 and 111.9dBL. (vi) Setting adjustment as necessary. (vii) Print off a confirmation of settings from the logger.	Not relevant as Datamaster system now used	Not Applicable			
Blast Monitoring Program	6.4	In order to ensure consistency, a sheet identifying the monitor settings for each blast monitoring site will be retained at the site office and adjusted as necessary to reflect, for example, increasing or decreasing distances to the blast site and monitoring results, e.g. false trigger frequency.	Not relevant as Datamaster system now used	Not Applicable			
	6.5 Field Set-Up						
Blast Monitoring Program	6.5	The instrument will be set up at the pre-selected location at each monitoring point, between 3.5m and 30m from the residence or building to be monitored. The instrument set-up procedures involve the following steps. (i) Insert the soil spike into the ground and level the geophone. (ii) Set up microphone. (iii) Connect microphone and geophone to the monitor. (iv) Turn power on. Powering up the monitor initiates a self check culminating in a VDU advice that the instrument is functioning properly. (v) Press "start". Following a countdown sequence, the monitor moves into a "standby mode" awaiting triggering.	Not relevant as Datamaster system now used	Not Applicable			
Blast Monitoring Program	6.5	Prior to initiation, a mine Site Blast Check List form which includes verification that the monitors are in place and records salient weather data, e.g. wind direction and cloud cover, will be completed and signed by the shot-firer.	Not relevant as Datamaster system now used	Not Applicable			
	6.6 Post Blast - In the			,			
Blast Monitoring Program	6.6	Following the completion of each blast, the following activities will be undertaken prior to the monitor being returned to the office for data downloading. (i) Press "stop" button. (ii) Turn power off. (iii) Disconnect microphone and geophone (iv) Remove soil spike.	Not relevant as Datamaster system now used	Not Applicable			
	6.7 Post Blast - At the	(v) Pack instrument up.		!	<u> </u>		





Doforence	Claves	Dominion	Evidones	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Blast Monitoring Program	6.7	On the return of each monitor to the office, the blasting contractor will: (i) retrieve / download the data to the office computer; (ii) review the data and delete any data pertaining to false triggers, ie. triggers before the blast initiation time; (iii) generate a results print-out sheet (in Microsoft Word) and insert relevant data relating to the blast, e.g. blast pattern, hole spacing, number of rows, number of holes, blast-hole diameter, stemming, MIC, explosives type and weight, delay type (interval and duration (ms)) and any relevant comments or observations. An example of a typical results sheet from the Tarrawonga Coal Mine is attached as Appendix 2; (iv) print off and distribute the results to the nominated recipients. Prior to the commencement of blasting activities, the owner of each building where monitoring is undertaken, will be asked if they would like to receive copies of the relevant blast results. Result distribution may be by email, fax or in hardcopy as appropriate or requested. Copies of the printouts, the mine site blast checklist and details such as blast design, charging and tie-in pattern are retained on the mine site files; (v) transfer the data for the blast to an Excel spreadsheet; and	Not relevant as Datamaster system now used	Not Applicable			
Blast Monitoring Program	6.7	(vi) place monitor battery on charge to await the next blast. (vii) In the event of an exceedance of blast criteria, or if a complaint is received, the blast contractor, in consultation with the Environmental Officer, will review video footage of the blast to assist in determination of causes or conditions which may have contributed to the exceedance.	Not relevant as Datamaster system now used	Not Applicable			
	7 Flyrock Distribution		Configured at interview DOD and	T	T I		
Blast Monitoring Program	7	Following each blast, the area surrounding the blast site will be inspected and flyrock distribution to the front, rear and both sides of the blast site observed.	Confirmed at interview D&B eng	Compliant			
	8 Blast Criteria Excee	dance Management Reporting					
Blast Monitoring Program	8	In the event that the monitoring results of a blast identify an excedance of: • peak vector sum velocity (ground vibration) – 5mm/s (ppv); and/or • peak overpressure – 115dBL,					
Blast Monitoring Program	8	WCMPL, as the holder of an Environment Protection Licence under the Protection of the Environment Operations Act 1997, will report the incident to DECC (EPA) Armidale (6773 7000) and initiate investigations as to the cause. In accordance with the PA 06_0198, Schedule 4, Conditions 3, 4 and 5, any excedance in blast criteria will also result in notification to the Department of Planning (DoP) and DECC within 24 hours of the occurrence of the excedance. Additionally, within 6 days of the notification, WCMPL will provide a written report to both DoP and DECC identifying the date, time and nature of the excedance, the cause or likely cause of the excedance, the action that has been taken to date relating to the excedance, and proposed measures to minimise potential for excedance in future blasts. WCMPL will also report any excedance in blasting criteria in the Annual Environmental Management Report (AEMR).		Not Applicable			





Reference	Clause	Doguiroment	Evidence	Audit Finding		Risk	
Kererence	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Blast Monitoring Program	8	It is noted that the above criteria are able to be exceeded for up to 5% of blasts in any one year, but not to exceed a: • peak vector sum velocity (ground vibration) – 10mm/s (ppv); and/or • peak overpressure – 120dBL. In the event any blast causes these criteria to be exceeded, the DECC and DoP will similarly be notified in accordance with the requirements identified above.	5% not exceeded, no 5-10mm/s and 2 in 115-120dBA both investigated even though they were on company owned land.	Compliant			
	9 Blast Complaint Ma	nagement Procedures					
Blast Monitoring Program	9	Any general complaint received relating to any blast will be managed in accordance with the complaints receipt and response procedure presented in the Tarrawonga Coal Mine Environmental Management Strategy and summarised below. (i) Details of the complainant and complaint will be recorded in the complaints log. (ii) TCPL representatives will liaise with the complainant to ascertain all details, to identify the nature and source of the complaint and provide supplementary information for the log. (iii) Investigations will be initiated to verify or otherwise the basis for the complaint. (iv) Results of the investigation will be provided to the complainant together with advice as to any changed blast management practices to be implemented as a consequence of the investigation.	The complaints log supports this	Compliant			
Blast Monitoring Program	9	Complaints pertaining to blast-related damage will be managed in accordance with Condition 4(22) of DA 88-4-2005	No complaints in the audit period regarding damage associated with blasting, only loss of amenity	Not Applicable			





Reference	Clause	Requirement	Evidence	A		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	WHC_PLN_TAR_Road	Closure Management Plan					
oad Closure Management Plan	NA	Ensure the public are well informed of any temporary road closures on Goonbri Road and Dripping Rock Road due to blasting at Tarrawonga and ensure minimal disturbance to traffic during blasting periods.	Noted, no road closures in the audit period to test.	Not Applicable			
oad Closure Management Plan	NA	Tarrawonga neighbours Those neighbours who have requested to be notified of blasts including the closure of local roads, will be advised by telephone on the day of the blast by the Environmental Officer or Site Clerk for Tarrawonga Coal.	Closure list of neighbours similar to blast list are contacted (note no road closures in the audit period)	Compliant			
oad Closure Management Plan	NA	Unless specifically requested, NSC will not receive advice for every road closure.	Noted				
oad Closure Management Plan	NA	All road closures will be performed, where possible, at a time to minimise impact on road users and the local school bus run. Blasting is not permitted before 9am which will avoid peak traffic times in the morning. Whitehaven will avoid blasting during the peak afternoon traffic time between 3:30pm and 5:30pm but reserves the right to blast until 5pm, in accordance with the Project Approval.	Blasting is generally between 11 and 2.	Compliant			
oad Closure Management Plan	NA	Where possible, blasts will be scheduled to occur at approximately 12 noon to maintain consistency of blast times.	Blasting is generally between 11 and 2.	Compliant			
oad Closure Management Plan	NA	Emergency Services Should any emergency services approach a road closure point and be required to pass to attend an emergency, the traffic controller will arrange for the emergency service vehicles to be escorted through the road closure station as soon as the route can be made safe. This may include contacting the blast controller to temporarily delay the firing of the blast.	Noted				
oad Closure Management Plan	NA	Traffic Control Traffic control points will be established on the affected road. These locations will remain relatively constant, however may require some modification due to prevailing conditions on the day of blasting. At a minimum a control point will be established at the intersection of the Goonbri Road and the Tarrawonga Haul Road and at the intersection of Goonbri Road and Dripping Rock Road. Control points will be revised as the pit progresses in an easterly direction. Any changes will be undertaken in consultation with NSC.	Noted				
oad Closure Management Plan	NA	As a minimum, the traffic control points will include one sentry for the purpose of traffic control, together with all equipment necessary for the safe control of the road. All sentries will be in two-way contact with the blast controllers.	Noted				
oad Closure Management Plan	NA	All road sentries will travel to the road closure points and open all necessary signage advising of the temporary road closure due to blasting. Once all signage is positioned the sentries will await direction from the blast controller.	Noted				
oad Closure Management Plan	NA	Traffic control point signage will consist of permanent fold out signs situated adjacent to the road.	Noted				





Defenses	Clause	Requirement	Evidence	Andte Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Road Closure Management Plan	NA	Once the blast is ready to be fired, the traffic controllers will be directed to close the road. The road sentry shall drive the entire route to ensure all vehicles are clear of all road closure points. Once the area has been confirmed to be clear of vehicles the road sentry will notify the blast controller and the blast will be fired in accordance with the site procedure for blasting. At the completion of the blast, the "all clear" is given by the Shotfirer.	Noted				
Road Closure Management Plan	NA	All necessary signage will be closed at the conclusion of the road closure and be padlocked shut by the road sentry.	Noted				
Road Closure Management Plan	NA	Blasting frequency and requirements for road closure will vary with mine development. There will be no more than one blast per day, unless in the case of a misfire, and a maximum of four blasts per week, averaged over 12 months. Where road closure is required, it is expected that the road will be closed for approximately 15 minutes for each blast.	Noted, blast frequency complies with this				
Road Closure Management Plan	NA	Incident Reporting Incident reporting as per procedures identified in the Environmental Management Strategy for the Site.	No incidents	Not Applicable			
Road Closure Management Plan	NA	Complaints Management Complaints management as per procedures identified in the Environmental Management Strategy.	No incidents	Not Applicable			
Road Closure Management Plan	NA	Incident Response – Exceedance or Complaint Investigate event and identify operational location, timing and climatic conditions to determine if additional management measures are required.	No incidents	Not Applicable			
Road Closure Management Plan	NA	Any incidents will be reported in the Annual Environmental Management Report/Annual Review	No incidents	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
		uality & Greenhouse Gas Management Plan					
	•	and Management Procedures					
	Vegetation Clearing a						
Air Quality & Greenhouse Gas		Cleared trees and branches are retained where possible for the	This done as sighted on site				
Management Plan	3	use in stabilising slopes identified for rehabilitation with native		Compliant			
Wanagement Han		woodland communities.					
Air Quality & Greenhouse Gas		Where practicable, soil stripping is undertaken at a time when	Soil stripping procedure.				
Management Plan	3	there is sufficient soil moisture to prevent significant dust lift-off.		Compliant			
-							
Air Quality & Greenhouse Gas	3	Stripping soil is avoided in periods of high winds.	Soil stripping procedure.	Compliant			
Management Plan	-						
Air Quality & Greenhouse Gas	_	Dust suppression by water application is used to increase soil	Soil stripping procedure.				
Management Plan	3	moisture if stripping occurs during periods of high wind or low soil		Compliant			
		moisture.					
		Vegetation clearance and soil stripping will be confined to an	Mine planning, Soil stripping procedure,				
Air Quality & Greenhouse Gas		area sufficient for the following 12 months mining activities. This	Biodiversity Management Plan.				
Management Plan	3	will minimise impact on hibernating/roosting/nesting fauna, and		Compliant			
J		minimise the areas exposed to dust lift off as much as practically					
	D :II: 1 D1 1: A	possible.					
	Drilling and Blasting A		Water interstan			1	
Air Quality & Greenhouse Gas	3	Drill rigs utilise water injection or alternatively, are fitted with	Water injection	Compliant			
Management Plan		dust collectors.	Ciabtod				
Air Quality & Greenhouse Gas Management Plan	3	Blast hole stemming is used to prevent venting of explosion gases.	Sighted	Compliant			
ivialiagement Flan		Blasting is conducted both before the establishment, and after	Blast assessment enviro checklist				
Air Quality & Greenhouse Gas	_	the break-up of low-level atmospheric temperature inversions.	blast assessment enviro enceklist				
Management Plan	3	the break up of low lever atmospheric temperature inversions.		Compliant			
-							
		Blasting will be subject to a regular assessment of weather	Blast assessment enviro checklist				
Air Quality & Greenhouse Gas	3	conditions prior to and on the day of a blast, to ensure that wind		Compliant			
Management Plan	J	speed and direction will not result in significant dust emissions or		Compilant			
		blast fume impacts on neighbouring residences.					
Air Quality & Greenhouse Gas		Review of geological conditions to inform blast design;	Geology considered in design including				
Management Plan	3		faults and anomolies and general strata	Compliant			
e e			information				
Air Quality & Greenhouse Gas	3	Minimise the time between drilling, loading and shooting the	Noted				
Management Plan		blast	N				
		• The following factors contributing to non-ideal detonation	Noted				
		behaviour and higher emission (principally NO2) concentrations					
		are avoided whenever possible.					
Air Quality & Greenhouse Gas	2	o weak overburden which reduces the necessary explosive					
Management Plan	3	confinement is ripped in preference to blasting.					
		o water infiltration.					
		o long explosive columns.					
		o explosive pre-compression, caused by hole-to-hole shock					
		propagation due to wet overburden and clay veins.	<u> </u>	<u> </u>			
	Overhurden Dinning						
Air Quality & Greenhouse Gas	Overburden Ripping a	Ripping of softer overburden material is avoided during periods	OCE discussion		I		





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	3	Low moisture coal is sprayed with water prior to excavation to raise moisture content to > 5.5%.	OCE discussion	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Use predictive meteorological forecasting and predictive air dispersion modelling, together with real time monitoring data to inform operational practices in advance of the commencement of shift. This will aid in the day to day planning of mining activities giving due consideration to prevailing weather conditions and potential for dust generation and dispersion.		Not Compliant	D	2	Medium
	Internal Road and Ha	rdstand Area Construction					
Air Quality & Greenhouse Gas Management Plan	3	Clearing ahead of construction activities is minimised.	All clearing is minimised, construction activities minimal apart from some road construction for the northern box cut.	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Cleared areas are watered regularly during any construction activities, where appropriate.	Noted				
Air Quality & Greenhouse Gas Management Plan	3	Unsealed roads used regularly for access will be watered utilising site water carts to minimise dust generation. Where necessary additional water applications will be made, and/or chemical dust suppressants will be used to minimise dust lift off from unsealed roads.	Dust suppressants trialled, not widely used, water cart is main dust suppression equipment.	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Mine vehicles and coal haulage trucks will be subject to speed limits in place at the site. Speed limits are 60km/hr on the access road and in pit area, 10km/hr at the workshop area and 30km/hr at the ROM and load-out bin. At all times, operators are to drive to conditions and obey speed restriction signs.	Speed limit signage comprehensive, OCE manages open cut, workshop manager manages hardstand etc and ROM is also managed by the OCE.	Compliant			
	Coal Processing Area	·					
Air Quality & Greenhouse Gas Management Plan	3	Water is applied to the coal at the feed hopper, crusher and at all conveyor transfer and discharge points.	Water injection at crusher and on loader and conveyors have water sprays fitted.	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	All conveyors are fitted with appropriate cleaning and collection devices to minimise the amount of material falling from the return of conveyor belts.	Conveyors did not exhibit excessive accumulations of spilt coal nor did they appear to be creating excessive particulates	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Some flexibility exists to temporarily cease operation in the event of protracted dry periods, high winds, or significant dust generation and dispersal towards the surrounding residences.	noted				
Air Quality & Greenhouse Gas Management Plan	3	Trucks transporting coal offsite from the Coal Processing Area must be covered immediately after loading to prevent windblown emissions and spillage. The covering must be maintained until immediately before unloading the trucks (as per Condition O3.2 of EPL 12365).	Observed on site and confirmed at interview.	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	In dry windy conditions, trucks on the loading bin loop will be speed limited to 10km/hr to reduce potential for dust lift off.	Trucks are GPS tracked and under instruction regarding relevant speeds under different conditions	Compliant			
	Wind Erosion Manag	ement					
Air Quality & Greenhouse Gas Management Plan	3	The extent of clearing/site preparation in advance of mining is minimised.	Checked above	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	 Progressive rehabilitation of areas of disturbance, including topsoil and subsoil stockpiles is undertaken. 	site visit confirms this	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk		
Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	3	Bund walls and windbreaks are constructed as required.	No evidence of wind erosion (exception southern emplacement and trees or bunds will not protect this area), some bund walls and tree wind breaks around the site.	Compliant			
	Internal Transport				•	•	
Air Quality & Greenhouse Gas Management Plan	3	The road for the transportation of coal product between the mine facilities area and mine entrance is sealed.	Observed on-site	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	 Unsealed roads regularly used on the mine site will be watered utilising site water carts to minimise dust generation. 	Observed on-site	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Dust suppressants may be used from time to time to reduce dust and particulate generation, especially during excessive dry weather periods or when site water levels are low or restricted.	Trialled but not used extensively, may be used more extensively if water availability becomes critical.	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	All roads are speed limited and enforced to ensure that dust generation is at acceptable levels.	All roads limited	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Earthmoving equipment and on-site vehicles: o are fitted with exhaust controls which satisfy NSW EPA emission requirements; o are properly maintained and any mobile equipment which does not comply with NSW EPA guidelines is removed; and o have the exhausts directed upwards or to the side (where applicable) so as not to cause dust lift-off.	Discussed with Maintenance team, 103 inspections and monthly or more frequent equipment inspections	Compliant			
air Quality & Greenhouse Gas Management Plan	3	The controls and management procedures will be reviewed in response to the results of air quality monitoring, complaints or comments identified through TCPL's consultation effort. Any changes made will be noted as part of annual environmental reporting in the AEMR/Annual Review.	No requirement for review based on the parameters listed.	Not Applicable			
	Pollution Reduction	Programs					
Air Quality & Greenhouse Gas Management Plan	3	Of relevance to the overall objective of minimising dust emissions from the Tarrawonga site, EPL 12365 has some Pollution Reduction Program (PRP) measures that the site is implementing. These include: • Wheel generated dust control efficiencies on active haul roads	Provided to EPA and PRP adapted in the EPL - 89% 94% efficiency compared to untreated	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Assessment and implementation of best practice measures to minimise dust from disturbing and handling overburden in adverse weather conditions. Appendix 2 provides the Daily Dust Inspection which includes triggers for adverse weather conditions and the response taken by site to minimise dust impacts.	Provided to EPA and PRP adapted in the EPL 15-08-14	Compliant			
Air Quality & Greenhouse Gas Management Plan	3	Participation in an Australian Coal Association Research Project (ACARP) study to trial best practice measures for disturbing and handling overburden.	No evidence Provided	Not Compliant Administrative			
Air Quality & Greenhouse Gas Management Plan	3	Where relevant, appropriate revisions/updates will be made to this management plan following the outcomes of the PRPs nominated above.	No changes to AQGGMP, the EPA signoff of the PRPs occurred in August, the onth before the audit took place	Not Applicable			
	4 Monitoring and Re	. •					
	4.1 Risk/Response N	Matrix					





Defenses	Classa	Requirement	Evidence	A codita filo di con		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	4.1	A risk/response matrix has been developed for the site to identify operational practices or weather conditions that require specific action to mitigate potential impacts. The intent behind the risk/response matrix is to understand the prevailing conditions and implement management practices accordingly to avoid subsequent air quality impacts. The risk/response matrix has been developed based on colour coding and key operational and weather conditions, and proximity of privately owned receptors to mining operations. The Environmental Officer will be responsible for implementation of the risk/response matrix in consultation with the Operations Manager.		Compliant			
	4.2 Monitoring Progra						
	4.2.1 Paramaters Mea						
Air Quality & Greenhouse Gas Management Plan	4.2.1	Activities on the mine site will emit dust in various forms, namely total suspended particulate matter (TSP), particulate matter with aerodynamic diameters less than 10μm (PM10), particulate matter with aerodynamic diameters less than 2.5μm (PM2.5), and deposited dust (which is assessed as insoluble solids as defined in Australian Standard AS 3580.10.1-2003 Methods for Sampling and Analysis of Ambient Air - Determination of Particulates - Deposited Matter - Gravimetric Method).	Noted, Dust gauges, HVAS with PM10 head and TEOM for use as an operational management tool	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.2.1	Monitoring implemented to ensure compliance with these criteria will include monitoring of dust deposition rates as well as PM10 and PM2.5 concentrations at nominated residences surrounding the mine site. The location of monitoring points is shown on Figure 2. As PM10 and PM2.5 is being monitored continuously at nominated TEOM sites it is not intended to undertake monitoring by High Volume Air Samplers at 6 day cycles. No monitoring of TSP is proposed as PM10 concentrations are considered of greater significance given its synergies with health-related issues. This approach was accepted by DPE for all Whitehaven sites in a letter dated 5th August 2011.	Monitoring is conducted as per the AQ Management Plan.	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.2.1	In addition to these monitoring sites, Tarrawonga will implement, in consultation with the other mining operations in the precinct and in accordance with the Air Quality Management Strategy for the BTM Complex, a predictive meteorological forecasting tool and dispersion model, to inform mine planning on a day to day basis, giving due consideration to the potential cumulative impacts of all mining operations in the area. Additional monitoring points to inform cumulative air quality management will also be established.	This is not yet done, Strategy is not approved.	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	4.2.1	A record of site activities undertaken and meteorological records during the period of recording will also be retained. This will initially be undertaken in accordance with the Daily Dust Inspection (Appendix 2) and updated once the cumulative network is implemented.	Evidence provided	Compliant			
	4.2.2 Monitoring Lo	cations					
Air Quality & Greenhouse Gas Management Plan	4.2.2	Regional monitoring of PM2.5 concentrations will also be undertaken via a real time monitor located at the "Will-gai" Property, approximately 10km from the Tarrawonga mine.	Conducted, shared resource with Vickery	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.2.2	The real time monitors are integrated with existing real time noise monitoring equipment and allow air quality results to be viewed from the site office in real time. This system also provides for site alerts in the event that air quality measurements approach compliance thresholds. The monitoring locations identified include mine owned land, that may, from time to time, be occupied under lease. Any lessees will be informed of any potential air quality impacts at the residence, and provided with any monitoring data as it comes available in accordance with the requirements of Condition 3(27).	Sentinex system, Sighted	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.2.2	Not withstanding the above monitoring activities specific to Tarrawonga operations, under the terms of the Air Quality Management Strategy for the BTM Complex, it is intended to install additional E-Sampler (or similar) monitors in proximity to operations at Maules Creek, Boggabri and Tarrawonga. Data from these monitors will be used to inform performance against criteria, and the capacity to identify source contribution to enable active management. The specific detail around activation of these E-Samplers will be included in the BTM Complex Plan once a supplier has been confirmed.		Compliant			
	4.2.3 Monitoring Fre	equency					
Air Quality & Greenhouse Gas Management Plan	4.2.3	The monitoring frequency for deposited dust, PM10 and PM2.5 is as follows: • Deposited dust – continuous • PM10 – continuous (TEOM) • PM2.5 – continuous (TEOM)	Noted				
	4.3 Monitoring Proc	cedures, Data Recording and Reporting					
	4.3.1 Monitoring Pro	ocedures					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	4.3.1	Monitoring will be undertaken according to the DEC (2006) document Approved Methods for the Sampling and Analysis of Air Pollutants in New South Wales 2006. Specifically, monitoring will be conducted in accordance with the following Australian Standards: • AS/NZS 3580.1.1:2007 "Methods for sampling and analysis of ambient air – Guide to siting air monitoring equipment". • AS / NZS 3580.9.6 – 2003 "Methods for sampling and analysis of ambient air – Determination of suspended particulate matter PM10 high volume sampler with size—selective inlet – Gravimetric Method". • AS 3580.10.1-2003 "Methods for Sampling and Analysis of Ambient Air - Determination of Particulates - Deposited Matter - Gravimetric Method" (NSW DEC Method AM-19). • AS 3580.9.8-2001 "Methods for the Sampling and Analysis of Ambient Air (NSW DEC Method AM-22).	compliance, note the equipment reviewed at site appeared to be sited in accordance with the AS.	Compliant			
	4.3.2 Data Recording		, 				
Air Quality & Greenhouse Gas Management Plan	4.3.2	For each deposited dust monitoring location, once each month the glass container used to capture the deposited dust will be removed, replaced and sent to a NATA accredited laboratory for analysis.	NATA certification noted in laboratory reports Note the laboratory report provided as evidence notes that the results could not be certified as they claimed TCL had collected the samples therefore the exposure time could not be verified, but the samples were collected by ALS staff who should have been able to verify exposure times from their records.	Not Compliant Administrative			
Air Quality & Greenhouse Gas Management Plan	4.3.2	Condition M1.3 of EPL 12365 requires the following records to be kept: • The date(s) on which the sample was taken; • The time(s) at which the sample was collected; • The point at which the sample was taken; and • The name of the person who collected the sample.	Records comply	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.2	In addition to these requirements, any notable activities or conditions at or around the monitoring location should be noted at the time of sample collection. Site activities that could impact on air quality results as well as any relevant regional conditions (eg. bushfires, dust storms) should be noted when they occur.	Field sheet includes this content	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	4.3.2	Data obtained from the real-time air quality monitoring network will be available in real time via access to the web interface, as well as through daily reporting from the monitoring unit to selected personnel. Key personnel for the provision of daily reports will be the Group Environment Manager, the Operations Manager for the Tarrawonga site and the Environmental Officer. In addition to daily reports, the monitor will be downloaded a regular intervals with data available on an archival basis. Real time data will be maintained on an archival basis at a central repository that can be retrieved on an as needs basis.	Observed onsite, web information is provided to EO, Ops M and Genv Mng plus the real time data is provide dto OCEs for operations management in real time.	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.2	Upon establishment of the BTM Complex monitoring network, data from the additional Esamplers (or similar) and other monitoring equipment will be accessible via the same methodology as proposed for the Tarrawonga monitoring program.	Noted, not yet implemented	Not Applicable			
	4.3.3 Data Reporting	, · · ·					
Air Quality & Greenhouse Gas Management Plan	4.3.3	A summary of air quality monitoring results will be reported internally on a monthly basis as well as on a quarterly basis to the Community Consultative Committee (CCC) via the Environment Monitoring Report. This report will be uploaded onto the company's website (www.whitehavencoal.com.au).	See CCC reports/minutes	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.3	Air quality monitoring required by the EPL must be published on the Whitehaven website within 14 days of receiving the results.	Results go up within 14 days	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.3	Each year, the results of the air quality monitoring program will be summarised and presented in the AEMR/Annual Review together with reference to the prevailing meteorological data and site activities during the measurement period(s). Reporting will also include an analysis of the monitoring results against the exceedance criteria, previous monitoring results and predictions made in the EA.	·	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.3	Dust monitoring results will be issued to EPA via the Annual Return for EPL 12365 for licenced monitoring points.	Included in the AR	Compliant			
Air Quality & Greenhouse Gas Management Plan	4.3.3	In addition to the reporting requirements listed above, air quality monitoring data will be made available to the public upon request. The real time air quality monitoring data, in the form of the 24hr average PM10 concentration for the preceding period, daily weather forecasts for the week ahead, and operational responses to forecasts and monitoring data for the preceding period will also be made available on the Whitehaven Coal website on a daily basis, in accordance with Project Approval requirements. It should be noted that the 24hr average PM10 concentration result is from all sources and not wholly related to mining activities, and therefore not validated to confirm mine related contributions.		Not Compliant Administrative			
	4.4 Cumulative Air Qu	Ll uality and GHG Management Strategy					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Neierence	Ciause			Addit Filluling	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	4.4	TCM has been in consultation with the nearby Boggabri Coal Mine and Maules Creek Project in developing the Air Quality Management Strategy for the BTM Complex in order to minimise cumulative impacts on air quality. The Strategy has been submitted to DPE for approval and will be appended to this Plan once approved.	Noted, not yet approved.				
Air Quality & Greenhouse Gas Management Plan	4.4	TCM and Boggabri Coal currently continue to operate shared air quality monitors, whilst Boggabri Coal, Tarrawonga and the Maules Creek Project participate in monthly meetings to discuss cumulative impact management measures.	Noted				
	4.5 Community Const	ultative Committee					
Air Quality & Greenhouse Gas Management Plan	4.5	As part of the quarterly Community Consultative Committee meeting, an environmental monitoring report is presented to the members with detailed explanation of the air quality monitoring results for the corresponding quarterly period. Dust deposition results and PM10 monitoring are summarised by month with an annual average also provided alongside the compliance criteria. Any exceedance is clearly explained as to the location and type of exceedance, as well a discussion on any queries or actions from the members of the committee.	CCC minutes	Compliant			
	5 Management of Exc	ceedances, Complaints and Non-Compliance					
		liance Criteria Exceedance					
	1 Confirmation of Exc						
Air Quality & Greenhouse Gas Management Plan	5.1	For deposited dust results, the analysing laboratory will be contacted to ensure no error has been made in storing, analysing or recording the sample or result. Should this investigation conclude the treatment, analysis and result recording for the sample are satisfactory, TCPL will proceed to response point 2.	Minor exceedances generally explained by bottle location and wind direction for the monitoirng period, no necessity for checking laboratory. Lab does run QA on all analyses.	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.1	In terms of the triggering of alerts at site in relation to dust levels from the real time monitoring network, a trigger level of 40µg/m3 against the PM10 24hr criteria has been set where this level is maintained over a 30 minute period. A second trigger is set for peak 1hr PM10 concentrations at 90µg/m3 for a 1 hour period in order to take any management measures prior to potential impacts on maintaining compliance with the 24hr PM10 criteria. The alert triggers are constrained by wind direction so as to only trigger when within the zone of influence from the Tarrawonga site, being 310-350°. These alerts provide site with opportunities to modify operations where appropriate in order to minimise dust generation and potential exceedances of compliance criteria. The implementation of the BTM Complex Air Quality Monitoring network will impact on site alert arrangements, and any revisions to alert levels or monitoring networks will be reflected in an update to this plan.	Noted and observed on site	Compliant			
	2 Notification (of exc	eedance)					





The Group Environment Manager must determine whet is a risk of material harm to the environment as a result exceedance. If so, the exceedance must be reported im as per the site's Pollution Incident Response Manageme and Condition 8 of Schedule 5 of the Project Approval. 147 of the POEO Act defines material harm as follows: 147 Meaning of material harm to the environment (1) For the purposes of this Part: (a) harm to the environment is material if: (i) it involves actual or potential harm to the health or so human beings or to ecosystems that is not trivial, or (ii) it results in actual or potential loss or property dama amount, or amounts in aggregate, exceeding \$10,000 (conther amount as is prescribed by the regulations), and (b) loss includes the reasonable costs and expenses that incurred in taking all reasonable and practicable measure prevent, mitigate or make good harm to the environment (2) For the purposes of this Part, it does not matter that the environment is caused only in the premises where the state of the premises where the cause of the premise of the cause of the project	t of the inmediately ent Plan, Section safety of large of an or such lat would be lares to lent. t harm to Noted	Audit Finding Not Applicable	Consequence	Likelihood	Risk
is a risk of material harm to the environment as a result exceedance. If so, the exceedance must be reported im as per the site's Pollution Incident Response Manageme and Condition 8 of Schedule 5 of the Project Approval. \$147 of the POEO Act defines material harm as follows: 147 Meaning of material harm to the environment (1) For the purposes of this Part: (a) harm to the environment is material if: (i) it involves actual or potential harm to the health or shuman beings or to ecosystems that is not trivial, or (ii) it results in actual or potential loss or property dama amount, or amounts in aggregate, exceeding \$10,000 (cother amount as is prescribed by the regulations), and (b) loss includes the reasonable costs and expenses that incurred in taking all reasonable and practicable measure prevent, mitigate or make good harm to the environme	t of the inmediately ent Plan, Section safety of large of an or such lat would be lares to lent. t harm to Noted	Not Applicable			
pollution incident occurs.					
If the exceedance has not caused, or will not cause, mai to the environment, and the exceedance is determined related to operations at Tarrawonga, the EPA should be as soon as practicable of exceedances for licenced mon points. Dust monitoring results are also provided in the Annual Return and AEMR/Annual Review.	to be e notified aitoring	Not Applicable			
ctive Action Plan and Re-assessment		<u> </u>			
5.1 deposition indicate compliance related problems, a corr		Not Applicable			
Where it is determined that a corrective action plan is n TCPL will prepare the plan to reduce dust generation an reduce dust deposition, PM2.5 and/or PM10 concentral around the mine site and return the operation to compl Preparation of the plan may require the assistance of a air quality consultant and the plan should include a sche assessment. Details on the preparation of the correctiv plan and re-assessment will be included in the relevant	nd thereby itions iliance. specialist edule for re- ve action	Not Applicable			
cation (of Compliance)			<u> </u>		
TCPL will notify EPA and other relevant government age and local stakeholder(s) of the return to compliance fol	** *	Not Applicable			
5	If review of monitoring real time data or measured dus deposition indicate compliance related problems, a cor action plan will be initiated. Where it is determined that a corrective action plan is a TCPL will prepare the plan to reduce dust generation at reduce dust deposition, PM2.5 and/or PM10 concentra around the mine site and return the operation to comp Preparation of the plan may require the assistance of a air quality consultant and the plan should include a sch assessment. Details on the preparation of the corrective plan and re-assessment will be included in the relevant AEMR/Annual Review and EPL Annual Return and to EF implementation. TCPL will notify EPA and other relevant government agand local stakeholder(s) of the return to compliance fo successful completion of Step 3.	If review of monitoring real time data or measured dust deposition indicate compliance related problems, a corrective action plan will be initiated. Where it is determined that a corrective action plan is necessary, TCPL will prepare the plan to reduce dust generation and thereby reduce dust deposition, PM2.5 and/or PM10 concentrations around the mine site and return the operation to compliance. Preparation of the plan may require the assistance of a specialist air quality consultant and the plan should include a schedule for reassessment. Details on the preparation of the corrective action plan and re-assessment will be included in the relevant AEMR/Annual Review and EPL Annual Return and to EPA prior to implementation. TCPL will notify EPA and other relevant government agency(ies) and local stakeholder(s) of the return to compliance following the successful completion of Step 3.	If review of monitoring real time data or measured dust deposition indicate compliance related problems, a corrective action plan will be initiated. Where it is determined that a corrective action plan is necessary, TCPL will prepare the plan to reduce dust generation and thereby reduce dust deposition, PM2.5 and/or PM10 concentrations around the mine site and return the operation to compliance. Preparation of the plan may require the assistance of a specialist air quality consultant and the plan should include a schedule for reassessment. Details on the preparation of the corrective action plan and re-assessment will be included in the relevant AEMR/Annual Review and EPL Annual Return and to EPA prior to implementation. TCPL will notify EPA and other relevant government agency(ies) and local stakeholder(s) of the return to compliance following the successful completion of Step 3.	If review of monitoring real time data or measured dust deposition indicate compliance related problems, a corrective action plan will be initiated. Where it is determined that a corrective action plan is necessary, TCPL will prepare the plan to reduce dust generation and thereby reduce dust deposition, PM2.5 and/or PM10 concentrations around the mine site and return the operation to compliance. Preparation of the plan may require the assistance of a specialist air quality consultant and the plan should include a schedule for reassessment. Details on the preparation of the corrective action plan and re-assessment will be included in the relevant AEMR/Annual Review and EPL Annual Return and to EPA prior to implementation. TCPL will notify EPA and other relevant government agency(ies) and local stakeholder(s) of the return to compliance following the	If review of monitoring real time data or measured dust deposition indicate compliance related problems, a corrective action plan will be initiated. Where it is determined that a corrective action plan is necessary, TCPL will prepare the plan to reduce dust generation and thereby reduce dust deposition, PM2.5 and/or PM10 concentrations around the mine site and return the operation to compliance. Preparation of the plan may require the assistance of a specialist air quality consultant and the plan should include a schedule for reassessment. Details on the preparation of the corrective action plan and re-assessment will be included in the relevant AEMR/Annual Review and EPL Annual Return and to EPA prior to implementation. TCPL will notify EPA and other relevant government agency(ies) and local stakeholder(s) of the return to compliance following the successful completion of Step 3.





Deference	Cleves	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	5.1	Within 3 months of receiving a written request from a landholder with acquisition rights, TCPL shall make a binding written offer as detailed in Condition 4(8) of PA 11_0047. Air quality acquisition criteria are specified in Condition 3(26) while Condition 4(8) requires TCPL to pay all reasonable costs associated with the land acquisition process.	No Such requests.	Not Applicable			
Air Quality & Greenhouse Gas Management Plan	5.1	It should be noted that under PA 11_0047, property ID 49 retains existing acquisition rights on the basis of air quality impacts, and can, on making a written request as per Condition 3(1), seek acquisition under the terms of Conditions 4(8) and 4(9). Property ID's 44 and 45, listed in PA 11_0047, have been acquired and are now project related properties.	No Such requests.	Not Applicable			
	6 Reporting						
Air Quality & Greenhouse Gas Management Plan	5.1	The recorded exceedance, corrective actions and reassessment will be reported to the CCC and included in each relevant AEMR/Annual Review.	AEMRs and CCC Minutes confirm this occurs	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.1	The implementation of real time air quality monitoring will provide TCPL with additional information on which to make operational decisions on a day to day basis. The results of predictive meteorological forecasting and predictive real time air dispersion modelling, coupled with real time air quality monitoring will provide site management with appropriate tools to plan daily mining activities, identify causes of any elevated dust results and make changes to site operations accordingly to ensure compliance. This management tool should reduce the overall potential for TCPL to exceed operational air quality criteria on an annual average and 24 hour basis through the provision of alerts and active management on site where dust contributions are related to site activities. Exceedances in 24 hour or annual average results may still occur due to regional and/or other dust sources, however, the data available in real time, coupled with predictive forecasting will provide the relevant information to assess contributing sources and appropriate response at a site level.	Noted, observed to a certain extent on site though weather was benign at the time of the site inspection.	Compliant			
	5.2 Complaints						
Air Quality & Greenhouse Gas Management Plan	5.2	In order to ensure an appropriate and consistent level of reporting, response and follow-up to any complaints is adopted by TCPL, the following complaints management protocol will be followed: • A publicly advertised telephone complaints line will be in place to receive complaints during operating hours and record complaints at other times.	Advertised on the website.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	5.2	Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. The nature of the incident that led to the complaint, including the time of the dispersal and its duration. The action taken by TCPL in relation to the complaint, including any follow-up contact with the complainant. If no action was taken by TCPL, the reason why no action was taken.	Complaints register was reviewed and found to be compliant.	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	 The Group Environment Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational). 	Noted, complaints register and response and follow provided as evidence confirm timing	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	Data from the site weather station will be obtained for the time applicable to the complaint for use in determination of cause and identification of future remedial actions.	Complaint investigation	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site.	Evidence of visits being organised presented	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	Once the identified measures are undertaken, the Group Environment Manager will sign off on the relevant complaint within the Complaints Register.	Sighted on site.	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	• If necessary, follow-up monitoring will take place to confirm the source of the complaint is adequately mitigated.	Not required for any of the complaints that were followed through the system	Compliant			
Air Quality & Greenhouse Gas Management Plan	5.2	A copy of the Complaints Register will be kept by TCPL and made available to the CCC and the complainant (on request). A summary of complaints received every 12 months will be provided to DPE, GSC, NSC, EPA, DRE and the CCC through the AEMR/Annual Review.	·	Compliant			
	5.3 Non-Compliance						
Air Quality & Greenhouse Gas Management Plan	5.3	With the exception of air quality criteria exceedances (as discussed in Section 5.1), noncompliances relating to air quality would most likely relate to issues with monitoring such as monitor interference/malfunction. Such non-compliances will be detailed in the EPL Annual Return and/or AEMR/Annual Review.	No exceedences in the audit period	Not Applicable			
	5.4 Unpredicted Imp	act Protocol					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Air Quality & Greenhouse Gas Management Plan	5.4	In the event that unpredicted or unforeseen air quality impacts are identified, the following protocol will be adopted. Table 3 Unpredicted Impact Protocol Table 3 Unpredicted Impact Protocol Review the unpredicted impact including consideration of: 1 Review the unpredicted impact including consideration of: 2 Any relevant monitoring data; and Current mina activities as well as activities in the vicinity of the issue. Commission an investigation by an appropriate specialist into the unpredicted impact, if considered appropriate. 3 Develop appropriate ameliorative measures based on the results of the above investigations, in consultation with relevant government departments. Implement additional monitoring, where relevant, to measure the effectiveness of the improvement measures.	No exceedences in the audit period	Not Applicable			
	6 Greenhouse Gas Ma	anagement	<u> </u>				
Air Quality & Greenhouse Gas Management Plan	6	TCPL will continue to record emissions generated by the site and assess and implement any identified feasible efficiencies in the future.	Noted				
	7 Document Review a	and Continuous Improvement					
Air Quality & Greenhouse Gas Management Plan	7	This document will be reviewed internally on an annual basis following completion of the AEMR/Annual Review, and/or following an air quality related incident, audit, or modification to the conditions of approval. In addition, every 2 years, the plan will be subject to review with consultation with the relevant agencies.	Noted. This review takes place but there is no auditable documented trail to provide evidence that the review has occurred. Recommendation made elsewhere	Compliant			
Air Quality & Greenhouse Gas Management Plan	7	TCPL will investigate and implement ways to improve the environmental performance of the project over time. This will be achieved by keeping abreast of best practice in the industry for air quality controls and reporting on outcomes of air quality monitoring annually in the AEMR/Annual Review.	Networking, working with other mines in the area, input from specialists, input from regulatory authorities and audits. Monitoring is reported in the AEMR.	Compliant			





D. C	Cl	Post to most	E Maria	a dra et altra		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
		ent Strategy - March 2014					
	3 Air quality strategy						
Air Quality Management Strategy - March 2014	3.1 Air quality assessn	The conditions require that BCPL,MCC and TCM must ensure particulate emissions generated by BTM Complex operational activities do not exceed the criteria listed in Tables 3.1 to 3.3 at any residence on privately-owned land or on more than 25 per cent of any privately owned-land. Table 3.1 Long term criteria for particulate matter Poliulated	Same criteria as PA, no exceedences.	Compliant			
	4 Monitoring						
Air Quality Management Strategy - March 2014	4	The mines of the BTM Complex already have comprehensive air quality monitoring systems in place. It is proposed that the existing air quality monitoring network will be upgraded to reflect the implementation of the BTM Complex cumulative air quality monitoring network.	Noted, not yet implemented.	Not Applicable			
	4.2 Proposed cumulat	tive monitoring network					
Air Quality Management Strategy - March 2014	4.2	The requirements of the cumulative monitoring network at the BTM Complex are to: - facilitate compliance with existing and likely future consent conditions - allow proactive management and real-time dust monitoring to assist in day to day operations of each mine site - develop an integrated and coordinated approach to air quality management of the BTM Complex - consolidate existing monitoring - allow for predictive meteorological forecasting - include procedures for identifying and apportioning the source(s) and contribution(s) to cumulative air impacts for mines and other sources, using the air quality and meteorological monitoring network - include appropriate investigative tools such as modelling of post incident plume dispersion.	Noted.				





Poforonco	Clause	Poguiroment	Evidonos	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Air Quality Management Strategy - March 2014	4.2	The BTM Complex monitoring network will include: Installation of four TEOMs, which include: - one TEOM at Bradshaw's, next to the Fairfax Public School - one TEOM at the Flixton property south east of TCM - one TEOM south of the Boggabri Coal Mine and TCM at the "Tarrawonga" property and - one TEOM (proposed to be located) at the northwest of the Maules Creek Coal Mine at the "Murphy" property. All TEOMs will measure PM ₁₀ , with at least one capable of measuring PM _{2.5} . All but the "Murphy" TEOM has been installed to date.	Noted, not yet implemented.	Not Applicable			
			Noted, not yet implemented.				
Air Quality Management Strategy - March 2014	4.2	- Installation of four portable real-time ${\rm PM_{10}}$ monitors for day to day operational dust management (e.g. e-samplers or equivalent).		Not Applicable			
Air Quality Management Strategy - March 2014	4.2	- Implementation of a web based system to manage real-time monitoring data (as well as weather, emissions and modelled predictions for air quality and noise).	Noted, not yet implemented.	Not Applicable			
Air Quality Management Strategy - March 2014	4.2	Relocation of one HVAS: - the Boggabri HVAS should be moved from the "Merriown" residence to a location in the proximity of the "Roma / Glenhope" residences to the southwest of its current location. - the exact locations of the HVAS will need to be negotiated with landowners.	Not able to be verified as the online system is not set up yet and this is a Boggabri Mine initiative.	Noted			
Air Quality Management Strategy - March 2014	4.2	Review of the locations for dust deposition gauges. This will consider recent mine plan modifications, which may require the relocation of existing dust deposition gauges to accommodate mining activity, as well as improvements to the coverage of the dust deposition matrix around the BTM Complex by sharing the available network. It is likely that some of the dust deposition gauges that are currently located within mining leases or along the project boundaries could be moved to locations between the BTM Complex operations and nearest residences such that the array of dust deposition gauges better captures potential emissions under all wind directions.	Noted, not yet implemented.	Not Applicable			
	4.2.1 Real-time mon	itors	T	1			
Air Quality Management Strategy - March 2014	4.2.1	It is important to understand the different roles of monitors in the cumulative network. The real-time monitors for compliance purposes (e.g. TEOMs) will be fixed at the locations outlined above (and shown in Figure 4-1) and will be capable of measuring PM_{10} and $PM_{2.5}$. The monitors will be used to demonstrate compliance with air quality criteria and be fixed at relevant locations of exposure. They will also be used to determine (in real time) if pre-defined trigger levels have been breached and when additional dust control is required.					





Reference Clause Requirement Prof. Mr. Conjule requirement Prof. April (Prof. April 1975) Prof. Appril (Prof. April 1975) Prof. Appril (Prof. April 1975) Prof. Appril (Prof. Appril 1975) Prof. Appril (Pr	Deference	Classes	Demilionent	Friid-was	A dia Fin dia		Risk	
PM unosition (se-samplers or equivelent) initiately for day to day dust management. It is intered but these portable monitors will be phone at appropriate locations dispers from the programment of the pr	Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
As an interim control site, reference will be made to EFA monitoring data collected at Tamworth, which will provide an indication of regional air quality not influenced by mining from the BTM Complex. 4.4 Responsibility of the individual mines. Air Quality Management Strategy - March 2014 A.5 Data management and interim control management and interim proposed that a proposed that an appropriate and interim control management and interim	, -	4.2.2	PM ₁₀ monitors (e-samplers or equivalent) initially for day to day dust management. It is intended that these portable monitors will be placed at appropriate locations closer to mining operations. The portable monitoring locations will move periodically as BTM Complex mining operations progress. Their locations will take account of a number of factors, such as: - seasonally predominant daily wind patterns - the relative locations of each mines highest controllable dust generating sources - practicality of locating monitoring equipment close to the mining		Not Applicable			
Air Quality Management Strategy - March 2014 A. Responsibility of the included an indication of regional and quality not influenced by mining from the BTM Complex. Air Quality Management Strategy - March 2014 A. Data managements of the operation of the monitoring network. Arrangements have been confirmed between the mines regarding the strategy - March 2014 A. Data management and interpretation. A. S Data management and interpretation. Air Quality Management Strategy - March 2014 Air Quality Management A. S Data management and interpretation. Air Quality Management A. S Data management and interpretation. Air Quality Management A. S Data management and interpretation. Air Quality Management Strategy - March 2014 Air Quality Management A. S Data management and interpretation. Air Quality Management Strategy - March 2014 Air Quality data will be summanased, validated and available for the love viewed in various formats on a secure website to display the data in read-time. Air Quality Management Strategy - March 2014 Air Quality Management Strategy - March 2014 A B Predictivem and read-time air quality management A. S Predictivem and read-time air quality management Strategy - March 2014 A B Predictivem and read-time air quality management A. S proposed that a predictive and reactive air quality management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in read-time air quality management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in read-time and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide alerts and recommendations with respect to abstement or avoidance of potentials usess and		4.3 Regional monitori	ng (control site)					
Each mine shares responsibility for the maintenance, calibration, repair, operating costs and site access agreements for the poeration of the monitoring network. Arrangements have been confirmed between the mines regarding the nogonic polistics of operating the monitoring network. Arrangements have been confirmed between the mines regarding the nogonic polistics of operating the monitoring network. Air Quality Management Strategy - March 2014 Air Quality Management Air Quality Management Air Quality Management 4.5 public and agencies on a monthly basis, via each mine site and various formats on a secure website to display the data in real-time. Air Quality Management Strategy - March 2014 A.5 Predictiven and real-time are quality monitoring system is installed, commissioned and proven. 4.5 revents of requirements 4.6.1 Overview of requirements 4.6.1 Overview of requirements Air Quality Management Air	Air Quality Management	-	As an interim control site, reference will be made to EPA monitoring data collected at Tamworth, which will provide an indication of regional air quality not influenced by mining from the	Noted, not yet implemented.	Not Applicable			
Air Quality Management Strategy - March 2014 4.4 4.5 Data management store the operation of the monitoring network. Arrangements have been confirmed between the mines regarding the ongoing logistics of operating the monitoring network. 4.5 Data management and interpretation Air Quality Management Strategy - March 2014 Air Quality Management A.5 Under the service of the ser		4.4 Responsibility of t	he individual mines					
Air Quality Management Strategy - March 2014 Air Quality Management		4.4	repair, operating costs and site access agreements for the operation of the monitoring network. Arrangements have been confirmed between the mines regarding	Noted				
Air Quality Management Strategy - March 2014 4.5 Air Quality Management Strategy - March 2014 4.6 Air Quality Management Strategy - March 2014 4.6 Air Quality Management Strategy - March 2014 4.6 Air Quality Management Strategy - March 2014 Air Quality Management Air Qualit		4 5 Data managemen						
Air Quality Management Strategy - March 2014 4.5 website. The availability of this data will be staged, as detailed in Section 6.1, as the air quality monitoring system is installed, commissioned and proven. 4.6 Predictiven and real-time air quality management 4.6.1 Overview of requirements It is proposed that a predictive and reactive air quality management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in real-time - perform scenario modelling under predicted adverse or other operating conditions Air Quality Management Strategy - March 2014 4.6.1 - develop a log of community complaints and air quality events - evaluate community complaints and determine if BTM Complex activities may have caused an impact - accept information and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide alerts and recommendations with respect to abatement or avoidance of potential issues and	Air Quality Management		It is proposed that air quality monitoring data from the three mine sites will be stored in a central data repository. The data will be available for use by each mine site and will be able to be viewed in various formats on a secure website to display the data in real-	Noted				
Air Quality Management Strategy - March 2014 Air Quality Management	Strategy - March 2014		public and agencies on a monthly basis, via each mine site's website. The availability of this data will be staged, as detailed in Section 6.1, as the air quality monitoring system is installed, commissioned and proven.	Noted				
It is proposed that a predictive and reactive air quality management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in real-time - perform scenario modelling under predicted adverse or other operating conditions - develop a log of community complaints and air quality events - evaluate community complaints and determine if BTM Complex activities may have caused an impact - accept information and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide alerts and recommendations with respect to abatement or avoidance of potential issues and								
management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in real-time - perform scenario modelling under predicted adverse or other operating conditions - develop a log of community complaints and air quality events - evaluate community complaints and determine if BTM Complex activities may have caused an impact - accept information and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide alerts and recommendations with respect to abatement or avoidance of potential issues and		4.6.1 Overview of req		les es		<u> </u>		
4.6.2 Components	Strategy - March 2014		management system will be implemented for the BTM Complex that personnel will use to: - assess potential offsite impacts and evaluate community risk in advance and in real-time - perform scenario modelling under predicted adverse or other operating conditions - develop a log of community complaints and air quality events - evaluate community complaints and determine if BTM Complex activities may have caused an impact - accept information and data inputs from various instruments and data sources (e.g. web services, real-time monitoring, and/or emissions estimates based on activity data) - provide alerts and recommendations with respect to abatement or avoidance of potential issues and	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	LVIdelice	Addit Fillding	Consequence	Likelihood	Risk
Air Quality Management Strategy - March 2014	4.6.2	It is proposed that the predictive and reactive air quality management system will include: - a predictive component: using forecast weather data and dispersion modelling - a reactive component: using real-time meteorology, air quality monitoring and dispersion modelling - a non-steady state air quality dispersion model (that is capable of processing data at a sub-hourly time interval) - short term tiered trigger levels and notifications for managing potential impacts - a daily forecast report: providing information on temperature inversions, wind conditions at various heights, dust risk, and recommended control actions.	Noted, not yet implemented.				
Air Quality Management Strategy - March 2014	4.6.2	It is extremely important to maintain periodic review of any forecast and real-time air quality system to ensure that the system is operating using: - validated meteorological forecasts - data from calibrated monitoring equipment - accurate varying emission rates, informed by campaign monitoring where necessary - accurate emission source parameters, i.e. updated as the mine plan evolves.	Noted				
	4.6.3 Predictive fore						
Air Quality Management Strategy - March 2014	4.6.3	It is proposed that a predictive forecast meteorology system be implemented based on the Weather Research & Forecasting (WRF) model and CALMET, specifically for the BTM Complex, and a website be developed to make data immediately available for each of the mine sites, with half hourly forecasts up to 48 hours in advance. This system will download global meteorological data and forecasts on a daily basis and process and run the WRF model to produce the information required for input to a real-time 3D dispersion model. An example of a forecast and dust risk summary is presented in Figure 4-2.					
Air Quality Management Strategy - March 2014	4.6.3 4.6.4 Local observed	Once this meteorological system is configured and operating, the outcomes will be evaluated by a competent meteorologist or atmospheric science professional against actual meteorological and dust measurements and the meteorological system will be validated and improved, where possible.	Noted, not yet implemented.				
Air Quality Management		Data from local automatic weather stations will be used to	Noted.				
Strategy - March 2014 Air Quality Management Strategy - March 2014	4.6.4	validate weather forecasting model performance over time. Meteorological instrumentation or data communications equipment will be reviewed to confirm that the right quality of	Noted.				
	4.6.5 Integrated real	data is available to the systemtime monitoring data					
Air Quality Management Strategy - March 2014	4.6.5	To enable real-time reactive feedback from the system, it is proposed that a connection be established to receive a data feed from weather stations and air quality monitoring equipment in the BTM Complex network.	Noted.				
	4.7 Predictive and re	active triggers					
Air Quality Management Strategy - March 2014	4.7	Predictive and 'near real-time' reactive triggers will be configured in the system. These triggers will be initially set based on analysis of the available monitoring data and experience from other similar operations where these systems are operating.	Noted				





2.6	GI	Dominous de	F 11	A discrete discre	Risk			
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk	
Air Quality Management Strategy - March 2014	4.7	An example of Investigation and Action trigger levels are shown in Table 4-1. These trigger levels have been set based on real-time monitoring data recorded at the Fairfax Public School. The relationship between peak 1-hour PM ₁₀ concentrations and mean 24-hour PM ₁₀ concentrations are analysed to determine the level of 1-hour PM ₁₀ concentrations that may result in elevated 24-hour PM ₁₀ concentrations. Table 4.1 Investigation and trigger levels Table 4.1 Investigation 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are required to identify what activities are occurring varieties 1-hour everage Pk10 Relevant personnel are re	Noted, not yet implemented.					
Air Quality Management Strategy - March 2014	4.7	It is important to note that once the real-time air quality management system is operational, trigger levels will be reviewed, updated and refined following a review of the data and calibration of the system.	Noted					
Air Quality Management Strategy - March 2014	4.7	If the trigger levels are not appropriate to site operations, for example, there are too many or too few investigation or action responses, they will be reviewed and updated. Different trigger levels may be set for each monitoring location within the cumulative network. For example, they may be set higher for monitoring locations closer to dust sources. Trigger levels will also be regularly assessed as part of the ongoing review of this plan.	Noted					
Air Quality Management Strategy - March 2014	4.7	Actual predictive and reactive triggers will be reviewed regularly, and be based on the initial air quality data collected during commissioning of air quality monitoring equipment as well as ongoing monitoring results.	Noted					
Air Quality Management Strategy - March 2014	4.7	SMS and email alerts will be sent to relevant personnel and monitoring data will be displayed in near realtime on a customised web based reporting system.	Noted					
Air Quality Management Strategy - March 2014	4.7	Real-time dust management alerts are sent if the trigger conditions outlined above are met. The notification will also identify which criteria have triggered the alert.	Noted					
Air Quality Management Strategy - March 2014	4.7	Alerts will be sent when a new level is triggered, i.e. subsequent time periods that result in the same dust level will not generate multiple warnings. When the conditions increase to a higher alert level or when conditions return to a lower alert level, the system will send a new notification alerting all relevant personnel to the new dust management alert level.	Noted					
Air Quality Management Strategy - March 2014	4.7	All alerts are recorded by the system in an alert log that can be analysed at any time to identify trends or patterns in alerts that may lead to improvements in operational planning and/or dust control that is focussed on certain areas of operations or times of the day.	Noted					
- 5	Corrective and prev	ventative actions						

Air Quality Management Strategy





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Addit Fillding	Consequence	Likelihood	Risk
Air Quality Management Strategy - March 2014	5.1	It is proposed that the reactive component of the dust management system will be designed to process real-time data from PM ₁₀ monitors and weather stations. It will generate outputs (such as those outlined in Section 4.7.1) that are used with predetermined triggers to assess the potential for dust impacts from operations. The system will notify operators when triggers are activated. The system will be used to analyse and provide information on potential dust sources that are responsible for the increase in monitored dust.	Noted				
			Noted				
Air Quality Management Strategy - March 2014	5.1	For the BTM Complex, real-time monitors will be used to measure $\rm PM_{10}$ concentrations at a number of locations around the operations (for example as shown in Figure 4-1). The dust monitoring data will be sent in short time steps to a web server where it will be processed by the air quality management system. Trigger levels will be set for the real-time monitors (i.e. TEOMs and portable samplers). As the system operates over time the trigger levels will be refined through consideration of historical data and any other relevant observations.					
			Noted				
Air Quality Management Strategy - March 2014	5.1	If a real-time monitor triggers an alert, the system will query the monitoring data to determine if mining operations are upwind of the triggered monitor. If so, it will be used to assess whether activities occurring between upwind and downwind monitors are creating an increased level of dust that has set off an alert. The system will use available weather data to determine the likely area of the operations that contains the dust generating source. This can be done by activating a reverse trajectory analysis of the plume that has triggered an alert.					
	5.2 Mitigation	<u> </u>		•			
Air Quality Management Strategy - March 2014	5.2	Processes to mitigate air quality outcomes associated with operations are addressed in each mine sites individual AQGHGMPs.	These processes are noted in the Tarrawonga AQGGMP				
Air Quality Management Strategy - March 2014	5.2	Each mines rankings will be used as the basis for scheduling operational activities or increasing dust control measures to mitigate risks when dust generation is predicted to reach trigger levels.	Noted				
Air Quality Management Strategy - March 2014	5.2	Dust generation assessment will be undertaken by experienced site personnel with the assistance of various specialists (e.g. operations, environment and air quality specialists) as required.	Noted				
	5.3 Communication	To a sure and a sure as a	lea e e e e e		1		
Air Quality Management Strategy - March 2014	5.3	Regular meetings will be held by nominated personnel representing each of the mines in the BTM Complex (at least quarterly) to discuss predictive model outcomes, monitoring results and future operational events. Meeting minutes will be documented and retained at each mine site.	These meetings currently occur but without specific focus on the issues here.				
Air Quality Management Strategy - March 2014	5.3	The trigger levels will initiate internal communication within the BTM Complex to allow the complex to implement management measures in order to reduce dust generation.	Noted				
Air Quality Management Strategy - March 2014	5.3	When air quality criteria are identified as exceeded, discussions will be held within the BTM Complex, regulatory agencies and affected landholders (where an exceedance occurs on privatelyowned land).	Noted				
	5.4 Reporting						





Deference	Clause	Dogwiromont	Evidence	Audit Finding		Risk	
Reference	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Air Quality Management	5.4	Internal management reports will be prepared regularly, noting	Sighted on site, included in presentations to	Compliant			
Strategy - March 2014		performance against triggers and criteria.	crews, CCC and management meetings				
		External reporting will include:	Evidence quoted elsewhere inthis report				
Air Quality Management		- updates on individual company websites - presentations to Community Consultative Committees (CCCs)	indicates compliance.				
Strategy - March 2014	5.4	- Annual Environmental Management Reports (AEMRs)/Annual		Compliant			
Strategy - Waren 2014		Reviews					
		- exceedance reporting.as required.					
	6 Implementation	<u> </u>			•		
	6.1 Staged approach					ľ	
		It is proposed that a staged approach will be taken to install the	Noted				
		equipment and systems which are additional to individual mine's					
		existing air quality monitoring systems.					
		Stage 1					
		Equipment acquisition, installation and commissioning					
Air Quality Management		Stage 2					
Strategy - March 2014	6.1	Review equipment and processes					
		Stage 3					
		Implement predictive modelling and management					
		Stage 4					
		Publish Webpages					
		Figure 6-1 Staged approach to implementation					
	6 1 1 Stage 1 - Equip	ment acquisition, installation and commissioning					
	0.1.1 Stage 1 - Equipi	ment acquisition, installation and commissioning	Noted		1		
		Four real-time portable PM10 monitors are proposed that will be					
		acquired and commissioned at locations relevant to current					
Ali Carlita Managara		mining operations, as part of the day to day management of real-					
Air Quality Management	6.1.1	time dust. An indicative layout of these monitors is also shown in					
Strategy - March 2014		Figure 4-1. The installation of these portable PM10 monitors will					
		be reviewed as part of Stage 2, to ensure they allow sufficient					
		coverage to achieve the required monitoring goals. The					
		configuration may change over time as each mining pit develops.					
<u> </u>	6.1.2 Stage 2 - Reviev	w equipment and processes Within three months of the installation and commissioning of	Noted not not implemented			ı	
		Stage 1 equipment the following will be reviewed:	Noted, not yet implemented.				
Air Quality Management		- performance and reliability of the cumulative air quality					
Strategy - March 2014	6.1.2	monitoring equipment		Not Applicable			
Strategy - March 2014		- triggers proposed in this protocol					
		- central data repository and data interface.					
		Each mine's respective air quality management plans will be	Noted, not yet implemented.				
Air Quality Management		updated at the end of each stage to ensure consistency with the	,, p				
Strategy - March 2014	6.1.2	AQMS. This will include a review of trigger action response plans		Not Applicable			
		(TARPs).					
	6.1.3 Stage 3 - Implei	ment predictive modelling and management					_
		Within six months of the installation and commissioning of Stage 1	Noted, not yet implemented.				
Air Quality Management	6.1.3	equipment the predictive modelling system will have been		Not Applicable			
Air Quality Management Strategy - March 2014			1				
Strategy - March 2014		acquired, installed and commissioned.					
	6.1.3	acquired, installed and commissioned. This system's performance will be reviewed every three months and validation reports will be produced.	Noted, not yet implemented.	Not Applicable			





Reference	Clause	Doguiroment	Evidence	Audit Finding		Risk	
Kererence	Clause	Requirement	Evidence	Audit Finding	Consequence	Likelihood	Risk
Air Quality Management Strategy - March 2014	6.1.4	Each mine site will establish or update an existing Company webpage.	Noted, Tarrawonga has a suitable webpage	Compliant			
Air Quality Management Strategy - March 2014	6.1.4	The webpage will present the summarised and validated results of the real-time air quality monitoring on a monthly basis.	Tarrawonga post up results and management actions where applicable	Compliant			
Air Quality Management Strategy - March 2014	6.1.4	Air quality monitoring data must be reviewed via formal quality assurance processes before it can be considered valid data.	Noted, this occurs .	Compliant			
Air Quality Management Strategy - March 2014	6.1.4	Continuous data collected by the real-time monitors will undergo preliminary data validity checks (for example, to identify outliers, negatives etc.), however until formal validation/ratification has been conducted any continuous data reported to the community will need to be considered preliminary and subject to further validation.	Noted				
	7 Document Control						
	7.1 Review and revis	·					
Air Quality Management Strategy - March 2014	7.1	This AQMS, its operation and implementation, will be reviewed and revised at least every two years or on an 'as required' basis to incorporate improvements identified by the BTM Complex or appropriate requirements of government agencies. The AQMS will be reviewed and updated at the end of each stage of the project rollout, as described in Section 6.1.	Noted, not yet implemented.	Not Applicable			
Air Quality Management Strategy - March 2014	7.1	In accordance with the project approvals, the AQMS will also be revised within three months of: - an annual review - incident threatening material harm, requiring notification of the Director-General / relevant agencies - statutory audit - modification of project approval.	Noted, not yet implemented.	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	WHC_PLN_TAR_Wa	ter Management Plan					
	2.2.3 Contaminated	Water Management					
Water Management Plan	2.2.3	These dirty water areas include the: • Wash-down area; • Workshop; • Fuel, oil and grease storages; and • Refuelling bays. These areas will be managed as follows: • Runoff from these areas will drain to an oil separator to remove hydrocarbons before draining to SB16. The oily fraction enters a containment system for pump out and removal from site as necessary;	Yes observed onsite	Compliant			
Water Management Plan	2.2.3	The surface facilities area has an impermeable surface and bunding that captures all flows and directs them to the oil separator;	Yes observed onsite	Compliant			
Water Management Plan	2.2.3	All oil, grease, fuel and hydrocarbon products will be securely stored; and	Yes observed onsite	Compliant			
Water Management Plan	2.2.3	Refuelling, oiling and greasing will take place in designated areas only.	Yes observed onsite	Compliant			
Water Management Plan	2.2.3	The following actions will be taken if a major hydrocarbon spill on an unsealed area occurs: • The contaminated soil at the site of the spill will be collected and transported to an approved waste depot or remediated on the mine site;	No records of major spills on unsealed areas	Not Applicable			
Water Management Plan	2.2.3	Pits would be constructed around the spill with sufficient hydraulic gradient to capture seepage water and contaminated material, enabling the pits to be pumped out; and	Noted				
Water Management Plan	2.2.3	The local groundwater will be monitored for signs of further contamination.	Noted				
	2.3 Groundwater M	-					
	2.3.1 Management						
Water Management Plan	2.2.3	To avoid, and significantly reduce the risk of, higher groundwater inflows to the open cut mine pit from the alluvial groundwater system during the life of the Project, and in the long term, a low permeability barrier is proposed.	The OC has not progressed to the point where an alluvial barrier is required.	Not Applicable			
	2.3.2 Management	of Extracted Pit Water	<u> </u>	1	<u> </u>		





Deference	Clause	Requirement	Evidence	A. dit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	2.3.2	Saline groundwater, pumped from the mine pit, is deposited in the storage dams PW2, PW3 within the site. This water is used in dust suppression activities for the coal crusher or is allowed to evaporate. The use of pit water for dust suppression minimises the need to use clean water for this purpose and is seen as suitable way to recycle water onsite. The water will be applied at a suitable rate to ensure negligible runoff. Any water falling/applied onto the mine site would be classed as 'dirty water' and collected in dirty water dams as discussed in Section 2.5. These dams are sized and operated to ensure that they only discharge in extreme high flows thus the water used for dust suppression has a minimal chance of discharging offsite. The water in these dirty water dams are regularly tested and monitored.	The area around the coal loader drains to dams that are not nominally dirty water dams and that subsequently drain through a number of other dams offsite. Management of pit water is generally acceptable based on site observations.	Not Compliant	D	1	High
Water Management Plan	2.3.2	Should salt levels in these dams exceed acceptable limits; the water will be pumped into PW2 or PW3 and recycled.	Noted, Recommend that "acceptable limits" be quantified				
Water Management Plan	2.3.2	Pump meters are currently used to measure flow extracted from the pit; however they are periodically compromised by sediments in the flow, creating an unreliable data record. Investigations are currently underway to find improved methods to measure pit water extraction more reliably.	Noted, not yet implemented. See also non-compliance in 2.3.2 below	Not Applicable			
Water Management Plan	2.3.2	The groundwater model will be updated every 2 years, using data recorded on site. If the estimates should show that additional groundwater licencing is required, this will be sought from the water markets.	Noted, given the implementation of the more reliable meters in 2.3.2 above has not occurred, this is problematic for accurate measurement of licensed ground water extraction in the open cut. The groundwater model was last updated for the EA for the current approval, report on modell update is dated January 2012, an up date of the groundwater model is overdue.	Not Compliant	E	2	Low
	3 Erosion and Sedim	ent Control					
	3.2 On-going Erosion	and Sediment Management Practices					
Water Management Plan	3.2	The following practices will be implemented to minimise potential for erosion and sedimentation: • Installation of all erosion and sediment control (ESC) and water management structures prior to any ground disturbance taking place;	Structures observed on-site, no ground disturbance observed that did not have erosion and sed control. Information from interviews	Compliant			
Water Management Plan	3.2	Land disturbance will be minimised by clearing the smallest practical area of land ahead of disturbance activities;	Observed on site.	Compliant			
Water Management Plan	3.2	Disturbance areas which will not be actively utilised will be revegetated as soon as practical following completion of works in that area;	No disturbed ares noted on site that were not being at least intermittently utilised.	Compliant			
Water Management Plan	3.2	Where practical, disturbance areas will be shaped such as to provide a free draining surface to direct dirty water runoff into the dirty water drains for collection in the dams;	No dirty water from disturbed areas entering clean water dams	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	3.2	Where localised flooding or ponding occurs, access will be restricted until such time as the ground is no longer waterlogged in order to reduce the potential for additional sediment mobilisation;	No water logged areas observed	Compliant			
Water Management Plan	3.2	The erosion and sediment control structures will be cleaned of accumulated sediment material (or extended or replaced) as soon as 20% capacity is lost due to the accumulation of material such that the specified capacities are maintained. This is reliant on the structures being empty to allow for access;	Dams are desilted when empty and are inspected monthly.	Compliant			
Water Management Plan	3.2	The erosion and sediment control structures will be inspected monthly, or after a rainfall event of >25mm/24hr, to assess their success in preventing erosion, identify signs of potential erosion and determine the retained capacity, especially within the sediment basin;	Inspections sighted including regular monthly and post rainfall events	Compliant			
Water Management Plan	3.2	As part of the surface water monitoring program, water flowing from all discharge points will be sampled for suspended sediment;	Sampling and analysis reports sighted	Compliant			
Water Management Plan	3.2	If a high risk of site discharge is identified, excess surface water may be pumped into the newly created Box Cut PWD. This would rely on the availability of storage capacity within the pit water dam and all water moved to the dam would be treated as pit water.	Noted, this has not been required in the audit period.	Not Applicable			
Water Management Plan	3.2	• In the event the suspended sediment concentration in any discharged water exceeds 50mg/L: o The EPA will be advised and salient preceding weather information will be provided; o The upstream structures will be inspected and cleaned of consolidated sediment as required; and/or o A flocculant will be added to the water contained within the sediment basin or storage dam to increase the efficiency of sediment settlement; and/or o The sediment basin(s) will be enlarged to provide greater settlement time for the sediment containing water; and/or o An additional storage dam will be constructed downstream with this becoming the new site discharge point and monitoring location. EPA will be advised to enable amendment to the Environment Protection Licence.	No exceedences in the audit period	Not Applicable			
Water Management Plan	3.2	All discharge from flumes will flow to sediment basins;	Observed on site.	Compliant			
Water Management Plan	3.2	Water captured in the open cut void will be allowed time to settle within the sumps before being pumped to one or more of the pit water dams;	Observed on site.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	3.2	If, following heavy rain, erosion is identified on the rehabilitated landform or in operational areas, it will be remediated as quickly as possible using one or a combination of the following: o Filling the erosion channels; o Cross-ripping (along the contour) to assist infiltration; and/or o Installation of additional controls, eg banks sown with a non-perennial cover crop. Areas previously identified as exhibiting and treated to prevent further erosion will be monitored on a minimum monthly basis or following a rainfall event of >25mm/24hr.	Inspections sighted including regular monthly and post rainfall events Erosion control requirements noted, the only erosion in rehab observed on site that required maintenance was programmed to be addressed.	Compliant			
Water Management Plan	3.2	If erosion is identified on a rehabilitating landform or in the operational area, it will be remediated as quickly as practical to reduce the potential for significant erosion to develop. Areas previously rehabilitated will be inspected regularly to ensure rehabilitation works are effective; and	Some areas of rilling observed on the south east emplacement that require work, EO indicated that work to repair was planned. Planted quarter 4 2013. Difficult to remediate without damaging surrounding establishing vegetation.	Compliant			
Water Management Plan	3.2	Where necessary, temporary ESC measures will be utilised to prevent and/or reduce the potential for adverse erosion developing. Temporary ESC structures and management practices will be constructed in accordance with the construction principles presented in the Blue Book (Landcom) and may include sediment fences, check dams, surface protection and advanced revegetation methods such as hydromulching.	Noted, observed in site inspection.	Compliant			
Water Management Plan	3.2	Improved water balance modelling is currently under way and this will allow a more proactive approach to water management. The modelling will allow the level of service of the system to be better understood, leading to improved management practices. The results of this modelling will be used to update future versions of this report.	Noted, consultants on site immediately after the audit.				
	3.3 Site Rehabilitatio	n Management					
Water Management Plan	3.3	When an area of disturbance is no longer required by mine processes, it is re-vegetated in an effort to minimise erosion. Where rehabilitation is impractical disturbed ground will be covered with a geofabric or similar means.	Noted, no geofabric observed though.				
Water Management Plan	3.3	Once rehabilitated areas are established and stable, any operational sediment control structures will either be removed or left as passive water storages. This is detailed further in the Rehabilitation Management Plan.	Noted				
	4 Site Water Balance						





Reference	Clause			Audit Einding		Risk	
				Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	4	Condition 39 of the project approval (application number 11_0047) requires a site water balance model to be produced. The existing model, developed for the conditions of the previous Notice of Modification (DA 88-4-2005 MOD 1), does not fulfil all of the requirements of the latest project approval. A new water balance model that fulfils these requirements will be developed. This is a significant undertaking and could not be completed within the time limits imposed on the completion of this Plan. The water balance model will be provided separately and incorporated into a future update of this WMP.					
Nater Management Plan	4	This section summarises the work carried out to develop the previous water balance model and has been taken from the previous version of this WMP. The water balance model was constructed in Microsoft Excel and used a daily time step.	Noted				
Water Management Plan	4	The water balance model forecasts the following water movements within the water cycle: Rainfall water source and security; Demand of the water uses on site; Export with the final coal product; Import from bores; Controlled release; and Spillway discharge from the site.	Noted, this will be superceded by the new water balance.				
	4.1 Management Pr	inciples					
Water Management Plan	4	The water balance model forecasts the following water movements within the water cycle: Rainfall water source and security; Demand of the water uses on site; Export with the final coal product; Import from bores; Controlled release; and Spillway discharge from the site.	Noted				
Water Management Plan	4	As with any modelling, a water balance model incorporates a number of assumptions to represent the operations on site. These vary for each site and can also vary across a single site. The following general management principles were adopted for all elements of the Tarrawonga site: • Limiting the extent of site disturbance; • Using the minimum volume of water necessary during dry periods; • Optimising the storage available on site; • Segregation of water by quality or source; • Reducing contaminant concentrations by suitable treatment methods; • Avoiding the accumulation of large volumes of contaminated water on-site, and; • Protecting groundwater resources from contamination.	Noted				





Reference	Clause	Requirement	Evidence	Audit Einding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	5.4 Goonbri Creek Di	version, Low Permeability Barrier and Permanent Flood Bund					
		It should be noted that the detailed design of these elements has	Noted, not yet implemented.				
Mater Management Dlan	F 4	not been undertaken. This work will be carried out at a later date					
Water Management Plan	5.4	when mining has progressed and the requirements of these					
		elements are better understood.					
	5.4.1 Low Permeabili	ty Barrier					
		A low permeability barrier would be constructed in the alluvium,	Noted, not yet implemented.				
		approximately 50-100 m to the east and south-east of the final					
Water Management Plan	5.4.1	open cut extent. Construction of the low permeability barrier					
		would be completed before the open cut intersects the alluvium					
		that feeds Goonbri Creek (approximately Year 12).					
		The low permeability barrier would be constructed using a soil-	Noted, not yet implemented.				
		bentonite mixture to meet the following design objectives:					
Water Management Plan	5.4.1	Minimise the potential for local drainage of alluvial groundwater					
water wanagement rian	5.4.1	into the open cut during operations and post-mining;					
		Minimise the potential for future instability of the open cut	Noted, not yet implemented.				
Water Management Plan	5.4.1	batters formed in the alluvium (achieved by reducing the					
water wanagement ran	5.4.1	groundwater hydrostatic head in the alluvium immediately					
		adjacent to the open cut/final void);					
Water Management Plan	5.4.1	Maintain the hydraulic character of Goonbri Creek by minimising	Noted, not yet implemented.				
water wanagement ran	3.4.1	the potential loss of baseflow; and					
Water Management Plan	5.4.1	Maintain the value of alluvial groundwater, by minimising	Noted, not yet implemented.				
		potential interactions with the mine final void, post-mining.					
	5.4.2 Permanent Floo		In				
		A permanent flood bund would be constructed to prevent	Noted, not yet implemented.				
Water Management Plan	5.4.2	inundation of the open cut during operations and post-mining.					
		The permanent flood bund would generally coincide with the					
		alignment of the low permeability barrier	Noted, not yet implemented.				
		The permanent flood bund would be designed to a height that would provide protection against the peak flood height associated					
		with a Probable Maximum Precipitation rainfall event. The width					
Water Management Plan	5.4.2	and geometry of the permanent flood bund would be such that it					
		is stable under these extreme flow conditions.					
		is stable under these extreme now conditions.					
		The permanent flood bund would consist of an engineered clay fill	Noted not vet implemented				
		core, which would be sub-excavated into the natural surface. Rock					
Water Management Plan	5.4.2	fill armouring would be placed on the eastern side of the clay fill					
water management ran	51.112	core. The bund would then be topsoiled for revegetation.					
		The state of the s					
		For sections of the permanent flood bund running parallel to	Noted, not yet implemented.				
Water Management Plan	5.4.2	internal haul roads, the height of the bund would be increased to	, , , ,				
· ·		6 m as a noise mitigation measure.					
	5.4.3 Goonbri Creek						
		In approximately Year 15 of the approved works, open cut mining	Noted, not yet implemented.				
		would progress through a 3 km section of Goonbri Creek. Prior to					
Water Management Plan	5.4.3	the open cut advancing into this section of the creek, the					
		permanent Goonbri Creek alignment would be established to the					
	ĺ	east of the open cut.					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Fillding	Consequence	Likelihood	Risk
Water Management Plan	5.4.3	The permanent Goonbri Creek alignment would be constructed to meet the following design objectives: • Construct a low flow channel that approximates the existing section of Goonbri Creek upstream of the Project in terms of stream geometry, hydrology and geomorphology; • Mimic the meandering path of the existing alignment of Goonbri Creek, such that the length of the permanent Goonbri Creek alignment is approximately the same length as the section of Goonbri Creek being removed; • Minimise the disturbance to the reaches of Goonbri Creek upstream of the permanent Goonbri Creek alignment; and • Provide a stable transition back to the existing Goonbri Creek alignment which results in no detectable change to the hydraulic conditions in the reaches of Goonbri Creek or the Bollol Creek floodplain area downstream.	Noted, not yet implemented.				
	5.5 Rehabilitation Po	llution Control					
Water Management Plan	5.5	It is expected that the water quality of runoff from portions of the mine waste rock emplacement which have been rehabilitated and where sufficient time has elapsed for vegetation to establish, would reflect runoff water quality from similar un-mined areas. Once this has occurred, runoff would be released to the environment. Prior to the stabilisation of rehabilitated areas, runoff would be managed via the actively managed sediment and erosion control system.	, ,	Not Applicable			
	5.6 Leard Forest Min	ng Precinct Water Management Strategy					
Water Management Plan	5.6	TCM has been in consultation with the nearby Boggabri Coal mine and Maules Creek Project in developing a Leard Forest Mining Precinct Air Quality Management Strategy. The aim of this strategy is to minimise cumulative impacts on the quality and availability of water resources in the catchment. At the time of developing this management plan, the Leard Forest Mining Precinct Water Management Strategy was currently in draft form awaiting approval. Upon its completion, the Leard Forest Mining Precinct Water Management Strategy will be appended to this management plan. TCM and Boggabri Coal currently continue to share surface and groundwater monitoring data, whilst Boggabri Coal, Tarrawonga and the Maules Creek Project participate in monthly meetings to discuss cumulative impact management measures.	Noted				
	6 Surface Water mor	itoring plan					
Water Management Plan	6.1	Tarrawonga Coal has a comprehensive monitoring program in place across the mine site that incorporates: • The collection of rainfall and meteorological data; and • The collection of water quality data in the onsite dams and surrounding creeks.	Noted, observed in monitoring results.	Compliant			
	6.3 Monitoring Locat	ions					





Deference	Clause	Requirement	Evidence	A dit Fin din a		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	6.3	The water quality monitoring program has been designed to provide data on the flows leaving site as well as the flows in the receiving creeks. As the approved works are undertaken, the potential for impacting Goonbri Creek, Bollol Creek and the downstream watercourses is increased. For this reason the surface water monitoring program will be updated to include the following locations: • Two new surface water flow gauging stations on Goonbri Creek upstream (GC-U) and downstream (GC-D) of the permanent Goonbri Creek alignment; • Two water level (flood) gauges in the lower reaches and overflow areas of Bollol/Goonbri Creek (FG1 and FG2); • Water quality monitoring on Goonbri Creek at the two new gauging stations; • Water quality monitoring in new on-site water storages and licensed discharge points; and • Two pluviometers in the upper (PV1) and mid (PV2) reaches of the Goonbri Creek catchment. These locations will be introduced as soon as possible and will begin to establish baseline data.	This has generally not been conducted.	Not Compliant	D	2	Medium
Water Management Plan	6.3	There is a requirement to establish baseline data on soils within areas irrigated with pit water. This practice has not been used on the site and there are no current plans to introduce it in the future. The practice was kept as an option to be used should the available storage for pit water become restricted. If this option becomes likely in the future, more detailed plans will be developed and a system for testing soils introduced.	Noted				
	6.4 Water Quality Mo						
Water Management Plan	6.4	The proposed water quality monitoring program provides for the assessment of background data for flow events in the various creeks as well as regular grab samples from the mine water storages on-site to determine whether mine site runoff meets the adopted water quality compliance criteria. The samples will be collected during every flow event as well as quarterly for routine monitoring. The samples should be collected in a manner consistent with the Approved Method for Sampling and Analysis of Water Pollutants in NSW (OEH, 2004). • The event-based sampling should enable quantification of any pollutant loads from the mine site and their corresponding impact on the local creek water quality. • On-site regular sampling from the water storages allows for any potential problem areas with respect to pollutant generation onsite to be identified in advance ensuring appropriate remedial action can be taken.	The ALS monitoring reports and AEMR appendices indicate that this monitoring is occurring	Compliant			





Water Management Plan 6.4 Monitoring is the responsibility of the Environmental Officer and should be conducted by a suitably qualified professional in accordance with the relevant Australian Standards. The frequency of monitoring and range of parameters analysed during flow and routine monitoring should be reviewed as part of the revision process for this document. 6.5 Surface Water Impact Assessment Criteria Impact Assessment Criteria Impact assessment Criteria are as prescribed by the EPA and specified as concentration limits within Environment Protection Licence (EPL) 12365. Condition L2 of EPL 12365 specifies the following concentration limits for licensed points/FPL ID No 1 (SD17), 2 (SD9), 3 (SB14), 24 (SD16), 25 (SB22), 26 (SB23B) and 27 (SB24B) (Table 6 3):	Deference	Clause	Requirement	Evidence	A dik Fin din a		Risk	
Should be conducted by a suitably qualified professional in accordance with the relevant Australian Standards. The frequency of monitoring and range of parameters analysed during flow and routine monitoring should be reviewed as part of the revision process for this document. 6.4 Surface Water Impact Assessment Criteria Impact assessment criteria for surface water are only relevant to water actually discharged from the mine site. The surface water compliance criteria are as prescribed by the EPA and specified as concentration limits within Environment Protection Licence (EPL) 12365. Condition L2 of EPL 12365 specifies the following concentration limits for licensed points/EPL ID No 1 (SD17), 2 (SD9), 3 (SB14), 24 (SD16), 25 (SB22), 26 (SB23B) and 27 (SB24B) (Table 6 3):	Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Impact assessment criteria for surface water are only relevant to water actually discharged from the mine site. The surface water compliance criteria are as prescribed by the EPA and specified as concentration limits within Environment Protection Licence (EPL) 12365. Condition L2 of EPL 12365 specifies the following concentration limits for licensed points/EPL ID No 1 (SD17), 2 (SD9), 3 (SB14), 24 (SD16), 25 (SB22), 26 (SB23B) and 27 (SB24B) (Table 6 3):	Water Management Plan	6.4	should be conducted by a suitably qualified professional in accordance with the relevant Australian Standards. The frequency of monitoring and range of parameters analysed during flow and routine monitoring should be reviewed as part of the revision	sampler is a suitably qualified professional . Letters of competency show no indication of professional education. This is probably not necessary Recommendation - modifying the WMP to remove the requirement for sampling to				
water actually discharged from the mine site. The surface water compliance criteria are as prescribed by the EPA and specified as concentration limits within Environment Protection Licence (EPL) 12365. Condition L2 of EPL 12365 specifies the following concentration limits for licensed points/EPL ID No 1 (SD17), 2 (SD9), 3 (SB14), 24 (SD16), 25 (SB22), 26 (SB23B) and 27 (SB24B) (Table 6 3):		6.5 Surface Water Imp		I.,	T	T T		
Table 6-3 Surface Water Assessment Criteria Pollulant Units of Sopercentile Concentration Illimit Concentrati	Water Management Plan	6.5	water actually discharged from the mine site. The surface water compliance criteria are as prescribed by the EPA and specified as concentration limits within Environment Protection Licence (EPL) 12365. Condition L2 of EPL 12365 specifies the following concentration limits for licensed points/EPL ID No 1 (SD17), 2 (SD9), 3 (SB14), 24 (SD16), 25 (SB22), 26 (SB23B) and 27 (SB24B) (Table 6 3): Table 6-3 Surface Water Assessment Criteria Poliularia Units of So percential SO percential SO percential Institution Instituti	Noted				
All monitoring events will be analysed against this criteria, and any surface water discharge events reported to the EPA as required. Tarrawonga Coal notes Condition L1.1 and Condition L2.3 of the EPL, as follows: Condition L1.1 Except as may be expressly provided in any other condition of this licence, the licensee must comply with section 120 of the Protection of the Environment Operations Act 1997. Condition L2.3 To avoid any doubt, this condition does not authorise the pollution of waters by any pollutant other than those specified in the tables. (see Table 6 3 above). ALS monitoring reports and AEMR, there have been some dischargesincluded in the AEMRs during the audit period but none were reportable either because they were below criteria or because the rainfall event exceeded the criteria for reporting uncontrolled discharges. Compliant Compliant	Water Management Plan	6.5	surface water discharge events reported to the EPA as required. Tarrawonga Coal notes Condition L1.1 and Condition L2.3 of the EPL, as follows: Condition L1.1 Except as may be expressly provided in any other condition of this licence, the licensee must comply with section 120 of the Protection of the Environment Operations Act 1997. Condition L2.3 To avoid any doubt, this condition does not authorise the pollution of waters by any pollutant other than those specified in	have been some dischargesincluded in the AEMRs during the audit period but none were reportable either because they were below criteria or because the rainfall event exceeded the criteria for reporting	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Fillullig	Consequence	Likelihood	Risk
Water Management Plan	6.6	The Environmental Officer for the mine site will retain an active database of monitoring results which will be updated on a regular basis. Any discharge event off-site will result in the triggering of a sampling event. All sampling results will be made available in the Annual Return (where required) for the site, as well as on the TCM website via the Annual Environmental Management Report (AEMR)/Annual Review. Tarrawonga Coal will notify the EPA as soon as practicable after identification of an exceedance in concentration criteria. Tarrawonga Coal will also provide a written report within 7 days of the event to the relevant agencies which will: • Describe the date, time, duration and nature of the exceedance/incident; • Identify the type and concentration of every pollutant discharged; • Identify the cause (or likely cause) of the exceedance/incident; • Name and contact details of witnesses to the event; • Describe what action has been taken to date; and • Describe the proposed measures to address the exceedance/incident.	·	Compliant			
Water Management Plan	6.6	The ongoing monitoring program and collation of relevant data will provide the basis for continuing improvement in surface water management across the site.	Noted				
	7 Groundwater Mon						
Water Management Plan	7	The additional monitoring, carried out for the investigation into the mine extension, will be formally incorporated into the site monitoring network. This will occur as the project proceeds. • MW-1: located in the Permian coal and measures (interburden) aquifer on the "Thuin" property, • MW-2: located in the alluvial aquifer on the "Thuin" property, • MW-3: located in the Boggabri volcanics aquifer on the "Nagero" property, • MW-4: located in the alluvial aquifer on the "Tarrawonga" property, • MW-5: located in the surface water drainage line to the southwest of the mine. This location would also serve as an indicator to impacts on the alluvial aquifer servicing the "Bollol Creek Station" aquifer, • MW-6: located in the alluvial aquifer of Nagero Creek, • MW-7: located in the Permian sediment up hydraulic gradient of the mine area, and • MW-8: located on the western side of the mining lease and is the most recent monitoring point.	Based on the content of the monitoring reports these sample points have not yet been incorporated into the monitoring network.	Not Applicable			
Water Management Plan	7	All bores used for groundwater monitoring will be licenced by the NOW.	licenses provided along with WALs	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	7	The existing TCM network of piezometers will be augmented, particularly prior to and during Years 12 to 17 of the extension and remain in place for 2 years post-mining. (i.e. coincidental with the anticipated open cut intersection with the alluvial groundwater system inside the low permeability barrier). This network of piezometers would be installed for the purposes of monitoring: • the construction of the low permeability barrier (to quantify and validate the predicted short-term/localised dewatering impacts); • groundwater levels and water quality in the alluvial groundwater system on the inside of the low permeability barrier as mining advances (to validate the predicted mine inflow and dewatering rates); • groundwater pressures in the porous rock groundwater system/coal measures (to validate the predicted depressurisation effects at depth); and • groundwater levels and water quality in the alluvial groundwater system on the outside of the low permeability barrier as mining advances (to validate the predicted negligible impacts).	Not yet at years 12 to 17 of the extension.	Not Applicable			
	7.3 Baseline Data						
Water Management Plan	7.3	Pre-mining baseline groundwater level and quality information has been collected routinely since mid-2006. This is not true for all the monitoring locations; however readings are taken and maintained as soon as a bore is operational.	Noted				
	7.3.1 Groundwater Le	evels					
Water Management Plan	7.3.1	Groundwater levels are monitored, and will be monitored at all current and future locations in the monitoring system. The piezometer bores are monitored for groundwater level/pressure and quality information.	AEMR and monitoring reports, this occurs	Compliant			
	7.3.2 Groundwater Q						
Water Management Plan	7.3.2	Groundwater quality monitoring has been conducted by TCM, with samples from all available standpipes in the monitoring network submitted for laboratory analysis of pH, electrical conductivity (EC), total dissolved solids (TDS), major anions, major cations and selected heavy metals (As, Cd, Cr, Cu, Ni, Pb, Zn, Hg, V, Mn). The main water quality characteristics (pH, TDS) in the mine area are summarised in Table 7 2.	AEMR and monitoring reports, this occurs	Compliant			
	7.4 Monitoring Sched						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	7.4	Key components of the groundwater monitoring program will include: • Collaboration with surrounding mine (Boggabri Coal Mine and Maules Creek Coal Project) to establish a combined monitoring regime. • Quarterly measurement of water levels, field pH and EC in the existing network of piezometers to be monitored through the life of the project; • Quarterly measurement of water levels, field pH and EC in existing registered groundwater bores within 5 km of the Mine Lease as indicated in Table 7 1; • Six monthly collection of water samples from all standpipe piezometers for laboratory analysis of a broader suite of parameters including: o Physical properties (EC, TDS, TSS and pH) o Major cations and anions (Ca, Mg, Na, K, Cl, SO4, HCO3 and CO3) o Nutrients (ammonia, nitrate, phosphorus, reactive phosphorus) o Dissolved metals (aluminium, antinomy, arsenic, boron, cobalt, cadmium, chromium, copper, iron, lead, manganese, mercury, molybdenum, nickel, silver, selenium, zinc)	AEMR and monitoring reports, this occurs apart from completion of the collaboration with the neighbouring mines which will occur following the approval of the Water Strategy for all three mines.	Compliant			
Water Management Plan	7.4	Data collected from the piezometers will be compared against initial groundwater modelling predictions.	This occurs, see AEMR	Compliant			
Water Management Plan	7.4	The frequency of monitoring and range of parameters analysed during flow and routine monitoring should be reviewed after every two years of operations.	This has not occurred, the WMP is not approved and is not two years old.	Not Applicable			
Water Management Plan	7.4	All monitoring results will be reviewed annually and summarised in each Annual Return (where required) as well as the AEMR/Annual Review, together with an assessment of the need to modify the parameters measured or monitoring frequency. In addition to this, the raw monitoring data is published on the mine's website within two weeks of sampling. The groundwater model will also be re-calibrated using the latest data collected.	Data is published on website and included in AEMR. Monitoring results are reviewed. The groundwater model has not been updated every second year as required by the WMP.	Compliant			
	7.5 Groundwater Sar	-		T			
Water Management Plan	7.5	The method of groundwater monitoring will be undertaken following the procedure outlined below. Note that these procedures are not exhaustive and will be complemented with 'best practice' sampling protocols. • Depth to water table will be measured using a calibrated water level meter or dip tape.	this is recorded	Compliant			
Water Management Plan	7.5	Bores will be purged of at least three well-volumes of groundwater using a submersible pump or bailer. The method of purging will be tailored for each site, dependant on the parameters being analysed to ensure the accuracy of results.	this does not happen, reference the field sheet 20-02 year not recorded but samples were submitted 24-02-14				





Deference	Clause	Requirement	Evidence	Oudit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	7.5	Collection of groundwater samples will be undertaken following the purging of each bore.	this does not happen, reference the field sheet 20-02 year not recorded but samples were submitted 24-02-15	Not Compliant	D	2	Medium
Water Management Plan	7.5	For groundwater samples, sampling devices will be dedicated and/or disposable for each sample or otherwise decontaminated between sampling locations. If rinsing is used, rinsate samples should be included in the QA/QC program as appropriate.	This is not clear from the analysis and sampling data provided, there is no record in the QA report from the lab of rinsate samples being tested. The water samped shold be from the aquifer not that exposed to the atmosphere in the well.				
Water Management Plan	7.5	Groundwater samples will be collected in laboratory supplied sampling containers that will be appropriately dosed with the preservative for the analysis required.	This occurs	Compliant			
Water Management Plan	7.5	The samples will be submitted for analysis to a NATA accredited laboratory within the relevant holding times with completed chain of custody documentation.		Compliant			
Water Management Plan	7.5	All sampling events will have a QA/QC program and the QA/QC sample analysis will be checked to validate the integrity of the collected data.	This occurs	Compliant			
	7.5.1 Document of r	nonitoring and reporting on monitoring results					
Water Management Plan	7.5.1	All documentation related to groundwater will be kept on archive, and be readily accessible to facilitate ease of data analysis against regulatory criteria and monitoring trigger values. Depending on the medium for recording of field data, relevant monitoring documentation may include (but not limited to): • Hand recorded field observations and data recording; • Electronic data records and downloaded information; • Calibration records for field monitoring equipment; • Photographs of monitoring sites or potential issues of concern; • Laboratory analytical results reports, including chain of custody records; • Summary and records of quantities of releases of hazardous materials to the environment; • Internal technical memorandums detailing the results of monitoring programs; and • Monitoring reports prepared for submission to regulatory authorities.	· · · ·	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	7.5.1	An organised internal approach to data management and monitoring documentation will significantly enhance the intended benefits of the monitoring program. When set up, it will facilitate the identification of potential issues of concern in a timely manner, such that appropriate contingency actions can be implemented if warranted. In addition to this, the raw monitoring data will be published on the mine's website within two weeks of sampling and will be given to the local NOW hydrogeologist.	The data is not posted up within two weeks and is not supplied to NOW.	Not Compliant Administrative			
	7.6 Groundwater Im	pact Assessment Criteria					
Water Management Plan	7.6	An observed inflow rate 100% in excess of the predicted mean monthly inflow rate at any stage during the mine life sustained for three consecutive months would require a response plan as detailed in Section 8.	This has not been observed.	Not Applicable			
	7.6.2 Impacts to Lice						
Water Management Plan	7.6.2	The groundwater modelling carried out for the EA suggests that all but one of the surrounding bores draw water from the Namoi Alluvium and will experience a maximum draw down of 1m. The one remaining bore is located in the Leard State Forest (GW967859) and draws water from the shallow porous rock source. This bore will experience approximately 8m of drawdown as a result of the TCM and approximately 20m from all surrounding mines. TCM will consult with NOW to ascertain the use of this bore and to agree on any mitigating action, if required.	Noted				
	7.6.3 Mine Inflow W	ater Quality					
Water Management Plan	7.6.3	Should the water quality of the mine inflows or dewatering discharge indicate an inflow salinity of more than 20% above that predicted by the groundwater modelling, all relevant monitoring data will be provided to an approved experienced hydrogeologist for review and assessment of the impact on other users or the environment.	This has not occurred	Not Applicable			
	7.6.4 Groundwater C	Quality Criteria					
Water Management Plan	7.6.4	Groundwater quality would be assessed predominantly against the National Environment Protection Council (NEPC) livestock guidelines, given that this is the predominant use of groundwater in the vicinity of the mine site. These are outlined in Table 7 6. Impacts on the water quality parameters of pH, TDS, other anions and heavy metals (not considered by the NEPC criteria) would be based on comparisons to baseline monitoring of groundwater quality taken from all groundwater bores within the mine site.	AEMR reports against these criteria	Compliant			
Water Management Plan	7.6.4	A trigger to assess the cause and effects on groundwater quality will be implemented when there is a prolonged and extended non-conformance of the outlined criteria at a particular piezometer.	Noted (this has not occurred)				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Fillding	Consequence	Likelihood	Risk
Water Management Plan	7.6.4	If a parameter is outside the designated criteria for at least three months in a sequence, or alternatively, exceeds its previous range of results by greater than a 20% variation for at least three months, then the cause will be investigated, and a remediation strategy proposed, if warranted.	Noted (this has not occurred)				
	7.7 Reporting Procedu	ires					
Water Management Plan	7.7	An active database of monitoring results, to be retained by the Environmental Officer, will be updated on a regular basis. Sampling results will be made available in the Annual Return (where required), as well as on the TCM website via the AEMR/Annual Review and Community Consultative Committee (CCC) report. The AEMR/Annual Review will present an overview of the performance of the groundwater monitoring network during the preceding 12 months and identify the proposed extraction, processing and rehabilitation activities and environmental management planned for the following 12 months.	This is done see AEMR/ARS	Compliant			
Water Management Plan	7.7	The ongoing monitoring program and collation of relevant data will provide the basis for continuing improvement in groundwater management across the site.	Noted				
	7.7.1 Community Com	pplaint Protocol					
Water Management Plan	7.7.1	In addition to monitoring of surface and groundwater quality, TCM records all complaints made by the community. For each complaint, the following information is recorded • Date and time of complaint; • Method by which the complaint was made; • Personal details of the complainant which were provided by the complainant or, if no such details were provided, a note to that effect; • Nature of the complaint; • The action(s) taken by TCM in relation to the complainat, including any follow-up contact with the complainant; and • If no action was taken, the reason why no action was taken.	See complaints from other sections of audit	Compliant			
	7.8 Further Developm	ent of the Groundwater Model		1			
Water Management Plan	7.8	The groundwater model will be recalibrated 2 years after the approval of this Plan. Recalibration will then occur every 5 years thereafter and at least 12 – 18 months prior to cessation of mining. Following the first review, if necessary, the groundwater model will be recalibrated and confirmatory forward impact predictions made.	This has not occurred, Ground water model updated for the EA in January 2012.	Not Compliant	E	2	Medium
Water Management Plan	7.8	Should any review or audit indicate a significant variance from the model predictions with respect to either water quality or groundwater levels, then the implications of such variance will be assessed, and appropriate response actions implemented in consultation with NOW and EPA as appropriate.	Noted, no significant variations to date.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Keierence	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	7.8	If in-flows deviate significantly from predictions, regular review of the groundwater model predictions against monitoring data will be carried out. Should the recalibrated model show groundwater inflows beyond these cases described in the EA, a separate detailed impact assessment will be conducted and mitigating measures determined.	This has not occurred.	Not Applicable			
Water Management Plan	7.8	It is proposed to calibrate this model with ongoing monitoring data from the site. Other circumstances which may trigger further development or refinement of the groundwater model include: • A significant change to the mine plan; • Acquisition of new hydrogeological information, such as groundwater levels and aquifer properties (i.e. hydraulic conductivity) which are different to calibrated values used in the model; and • Groundwater drawdown and inflows which significantly exceed model predictions for that stage of mining.	Noted, none of these circumstances have eventuated.	Not Applicable			
	8 Surface and Groun	dwater Response Plan					
Water Management Plan	8	The surface and ground water response plan will include a protocol for managing and reporting any: • incidents; • complaints; • non-compliances with statutory requirements; and • exceedances of the impact assessment criteria and or performance criteria; and; • a protocol for periodic review of the plan.	There is no Surface and Groundwater Response Plan, there is a TARP but it does not detail the management of complaints.	Not Compliant Administrative			
Water Management Plan	8 8.1 Surface Water	Records of any of these items along with the data collected as outlined in this section are to be maintained and available for review by the appropriate agency/ authorities.	Records are kept	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
Water Management Plan	8.1	If an exceedance of the monitoring criteria for wet weather discharges (listed in Section 6) is identified then Tarrawonga Coal will follow the procedure listed below: • Exceedance in monitoring criteria identified; • Record the timing, location, environmental conditions and any contributing factors to the exceedance; • Advice issued to relevant agencies as soon as practicable; • Sampling point and areas upstream inspected to ascertain cause of exceedance; • Operational practices reviewed to determine if any current operational practice contributed to the exceedance; • Implementation of ameliorative measures on site to minimise the potential for future exceedance, which may include clean out, redesign or alteration to structures and/or operational practice; • Further written advice to relevant agencies identifying actions undertaken to reduce future risk of exceedance, if not included in initial communication; • Where specific cause of exceedance cannot be identified, external advice may be sought; and • Ongoing future monitoring to ensure ameliorative measures have been successful with concentration criteria being met.	This has not occurred in the audit period	Not Applicable			
Water Management Plan	8.1	All other surface water monitoring will be discussed in the AEMR/Annual Review with reference to relevant guidelines, such as the ANZECC guidelines.	This is included in the AEMR	Compliant			
	8.2 Groundwater						
Water Management Plan	8.2	In the event of any exceedance in groundwater monitoring criteria, the following response action may be initiated: • TCM assessment to determine the reason for the exceedance. • Refer the matter to an independent hydrogeologist for review if TCM investigation is unable to identify reason for exceedance. • If assessed as being caused by the mining operation, and it is further assessed to be likely to cause an adverse impact on an existing beneficial or environmental use of surface water or groundwater, then an appropriate preventative and/or remedial strategy would be recommended, which may comprise: o Additional monitoring; o Modification to mine plans; o Provision of alternative water supply or other agreed compensation; o Introduction of engineering structures to break hydraulic coupling caused by mining; or o (If appropriate) no change to operations.	This has not occurred in the audit period	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Plan	8.2	The above response program would be carried out in consultation with EPA, NOW and other relevant government agencies.	Noted				
Water Management Plan	8.2	Specific trigger levels have been designed to alert TCM to observed parameter responses which are outside of normal variation and predicted responses, or where observed parameter values do not follow anticipated trends.	Noted				
Water Management Plan	8.2	The triggers for instigation of response actions would occur when observed changes to monitored parameters exceed specified levels. Such changes in observed parameters or conditions include: • Sudden inrush of groundwater into the mine pit in exceedance of predicted inflows; • Significant change in observed water quality or groundwater levels between sampling rounds; • Changes in trends over an extended period for groundwater levels and quality; and • A significant increase or variation from predicted models.	Noted, covered in the TARP	Compliant			
	8.3 Trigger Action Re	esponse Plan (TARP)					
Water Management Plan	8.3	The Trigger Action Response Plan (TARP) has been developed to focus upon appropriate trigger and response actions for mitigation of impacts to the natural environment as a result of mining.	Noted				
Water Management Plan	8.3	Monitoring serves to advise of changes to water levels or quality that occur, or to raise alert that an abnormal condition relating to mining has developed.	Noted				
Water Management Plan	8.3	Each monitoring program has established trigger levels for particular impacts at which a response is needed, and to help define an appropriate response (Table 8 1). Management of impacts within predictions follows standard assessment review and response protocols.	Noted				
Water Management Plan	8.3	The TARP has been designed to allow reference to risks of impact from mining to environmental aspects identified within the mining area and surrounds. These include both predicted and unpredicted impacts, and include: • Storage dam water quality; • Site discharge water quality; • Creek water quality; • Groundwater level; • Groundwater quality; • Unexpected impacts to alluvium associated with the Namoi River; and • Groundwater users (private bores).					
	8.4 Management Pla	an Review					





Deference	Clause	Requiremen	t	Evidence	Audit Finding		Risk	
Reference	Clause	Liduse		Audit Finding	Consequence	Likelihood	Risk	
Water Management Plan	8.4	be reviewed as par revised to describe implemented over exceedances of wa contingent mitigat	It of the review cycle of this document. It will be any additional measures/procedures that are the life of the project to respond to potential ater related criteria. It would also describe the ion/compensation/offset options that would be nt that surrounding users are adversely	and supporting documents in the coming weeks.	Compliant			
	8.5 Unforseen Impact				•			
	a S	accordance with the Section 5 and Section 5 are groundwater im	the procedure to be followed (in general ne criteria exceedance protocols detailed in ion 7) in the event that any unforseen surface spacts are detected.	Noted, covered in TARP.				
		Stage	Procedure					
Water Management Plan	8.5	Υ.	Review the uniforesien impact including consideration of: • Any relevant mentioning data, and • Current mine activities and land management practices in the relevant curchiment		Compliant			
		2	Commission an investigation by an appropriate specialist into the unforeseen impact, if considered appropriate by the Environmental Specialist.					
		3	Develop appropriate ameliorative measures based on the results of the above investigations, in criniculation with the relevant authorities.					
		4	Implement additional mentioning where relevant to measure the effectiveness of the improvement measures.					





Poforonco	Clause	Requirement	Evidonco	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
	Groundwater Conting						
	2.3 Monitoring Location	ons					
Groundwater Contingency Plan	2.3	The GWMP applies to a total of sixteen locations where both groundwater levels and quality will be measured (see Figure 1). Eight of these locations will comprise specifically installed piezometers, positioned to provide relevant information on groundwater in close proximity to the mine.	Noted				
Groundwater Contingency Plan	2.3	In addition to these eight piezometers, TCPL will monitor groundwater levels within five representative bores on and surrounding the mine site, namely: • GW002501 – within the water bearing zone(s) of the mine site; • GW002129 – within the mine site for as long as this bore exists (prior to destruction by the advancing footprint of the mine); • GW031856 – within the alluvial aquifer on the "Ambardo" property; • GW052266 – within the alluvial aquifer on the "Tarrawonga" property; and • GW000507 – within the Permian coal measures and interburden on IBC's "Merriown" property	Results for these piezos are recorded in the AEMRs	Compliant			
Groundwater Contingency Plan	2.3	In order to monitor for natural variation in groundwater level and quality, the following bores will also be monitored. • GW044997 – within the alluvial aquifer on the "Templemore" property. • GW020432 – within the volcanics aquifer on the "Merriown" property. • GW006013 – within the Permian Coal Measures on the "Bollol Creek Station" property.	2012-13 AEMR indicates that GW044997 is monitored but GW020432 and GW006013 are not recorded, no explanation is given in the text.	Not Compliant	E	2	Low
	2.4 Monitoring Param						
Groundwater Contingency Plan	2.4.1 Baseline Monito	Baseline monitoring of water chemistry (see Table 1) is to be conducted within GW002129 and GW002501 prior to the commencement of mining given the proximity of these to the mine and the predicted direction of groundwater flow. Baseline monitoring of Standing Water Level (SWL), electrical conductivity and pH will be undertaken at the remaining 14 piezometers and groundwater bores. Groundwater levels will be assessed to the nearest 0.01m and all monitoring locations surveyed to AHD so relative levels can be determined.	This was done prior to the audit period	Not Applicable			
Groundwater Contingency Plan	2.4.1 2.4.3 Monitoring of O	For those groundwater bores that are currently a source of water for one or more local properties, the yield will be tested prior to commencement of mining. This yield will reflect the pumping rate at the time of measurement, which may not be the maximum pumping rate, but will provide a reference for comparison should a complaint be received over future bore yields.	Noted, no evidence of the testing was provided and it occurred prior to the audit period.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk	
Groundwater Contingency Plan	2.4.3	Following the establishment of baseline conditions for standing water level (SWL), water chemistry and yield, the sixteen groundwater bores and piezometers will be monitored at varying frequencies to assess changes to groundwater SWL's, water chemistry and yield. Data loggers will be installed in the nested piezometer pair of MW-1 and MW-2 to obtain continuous data on fluctuations in the standing water level in the alluvial and coal measures aquifers. The SWL will be measured quarterly in the remaining fourteen bores and piezometers.	AEMRs confirm this monitoring	Compliant				
Groundwater Contingency Plan	2.4.3	The water chemistry parameters of electrical conductivity (EC), pH and Lead (Pb) will be measured every 6 months as required by Condition M2.1 of EPL 12365 and a complete assessment of water chemistry will be undertaken on a yearly basis to review any changes to assess any trends in groundwater chemistry over time. These parameters will include the following. • Total Petroleum Hydrocarbons (TPH) – these contaminants (typically oils and diesel) will be used during mining.	AEMRs confirm this occurs	Compliant				
Groundwater Contingency Plan	2.4.3	Heavy Metals – some heavy metals may be associated with waste oils. These may include arsenic, cadmium, chromium, nickel, lead (already included in Table 7.1), copper, manganese and zinc.	AEMR confirms these analytes are covered	Compliant				
Groundwater Contingency Plan	2.4.3	Major cations and anions – to assess overall changes in groundwater chemistry	AEMR confirms these analytes are covered	Compliant				
Groundwater Contingency Plan	2.4.3	In addition to the program identified in Table 1, and in the event that a complaint be made over deceasing yields within bores closer to the mine, TCPL will undertake monitoring of the yield within the background bores to establish whether the decrease is a consequence of mine related activities or due to natural fluctuation.	No complaints regarding this issue	Not Applicable				
Groundwater Contingency Plan	2.4.3	Monitoring will continue for a period of up to 10 years after mining has ceased. The frequency will be reassessed after mining is complete, however, as it may be possible, depending on results, to lengthen the intervals between monitoring campaigns.	Noted					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
Groundwater Contingency Plan	3	Table 2 presents the trigger levels for changes in groundwater level and chemistry. The trigger levels will be assessed against a benchmark of the natural conditions which have been and/or are currently being established through the baseline monitoring program, or ANZECC criteria. Table 2 Trigger Levels and Benchmarks Parameter Measure Benchmark* Standing Water Saturated Established baseline +/- natural fluctuation 15% Reduction 15% Reduction EG Established baseline +/- natural fluctuation 15% Increase	Noted				
Groundwater Contingency Plan	3	In the event that routine monitoring indicates that a trigger has been reached or is being approached, TCPL will commission a hydrogeologist to review the data. The outcomes of that review, including any recommendations, will be subject to discussion and agreement with the Department of Natural Resources (DNR) hydrogeologists.	No triggers to date of the GCP	Not Applicable			
Groundwater Contingency Plan	3	A trigger of pH or EC would initially lead to an increase in the analytes monitored and/or frequency of sampling to confirm the magnitude and extent of the change in water chemistry and verify the change is a consequence of mining.	No triggers to date of the GCP	Not Applicable			
Groundwater Contingency Plan	3	Should the saturated thickness trigger level be achieved in any bore, TCPL will notify the affected landowner(s) and, if TCPL's and DNR's hydrogeologists are of the opinion that the reduction is a consequence of mining, initiate the mitigation measures identified in Section 4.2. An independent authority may also be used where a dispute arises as to the cause of the change, given that groundwater supply and quality can be affected by non-mining related factors such as bore siltation, aquifer depletion by large scale agricultural users, bacterial infection, fertilizer contamination etc.	No triggers to date of the GCP	Not Applicable			
		res, Negotiated Agreements and Compensatory Measures					
Groundwater Contingency Plan	4.1 Groundwater Qu 4.1	The only potential major contaminant of groundwater on the mine site will be liquid hydrocarbons such as diesel fuel, oils and lubricants. In the event a major liquid hydrocarbon/contaminant spill does occur, TCPL will implement the following 3-phase mitigation remediation plan.	Noted - no hydrocarbons detected at significant levels to date	Not Applicable			
Groundwater Contingency Plan	4.1	Phase 1 – Recovery (of surface) Recover as much as possible at the source by pumping the contaminant from the surface and excavating the contaminated materials and stockpiling them on site, under cover and on an impermeable surface, eg. a high density polyethylene sheet. This material will later be bioremediated on site and/or transported to an approved waste depot.	Noted - no hydrocarbons detected at significant levels to date	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
Groundwater Contingency Plan	4.1	Phase 2 – Source Control Begin hydraulic control of the source to prevent spreading of contamination. This will involve digging one or more holes close to the centre of the spill area, and pumping from these holes to create a cone of depression with a hydraulic gradient towards the holes. This will prevent movement of contamination away from the area of the spill.	Noted - no hydrocarbons detected at significant levels to date	Not Applicable			
Groundwater Contingency Plan	4.1	Phase 3 – Recovery (from groundwater) If necessary, install boreholes to remove and treat contaminated groundwater. Any contaminated surface water or groundwater recovered will be treated on site prior to release (or use for processing or dust suppression) under an EPA licence.	Noted - no hydrocarbons detected at significant levels to date	Not Applicable			
Groundwater Contingency Plan	4.1	No other potential contaminants will be used on the site in sufficient quantity as to require development of a specific contingency plan. However, any spillages of such materials, eg. explosives, will immediately be cleaned up and disposed of in an appropriate manner approved by DEC.	Noted				
Groundwater Contingency Plan	4.1	Notwithstanding the measures identified above, TCPL recognise that the potential remains for changes in groundwater quality to occur which may or may not be a consequence of potential contaminants used on the mine site. Should such a situation be demonstrated by monitoring, appropriate measures to mitigate impacts on groundwater quality will be developed in consultation with DNR's hydrogeologists, with the nature of the "appropriate solution(s)", eg. pumping and treatment, isolation or remediation, being dependent on the nature of the issue.	Noted				
Groundwater Contingency Plan	4.2	In the event that monitoring identifies a reduction in the saturated thickness and/or bore yield within any bore which is in excess of the identified trigger level and a consequence of mining, TCPL will enter into negotiations with the affected landowners with the intent of formulating an agreement which provides for one or a combination of: • re-establishment of saturated thickness in the affected bore(s) through bore deepening; • establishment of additional bores to provide a yield at least equivalent to the affected bore prior to mining; • provision of access to alternative sources of water; and • monetary compensation to reflect increased water extraction costs (if any), eg. as a consequence of lowering pumps or installation of additional or alternative pumping equipment.	This has not occurred to date	Not Applicable			





Deference	Clause	Requirement	Fuidones	Audit Finding		Risk	
Reference	Clause		Evidence	Audit Finding	Consequence	Likelihood	Risk
Groundwater Contingency Plan	4.2	The proposed mitigation or compensation measures will be developed by TCPL following a review of all available data by a qualified hydrogeologist. Should the affected land owner consider the proposed measures to be inadequate, TCPL will request a review of these by a DNR hydrogeologist and/or the Director-General of the Department of Planning. Any recommendations provided by the Director-General in relation to the proposed mitigation or compensation measures will be reviewed by the TCPL's hydrogeologist and unless considered unreasonable will be incorporated and resubmitted to the affected land owner.	This has not occurred to date	Not Applicable			
Groundwater Contingency Plan	4.2	In the event, the affected land owner still rejects the proposed mitigation or compensation measures, or the recommendations provided by the Director-General are considered unreasonable or unlikely to benefit the affected land owner, TCPL will request the matter be referred to an Independent Dispute Resolution Process, as described under schedule 56 of DA-188-4-2005.	This has not occurred to date	Not Applicable			
	5 Rocumentation and	Record Keeping					
Groundwater Contingency Plan	5	Monitoring undertaken by TCPL personnel, eg. monthly data downloads from the continuous data loggers and quarterly monitoring of standing water levels and chemical parameters within the nominated bores, will also be recorded in an excel spreadsheet and the results graphed against local rainfall.	Spreadsheet provided as evidence, complies	Compliant			
Groundwater Contingency Plan	5	The following information will be recorded at each groundwater monitoring location. • Date and time of SWL monitoring or water sampling. • Condition of the monitoring bore. • Notable ground disturbances or activities ongoing in the general activity. • Information on recent bore pumping (if available). • Colour, odour or other noticeable characteristics of the water.	Included in monitoring field sheets	Compliant			
Groundwater Contingency Plan	5	In the event that pump testing is required from a bore operated on a neighbouring property for domestic or stock purposes, a hydrogeological consultant to TCPL will be commissioned to undertake the test and interpret the results. TCPL will require the results of the test within 1 month of the test.	Not been required to date	Not Applicable			
Groundwater Contingency Plan	5	The nominated Environmental Officer for the Tarrawonga Coal Mine will manage the scheduling, completion and interpretation of results with copies of monitoring results to be forwarded quarterly to the TCPL mine manager and area manager. An up to date copy of the excel spreadsheet in which the monitoring results are maintained is to be kept at both the mine site office and TCPL's Boggabri Office.	This occurs	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Reference	Clause				Consequence	Likelihood	Risk	
Groundwater Contingency Plan	5	The results of all groundwater monitoring will be made publicly available at the offices of Narrabri Shire Council and Gunnedah Shire Council and at TCPL's Boggabri office. These results will be updated at least every three months. Each year, the results of the groundwater monitoring program will be summarised and presented in the AEMR. This will include an analysis of the monitoring results against the trigger levels listed in Table 2, previous monitoring results and the monitoring results of background variance. Based on these results, trends in the groundwater levels, quality and availability will be identified and the requirement to initiate the contingency measures identified in Section 4.		Not Compliant Administrative				





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
		Strategy - March 2013					
	3.3 Water licences an	_ ·					
Water Management Strategy - March 2013	3.3 4.1 Surface water	Table 3.1 below sets out the water access licence (WAL) types and volumes currently held by BCM, TCM and MCC. Table 3.1 Current WAL associated with the BTM Complex nines	Noted				
	4.1.1 Contaminant ex	nort					
Water Management Strategy - March 2013	4.1.1 Contaminant ex	Contaminated water, containing suspended solids and soluble salts, will be generated from coal stockpiles and the mining void, as well as groundwater inflows to the mining void. In addition, dirty water containing suspended solids will be generated from	The water collected from the coal loading area is located in Dam SB4, which then flows through a series of dams to Licensed Discharge Point 1 (LDP1 - EPL 12365) and is discharged off-site.	Not Compliant	D	1	High
		mines, contaminated water will be retained onsite for use, and dirty water will be retained in settlement ponds prior to discharge or on-site use.					
	4.3 Supply and demai	na					





Reference	Clause	Requirement	Evidence	Audit Finding	Risk		
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	4.3	All site water management plans for the BTM Complex mines aim to: - divert clean runoff from undisturbed catchment areas around the mine workings into local creeks ('Nagero Creek' for BCM, 'Nagero', Bollol and Goonbri Creeks for TCM, and Back Creek for MCC) or for use on-site, where appropriate licences are held. Clean water may also be stored temporarily for subsequent controlled discharge into local creeks if immediate diversion is not feasible	This was observed on-site.	Compliant			
Water Management Strategy - March 2013	4.3	- capture dirty water from disturbed areas in sedimentation dams. If the water quality meets licence requirements, and the water is not required for use on site, it will be discharged into the local creeks	Noted				
Water Management Strategy - March 2013	4.3	- use dirty water wherever possible for coal processing and dust suppression	Observed on site	Compliant			
Water Management Strategy - March 2013	4.3	- use imported water as follows:> for BCM, use imported raw water for potable water, vehicle wash down and construction activities, as well as to meet dust suppression and coal processing demands when there is a site water deficit> for TCM, use imported raw water for portable water, and use groundwater from a licensed production bore with an annual entitlement of 50ML during protracted dry periods> for MCC, use imported raw water from an existing high security licence for 3000 ML/yr from the Namoi River for vehicle wash-down, construction activities and potable water, as well as for other site water deficits.	Noted				
	5.2.1 Surface water						
	Surface water flow a	nd use					
Water Management Strategy - March 2013	5.2.1	The objectives of surface water flow and use monitoring are: - to provide baseline surface water flow data upstream of each mining operation over time	Noted				
Water Management Strategy - March 2013	5.2.1	- to record changes in surface water flows downstream of each mine over time	Noted				
Water Management Strategy - March 2013	5.2.1	- to record changes in downstream surface water flows in local creeks resulting from the cumulative impacts of mining operations	Noted				
Water Management Strategy - March 2013	5.2.1	- to record changes in downstream surface water flows in the Namoi river resulting from the cumulative impacts of all BTM Complex mine operations	Noted				
Water Management Strategy - March 2013	5.2.1	- to facilitate surface water hydraulic and/or hydrologic model evolution	Noted				
Water Management Strategy - March 2013	5.2.1	- to allow assessment of surface water access impacts on other users	Noted				
Water Management Strategy - March 2013	5.2.1	- as required by licence conditions, to record surface water volumes pumped from the Namoi River accurately in terms of volume and timing of pumping for input into mine water balances, and to quantify opportunities for water sharing between mines.	Noted				
	surface water quality		I.	<u> </u>			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Addit Fillding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	5.2.1	The objectives of surface water quality monitoring are: - to provide baseline surface water flow data upstream of each mining operation over time for a range of informative and diagnostic parameters	Noted				
Water Management Strategy - March 2013	5.2.1	- to record mining-induced changes in surface water quality in space and time	Noted				
Nater Management Strategy - March 2013	5.2.1	to assess whether any changes in surface water quality with time occur during and after mining, and whether such changes are likely to have a material effect on environmental values.	Noted				
	5.2.2 Groundwater r	nonitoring objectives					
	Groundwater levels						
Water Management Strategy - March 2013	5.2.2	The objectives of groundwater level monitoring are (as outlined by Heritage Computing, 2012): - to provide baseline pre-mining groundwater levels in space and time	Noted				
Water Management Strategy - March 2013	5.2.2	- to quantify natural time variations in groundwater levels	Noted				
Water Management Strategy - March 2013	5.2.2	- to record mining-induced changes in groundwater levels in space and time	Noted				
Nater Management Strategy - March 2013	5.2.2	- to provide a foundation for characterisation of aquifer and aquitard properties by numerical model calibration	Noted				
Water Management Strategy - March 2013	5.2.2	- to facilitate groundwater model evolution through verification of simulated heads against those measured	Noted				
Water Management Strategy - March 2013	5.2.2	- to reveal mining-induced changes in groundwater flow directions and hydraulic gradients	Noted				
Water Management Strategy - March 2013	5.2.2	- to provide evidence for the degree of stream-aquifer interaction, especially losses of stream water, and whether the losses are permanent or temporary	Noted				
Water Management Strategy - March 2013	5.2.2	- to allow assessment of potential impacts on groundwater dependent ecosystems	Noted				
Water Management Strategy - March 2013	5.2.2	- to allow assessment of yield/drawdown impacts on other groundwater users	Noted				
Water Management Strategy - March 2013	5.2.2	- to monitor post-mining rates of groundwater pressure recovery.	Noted				
	Groundwater quality						
Water Management Strategy - March 2013	5.2.2	The objectives of groundwater quality monitoring are (as outlined by Heritage Computing, 2012): - to provide baseline pre-mining groundwater quality data in space and time	Noted				
Water Management Strategy - March 2013	5.2.2	'	Noted				
Vater Management Strategy - March 2013	5.2.2	- to record mining-induced changes in groundwater quality in space and time	Noted				
Nater Management Strategy - March 2013	5.2.2	- to facilitate confirmation or revision of the conceptual model for chemical evolution and groundwater flow directions	Noted				
Nater Management Strategy - March 2013	5.2.2	- to assess whether any changes in groundwater quality with time occur during and after mining, and whether such changes are likely to have a material effect on beneficial uses	Noted				





5.2.2	- in the case of open cut pits, to establish whether enhanced rainfall recharge through backfill provides a freshening effect on groundwater, or instead mobilises latent chemicals - in the case of a water-filled final void, to assess the risk of	Noted	Audit Finding	Consequence	Likelihood	Risk
5.2.2	rainfall recharge through backfill provides a freshening effect on groundwater, or instead mobilises latent chemicals - in the case of a water-filled final void, to assess the risk of	Noted			Ì	
5.2.2						
	migration of saline void waters during the post-mining recovery phase whenever such waters are not contained as a groundwater sink	Noted				
5.2.2	- to assess whether acid rock drainage has occurred	Noted				
5.2.2	- to assess whether tailings dams are leaking.	Noted				
roundwater seepage						
5.2.2	The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs)	Noted				
5.2.2	- to develop a profile of mine inflow rates and variations with time as input to the mine water management systems	Noted				
5.2.2		Noted				
roundwater use						
	The objectives of groundwater usage monitoring are (as outlined by Heritage Computing, 2012): - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use	Noted				
	- to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine	Noted				
5.2.2	- to provide input information into ongoing groundwater model simulations	Noted				
	 if declining water levels in a monitoring bore occur, to allow informed cause-and-effect analysis in establishing whether the cause is mining or abstraction from a bore. 	Noted				
	mplex cumulative impact monitoring networks					
		L	T		T	
	not supersede surface water management plans currently in place. Rather the program has been developed to work in parallel.	Noted				
3	5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 5.2.2 Froposed BTM Coi.1 Surface water	5.2.2 - to assess whether acid rock drainage has occurred 5.2.2 - to assess whether tailings dams are leaking. The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs) 5.2.2 - to develop a profile of mine inflow rates and variations with time as input to the mine water management systems - to facilitate groundwater model evolution through verification of simulated inflows against those measured or estimated. Doundwater use The objectives of groundwater usage monitoring are (as outlined by Heritage Computing, 2012): - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use - to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine - to provide input information into ongoing groundwater model simulations - if declining water levels in a monitoring bore occur, to allow informed cause-and-effect analysis in establishing whether the cause is mining or abstraction from a bore. Proposed BTM Complex cumulative impact monitoring networks 1 Surface water The proposed cumulative impact monitoring networks The monitoring locations selected for the cumulative monitoring network are part of the existing networks of each mine, with the	5.2.2 - to assess whether acid rock drainage has occurred Noted 5.2.2 - to assess whether tailings dams are leaking. Noted 5.2.2 - to assess whether tailings dams are leaking. Noted The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs) 5.2.2 - to develop a profile of mine inflow rates and variations with time as input to the mine water management systems - to facilitate groundwater management systems - to facilitate groundwater management systems 5.2.2 - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use Demonstructure and timing of pumping where dewatering bores are in use 5.2.2 - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use 5.2.2 - to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine 5.2.2 - to provide input information into ongoing groundwater model simulations - if declining water levels in a monitoring bore occur, to allow informed cause-and-effect analysis in establishing whether the cause is mining or abstraction from a bore. Proposed BTM Complex cumulative impact monitoring networks 1. Surface water The proposed cumulative surface water monitoring program does not supersede surface water management plans currently in place. Rather the program has been developed to work in parallel. The monitoring locations selected for the cumulative monitoring network are part of the existing networks of each mine, with the	5.2.2 - to assess whether acid rock drainage has occurred Noted 5.2.2 - to assess whether tailings dams are leaking. Noted The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs) 5.2.2 - to develop a profile of mine inflow rates and variations with time as input to the mine water management systems - to facilitate groundwater model evolution through verification of simulated inflows against those measured or estimated. 5.2.2 - The objectives of groundwater wasge monitoring are (as outlined by Heritage Computing, 2012): - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use - to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine - to provide input information into ongoing groundwater model simulations - if declining water levels in a monitoring bore occur, to allow informed cause-and-effect analysis in establishing whether the cause is mining or abstraction from a bore. Proposed BTM Complex cumulative impace monitoring program does in outsupersed surface water management plans currently in place. Rather the program has been developed to work in parallel. The monitoring locations selected for the cumulative monitoring networks are part of the existing networks of each mine, with the	5.2.2 - to assess whether acid rock drainage has occurred Noted 5.2.2 - to assess whether tailings dams are leaking. Noted 5.2.2 - to assess whether tailings dams are leaking. Noted Dundwater seepage The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs) 5.2.2 in the develop a profile of mine inflow rates and variations with time as input to the mine water management systems - to facilitate groundwater management systems - to facilitate groundwater management systems - to facilitate groundwater usage monitoring are (as outlined by Heritage Computing, 2012): - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use - to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use - to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine - to record pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine - to provide input information into ongoing groundwater model simulations - if declining water levels in a monitoring bore occur, to allow informed cause and effect analysis in establishing whether the cause is mining or abstraction from a bore. - Proposed BTM Complex cumulative impact monitoring program has been developed to work in parallel. The monitoring locations selected for the cumulative monitoring in etworks are part of the existing networks are attention. - The monitoring locations selected for the cumulative monitoring networks are part of the existing networks are attention.	5.2.2 -to assess whether acid rock drainage has occurred Noted 5.2.2 -to assess whether tailings dams are leaking. Noted 5.2.2 -to assess whether tailings dams are leaking. Noted outwinder seepage The objectives of groundwater seepage monitoring are (as outlined by Heritage Computing, 2012): - to accurately record water volumes pumped from open cut pits, in terms of volume and timing of pumping, so that an assessment can be made of groundwater seepage through water balance modelling (allowing for evaporative losses and surface water inputs) 5.2.2 -to develop a profile of mine inflow rates and variations with time as input to the mine water management systems -to facilitate groundwater model evolution through verification of simulated inflows against those measured or estimated. 5.2.2 -to record pumped groundwater usage monitoring are (as outlined by Heritage Computing, 2012): 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping where dewatering bores are in use -to tarcord pumped groundwater volumes accurately in terms of volume and timing of pumping from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in from production bores operated or owned by the mine 5.2.2 -to record pumped groundwater volumes accurately in terms of volume and timing of pumping in to pumping to t





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	5.3.1	The monitoring parameters and frequencies proposed for the cumulative surface water monitoring network are preliminary and draft in nature. The proposed cumulative monitoring program will form part of discussions with NOW and each mine as part of the consultation process required in the Project Approval. Table 5.3 Proposed cumulative surface water monitoring program	Noted				
	5.3.2 Groundwater						
	Groundwater levels a		I		ı		
Water Management Strategy - March 2013	5.3.2	A program has been developed by Heritage Computing (2012) to monitor and manage the cumulative impacts on groundwater conditions as a result of mining within the BTM Complex.	Noted				
Water Management Strategy - March 2013	5.3.2	to monitor regional groundwater conditions, in addition to existing monitoring bore networks (see Table 5.2 in Section 5.1.2). Some sites selected for groundwater quality monitoring are aligned with existing bores that monitor alluvial aquifers, with the new bores placed to monitor underlying aquifers at the same location. Some sites are selected for water level monitoring only. All bores are recommended to have automatic dataloggers installed to collect continuous water level data, which can be downloaded during sampling rounds.	Noted				
	Groundwater seepag	e and use					
Water Management Strategy - March 2013	5.3.2	Monitoring of water accumulating in open cut pits is proposed across the BTM Complex to provide a dataset for periodic water balance modelling.	Noted.				
	6.1 Existing trigger lev	vels and responses					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Filiding	Consequence	Likelihood	Risk
	6.1.1 Surface water						
Water Management Strategy - March 2013	6.1.1	A trigger event is also considered to occur when water quality between SW2 (upstream) and SW1 (downstream) differ by more than 20%. The intent of the variance check is to ensure trigger events occur where significant changes in water chemistry are evident over a short section of the creek line even though the actual stated parameter may be exceeded (Parsons Brinckerhoff, 2012). Table 6.1 Boggabri surface water triggers Parameter Discharge quality	Noted, note the table has been formulated fro Boggabri and may not be applicable for all three sites.				
Water Management Strategy - March 2013	6.1.1	and relate to discharge water quality. These concentration levels are the same as those included in the Boggabri EPL 12407, listed above in in Table 6.1.	Noted				
Water Management Strategy - March 2013	6.1.1	The Tarrawonga Surface Water Management Plan (Whitehaven, 2011) also includes a triggering EC value of 4000 μ S/cm, based on the ANZECC (2000) concentration for moderately tolerant crops.	Noted				
Water Management Strategy - March 2013	6.1.1 Response Plans	Surface water flow monitoring is not part of the current TCM Surface Water Management Plan (Whitehaven Coal, 2011), and has not specifically been recommended for Maules Creek in the EA (WRM, 2011). TCM is updating its WMP to include monitoring of surface water flow in Goonbri Creek as a means of identifying the contribution of surface water discharge that TCM makes to stream flows in Goonbri Creek during significant rainfall events.	Noted				





Deference	Classa	Requirement	Evidence	A coding to altern		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Vater Management Strategy - March 2013	6.1.1	The surface water response plan for TCM is generally consistent with the overall BTM Complex surface water response strategy (outlined in Section 0). Key features of the TCM surface water response plan are: - exceedance in criteria identified - record of timing, location, conditions and contributing factors taken - advice issued to regulatory authorities as soon as possible - sampling points and areas upstream investigated to ascertain cause - operational practices reviewed to determine if any operational contribution to cause - implementation of ameliorative measures on site to minimise potential for future occurrences - written advice issued to regulatory authorities confirming actions taken to reduce potential for future incidences - ongoing monitoring to confirm return to expected parameters.	Noted				
Vater Management Strategy - March 2013	6.1.1	In the event that any unforeseen surface water impacts are detected the aforementioned protocol will extend to include the following steps: Review the unforeseen impact, including consideration of: > Any relevant monitoring data; and > Current mine activities and land management practices in the relevant catchment - Commission an investigation into the unforeseen impact by an appropriate specialist selected in consultation with appropriate regulatory authorities - Develop appropriate ameliorative measures based on the results of the above investigations, in consultation with the relevant authorities - Implement additional monitoring where relevant to measure the effectiveness of the ameliorative measures.					
	.1.2 Groundwater	The area of the area of the area included.	1	ı			
Т	rigger levels						





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.1.2	The Tarrawonga Groundwater Contingency Plan (Corkery & Co, 2006) specifies triggers for groundwater quality, established through baseline information and a review of natural fluctuation. The triggers are listed in Table 6.3. A revision to the Groundwater Contingency Plan is currently being prepared for TCM, and this will be revised based on the approval conditions for the extension of TCM. Table 6.3 TCM groundwater quality triggers Parameter Trigger level pH 15% increase E0 15% increase or decrease Lead 0.1 mgL* a. Based on orteria for livestock watering from ANZECC (2000)	Noted				
	Groundwater levels	<u></u>					
Water Management Strategy - March 2013	6.1.2	The Tarrawonga Groundwater Contingency Plan (Corkery & Co, 2006) sets a trigger for groundwater levels at 15% reduction, based on baseline information and a review of natural fluctuation.	Noted				
Water Management Strategy - March 2013	6.1.2	To counteract spurious measurements, which could occur for example during maintenance of a sensor or downloading or water sampling, a 7-day average will be calculated to cover such events. In addition, to ensure the "breach" of a trigger is sustained and is therefore significant, a 1-month exceedance duration will be adopted to allow water levels to stabilise. This would "trigger" an investigation in the first instance, not an immediately reportable incident.					
	Response Plans		<u>, </u>				
Water Management Strategy - March 2013	6.1.2	Corkery & Co (2006) outlines TCM's groundwater response plan. In summary, the plan dictates that if monitoring indicates that a trigger has been reached or is being approached, Tarrawonga will commission a hydrogeologist to review the data. The outcome of that review, including any recommendations, will be subject to discussion and agreement with NOW. If water quality triggers are exceeded, additional water quality monitoring is to be carried out to determine if the change is a consequence of mining.	Noted				
Water Management Strategy - March 2013	6.1.2	If an oil/chemical spill occurs, the response strategy is summarised as follows: - recover as much of the spill as possible - control the source through pumping - (if necessary) install boreholes to remove and treat contaminated groundwater.	See response in WMP, no major spills	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.1.2	If monitoring identifies a reduction in aquifer saturated thickness and/or bore yield, the response strategy is as follows: - re-establishment of saturated thickness in the affected bore(s) through bore deepening - establishment of additional bores to provide a yield at least equivalent to the affected bore prior to mining - provision of access to alternative sources of water - monetary compensation to reflect increased water extraction costs (if any), e.g. as a consequence of lowering pumps or installation of additional or alternative pumping equipment.	This has not occurred - noted.	Not Applicable			
	6.2 Cumulative impac	t management objectives and triggers					
	6.2.1 Surface water						
	Surface water quanity	у					
Water Management Strategy - March 2013	6.2.1	The surface water quantity objectives adopted by this strategy are as follows, and have been adapted from the objectives contained in the WSP for the Namoi Unregulated and Alluvial Water Sources 2012, and from the Namoi Catchment Action Plan: to protect, preserve, maintain and enhance the important river flow dependent ecosystems	Noted				
Water Management Strategy - March 2013	6.2.1	- to ensure mine water use does not result in average surface water flow in local creeks being less than 66% of natural (pre- development) condition, with a sensitivity to natural frequency and duration	Noted				
Water Management Strategy - March 2013	6.2.1	- to minimise mine impacts on basic landholder rights, including:> landholder extractions from local creeks> flows to harvestable rights stock watering dams	Noted				
Water Management Strategy - March 2013	6.2.1	- to ensure mine water use does not result in the total water use within the Maules Creek unregulated river water source exceeding the long term average annual extraction limit identified for that water source in the WSP for the Namoi Unregulated and Alluvial Water Sources	Noted				
Water Management Strategy - March 2013	6.2.1	- to minimise the impacts of altered flood flows on catchment landholders	Noted				
Water Management Strategy - March 2013	6.2.1	- to provide opportunities for enhanced market based trading of access licences and water allocations between BTM Complex mines	Noted				
Water Management Strategy - March 2013	6.2.1	- to contribute to the maintenance of water quality in the local creeks and the Namoi River.	Noted				
	6.2.2 Groundwater						
	Groundwater quality			1	-		
Water Management Strategy - March 2013	6.2.2	The groundwater quality objectives of this strategy are: - to maintain the most sensitive identified beneficial use (or EV) of all groundwater systems potentially affected by the BTM Complex operations, consistent with the NSW State Groundwater Quality Protection Policy	Noted				





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.2.2	- within this, to maintain the annual average EC values within the historical 95th percentile.	Noted				
	Groundwater quantity						
Water Management Strategy - March 2013	6.2.2	The groundwater quantity objectives adopted by this strategy are as follows, and have been adapted from the objectives contained in the WSPs for the Upper and Lower Namoi Groundwater Sources, and the Murray-Darling Basin Porous Rock Groundwater Sources, as well as from the Namoi Catchment Action Plan: - to protect the structural integrity of the alluvial aquifers by ensuring mining activities and their groundwater extraction do not result in more than minimal alluvial aquifer compaction, aquitard compaction or land subsidence					
Water Management Strategy - March 2013	6.2.2	- to account for all BTM Complex groundwater use and quantity impacts through the statutory licensing and allocation systems	Noted				
Water Management Strategy - March 2013	6.2.2	- to ensure there are no long-term declines in alluvial groundwater levels resulting from BTM Complex mining activities and their groundwater extraction	Noted				
Water Management Strategy - March 2013	6.2.2	- to maintain basic landholder rights access to groundwater sources that may be impacted by the BTM Complex mines	Noted				
Water Management Strategy - March 2013	6.2.2	- to identify and offset any unacceptable interference to irrigation bore supply in groundwater sources that may be impacted by the BTM Complex mines	Noted				
Water Management Strategy - March 2013	6.2.2	- to provide opportunities for enhanced market based trading of groundwater access licences and groundwater allocations between BTM Complex mines.	Noted				
Water Management Strategy - March 2013	6.2.2	- to minimise the impact of changes to groundwater levels and flows resulting from BTM Complex activities on groundwater dependent ecosystems, including vegetation	Noted				
		mplex cumulative impact mitigation					
	6.3.1 Surface water						
	Trigger levels Groundwater quantity	,					
	Groundwater quantity	Proposed trigger levels for ambient surface water quality are	Noted				
Water Management Strategy - March 2013	6.3.1	based on the WQOs, ANZECC (2000) guidelines and available ambient data for the BTM Complex, as outlined in Table 6.5. Table 6.5 sets out the concentrations of key water quality parameters given in each of the WQOs, ANZECC (2000) guidelines and available ambient data. The proposed trigger values are based on the cumulative management objectives discussed in Section 6.2.1. It is proposed that the ambient surface water trigger levels be reviewed once further data becomes available with continued monitoring.					





Reference	Clause		Requi	rement			Evidence	Audit Finding		Risk	
	Clause							Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.3.1	Parameter Total Phosphorus (upt.) Total Niropen (upt.) Total Niropen (upt.) Trutidity (NTU) EC (upt.om) Dissolved oxygen (% saturation) pH Total suspended seldes (mg.t.) Nerate (mg.t.) Reactive phosphorus (mg.t.) Aluminum (upt.) Arsenic (As 19 (upt.) Arsenic (As 19 (upt.) Codemium (upt.) Codemium (upt.) Copper (upt.) Ivon (upt.) Lead (upt.) Lead (upt.) Manganese (upt.) Marquese (upt.) Marquese (upt.) Marquese (upt.)	250 2-255 3-3560 90-110 6-5-5-0 6-5-5-0 NA NA Refer to ANZECC	AMERICO 120 25	distorical application (1) 10-300 (10-	Proposed trigger 110-360 110-360 110-360 150-360	Noted				
Water Management Strategy - March 2013	6.3.1	Proposed triggers for responses to achieve in Section 6.2.1 are si in Section	the objection of the control of the	ives describ ble 6.6. r quantity triggers sonale Intribute to maintainin Maules Creek urree pin the identified intribute to maintainin makes Creek urree pin the impacts or coal catchments simise the impacts of chment landholders	g total water us g total water us g total water us stated fiver water and dom altered flood fic	s within or source nnual sestic rights ws on t based scattors,	Noted				
	6.3.2 Groundwater Triggers										





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.3.2	Groundwater quality triggers for the BTM Complex are set out in Heritage Computing (2012), and relate to the measured EC values in proposed monitoring bores Reg1, Reg2, Reg3, Reg6 and Reg7 (see Table 5.1 and Figure 3.2). A response is triggered is when the annual average EC value exceeds the historical 95th percentile.	Noted				
	Groundwater levels						
Water Management Strategy - March 2013	6.3.2	Heritage Computing (2012) has developed groundwater level triggers for the BTM Complex to detect potential regional impacts to the hydrogeological environment as a result of the BTM Complex. These triggers cover groundwater levels in the alluvial aquifers and Maules Creek Formation (hard rock aquifer).	Noted				
Water Management Strategy - March 2013	6.3.2	In addition to the above triggers proposed by Heritage Computing, the following are recommended: - Groundwater take or volumetric impacts not accounted for within the long-term average extraction limit of the applicable Groundwater Sharing Plan Complaints regarding groundwater stock and domestic or irrigation supply impacts	Noted				
	Responses						
	Groundwater quality						
Water Management Strategy - March 2013	6.3.2	Heritage Computing (2012) has proposed the following response to groundwater quality triggering events. If groundwater triggers are exceeded, measured values are to be compared between sites. If the cause of cannot be directly attributed to natural seasonal variations, a groundwater specialist is to be engaged to determine the reason for the exceedance, and advise on corrective action.	Noted				
	Groundwater levels						
Water Management Strategy - March 2013	6.3.2	Site specific groundwater trigger levels for groundwater levels are set out in the respective water management plans for each operation.	Noted				
Water Management Strategy - March 2013	6.3.2	In the event that trigger levels are exceeded, preventative actions will be identified, communicated and agreed on with proponents of the BTM complex. Actions will likely occur in the following sequence: - Compare water levels to control site to determine if the cause cannot be directly attributed to natural seasonal variations,	Noted				
Water Management Strategy - March 2013	6.3.2	- Engage the services of a groundwater specialist to ascertain cause for the decline in water level,	Noted				
Water Management Strategy - March 2013	6.3.2	- If deemed that activities of the BTM complex are contributing to the decline in water level, potential impacts on groundwater dependent ecosystems are to be assessed in accordance with the BTM Complex biodiversity strategy, and appropriate groundwater management responses developed in consultation with NOW,	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	6.3.2	- An action plan to reduce the impact will be developed in consultation with NOW, with additional monitoring implemented as necessary.	Noted				
Water Management Strategy - March 2013	6.3.2	- Reporting of incidents and responses will form part of the Annual Environmental Management Report.	Noted				
Water Management Strategy - March 2013	6.3.2	In addition: - If groundwater take or volumetric impacts not accounted for within the long-term average extraction limit of the applicable Groundwater Sharing Plan, then: - The relevant mine should report status to BTM Complex, and the overall precinct status should be determined. - NOW should be consulted to determine the need to offset volumetric impacts, and mechanisms for such. - An assessment of market opportunities within the applicable groundwater source should be undertaken	Noted				
Water Management Strategy - March 2013	6.3.2	If there is a complaint regarding groundwater stock and domestic or irrigation supply impacts: - It should be reported to the BTM Complex. - The complaint should be investigated both technically, and with landholder/s. - If impacts are verified, landholders should be compensated, consistent with an agreed BTM Complex landholder compensation 4 policy and strategy.					
	7 Modelling	, , ,					
	7.1 Existing models						
	7.1.1 Surface water						
Water Management Strategy - March 2013	6.3.2	Hydrologic and hydraulic modelling has also been carried out for the TCM by Gilbert & Associates (2011). The purpose of the modelling was to design the realignment of Goonbri Creek around the eastern edge of the proposed extended open cut pit. Hydrological modelling was carried out using RORB software, to predict peak design flows for the 2, 20 and 100 ARIs and the Probable Maximum Flood (PMF). 10 hydraulic modelling was carried out using HEC-RAS software to assess the viability of the channel design.	Noted, the Goonbri diversion is unlikely to happen in the near future and this work is unlikely to continue in the medium term.	Not Applicable			
Water Management Strategy - March 2013	6.3.2	Gilbert & Associates (2011) recommend further hydraulic modelling be carried out using additional survey data, to model the hydraulic characteristics of Goonbri Creek down to Bollol Creek, and downstream to Barbers Lagoon and the Slush Holes. The results of the recommended additional modelling could then be used to provide a more accurate baseline characterisation of existing conditions in Goonbri Creek and Bollol Creek in the final design of the realigned section of Goonbri Creek. Recommendations of the report (among others) include the placement of flow gauging stations on Goonbri Creek to verify and calibrate the completed models, and to assist in performance evaluation during and post construction.	Noted				





Reference Clause Note No	Reference	Clause	Requirement	Evidence	Audit Einding		Risk	
Water Management Strategy March 2013 Water Management Strategy To a cheeve the objectives set out above, it is proposed that a design by design the management proposed that a design by design the management proposed that a design the management proposed that the review who the individual mines and contact proposed that a design the management proposed that the review was been used to scope the need for the management proposed that the review was been used to a charment surface water model (incorporating groundwater baseflows) for the BTM Complex countables water design the management proposed	Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy- March 2013 Water Management Strategy- March 2014 Water Management Strategy- March 2015 Water Management Strategy- March 2015 Water Management Strategy- March 2015 Water Management S		7.2 Cumulative mode	Illing objectives					
Water Management Strategy- March 2013 Water Management Strategy- March 2013 To achieve the objectives set out above, it is proposed that a detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, timing, depth and velocities throughout the BTM Complex. Water Management Strategy- March 2013 Water Management Strategy- March 2013 It is recommended that the review also be used to scope the need for, and practicality and efficacy of, a "whole of catchment" and efficacy of, a "whole of catchment water behaviour from the BTM Complex cumulatively and consistently """. """ one accurately predict required ileance volumes - attribute surface water impacts to individual mines used to be used to: """ or provide a quantitative basis for complaint verification and """ or provided a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a quantitative basis for complaint verification and """ or provide a qua		7.2	understanding of the cumulative behaviour of surface and groundwater resources in the BTM Complex area, and as such: 1. to estimate quantitatively the cumulative impacts from the BTM Complex on groundwater and surface water resources, so as to determine appropriate management responses (e.g. licence acquisition, compensatory measures for affected landholders,					
water Management Strategy - March 2013 7.3 Proposed BTM Complex modelling 7.3.1 Surface water To achieve the objectives set out above, it is proposed that a detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, triming, depth and velocities throughout the BTM Complex. Water Management Strategy - March 2013 Water Management Strategy - March 2013 Water Management Strategy - Arabic - Complex cumulative flow behaviour has been incorporated, in terms of flow distribution, triming, depth and velocities throughout the BTM Complex area and downstream. The models should be assessed for their capabilities and limitations with respect to prediction of cumulative surface water impacts as a result of the BTM Complex. It is recommended that the review also be used to scope the need for, and practicality and efficacy of, a "whole of catchment" surface water model (incorporating groundwater baseflows) for the BTM Complex. A whole of catchment of the BTM Complex. A whole of catchment of the BTM Complex. A whole of catchment surface water model (incorporating groundwater baseflows) for the BTM Complex and water behaviour from the BTM Complex accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and	•	7.2	order to determine appropriate responsibilities for management	Noted				
To achieve the objectives set out above, it is proposed that a detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, timing, depth and velocities throughout the BTM Complex area and downstream. The models should be assessed for their capabilities and limitations with respect to prediction of cumulative surface water impacts as a result of the BTM Complex. It is recommended that the review also be used to scope the need for, and practicality and efficacy of, a 'whole of catchment' surface water basellows) for the BTM Complex. A whole of catchment model may be used to: - predict changes to surface water behaviour from the BTM Complex caudatively and consistently - more accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and	March 2013		operations through evaluation of design hydraulic behaviour, mine inflows and groundwater drawdown magnitude/extent, with this information feeding back into the above management responses.					
To achieve the objectives set out above, it is proposed that a detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, timing, depth and velocities throughout the BTM Complex area and downstream. The models should be assessed for their capabilities and limitations with respect to prediction of cumulative surface water impacts as a result of the BTM Complex. It is recommended that the review also be used to scope the need for, and practicality and efficacy of, a 'whole of catchment' surface water model (incorporating groundwater baseflows) for the BTM Complex. A whole of catchment model may be used to: - predict changes to surface water behaviour from the BTM Complex cumulatively and consistently - more accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and		7.3 Proposed BTM Co	omplex modelling					
detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, timing, depth and velocities throughout the BTM Complex area and downstream. The models should be assessed for their capabilities and limitations with respect to prediction of cumulative surface water impacts as a result of the BTM Complex. It is recommended that the review also be used to scope the need for, and practicality and efficacy of, a "whole of catchment" surface water model (incorporating groundwater baseflows) for the BTM Complex. A whole of catchment model may be used to: - predict changes to surface water behaviour from the BTM Complex cumulatively and consistently - more accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and		7.3.1 Surface water						
for, and practicality and efficacy of, a 'whole of catchment' surface water model (incorporating groundwater baseflows) for the BTM Complex. A whole of catchment model may be used to: - predict changes to surface water behaviour from the BTM Complex cumulatively and consistently - more accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and		7.3.1	detailed review of the individual mine surface water models be undertaken to determine if and how cumulative flow behaviour has been incorporated, in terms of flow distribution, timing, depth and velocities throughout the BTM Complex area and downstream. The models should be assessed for their capabilities and limitations with respect to prediction of cumulative surface	is approved the revision will go ahead.	Not Applicable			
	•	7.3.1	for, and practicality and efficacy of, a 'whole of catchment' surface water model (incorporating groundwater baseflows) for the BTM Complex. A whole of catchment model may be used to: - predict changes to surface water behaviour from the BTM Complex cumulatively and consistently - more accurately predict required licence volumes - attribute surface water impacts to individual mines and direct mitigation measures - provide a quantitative basis for complaint verification and	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	7.3.2	The following recommendations of Heritage Computing (2012) in relation to BTM Complex groundwater modelling are adopted for this strategy: - The MCC and TCM models be maintained separately, as they give similar predictions for the one metre drawdown extent. Both models are at numerical stability limits, and expanding their functionality might be counteractive. - Each of the BCM, TCM and MCC models undergo regular maintenance and recalibration as additional data on groundwater responses to progressive mining improves the understanding of the groundwater systems.	Noted, note that the TC GW model has not been updated since the EA and is now overdue to be revised.	Compliant			
	8 Complaint Manager						
Water Management Strategy - March 2013	8	It is recognised that BCM and TCM have their own well developed complaint management systems, including hotlines, and that these will continue to operate independently. MCC will also establish its own protocols. This strategy supports the use of the existing mechanisms, with individual mining operations investigating complaints that are raised with them. If the investigating mine considers the complaint to be potentially related to a cumulative impact, it will: - seek comments from other mines on the complaint/issue - refer any draft response to the complaint to other mines for comment - resolve any differences prior to finalising response - log the complaint and its resolution on a cumulative impact response register.	See complaints section elsewhere in the audit	Compliant			
	9 Implementation						
Water Management Strategy - March 2013	9	This strategy is one of a number of cumulative environmental impact management strategies being developed in response to current, and draft mine development approval conditions. As such, the following will be established outside of this strategy and will apply to this and all other cumulative environmental impact management strategies: - governance and communication protocols - data sharing protocols and/or shared databases - reporting procedures, formats and frequencies.	Noted				
	9.1 Summary of strate	egic actions					
Water Management Strategy - March 2013	9.1	Specific groundwater data management recommendations have also been made by Heritage Computing (2012), as follows - Groundwater monitoring data from the three sites should be stored in a central data repository, available for use by each site. The repository should hold data from the regional monitoring network and from the individual mine monitoring networks.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Water Management Strategy - March 2013	9.1	- Current site data management is through Excel spreadsheet software rather than database software. While this is workable for a single mine, and easy for mine personnel to maintain and interrogate, there would be an advantage in standardisation of data formats and data management software as part of a cumulative monitoring strategy. Relational database software is recommended for adoption, preferably a system that is tailored to groundwater data and includes automatic quality control procedures and automatic graphics and report production. It should be able to import directly from field equipment formats (e.g. data logger records) and chemical laboratory templates, and should support linkages to common GIS and graphics software packages.	Noted				
	10 Document control						
Water Management Strategy - March 2013	10	This BTM Complex Water Management Strategy has been developed with the input of representatives of BCM, TCM, MCC and Parsons Brinckerhoff.	Noted				
Water Management Strategy - March 2013	10	This Strategy, its operation and implementation, will be reviewed and revised at least every two years or: - in response to strategic actions set out in the Plan - to incorporate proposed new or expanded mining operations at an early stage - on an 'as required' basis to incorporate improvements identified by BTM Complex, or - as required by government agencies.		Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
		diversity Management Plan					
		ement within the Project Area					
		ality of Existing Vegetation and Fauna Habitat					
	4.1.2 Management St	Land clearing for the Project will be undertaken progressively.	There were no clearing activities in progress				T
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	Areas to be cleared will be delineated, restricting clearing to the minimum area necessary to undertake the approved activities.	at the site during the inspection. Some had been recently completed. It was not possible to verify on the ground that delineation had taken place. It would bebetter practise to control ground disturbance by a permit system.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	A seed collection and propagation program will be implemented so that areas to be revegetated are genetically comparable with those in the surrounding woodland areas.	There is no seed collection program at Tarrawonga but the Biobank site has a seed collection program and the offset site will have a seed collection program. The ecologist in the audit team indicated that the distance between the seed collection area and the site could be significant in terms of the risk of genetic variation, local seed should be sought.	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	The faunal value of farm dams surrounding the Project area will be improved by: o excluding livestock grazing planting a range of submerged and fringing vegetation (rushes) o placing a partially submerged log in the dam for use by a variety of fauna o placing a log pile for refuge habitat on the dam shores o limited woodland tree plantings, including some shrub species.	This has not been done as yet.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL.	Noted, no work on this to date.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	Felled timber that is not mulched and salvaged during land clearing will be replaced on topsoiled areas to provide habitat for native reptiles.	This has been observed in the rehabilitation areas on the site.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	A Farm Management Plan will be developed by WCL.	This has not been developed yet.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	Implement weed control measures (see Section 4.7 below) to prevent the establishment and spread of weeds throughout the mining tenement.	Some weed control was conducted.	Compliant			





Defenses	Classes	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.2	All revegetation works will be undertaken in accordance with the Rehabilitation Management Plan.	No works were observed that did not comply with the rehab management plan. Revegetation quality could be improved, some trials have been conducted, further trials should be implemented. Recommendation - conduct further rehbilitation trials to research optimal rehabilitation results.	Compliant			
	4.1.3 Performance / C	•					
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.3	The progress and effects of measures to enhance the quality of existing vegetation and fauna habitat are monitored and reported annually as part of the Annual Environmental Management Report (AEMR)/Annual Review.	Included in the AEMR	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.3	Commencement of revegetation in the riparian zone along 3.1 km of Goonbri Creek south of the Project area.	Not commenced, Goonbri Creek diversion postponed	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.3	Adherence to Farm Management Plan.	Noted. Plan not developed yet	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.1.3	A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval by the Director-General. An implementation plan will also be provided for approval at the same time, to ensure delivery of suitable areas of viable habitat for the nominated species (being Speckled Warbler, Brown Treecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittella, Turquoise Parrot, Masked Own, Yellow-bellied Sheath Tail Bat and Squirrel Glider). This plan will then be incorporated into the revised Biodiversity Management Plan.	Incorporated into the next revision of the Biodiversity MP, not yet acted on.	Not Applicable			
		/egetation and Fauna Habitat					
	4.2.2 Management St						
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	A Rehabilitation Management Plan (RMP) will be developed and implemented in accordance with Schedule 3, Condition 64 of the Project Approval. Ecosystem establishment within the rehabilitation area will take place according to the measures described in the RMP.	Completed and submitted	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	The disturbance areas will be progressively rehabilitated and revegetated to either native bushland and/or agricultural land.	The MOP details this.	Compliant			





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	The rehabilitated areas shall be protected by the following measures: o Establishment and maintenance of perimeter fencing around the active mine site and rehabilitated areas to exclude stock. o Installation of biodegradable plastic or cardboard tree guards around the planted seedlings to protect against wind and cold and rabbit/hare grazing (if required). The necessity for tree guards will be determined during each tree planting campaign and based on likely impacts of grazing. o Temporary fencing around tree lots on the re-created agricultural land. Fencing will be removed at the completion of mining or once the trees attain a height of 4 - 5 m and are not liable to damage from stock. o Permanent fencing of native bushland re-establishment areas to prevent stock access. o Exotic vertebrate pest control (see Section 4.7). o Fire protection (see Section 4.10).	All of these measures were observed on site except the tree protection measures (hares and Rabbits), these were deemed unneccessary. The agricultural land has not yet been established. Cattle are kept off the site by boundary fencing.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	Revegetation of woodland/forest areas would include the planting of species characteristic of the local vegetation communities, including species from the Box-Gum Woodland EEC/CEEC (e.g. White Box overstorey as well as appropriate understorey). Other species would include Bulloak (Allocasuarina luehmannii) and Belah (Casuarina cristata).	This is conducted.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	Irrigation shall be used to promote revegetation where practically possible.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	A Goonbri Creek Management Plan will be developed and implemented to describe the measures to revegetate the permanent Goonbri Creek alignment, including the revegetation of the low flow channel (within and approximately 40 m either side) using species characteristic of the Bracteate Honeymyrtle (Melaleuca bracteata) community. The design of the creek alignment will include in-stream and riparian habitats including the establishment of a pool-riffle system and wetland spaces.	Not commenced, Goonbri Creek diversion postponed	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL.	Noted, no work on this to date.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.2	Implement weed control measures (see Section 4.7) to prevent the establishment and spread of weeds throughout the mining tenement.	Conducted internally, no extensive spread of weeds apart from skeleton weed (turnip weed) in the rehab. Note that the skeleton weed is generated from the topsoil seedbank.	Compliant			
	4.2.3 Performance / 0	Completion Criteria					
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Native vegetation species associations are established in accordance with the approved species mix and tubestock and seed planting density specifications.	This occurs though density has been reduced by attrition, this is being corrected by interplanting with tubestock when conditions are suitable.	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Vegetation health is within the range of analogue sites.	Rehab is not advanced enough to assess this requirement.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Minimum tree height and girth standards for selected indicator species of the vegetation association are within the range of analogue site benchmarks at 1, 5 and 15 years.	Site rehab establishment is poor in some areas and better established in others. The scope of the audit does not allow a full rehabilitation survey to be conducted. The annual rehabilitation review (perviously conducted by EcoLogical) addresses this requirement.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Canopy cover is within the range of analogue sites for the vegetation association at 1, 5 and 15 years.	There is no canopy, to be determined in future surveys	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Species are capable of setting viable seed, flowering or otherwise reproducing.	Not yet advanced enough	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Vegetation develops and maintains a litter layer evidenced by a consistent mass and depth of litter over subsequent seasons	Not yet advanced enough	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Rehabilitation of 752 ha of the woodland/forest post-mine landform.	Not yet advanced enough	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.2.3	Performance criteria are provided in Table 4-1. Table 4-1: Performance criteria for restoration of native vegetation and fauna habitat (TCPL 2012) TIMING PERFORMANCE CRITERIA Long-term (greater than 12 years) To final landform and revegetation program will provide for approximately 752 has of native vocalized from the providence of the providence o	Noted				
	4.3 Maximising the S						
	4.3.2 Management S	-					
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Soil resource management will be undertaken in accordance with measures provided in the RMP.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Identify and quantify potential soil resources for rehabilitation.	Soil stripping map done for the EA provides minimum depths then any extra depths are assessed through observation of the stripping operation.	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Depth for soil stripping for the disturbance area are as follows (Figure 4-1): o Cleared creek flats (relatively recent alluvium; Stratic Rudosols), there is potential to collect soil (from an average depth of 3 m) with the high quality soil to be used as topdressing material for agricultural post-mining land use, following treatment with coarse-grade gypsum (approximately 80 ha total). o Sub-sections of the vegetated areas in the north of the Project site have soil conditions that allow a cut of 0.25 m (approximately 30 ha total). o Due to major subsoil constraints, a cutting depth of 0.10 m is recommended elsewhere in the remaining Project disturbance areas (approximately 405 ha). These soils could be used for woodland/forest rehabilitation (McKenzie Soil Management 2011). o In addition to the high quality soil resources described above, large volumes of other soils could be used in rehabilitation and without amelioration to provide conditions suitable for the native woodland/forest. This additional soil could be obtained from the Class 3 Agricultural Suitability areas that are not Stratic Rudosols to a depth of approximately 1 m.	No evidence of the site exceeding these requirements	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Areas of disturbance will be stripped progressively to reduce potential erosion and sediment generation, and to minimise the extent of topsoil stockpiles and the period of soil storage.	noted in site inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Areas of disturbance requiring soil stripping will be clearly defined following vegetation clearing.	Sites to be cleared are pegged by surveyors	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Topsoil and subsoil stripping during periods of high soil moisture content (i.e. following heavy rain) will be avoided to reduce the likelihood of damage to soil structure.	None done that has damaged soil structure, soil dries quickly	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Stripped soil will either be directly re-emplaced in rehabilitation areas or stockpiled for future reuse. Preference will be given to placing subsoils and topsoil (in reverse stripping order) directly onto re-contoured areas.	Noted onsite thought here are large stockpiles of soils that have been there since the inception of the site that could be preferentially used.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Subsoils and topsoils will be characterised prior to re-spreading to determine the type and application rates for any required soil ameliorants (e.g. lime, gypsum, fertiliser and organics) to maximise the availability of soil reserves for rehabilitation works.	Not done in the past, but will be in the future, minor amount of topsoil spread in the audit period	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Erosion and sediment control measures will be installed prior to the commencement of soil stripping and rehabilitation activities. Erosion control measures, including the management of soil stockpiles, are outlined in Section 4.8.	Sighted in inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	All cleared vegetation not retained for habitat augmentation is mulched on site and mixed into topsoil as a soil conditioner.	Sighted in inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	The ground-layer vegetation and low shrubs will be incorporated into the topsoil when it is stripped to assist rehabilitation by increasing the seed bank and organic matter within the stockpiled soil.	Sighted in inspection	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Recover vegetation and habitat resources during clearing activities where available and practical for re-use in rehabilitated areas to provide habitat resources for fauna.	Sighted in inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Vegetation from Project areas in Leard State Forest, plus the vegetation within other Project disturbance areas that is not salvaged for other purposes and is suitable for habitat enhancement, will be re-used in the mine rehabilitation program. Where insufficient area is available for the direct transfer of cleared debris, the material will be stockpiled for later use. Tree hollows and logs will be selectively chosen for placement in areas where habitat enhancement is required. These features will be fixed to mature trees or placed on the ground.					
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Habitat features (e.g. trunks, logs, branches, small stumps and roots) will be salvaged during vegetation clearance activities where practical, and relocated to areas undergoing rehabilitation.	Sighted in inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Hollow tree trunks and branches used for roosting by threatened fauna will be relocated into alternative locations close by. These resources will be identified during pre-start inspections.	During the audit period, some clearing has occurred with an ecologist in attendance, reports by the ecologist were sighted.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	TCPL will develop and adopt an Aboriginal Heritage Conservation Strategy and a Heritage Management Plan. These plans will detail methods for salvaging Cultural Heritage Resources, as well as protocols for surface works to reduce the risk of accidental damage to known sites.	These have been developed	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Where practical, known Aboriginal and Cultural Heritage sites will be avoided, or sites marked on site plans and relevant Project documentation.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Where avoidance of sites is not practicable, site(s) will be subject to baseline recording in consultation with representatives of the Aboriginal community prior to disturbance and artefacts salvaged for safekeeping in consultation with the Aboriginal community. Baseline recording of sites will include completing an Aboriginal Site Impact Recording Form (or its equivalent) and submitting it to the AHIMS Registrar for each site.	See Heritage Management plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Culturally modified trees located outside (but in close proximity) of any Project disturbance areas will be suitably fenced and signed to reduce the risk of incidental damage.	See Heritage Management plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	If appropriate in the context of the tree condition, culturally modified trees subject to direct disturbance will be considered for salvage. A suitable location for the storage and/or display of the salvaged sections will be identified and managed in consultation with the Aboriginal community.	See Heritage Management plan	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	The Aboriginal community will, where necessary, provide advice on the storage of collected artefacts, management of artefacts at the completion of Project activities (e.g. artefact replacement onto the post-mining landscape) and the implementation of management measures for salvaged culturally modified trees.	See Heritage Management plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.2	Appropriate Aboriginal representation will be provided during archaeological fieldwork (e.g. collection of artefacts prior to disturbance).	See Heritage Management plan	Compliant			
	4.3.3 Performance /	Completion Criteria	•				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	The progress and effects of salvaging and reusing habitat resources are monitored annually.	The Tarrawonga Monitoring Report 2013 notes the reuse of materials in the plots used for rehabilitation monitoring but there is no assessment of or comment on the effects of reusing these materials. The AEMR also does not address this issue.	Not Compliant Administrative			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Topsoils and subsoils are mapped and soil resources for Agricultural and Woodland land-uses are selectively stripped and managed.	This is conducted and will be documented in the topsoil register	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Habitat features (hollow bearing logs, felled timber and large rocks) are salvaged during prestripping operations for re-use where practical.	This occurrs, sighted stockpiles and materials in the rehab areas during the site inspection.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Photographic evidence and documentation of salvaging resources and reusing in the rehabilitation area for beneficial reuse is to be monitored and reported annually as part of the Biodiversity Monitoring Program.	Photos are taken and reported	Compliant			
	4.4 Collecting and Pr						
Tarrawonga Mine Biodiversity Management Plan 2013	4.4.2 Management S 4.3.3	TCPL will arrange for the collection of seed at the site at appropriate times after flowering.	This has not occurred at the mine site	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	The seed collection program will be aimed at producing the maximum quantity and quality of seed incorporating the correct degree of maturity for maximum germination.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Trees will not be felled solely for seed collection reasons. Where trees are cleared for mining purposes, available seed will be collected. Wherever practicable, tree felling will be timed to coincide with maximum seed load (but avoiding times when native fauna would be using trees for roosts, nests).	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Filluling	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Seed collection undertaken to date has involved hand collection from the ground and collection from felled trees; however, a variety of additional seed collection techniques may be adopted, including: o For plants with a low density of occurrence, collection may require the placing of bags over the flowers after pollination. Alternatively, dropped seed may be collected by placing plastic on the ground below. o Seed from small plants close to the ground could be collected by machine harvesters or hand picking/pruning. o Seed pods from the upper and middle portions of the crown of trees are more desirable as they tend to be of better quality. Collection techniques for trees may include cherry picker, ladder or hoist attached to a vehicle, long handled pruners, rope saw, etc. o For grasses and other groundcover species, use of a forage harvester to collect seed and mulch material after seed maturity and the direct placement of this material on the post-mining landscapes is an appropriate means of collection and transferral of seed where safe machine access is possible.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Seed handling and storage varies with the type of seed and will require specialised techniques and equipment. This task will be contracted to the personnel undertaking seed collection and/or propagation works.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Seedling propagation activities will be undertaken by personnel who are experienced in the propagation of native species.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	TCPL will advise the nursery / propagation contractor of its rehabilitation requirements (in terms of area) sufficiently in advance of the programmed planting time to enable seedling propagation, hardening-off (if required) and planting each year.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.3.3	Planting will preferentially be undertaken in the Autumn period on previously established riplines followed by mulching around each seedling.	This does not occur, no mulching	Not Compliant	D	3	Medium
	4.4.3 Performance / C						
Tarrawonga Mine Biodiversity Management Plan 2013	4.4.3	Monitoring of plant flowering and seeding cycles by the seed collectors and their establishment of a comprehensive database and detailed strategy.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.4.3	Adaptive management undertaken and documented if required.	Noted				
	4.5 Minimising the Im						
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2 Management St 4.5.2	Some potential impacts have been avoided/reduced through refinement of the mine design, and other impacts are likely to be mitigated by progressive rehabilitation as well as local habitat restoration, management and supplementation strategies.	Noted				





Deference	Classes	Requirement	Evidence	A coults file alter a		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Residual impacts will be addressed by the long-term conservation and enhancement of significant areas of fauna habitats in the offset area that can be enhanced by appropriate management and/or the creation of significant areas of fauna habitat resulting from the revegetation programme.	Noted				
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	Progressive backfilling of the open cut mine voids instead of only out-of-pit dumping to avoid additional native vegetation clearance.	Noted and observed on site	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Land clearance for the Project will be undertaken progressively, and the area cleared at any particular time would generally be no greater than that required to accommodate the mine"s needs for the following twelve months.	Noted and observed on site	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Areas requiring clearing will be clearly delineated and would be restricted to the minimum area necessary to undertake the approved activities.	this occurs, reviewed the northern extension cleared area during the site inspection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Clearing will be subject to a pre-clearance survey by an appropriately qualified ecologist to ensure clearing activities are managed to minimise impacts on fauna. The Vegetation Clearance Protocol will be implemented for clearing activities in accordance with Appendix A of this BMP.	This occurs, reviewed Ecologicals report on the Northern Extension area	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Clearing of trees and shrubs will, where practical, be restricted to late summer and autumn in order to avoid the spring when birds are nesting, winter when bats are hibernating and early to midsummer when bats are bearing young.	Noted				
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	Suitably trained or qualified person(s) will be present during the felling of identified hollow bearing trees to provide assistance with the identification, and if necessary, rescue and care of any injured fauna.	This occurrs - ecologist, archaeologist and aboriginal groups present dueing northern extension clearing, evidence presented for all.	Compliant			
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	The species, number and condition of fauna identified during clearing activities will be recorded and a summary provided in the Annual Environmental Management Report (AEMR)/Annual Review.	Noted, none in the audit period	Not Applicable			
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	The chainsaw felling and/or bulldozing of larger vegetation to just above the ground to minimise soil disturbance. Groundcover will be retained and subsequently collected when the topsoil is stripped.	Not able to observe this during the audit. EO states that this occurs.	Compliant			
arrawonga Mine Biodiversity Management Plan 2013	4.5.2	Translocation of fauna will not be undertaken unless appropriate licences have been obtained from OEH. Although unlikely, if a Koala is found, it will be left to move away from the clearance area on its own accord.	Noted				
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	The salvage of cleared trunks, logs, branches, small stumps and roots will be transferred directly to an area that has been prepared to the post mining landform, or alternatively stockpiled for later use in the rehabilitation.	Observed in the site inspection	Compliant			
Farrawonga Mine Biodiversity Management Plan 2013	4.5.2	The salvage of hollow tree trunks and branches will be placed on rehabilitated areas to provide fauna habitat.	Observed	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Pre-clearance surveys of ancillary infrastructure areas (e.g. water management structures, monitoring equipment areas) for threatened flora species will be undertaken. Relocation (if practicable) of ancillary infrastructure areas (e.g. water management structures, monitoring equipment) should be considered to avoid any threatened species.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Feral animal and weed control shall be undertaken.	This occurrs - property manager	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	Adequate noise and dust management, controlling the use of artificial lighting and fire management will also assist in minimising the impact on fauna within the Project Area. Lighting strategies/control measures to minimise potential artificial lighting impacts will include the use of unidirectional lighting fixtures.	Noted 3				
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.2	The on-site speed limit of 40 km/hr will continue to be applied to new haul roads and internal roads.	Noted				
	4.5.3 Performance /				<u> </u>		
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.3	Pre-clearing reports verify that clearing has been undertaken in accordance with the measures described above.	Evidence provided for arch and ecology pre- clearance reports for the northern extension	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.5.3	The species, number and condition of fauna identified during clearing activities would be recorded and a summary provided in the Annual Environmental Management Report (AEMR)/Annual Review.	None identified to date	Not Applicable			
	4.6 managing Salinity						
	4.6.2 Management S	itrategies					
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.2	Overburden and interburden will be characterised prior to emplacement in waste dumps to ensure the final outer surfaces of the overburden emplacements (and structures such as drainage elements) are constructed with suitable non-sodic or low sodicity material where possible.	This does not occur, no characterisation of overburden.	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.2	Any sodic materials that may be exposed or near-surface will be ameliorated (e.g. treated with gypsum) to minimise dispersivity	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.2	Irrigation activities of pasture areas will be undertaken to maximise evapo-transpiration but avoid surface runoff to minimise the risk of impacts on downstream water resources	This doesn't happen on the mine site, no pasture areas established in the rahab to date. The times when water has been in excess have conincided with saturated soils preventing the irrigation option being used.	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.2	Water quality monitoring of sites on Goonbri Creek shall continue to be undertaken on an event based frequency	This occurs	Compliant			
	4.6.3 Performance /	Completion Criteria					
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.3	Soil tested for chemical properties, including salinity.	The site has not progressed to the point where areas are being considered for relinquishment.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.6.3	Salinity down the soil profile is within the range of analogue sites at year 5.	Not at year 5	Not Applicable			
	4.7 Controlling Weed 4.7.2 Management S						





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Farrawonga Mine Biodiversity Management Plan 2013	4.7.2	Weeds from topsoil stockpiles will be scalped prior to respreading in rehabilitation areas	topsoil stockpiles are sprayed	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Maximising the retention of ground cover (cover crop stubble) when planting tubestock to minimise opportunities for weed activity.	This was observed	Compliant			
Farrawonga Mine Biodiversity Management Plan 2013	4.7.2	Regular inspections (at least six monthly) of revegetated areas and other parts of the mining tenements will be undertaken to identify and demarcate areas of noxious and environmental weeds	Monthly inspections by the EO provide this information.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Mechanical removal and/or the application of approved herbicides in areas identified as being affected by weeds (in accordance with the Pesticides Act 1999).	This was observed	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Follow-up site inspections to evaluate the effectiveness of weed control programs.	This occurs - Monthly inspections	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Follow-up weed control in previously treated areas where weed management has been suboptimal	This occurs - Monthly inspections and actions resulting	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Minimise the potential for seed and organic matter transport to or from the rehabilitation area by ensuring all plant and equipment are weed free (via inspection of vehicles and plant and use of the site's vehicle wash bay).	This does not occur - no inspections	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Focused weed management/suppression measures will be undertaken on stockpiles, roadsides and disturbance areas.	This occurs	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Regular liaison with local landholders and relevant government agencies to monitor the spread and management of weeds within the local area.	This occurs	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Installation of fauna exclusion fencing and/or tree guards for newly planted tubestock.	This does not occur	Not Compliant	D	3	Medium
Farrawonga Mine Biodiversity Management Plan 2013	4.7.2	Trapping and/or baiting of animal pests (e.g. Rabbits and Red Foxes), in accordance with the requirements of the Livestock Health and Pest Authorities	This occurs - Property Manager	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.2	Follow-up site monitoring to determine the effectiveness of trapping and/or baiting programs.	This occurs - Property Manager	Compliant			
4	1.7.3 Performance /	Completion Criteria					
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.3	Known weed and feral animal populations do not increase in size or distribution as a result of Project activities.	No evidence to confirm spread of ferals or weeds	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.7.3	Weed species presence and densities are monitored and control programs implemented when required.	This occurs. No evidence to confirm spread of ferals or weeds	Compliant			
Farrawonga Mine Biodiversity Management Plan 2013	4.7.3	Faunal exclusion fencing and/or tree guards are installed to exclude vertebrate pest species from rehabilitation areas / juvenile vegetation.	This does not happen	Not Compliant	D	3	Medium
Farrawonga Mine Biodiversity Management Plan 2013	4.7.3	Vertebrate pest species presence and densities are monitored, and control programs implemented when required.	No evidence to confirm spread of ferals or weeds	Compliant			
Farrawonga Mine Biodiversity Management Plan 2013	4.7.3	Monitoring results and any control activities will be reported in the Annual Environmental Management Report (AEMR)/Annual Review.	Included in the AEMR	Compliant			
	1.8 Controlling Erosi	ion					
L	1.8.2 Management	Strategies					
·			·				





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Erosion and sediment control plans will be developed over the life of the Project as part of the Water Management Plan. These plans will describe the specific controls (including locations, function and water monitoring structure capacities) that will be used to minimise the potential for soil erosion and transport of sediment off-site. These plans will be updated periodically to meet the particular changes to the Project over the life of the mine. All erosion and sediment control storages will have sufficient capacity to manage disturbed area runoff in accordance with design criteria recommended in the Landcom (2004) guidelines.		Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Approved drainage and sediment and erosion controls will be implemented progressively, including sediment basins and contour banks	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	General soil resource management practices will include the stripping and stockpiling of soil resources prior to any minerelated disturbance for use in rehabilitation, including. o Identify and quantify potential soil resources for rehabilitation o Optimise the recovery of useable topsoil and subsoil during stripping operations o Manage topsoil and subsoil reserves so as not to degrade the resource when stockpiled o Establish effective soil amelioration procedures to maximise the availability of soil reserves for future rehabilitation works o Disturbance areas will be stripped progressively in order to reduce sediment generation and the extent of topsoil stockpiles, and to enable use of stripped topsoil as soon as possible for rehabilitation.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Erosion and sediment control measures will be installed prior to the commencement of soil stripping and rehabilitation activities.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Areas of disturbance will be stripped progressively, as required, to reduce potential erosion and sediment generation, and to minimise the extent of topsoil stockpiles and the period of soil storage.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Areas of disturbance requiring soil stripping will be clearly defined following vegetation clearing.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Cover crops will be planted on newly rehabilitated mine landform areas as soon as possible after completing earthworks.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	New infrastructure disturbance areas (e.g. road and dam embankments) will be stabilised as soon as possible by topsoiling and seeding.	See WMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Topsoil and subsoil stripping during periods of high soil moisture content (i.e. following heavy rain) will be avoided to reduce the likelihood of damage to soil structure.	See WMP, soil stripping procedure	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	Any long-term soil stockpiles will be managed to maintain long-term soil viability through the implementation of the following management practices: o Topsoil and subsoil stockpiles will be retained at a height of 3 m, with slopes no greater than 1:2 (vertical to horizontal [V:H]) and a slightly roughened surface to minimise erosion o Topsoil stockpiles will be constructed to minimise erosion, encourage drainage, and promote revegetation o Where additions such as lime, gypsum and fertiliser are needed to improve the condition of cut soil, they will be applied to the stockpiles in-between the application of separate layers from the scrapers o Wherever practicable, soil will not be trafficked, deep ripped or removed in wet conditions to avoid breakdown in soil structure o All topsoil and subsoil stockpiles will be seeded with a nonpersistent cover crop to reduce erosion potential as soon as practicable after completion of stockpiling o Where seasonal conditions preclude adequate development of a cover crop, stockpiles will be treated with a straw/vegetative mulch to improve stability o Soil stockpiles will be located in positions to avoid surface water flows. Silt stop fencing will be placed immediately down-slope of stockpiles until stable vegetation cover is established	See WMP, soil stripping procedure	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.2	o An inventory of soil resources (available and stripped) on the Project site will be maintained and regularly reconciled with rehabilitation requirements o In preference to stockpiling, wherever practicable, stripped topsoil and subsoil will be directly replaced on completed sections of the final landform.	See WMP	Compliant			
	4.8.3 Performance /	Completion Criteria					
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.3	Surface Water Management Plan is developed and implemented, including erosion and sediment control plans. Periodic review of these plans is undertaken regularly.	See WMP, developed and implemented but not approved. Plan is not old enough to require review but there are comments on reviews elsewhere in the audit.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.3	Sediment control structures are in accordance with the relevant site management plans and standards.	See WMP - design documentation sighted and inspections of the operation of the structures sighted.	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.3	Erosion and sediment control measures meet the requirements of the Landcom (2004) guideline.	- '	Compliant		_	
Tarrawonga Mine Biodiversity Management Plan 2013	4.8.3	Absence of slumping and/or uncontrolled erosion.	See WMP - no evidence of either slumping or erosion (uncontrolled) in the site visit.	Compliant			
	4.9 Controlling Acces						
Tarrawonga Mine Biodiversity Management Plan 2013	4.9.2 Management S 4.9.2	Maintain access roads and tracks within the Project Boundary	This occurs	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	4.9.2	Maintain boundary fencing	This occurs	Compliant			





Deference	Claves	Requirement	Evidence	Audit Findin	Risk			
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk	
Farrawonga Mine Biodiversity Management Plan 2013	4.9.2	Maintain signage to advise not to enter Project Boundary.	This occurs	Compliant	-			
	4.9.3 Performance / 0	Completion Criteria						
Tarrawonga Mine Biodiversity Management Plan 2013	4.9.3	All fences are maintained in good working order	This occurs	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.9.3	Designated access roads and tracks are maintained	This occurs	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.9.3	Signage is installed and clearly visible.	This occurs	Compliant				
	4.10 Managing Bushfi	re Risk						
	4.10.2 Management S	Strategies						
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	Clearing will not be undertaken during periods of extreme fire danger as defined by the Bureau of Meteorology	This occurs	Compliant				
-		Controlled high intensity, short-term grazing will be employed to	This has not yet happened as there are no					
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	assist in the reduction of vegetative fuel loads on areas on which active mining operations are not occurring and appropriate fencing is available	large areas where cattle could access rehab that would not place then into contact with site equipment.	Not Applicable				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	All personnel and contractors would be required to use diesel vehicles and/or remain on defined roads or tracks	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	A fire break shall be established and maintained around the perimeter of the mining leases.	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	Prohibition of smoking in fire prone areas	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	Fire fighting equipment shall be provided on-site including water carts equipped with water cannons to provide immediate response to a bushfire	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	Appropriate mine personnel will receive basic fire control training	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.2	Fire prevention and fuel load reduction in rehabilitated mine areas will be undertaken if required and will involve measures such as a combination of fire breaks and short periods of high intensity grazing. Controlled burns may also be used through consultation with Narrabri Shire Council and the local fire brigades.	See Bushfire Management Plan	Compliant				
	4.10.3 Performance /	Completion Criteria						
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.3	Fuel load/fire security will be inspected at a minimum of 12 monthly intervals, occurring prior to the commencement of each bushfire season	This is not documented	Not Compliant Administrative	D	2	Medium	
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.3	All fire equipment will be kept in a serviceable condition and inspected at least once every 6 months by a fire equipment service provider (3 months for mining equipment)	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.3	Bushfire incidents will be recorded and managed via the Whitehaven incident management process. Occurrence of bushfires will be reported to the relevant authorities and discussed in the Annual Environmental Management Report (AEMR)/Annual Review.	See Bushfire Management Plan	Compliant				
Tarrawonga Mine Biodiversity Management Plan 2013	4.10.3	The bushfire management plan will be reviewed every two years and following any significant changes in bushfire management at the site.	See Bushfire Management Plan	Compliant				





Reference	Clause	Requirement	Evidence	Andia Finalia		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
į.	5.2.1 Weed and Fera						
Tarrawonga Mine Biodiversity	5.2.1	Weeds will be controlled and monitored by an appropriately	See Biodiversity Offset Management Plan -	Compliant			
Management Plan 2013	5.2.1	qualified contractor.	Property Manager	Compliant			
Tarrawonga Mine Biodiversity		Mechanical removal and/or the application of approved	Noted				
Management Plan 2013	5.2.1	herbicides may be used in areas identified as being affected by					
Wanagement Flam 2013		weeds.					
		Follow-up site inspections to evaluate the effectiveness of weed	See Biodiversity Offset Management Plan -				
Tarrawonga Mine Biodiversity	5.2.1	control programs. Follow-up weed control in previously treated	Property Manager	Compliant			
Management Plan 2013	312.1	areas will be undertaken where weed management has been sub-		Compilation			
		optimal.					
		Potential carriers of weed seeds and propagules (including	See Biodiversity Offset Management Plan -				
Tarrawonga Mine Biodiversity	5.2.1	vehicles, machinery and boots) will be cleaned before entering	not yet implemented	Noted			
Management Plan 2013	512.1	BOAs and when moving between weed infested sites to minimise		110100			
		the potential for the spread of weeds.					
		Three noxious weeds listed under the NSW Noxious Weeds Act,	Noted				
		1993 for the Narrabri Shire Council area were recorded in the					
		BOA: Galvanised Burr (Sclerolaena birchii), Noogoora Burr					
Tarrawonga Mine Biodiversity		(Xanthium occidentale) and Prickly Pear (Opuntia stricta). Two					
Management Plan 2013	5.2.1	environmental weeds that are common in the offset area include					
		Coolatai Grass (Hyparrhenia hirta) and Sweet Briar (Rosa					
		rubiginosa). Where available, weed management for these					
		species should be undertaken in accordance with the Weeds					
		Management Plans developed by the Narrabri Shire.					
		Regular liaison with local landholders and relevant government	Biodiversity Offset Management Plan.				
Tarrawonga Mine Biodiversity	5.2.1	agencies to monitor the spread and management of weeds within	Various levels of contact with local	Compliant			
Management Plan 2013		the local area.	landholders, no high risk weed outbreaks in				
			the audit period.				
		Eight introduced species have been identified within the offset	See Biodiversity Offset Management Plan,				
		area: Common Starling, Domestic Dog (Canis lupus familiaris),	the Property Manager will conduct the				
Tarrawonga Mine Biodiversity		Goat (Capra hircus), House Mouse, Rabbit, Black Rat, Pig and Red	monitoring and control of feral species				
Management Plan 2013	5.2.1	Fox. Animal pests would be controlled and monitored by an		Compliant			
		appropriately qualified contractor, and will be undertaken in					
		accordance with the requirements of the Livestock Health and					
T		Pest Authorities.	C D: I: ': O' +AA +BI				
Tarrawonga Mine Biodiversity	5.2.1	Follow-up site monitoring to determine the effectiveness of	See Biodiversity Offset Management Plan,	Compliant			
Management Plan 2013		trapping and/or baiting programs.	Property Manager				
Tarrawanga Mina Diadiyarsity		Known weed and feral animal populations do not increase in size	See Biodiversity Offset Management Plan,				
Tarrawonga Mine Biodiversity	5.2.1	or distribution as a result of Project activities.	to early to determine at the offset site, no increase on the mine site or the biobank	Compliant			
Management Plan 2013			site				
		Weed species presence and densities are monitored and control	See Biodiversity Offset Management Plan,				
Tarrawonga Mine Biodiversity	5.2.1	programs implemented when required.	Property Manager for offset areas and EO	Compliant			
Management Plan 2013	3.2.1	programs implemented when required.	for site areas, both run inspection regimes.	Compliant			
		Fauna exclusion fencing and/or tree guards are installed to	This is not conducted				
Tarrawonga Mine Biodiversity	5.2.1	exclude vertebrate pest species from rehabilitation areas /	This is not conducted	Not Compliant	D	3	Medium
Management Plan 2013	J.2.1	juvenile vegetation.		NOT COMPHAIL	<i>U</i>	J	ivieuluili
		Vertebrate pest species presence and densities are monitored,	See Biodiversity Offset Management Plan,				
Tarrawonga Mine Biodiversity	5.2.1	and control programs implemented when required.	Property Manager for offset areas and EO	Compliant			
, ,							





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.1	Monitoring results and any control activities will be reported in the Annual Environmental Management Report (AEMR)/Annual Review.	AEMR covers both weeds and feral animal controls and moniotoring in the reporting year.	Compliant			
	5.2.2 Bushfire Manag	ement	•				
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	A Bushfire Management Plan has been developed for the Tarrawonga Coal Mine and referred to the Rural Fire Service Narrabri and Narrabri Shire Council. The Bushfire Management Plan provides bushfire controls (including fire equipment and locations), emergency response (community/mine personnel), and bushfire training. The Bushfire Management Plan should be updated to address control measures for the BOA.	See Bushfire Management Plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	The Offset Area Management Plan will also address fire management procedures applicable to the BOA. Specifically, within areas of intact vegetation communities: Access tracks through the BOA will be maintained for fire management. Any fire regime implemented will require the approval from relevant authorities and a site specific Fire Management Plan should be developed. Mosaic burning shall be prescribed to reduce potential negative impacts.	See Bushfire Management Plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	Fuel load/fire security will be inspected at a minimum of 12 monthly intervals, occurring prior to the commencement of each bushfire season	See Bushfire Management Plan, this is not recorded	Not Compliant Administrative	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	All fire equipment will be kept in a serviceable condition and inspected at least once every 6 months by a fire equipment service provider (3 months for mining equipment)	See Bushfire Management Plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	Bushfire incidents will be recorded and managed via the Whitehaven incident management process. Occurrence of bushfires will be reported to the relevant authorities and discussed in the Annual Environmental Management Report (AEMR)/Annual Review.	See Bushfire Management Plan	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.2	The bushfire management plan will be reviewed every two years and following any significant changes in bushfire management at the site.	See Bushfire Management Plan, this is not yet due	Not Applicable			
	5.2.3 Controlling Acce	ess					
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Access tracks through the offset area will be maintained.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Offset areas will be designated "Restricted Access" areas to members of the community and all personnel and contractors (with the exception of those with approval of TCPL management to enter for legitimate purposes including weed and feral animal control, or for monitoring purposes).	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Maintain boundary fencing.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Maintain signage at primary access points to advise not to enter project boundary.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Access gates will be locked. Key distribution will be controlled by the Environmental Officer.	See Above	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	All fences are maintained in good working order.	See Above	Compliant	-		
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Designated access roads and tracks are maintained.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.3	Signage is installed and clearly visible.	See Above	Compliant			
	5.2.4 Collecting and F	Propagating Seed					
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	TCPL will arrange for the collection of seed at the site at appropriate times after flowering and will extend this program to involve other contractors if required.	See above, seed not collected at the mine site	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	The seed collection program will be aimed at producing the maximum quantity and quality of seed incorporating the correct degree of maturity for maximum germination.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Trees will not be felled solely for seed collection reasons.	Noted, no evidence of this occurring				
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Seed collection undertaken to date has involved hand collection from the ground and collection from felled trees; however, a variety of additional seed collection techniques may be adopted, including: o For plants with a low density of occurrence, collection may require the placing of bags over the flowers after pollination. Alternatively, dropped seed may be collected by placing plastic on the ground below. o Seed from small plants close to the ground could be collected by machine harvesters or hand picking/pruning. o Seed pods from the upper and middle portions of the crown of trees are more desirable as they tend to be of better quality. Collection techniques for trees may include cherry picker, ladder or hoist attached to a vehicle, long handled pruners, rope saw, etc. o For grasses and other groundcover species, use of a forage harvester to collect seed and mulch material after seed maturity and the direct placement of this material on the post-mining landscapes is an appropriate means of collection and transferral of seed where safe machine access is possible.					
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Seed handling and storage varies with the type of seed and will require specialised techniques and equipment. This task will be contracted to the personnel undertaking seed collection and/or propagation works.	Noted, seed is passed onto the nursery immediately following collection	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Seedling propagation activities will be undertaken by personnel who are experienced in the propagation of native species.	See above, The nursery used is familiar with the required processes for native seed propagation	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	TCPL will advise the nursery / propagation contractor of its rehabilitation requirements (in terms of area) sufficiently in advance of the programmed planting time to enable seedling propagation, hardening-off (if required) and planting each year.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Planting will preferentially be undertaken in the Autumn period on previously established riplines followed by mulching around each seedling.	See above, no mulching done	Not Compliant	D	3	Medium





Doforance	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Monitoring of plant flowering and seeding cycles by the seed collectors and their establishment of a comprehensive database and detailed strategy.	See above, this has not yet occurred	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.2.4	Adaptive management will be undertaken and documented if required.	See above - Noted				
	5.3 Degraded or Rege	eneration Areas					
	5.3.1 Weed and Fera	l Animal Control					
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.1	Known weed and feral animal populations do not increase in size or distribution as a result of Project activities.	See above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.1	Weed species presence and densities are monitored and control programs implemented when required.	See above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.1	Faunal exclusion fencing and/or tree guards are installed to exclude vertebrate pest species from rehabilitation areas / juvenile vegetation.	See above	Not Compliant	D	3	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.1	Vertebrate pest species presence and densities are monitored, and control programs implemented when required.	See above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.1	Monitoring results and any control activities will be reported in the Annual Environmental Management Report (AEMR)/Annual Review.	See above	Compliant			
	5.3.2 Bushfire Manag						
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	"Hot fires" may be used to stimulate regeneration from the seed bank in the soil	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	"Cool fires" may be used to control established weeds.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	Fire management is an important tool for biodiversity management, any fire management (e.g. implemented to control biomass, manage nutrients and manipulate species germination) must be carefully considered in light of detrimental impacts. Suitable alternative management measures should be considered in the first instance.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	Any fire regime implemented will require the approval from relevant authorities and a site specific Fire Management Plan should be developed. Mosaic burning shall be prescribed to reduce potential negative impacts.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	Fuel load/fire security will be inspected at a minimum of 12 monthly intervals, occurring prior to the commencement of each bushfire season.	This is not documented	Not Compliant Administrative	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	All fire equipment will be kept in a serviceable condition and inspected at least once every 6 months by a fire equipment service provider (3 months for mining equipment).	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2	Bushfire incidents will be recorded and managed via the Whitehaven incident management process. Occurrence of bushfires will be reported to the relevant authorities and discussed in the Annual Environmental Management Report (AEMR)/Annual Review.	Noted				
Tarrawonga Mine Biodiversity Management Plan 2013	5.3.2 5.3.3 Controlling Acco	The bushfire management plan will be reviewed every two years and following any significant changes in bushfire management at the site.	Noted				





Clause	Requirement	Requirement Evidence	Audit Finding	Risk			
Clause			Audit Finding	Consequence	Likelihood	Risk	
5.3.3	All fences are maintained in good working order.	See above	Compliant				
5.3.3	Designated access roads and tracks are maintained.	See above	Compliant				
5.3.3	Signage is installed and clearly visible.	See above	Compliant				
5.3.4 Enhancing the	Quality of Existing Vegetation and Fauna Habitat	<u> </u>					
5.3.4	Ecological thinning of White Cypress Pine regrowth will be undertaken as required to prevent the establishment of a dense locked growth monoculture.	Not yet required					
5.3.4	Plantings or seeding of White Box using local seed sources will also be undertaken as necessary following procedures prescribed in the RMP.	RMP	Compliant				
5.3.4		Not yet required					
5.3.4	The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL.	Noted - occurs at the site, logs and rocks in the rahab as additional structure					
5.3.4	Permanent fencing of the BOA to exclude livestock.	Noted					
5.3.4	Ongoing monitoring of the landscape of the available range of habitat types, stages and niches present in the landscape to demonstrate enhancement of native vegetation.	Noted - Ecological Monitoring Report	Compliant				
5.3.4	Implementation of the revised offset strategy, if required, following approval to Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Offset Strategy. Annual revegetation surveys undertaken with a detailed monitoring report provided	Noted - Leard Forest Mining Precinct Regional Biodiversity Offset Strategy notapproved or implemented	Not Applicable				
5.3.4	A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset	Noted - not yet developed	Not Applicable				
5.3.4	The proposed offset area would be independently audited at intervals agreed with relevant authorities.	Noted - This audit	Compliant				
5.3.4	A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval by the Director-General. An implementation plan will also be provided for approval at the same time, to ensure delivery of suitable areas of viable habitat for the nominated species (being Speckled Warbler, Brown Treecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittella, Turquoise Parrot, Masked Own, Yellow-bellied Sheath Tail Bat and Squirrel Glider). This plane will then be incorporated into	These have been provide to the DG	Compliant			_	
	5.3.3 5.3.4 Enhancing the 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4 5.3.4	5.3.3 All fences are maintained in good working order. 5.3.3 Designated access roads and tracks are maintained. 5.3.3 Signage is installed and clearly visible. 5.3.4 Enhancing the Quality of Existing Vegetation and Fauna Habitat Ecological thinning of White Cypress Pine regrowth will be undertaken as required to prevent the establishment of a dense locked growth monoculture. Plantings or seeding of White Box using local seed sources will also be undertaken as necessary following procedures prescribed in the RMP. Selective thinning of regenerating woodland that grows back in forest formation will be undertaken to create an optimal outcome for species diversity. The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL. Permanent fencing of the BOA to exclude livestock. Ongoing monitoring of the landscape of the available range of habitat types, stages and niches present in the landscape to demonstrate enhancement of native vegetation. Implementation of the revised offset strategy, if required, following approval to Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Offset Strategy. Annual revegetation surveys undertaken with a detailed monitoring report provided A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset 5.3.4 The proposed offset area would be independently audited at intervals agreed with relevant authorities. A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval by the Director-General. An implementation plan will also be provided for approval at the same time, to ensure delivery of suitable areas of viable habitat for the nominated species (being Speckled Warbler, Brown Treecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittella, Turquoise Parrot, Masked Own, Yellow-bellied Sheath Tail	See above 5.3.3 All fences are maintained in good working order. See above 5.3.3 Designated access roads and tracks are maintained. See above 5.3.4 Enhancing the Quality of Existing Vegetation and Fauna Habitat Ecological thinning of White Cypress Pine regrowth will be undertaken as required to prevent the establishment of a dense locked growth monoculture. Plantings or seeding of White Box using local seed sources will also be undertaken as necessary following procedures prescribed in the RMP. Selective thinning of regenerating woodland that grows back in forest formation will be undertaken to create an optimal outcome for species diversity. The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL. 5.3.4 Permanent fencing of the BoA to exclude livestock. Ongoing monitoring of the BoA to exclude livestock. Noted Ongoing monitoring of the landscape of the available range of habitat types, stages and niches present in the landscape to demonstrate enhancement of native vegetation. Implementation of the revised offset strategy, if required, following approval to Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Offset Strategy andertaken with a detailed monitoring report provided A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset The proposed offset area would be independently audited at intervals agreed with relevant authorities. A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval at the same time, to ensure delivery of suitable areas of viable habitat for the nominated species (being Speckled Warbler, Brown Trecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittela, Turquoise Parrot, Masked Own, Yellow-bellied Sheath Tall Bat and Squirrel Glider). This plan will then be incorporated into	Clause 5.3.3 All fences are maintained in good working order. 5.3.4 Designated access roads and tracks are maintained. 5.3.5 Signage is installed and clearly visible. 5.3.4 Signage is installed and clearly visible. 5.3.4 Enhancing the Quality of Existing Vegetation and Fauna Habitat 5.3.4 Ecological thinning of White Cypress Pine regrowth will be undertaken as required to prevent the establishment of a dense locked growth monoculture. 5.3.4 Plantings or seeding of White Edox using local seed sources will also be undertaken as necessary following procedures prescribed in the MMP. 5.3.4 The Manual of the Compliant of the Seed Swersty. 5.3.4 The Seed Swersty. 6.5.3.4 The Seed Swersty. 7.5.3.4 The Seed Swersty. 7.5.3.4 The Seed Swersty. 8.5.3.4 Permanent fencing of the BOA to exclude livestock. 8.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Ongoing monitoring of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Permanent fencing of the BOA to exclude livestock. 9.5.3.4 Regional Blodwersity Offset Strategy, if required, following approval to Strage 2 of the Leard Forest Mining Precinct Regional Blodwersity Offset Strategy and Precince Regional Blodwersity Offset Strategy and Pr	Clause 5.3.3 All fences are maintained in good working order. 5.3.3 Designated access roads and tracks are maintained. 5.3.3 Signage is installed and clearly visible. 5.3.4 Echancing the Quality of Existing Vegetation and Fauna Habitat Ecological thinning of White Cypress Pine regrowth will be undertaken as required to prevent the establishment of a dense locked growth monoculture. Plantings or seeding of White Box using local seed sources will also be undertaken as necessary following procedures prescribed in the RMP. 5.3.4 also be undertaken as necessary following procedures prescribed in the RMP. 5.3.5 Selective thinning of regenerating woodland that grows back in forest formation will be undertaken to create an optimal outcome for species diversity. The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL. 5.3.4 Permanent fencing of the BOA to exclude livestock. Ongoing monitoring of the BOA to exclude livestock. Ongoing monitoring of the BOA to exclude livestock. Implementation of the revised offset strategy, if required, following approval to Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Offset Strategy, Annual revegetation surveys undertaken with a detailed monitoring report provision of suitable habitat, shall be submitted for approval to the effectiveness of the measures and the performance of the offset 5.3.4 The proposed offset area would be independently audited at increasing agend with relevant authorities. A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset 5.3.4 The proposed offset area would be independently sudited at increasing agend with relevant authorities. A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval by the Director-General. An implementation plan will also be pr	See above Compliant 5.3.3 All fences are maintained in good working order. See above Compliant 5.3.3 Designated access roads and tracks are maintained. See above Compliant 5.3.3 Signage is installed and clearly visible. See above Compliant 5.3.4 Compliant Compliant 5.3.4 Experimentation of the control of the set and tracks are maintained. See above Compliant 5.3.4 Experimentation of the control of the set and tracks are maintained. See above Compliant 5.3.4 Experimentation of the control of the set and tracks are maintained. See above Compliant 5.3.4 Configuration of the control of the set and tracks are maintained. See above Compliant 5.3.4 International of the control of the set and tracks are maintained. See above Compliant 5.3.4 Plantings or seeding of White Cypress Pine regrowth will be undertaken as received prolonying procedures prescribed in the MMP. 5.3.4 Plantings or seeding of White Bou using local seed sources will be undertaken as received prolonying procedures prescribed in the MMP. 5.3.4 International of the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of sources will be interesting to the seed of the seed of seed of the seed of seed of the seed of sources will be interesting to the seed of the seed of seed of the seed of seed of seed of the seed of s	





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	5.4	Intensive restoration management activities are required within the grassland areas to restore native vegetation and fauna habit, focussing on assisted natural regeneration, targeting vegetation establishment and the introduction of naturally scarce fauna habitat features. Areas requiring restoration include derived native grasslands (c) and grasslands dominated by exotics (d)	These are not yet planned or commenced.	Not Applicable			
	5.4.1 Weed and Fera	I Animal Control					
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.1	Known weed and feral animal populations do not increase in size or distribution as a result of Project activities.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.1	Weed species presence and densities are monitored and control programs implemented when required.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.1	Faunal exclusion fencing and/or tree guards are installed to exclude vertebrate pest species from rehabilitation areas / juvenile vegetation.	See Above	Not Compliant	D	3	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.1	Vertebrate pest species presence and densities are monitored, and control programs implemented when required.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.1	Monitoring results and any control activities will be reported in the Annual Monitoring Program.	See Above	Compliant			
	5.4.2 Bushfire Manag	gement					
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.2	Fuel load/fire security will be inspected at a minimum of 12 monthly intervals, occurring prior to the commencement of each bushfire season.	See Above	Not Compliant	D	2	Medium
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.2	All fire equipment will be kept in a serviceable condition and inspected at least once every 6 months by a fire equipment service provider (3 months for mining equipment).	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.2	Bushfire incidents will be recorded and managed via the Whitehaven incident management process. Occurrence of bushfires will be reported to the relevant authorities and discussed in the Annual Environmental Management Report (AEMR)/Annual Review.	See Above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.2	The bushfire management plan will be reviewed every two years and following any significant changes in bushfire management at the site.	See Above	Compliant			
	5.4.3 Controlling Acco	ess					
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.3	All fences are maintained in good working order.	See above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.3	Designated access roads and tracks are maintained.	See above	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.3	Signage is installed and clearly visible.	See above	Compliant			
	5.4.4 Restoring Nativ	e Vegetation and Fauna Habitat					
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	Revegetation of derived grassland areas (refer to RMP).	See RMP	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	Management threatening processes that inhibit natural regeneration, including destocking by implementing ongoing stock exclusion from grassland areas.	See RMP - destocking complete	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	Ongoing monitoring of Eucalypt regeneration in the derived grasslands to identify any shortterm suppression due to the dense grassy layer in the derived grasslands may indicate that management options causing disturbance to the grassland may be required. Management options could include slashing or low-intensity controlled burning around paddock trees before seed fall and seasonal rains. In areas with no paddock trees, disturbance could be caused before seasonal rains to encourage regrowth from soil seed stores.		Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	The use of supplementary habitat measures to provide suitable habitat, including the use of naturally scarce fauna habitat features will be investigated by WCL.	See RMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	Revegetation of derived native grasslands and cleared land to revegetated woodland/open woodland habitat areas.	See RMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	Implementation of the revised offset strategy if necessary following approval of the Stage 2 Leard Forest Mining Precinct Regional Biodiversity Offset Strategy. Annual revegetation surveys undertaken with a detailed monitoring report provided.	See RMP	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset area.	See RMP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.4	A report detailing the investigation of factors likely to enhance or impede the effective long term provision of suitable habitat(s) shall be submitted for approval by the Director-General. An implementation plan will also be provided for approval at the same time, to ensure delivery of suitable areas of viable habitat for the nominated species (being Speckled Warbler, Brown Treecreeper, Grey-crowned Babbler, Hooded Robin, Varied Sittella, Turquoise Parrot, Masked Own, Yellow-bellied Sheath Tail Bat and Squirrel Glider). This plan will then be incorporated into the revised Biodiversity Management Plan.	These have been provide to the DG	Compliant			
	5.4.5 Enhancing the	Quality of Existing Vegetation and Fauna Habitat	<u>, </u>				
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.5	Implementation of the revised offset strategy if necessary following approval of the Stage 2 Leard Forest Mining Precinct Regional Biodiversity Offset Strategy	Noted	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.5	Annual revegetation surveys undertaken with a detailed monitoring report provided	Ecological report	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.5	A programme would be undertaken to monitor and report on the effectiveness of the measures and the performance of the offset	Not yet developed or implemented	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.5	The proposed offset area would be independently audited at intervals agreed with relevant authorities.	This audit	Compliant			
	5.4.6 Managing any						
Tarrawonga Mine Biodiversity Management Plan 2013	5.4.6	No surface disturbance works are planned for the BOA. Activities that will be undertaken include revegetation, stock exclusion and bushfire management. The Aboriginal Heritage Information Management System (AHIMS) register will be referred to should any surface disturbance works be undertaken.	No actions yet	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
9	9 Reporting						
9	9.1 Incident Reportin	ng					
Tarrawonga Mine Biodiversity Management Plan 2013	9.1	TCPL shall notify, at the earliest opportunity, the Director-General and any other relevant agencies of any incident that has caused, or threatens to cause, material harm to the environment. For any other incident associated with the project, TCPL will notify the Director-General and any other relevant agencies as soon as practicable after the TCPL becomes aware of the incident. TCPL will provide the Director-General and any relevant agencies a report on the incident, within seven days of the date of the incident, and such further reports as may be requested.	No incidents in the audit period	Not Applicable			
9	9.2 Annual Biodiversi	ity Monitoring			•		
Tarrawonga Mine Biodiversity Management Plan 2013	9.2	Annual biodiversity monitoring will be undertaken to assess the effectiveness and progress of biodiversity management against the performance/completion criteria, and identify triggers for adaptive management.	Ecological monitoring reports	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.2	Monitoring will be undertaken within both the Rehabilitation Area and the BOA, as described in Section 6. An annual monitoring report will be prepared that includes record monitoring results and discusses performance against key performance/completion criteria and TCPL commitments. Trends will also be identified and any non-compliances or triggers identified. Results of the annual biodiversity monitoring will be reported in the Annual Environmental Management Report.	Ecological monitoring reports	Compliant			
	9.3 Regular Reporting	g and Review	L				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kelelelice	Clause			Addit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	9.3	By the end of June each year, TCPL will prepare an Annual Review which reviews the environmental performance of the project for the previous calendar year (Schedule 5, Condition 5 of Project Approval (PA 11_0047) Approval. This review will: Describe the development, including any rehabilitation, that was carried out in the past calendar year, and the development that is proposed to be carried out over the current calendar year Include a comprehensive review of the monitoring results and complaints records of the Project over the past year, which includes a comparison of these results against the: o Relevant statutory requirements, limits or performance measures/criteria o Monitoring results of previous years o Relevant predictions in the EA Identify any non-compliance over the last year, and describe what actions were (or are being) taken to ensure compliance Identify any trends in the monitoring data over the life of the project Identify any discrepancies between the predicted and actual impacts of the project, and analyse the potential cause of any significant discrepancies Describe what measures will be implemented over the next year to improve the environmental performance of the project.		Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.3	The AEMR/Annual Review also addresses the matters identified in the Department of Trade and Investment (Resources and Energy) document entitled "Guidelines to the Mining, Rehabilitation and Environmental Management Process" (n.d).	SEE AEMR results in PA and MOP	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.3	The AEMR/Annual Review is provided to relevant agencies, the CCC members and is made available for public viewing on the Whitehaven Coal website. Copies of all management plans / strategies or monitoring programs, together with the results of independent audits undertaken in accordance with PA 11_0047 are provided to relevant authorities, and made publicly available on the Whitehaven Coal Website.	Sited CCC minutes, agencies distribution message and is available on the website.	Compliant			
	9.4 Complaint Receip	t and Response Procedure					
Tarrawonga Mine Biodiversity Management Plan 2013	9.4	In order to receive, record and respond to any complaints in a timely manner, TCPL maintains a telephone complaints line for the purpose of receiving complaints from any member of the public in relation to its activities. The complaints line operates 24 hours per day, seven days per week, is publicly advertised and the details are supplied to adjacent landowners (TCPL 2009).	See elsewhere	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	9.4	All complaints are recorded and responded to within 24 hours of the receipt of a complaint. A message bank will capture calls when they cannot be answered. The recorded message records the time of receipt and requests the following information. 1. Complainant's name 2. Telephone number 3. Preferred contact time 4. Nature of complaint.	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.4	The nature of the response depends on the nature and source of complaints but includes one or more of the following actions. 1. Liaison with the complainant to ascertain all details and to identify the nature and source of the complaint and provide supplementary details for the log. Details recorded in the log include: a. the date and time of the complaint b. the method by which the complaint was made c. personal details d. the nature of the complaint e. action taken by TCPL in relation to the complaint including any follow-up contact and f. if no action, the reason why. This activity may extend to other landowners /residents to determine the overall extent of the perceived problem. 2. As appropriate, the initiation of monitoring or other investigations to verify or otherwise the exceedance or noncompliance with consent, licence or lease conditions 3. Initiation of appropriate changes in operating practices or procedures 4. Conducting a follow-up interview with the resident to determine their level of satisfaction with the mine's response and the resultant outcome.	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.4	A copy of the report sheet is supplied to the complainant, if requested.	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.4	A summary of the complaints received in each 12 month period is included in each AEMR/Annual Review, together with a comparison with the number and nature of complaints received in the previous five years (TCPL 2009).		Compliant			
	9.5 Response to Non						
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	Compliance with all approvals, plans and procedures is the responsibility of all personnel and contractors employed on or in association with the mine, and is developed through promotion of project ownership under the direction of the Operations Manager and Environmental Manager (TCPL 2009).	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	The Environmental Manager undertakes regular inspections, internal audits and initiates directions identifying any remediation/rectification work required, and areas of actual or potential non-compliance, which have the potential to cause environmental harm or result in complaints are also reported to the relevant authority (TCPL 2009).		Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	Non-compliances with the requirements of the mine site's Environment Protection Licence (EPL) are reported on each Annual Licence Return (TCPL 2009).	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	A review of TCPL"s compliance with all conditions of the Development Consent, mining lease and all other approvals and licences is undertaken as part of each AEMR/Annual Review submitted to the Director-General and DPI. and also made publically available on the Whitehaven Coal website.	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	Additionally an independent audit is undertaken a minimum of once every three years and report submitted to the Director-General, and any other relevant authorities and made available to the public on the Whitehaven Coal Website.	See elsewhere	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	9.5	A set of environmental procedures that deal with the renewal of licences, leases and approvals, and also prescribe the means to ensure compliance with the regulatory requirements has been established to ensure ongoing compliance in the event of personnel changes (TCPL 2009).	See elsewhere	Compliant			
	10 Auditing	, ,					
Tarrawonga Mine Biodiversity Management Plan 2013	10	Whitehaven will undertake annual internal audits to align with its AEMR/Annual Review and MOP reporting requirements, as well as independent audits every 3 years	Internal audits not conducted in the audit period	Not Compliant Administrative			
	10.1 Annual Internal	Auditing	<u> </u>				
Tarrawonga Mine Biodiversity Management Plan 2013	10.1	An internal annual environmental audit will be undertaken to align with reporting requirements for the MOP and AEMR/Annual Review. The annual audit will focus on the compliance with the requirements of the BMP and where corrective action is required.	Internal audits not conducted in the audit period	Not Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	10.1	An annual audit and compliance report will be prepared which will include the methodology of the audit, actions implemented in accordance with the BMP, non-compliances with the BMP, and corrective actions taken. A summary of this report will be included in the AEMR/Annual Review.	period	Administrative			
	10.2 Independent Au	dit	<u> </u>				
Tarrawonga Mine Biodiversity Management Plan 2013	10.2	In accordance with Schedule 5, Condition 10 of the Project Approval, an independent audit will be undertaken by the end of June 2014 and every 3 years thereafter, unless the Director-General directs otherwise.	This audit, not compliant with timing	Not Compliant Administrative			
Tarrawonga Mine Biodiversity Management Plan 2013	10.2	The audit will be led by a suitably qualified auditor, and include experts in noise, air quality, water, ecology, and any other fields specified by the Director-General, who are endorsed by the Director General.	This audit	Compliant			
Tarrawonga Mine Biodiversity Management Plan 2013	10.2	Specifically in relation to this BMP, the audit will include: Consultation with relevant agencies Assessment of the environmental performance in complying with this BMP Review of the adequacy of this BMP and Recommend measures or actions to improve the environmental performance in relation to the requirements off this BMP.	This audit	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Tarrawonga Mine Biodiversity Management Plan 2013	10.2	A document will be prepared within 3 months of commissioning the audit, or as otherwise agreed with the Director-General, which includes the audit report, together with responses to recommendations contained in the audit report.	This audit, agreement re the extension of time noted.	Compliant			
	11 Review of this BM						
Tarrawonga Mine Biodiversity Management Plan 2013	11	Consequently, it is proposed that this BMP is reviewed and, if required, revised following completion of the related State and Commonwealth approval requirements, including: Stage 2 of the Leard Forest Mining Precinct Regional Biodiversity Management Strategy (State Condition 41 and Commonwealth Condition 22) Revised Biodiversity Offset Strategy (State Condition 42) Independent verification of Offset areas (Commonwealth Condition 7) Offset Management Plan (Commonwealth Condition 12 to15) Mine Site Rehabilitation Plan (Commonwealth Condition 23 to 26).	Will be reviewed following its approval	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	11	Following the review described above, the BMP will be reviewed and, if required, revised every 3 years following the findings of the independent audit (Section 10.2). The review will be undertaken in consultation with key stakeholders (including OEH, SEWPaC, Forests NSW, the CCC, DPI Catchments and Lands and the Namoi CMA) and approved by DP&I.	Will be reviewed following its approval	Not Applicable			
Tarrawonga Mine Biodiversity Management Plan 2013	11	A review of this BMP should be undertaken after 12 months and from then on at least every three years. The BMP review should examine trends, investigate the sampling effort in terms of redundancy or shortfalls, and provide the opportunity to incorporate new monitoring technologies or techniques if required. Regular review of this BMP will allow for adaptive management of both the Project Area and the BOS to be adopted and ensure that the BMP is consistent to overarching management plans, environmental policies, objectives and legal requirements. If preventative measures indicate that amendments to the BMP are necessary, the BMP will also be reviewed. Any review will be undertaken in consultation with key stakeholders and with the approval of relevant agencies.		Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
		offset Area Management Plan					
	3 Offset Management		1		ı		
Willeroi Biodiversity Offset Area Management Plan	3	For the purposes of biodiversity and offset management, the Willeroi BOA has been divided into three management zones according to condition and resilience (Figure 3 1): • Management zone 1 - maintenance of remnant vegetation • Management zone 2 - enhancement of regenerating native vegetation • Management zone 3: o 3a - restoration of DNG o 3b - restoration of exotic dominated grasslands o 3c - restoration of DNG and erosion areas.	Noted				
Willeroi Biodiversity Offset Area Management Plan	3	Each management zone has strategic ecological management objectives and relevant actions to achieve these objectives for a sustainable landscape in perpetuity. Each management zone and their broad management actions/objectives is described briefly below.	Noted				
	3.1.1 Management Zo	one 1 - Maintenance of remnant Vegetation					
Willeroi Biodiversity Offset Area Management Plan	3.1.1	The primary management objective of MZ1 is to maintain vegetation structural and species diversity, exclude stock, remove threats to biodiversity and implement ongoing monitoring of condition to detect changes in vegetation condition to inform adaptive management procedures.					
Willeroi Biodiversity Offset Area Management Plan	3.1.1	Management actions in this zone include: • Conduct baseline monitoring for flora and fauna to identify original reference condition for measuring future changes in condition, and to inform extent and location of weed management interventions	Done in the original assessment, included in this document	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.1	Grazing exclusion to promote understorey recovery and reduce competition for food with native fauna species	Grazing exclusion has been recently completed	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.1	Management of human disturbance including signage (and compliance checks) to prevent unauthorised entry/use	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	3.1.1	Retain dead timber (prevent fire wood collection)	Gates locked, no evidence of firewood collection	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.1	 Minor weed removal and ongoing monitoring to compliment adjacent areas. Opuntia stricta (Prickly Pear) does occur within areas of intact vegetation 	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	3.1.1	whole BOA.	Not yet completed	Not Applicable			
	3.1.2 Management Zo	one 2 - Enhancement of regenerating native vegetation					
Willeroi Biodiversity Offset Area Management Plan	3.1.2	The management objectives of MZ2 are to restore native species richness to all strata, initially through passive regeneration assisted by removal and treatment of threats, followed by active regeneration if required. The high resilience of the vegetation in MZ2 should enable natural regeneration in structural and species diversity to improve fauna habitat capacity over time (McIntyre et al. 2001). MZ2 will augment and buffer the core conservation areas in MZ1.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	3.1.2	Periodic, light grazing permitted e.g. cell grazing. Grazing removal to be timed to coincide with maximum native ground cover (typically associated with seasonal rains in mid-late summer) and minimum exotic species dominance of ground cover to avoid interim deterioration of condition and avoid resource competition for native fauna. Grazing must be actively managed to balance the objectives of overstorey regeneration and agricultural weed suppression. Temporary stock exclusion fencing will be required to protect high conservation areas (MZ1). In the mid-term (> 5yrs) grazing will be excluded to accommodate supplementary plantings or promote further maturation of natural regeneration.	on the Willeroi property. Tarrawonga will continue to investigate the	Not Applicable			
Villeroi Biodiversity Offset Area Management Plan	3.1.2	Management of human disturbance including signage at external access points to the Willeroi Offset Area (and compliance checks) to prevent unauthorised entry/use	see above	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.2	Retain dead timber including standing and fallen (prevent fire wood collection) to protect habitat niches	see above	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.2	Minor weed removal and ongoing monitoring to compliment adjacent areas. Opuntia stricta (Prickly Pear) does occur within areas of intact vegetation	see above	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.2	Targeted control of foxes, rabbits, goats and pigs required across whole Offset Area	see above	Compliant			
Willeroi Biodiversity Offset Area Management Plan	3.1.2	Revegetation will be undertaken in instances where monitoring indicates that natural regeneration is poor after three years.	see above	Compliant			
	3.1.3 Management 2	Zone 3 - Restoration of DNG and exotic dominated grasslands					
Willeroi Biodiversity Offset Area Management Plan	3.1.3	While natural regeneration is optimal for species integrity, a revegetation regime to compliment threat abatement and conservation management is recommended to reduce establishment times. Direct seeding is recommended in MZ3 as a cost-effective way to re-establish native vegetation over broad areas that often results in a more natural distribution of species across a site than planting. If direct seeding proves to be ineffective, then the revegetation works will be supplemented with tube stock planting to meet the completion criteria for MZ3.	Noted				
Willeroi Biodiversity Offset Area Management Plan	3.1.3	Management actions in MZ3 include: • Develop and implement a weed control regime (prior to any revegetation planting). Note additional environmental and noxious weed species may be present at the time of survey	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	3.1.3	Conservation grazing to reduce weed infestations and fire hazards until direct seeding events, thereafter grazing exclusion	Grazing has been removed	Compliant			
Villeroi Biodiversity Offset Area Management Plan	3.1.3	Control of over abundant native herbivores in consultation with adjacent landholders (including OEH)	Not yet completed	Not Applicable			
Villeroi Biodiversity Offset Area Management Plan	3.1.3	 Develop and implement a feral animal management control plan. Targeted control of foxes, rabbits, goats and pigs required across whole Offset Area 	Not yet completed	Not Applicable			
Villeroi Biodiversity Offset Area Management Plan	3.1.3	 Targeted revegetation of the DNG and exotic dominated grassland areas using direct seeding with local endemic species of pre-clearing vegetation type to enhance native flora diversity and increase fauna habitat values 	Not yet completed	Not Applicable			
Villeroi Biodiversity Offset Area Management Plan	3.1.3	If direct seeding proves to be ineffective, then the revegetation works will be supplemented with tube stock planting	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area	3.1.3	Active erosion control measures (MZ3c only)	Not yet completed	Not Applicable			
Management Plan	3.1.3			Not Applicable			
Willeroi Biodiversity Offset Area	3.1.3	 Management of human disturbance including signage (and 	Not yet completed	Not Applicable			
Management Plan		compliance checks) to prevent unauthorised entry/use.		140t Applicable			
	1 Land Management				l l		
		Unauthorised access to the whole Willeroi property will be	Not yet completed				
Willeroi Biodiversity Offset Area		prohibited and actively enforced. This will reduce the risk of					
Management Plan	4	disturbance to intact vegetation, regenerating areas, or		Not Applicable			
-		revegetated areas and prevent soil disturbance, weed dispersal,					
		fauna habitat disturbance and illegal rubbish dumping.	Decreate Menorar consists d				
		TCPL will appoint a Property Manager who will be responsible for	Property Manager appointed.				
		coordinating the implementation and reporting on all aspects of					
Willeroi Biodiversity Offset Area	4	this management plan. The approval of the Property Manager must be obtained for any access to or to undertake any activities		Compliant			
Management Plan	4	on or in the BOA. Personnel will only be permitted into the offset		Compilant			
		area to undertake management and monitoring actions identified					
		through this plan					
		TCPL will maintain external fencing around the entire Willeroi BOA	Not yet completed				
		to restrict human and stock access. Signage will be erected to	l l l l l l l l l l l l l l l l l l l				
		identify the site as an Offset Area and prohibit 4WD, trail bikes,					
Willeroi Biodiversity Offset Area		rubbish dumping, camping, shooting, fires and unauthorised					
Management Plan	4	access. The signage will be supplemented by a direct		Not Applicable			
J.		communication strategy with adjoining landholders to encourage					
		compliance and support for conservation objectives.					
		,					
		Some fencing will be required to protect revegetation works and	Not yet completed				
		to meet interim conservation grazing objectives. Internal fencing					
		will be maintained as an interim management measure where it					
		excludes stock to enable natural regeneration. Temporary stock					
Willeroi Biodiversity Offset Area	4	exclusion fencing is recommended to supplement existing fencing		Not Applicable			
Management Plan	7	to manage grazing pressure appropriate to each zone. Barbed		Not Applicable			
		wire is not recommended for new temporary, internal fencing.					
		The long term objective will be to remove all internal fencing to					
		minimise obstructions to fauna movements.					
		Fencing that is erected within or on the boundary of the offset	Not yet completed				
		area should be in accordance with the following 'wildlife friendly'					
		specifications (Rawlings et al. 2010):					
		Use plain, non-barbed wire only Limit the graph of chiral through a standard on the force.					
		Limit the number of wire strands used on the fence December 16 peace provides bashed wire to protect the offset.					
Willeroi Biodiversity Offset Area		 Boundary fences may use barbed wire to protect the offset areas from grazing cattle but the top two strands should be plain, 					
Management Plan	4	non-barbed wire		Not Applicable			
ivianagement rian		A night-visible fencing material should be installed along the					
		fence adjacent to high risk areas such as woodlands and wetlands					
		If new internal or temporary fencing is required, it should be					
		polywire electric using either tape or braid.					
Willeroi Biodiversity Offset Area	4	All access tracks, fence lines and signage erected within the offset	Not yet completed	Not Applicable			
Management Plan		areas will be inspected and maintained regularly.	1		1		





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.2	Grazing management on Willeroi property will be conducted to support primary conservation objectives in a deliberate shift from historical productivity objectives. Grazing by domestic stock reduces food and habitat resources for native animals, however, light to moderate grazing by stock can be compatible with native grassland management as it may be used to suppress weeds (McIntyre et al. 2002).	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.2	Stock should be excluded from the offset area prior to revegetation works commencing. Exclusion of stock will continue in areas where the removal of stock has allowed natural regeneration to occur without an increase in weed density. However, where stock exclusion has seen a marked increase in weed density and limited natural regeneration of native vegetation, cattle may be reintroduced to these areas.	This has occured	Compliant			
Willeroi Biodiversity Offset Area Management Plan	4.2	Grazing activities within offset areas will adhere to the specifications shown in <i>Grassy Guide</i> (Barlow 1998) and <i>A Guide to Managing Box Gum Grassy Woodlands</i> (Rawlings et. al. 2010). Strategic grazing should: • Be undertaken as 'pulse' or 'crash' grazing, where fencing restricts stock movement to a limited area for a brief amount of time at deliberate intervals • Take into account variability within the Biodiversity Offset and Management Areas. Pulse grazing is most effective in grasslands and less effective in woodlands • Be undertaken in such a way as to promote patchiness (i.e. areas of different age, density, and species) • Take into account the ecological requirements of the native species being targeted for promotion • Not be undertaken during spring unless for control of annual exotics • Not be undertaken during wet conditions • Utilise 'clean' (i.e. weed free) stock. For instance, if sheep are used they should be introduced off-shears and with empty stomachs • Incorporate adequate rest time for the patch after grazing to promote native recovery. Weed and erosion control may also be required.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.2	Monitoring of the impact of grazing will be required. The objectives of grazing and grazing monitoring criteria should be developed in a 'Farm Management Plan'. This will include stocking rates, and triggers to initiate and remove grazing.	Farm Management Plan not completed	Not Applicable			
	4.3 Enhancing the qu	ality of existing vegetation and fauna habitat					
Willeroi Biodiversity Offset Area Management Plan	4.3	The quality of existing vegetation and fauna habitat within the areas marked for maintenance or enhancement within the BOA (MZ1 and MZ2) will be enhanced through specific land management activities including stock management and weed control. These activities will help to restore internal connectivity of woodland and forest habitats.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.3	Specific strategies that will be implemented in the BOA Area to enhance the quality of existing vegetation include: • MZ1 - No active planting is proposed within these areas, active regeneration will be encouraged through weed and stock management.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.3	MZ2 - No initial plantings are proposed; if natural regeneration is poor after the first management period (as indicated by the monitoring program), targeted planting should be undertaken. Thinning of the White Cypress Pine regrowth communities to prevent a dense locked growth monoculture becoming established should be undertaken in areas that are dominated by thick regrowth to create an optimal outcome for species diversity	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.3	MZ3c – restoration of native vegetation	Noted				
	4.4 Restoring native	vegetation					
Willeroi Biodiversity Offset Area Management Plan	4.4	Restoring native vegetation and fauna habitat will be undertaken within MZ3 only. Activities to restore native vegetation and fauna habitat within the grasslands of the Offset Area will focus on assisted natural regeneration complemented by targeted vegetation establishment. Areas of DNG and grasslands dominated by exotics are planned to be revegetated to woodland/open woodland habitat areas.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.4	Generic management strategies for restoring native vegetation include: • Management of threatening processes that inhibit natural regeneration (including stock management, vertebrate pest control and weed control) • Ongoing monitoring of Eucalypt regeneration to identify whether the dense grassy layer is inhibiting growth of eucalypt seedlings. Monitoring results would be used to determine whether disturbance to the grassland is required to encourage Eucalypt growth. Management options include o Slashing around paddock trees o Low-intensity controlled burning around paddock trees o Disturbance to soil before seasonal rain.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.4	Wherever possible, the seed for the planting program will be collected from the BOA. Seeds should be prepared for direct seeding (% mix) or propagated for planting as 'Hiko' tube stock by a specialist native species propagation nursery.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.4	Plant density will depend on the amount of existing vegetation and the results of natural regeneration. All vegetation types are to be managed towards the benchmark range canopy cover per Biometric vegetation type (Table 4 1). As a guide, a density of one tree per 100 m2 (one every 10 metres) will achieve a final density of one tree every 200 m2 or a canopy cover of between 5-25% as per the benchmark for the 'White Box grassy woodland' vegetation type, assuming a 50% success rate. This planting density allows for time lag to maturation and mixed success of plantings.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.4 4.4.1 Management a	Species selection will be determined by parent vegetation type	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.4.1	Areas of DNG that show low rates of natural regeneration establishment after the first three years (MZ3a) shall be direct seeded in improve establishment. Areas to be direct seeded will be slashed prior to planting (MZ3a). Direct seeding is only to be undertaken during the optimum period of growth for the majority of species being seeded (typically late winter / early spring). Methods of direct seeding depend on safe machine access with minimal disturbance of existing native vegetation and soil erosion. Two main methods may be used for revegetation: • Mechanised direct seeding: A direct seeding machine, or seed box, is used; typically towed by a tractor on a three point linkage system. This approach is useful as it prepares the seed bed and plants the seed in one operation. Limitations include that it is only suited to seed which can easily pass through the machinery, such as Acacia species, and to areas with a slope of 3:1 or less • Hand seeding: Hand seeding (or other air delivery systems) is useful for native grass and eucalyptus seed which is unable to pass through a seed box. Hand seeding also allows for revegetation in restricted areas. The approach is usually more labour intensive than using a seed box.					
Willeroi Biodiversity Offset Area Management Plan	4.4.1	The following rates for direct seeding of native species (per ha) is recommended based on experience with broad-acre revegetation in similar climatic regions (ELA 2011): • Tree species - 200g/ha • Shrub species - 300g/ha • Grass species - 5,000g/ha.	Noted				
	4.4.2 Management z				1		
Willeroi Biodiversity Offset Area Management Plan	4.4.2	Targeted vegetation establishment will require tubestock planting with MZ3b and MZ3c in areas where natural regeneration of native species is unlikely to occur in a timely mannr. Revegetation of woodland/forest areas would include the planting of species characteristic of the local vegetation communities, including species from the BGW (e.g. White Box overstorey as well as appropriate understorey).					





Deference	Clause	Requirement	Evidence	A dik Finadina		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.4.2	Ripping will be undertaken in revegetation areas prior to tubestock planting when appropriate. The following guidelines should be followed when conducting ripping activities. • The preferred method of ripping is with a dozer (or similar piece of equipment) and a single tine, with planting occurring shortly after • Rip line spacing will be determined by the planting density of the tube stock required • Rip lines will follow the contours of the site in sloping areas to avoid formation of erosion channels and loss of topsoil • Ripping should be undertaken when the soil is moist but not saturated to allow a suitable medium for seedling establishment • Ripping prior to forecasted rainfall is preferable to allow moisture penetration within the rip line prior to tubestock planting • Operation of heavy equipment on wet areas will result in excessive soil compaction and should be avoided • In areas where ripping is likely to result in further weed infestation, additional weed management activities will be undertaken prior to ripping occurring.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.4.2	Tube stock are to be planted in 'Hiko' cells in autumn and spring to capitalise on greater soil moisture and reduced heat stress with follow up watering as required. Weed control including knockdown and pre-emergent herbicide are likely to be required prior to and after planting.	Not yet completed	Not Applicable			
	4.5 Collecting and pr	A seed collection program targeting local provenance species	Not yet completed			<u> </u>	
Willeroi Biodiversity Offset Area Management Plan	4.5	required for revegetation of the Willeroi property is required to ensure that the biodiversity values of the offset area is maintained. Table 4 2 contains the list of species associated with each vegetation community planned for revegetation within Willeroi Offset Area. The species list should be used as a guide for seed collection and propagation.		Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.5	Within the offset areas, local provenance seed will be targeted for collection. Seed collection will occur in other suitable areas when viable seed is considered to be available. Supplementation with non-local provenance seed (with the exception of horticultural varieties) may be required when adverse seasonal conditions (i.e. drought) affect the availability of local provenance seed. Alternatively, revegetation works may be delayed until sufficient stocks of local provenance species are available.	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.5	The seed collection program will focus on: Collection of seed at an appropriate time post flowering Aiming for the maximum quantity and quality of seed to be collected, and incorporate the correct degree of maturity to maximise germination Collection of seed from the ground and felled trees. However, where additional seed collection methods are required, these may include: For plants with a low density of occurrence, collection may require the placing of bags over the flowers after pollination. Alternatively, dropped seed may be collected by placing plastic on the ground below Seed from small plants close to the ground could be collected by machine harvesters or hand picking/pruning Seed pods from the upper and middle portions of the crown of trees are more desirable as they tend to be of better quality. Collection techniques for trees may include cherry picker, ladder or hoist attached to a vehicle, long handled pruners, rope saw etc o For grasses and other groundcover species, use of a forage harvester to collect seed and mulch material after seed maturity and the direct placement of this material on the post-mining landscapes is an appropriate means of collection and transferral of seed where safe machine access is possible Meeting the volume of local provenance seed required in order to propagate the numbers of tubestock required for revegetation works.		Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.5	A detailed recording of the seed collection and propagation process should be maintained. Details to be recorded include the seed collection methods, timing, seed storage, seed quantity, viability testing, propagation methods and successes. Such records will be fed back into the process to allow for continual improvement.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.5	Propagation of seed collected will be undertaken by personnel who are experienced in the propagation of native species. Propagation requirements will be determined by TCPL and communicated to the contractor responsible for seed propagation to ensure that seedlings are ready for planting each year. Planting activities will occur in autumn each year.		Not Applicable			
ı	4.6 Minimising the in		I		Г		
Willeroi Biodiversity Offset Area Management Plan	4.6	Impacts on fauna during activities being undertaken within the BOA are expected to be minimal. Any impacts will be minimised through implementation of the following management actions. • Creation of significant areas of fauna habitat resulting from the revegetation program • Weed control (Section 4.8) and feral animal control (Section 4.9) shall be undertaken • The on-site speed limit of 40 km/hr will be applied to access roads within the BOA.	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.7	Minimal surface disturbance works are planned for the BOA. Activities that will be undertaken include restoration and revegetation, stock exclusion and bushfire management. The Aboriginal Heritage Information Management System (AHIMS) register will be referred to should any surface disturbance works be undertaken.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.7	Any surface disturbance works will be managed as it is on the mine site in relation to Aboriginal cultural heritage, including: • In areas where there is a low potential for archaeological material and the disturbance footprint is considered minor, e.g. soil stripping in areas previously cleared for agriculture and >100 meters from a drainage line, works will be undertaken without the presence of site monitors. In areas other than those outlined previously, a representative member of the Narrabri LALC and representative member of the Gomeroi Narrabri People will be invited to monitor areas subject to any soil stripping activities	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.7	In the event of the discovery of a potential site or artefact, the following procedure will be followed: Work will cease in the area of discovery If the area of discovery is in deposited material, then work will also cease in the area where the material originated from The person discovering the artefact will notify the Property Manager who will ensure that work has ceased and area(s) is(are) cordoned off with tape The Property Manager will: Request a qualified archaeologist to attend the site and advise on its archaeological significance; Request the site monitor from the Narrabri LALC and the Gomeroi Narrabri People if not already present to attend and advise on its cultural significance in consultation with the qualified archaeologist Cif the find is determined to be a site, notify the OEH with the advice from the archaeologist and the Narrabri LALC and Gomeroi Narrabri People for determination of further procedures. If the find is confirmed as a site, the archaeologist will complete a Sites Register Card and forward to the OEH for inclusion on the Aboriginal Heritage Information Management System (AHIMS) database		Not Applicable			





Poforonco	Clause	Requirement	Evidence	Audit Einding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.7	6. Subject to the recommendations of the archaeologist, the Narrabri LALC and Gomeroi Narrabri People, the appropriate permit to transfer (under Section 85A of the National Parks and Wildlife Act 1974 (NPW Act)) will be applied for prior to further work being undertaken in the vicinity of the site. Any such action to disturb or transfer Aboriginal items will also require the development of excavation and/or salvage plans, which require the approval of the Director-General, in consultation with OEH. 7. The Property Manager will implement any other procedures or recommendations issued by the OEH 9. In addition to the procedure outlined above, any trees identified as having a culturally made scar(s) will be retained in situ. Where this is not possible, the tree will be cut down to preserve the scar and relocated to a designated protected area in consultation with the Aboriginal stakeholders and archaeologist 9. Should any bone(s) be uncovered, work should cease immediately in the area of the find, a suitably qualified person should be contacted to clearly identify the bone(s). If the bone(s) cannot be identified as animal bones, the police and Aboriginal stakeholder representatives will be advised of the discovery. Work will not recommence until both the police and the representatives of the Aboriginal community have given their permission to do so 9. Failure to report a discovery and those responsible for the damage or destruction occasioned by unauthorised removal or alteration to a site or to archaeological material may be prosecuted under the NPW Act (as amended).		Not Applicable			
	4.8 Controlling weeds						
Willeroi Biodiversity Offset Area Management Plan	4.8	Weed species within revegetation and regeneration areas have the potential to be a significant impediment to the success of revegetation and restoration activities as weeds compete with the tubestock species for moisture, nutrients and light. There are two aspects to weed control at the Willeroi Offset Area: 1. General control of environmental weeds 2. Targeted control of noxious weeds.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.8	Weeds will be controlled and monitored by an appropriately qualified contractor. Weeds may be removed from offset areas using mechanical means, or through the application of an approved herbicide. Herbicide application will occur in accordance with the requirements detailed on the relevant Material Safety Data Sheet.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.8	Once weeds have been removed from offset areas, inspections of weed controlled areas should be conducted to establish whether weed control programs have been successful. Where weed control activities have not been successful, further treatment of these areas will be undertaken. Seasonal conditions influence the extent of weed infestations and as such TCPL will continually monitor and update weed management areas.	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk Consequence Likelihood		
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.8	Noxious weeds that have been identified within the offset area should be treated in accordance with the Narrabri Shire Council's Weeds Management Plan specific to each weed species. Liaison should occur with local landholders, Narrabri Shire Council and other relevant government agencies to monitor the spread of weeds locally, and the extent of management activities being undertaken.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.8	Prior to entry into offset areas and when moving between sites of weed infestations, vehicles, machinery and boots will need to be cleaned in order to minimise the potential for further spread of weeds.	Not yet completed	Not Applicable			
	4.9 Controlling feral						
Willeroi Biodiversity Offset Area Management Plan	4.9	Feral pests will be controlled and monitored by an appropriately qualified and experienced contractor. All feral pest control activities will be undertaken in accordance with the requirements of the North West Local Land Services.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.9	Fencing of the BOA may be required to exclude feral pests from revegetation and regeneration areas. However, not all feral species will be controlled through this method and further implementation of management actions will need to occur. These actions may include trapping and/or baiting programs. Follow-up monitoring of the offset areas will be used to determine the effectiveness of trapping and/or baiting programs. Where feral species are detected within the monitoring program, these species will be targeted as part of the feral species management program in the following year.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.9	A 'Vertebrate Pest Management Program' will be developed in consultation with adjoining landholders (including OEH) for maximum efficacy. The proposed strategy includes: • Shooting feral goats using a high powered (calibre) rifle as humanely as possible (i.e. a single shot to head) • Joint aerial control of feral goats with OEH • Fox baiting using 1080 poison or pindone baits with safety precautions. Fox control will be undertaken through a baiting programme across the Offset Area as necessary. Fox control should be undertaken in autumn, when young foxes are leaving the den and dispersing and in spring, when foxes are breeding • Rabbit baiting (as fox) with warren destruction if found.	Not yet completed	Not Applicable			
	4.10 Controlling eros	ion			l l		
Willeroi Biodiversity Offset Area Management Plan	4.10.	Erosion and sediment control is required in MZ3c and along drainage lines that will undergo restoration works as part of MZ3. The proposed treatment is to: Remove stock grazing in riparian areas to prevent further erosion Actively manage areas mapped as MZ3c including jute matting to limited further erosion Periodically monitor the status of creek lines to identify significant erosion hotspots that require remediation (methods may include retaining natural snags to slow water velocity, battering off steep areas, jute matting and rock lining of creek banks) Increase vegetation cover in the creek lines and on the banks, making them more stable and less vulnerable to soil erosion.	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.10.	Control of grassy or herbaceous weeds along the creek lines will be selective until there is greater than 50% cover of native vegetation, as all vegetation cover (native and exotic) will aid in the stabilisation of the creek line. Selective weed control works will be required to prevent the creation of bare areas of soil. All woody weeds found occurring in the creek line will be controlled by the cut and paint method as this will retain the root structure of the plant in the bank and assist in holding the soil together. Woody debris from weed control is to be left in the creek bed provided the species cannot shoot from vegetative material.	Not yet completed	Not Applicable			
	1.11 Managing bush			ı	ı		
Willeroi Biodiversity Offset Area Management Plan	4.11	TCPL has developed a Bushfire Management Plan which outlines bushfire controls, emergency response actions and bushfire training requirements. This Offset Management Plan outlines further fire management procedures that are applicable to offset areas and should be incorporated into the Tarrawonga Bushfire Management Plan.	Noted				
Willeroi Biodiversity Offset Area Management Plan	4.11	The actions specific to the Willeroi property are: • Maintenance of access tracks within the offset area is required for fire management activities. • Use of a fire regime as a management strategy within the Willeroi Offset Area will require approval from relevant authorities prior to activities commencing. A site specific Fire Management Plan will be required in addition to the approvals required • Small fire breaks will be maintained along existing or new fence lines installed as part of this OMP. These breaks will double as access for the Rural Fire Service (RFS) in the event of wildfire, and will be maintained annually by slashing the fire breaks • Any wildfires that do occur in the Willeroi BOA in the interim will be managed through an appropriate response from the RFS to extinguish or contain the spread of the fire, including to let the fire burn to minimise potential damage caused by fire suppression activities • Mosaic burning shall be prescribed to reduce potential negative impacts. No prescribed burning will be implemented in MZ3 until planted trees are mature and able to withstand the impact of fire.		Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.11	Inspections of fuel load in offset areas will occur at a minimum of 12 monthly intervals, prior to the commencement of each bushfire season. Additional inspection of fire equipment is required every 6 months, and must be conducted by a fire equipment service provider.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.11	Any bushfire incident that occurs will be managed via the Whitehaven incident management process, and reported to relevant authorities as they occur. Reporting of bushfire incident details will also form a part of the Annual Environmental Monitoring Report (AEMR).	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	4.11	Access tracks through the BOA will be maintained for fire management.	Not yet completed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	4.12	Rubbish is present around the homestead site on the northern boundary of the Willeroi property. Rubbish within the BOA should be removed where it is appropriate. In some cases, rubbish may need to be left in-site as it presents major health concerns if removed, or it is providing an important habitat value.	Not yet completed				
	5 Ecological Monitori	ng					
Willeroi Biodiversity Offset Area Management Plan	5	Monitoring will be undertaken over a period of 20 years or until the completion criteria are met. Should the key completion criteria not be met in this timeframe or if some environmental event delays the recovery of the offset area, for example bushfire, TCPL will continue to manage the BOA until the completion criteria are met.	Noted, not commenced as yet apart from initial assessment	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	5	Once this is achieved, the offset area will then be managed in perpetuity via the security mechanism which is yet to be finalised. The monitoring program is based on sound statistical principals and is generally consistent with the standard flora and fauna assessment guidelines (DECC 2004). The monitoring program is broken up into the following components of quarterly visual inspections of disturbance factors including weeds and feral animals and annual flora and fauna monitoring.	Not yet completed	Not Applicable			
	5.1 Visual inspections	5					
Willeroi Biodiversity Offset Area Management Plan	5.1	Visual inspections of the BOA will be undertaken by the Property Manager quarterly, or in response to incidents, to detect: • Physical condition of fencing and gates • Disturbance factors including fire and unauthorised access e.g. fire wood collection • Presence/activity of feral animal species informed by regular inspections of the property (at least quarterly) and analysis of results of control measures (i.e. number of baits taken, number of animals shot) • Grazing pressure from over-abundant native herbivores (quarterly) • Presence of exotic weed species (as part of weed control measures).	Visual inspections have commenced	Compliant			
Willeroi Biodiversity Offset Area Management Plan	5.1	Inspection results will be reported annually as part of the Reporting Protocol	Noted, Offset has not been running for 12 months as yet.	Not Applicable			
	5.2 Flora monitoring						
Willeroi Biodiversity Offset Area Management Plan	5.2	The flora monitoring program outlined is adapted from DEC (2004) and ELA (2012a, 2012b). It is proposed that field surveys in permanent plots will occur annually and will be supplemented with additional photo monitoring. Field survey is to be undertaken in spring to maximise the detection of cryptic species.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	5.2	The location of permanent monitoring plots will be determined during the first year of monitoring to establish baseline data. Plots will be located by applying a random, stratified sampling regime. The Willeroi BOA will be divided into 26 strata (environmental sampling units) based on vegetation type, condition and management zone. To determine plot locations in the first round of survey, random points are allocated in each strata on a field map prior to field surveys to avoid potential sampling bias.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk	
Willeroi Biodiversity Offset Area Management Plan	5.2	Flora monitoring sites will be established to measure changes to vegetation condition in control (reference) sites (one in each of the seven vegetation communities in benchmark condition) and intervention sites (two in each vegetation type/management zone combination). The total number of permanent sites may be varied following baseline surveys.						
Willeroi Biodiversity Offset Area Management Plan	5.2	Flora monitoring plots are undertaken in a standard 50 m by 20 m plot	Noted					
	5.3 Fauna monitoring						_	
Willeroi Biodiversity Offset Area Management Plan	5.3	The fauna monitoring program comprises three elements: • Targeted surveys for the Regent Honeyeater, Swift Parrot and Greater Long-eared Bat in preferred habitat • Quarterly inspections and monitoring for presence of feral animal species.	Noted Note that there are only two elements listed in the management plan					
Willeroi Biodiversity Offset Area Management Plan	5.3	Except for the quarterly visual inspections, any fauna monitoring, including targeted surveys, will be undertaken by appropriately qualified, licensed and trained ecologists.	Noted					
Willeroi Biodiversity Offset Area Management Plan	5.3	The first year of monitoring will establish baseline data sets for ongoing comparison (i.e. ongoing data collected will be directly comparable using the same methodology). An adaptive management approach will be applied if early surveys detect significant species).	Noted					
Willeroi Biodiversity Offset Area Management Plan	5.3	Targeted surveys for the Swift Parrot and Regent Honeyeater in their preferred habitat (i.e. wintering flowering eucalypt species such as White Box) will be included in the monitoring program. Winter surveys will be undertaken at times when White Box is flowering (as informed by other inspections and monitoring, to be prescribed in the Offset Management Plan), and will consist of a suitably experienced ecologist traversing the offset area when White Box is confirmed flowering to detect the presence and habitat utilisation of either species. Remote recording devices (e.g. bird song meters) will be set up in areas of suitable habitat and left to record bird activity over a three to four week period during likely peak visitation periods (i.e. when there is mass flowering of White box). Targeted surveys will be conducted when local information (i.e. local bird observers and/or information from the National Recovery teams indicates Swift Parrots and/or Regent Honeyeaters are in the area).	Not yet completed	Not Applicable				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	5.3	Targeted surveys for the Greater Long-eared Bat will be undertaken within a representation of the potential habitat across the offset property (maximum of five sites). The targeted surveys will consist of both Anabat ultrasonic detection devices and harp trapping. The use of Anabat detection devices will identify the use of an area by Long-eared bats; however, they cannot be identified down to species level via Anabats. Two Anabat detection devices will be placed within potential habitat (fly-ways within woodland or over open bodies of water) over a period of two nights during spring. At each targeted site, harp traps will also be set up, given that the species forages below the tree canopy, often to ground level. Two harp traps will be placed within potential habitat (flyways within woodland) over a two over a period of two nights during spring.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	5.3	The results of monitoring will be analysed and compared to previous survey results to determine general population trends, including trends in the presence of feral animals. In the event that negative trends are identified indicating the decline of particular threatened species, appropriate amelioration measures will be recommended.	Noted				
Willeroi Biodiversity Offset Area Management Plan	5.3	If new populations of existing threatened species or additional threatened species are identified through the course of monitoring or other observations, records will be assessed by a qualified ecologist to advise on any changes required to the management of the BOA. This may result in: • Certain planned activities being postponed or modified • Modification of the monitoring program • Immediate survey and assessment of the new record, and an assessment of the impact of any proposed revegetation activities on it.	Noted				
Willeroi Biodiversity Offset Area Management Plan	5.3	If further threatened species or significant new records of existing threatened species are collected, the significance of such records will be reviewed, as will the likely impact of existing or proposed management activities, and any options for minimising impacts on these species.	Noted				
The state of the s	6 Contingency Plann		Noted]	1	
Willeroi Biodiversity Offset Area Management Plan	6	Condition 13.e of EPBC Act Approval 2011/5923 requires TCPL to describe the potential risks to successful management against the performance criteria, and provide a description of the contingency measures that would be implemented to mitigate against these risks.	Noted				
Willeroi Biodiversity Offset Area Management Plan	6	The following risk assessment considers the impact of risks from management actions in this OMP. The risk model is based on the Australian Standard ISO 31000:2009 (Standards Australia 2009).	Noted				





Deference	Clause	Requirement	Evidence	A alik Finadin -		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	6	To address the risks identified in Table 6 1 the Property Manager will: • Educate TCPL staff, contractors and neighbouring property owners/managers of the BOA objectives and location • Avoid potential impacts by following the recommended protocol in this OMP for each management action • Mitigate potential impacts by regular monitoring (Section 5) and applying corrective action under an adaptive management framework • Report incidences and responses in the Annual Report to facilitate managerial review and if necessary trigger systemic change of practices.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	6	Contingency plans are required if issues of non-compliance, exceedence, or new information about the management of the offset area present comes to light. The responsible agent for each incident, response management actions and reporting requirements are outlined in Table 6 3. Risks and their associated controls that relate to rehabilitation works on the Project area are provided in the RMP (GSS Environmental 2013).	Noted				
	7 Funding Requireme	nts					
Willeroi Biodiversity Offset Area Management Plan	7	Condition 13.g of EPBC Act Approval 2011/5923 requires TCPL to provide details of the funding requirements for the ongoing management activities within the BOA. TCPL will be responsible for funding the offset management activities on the Willeroi BOA.	Noted				
	8 Incidents and Non-0	•					
	8.1 Incident Reporting						
Willeroi Biodiversity Offset Area Management Plan	8.1	TCPL shall notify, at the earliest opportunity, the Director-General and any other relevant agencies of any incident that has caused, or threatens to cause, material harm to the environment. For any other incident associated with the project, TCPL will notify the Director-General and any other relevant agencies as soon as practicable after the TCPL becomes aware of the incident. TCPL will provide the Director-General and any relevant agencies a report on the incident, within seven days of the date of the incident, and such further reports as may be requested.	Noted, no such incidents	Not Applicable			
	8.2 Complaint Receipt	t and Response Procedure					
Willeroi Biodiversity Offset Area Management Plan	8.2	In order to receive, record and respond to any complaints in a timely manner, TCPL maintains a telephone complaints line for the purpose of receiving complaints from any member of the public in relation to its activities. The complaints line operates 24 hours per day, seven days per week, is publicly advertised and the details are supplied to adjacent landowners (TCPL 2013).	See other sections of the audit	Compliant			





Reference	Clause	Requirement	Evidence	Audit Einding		Risk	
Keierence	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	8.2	All complaints are recorded and responded to within 24 hours of the receipt of a complaint. A message bank will capture calls when they cannot be answered. Each complaint will be recorded on the Complaints Register, which will include the following details: 1. Date and time of complaint 2. Any personal details the complainant wishes to provide (alternatively, if no such details are provided then a note to that effect is recorded) 3. Nature of the complaint 4. The action taken by TCPL in relation to the complaint, including any follow-up contact with the complainant 5. If no action was taken by TCPL, the reason why no action was taken.	See other sections of the audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	8.2	The following protocol will be followed on all complaints received TCPL 2013): The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational) The cause of the complaint and any required remedial actions identified Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register. If necessary, the Environmental Manager will follow-up to confirm the source of the complaint is adequately mitigated.	See other sections of the audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	8.2	A copy of the Complaints Register will be kept by TCPL and made available to the CCC and the complainant (on request). A summary of complaints received every 12 months will be provided in the AEMR.	See other sections of the audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	8.2	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Environmental Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately.	See other sections of the audit	Compliant			
	8.3 Response to Nor						
Willeroi Biodiversity Offset Area Management Plan	8.3	Compliance with all approvals, plans and procedures is the responsibility of all personnel and contractors employed on or in association with the mine, and is developed through promotion of project ownership under the direction of the Operations Manager and Environmental Manager.	See other sections of the audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	8.3	The Environmental Manager or Environmental Officer and/or Operations Manager undertakes regular inspections, internal audits and initiates directions identifying any remediation/rectification work required, and areas of actual or potential non-compliance, with all directions provided to the relevant party in writing and/or diarised (TCPL 2013).	See other sections of the audit	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Ciause			Addit Filluling	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	8.3	Non-compliances with regulations, licences or approvals will be reported to the relevant authority, together with details of the corrective actions taken to avoid future occurrences (TCPL 2013).	See other sections of the audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	8.3	A review of TCPL's compliance with all conditions of the Development Consent, mining lease and all other approvals and licences is undertaken as part of each AEMR/Annual Review.	See other sections of the audit	Compliant			
	9 Reporting and Doc	umentation					
Willeroi Biodiversity Offset Area Management Plan	9	EPBC Approval (2011/5923) requires TCPL to prepare and submit reports that detail the progress of management activities undertaken in the BOA, and the outcomes of those activities. These reports should contain the results of any monitoring, inspections or surveys undertaken in the previous 12 months including details of date, time, location, nature of inspection/activity, name of person undertaking inspection/activity, observations and the photographs from the monitoring plots. Annual reports shall be provided every 12 months by March 31. The annual report will also include the results of monitoring undertaken as required by this OMP	Not twelve months into the BOMP	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	9	The annual 'Willeroi' BOA report will be incorporated into TCPL's Annual Environmental Management Report (AEMR) and Annual Review reporting period, i.e. 1 April-31 March, and will consist of two parts: • Summary of the implementation of management actions within the BOA including inspection results and review findings; and • The results of the flora and fauna monitoring program.	Not twelve months into the BOMP	Not Applicable			
Willeroi Biodiversity Offset Area	9	A series of reporting protocols also need to be completed	Not yet completed	Not Applicable			
Management Plan	9 1 Vegetation moni	Litoring and reporting					
Willeroi Biodiversity Offset Area Management Plan	9.1	Annual vegetation monitoring reporting is to include a written summary of the methodology and the current year's findings for each vegetation plot, including the average results of each variable recorded. Photo monitoring records and field data sheets will be included in an appendix. A flora species list with corresponding plot numbers where species were recorded is also to be included in an appendix.	Not yet completed	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	9.1	A discussion section is to be included that compares the results from the current monitoring year with previous years either through actual annual records or a mean value where several previous years are being compared. The annual report is to include statistical graphs illustrating changes in diversity and cover/abundance of each attribute recorded within each management zone.	Not yet completed	Not Applicable			
	9.2 Fauna monitorin						
Willeroi Biodiversity Offset Area Management Plan	9.2	The report is to include the results of any surveys/monitoring undertaken for the Swift Parrot, Regent Honeyeater and Greater Long-eared Bat, and habitat assessments. The results of monitoring will be analysed and compared to previous survey results to determine general habitat usage trends (if any).	Noted				
	9.3 Report recomme	endations and conclusion					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	9.3	The annual OMP implementation report will provide a summary of all the actions implemented during the previous year. Any significant events that occurred during the year (e.g. wildfire, outbreak of any weeds or incidence of any new threats) and any recommended changes to the management actions, their duration, intensity or relative priority identified through the annual review will be included.	Noted				
Willeroi Biodiversity Offset Area Management Plan	9.3	A concluding section is to be prepared within the annual on-site BOA report that highlights and describes significant findings, either positive or negative. Changes to any management recommendations for the following year will also be suggested.	Noted				
	9.4 Review and Audi						
Willeroi Biodiversity Offset Area Management Plan	9.4	The annual OMP implementation report will provide a summary of all the actions implemented during the previous year. Any significant events that occurred during the year (e.g. wildfire, outbreak of any weeds or incidence of any new threats) and any recommended changes to the management actions, their duration, intensity or relative priority identified through the annual review will be included.	Noted				
Willeroi Biodiversity Offset Area Management Plan	9.4	The OMP is to undergo a review and revision every five years. It is expected that baseline environmental data can be supplemented over time as seasonal trends are recorded. These results should be considered to provide for adaptive management	Noted				
Willeroi Biodiversity Offset Area Management Plan	9.4	The land management activities (Section 4) should be the focus of the review: Investigate how these strategies are measuring up against the overall objectives and the performance criteria set in the previous three years Identify any poor results / non-conformances and rectify any problems encountered or better approaches identified in reporting and auditing processes Revise or set performance criteria for the next three year period.					
Willeroi Biodiversity Offset Area Management Plan	9.4	Improved information gathered as a result of BOA monitoring or recent studies should be used to refine land management activities and performance criteria. Any updates to the OMP will be provided to DP&I and DoE for approval.	Noted				
Willeroi Biodiversity Offset Area Management Plan	9.4	TCPL shall review and if necessary revise this OMP within six months of the completion of the Leard Forest Mining Precinct Regional Biodiversity Strategy (condition 47 of the PA).	Not approved	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	9.4	In general, the OMP will be revised due to: • Deficiencies being identified • Recommendations from the annual report • Changing environmental requirements (including any project conditions of approvals) • Improvements in knowledge or new technology becomes available • Change in legislation or relevant approvals • Change in the activities or operations associated with the Tarrawonga Mine.	Noted				
Management Plan	9.4 9.5 Auditing	In general, the OMP will be revised due to: • Deficiencies being identified • Recommendations from the annual report • Changing environmental requirements (including any project conditions of approvals) • Improvements in knowledge or new technology becomes available • Change in legislation or relevant approvals • Change in the activities or operations associated with the	Noted				





Deference	Clause	Requirement	Evidence	Audia Sindina		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Willeroi Biodiversity Offset Area Management Plan	9.5	TCPL will undertake annual internal audits to align with its AEMR/Annual Review and Mining Operations Plan reporting requirements, as well as independent audits every 3 years. The annual audit will focus on the compliance with the requirements of the OMP and where corrective action is required.	Noted, not yet required	Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	9.5	An annual audit and compliance report will be prepared which will include the methodology of the audit, actions implemented in accordance with the OMP, non-compliances with the OMP, and corrective actions taken. A summary of this report will be included in the AEMR/Annual Review.		Not Applicable			
Willeroi Biodiversity Offset Area Management Plan	9.5	In accordance with Schedule 5, Condition 10 of the DP&I PA, an independent audit will be undertaken by the end of June 2014 and every 3 years thereafter, unless the Director-General directs otherwise.	This audit, timing not compliant	Not Compliant Administrative			
Willeroi Biodiversity Offset Area Management Plan	9.5	The audit will be led by a suitably qualified auditor, and include experts in noise, air quality, water, ecology, and any other fields specified by the Director-General, who are endorsed by the Director-General. Specifically in relation to this OMP, the audit will include: • Consultation with relevant agencies • Assessment of the environmental performance in complying with this OMP • Review of the adequacy of this OMP and • Recommend measures or actions to improve the environmental performance in relation to the requirements off this OMP.	This audit	Compliant			
Willeroi Biodiversity Offset Area Management Plan	9.5	A document will be prepared within 3 months of commissioning the audit, or as otherwise agreed with the Director-General, which includes the audit report, together with responses to recommendations contained in the audit report.	This audit	Compliant			





Rehabilitation Management Plan 3.1 Disc Mining Land Use Rehabilitation Management Plan 3.1 and General Consequency (somewhat or substantial productivity comparable with the pre-mining and use vision is to reinstate a sustainable balance of pre-mining land use, showing land use,	Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
3.1 Post Mining Land Use and Rehabilitation Objectives 3.2 Post Mining Land Use Tarrawongs's post-mining land use vision is to reinstate a sustainable balance of pre-mining land uses, being biodiversity conservation (Leard State Forest) and agriculture. Consequently, the post mining and agriculture. Consequently, the post mining and agriculture acreas of native vegetation communities, and agricultural areas capable of sustainable agricultural and agricultural areas capable of sustainable agricultural and agricultural areas capable of sustainable agricultural areas data for a sustainable agricultural and agricultural areas capable of sustainable agricultural areas data for a sustainable agricultural areas will be proximate to existing agricultural areas will be designed and constructed to have a minimal adverse impact upon the desired land use outcomes and minimise impacts to surface and groundwater quality and hydrology 3.2 Tarrawongs Rehabilitation Goal and Objectives Rehabilitation Management Plan 3.2 Tarrawongs's post-mining land uses vision is to reinstaite as surface and groundwater mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform reconstruction, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise	Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk
See MOP Plans - compiles								
Tarrawonga's post-mining land use vision is to reinstate a sustainable balance of pre-mining land use, being biodiversity conservation (Leard State Forest) and agricultura. Consequently, the post mining and ground uses, being biodiversity conservation (Leard State Forest) and agricultural constitution. In another will include areas of paths of sustainable agricultural productivity comparable with the pre-mining capability of agricultural areas capable of sustainable agricultural productivity comparable with the pre-mining capability of agricultural areas subject to stude by the project. Rehabilitation Management Plan 3.1 and enhance the regional biodiversity outcomes. Agricultural areas will be proximate to existing agricultural adverse impact upon the desired land use uncomes and minimise impacts to surface and groundwater quality and hydrology uncomes and minimise impacts to surface and groundwater quality and hydrology uncomes and minimise impacts to surface and groundwater quality and hydrology and proposed to the project is that the status of rehabilitation Goal and Objectives are an applicated for the project is that the status of rehabilitation of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Behabilitation Management Plan 3.2 Tarrawonga Rehabilitation of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the norminated standards and/or completion criteria for land use, landform reconstruction, landform between met or the relevant Minister(s) otherwise		•						
Rehabilitation Management Plan 3.1 3.2 Rehabilitation Management Plan 3.3 Rehabilitation Management Plan 3.4 Rehabilitation Management Plan 3.5 Rehabilitation Management Plan 3.6 Rehabilitation Management Plan 3.7 Rehabilitation Management Plan 3.8 Rehabilitation Management Plan 3.9 Rehabilitation Management Plan 3.1 3.1 3.1 3.1 3.1 3.2 Tarrawonga Rehabilitation Management Plan 3.2 Tarrawonga Rehabilitation folial mode service of the satisfaction of the relevant Minister(s) and minister(s) and that all relevant mining tenement and Project Approval conditions have been met or the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met or the relevant Minister(s) and that all relevant inardier reconstruction, landform tability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise		3.1 Post Mining Land			1	1		
Rehabilitation Management Plan 3.1 Leard State Forest to provide connectivity and enhance the regional biodiversity outcomes. Agricultural areas will be proximate to existing agricultural lands The final void will be designed and constructed to have a minimal adverse impact upon the desired land use outcomes and minimise impacts to surface and groundwater quality and hydrology 3.2 Tarrawonga Rehabilitation Goal and Objectives Rehabilitation Management Plan Tarrawonga's mine closure goal for the Project is that the status of rehabilitation at relinquishment will be to the satisfaction of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise Not Applicable	-	3.1	sustainable balance of pre-mining land-uses, being biodiversity conservation (Leard State Forest) and agriculture. Consequently, the post mining landform will include areas of native vegetation analogous to the adjacent native vegetation communities, and agricultural areas capable of sustainable agricultural productivity comparable with the pre-mining	Noted				
Rehabilitation Management Plan 3.1 final void will be designed and constructed to have a minimal adverse impact upon the desired land use outcomes and minimise impacts to surface and groundwater quality and hydrology 3.2 Tarrawonga Rehabilitation Goal and Objectives Tarrawonga's mine closure goal for the Project is that the status of rehabilitation at relinquishment will be to the satisfaction of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise	_	3.1	Leard State Forest to provide connectivity and enhance the regional biodiversity outcomes. Agricultural areas will be proximate to existing agricultural	See MOP Plans - complies	Compliant			
Rehabilitation Management Plan 3.2 Tarrawonga's mine closure goal for the Project is that the status of rehabilitation at relinquishment will be to the satisfaction of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform reconstruction, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise	_	3.1	final void will be designed and constructed to have a minimal adverse impact upon the desired land use outcomes and minimise impacts to surface and groundwater	Final design not yet required but it will further detail how the final void will meet	Compliant			
Rehabilitation Management Plan 3.2 Tarrawonga's mine closure goal for the Project is that the status of rehabilitation at relinquishment will be to the satisfaction of the relevant Minister(s) and that all relevant mining tenement and Project Approval conditions have been met. Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform reconstruction, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise		3.2 Tarrawonga Rehal	Dilitation Goal and Obiectives	l.				
Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform reconstruction, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise	Rehabilitation Management		Tarrawonga's mine closure goal for the Project is that the status of rehabilitation at relinquishment will be to the satisfaction of the relevant Minister(s) and that all relevant mining tenement and Project Approval	Noted				
	Plan		Disturbed land would be considered suitable for surrender when the nominated standards and/or completion criteria for land use, landform reconstruction, landform stability, revegetation, and beneficial water use have been met or the relevant Minister(s) otherwise accept the rehabilitation status.	No rehab is yet suitable for relinquishment.	Not Applicable			
4.1.1 Tarrawonga Environmental Risk Management		4.1.1 Tarrawonga Env			1	1		
Tarrawonga has implemented an Environmental Management Strategy (EMS), supported by a comprehensive set of Environmental Management Plans (EMPs) to manage environmental risks on all Tarrawonga lands, including rehabilitation areas. These plans have been developed and implemented by TCPL in accordance with Project Approval PA 11_0047 and regulatory requirements including EPL 12365. EMPs are supported by an environmental monitoring program which includes monitoring meteorological conditions, air quality, noise, blasting, surface water and groundwater.	_	4.1.1	Strategy (EMS), supported by a comprehensive set of Environmental Management Plans (EMPs) to manage environmental risks on all Tarrawonga lands, including rehabilitation areas. These plans have been developed and implemented by TCPL in accordance with Project Approval PA 11_0047 and regulatory requirements including EPL 12365. EMPs are supported by an environmental monitoring program which includes monitoring meteorological conditions, air quality, noise,					
4.2 Key Risks to Rehabilitation		4.2 Key Risks to Rehal			<u> </u>			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	4.2	Key risks to rehabilitation and mine closure will be formally assessed in a Mine Closure Risk Assessment undertaken prior to submission of the draft Final Void and Mine Closure Plan in 2019.	Noted	Not Applicable			
	4.2.1 Landform Stabil	ity					
	Surface Water Manag	gement and Erosion Control			Ī	·	
Rehabilitation Management Plan	4.2.1	Erosion and sedimentation is managed in accordance with the TCPL Erosion and Sediment Control Plan, which is a component of the Water Management Plan. Failure to adequately manage surface water in rehabilitation areas may result in landform instability, significant erosion features and offsite transport of sediment.	Noted				
Rehabilitation Management Plan	4.2.1	Management and mitigation measures include but are not limited to: Installation of all erosion and sediment controls in accordance with an area-specific Erosion and Sediment Control Plan (ESCP);	Sighted on-site	Compliant			
Rehabilitation Management Plan	4.2.1	Selective handling and amelioration of spoils and soils to minimise exposure of unsuitable (e.g.: dispersive) materials;	Sighted on-site and noted in a number of management pans and procedures	Compliant			
Rehabilitation Management Plan	4.2.1	[■] Constructing final landform drainage structures to meet design storm events in accordance with regulatory requirements and industry guidelines (refer to Section 6.3.2); and	MOP plans, approved by DRE and EA plans approved by DP&E, no steep slopes observed on-site, no long slopes observed, some erosion but not unmanagable and not ignored.	Compliant			
Rehabilitation Management Plan	4.2.1	Rapid stabilisation of topsoiled rehabilitation areas with cover crops and / or mulches.	Sighted on-site, note no mulches are used at present	Compliant			
	Saline and Sodic Mate		<u>, </u>				
Rehabilitation Management Plan	4.2.1	Management and mitigation measures include but are not limited to: ① Overburden and interburden selective handling to ensure the final outer surfaces of overburden emplacements (and structures such as drainage elements) are constructed with suitable non-sodic or low sodicity material where possible;	Overburden not tested for sodicity, no evidence of dispersive materials destabilising slopes. Slopes are generally holding up well even with some poorly vegetated areas.	Compliant			
Rehabilitation Management Plan	4.2.1	② Ameliorating any sodic materials that may be exposed or near- surface to minimise dispersivity. Management of sodic materials is discussed further in Section 6.3.2; and	No amelioration has been required in the audit period.	Not Applicable			
Rehabilitation Management Plan	4.2.1	Treating any identified occurrence of sodic material in the final void walls in accordance with procedures to be documented in Section 10.	Noted, not yet required.	Not Applicable			
	Major Storms and Flo	ooding (Goonbri Creek)					
Rehabilitation Management Plan	4.2.1	Management and mitigation measures include: All landforms and drainage structures will be designed to withstand major storm events in accordance with regulatory requirements and industry guidelines	Erosion and sediment control design report sighted. Report stated complaince with the Blue Book and other relevant guidelines.	Compliant			
Rehabilitation Management Plan	4.2.1	☑ The final void design will consider the optimal catchment reporting to the void to minimise the potential for flooding the void; and	Noted, design not finalised to this degree yet	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Filluling	Consequence	Likelihood	Risk
Rehabilitation Management Plan	4.2.1	Pingarian corridors (Goonbri Creek alignment and key drainage paths) will be stabilised with strategic placement of structures (including large rocks and woody debris). Riparian vegetation species able to withstand high velocity flows (e.g.: lomandras) will be established in riparian corridors.	This work has not been completed.	Not Applicable			
	Bushfire						
Rehabilitation Management Plan	4.2.1	Bushfire risk on all Tarrawonga owned lands is managed in accordance with the Bushfire Management Plan and procedures in the Biodiversity Management Plan. Management and mitigation measures to minimise the impacts of bushfire are outlined in Section 6.5.7 and include: ② Selection of fire-tolerant species in native vegetation areas; ② Maintaining fire breaks by maintaining cleared easements and fence-lines; ② Maintain fire-fighting access tracks and water sources (sediment dams) in rehabilitation areas; ② Monitoring fuel loads and considering mosaic back-burning programs (where appropriate) to reduce fuel loads; and ② Maintaining sufficient viable seed for key native vegetation	Generally compliant with the Bushfire MP though fuel load monitoring is not done. To date no fuel load reduction programs have been undertaken, there was no evidence of excessive fuel load build up in the site inspection.	Compliant			
	Severe or Prolonged	species to re-seed areas after bushfire events.					
Rehabilitation Management Plan	4.2.1	Management and mitigation measures include but are not limited to: ② Optimise use of local provenance seed tolerant of the region's drought regime within the species mix for rehabilitation areas. Deep rooted native grassland species will be included in pasture mixes to maximise retention of ground cover in pasture and native grassland areas during drought events; ② Ameliorate soils where necessary to optimise infiltration and water holding capacity in the soil profile prior to tubestock planting and direct seeding. Tarrawonga is also trialling tubestock planting methods including mounding and direct seeding / oversowing through cover crop stubble to increase drought tolerance by maximising moisture retention; and ② Monitor rehabilitation following drought periods to identify areas where re-seeding and replanting is required. Tarrawonga will maintain a contingency supply of viable seed for key native vegetation species until relinquishment that is sufficient to reseed rehabilitation areas in the event of a widespread rehabilitation failure.	elsewhere in audit	Compliant			
	Regional Insect Attac	k and Disease					
Rehabilitation Management Plan	4.2.1	Management and mitigation measures include but are not limited to: Source disease-free local provenance seed and trial seed from multiple sources to identify seed sources that produce resilient vegetation;	This occurs	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Filluling	Consequence	Likelihood	Risk
Rehabilitation Management Plan	4.2.1	Monitor vegetation health and use controls such as fungicide / insecticide spraying to protect juvenile vegetation where appropriate; and	Not required in the audit period	Not Applicable			
Rehabilitation Management Plan	4.2.1	Monitor rehabilitation following periods of disease or insect attack to identify areas where re-seeding and replanting is required.	Property Manager, monthly inspections	Compliant			
		or Agricultural Suitability Class 3 Land					
	Agricultural Soil Mana	agement					
Rehabilitation Management Plan	4.2.3	Management and mitigation measures include but are not limited to: Characterise soil resources prior to stripping to identify suitable growth media resources and any amelioration required to meet rehabilitation objectives; and	See comments elsewhere, no agricultural class soil stripped as yet.	Not Applicable			
Rehabilitation Management Plan	4.2.3	☑ Topsoil stripping, stockpiling and respreading will be undertaken in accordance with the procedures documented in this plan and the MOP to protect and enhance soil biochemical and physical properties.		Compliant			
	4.2.4 Water Quality						
	Acid Mine Drainage						
Rehabilitation Management Plan	4.2.4	Tarrawonga will identify and emplace all PAF materials under at least 15 m of non-acid forming (NAF) material to minimise the potential for AMD.	No PAF materials identified in the audit period	Not Applicable			
	4.2.5 Regulator and /	or Community Expectations					
	Failure to Integrate R	ehabilitation with Boggabri Coal mine					
Rehabilitation Management Plan	4.2.5	Tarrawonga and Boggabri have committed to integrating the final landform and vegetation communities across the shared mine lease boundary. Failure to adequately consult with Boggabri regarding landform design and native vegetation species selection presents a risk to realising the Tarrawonga Rehabilitation Strategy	TCL and BC have been in consultation and mine planning is well advanced for the integration of the two dumps.	Compliant			
Rehabilitation Management Plan	4.2.5	Tarrawonga will work closely with Boggabri mine planners and engineers to design and construct an integrated final landform across the shared mine lease boundary. Tarrawonga will also liaise with Boggabri regarding native vegetation species selection and propagation. Progressive rehabilitation planning for the Northern Emplacement Area will be developed well in advance of the integration of the overburden dumps. The detail relating to rehabilitation actions within this shared zone will then be included in a future revision of this Rehabilitation Management Plan.	This is occurring	Compliant			
	Regulator and Comm	unity Expectations					
Rehabilitation Management Plan	4.2.5	Evolution of regulator or community expectations regarding post- mining land-uses may present a risk to achieving relinquishment. Tarrawonga propose to undertake ongoing consultation with key stakeholders as mining progresses and adopt an open approach to evolving expectations regarding post-mining landuse at Tarrawonga.	CCC, annual DRE inspections, DP&E inspections, meetings and managemtn plus coordination with surrounding sites.	Compliant			





Defenses	Clause		Requirement	Evidence	A codia Pio dio o		Risk	
Reference	Clause				Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	4.2.5	rehabilitation implem proactively identify a literature, and indust	dvances in technical knowledge and scientifi	Noted c				
	5 Rehabilitation Plann	ning						
	5.1 Overview							
Rehabilitation Management Plan	5.1	with the project feas process. Initial rehab were established in t	oilitation objectives and land-use outcomes he stakeholder undertaken for the Project EA, and initial i identified and	Noted				
Rehabilitation Management Plan	5.1	Development of rehacriteria draws upon; research projects, inc Whitehaven's experioperations. Tarrawo process to assess the rehabilitation works completion criteria ti analysis, regular risk engagement.	abilitation methodologies and completion scientific knowledge and dustry best management practices and ences across all mining inga adopts a continuous improvement	Noted				
	5.2 Stakeholder Enga							
	5.2.1 Consultation in	the Project EA Process				1		
Rehabilitation Management Plan	5.2.1	during preparation at the Project EA. Agency Date	Table 5.1 – Summary of Consultation Rehabilitation Themes Discussed pill and 5 Rehabilitation 8 Rehabilitation Themes Discussed 10 Rehabilitation Themes Discussed 10 Rehabilitation Themes Discussed 11 Rehabilitation Themes Discussed 12 Rehabilitation Themes Discussed 12 Rehabilitation Themes Discussed 12 Potential Impacts on surface water and groundwater resources including, adapticate licensed water users and groundwater dependent ecosystems.	S Noted				
	F. 2.2 Stakeholder Fra	ragement for the D-b-	hilitation Management Dlan					
	5.2.2 Stakeholder Eng	gagement for the Reha	bilitation Management Plan					





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Kelefelice	Clause			Addit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	5.2.2	In accordance with Project Approval PA 11_0047 this Rehabilitation Management Plan was prepared in consultation with the Department of Planning and Infrastructure (DP&I), DTIRIS-DRE, New South Wales (NSW) Office of Water (NOW), NSW Office of Environment and Heritage (OEH), Namoi Catchment Management Authority (Namoi CMA) and Gunnedah Shire Council.	Noted				
Rehabilitation Management Plan	5.2.2	Stakeholders and agencies identified above were contacted in May 2013 and provided a draft of this Rehabilitation Management Plan and invited to make comment. The first version of the Rehabilitation Management Plan was also referred to DTIRIS_DRE and DP&I during May, with a commitment to provide a revised draft for approval upon receipt of relevant comments from the above mentioned stakeholders and agencies.	Noted				
Rehabilitation Management Plan	5.2.2	Tarrawonga will engage in further ongoing consultation with all key stakeholders during the life of mine specific to: Rehabilitation methodology and management of rehabilitation areas; Mine closure requirements and activities; and Post-mining land use.	See above - Noted, the rehab methodology consultation is on-going, the others are not yet required.	Compliant			
	5.3 Rehabilitation Do						
	Final Void						
Rehabilitation Management Plan	5.2.2	Tarrawonga has approval to retain one final void at the eastern extent of the project site (Figure 1.3). Tarrawonga has developed a conceptual final void design described in the Project EA. To provide landform stability the void highwall (at the eastern and northern faces) will be cut at an angle of approximately 60° while the lowwalls will be re-graded to between (approximately) 10 and 15°.					
Rehabilitation Management Plan	5.2.2	The final void design will be developed to optimise the void size and catchment reporting to the void to minimise long term surface water and groundwater impacts. The void will be designed to slowly fill to an initial target equilibrium water level (based on pre-mining groundwater data) of 240 to 260 AHD. A low permeability groundwater barrier will be constructed to minimise groundwater losses into the final void	Noted				
	Water Management	Areas			<u> </u>		
Rehabilitation Management Plan	5.2.2 Agricultural Rehabilit	The Water Management Area domain constitutes all areas with a final land-use of water diversion and collection, including the major rock lined channels and drop structures, and dams that are retained in the final landform.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	5.2.2	A total of 210 ha of agricultural land (Class 3 Agricultural Suitability) will be re-established in the final rehabilitated landform. Up to 160 ha of agricultural land capable of being used for a combination of pasture production for grazing and occasional cropping will be reestablished on the lower terraces of the open cut infill area (Figure 1.2). This area, combined with the rehabilitated mine facilities area (40 ha) and the temporary soils stockpile location near the permanent Goonbri Creek realignment (10 ha) will provide for the total of 210 ha of Class 3 Agricultural Suitability land in the final landform.	Noted, no yet established, nor is the soil stripped from the agricultural area adjacent to Goonbri creek	Not Applicable			
Rehabilitation Management Plan	5.2.2	Agricultural soil resources are not anticipated to be disturbed prior to Year 12 of the project. Consequently this Rehabilitation Management Plan includes conceptual methodologies for rehabilitating Class 3 agricultural land only; detailed methodologies to rehabilitate Class 3 agricultural lands will be developed following further research and consultation with stakeholders and will be documented in a revision to this plan and in future MOPs.		Not Applicable			
	Woodland Rehabilitat	ion Areas					
Rehabilitation Management Plan	5.2.2	In accordance with EPBC Referral 2011/5923 Condition 23, Tarrawonga's rehabilitation program will result in no less than 752 ha of native vegetation open woodland and forest communities within the Project boundary. Woodland Rehabilitation Areas will include at least 13 ha commensurate with White Box Yellow Box Blakely's Red Gum Grassy Woodland and Derived Native Grassland Endangered Ecological Community (Box-Gum Woodland EEC/CEEC) on areas disturbed by mining.	Noted, see Ecological report - Tarrawonga Disturbance Limits Analysis (2014)	Compliant			
Rehabilitation Management Plan	5.2.2	Woodland areas will be established on slopes and upper terraces of the Northern and Southern Emplacement Areas. Tree species selection and planting densities adjacent to Boggabri and Leard State Forest will be determined considering integration with the Boggabri waste emplacement area and Leard State Forest. The Biodiversity Management Plan provides further detail on the rehabilitation of woodland species and integration with Boggabri Mine.	the MOP details these measures though some are outside the current MOP period. Not yet established	Not Applicable			
Rehabilitation Management Plan	5.2.2	The Woodland Rehabilitation Area domain also includes riparian corridors along constructed channels and the permanent Goonbri Creek Realignment	Noted				
	Permanent Goonbri C						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit I IIIdilig	Consequence	Likelihood	Risk
Rehabilitation Management Plan	5.2.2	Works will be undertaken between approximately Year 12 and Year 15 of the project to construct a realigned section of Goonbri Creek. The low flow channel of the alignment, and an (approximate) 40 m buffer on either bank will be revegetated with species that are commensurate to the existing Goonbri Creek riparian and floodplain vegetation (refer to Table 7.1) prior to	Currently in year 1 of the project	Not Applicable			
Rehabilitation Management Plan	5.2.2	commissioning in (approximately) Year 15. The Goonbri Creek Realignment works will include construction of a low permeability barrier between the creek alignment and final void to limit alluvial groundwater drawdown into the void. A flood bund will also be constructed to minimise risks of flooding the void. As outlined in Section 4.1.2 Tarrawonga will develop a Permanent Goonbri Creek Alignment Management Plan prior to commencing construction that will include: Detailed design specifications; Rehabilitation objectives and activities including vegetation selection and geomorphic performance and completion criteria based on baseline data from Goonbri Creek; and	Not yet required	Not Applicable			
	6 Rehabilitation Metl	A monitoring and maintenance program.					
	6.2 Recommissioning						
	6.2.1 Decommissioni						
Rehabilitation Management Plan	6.2.1	Tarrawonga will decommission all mining related infrastructure (except some sediment dams retained in the final landform) prior to relinquishment of the mine lease. Fixed plant, built infrastructure and services will be rehabilitated progressively as infrastructure items and plant become redundant to facilitate timely rehabilitation and relinquishment.	Not yet required	Not Applicable			
Rehabilitation Management Plan	6.2.1	A decommissioning plan will be prepared for all significant infrastructure and fixed plant. The decommissioning plan will include a risk assessment to identify potential risks such as land contamination. If required a land contamination assessment will be completed and remediation will be undertaken in accordance with the recommendations of the assessment.	Not yet required	Not Applicable			
Rehabilitation Management Plan	6.2.1	Decommissioning works will be undertaken in accordance with the decommissioning plan and will be verified prior to issuance of a decommissioning certificate.	Not yet required	Not Applicable			
	6.3 Landform Establis						
Rehabilitation Management Plan	6.3.1 Landform Estab	The post-mining landform will be profiled to produce a safe, structurally stable landform that complements the surrounding landscape. Final landforms will have acceptable slopes and unimpeded drainage lines. Slopes are generally to be graded to no more than 10 degrees and no more than 18 degrees without DTIRIS approval, as per DTIRIS guidelines.	Noted, no steep slopes identified in the site inspection	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.3.1	The Northern Emplacement will be shaped to integrate with the Boggabri waste rock emplacement to the north and the Leard State Forest to the east, providing continuity between the Tarrawonga and Boggabri Coal post-mining landforms, and habitat connectivity with undisturbed vegetation communities at Leard State Forest. Both the Northern Emplacement and Southern Emplacement will be shaped to integrate with the open cut infill area and include lower terraces to be graded to achieve a Class 3 agricultural land class suitable for the desired agricultural land-use.	See MOP, planned but implementation is a number of years from from the audit date.	Not Applicable			
Rehabilitation Management Plan	6.3.1	Waste rock will be selectively handled and blended to enhance landform stability and avoid emplacement of potentially acid forming material in concentrated areas. As outlined in Section 3.2.5 where PAF-LC material is identified it will be encapsulated with at least 15 m of NAF to minimise potential for acid leachate.	No PAF materials identifed to date (Mine Planner)	Not Applicable			
Rehabilitation Management Plan	6.3.1	Dispersive materials will be avoided for material layers at the final landform surface where practical to minimise potential for significant scouring or land slumping. Non-sodic and low sodicity materials will be selectively handled and preferentially emplaced at or near the surface. Where there are potentially dispersive materials emplaced at or near the surface, appropriate amelioration with lime or gypsum will be undertaken to stabilise soils, particularly foundation materials for earthworks structures such as contour banks.	dealt with differently to general overburden	Compliant			
	Agricultural Rehabili						
Rehabilitation Management Plan	6.3.1	By definition Agricultural Suitability Class 3 land requires soil conservation practices to support grazing and rotational cropping land-use. Lower terraces to be rehabilitated to an Agricultural Suitability Class 3 standard will be graded to be no more than 2 per cent and incorporate contour banks to: Produce free draining landforms that minimise soils at depth becoming water logged; and Impede runoff velocities to sub-erosive levels.	SEE MOP, not yet required	Not Applicable			
Rehabilitation Management Plan	6.3.1	To optimise growing conditions for deeper rooted native grasses and crops, the surface layers of spoil will be: Selectively emplaced with non – sodic or low sodicity spoils and subsoils; Rock-raked to remove any large rocks that may impede ripping tynes and agricultural equipment; and Deep ripped along the contour and ameliorated with coarse gypsum prior to soil re-spreading.	This is done, detailed in MOP and confirmed at site interview	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.3.1	Woodland rehabilitation areas will be graded to generally 10° and incorporate contour banks as per agricultural rehabilitation areas for surface water control. To optimise growing conditions for deeper rooted native trees, the surface layers of spoil will be: Selectively emplaced with non – sodic or low sodicity spoils and subsoils (refer Section 6.2.3); and Deep ripped along the contour and ameliorated with coarse gypsum (where required) prior to soil respreading.	This is done, detailed in MOP and confirmed at site interview	Compliant			
	Final Void						
Rehabilitation Management Plan	6.3.1	The principal objective of landform establishment activities associated with the final void is to create a safe and stable landform that is non-polluting and has no long term impacts on surface and groundwater resources. A detailed methodology to treat the final void will be developed in a Final Void and Mine Closure Plan that will be developed prior to mine closure when more certainty about the final void conditions will be known.	Noted				
Rehabilitation Management Plan	6.3.1	General principles that may be adopted to make the final void safe and stable include: Battering back the lowwalls and highwall to minimise potential for failures and mass movement; Capping (or excavating) exposed coaliferous material with inert material to prevent ignition from spontaneous combustion, bushfires or human interference; Constructing a physical barrier to isolate the perimeter of the void to prevent human access. The highwall areas will be secured by the construction of a trench and a safety berm, as well as a security fence along the entire length of the eastern and northern highwalls; Suitable signs, clearly stating the risk to public safety and prohibiting public access will be erected at intervals along the entire length of the fence; and Surface runoff from land surrounding the void will be diverted so as to prevent any potential development of instability of the void walls.	Noted				
	Water Management I	Rehabilitation Areas					
Rehabilitation Management Plan	6.3.1	Sediment dams will be designed and constructed to meet the relevant criteria as prescribed by NSW industry guidelines Managing Urban Stormwater Volume 1: Soils and Construction (Landcom 2004) and Managing Urban Stormwater Volume 2E: Mines and Quarries (DECC 2008) (the 'Blue Book'). Sediment dams are maintained throughout rehabilitation to capture runoff from rehabilitated areas until discharge water quality meets regulatory criteria and dams can be decommissioned.	Reviewed latest sed dam design report , design states complaince with the required guidelines.	Compliant			





Poforonco	Clause	Requirement	Evidence	Audit Einding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	6.4 Growth Media Dev						
	6.4.1 Growth Media D						
	Agricultural Rehabilita	Subsoils and topsoils with the appropriate physical and chemical	See elsewhere in the audit, the agricultural				
Rehabilitation Management Plan	6.4.1	properties will be selectively stripped and handled for re-use in Agricultural Rehabilitation Areas. The Agricultural Resources and Productivity Assessment (McKenzie, 2011) undertaken for the Project EA identified Stratic Rudosols located within the alluvial creek flats at the south-eastern project disturbance boundary as a suitable resource for agricultural rehabilitation activities. This soil type was identified as suitable due to: ② Favourable pH values; ② Low Exchangeable Sodium Percentage (ESP) requiring minimal amelioration with gypsum; ② Predominantly non-saline soils; and ③ Cation Exchange Capacity (CEC) of the clay-rich subsoils is generally high enough to produce selfmulching soils through shrink-swell processes (McKenzie 2011).	areas are not scheduled to be mined for a number of years from the time of the audit.	Not Applicable			
Rehabilitation Management Plan		These agricultural soil resources will not be disturbed until mining advances to the east, in approximately Year 12 of the mine life. Conceptual methodologies to manage agricultural soil resources are presented in this Rehabilitation Management Plan. Detailed methodologies to handle, stockpile and reinstate agricultural soil resources require further consultation with stakeholders and will be developed prior to Year 12 and documented in subsequent revisions of this plan.	, in the second	Not Applicable			
	6.4.2 Topsoil Manager						
	Selective Soil Stripping	and Handling Topsoil and subsoil resources suitable for rehabilitation are	MOP and soil stripping procedure -		ı		
Rehabilitation Management Plan		stripped in accordance with the soil stripping procedures identified in this plan and the MOP. Soil resources appropriate for agricultural and native vegetation rehabilitation outcomes, and suggested stripping depths, have been assessed documented in the Agricultural Resources and Productivity Assessment (McKenzie, 2011) and are documented in the Biodiversity Management Plan.	otherwise noted.				
Rehabilitation Management Plan	6.4.2	Prior to soil stripping activities, additional investigations will be conducted to confirm the appropriate soil stripping depths for each soil type recommended for salvage (refer to Section 4.3.2 and Figure 7 in the Biodiversity Management Plan).	This is done during stripping, not prior to.				





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.4.2	The soil stripping and handling techniques described below have been developed to maximise the soil resource salvaged and minimise deterioration of the soil resource. ② Undertake soil testing to characterise subsoils and topsoils prior to stripping to determine the type and application rates for any required soil ameliorants (e.g. lime, gypsum, fertiliser and organics); ② Mulch all cleared vegetation not retained for habitat augmentation on site and mix into topsoils as a soil conditioner; ② Confirm the depth of soils to be stripped in accordance with the Biodiversity Management Plan and the MOP. Topsoils and subsoils are generally to be stripped and handled separately; ② Topsoil would be maintained in a slightly moist condition during stripping where possible. Material will not be stripped in either an excessively dry or wet condition; and ② Soil would be graded or pushed into windrows with excavators, graders or dozers for loading into rear dump trucks by front-end loaders, or will be collected by scraper fleets and replaced directly on rehabilitation areas where available or placed in temporary stockpile locations.		Not Compliant	D	1	Medium
	Soil Stockpiling						
Rehabilitation Management Plan	6.4.2	Where direct subsoil and topsoil re-spreading is not practicable, the stripped soils are stockpiled in accordance with the procedures documented in the Biodiversity Management Plan. Soils anticipated to be stockpiled for extended periods may be seeded with native vegetation. Prior to re-spreading, weed growth is scalped from stockpiles to minimise the introduction of weeds into rehabilitated areas.	See BOMP - site inspection indicated that the topsoil storage was adequate and compliant with these commitments.	Compliant			
Rehabilitation Management Plan	6.4.2	Preliminary topsoil handling and stockpiling processes include: Windrow topsoil stockpiles on relatively flat areas along the contour. Consider ease of access and minimising rehandling (i.e. avoid locations that will be required for future mining activities);	Complies based on site observations	Compliant			
Rehabilitation Management Plan	6.4.2	☑ Topsoil stockpile heights are designed to prevent biological and structural degradation. Generally stockpiles will be no higher than 3 m. Clayey soils (e.g.: Stratic Rudosols) will be stored in lower stockpiles for shorter periods of time (where feasible) compared to soils that have a coarser texture;	Complies based on site observations	Compliant			
Rehabilitation Management Plan	6.4.2	Shape stockpiles to be free-draining stockpiles to minimise the formation of anaerobic zones, with batters graded to achieve slopes approaching 18°. Retain a surface condition to enhance infiltration and minimise erosion until vegetation is established;	Complies based on site observations	Compliant			





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.4.2	☑ Where long-term stockpiling is planned (i.e. greater than six months), fertilise and seed stockpiles with an annual cover crop species that produces sterile florets or seeds. The annual pasture species would not persist in the rehabilitation areas but would provide sufficient competition for emerging weed species and enhance desirable microbial activity in the soil. Additional ameliorants may be periodically applied (e.g. biosolids) to assist retain desirable microorganism activity in the soil; and	Complies based on site observations	Compliant			
Rehabilitation Management Plan	6.4.2	© Clearly demarcate stockpiles with signage. Install appropriate erosion and sediment controls (e.g.: bunds along the upslope perimeter and sediment fences along the toes of the downslope batter;	Not observed in site visit	Not Compliant	E	1	Medium
Rehabilitation Management Plan	6.4.2	© Conduct regular inspections to identify erosion and weed infestation and maintain stockpiles where issues are identified. Prior to respreading stockpiled topsoil assess any weed infestation and if required treat weeds with herbicide application and/or "scalp" weed species prior to topsoil spreading.	EOs monthly inspections review weed status and drive control actions	Compliant			
	Soil respreading and	Seedbed Preperation	·				
Rehabilitation Management Plan	6.4.2	Where possible, topsoil would be re-spread directly onto reshaped landforms. Appropriate soil resources have been identified and quantified in the Agricultural Resources and Productivity Assessment (McKenzie, 2011). Topsoil and subsoil will be spread in accordance with the re-spreading depths indicated in the Agricultural Resources and Productivity Assessment (Figure 12) (McKenzie 2011) for the desired final land use outcome.	Noted, confirmed at interview. Recommendation - there are large amounts of soil stockpiled from the construction period that should be worked into the rehab process rather than leaving in storage.	Compliant			
	Agricultural Areas						
Rehabilitation Management Plan	6.4.2	All topsoils intended for re-use in agricultural rehabilitation areas will be assessed prior to stripping to determine their suitability for re-use and determine specific amelioration requirements. Once soils are respread, ameliorants such as gypsum are applied and the area is deep-ripped along the contour.	Not yet occuring	Not Applicable			
Rehabilitation Management Plan	6.4.2	To achieve the rehabilitation objective of an effective rooting depth (ERD) greater than 1.5 m, the soil profile will be reconstructed in a staged approach to achieve the necessary structure, chemical properties and microbes in the entire soil profile. A staged re-spreading methodology will be developed prior to the anticipated timing of the reinstatement of agricultural soils to optimise the structure, chemical and biological properties throughout the soil profile. The re-spreading methodology will be developed in consultation with stakeholders and agencies and documented in future revisions of the Rehabilitation Management Plan.	Noted, not yet required.	Not Applicable			
	6.4.3 Habitat Augmer	ntation	•				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Filluling	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.4.3	TCPL will develop procedures to salvage and re-use materials from site for habitat establishment in the Biodiversity Management Plan.	The BOMP details some of these measures but there is inadequate detail to ensure that on ground activities are properly targeted.	Compliant			
	6.5.2 Native Vegetati	on Species Selection					
Rehabilitation Management Plan	6.5.2	Native vegetation tubestock and seeds will be planted / seeded over the rehabilitation area once the initial cover crop dies down. Native vegetation areas are planted with native tubestock species, selected on a site by site basis considering factors such as adjacent remnant vegetation communities, aspect, slope, and soil types.	This occurs onsite, noted in site inspection and also in MOP	Compliant			
Rehabilitation Management Plan	6.5.2	In accordance with Project Approval PA 11_0047 Schedule 3 Condition 43 and EPBC Referral 2011/5923 Condition 23, Woodland Rehabilitation Areas will include at least 13 ha planted with species representative of the Box-Gum Woodland EEC/CEEC. The Final Void domain will be planted with native vegetation tubestock above the predicted permanent water line.	The woodland areas established to date include the species required to re-establish the EEC. There is a lack of understory species in the established rehab as discussed in the EcoLogical Monitoring Report. Recommendation - Using tubestock or seed additional understory species to be added into existing rehab and future rehab to have some trials established to determine the best way to ensure these species survive along with the canopy and grass species.	Not Applicable			
Rehabilitation Management Plan	6.5.2	TCPL will rehabilitate the permanent Goonbri Creek alignment and Goonbri Creek Enhancement Area with riparian species to enhance riparian habitat. Riparian species will be characteristic of the Bracteate Honeymyrtle (Melaleuca bracteata) Low Riparian Forest community. Final riparian species selection will be documented in the Permanent Goonbri Creek Alignment Management Plan.	This is some years away at the time of the audit.	Not Applicable			
	6.5.3 Seed Collection	and Propagation					
Rehabilitation Management Plan	6.5.3	Tarrawonga will engage appropriately qualified contractors to undertake seed collection from remnant native vegetation in onsite and offset areas. Seed collection and propagation will be conducted in accordance with the procedures described in the Biodiversity Management Plan.	This occurs, Field Nursery conduct seed collection.	Compliant			
	6.5.4 Tubestock Plant	ings and Direct Drill Seeding					
	Tubestock Planting						





2.6	CI.	Requirement	Evidence	A 12 51 12		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.5.4	Tubestock plantings will be mulched, watered and protected from grazing by tubestock bags and/or fencing where required.	This does not occur.	Not Compliant	D	3	Medium
	Direct Drill Seeding						
Rehabilitation Management Plan	6.5.4	TCPL will trial direct seeding of eucalypts and some understorey species for woodland areas. If these trials are successful consideration will be given to utilising direct seeding to rehabilitate open woodland rehabilitation areas. Direct seeding trials will assess factors including: ② Sowing rates and species proportions; ③ Seed pre-treatment; ③ Equipment selection; ② Soil amelioration and fertilizer types and application rates; and ② Timing.	This has not yet occurred but may be useful in light of the lack of understory plants in the existing rehab noted by EcoLogical.	Not Applicable			
	6.5.5 Weed and Verte	l ebrate Pest Control					
Rehabilitation Management Plan	6.5.5	Ecosystem Establishment includes initial management actions to limit the introduction of weeds and vertebrate pest species in rehabilitation areas as described in Section 3.2.7. Management measures are documented in the Biodiversity Management Plan and include: Minimise potential weed seed transport to or from the site by inspecting all vehicles and plant used in vegetation clearing and rehabilitation activities. Plant and vehicles are to be clean and weed-free before permitted on to site.	BOMP details this, no evidence of actual vehicle checks though the offset area has only recently been established. Recommendation - establish a weed vector management procedure for the site and offset areas or tie the requirements into existing procedures.	Compliant			
Rehabilitation Management Plan	6.5.5	Stabilise topsoil stockpiles (and disturbance areas where possible) with mulches and cover crops to suppress weed germination. Weeds will be scalped from stockpiles prior to re-use to minimise introduction of weeds in rehabilitation areas.	BOMP - based on site observations this is conducted.	Compliant			
Rehabilitation Management Plan	6.5.5	Prior to over-sowing cover crops with native vegetation spray isolated broadleaf weed occurrences with targeted herbicide application. If weed infestations occur on rehabilitation areas delay native vegetation sowing and spray weeds and sow repeat cover crops until the weed seedbank is exhausted and weed control will be manageable prior to establishing native vegetation.	AEMR and BOMP, this process needs better implementation or rewriting in the BOMP. The weed seed bank in the topsoil is quite heavy in some areas and competition with the planted/seed natives is strong.	Compliant			
Rehabilitation Management Plan	6.5.5	Monitor weeds at least every six months on rehabilitation areas and undertake appropriate control actions (mechanical removal or targeted herbicide spraying) using an appropriately qualified contractor.	This is done more regularly at monthly intervals.	Compliant			
Rehabilitation Management Plan	6.5.5	② Install fencing and/or tree guards to protect juvenile vegetation and tubestock from herbivores.	Not done	Not Compliant	D	3	Medium





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.5.5	Monitor vertebrate pest species on all Tarrawonga land at least every six months. Where required, TCPL will engage suitably qualified contractors to undertake appropriate control programs (such as baiting and trapping) in consultation with the North West Livestock Health and Pest Authority.	AEMR and BOMP, EO and Property Manager record occurrences of ferals and Property Manager or contractors implment controls.	Compliant			
	6.5.6 Erosion and Sed	iment Control					
Rehabilitation Management Plan	6.5.6	Temporary erosion and sediment controls (ESCs) will be installed in accordance with the Blue Book to minimise erosion prior to rehabilitation areas being considered stable (i.e. greater than 70% cover). Temporary ESCs will be installed in accordance with an areaspecific Erosion and Sediment Control Plan (ESCP) and will be retained and maintained until no longer required.	In WMP and reported in the AEMR, the iste inspection did not identify any areas where erosion and sed control was required and had not been implemented.	Compliant			
	6.5.7 Bushfire Mitigat						
Rehabilitation Management Plan	6.5.7	As outlined in Section 4.2.2 uncontrolled bushfire events have the potential to significantly damage biodiversity in rehabilitation areas. Bushfire risks in rehabilitation areas are managed in accordance with the Bushfire Management Plan and the Biodiversity Management Plan.	See Bushfire Managemrnt Plan	Compliant			
Rehabilitation Management Plan	6.5.7	Management and mitigation measures include but are not limited to: ② Selection of fire-tolerant species in native vegetation areas; ② Maintaining fire breaks by maintaining cleared easements and fence-lines; ② All mine personnel will receive basic fire-fighting training and appropriate fire fighting equipment will be maintained and available onsite; ② Maintain fire-fighting access tracks and water sources (sediment dams) in rehabilitation areas; ② All vehicles will be diesel fuelled only and are to be driven only on demarcated roads and tracks; ③ Controlled high intensity, short-term grazing will be employed to assist in the reduction of vegetative fuel loads on areas on which active mining operations are not occurring and appropriate fencing is available; ② Monitoring fuel loads and undertaking mosaic back-burning programs (where appropriate and in consultation with Narrabri Shire Council) to reduce fuel loads. Back burning frequency will be varied to minimise disadvantaging regeneration of particular ground cover and shrub species; and ② Maintaining sufficient viable seed for key native vegetation species to re-seed areas after major bushfire events.	grazing not conducted but otherwise compliant, fuel loads acceptable.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Filiding	Consequence	Likelihood	Risk
Rehabilitation Management Plan	6.6	In the context of this plan, the Ecosystem Development phase constitutes the medium to long term management of rehabilitation areas toward achieving the desired final land-use. This phase includes: ① Ongoing environmental management to minimise risks to rehabilitation; ② Comparing specific ecosystem characteristics such as soil profile development, floristic composition and structure and faunal diversity and abundance with the characteristics of appropriate analogue sites; and ② Undertaking adaptive management and remedial works where characteristics of the rehabilitation are not trending toward desired outcomes.	The rehab has been monoitored by EcoLogical in the audit period and measures recommended are undergoing implementation. With the poor soils and weed seed bank in topsoil this will require more attention as the rehab progresses.	Compliant			
	6.6.2 Intervention and	Adaptive Management					
Rehabilitation Management Plan	6.6.2	Where monitoring results reveal that indicators are not trending towards the completion criteria as predicted, plans should be executed to promptly modify management practices to achieve the desired rehabilitation results.	Rehab not yet at the point where completion should be considered	Not Applicable			
Rehabilitation Management Plan	6.6.2	Early intervention will be actioned in the form of a Trigger Action Response Plan (TARP), to be developed for each rehabilitation domain, to minimise the risk of failure to achieve the desired rehabilitation outcomes for each domain.	There is evidence of intervention in failed rehab on the site, the TARP is located in the Rehabilitation Management Plan	Compliant			
	6.8 Completion Criter	ia			•		
Rehabilitation Management Plan	6.8	Indicators and completion criteria specific to each phase of rehabilitation have been developed for each Rehabilitation Domain.	Noted				
Rehabilitation Management Plan	6.8	Domain rehabilitation indicators and completion criteria at Tarrawonga are based on: Review of regulatory and approval requirements (Section 2); Industry knowledge including industry guidelines and scientific literature; and Characteristics of comparable environments, determined by selecting and analysing appropriate analogue sites (Section 8.2.1).	Noted				
	6.8.1 Inspection and T	est plans and As Built Surveys					
Rehabilitation Management Plan	6.8	TCPL employs quality assurance tools such as hold points, Inspection and Test Plans (ITPs) and 'as built' surveys, to verify that key rehabilitation tasks have been completed in accordance with specifications. Tables 6.4 to 6.8 document the records, inspections and surveys required to verify satisfactory completion of each completion criteria. Rehabilitation records are discussed further in Section 8.1.	Noted				
	7 Rehabilitation Imple	mentation					





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk	
Rehabilitation Management Plan	7	In accordance with EPBC referral 2011/5923 Condition 3, TCPL must submit to SEWPaC an approach to limit the maximum disturbance of Box-Gum Woodland EEC/CEEC (for each of the years 5, 10, 15 and 17) within three months of commencement of construction of the Tarrawonga Coal Project (eastern extension).	The Tarrawonga Disturbance Limits Analysis report by EcoLogical addresses this requirment.	Compliant				
Rehabilitation Management Plan	7	TCPL will develop mine planning for life of the mine that will provide staged plans for years 5, 10, 15 and 17 showing anticipated disturbance and rehabilitation progress for the site. This Rehabilitation Management Plan will be revised prior to submission to SEWPaC for Commonwealth approval to include the staged rehabilitation plans for life of mine.	The Tarrawonga Disturbance Limits Analysis report by EcoLogical addresses this requirment.	Compliant				
Rehabilitation Management Plan	7	Progressive mine plan and rehabilitation plans are prepared for each year of the currently approved MOP period. Figure 7.1 below shows the anticipated mine plan and rehabilitation progress for the end of the current MOP period (2015).	See MOP	Noted				
	8 Monitoring, Trials ar	nd						
Rehabilitation Management Plan	7	In accordance with EPBC referral 2011/5923 Condition 3, TCPL must submit to SEWPaC an approach to limit the maximum disturbance of Box-Gum Woodland EEC/CEEC (for each of the years 5, 10, 15 and 17) within three months of commencement of construction of the Tarrawonga Coal Project (eastern extension).	The Tarrawonga Disturbance Limits Analysis report by EcoLogical addresses this requirment.	Compliant				





Audit Finding Consequence Ithe Bino Ithe Bino Consequence Ithe Bino Ither Bino Ithe Bino Ithe Bino Ither Bino Ithe Bino Ither	Doforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
3.1 Cutural Netrotage Assessments In the event that other items of potential heritage significance are This has not occurred in the audit period uncovered during mining operations, the following protocol will be a considered to the protocol of the proto	Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
in the event that other freeze exercisements In the event that other freeze or potential heritage significance are uncovered during mining operations, the following protocol will be followed: I. Work will case and office or provided in the second of the followed: I. Work will case and office or provided in the second office of the followed: I. Work will case and office or provided in the second of the followed: I. Work will case and office or provided in the second office of the followed: I. The generous will notify the Department will be called and seriel() is part outcomed off with the provided office or series or case of the followed: I. The generous will notify the Department will be called the second of the follower of the follower or series of the follower or series or case of the follower or series or case of the follower or series or case or series or case of the follower or series or case								
In the event that other interns of potential heritage significance are in the audit period uncovered during mining operations, the following protosis her following the protosis heritage will could be followed: 1. Work will clease in the area of discovery immediately; 2. The person discovering the items will result private assent and and earle() is pare (protoment of with tage; 3. The supervisor will notify the Operations Manager visits and protons on site; 4. The Operations Manager will: 5. In the find a determined to be of heritage significance at their earliest. 5. In the find a determined to be of heritage significance, and the second and earle() is pare (protoment of with tage; 3. The supervisor will notify the Operations Manager will: 5. In the find a determined to be of heritage significance, only of the operations will be protoned as their earliest. 5. In the find a determined to be of heritage significance, only of the operations and second will be protoned as their earliest. 5. In the find a determined to be of heritage significance, only of the protoned are easier to the find of the second of t								
uncovered during mining operations, the following protocol will be followed: 1. Work will cease in the area of discovery immediately; 2. The person discovering the most mining the supervisor immediately who will cease in the area of discovery immediately; 3. The supervisor discovering the formula forth the Supervisor immediately who will cease the formula forth the Supervisor immediately who will cease the formula forth the Supervisor immediately who will cease the formula for the Supervisor will north the Coperation Manager or senior (Tch person on step; 4. The Departure Namager ("Evenomental Manager will: a. arrange a qualified heritage management consultant to stread the site and above in its fertilage splits cance at the reliants opportunity, and but the site of the fertilage splits cance at the reliants opportunity, and but the site of the fertilage splits cance at the reliants opportunity, and but the site of the fertilage incursor and and seek referent members of software states of the fertilage incursor and interest of experiments. The importance of Heritage items and the associated obligations in terms of recordation and notification requirements. The importance of Heritage items and the associated obligations in terms of recordation and personal records of their requirements. The importance of Heritage items and the associated obligations in terms of recordation and seek reflevior advice as to appropriate protocols in terms of recordation and ordinated as the form the induction process are other controls in destring and ground for site so all personal requirements. The importance of Heritage items and the associated obligations in terms of accomplication of the requirements of their requirements. The Altogramment Plan 4. Mortagement Plan 4. Mortagement Plan 4. Condition of the control ground of their requirements of their requirements of the requirements		3.1 Cultural Heritage		This has not occurred in the audit period		<u> </u>		
Heritage Management Plan 4 Management Program TCPL recognises the importance of Aboriginal places and objects to the Aboriginal community and its obligations under the NPW Act the FP8A Act, and the provisions of the Project and the Project Act, and the provisions of the Project approval, and will undertake all activities in a manner that conserves the cultural heritage values of the area. Heritage Management Plan 4 Central to the satisfaction of this commitment, there will be the continuation of the existing open and honest relationship between TCPL management and the RAPs. The management and the RAPs. The management Resource Strategies, 2011) and in consultation with registered stakeholders. The management (Resource Strategies, 2011) and in consultation with registered stakeholders. Al Collection of Aboriginal Cultural Heritage This management plan focuses on the management of these sites only, with additional details to be developed for the management of the sea along the project in project approval. Table 3 lists the cultural heritage sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites sites with the project.	Heritage Management Plan	3.1	followed: 1. Work will cease in the area of discovery immediately; 2. The person discovering the item will notify their supervisor immediately who will ensure that work has ceased and area(s) is (are) cordoned off with tape; 3. The supervisor will notify the Operations Manager or senior TCPL person on site; 4. The Operations Manager/Environmental Manager will: a. arrange a qualified heritage management consultant to attend the site and advise on its heritage significance at their earliest opportunity; and b. if the find is determined to be of heritage significance, notify OEH Heritage Branch with the advice from the heritage consultant and seek relevant advice as to appropriate protocols in terms of		Not Applicable			
TCPL recognises the importance of Aboriginal places and objects to the Aboriginal community and its obligations under the NPW Act the EP&A Act, and the provisions of the Project Approval, and will undertake all activities in a manner that conserves the cultural heritage values of the area. Central to the satisfaction of this commitment, there will be the continuation of the existing open and honest relationship between TCPL management and the RAP's. The management Plan 4 Continuation of the existing open and honest relationship between TCPL management program has been prepared principally based upon the recommendations of the Tarrawonga Coal Project Environmental Assessment (Resource Strategies, 2011) and in consultation with registered stakeholders. 4.1 Collection of Aboriginal Cultural Heritage This management of the management of updated Heritage Management Plans every two years throughout the life of the project. The purpose of the collection program is to retrieve culturally significant items from these Aboriginal cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage stars requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the next two years whilst Table 4 lists the sites	Heritage Management Plan	3.1	under relevant legislation will be included in the induction process	are other controls in clearing and ground	Compliant			
Heritage Management Plan 4		4 Management Prog	ram					
Heritage Management Plan 4 continuation of the existing open and honest relationship between TCPL management and the RAP's. The management program has been prepared principally based upon the recommendations of the Tarrawonga Coal Project Environmental Assessment (Resource Strategies, 2011) and in consultation with registered stakeholders. **A1. Collection of Aboriginal Cultural Heritage** This management plan focuses on the management of these sites only, with additional details to be developed for the management of the project in consultation with the RAP's. This will be achieved through the development of updated Heritage Management Plans every two vers throughout the life of the project. The project in consultation with the RAP's and the project approval. Table 3 lists the cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage sites requiring collection over the life of the project.	Heritage Management Plan	4	to the Aboriginal community and its obligations under the NPW Act the EP&A Act, and the provisions of the Project Approval, and will undertake all activities in a manner that conserves the cultural					
Heritage Management Plan 4 upon the recommendations of the Tarrawonga Coal Project Environmental Assessment (Resource Strategies, 2011) and in consultation with registered stakeholders. 4.1 Collection of Aboriginal Cultural Heritage This management plan focuses on the management of these sites only, with additional details to be developed for the management of the project in consultation with the RAP's. This will be achieved through the development of updated Heritage Management Plans every two years throughout the life of the project. The purpose of the collection program is to retrieve culturally significant items from these Aboriginal cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the life of the project.	Heritage Management Plan	4	continuation of the existing open and honest relationship between	Noted				
This management plan focuses on the management of these sites only, with additional details to be developed for the management of the remaining identified sites over the life of the project in consultation with the RAP's. This will be achieved through the development of updated Heritage Management Plans every two years throughout the life of the project. The purpose of the collection program is to retrieve culturally significant items from these Aboriginal cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the life of the project.	Heritage Management Plan	4	upon the recommendations of the Tarrawonga Coal Project Environmental Assessment (Resource Strategies, 2011) and in	Noted				
only, with additional details to be developed for the management of the remaining identified sites over the life of the project in consultation with the RAP's. This will be achieved through the development of updated Heritage Management Plans every two years throughout the life of the project. The purpose of the collection program is to retrieve culturally significant items from these Aboriginal cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage sites requiring collection over the next two years whilst Table 4 lists the sites requiring collection over the life of the project.		4.1 Collection of Abo						
	Heritage Management Plan	4.1	only, with additional details to be developed for the management of the remaining identified sites over the life of the project in consultation with the RAP's. This will be achieved through the development of updated Heritage Management Plans every two years throughout the life of the project. The purpose of the collection program is to retrieve culturally significant items from these Aboriginal cultural heritage places to avoid their destruction prior to disturbance proceeding in accordance with the project approval. Table 3 lists the cultural heritage sites requiring collection over the next two years whilst Table 4 lists the sites	have been salvaged during the audit period				
Heritage Management Plan 4.1 Collection must be supervised by a suitably qualified archaeologist in partnership with the RAP's. Compliant Compliant		4.1		This occurred noted in RPS salvage report	Compliant			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.1.1	This research design provides for the recording, collection, curation and storage of identified Aboriginal cultural heritage items from areas that will be disturbed as a result of the construction and operational activities associated with expansion of the mine. Prior to active salvage occurring, the cultural values associated with the artefacts for salvage, and the specific area in which the salvage is undertaken, will also be considered and assessed, with specific input sought from the RAP's in relation to cultural values, how sites may have been used in the past, and their overall significance in the wider landscape. This information will be recorded and incorporated into the salvage report to be produced at the conclusion of each salvage event.	This occurred noted in RPS salvage report	Compliant			
Heritage Management Plan	4.1.1	All previously identified archaeological sites and isolated finds outside the mine disturbance areas will be avoided with protection measures, such as fencing, in place to avoid disturbance. This fencing will take place progressively, with those areas in closest proximity to current operations fenced first. All sites will have protective fencing in place within 12 months of approval of this management plan. Identified archaeological sites and isolated finds within the expanded open-cut pit, waste emplacements, infrastructure areas and associated disturbance areas will be collected (Table 3). Collection of all artefactual material will be undertaken with the relevant details recorded on the Aboriginal Site Impact Recording Form, as available from the Office of Environment and Heritage Website. Those site locations noted as being impacted by infrastructure will be subject to additional review, prior to disturbance, to ascertain if these sites can be practically avoided, and hence left in-situ, during the life of the project.	Sighted, photos on file	Compliant			
Heritage Management Plan	4.1.1	Prior to commencing salvage works, the archaeologist will review and consider the outcomes/findings from prior salvage works in the vicinity of the area, and use this information to develop the salvage program, with specific consideration to identifying key research questions/opportunities during the salvage. A key aspect of the research questions relevant to salvage will be consistency with other cultural heritage programs in the area, and in particular, how the knowledge gained from the salvage works can be utilized in the development of the broader Aboriginal Heritage Conservation Strategy required under Condition 51 of Schedule 3 of the Project Approval PA 11_0047, and in the context of any other cultural heritage programs being developed by the OEH.	This occurred noted in RPS salvage report	Compliant			
Heritage Management Plan	4.1.1	A broad research question for the project site relates to the identification of any patterns of stone artefact discard behaviour or location of culturally modified trees, and how it relates to the topographic features in the landscape. Refinement and consideration of specific questions for the salvage program will be determined in consultation with the archaeologist prior to commencement of on ground works.	This occurred noted in RPS salvage report	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.1.1	At the conclusion of the salvage works, the archaeologist will provide written clearance to the Tarrawonga mine that all known artefacts have been salvaged and that mining activities may commence in that area. Ongoing management of the area will include the opportunity for representatives of the RAP's to participate in a soil stripping monitoring program on a roster basis. This will provide capacity to monitor for additional uncovering of artefacts during the soil stripping program.	This occurred noted in RPS salvage report. The RAPs are engaged on a rotating basis managed by the site. Evidnence of the engagement basis and roatation provided by the site	Compliant			
Heritage Management Plan	4.1.1	Prior to any vegetation clearance, archaeological work at each disturbance site will include the following: 1. Stone Artefacts: The following methodology is required for collection of stone artefacts: • Commissioning of a suitably qualified archaeologist to inspect and record each known site to identify surface Aboriginal objects within the proposed disturbance area. The archaeologist will engage with the RAP's during this process who will be invited to observe and participate in the works (e.g. recording, excavation and collection). • In relation to each identified Aboriginal object, the archaeologist will, prior to any disturbance, record its position with a GPS and/or map, and measure and describe the site or find. All identified Aboriginal objects will be recovered and separately bagged, labelled, and stored temporarily in the Cumbo Gunerah Keeping Place at the Red Chief LALC office in Gunnedah, or other temporary keeping place as agreed with the RAP's. If appropriate, a temporary keeping place could be established at site to retain artefacts up to the period prior to in- situ replacement. A Care and Control Permit will be obtained for the storage of the artefacts in consultation with OEH and the RAP's. • The collected Aboriginal objects will be recorded in sufficient detail to allow description of the lithic technology and reduction strategies adopted. The regional and local contexts of the collected objects will also be analysed and described.		Compliant			
Heritage Management Plan	4.1.1	Following the completion of rehabilitation activities or during the rehabilitation program at a time when works within the specific area do not pose a threat, the salvaged Aboriginal objects will be replaced in a location as close as possible to their original location in consultation with OEH and the RAP's. Aboriginal object replacement will be conducted by a suitably qualified archaeologist, with the RAP's invited to observe and participate in replacement works.	Noted in salvage report All salvaged artifacts retained in shipping container on-site. Final location to be determined with RAPS when site is at closure.	Not Applicable			





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.1.1	2. Culturally Modified Trees: The following methodology is proposed for scarred tree removal and relocation. Prior to any actions to remove the tree, relevant advice will be obtained from a suitably qualified arborist to ensure the integrity of the tree prior to movement. Advice will be sought from an archaeologist to verify archival recording requirements of the trees prior to removal: Protection of the tree prior to removal, by padding the exterior with layers of foam rubber or similar padding, to protect the surface and structure of the tree during movement. It may also be necessary to fill the hollows of the tree with two part liquid polyurethane foam, which expands and fills the cavity, thereby further supporting the surface. The requirement for this treatment will be confirmed with the arborist.		Compliant			
Heritage Management Plan	4.1.1	Depending on the structural integrity of the tree, the tree may need to be cut back to a manageable size before attempting to remove the tree. Preparation of the tree for removal, by attaching lengths of rigid timber to the outside of the padding, to effectively create 'splints' that will spread pressure evenly along the tree surface during movement. Lifting of the tree using a crane, with the tree supported by straps placed around the padded trunk or a heavy cargo net. During the initial stages of lifting, the crane would also support the tree when the final cut is made near the ground surface. After the base cut is complete, the tree would be suspended by the straps and manoeuvred into a horizontal position so that it is possible to examine the underside of the tree. If the core is too fragile and is likely to drop out, a cap would be secured in place to prevent loss. During this stage, any termite debris and loose materials would also be removed from the hollow base, which would then be stabilised by insertion of expanding polyurethane foam to fill the cavity. Movement of the tree onto the back of a flat bed truck, where it would be placed on a bed of foam and secured with straps for transport to the Cumbo Gunerah Keeping Place at the Red Chief Local Aboriginal Land Council office in Gunnedah or another suitable location agreed with the RAP's. Unloading the tree using a crane, with the tree supported by straps. The tree will be supported on a well-drained bed of rubber or similar protective surface. Periodic treatment for termites or	methodology for protection of the scar tree. Recommendation - at the next review of the HMP, the scar tree section be make more generic to allow for site conditions when recovering and preserving scar trees.				
Heritage Management Plan	4.1.1	It may be necessary to undertake insect eradication of the section of tree removed to minimise the potential for termite or other insect damage to the tree. This will be assessed with the arborist at the time of tree removal, and where required, appropriate fumigation by a certified pest controller will be undertaken prior to its permanent relocation.	This was not required for the report reviewed	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Keierence	Clause			Addit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.1.1	Upon agreement with the RAP's as to suitable placement locations, specific measures will be taken depending on whether the tree is kept in an indoor or outdoor location. If the tree is to be retained in an outdoor location, it should be positioned such that it has shelter from rain and direct sunlight. Placement of a metal cap of the top of the trunk will also reduce potential for water ingress. If kept upright, an excavation will be made in the ground surface at least 3m in diameter and 1m deep. This will be filled with concrete and then covered in gravel to minimise moisture being drawn into the wood and discouraging termites from accessing the tree. A suitable mechanism for supporting the trunk will be established in consultation with the RAP's, which may include a steel shaft through the centre of the trunk, set in the concrete described above, or such other alternate measure that will not result in damage to the cultural scar. These methods are in accordance with the "Guidelines for the Preservation and Management of Carved and Scarred Trees" as produced by the Australian Museum 2010.	Noted, not yet completed and to be determined with RAPs as discussed.	Not Applicable			
	4.1.2 Reporting						
Heritage Management Plan	4.1.2	TCPL will commission an archaeologist to produce reports detailing the results of the investigation programs. These reports will include: • the aims and methodology of the investigation program; • the location and characteristics of sites; • the cultural values associated with the artefacts and the specific area from which the artefacts were salvaged; • the number of Aboriginal objects collected from each site; • the nature of the cataloguing system used; • the lithic technology and reduction strategies adopted; • an analysis of the regional and local contexts of the artefacts; and, • a summary of consultation undertaken with the RAP's.	This occurred noted in RPS salvage report	Compliant			
Heritage Management Plan	4.1.2	These reports will form part of the Annual Environmental Management Report/Annual Review (AEMR/Annual Review). They will be provided to the NSW DPI and OEH within 12 months of the completion of each investigation program. The reports will also be provided to the RAP's at the conclusion of each salvage event.	AEMR 2013-14 summarises the outcomes of these reports.	Compliant			
Heritage Management Plan	4.1.2	Following completion of the salvage event, the archaeologist will provide relevant notification to the OEH to enable updating of the AHIMs records.	This occurred noted in RPS salvage report	Compliant			
	4.2 Site Protection		I	1			
Heritage Management Plan	4.2	Twenty-five cultural heritage features comprising open scatters, isolated finds and scarred trees occur outside areas planned for future expansion and development of the mine.	Noted				





Deference	Clause	Requirement	Evidence	Audia Findina		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.2	In order to protect and conserve these features, TCPL will undertake the following: • Culturally modified trees will be fenced at a distance of 5m from the tree where practicable. The fence line location will be subject to physical constraints such as other vegetation and topography Fence posts will be marked white and set vertically apart at approximate 5m intervals; • Isolated Artefacts will be fenced with a 10m radius from the centre point from the identified GPS coordinate; • Artefact scatters will be assessed in terms of their extent based on the site record from the archaeological survey. Where the extent cannot be confirmed as within 20m of a central coordinate, Tarrawonga will consult with the RAP's as to determining the relevant extent for the purpose of protective fencing.	Sighted, photos on file	Compliant			
Heritage Management Plan	4.2	The fencing works will be completed within 12 months of the date of management plan approval. Sites that have already been fenced for protection are highlighted yellow.	Not twelve months since the approval of the plan (January 2014), most sites are fenced (if not all) already	Not Applicable			
Heritage Management Plan	4.2	All fenced sites will be subject to 12 monthly inspections to verify the sites remain secure. RAP's will be invited to participate in annual reviews of the protective measures. Inspections of these sites will also consider any potential impacts associated with drainage lines and erosive processes and any options to mitigate impacts.	Not yet required	Not Applicable			
Heritage Management Plan	4.2	Blasting will be a regular feature of operations at site. The nature of the known artefacts (isolated finds, artefact scatters and culturally modified trees) suggests that blasting is likely to have minimal impact on these features, with no predicted impacts from blast vibration and flyrock unlikely to impact on those artefacts to be retained in-situ. Known artefact sites will be subject to regular inspections, as part of the site monthly inspection regime, to verify no detrimental impact from blasting activities, and in accordance with the management measures specified in the Blast Management Plan.	No known sites within damage range of site blasting	Compliant			
Heritage Management Plan	4.2	Cultural heritage awareness will be included in the induction, as discussed in Section 4.7.	Included in the induction	Compliant			
	4.4 Contingency Plan	for the Management of Previously Unidentified Cultural Heritage					
Heritage Management Plan	4.4	Notwithstanding the recognised low potential for other archaeological material to be present within the areas to be disturbed by mining and mining-related activities, TCPL will continue to invite representatives of the RAP's on a rotational roster to monitor all ground disturbance works throughout the life of the mine.	This occurs, documnetation supporting this sighted includeding the observations of the RAPs for the clearance of the northern extension area	Compliant			





Reference	Clause	, ,	This occurs, though sites and artifacts that	Audit Finding	Consequence	Likelihood	Risk
		, ,	This occurs, though sites and artifacts that				IVION
Heritage Management Plan	4.4	· · · · · · · · · · · · · · · · · · ·	have been found in this way have generally been sites previously found but not able to be relocated during salvage.	Compliant			
Heritage Management Plan	4.4	6. If the find is confirmed as a site, the archaeologist will complete a Sites Register Card and forward to OEH for inclusion on the Aboriginal Heritage Information Management System (AHIMS) database. 7. If the find is confirmed as a site, TCPL will offer to facilitate inspection of the new site by other registered Aboriginal groups prior to salvage. 8. Subject to the recommendations of the archaeologist and the RAP's, artefacts from the site will be collected following the protocols outlined in Section 4.1 prior to further work being undertaken in the vicinity of the site.	Noted				
Heritage Management Plan	4.4	, , , ,	No human remains found in the audit period	Not Applicable			





Defenses	Clause	Requirement	Evidence	A dik Finadina		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.5	The biodiversity offset area ("Willeroi") established for the Tarrawonga extension project was not subject to a cultural heritage assessment, and as such is not covered by the provisions of the Part 3A Project Approval relating to cultural heritage. On this basis, the provisions of the NPW Act apply in full at the "Willeroi" property in relation to the protection of items of cultural heritage. The biodiversity offset area is being set aside for conservation purposes, in perpetuity, and is unlikely to be subject to any surface disturbance activity other than those activities associated with general land management and enhancement of vegetation communities. As a consequence, any disturbance activities within the "Willeroi" property will be subject to the Due Diligence Code of Practice for the Protection of Aboriginal objects in NSW. The completion of the Aboriginal Cultural Heritage Strategy, as required under the Project Approval, will include additional consideration to the Aboriginal heritage values associated with the area, and the potential opportunities to enhance and conserve Aboriginal heritage within the biodiversity offset site.	No work done at Willeroi to date so this has not been required.	Not Applicable			
	4.6 Aboriginal Horitag	ge Conservation Strategy					
Heritage Management Plan	4.6	Recognising the cumulative impact of proposed mining activities within the wider Leard State Forest area, in addition to this HMP, an Aboriginal Heritage Conservation Strategy will be prepared and implemented for the Boggabri-Tarrawonga-Maules Creek complex, to enhance and conserve the Aboriginal Cultural Heritage Values (both cultural and archaeological) of this area and provide for their long term management. The Strategy will be developed in accordance with the following requirements of the Tarrawonga Project Approval:	Noted				
Heritage Management Plan	4.6	The Aboriginal Cultural Heritage Strategy will focus on opportunities to conserve and enhance areas of cultural heritage significance that are not impacted by the approved projects, and how these areas can be used to enhance knowledge and cultural understanding in the wider community.	Noted				
Heritage Management Plan	4.6	Development of the Aboriginal Cultural Heritage Strategy will be undertaken progressively with the Boggabri and Maules Creek projects, with an approved expert, as endorsed by the Director General, and in consultation with the RAP's for each project. It is expected that the strategy will be submitted for approval by 30th June 2014.	Noted				
	4.7 Workforce Educat						
Heritage Management Plan	4.7	Prior to any person undertaking work on the mine site, they will undergo cultural awareness training as provided in the general mine site induction. Any person undertaking soil stripping activities will undertake such works in consultation with the Environmental Officer and RAP's undertaking site monitoring during pre-strip works.	The Induction includes information on arch heritage and there are further controls and training for those involved in ground distrubance.	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	4.7	The Environmental Officer at site will also provide general education to the workforce via toolbox talks on a range of environmental issues relevant to operations. This will include discussion on matters relating to cultural heritage and the obligations of personnel to preserve and protect items of cultural heritage significance. This will be further embodied in preclearance procedures at site that requires Environmental Officer sign off prior to site disturbance to ensure Aboriginal heritage items are considered in site operations.	This occurs, per OCE and EO	Compliant			
	5 Consultation, Moni						
Heritage Management Plan	5.1 Consultation Prot	Subject to any modifications initiated by the RAP's, the following protocol will be followed with respect to consultation: 1. Prior to any topsoil stripping campaign, the Operations Manager, or authorised delegate, will notify the RAP's and advise of the extent, location, timing and expected duration of the planned campaign and invite a representative to monitor the topsoil stripping activities. For safety and logistical reasons, RAP involvement will be rotated such that one representative from each of three groups will be on site each day as required (i.e. three rotations of three groups of representatives to ensure inclusion of the nine Aboriginal stakeholder organisations). A minimum of three days' notice will be provided in advance of each campaign to enable representation as required.	Evidence sighted of the RAPs involvement in this activity	Compliant			
Heritage Management Plan	5.1	2. At the request of the RAP's, the Operations Manager, or authorised delegate, will avail themselves to attend community meetings at the RAP's offices or to arrange site visits for interested personnel or members. A meeting will be held with RAP's prior to each salvage event to ensure the stakeholders are informed of the intended works and a clear line of communication is established for those works, outlining the expectations of the RAP's, the Archaeologist and TCPL throughout the process.		Not Applicable			
Heritage Management Plan	5.1	In the event that any possible archaeological sites are identified in the absence of the RAP's, the Operations Manager, or authorised delegate, will notify the RAP's following the initiation of the procedures identified in Section 4.4.	·	Not Applicable			
Heritage Management Plan	5.1	4. If requested by the RAP's, TCPL will facilitate access to known sites within the mining lease areas, along the Haul Route and in the Biodiversity Offset Area, subject to satisfaction of required safety protocols. The Aboriginal Heritage Conservation Strategy, required under Condition 51 of PA 11_0047 requires determination of cultural heritage values within the Biodiversity Offset Strategy Areas. Development of this strategy, in consultation with the other mining projects and Aboriginal Stakeholders will further define heritage values of the offset areas, and any ongoing access arrangements to those sites.	No such requests	Not Applicable			





Deference	Clause	Requirement	Evidence	Andie Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	5.1	5. If requested by members of the Aboriginal community, TCPL wil facilitate access to any artefact material salvaged from the TCM and stored within an agreed Keeping Place either at the site or other Whitehaven controlled property. Any material held at the Cumbo Gunerah Keeping Place would be subject to any access requirements of the Red Chief LALC.	No such requests	Not Applicable			
Heritage Management Plan	5.1	The effectiveness and value of the consultation with the RAP's will be periodically reviewed. In the event there is agreement that the approach to consultation needs to change, the changed procedures would be documented and a copy forwarded to the Director-General and recorded in the relevant AEMR/Annual Review. All consultation will be diarised.					
Heritage Management Plan	5.1	For the purposes of initial communication procedures, the following will be adopted: - At the first RAP consultation session, a contacts register will be distributed comprising the RAP's name, address, phone and mobile phone numbers, as well as the Tarrawonga mine representatives contact details. Upon confirmation that all details are correct, copies of the contacts register will be distributed between the RAP's and Tarrawonga Mine. - The first point of contact for matters pertaining to cultural heritage at the Tarrawonga Mine will be the Tarrawonga Environmental Officer. The Tarrawonga Environmental Officer will coordinate with the RAP's, the requirements for attendance at site for salvage activities, soil stripping monitoring and revisions to this management plan. The Tarrawonga Environmental Officer will also retain the register of work activities, and the status of the rotation system for engagement of the RAP's for monitoring works. - The Tarrawonga Environmental Officer will ensure that at a minimum, 3 days' notice will be provided to the RAP's on rotation for soil stripping monitoring at site. In the event that a RAP is not available for a monitoring event, the next RAP in rotation will be contacted for that event.		Compliant			
Heritage Management Plan	5.1	The responsibilities for each party in relation to these communication procedures are as follows: Tarrawonga will generate and distribute a contacts register for ratification at the first RAP consultation session prior to the initial salvage works.	Noted				
Heritage Management Plan	5.1	The RAP's will provide current contact details for inclusion on the register. In the event of any change of contact details, the RAP's will provide notification to the Tarrawonga Environmental Officer to update the contacts register accordingly.	Noted				





Reference	Clause	Requirement	Evidence	Audit Einding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	5.1	- Tarrawonga will provide reasonable notice to the RAP's (at a minimum 3 days) for soil stripping monitoring requirements. Upon provision of notice, and where a message has been left with the RAP's requesting confirmation of availability, and availability is not confirmed within 24hrs of notice, Tarrawonga will contact the next RAP for the work as per the rotation register. For salvage works undertaken under this plan, at least 1 weeks notice will be provided to the RAP's. All RAP's will be invited to participate in the cultural values assessment prior to salvage works commencing. Actual salvage works will be undertaken in accordance with the rotation system.		Compliant			
Heritage Management Plan	5.1	- The RAP's will attend the works programs at Tarrawonga site in accordance with the timeframes provided in the notification process. The RAP's will abide by all instructions given to them by Tarrawonga site personnel when on the Tarrawonga Mine Site. The RAP's will abide by Tarrawonga safety standards as advised by personnel and undertake required induction processes to access the site. The RAP's will provide information, advice and knowledge of artefactual material identified on the site and assist the archaeologist in the salvage procedure.	Noted				
Heritage Management Plan	5.1	The intent and suitability of this communication protocol will be discussed at the initial RAP's meeting to verify agreement with the process. Any suggested modifications to this process will be considered and where appropriate, modifications to the protocol will be made and this management plan updated accordingly.	All parties currently satisfied with the operation of the communication protocol, meeting minutes sighted during site visit	Not Applicable			
	5.2 Monitoring and R	eporting					
Heritage Management Plan	5.2	Monitoring undertaken by the RAP's will be documented. In the event that an Aboriginal artefact or site is identified, and in accordance with the procedure documented in Section 4.4, a site register card will be completed and forwarded to OEH for entry onto the AHIMS database.	Salvage report and RAPs soil stripping notes provided as evidence	Compliant			
Heritage Management Plan	5.2	In addition, the results of monitoring undertaken on the mine site will be summarised in each relevant AEMR/Annual Review, with the RAP's kept informed as to developments by virtue of the ongoing consultation with those groups. TCPL will discuss the measures implemented to preserve and protect Aboriginal cultural heritage in the AEMR/Annual Review.	The AEMR provides this information	Compliant			
		idents, Complaints and Non-Compliances					
	6.1 Compliance and II	· ·	lana ahawa				
Heritage Management Plan	6.1	To ensure compliance with the requirements of the NPW Act, EP&A Act and the conditions of PA 11_0047, the following measures will be implemented: • Continue to include information relating to Aboriginal cultural heritage in inductions for all site personnel (refer to Section 4.7);	see above	Compliant			
Heritage Management Plan	6.1	Maintain protection measures for sites;	see above	Compliant			
Heritage Management Plan	6.1	Maintain protection measures not sites, Maintain ongoing consultation with the RAP's over the life of the mine, including appropriate RAP representation during topsoil stripping, disturbance activities and archaeological fieldwork (e.g. collection of artefacts);		Compliant			





Poforonco	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Heritage Management Plan	6.1	 Maintain a record of known Aboriginal heritage sites (e.g. status and location) including identification of these sites on mine plans. Copies of the site cards completed for each heritage site will be retained in the site office for detailed information about each site; 		Compliant			
Heritage Management Plan	6.1	Require a site disturbance protocol be followed at site whereby site disturbance does not commence until sign off is provided by the Environmental Officer;	see above	Compliant			
Heritage Management Plan	6.1	Undertake all salvage works in accordance with requirements outlined in Section 4.1;	see above	Compliant			
Heritage Management Plan	6.1	Undertake all consultation as outlined in Section 5.1;	see above	Compliant			
Heritage Management Plan	6.1	Undertake regular reviews of this HMP as outlined in Section 7.	see above	Compliant			
Heritage Management Plan	6.1	Any incidents will be recorded and managed via the Whitehaven incident management process. Incident reporting will be conducted in accordance with the requirements of Condition 8 of Schedule 5 of PA 11_0047, requiring notification at the earliest possible opportunity, the Director General, and any other relevant agency, of any incident that has caused or threatens to cause material harm to the environment, or other breach or incident associated with the Project. Any notification would be followed by a written report within 7 days of the incident, and provision of any other information as may be requested.		Compliant			
Heritage Management Plan	6.1	A discussion of any incidents will also be provided in the AEMR/Annual Review.	see above	Compliant			
Heritage Management Plan	6.1	For the purposes of this plan, an incident is defined as any activity or action that results in the damage to, or destruction of heritage sites within the Tarrawonga project boundary.	see above	Compliant			
	6.2 Complaints	,					
Heritage Management Plan	6.2	A complaints management protocol has been developed to ensure an appropriate and consistent level of reporting, response and follow-up is adopted by TCPL. A complaint is verbal or written advice from any party in response to activities undertaken at the Tarrawonga Mine site. The following complaints management protocol will be followed on all complaints received relating to Aboriginal cultural heritage: A publicly advertised telephone complaints line will be in place to receive complaints during operating hours and record complaints at other times.	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint. o The action taken by TCM in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCM, the reason why no action was taken.	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational).	No complaints regarding Heritage issues	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Filluling	Consequence	Likelihood	Risk
Heritage Management Plan	6.2	 Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site. 	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	 Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register. 	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	• If necessary, follow-up monitoring or will take place to confirm the source of the complaint is adequately mitigated.	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	 A copy of the Complaints Register will be kept by TCM and made available to the Tarrawonga Coal Mine Community Consultative Committee (CCC) and the complainant (on request). Complaints are also listed on the Whitehaven website (www.whitehavencoal.com.au) and a summary of complaints received every 12 months will be provided to DPI, NSC, GSC, OEH/EPA, DRE and the CCC through the AEMR/Annual Review. 	No complaints regarding Heritage issues	Not Applicable			
Heritage Management Plan	6.2	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Environmental Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately.	No complaints regarding Heritage issues	Not Applicable			
	6.3 Dispute Resolution						
Heritage Management Plan	6.3	In the event of a dispute arising between TCPL and the RAP's in relation to implementation of this management plan, the following protocol will be followed: 1. The Environmental Manager will discuss the issue with the disputer within 2 weeks (10 working days) of becoming aware of the dispute. TCPL may engage a Heritage Advisor to assist in this process. 2. If a resolution is not reached from the above discussions, a meeting will be convened at site between TCPL, the RAP's and the Heritage Advisor to discuss the issue and reach consensus. This meeting is to be convened within 1 month of the discussion identified in 1 above. 3. If a resolution is still not achieved, the matter will be referred to the Director General for a decision within 2 weeks (10 working days) of the meeting identified in 2 above.		Not Applicable			
Heritage Management Plan	6.3	Where a dispute is technical in nature (ie. relates to methodology or interpretive issues) for implementation of the management plan, the same process will be followed, however if a consensus is not reached, TCPL may refer the matter to an independent third party for review. As TCPL will have the right of engagement of an independent third party, the costs associated with engagement of an independent third party will be borne by TCPL. The independent third party will be identified by writing to one of the following: The President Australian Archaeological Association School of Social Science University of Queensland Brisbane QLD 4072 or The President Australian Association of Consulting Archaeologists Inc PO Box 8382 Armadale Vic 3143		Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Reference	Clause			Addit Fillding	Consequence	Likelihood	Risk	
	7 Document Review a	nd Continuous Improvement						
Heritage Management Plan	7	This document will be reviewed internally on an annual basis following completion of the Annual Review, and/or following a heritage related incident, audit, or modification to the conditions of approval. In addition, every 2 years, the plan will be subject to review and update in consultation with the RAP's and relevant agencies.	Not yet required.	Not Applicable				
Heritage Management Plan	7	TCM will investigate and implement ways to improve the environmental performance of the project over time. This will be achieved by keeping abreast of best practice in the industry for cultural heritage management and controls and reporting on outcomes of surveys and monitoring in the AEMR/Annual Review.	see other comments in other sections of audit	Compliant				





Aborginal Heritage Conservation Strategy for the BTM Complex and Blodiversity Offset Areas 2.1 Environmental Planning and Assessment Act 1979 Aborginal Heritage Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 2.1 The MCAN have been used under Section 4.1 of FP&A Act, Part S. 3, 4 and 5 relate to Aborginal Heritage. Conservation Strategy for the 9TM Complex and Stockwesty Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 9TM Complex and Blodiversity Offset Areas Aborginal Heritage Conservation Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and Wildlife Act 1974 (as amended) (1974 Act) Offset Areas Description Strategy for the 1 National Parks and W	Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
2.1 Eventometral Planning and Assessment Act 1979 The PPAA Act regulates a system of environmental planning and Aboriginal Heritage Conservation Strategy for the IPA Act and Season of the PPAA Act and Pearly and Pearly and Pearly and Season of the PPAA Act and Pearly an	Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
Aboriginal Heritage Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 2.1 The MCA have been used under Section 4.1 of EPBA Act, as the mines in the ITM complex as existed Aboriginal Heritage. Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas 2.2 NSW Legislation Regulation Bendinger. 2.2 NSW Legislation Regulation Bendinger. Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 8TM Complex and Biocilversity Offset Areas Aboriginal Heritage Conservation Strategy for the 1 In Invest (in the PRIVA Act, and exceeding places) 1 In Invest (in the PRIVA Act, and exceeding doubt exceeding the Privategy of the Strategy for the 1 In Invest (in the PRIVA Act, and exceeding doubt exceeding the Privategy of the Strategy for the 1 In Invest (in the PRIVA Act, and exceeding doubt exceeding the Privategy of the Strategy for the Strate				reas				
Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Altonomical Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the STM Complex and Biodiversity Offset Areas Aboriginal Heritage C								
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the DTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the DTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the DTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the DTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the DTM Complex and Biodiversity Offset Areas DTM MCAD Associated Application Developments (\$50.9). Although a number of Acts and regulations protect and manage outbrack application through a number of Acts and regulations protect and manage outbrack and press and Wildlife Act 1374 (as amended) Offset Areas DTM Complex and Biodiversity Offset Areas DTM MCAD Associated Acts and DTM Act		2.1 Environmental Pla	_	I.e.	1	T T	T	
Aboriginal Heritage Conservation Strategy for the BTM Complex and Students and Students and Strategy for the BTM Complex and Students and Students and Strategy for the BTM Complex and Students an	Conservation Strategy for the BTM Complex and Biodiversity	2.1	assessment for New South Wales. Land use planning requires that environmental impacts are considered, including the impact on cultural heritage	Noted				
Offset Areas 2 2 NSW Legislation Regulating Aboriginal Cultural heritage Although a number of Acts and regulations protect and manage Cultural heritage in New South Wales, the Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2 1 National Parks and Wildlife Regulation 2009 3 Environmental Planning and Assessment Act 1979 1 In Interf. the NPW Act (as samended) protects Aboriginal heritage (places, sites and objects) within NSW, and the National Parks and Wildlife Regulation 2009 provides a framework for undertaking activities and exercising due diligence. 2 2.1 National Parks and Wildlife Rest 1974 The National Parks and Wildlife Rest 1975 The National Parks and Wildlife Rest 1974 The National Parks and Wildlife Regulation 2009 The Natio	Aboriginal Heritage Conservation Strategy for the	2.1	4 and 5 relate to Aboriginal heritage. The MCoA have been used under Section 4.1 of EP&A Act, as the mines in the BTM complex are State	Noted				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2 I National Parks and Wildlife Regulation 2009 B Environmental Planning and Assessment Act 1979 In Price In Interior In Price In Interior In Interior In Interior In Interior In Int								
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2 a lin brief, the NPW Act (as amended) 2.3 National Parks and Wildlife Regulation 2009 2.2.1 National Parks and Wildlife Regulation 2009 provides a framework for undertaking activities and exercising due diligence Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2.1 National Parks and Wildlife Act 1974 (as amended) provides a framework for undertaking activities and exercising due diligence Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 3 "A person must not harm or descrate an object with the person knows is an Aboriginal object" \$86(1). 3 "A person must not harm or descrate an Aboriginal place" \$86(2). 3 "A person must not harm or descrate an Aboriginal place" \$86(2). 3 "A person must not harm or descrate an Aboriginal place" \$86(2). 3 "A person must not harm or descrate an Aboriginal place" \$86(3). 4 "A person must not harm or descrate an Aboriginal place" \$86(3). 5 "A person must not harm or descrate an Aboriginal place" \$86(4). 5 "A person must not harm or descrate an Aboriginal place" \$86(4). 5 "A person must not harm or descrate an Aboriginal Parks and Wildlife Regulation 2009 (NPW Regulation) provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage (conservation Strategy for the BTM Complex and Biodiversity Offset Areas 4 "The National Parks and Wildlife Regulation 2009 (NPW Regulation) Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage (conservation Strategy for the BTM Complex and Biodiversity Offset Areas 5 "The National Parks and Wildlife Regulation 2009 (NPW Regulation) Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage (conservation Strategy for the BTM Complex and Biodiversity Offset Areas 6 "The National Parks and Wildlife Regulation 2009 (NPW Regulation) Provides a frame		2.2 NSW Legislation R	, , , , , , , , , , , , , , , , , , , ,		1	,		
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2.1 National Parks and Wildlife Regulation 2009 provides a framework for undertaking activities and exercising due diligence. The National Parks and Wildlife Act 1974 The National Parks and Wildlife Act 1974 The National Parks and Wildlife Act 1974 (as amended) (NPW Act) protects Aboriginal Heritage (places, sites and objects) within NSW. Protection of Aboriginal heritage is outlined in S86 of the NPW Act, as follows: 2.2.1 Bar Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2.2 National parks and Wildlife Regulation 2009 2.2.2 National parks and Wildlife Regulation 2009 The National Parks and Wildlife Regulation 2009 (NPW Regulation) provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas The National Parks and Wildlife Regulation 2009 (NPW Regulation) Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas The National Parks and Wildlife Regulation 2009 (NPW Regulation) Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage. 2.2.2 National parks and Wildlife Regulation 2009 (NPW Regulation) Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage. 2.2.2 Particular Provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage (Provides and Exercising Activities and Exercising Activitie	Conservation Strategy for the BTM Complex and Biodiversity	2.2	cultural heritage in New South Wales; the primary ones that apply to this report include: Rational Parks and Wildlife Act 1974 (as amended) Rational Parks and Wildlife Regulation 2009	Noted				
The National Parks and Wildlife Act 1974 (as amended) (NPW Act) protects Aboriginal heritage (places, sites and objects) within NSW. Protection of Aboriginal heritage is outlined in s86 of the NPW Act, as follows: 2.2.1 2.2.1 2.2.1 3 "A person must not harm or desecrate an object that the person knows is an Aboriginal object" s86(1), 3 "A person must not harm and Aboriginal object" s86(2) 4 "A person must not harm or desecrate an Aboriginal place" s86(4). 2.2.2 National parks and Wildlife Regulation 2009 The National Parks and Wildlife Regulation 2009 (NPW Regulation) provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal Conservation Strategy for the BTM Complex and Biodiversity Offset Areas The National Parks and Wildlife Regulation 2009 (NPW Regulation) their tage. The NPW Regulation utilines the recognised due diligence codes of practice which are relevant to this report, but it also outlines procedures for AHIP applications and ACHCRs (DECCW 2010a); amongst other	Conservation Strategy for the BTM Complex and Biodiversity	2.2	(places, sites and objects) within NSW; and the National Parks and Wildlife Regulation 2009 provides a	Noted				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2.1 B "A person must not harm or desecrate an object "s86(1), B "A person must not harm or desecrate an Aboriginal place" s86(4). The National Parks and Wildlife Regulation 2009 Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Abo		2.2.1 National Parks a	ind Wildlife Act 1974					
The National Parks and Wildlife Regulation 2009 (NPW Regulation) provides a framework for undertaking Aboriginal Heritage activities and exercising due diligence with respect to Aboriginal Conservation Strategy for the BTM Complex and Biodiversity Offset Areas 2.2.2 The National Parks and Wildlife Regulation 2009 (NPW Regulation) provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage. The NPW Regulation 2009 (NPW Regulation) heritage. The NPW Regulation 2009 (NPW Regulation) heritage activities and exercising due diligence with respect to Aboriginal heritage. The NPW Regulation 2009 (NPW Regulation) hoted	Conservation Strategy for the BTM Complex and Biodiversity Offset Areas		protects Aboriginal heritage (places, sites and objects) within NSW. Protection of Aboriginal heritage is outlined in s86 of the NPW Act, as follows: "A person must not harm or desecrate an object that the person knows is an Aboriginal object" s86(1), "A person must not harm an Aboriginal object" s86(2) "A person must not harm or desecrate an Aboriginal place" s86(4).					
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas Offset Areas Double a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage. The NPW Regulation outlines the recognised due diligence codes of practice which are relevant to this report, but it also outlines procedures for AHIP applications and ACHCRs (DECCW 2010a); amongst other		2.2.2 National parks a			·	1	·	
	Conservation Strategy for the BTM Complex and Biodiversity	2.2.2	provides a framework for undertaking activities and exercising due diligence with respect to Aboriginal heritage. The NPW Regulation outlines the recognised due diligence codes of practice which are relevant to this report, but it also outlines procedures					
2.2.3 Aboriginal Land Rights Act 1983		2 2 3 Aboriginal Land	regulatory processes.					





Deference	Clarras	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	2.2.3	The purpose of this legislation is to provide land rights for Aboriginal people within NSW and to establish Local Aboriginal Land Councils (LALCs). The land able to be claimed by LALCs, on behalf of Aboriginal people, includes Crown Land that (s36): Is able to be lawfully sold, leased, reserved or dedicated; Is not lawfully used or occupied; Does not comprise lands which, in the opinion of the Crown Lands Minister, are needed or are likely to be needed for residential purposes; Are not needed, nor likely to be needed for an essential public purpose; Does not comprise land under determination by a claim for native title; and	Noted				
	2.2.5	Title.					
	2.3 Federal Legislation 2.3.1 Native Title Act	n regulating Aboriginal Cultural Heritage					
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	2.3.1	The Commonwealth Government enacted the Native Title Act (1993) to formally recognise and protect native title rights in Australia following the decision of the High Court of Australia in Mabo & Ors v Queensland (No. 2) (1992) 175 CLR 1 ("Mabo").	Noted				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	2.3.1	Although the presumption of native title can be in any area where an Aboriginal community or group can establish a traditional or customary connection with that area, native title can be extinguished by a number of ways: Bland that was designated as having freehold title before 1 January 1994 Bany commercial, agricultural, pastoral or residential lease. Land that has been used for the construction or establishment of public works for as long as they are used for that purpose.					
	6 Aboriginal Values As	sessment Methodology			T T		
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	6	The methodology for Aboriginal consultation for the AHCS was mailed to the RAPs (7 April 2014) to allow all knowledge holders the opportunity for input into the proposed framework for consultation. The methodology adopted for this AHCS is provided below and is followed by a summary of the documentation of consultation.	Evidence in appendix to AHCS	Compliant			
	6.1 Methodology for A	Aboriginal Consultation					
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	6.1	The methodology for Aboriginal consultation followed the Ask First Principles and adhered to the guiding principles of the ACHCR process. Five formal opportunities for input into the strategy were provided to the RAPs as part of this AHCS process (Figure 13). Informal opportunities for input (written or verbal) were also provided throughout the six month process.	The reply letter from the RAPs disputes this. But the dopcumentation attached to the strategy regarding consultation supports this approach being followed. Note this strategy is not yet approved.	Compliant			
	7 Aboriginal Values As						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	7	Consultation for this process was undertaken in accordance with the OEH ACHCRs guideline and the Australian Heritage Commission's Ask First guideline. The BTM Complex had already completed Stage One of the ACHCR process and was able to provide RPS with a list of identified RAPs. The identification of RAPs was drawn from government regulatory bodies which included: Registrar (Aboriginal Land Rights Act, 1983), relevant OEH Environmental Protection Regulation Group (EPRG) Regional Office, National Native Title Tribunal, Native Title Services Corporation Limited, relevant Catchment Management Authority, Local Aboriginal Land Council and relevant local councils.	Noted, consultation is almost complete.				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	7	A RAP Tender Document, identifying the processes for the AHCS, was sent to the RAPs. The RAP tender document was the first opportunity for the RAPs to express interest in the AHCS. A second opportunity for participation in the AHCS was Workshop 1, which outlined the AHCS in more detail. Workshop 2, the third opportunity to participate in the AHCS, provided an opportunity to put forward conservation strategies and culturally map the BOAs. The draft report was sent to all RAPs and a 28 day review period was given (in accordance with the ACHCR process) so RAPs could comment on the draft AHCS. The review period was the fourth opportunity for the RAPs to comment on the AHCS. Following the draft report, the third workshop (and fifth opportunity) was for the RAPs to comment on the draft AHCS. The third workshop was the final opportunity for the RAPs to supply comments on the AHCS and inclusion into the final report. The table below summarises the five opportunities for the RAPs to supply comments on the AHCS.					
	9 Implementation						
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	9	This AHCS will be implemented using a staged approach (Figure 14). Stage 1 will involve evaluation of conservation options. This evaluation process would assess the effectiveness of the options in conserving archaeological and Aboriginal cultural values, as well as considering the practicalities of their implementation. Options would be prioritised on this basis. The most practicable and effective option(s) would be selected. Stage 2 will involve conducting the activity associated with the selected conservation option(s). Stage 3 will monitor the activity during implementation, to ensure it is meeting its purpose and will consider inputs from the RAPs and DEH, where relevant. Stage 3 would also evaluate if changes to the implementation of the activity are required, or if additional conservation options should be considered. Stage 4 will document the activities undertaken for the conservation option(s) in an appropriate format and provide this information to the RAPs and OEH at end of the implementation phase.					
	10 Conclusion			1			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	10	A Cultural Heritage Strategy for the BTM Complex and BOA's has been developed in compliance with the approval conditions for the three mines. Its objective is to enhance and preserve cultural heritage. The strategy has been developed on the basis of an extensive desktop analysis complemented by the provision of extensive opportunities for consultation.	Noted				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	10	The strategy includes a high level implementation plan which will be implemented in the future.	Noted				
Aboriginal Heritage Conservation Strategy for the BTM Complex and Biodiversity Offset Areas	10	The next step in finalising the strategy is another round of consultation followed by submission to DoPE.	Noted				





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
	Bushfire Managemer	nt Plan					
	3 Responsibilities	Territoria de la compansión de la compan	I			<u> </u>	
Bushfire Management Plan	3	The Operations Manager has the overall responsibility for the management of the mine, including the responsibility for compliance with the commitments within the Bushfire Management Plan and the mine's Emergency Management System developed in accordance with the Coal Mines Health and Safety Regulations 2006.	Noted				
Bushfire Management Plan	3	The Operations Manager must ensure that all the onsite safeguards and controls are in place and are effective for the prevention and control of bushfires, and that the desired environmental outcomes are achieved.	Noted				
Bushfire Management Plan	3	All mine personnel have responsibilities to observe and comply with the requirements of the Bushfire Management Plan, to participate in fire training if requested, and to observe best practice when handling flammable or hazardous materials and operating heavy machinery within the ML and MLAs.	The induction process addresses incident management and fires/bushfires	Compliant			
Bushfire Management Plan	3	A Fire Officer will be appointed by the Manager of Mining Engineering as required by the Coal Mines Health and Safety Regulations 2006. The Fire Officer's responsibilities will include: • Recruiting sufficient volunteers to establish fire fighting teams, of up to 5 persons per team, on each shift and training them in fire fighting theory and practice at 3 monthly intervals;	OCE on duty is the nominated fire officer for each shift and they have suitable levels of volunteers	Compliant			
Bushfire Management Plan	3	Carrying out inspections and arranging for the repair and maintenance of fire extinguishers in accordance with (Australian Standard) AS1851, part 1 (Portable fire extinguishers and fire blankets) and fire hoses in accordance with AS1851, Part 9 (Delivery lay flat fire hose);	Contractor checks site equipment, maintenance team supervise fire fighting equipment.	Compliant			
Bushfire Management Plan	3	 Arranging for the replacement of fire fighting equipment reported as damaged in the book kept for the purpose and reporting to the Operations Manager any matter unable to be rectified; 	Contractor checks site equipment, maintenance team supervise fire fighting equipment.	Compliant			
Bushfire Management Plan	3	Checking that fire fighting equipment at the mine, and following repair or replacement, is compatible with NSW fire brigades;	Contractor checks site equipment, maintenance team supervise fire fighting equipment.	Compliant			
Bushfire Management Plan	3	Clearly marking with signs all fire fighting equipment on the mine site; and	Observed on site	Compliant			
Bushfire Management Plan	3	Maintaining an up-to-date Fire Control Plan that delineates any fixed fire control installations and displaying a copy of the Plan in the Mining Supervisor's Office.	This is the case, sighted	Compliant			
Bushfire Management Plan	3	The Fire Officer will report directly to the Operations Manager and Manager of Mining Engineering and will be required to complete monthly reports of his/her inspections.	This occurs	Compliant			
	5 Controls	E	L				
Bushfire Management Plan	5	TCPL is conscious of the need to ensure adequate bushfire controls are maintained to minimise the potential bushfire hazard of the mine.	Noted				
		nt and Mine Operations					_
	5.2.1 Earthmoving M	,	T === 11		T.		
Bushfire Management Plan	5.2.1	 All earthmoving machinery will be maintained in good working order with efficient exhaust systems and spark arrestors. Regular inspections will be carried out. 	see EPL section of audit	Compliant			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk	
		 All earthmoving machinery and mobile equipment will be fitted with appropriate sized and approved fire extinguishers and/or fire suppression systems suitable for the control of flammable liquid and electrical fires. 	This occurs	Compliant				
	5.2.2 Clearing Opera	tions						
Bushfire Management Plan	5.2.2	TCPL will enforce clearing restrictions. Clearing operations will not be undertaken during periods of extreme fire danger as defined by the Bureau of Meteorology.	See MOP, BOMP	Compliant				
Bushfire Management Plan	5.2.2	 The retention of cleared debris for use in the rehabilitation program will be strictly adhered to and there will be no burning of debris. 	See MOP, BOMP	Compliant				
Bushfire Management Plan	5.2.2	All vehicle movements within ML and MLA areas will be confined to defined roads or tracks, unless otherwise authorised.	See MOP, BOMP	Compliant				
Bushfire Management Plan	5.2.2	Where appropriate, controlled high intensity short term grazing will be employed to assist in the reduction of vegetative fuel loads.	See MOP, BOMP this has not been utilised to date but here is no significant buildup of fuel on the site.	Not Applicable				
	5.2.3 Blasting Opera							
Bushfire Management Plan	5.2.3	 All flammable material will be removed by pre-stripping the topsoil prior to any drilling and blasting operations taking place. 	This occurs	Compliant				
Bushfire Management Plan	5.2.3	Blast design will be undertaken by qualified personnel	This occurs	Compliant				
Bushfire Management Plan	5.2.3	All blasting will be carried out in accordance with the Explosives Act 2003 and Coal Mines Health and Safety Regulations 2006.	This occurs	Compliant				
	5.2.4 Welding Opera	itions						
Bushfire Management Plan	5.2.4	 All welding activities will, as far as practicable, be conducted and confined to the main workshop area. In the event that welding or cutting is to be conducted outside the main workshop area the mine's Hot Work Permit system will be implemented. 	This occurs	Compliant				
	5.2.5 Fixed Plant and	Buildings						
Bushfire Management Plan	5.2.5	Maintenance of housekeeping by mine management.	Housekeeping was reasonable with respect to fire risk	Compliant				
Bushfire Management Plan	5.2.5	 All workshops and offices will be installed with an appropriate fire fighting equipment and their location will be indicated by signs compliant with Australian Standards. 	Observed on site	Compliant				
	5.2.6 Exploration Ac							
Bushfire Management Plan	5.2.6	 Exploratory drilling on any WHC exploration areas near TCM will not be undertaken during periods of extreme fire danger as defined by the Bureau of Meteorology or on total fire ban days as defined by the NSW Rural Fire Service. 	Inductions for contractors and prework risk assessments cover this issue	Compliant				
Bushfire Management Plan	5.2.6	The slashing or clearing of access tracks to exploratory drill sites will not be permitted during total fire bans days as defined by the NSW Rural Fire Service.	This has not occured in the audit period.	Not Applicable				
	5.2.6 Exploration Ac		Industrial Control of the Control of	I	T I			
Bushfire Management Plan	5.2.6	 Exploratory drilling on any WHC exploration areas near TCM will not be undertaken during periods of extreme fire danger as defined by the Bureau of Meteorology or on total fire ban days as defined by the NSW Rural Fire Service. 	Inductions for contractors and prework risk assessments cover this issue	Compliant				
	5.2.7 Fuel and Oil M							





Reference	Clause	Requirement	Evidence	Audit Einding		Risk	
Kererence	Clause			Audit Finding	Consequence	Likelihood	Risk
Bushfire Management Plan	5.2.7	 All fuel and oil storage will be located and constructed in accordance with the requirements of the Work Health and Safety Act 2011. 	The audit team were not qualified to review this requirement	Not Applicable			
Bushfire Management Plan	5.2.7	Undertake refuelling within designated fuel bays or within cleared areas of the mine site.	This occurs except for inpit refuelling	Compliant			
Bushfire Management Plan	5.2.7	Turn vehicles off during refuelling.	This occurs	Compliant			
Bushfire Management Plan	5.2.7	Fuel and oil storage areas will be signposted as to the contents of the storages and will be fitted with approved fire extinguishers.	The site inspection confirmed this	Compliant			
Bushfire Management Plan	5.2.7	All fuel tanks on-site will be fully or self bunded to ensure that in the event of a leak or rupture, no fuel escapes from the bunded area. Each bunded area will have the capacity of at least 110% of the largest tank. Bunds may be integrated, (ie. form part of the tank structure) or be external.	This occurs and was inspected	Compliant			
Bushfire Management Plan	5.2.7	 Drainage from the workshop, workshop apron and wash down areas will be directed to an oil separator and containment system for subsequent pump out and disposal. 	This occurs, observed in site inspection	Compliant			
Bushfire Management Plan	5.2.7	A foam generator and an adequate supply of high expansion foam will be available on-site for fighting oil-based fires.	This is available, no such fires in the audit period (interview with EO)	Compliant			
	5.2.8 Mine Site Prote	ocols					
Bushfire Management Plan	5.2.8	Designated "No Smoking" areas will be clearly marked and enforced onsite. These will include: o Fuel and oil storage areas; o Within areas flagged or barricaded areas in preparation for blasting; o When transporting explosives or within 20m of a vehicle transporting explosives; o Within workshops; o All buildings and offices; and o Any gas cylinder storage areas.	Observed on site	Compliant			
Bushfire Management Plan	5.2.8	Equipment will not be stored on uncleared ground.	Observed on site	Compliant			
Bushfire Management Plan	5.2.8	Vehicular access will be maintained around all areas of mining- related activities.	Observed on site	Compliant			
Bushfire Management Plan	5.2.8	The following measures will be adopted to minimise the potential for spontaneous combustion: Regularly inspect and water stockpiles. Control stockpile height and volume to limit the duration coal is retained in stockpiles. Routinely turn over coal in the ROM and product coal stockpile areas to minimise stockpile residence time. In longer term stockpiles, creation of a shallow stockpile batter face to the direction of the prevailing wind. Compaction of the stockpile area by use of mobile equipment. Visual inspections for evidence (visual or smell) of combustion.	No occurrances of spon com on record	Not Applicable			
Bushfire Management Plan	5.2.8	Fire prevention and fuel load reduction (if required) in rehabilitated mine areas or the biodiversity offset areas will be undertaken and will involve measures such as a combination of fire breaks and short periods of high intensity grazing. It is also conceivable that controlled burns may also be used through consultation with the NSC and the local fire brigades.	This has not been necessary to date	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Addit Finding	Consequence	Likelihood	Risk
Bushfire Management Plan	5.2.8	Fuel load / fire security will be inspected at minimum 12 monthly intervals, occurring prior to the commencement of each bushfire season.	This has not been documented	Not Compliant Administrative			
	5.3 Fire Fighting Equi	pment and Protection					
Bushfire Management Plan	5.3	Maintain fire extinguishers within all site vehicles.	DRE requirement	Compliant			
Bushfire Management Plan	5.3	Provision of fire equipment on-site will be in accordance with the requirements of the Coal Mines Health and Safety Regulation 2006.	DRE requirement	Compliant			
Bushfire Management Plan	5.3	All fire extinguishers will comply with Australian Standards and will be compatible with those of the Rural Fire Brigades.	DRE requirement	Compliant			
Bushfire Management Plan	5.3	All fire equipment will be kept in a serviceable condition and be inspected at least once every 6 months by a fire equipment service provider (3 months for mining equipment). Daily inspections will also be conducted by operators during pre-start inspections while monthly inspections will be conducted by the Fire Officer.	DRE requirement	Compliant			
Bushfire Management Plan	5.3	Fire response equipment will be maintained on site to provide	DRE requirement	Compliant			
	5.3.2 Fire Response E	immediate response to a bushfire.		<u> </u>			
	3.3.2 Fire Kesponse E	Fire response equipment will be:	Maintenance staff manage this			I	
Bushfire Management Plan	5.3.2	o available for use when the mine is in operation; o tested regularly for mobility and operation of the water pump; and o properly maintained, with operators trained in the use of all equipment.	The manufacture of the manufactu	Compliant			
Bushfire Management Plan	5.3.2	Additionally, the mine water trucks are fitted with water cannons for fire fighting purposes and are kept filled, even when parked. One water truck is also fitted with aqueous fire fighting foam.	Maintenance staff manage this	Compliant			
Bushfire Management Plan	5.3.2	All fire equipment will be compatible with that of the local Rural Fire Brigades.	Maintenance staff manage this	Compliant			
	5.3.3 Water Supply		The state of the s			T	
Bushfire Management Plan	5.3.3	 Ensure water cart is available to assist in extinguishing any fire ignited. 	This has occurred in the past.	Compliant			
Bushfire Management Plan	5.3.3	The water for fire fighting purposes will be sourced from the various water storages and licenced bores within the mine site or associated landholdings.	Noted				
	5.3.4 Fire Breaks		_				
Bushfire Management Plan	5.3.4	A suitable fire break will be established and maintained around the perimeter of the ML and MLA areas and/or associated landholdings. The fire breaks will be a minimum of 6m wide and kept free of flammable material. Additional fire breaks will be maintained around flammable materials storage areas.	Fire break maintained arount the site by slashing and grading	Compliant			
Bushfire Management Plan	5.3.4	Fire breaks will be inspected at a minimum of 6 monthly intervals.	This is not documented	Not Compliant Administrative			
	5.4 Training	Intervals.		Auministrative			
		All mine personnel will receive basic fire control training at the	This occurs, discussed in interview with				
Bushfire Management Plan	5.4	commencement of employment and regular refresher training.	safety officer	Compliant			
Bushfire Management Plan	5.4	Mine fire teams will be formed and trained in accordance with coal mining legislation.	This occurs, discussed in interview with safety officer	Compliant		-	





Deference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Bushfire Management Plan	5.4	All sub-contractors will receive basic fire training during their site induction.	This occurs, discussed in interview with safety officer	Compliant			
Bushfire Management Plan	5.4	All mine personnel, visitors and sub-contractors will be advised of the nominated muster area in the case of an emergency	This occurs, discussed in interview with safety officer	Compliant			
	5.5 External Assisstan	ce					
Bushfire Management Plan	5.5	TCPL management will liaise with the RFS and adjacent landowners regarding fire hazard minimisation at the mine and on associated landholdings.	EO has discussions with the RFS, there is an RFS person on the CCC	Compliant			
Bushfire Management Plan	5.5	In the event that a fire cannot be controlled by the mine fire fighting teams, or threatens public property, the mine emergency procedures, as defined in the mine's Emergency Management System, will be followed. These procedures include evacuation and notification of external emergency services.	Noted				
Bushfire Management Plan	5.5	Additionally TCPL will, where possible, respond to any community fire emergency situation and provide assistance by way of machinery, water, equipment and labour.	Noted				
	6 Record Keeping and						
Bushfire Management Plan	6	Records of all inspections of fire fighting equipment and fire prevention measures (eg. fire breaks) will be retained at the mine.	DRE requirement	Compliant			
Bushfire Management Plan	6	Bushfire management and performance will be reported on via the AEMR/Annual Review	The AEMR addresses this issue	Compliant			
	7 Management of Inc	idents and Complaints	I-m				
Bushfire Management Plan	7	Bushfire incidents will be recorded and managed via the Whitehaven incident management process. Occurrence of bushfires will be reported to the relevant authorities and discussed in the AEMR/Annual Review.	This occurs though ther have been none in the audit period	Not Applicable			
Bushfire Management Plan	7	Any complaints received will be managed in accordance with complaints management protocol described as follows: • A publicly advertised telephone complaints line will be in place to receive complaints during operating hours and record complaints at other times.	See Complaints elsewhere	Compliant			
Bushfire Management Plan	7	Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint, including the time of the dispersal and its duration. o The action taken by TCM in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCM, the reason why no action was taken.	See Complaints elsewhere	Compliant			
Bushfire Management Plan	7	The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational).	See Complaints elsewhere	Compliant			
Bushfire Management Plan	7	Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site.	See Complaints elsewhere	Compliant			





Deference	Classes	Requirement	Evidence	A codite Finalises		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Bushfire Management Plan	7	 Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register. 	See Complaints elsewhere	Compliant			
Bushfire Management Plan	7	• If necessary, follow-up monitoring will take place to confirm the source of the complaint is adequately mitigated.	See Complaints elsewhere	Compliant			
Bushfire Management Plan	7	A copy of the Complaints Register will be kept by TCM and made available to the Tarrawonga Coal Mine Community Consultative Committee (CCC) and the complainant (on request). Complaints are also listed on the Whitehaven website (www.whitehavencoal.com.au) and a summary of complaints received every 12 months will be provided to DoPI, NSC, GSC, EPA, DTIRIS – DRE and the CCC through the AEMR/Annual Review.		Compliant			
Bushfire Management Plan	7	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Environmental Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately.	See Complaints elsewhere	Compliant			
	8 Document Review	and Continuous Improvement					
Bushfire Management Plan	8	This document will be reviewed at least every two years and following any significant changes in bushfire management at the site.	Not yet required (BMP last revision 29-4-13)	Not Applicable			
Bushfire Management Plan	8	TCPL will investigate and implement ways to improve the environmental performance of the project over time. This will be achieved by keeping abreast of best practice bushfire management measures as well as maintaining regular contact with local brigades.	Noted				





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Keterence	Clause			Audit Finding	Consequence	Likelihood	Risk
		ste Management Plan					
	4.1 General Site Was	te Management					
		The following actions/strategies will be put into practice to	The induction addresses waste				
		minimise the accumulation/generation of waste onsite.	management				
Waste Management Plan	4.1	All personnel working on the mine site will undergo a site		Compliant			
		induction which will include details on environmental					
		management requirements;					
		All waste areas will be clearly identified as waste storage areas.	Observed on-site				
		This includes bins and other receptacles which will be marked					
Waste Management Plan	4.1	according to the type of waste accepted (eg. scrap metal, oil		Compliant			
		filters, waste oil, paper and cardboard, other recyclables, general					
		waste);	Ohannadan a'ta				
Wester Measure and Disc	4.4	Clear written instructions will be erected at appropriate locations	Observed on-site	Consultant			
Waste Management Plan	4.1	detailing recycling and waste separation information; and		Compliant			
		With the exception of mined waste rock, coal washery coarse	Observed on-site				
Waste Management Plan	4.1	reject and used tyres there will be no long term storage of any	Observed on-site	Compliant			
waste management rian	4.1	waste materials on the mine site.		Compilant			
	4.2 Waste Minimisat						
		The following methods will be used to minimise waste production:	This occurs, interview with EO				
		- The ordering of stock will be regularly reviewed to ensure					
Waste Management Plan	4.2	efficient stock control and to avoid wastage; and		Compliant			
Waste Management Plan	4.2	Alternate products or bulk storages will be considered in an effort	Generally, as observed on-site	Compliant			
waste wanagement Plan		to reduce the volume of packaging.		Compilant			
	4.3 Recycling and Re						
		TCM will provide appropriate storage areas or receptacles for all	Noted, these areas are set aside				
		materials that are suitable for recycling. The main recyclable					
		waste materials that will be generated by the mine and their					
		primary source(s), storage and collection requirements are as					
		follows:					
Waste Management Plan	4.3	Paper, cardboard, aluminium cans and plastic bottles: primarily		Compliant			
		generated within the administration facilities, crib rooms and to a					
		lesser extent in the workshop. Wheelie bins for these recyclables are located adjacent to these					
		facilities with collections occurring on a weekly basis by					
		Whitehaven's Environmental Field Technician for recycling at the					
		Gunnedah Recyclit centre.					
		Scrap metal: placed into large skip bins, which will be collected	Observed on-site				
Marta Managana Dlag	4.3	by a metal recycler as sufficient quantities are available.		Compliant			
Waste Management Plan	4.3	.,		Compliant			
		Oil filters and oily rags: generated at the maintenance workshop,	Observed on-site				
Wests Manager 2	4.2	with oil filters collected in designated, bunded 1000L pods for		Consultant			
Waste Management Plan	4.3	recycling and oily rags disposed		Compliant			
		of in general waste skip bins. Oil filters will be collected by a					
		licenced waste oil contractor during the collection of waste oil.	Observed on site				
		Waste oil and grease: waste hydrocarbons are stored in a hunded area at the maintanance workshop and are collected and	Observed on-site				
Waste Management Plan	4.3	bunded area at the maintenance workshop and are collected and disposed of by a licenced contractor approximately every 3		Compliant			
		ruisposed of by a licenced contractor approximately every 3	1		1		





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
Reference	Clause			Audit Finding	Consequence	Likelihood	Risk
Waste Management Plan	4.3	Waste material from the sump of the wash down bay is collected and analysed on an as needs basis to determine concentration thresholds for hydrocarbons. In the event that the material sampled is below concentration thresholds as specified by the EPA, the material will be disposed of in the open pit and covered with overburden material. The GPS location of disposed materials will be recorded. Where materials do not meet concentration threshold criteria for classification as general waste, they will be stored on site for bio-remediation within the designated bioremediation area within the overburden dump footprint.	Observed on-site, EO reviewes the hydrocarbon rehab site every two months	Compliant			
Waste Management Plan	4.3	Batteries: removed from site for delivery to a facility able to despatch them to an appropriate recycling facility.	Observed on-site	Compliant			
Waste Management Plan	4.3	Timber: timber products such as pallets will be reused or recycled where possible. If recycling or reuse is not feasible, waste timber products will be disposed of in general waste.	Observed on-site	Compliant			
Waste Management Plan	4.3	Miscellaneous recyclables: such as printer cartridges will be stored at appropriate locations prior to collection by, or delivery to, appropriate recycling facilities.	Observed on-site	Compliant			
Waste Management Plan	4.3	TCM's Environmental Officer will undertake regular inspections of all waste storage locations to ensure that the appropriate separation and collection of waste is being undertaken.	Monthly Inspections	Compliant			
Waste Management Plan	4.3	Opportunities for the re-use of materials on site will be evaluated on a regular basis.	Noted				
	4.4 Waste Disposal						
Waste Management Plan	4.4	General domestic wastes are collected on-site and placed into large storage receptacles. A contracted industrial waste collector removes the storage receptacles from the site on a fortnightly basis, or as required. Disposal of other wastes will be viewed as the last option in the management of waste, only if the avoidance, re-use or recycling of the waste in question is not practical. The following systems will be implemented at the mine in regard to waste disposal: • Only transport operators or companies that are licensed by the appropriate authorities will be contracted to remove waste from the mine site;	done by Contractor JR Richards	Compliant			
Waste Management Plan	4.4	Waste materials which cannot be either re-used or recycled will be sent to a landfill licenced to accept that category of waste; and	done by Contractor JR Richards	Compliant			
Waste Management Plan	4.4	Wastes, which are required to be tracked, will be done so in accordance with the relevant legislation.	done by Contractor JR Richards	Compliant			
	4.5 Effluent						
Waste Management Plan	4.5	TCM has installed adequate toilet and hand-washing facilities within the mine facilities and amenities area for the site workforce and visitors. These facilities incorporate a biocycle sewage treatment system approved by Narrabri Shire Council. The system is serviced by licensed waste collection and disposal contractors as required. Treated effluent disposal will be pumped to a small waste water utilisation area located close to the administration facilities on the southern side of the administration area.	Noted				
	4.6 Tyres						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
					Consequence	Likelihood	Risk
Waste Management Plan	4.6	Used mine equipment tyres are stockpiled on-site until they can be safely disposed of within the mined out areas of the open cut. Records on the number of tyres disposed of and the location of disposed tyres are maintained.	Observed on-site and noted in the AEMR	Compliant			
	4.7 Potentially Contar	minated Runoff					
Waste Management Plan	4.7	Potentially contaminated run-off from the workshop, wash-down bay and fuel storage areas is directed to an oil separator and containment system	Observed on-site	Compliant			
	4.8 Waste Rock						
Waste Management Plan	4.8	Waste rock removed during mining will be disposed of in the northern and southern waste emplacements and in-pit, as defined in the Tarrawonga Coal Project Environmental Assessment (2012) and Mining Operations Plan (MOP).	Observed on-site	Compliant			
	4.9 Whitehaven CHPF						
Waste Management Plan	4.8	TCPL has an approved Reject Emplacement Area designated within the open cut and approved under Section 100 of the CMHSA on the 6th January 2010. It is estimated that up to 2,000 tonnes/day of coarse reject may be transported to the Tarrawonga reject emplacement up to 5 days/week. As the emplacement area is within the pit void, the reject material will be covered by up to 20m of overburden material (and a minimum of 3m) within the in pit dump and will be subsequently rehabilitated consistent with the remainder of the in-pit dump area.	Reject emplacement is generally in accordance with this requirement.	Compliant			
	5 Monitoring, Reporti	ing and Review					
Waste Management Plan	5	As far as practical, TCM will maintain a register of recycled material and general waste at the mine site. The register will include the quantities and type of waste removed from site for recycling or disposal, the contractor engaged to remove the wastes, the date the waste or recyclables were removed from site, the final destination for all waste products and any other relevant information. All waste receipts will be retained. The register will enable TCM to review waste volumes and performance on an ongoing basis.	Contractor documentation retained by the site with quantities.	Compliant			
Waste Management Plan	5	Waste management information (including review of the effectiveness of waste minimisation and management) will be documented and reported in each AEMR/Annual Review, where applicable. Details will be provided on the implementation success of the Waste MP and any areas that require improvement.	The AEMR addresses waste management but does not address the requirement to comment on waste minimisation effectiveness.	Not Compliant Administrative			
		idents, Complaints and Non-Compliances					
	6.1 Incidents and Non				· · · · · · · · · · · · · · · · · · ·		
Waste Management Plan	6.1	Any incidents will be recorded and managed via the Whitehaven incident management process.	Noted, no incidents in the reporting period	Not Applicable			
Waste Management Plan	6.1	EPA and DoPI will be notified of all waste related incidents and non-compliances either after the incident/non compliance has occurred (for more significant issues) or via the AEMR/Annual Review and/or EPL Annual Return (for minor issues).	Noted, no incidents in the reporting period	Not Applicable			
	6.2 Complaints						





Reference	Clause	Requirement	Evidence	Audit Finding		Risk	
					Consequence	Likelihood	Risk
Waste Management Plan	6.2	Based on the proposed waste management measures, it is considered unlikely that the site will receive complaints relating to waste. Despite this, any complaints received will be managed in accordance with complaints management protocol	See Complaints Elsewhere				
		described as follows: • A publicly advertised telephone complaints line will be in place to receive complaints during operating hours and record complaints at other times.		Not Applicable			
Waste Management Plan	6.2	Each complaint received will be recorded on a Complaints Register, which will include the following details: o The date and time of complaint. o Any personal details the complainant wishes to provide or if no such details are provided a note to that effect. o The nature of the incident that led to the complaint. o The action taken by TCM in relation to the complaint, including any follow-up contact with the complainant. o If no action was taken by TCM, the reason why no action was taken.	See Complaints Elsewhere	Not Applicable			
Waste Management Plan	6.2	 The Environmental Manager will be responsible for ensuring that an initial response is provided within 24 hours of receipt of a complaint (except in the event of complaints recorded when the mine is not operational). 	See Complaints Elsewhere	Compliant			
Waste Management Plan	6.2	 Additional measures will be undertaken as required to address the complaint. This may include visiting the complainant, or inviting the complainant to the mine site. 	See Complaints Elsewhere	Compliant			
Waste Management Plan	6.2	 Once the identified measures are undertaken, the Environmental Manager will sign off on the relevant complaint within the Complaints Register. 	See Complaints Elsewhere	Compliant			
Waste Management Plan	6.2	If necessary, follow-up monitoring or will take place to confirm the source of the complaint is adequately mitigated.	See Complaints Elsewhere	Compliant			
Waste Management Plan	6.2	A copy of the Complaints Register will be kept by TCM and made available to the Tarrawonga Coal Mine Community Consultative Committee (CCC) and the complainant (on request). Complaints are also listed on the Whitehaven website (www.whitehavencoal.com.au) and a summary of complaints received every 12 months will be provided to DoPI, NSC, GSC, EPA, DRE and the CCC through the AEMR/Annual Review.		Compliant			
Waste Management Plan	6.2	Based on the nature of individual complaints, specific contingency measures may be implemented to the (reasonable) satisfaction of the complainant. The Environmental Manager retains ultimate responsibility to ensure that complaints received are properly recorded and addressed appropriately.	See Complaints Elsewhere	Compliant			
	7 Document Review	and Continuous Improvement					
Waste Management Plan	7	This document will be reviewed at least every two years and following any significant changes (ie. changes to consent/licence requirements or waste management measures). Each review will be undertaken in consultation with relevant stakeholders and will be submitted to the Director-General for approval.	Latest revision 1-5-13, not required yet.	Not Applicable			





Reference	Clause	Requirement	Evidence	Audit Finding	Risk			
				Audit Finding	Consequence	Likelihood	Risk	
Waste Management Plan	7	1	Noted					
		environmental performance of the project over time. This will be						
		achieved by keeping abreast of best practice in the industry for						
		waste management and recycling options and reporting on						
		outcomes of waste management in the AEMR/Annual Review.						